# Travaux mathématiques

**Special issue** 

dedicated to scientific contributions of the Centre for Quantum Geometry of Moduli Spaces QGM, Aarhus University, Denmark

Editor

Martin Schlichenmaier

Vol. XXV, 2017

Faculté des Sciences, de la Technologie et de la Communication



# Travaux mathématiques

Volume XXV, 2017

Published by the UNIVERSITY OF LUXEMBOURG

ISBN: 978-2-87971-168-3 ISSN: 1024-1833

UNIVERSITY OF LUXEMBOURG Mathematics Research Unit, FSTC Maison du Nombre 6, avenue de la Fonte L-4364 Esch-sur-Alzette Grand-Duchy of Luxembourg

Phone: (+352) 466 644 6236 Fax: (+352) 466 644 6237 Email: <u>martin.schlichenmaier@uni.lu</u>

#### Travaux mathématiques

#### Presentation

The journal *Travaux mathématiques* is published by the Mathematics Research Unit of the University of Luxembourg.

#### Editor-in-Chief

Martin Schlichenmaier (University of Luxembourg)

Address: University of Luxembourg Mathematics Research Unit, FSTC Maison du Nombre 6, avenue de la Fonte L-4364 Esch-sur-Alzette Grand-Duchy of Luxembourg Email: martin.schlichenmaier@uni.lu

#### Associate Editors

Jean Ludwig (University Paul Verlaine, Metz – University of Luxembourg) Martin Olbrich (University of Luxembourg) Norbert Poncin (University of Luxembourg) Anton Thalmaier (University of Luxembourg) Gabor Wiese (University of Luxembourg)

#### **Editorial Board**

Bachir Bekka (University of Rennes) Lionel Bérard-Bergery (University H. Poincaré, Nancy) Martin Bordemann (University of Haute Alsace, Mulhouse) Johannes Huebschmann (University of Lille) Pierre Lecomte (University of Liège) Jiang-Hua Lu (University of Hong Kong) Raymond Mortini (University Paul Verlaine, Metz) Claude Roger (University Claude Bernard, Lyon) Izu Vaisman (University of Haifa) Alain Valette (University of Neuchâtel) Robert Wolak (Jagiellonian University of Krakow) Ping Xu (Penn State University)

# Travaux mathématiques

Special issue

dedicated to scientific contributions of the Centre for Quantum Geometry of Moduli Spaces QGM, Aarhus University, Denmark

Editor

Martin Schlichenmaier

Vol. XXV, 2017



# Quantum Geometry of Moduli Spaces with applications to TQFT and RNA folding



A volume dedicated to Centre for Quantum Geometry of Moduli Spaces QGM at Aarhus University, Denmark

Edited by

Martin Schlichenmaier

#### Foreword

This volume of the *Travaux Mathématiques* is dedicated to the Centre for Quantum Geometry of Moduli Spaces (QGM) at Aarhus University, Denmark.

QGM was established in 2009 as a Center of Excellence funded by the Danish National Research Foundation. The research objective is to address fundamental mathematical problems at the interface between geometry and theoretical physics, in particular to further our understanding of quantum geometry of moduli spaces. One of its goals is to advance our knowledge about special kinds of quantum field theories such as Topological Quantum Field Theories and topological twisted theories. Furthermore, the mission of the centre is to explore applications of these theories and techniques in the understanding of macromolecular biology.

Based at Aarhus University and directed by Professor Jørgen Ellegaard Andersen, QGM hosts a strong team of high profile, internationally acclaimed researchers. With the continuous generation of ground-breaking results, the Centre together with its international collaborators are recognised throughout the mathematics community worldwide as one of the leading research institutions within its field of research.

The biannual International School and Conference on Geometry and Quantization (GEOQUANT) is now a well-established event in the scientific field. The main theme of the GEOQUANT conferences has strong ties to the research focus of QGM. This is further underlined by the fact that the next GEOQUANT conference will be hosted by QGM in August 2017. Travaux Mathématiques has several times before published volumes dedicated to the GEOQUANT conferences and intend to continue this tradition in connection with the upcoming GEOQUANT conference at QGM.

At the previous GEOQUANT school and conference at ICMAT in Madrid, Spain, 2015, the editor of this volume took, on the above background, the initiative to solicit articles for this volume of *Travaux Mathématiques* dedicated to the work of centre director Jørgen Ellegaard Andersen and other researchers at QGM. The intention is to give a snapshot of the very interesting spread of pure mathematics together with its application, which are being conducted currently by researchers at QGM.

Martin Schlichenmaier

#### **Collection of Summaries**

Jørgen Ellegaard Andersen and Niccolo Skovgård Poulsen Coordinates for the Universal Moduli Space of Holomorphic Vector Bundles (pp. 9–39)

In this paper we provide two ways of constructing complex coordinates on the moduli space of pairs of a Riemann surface and a stable holomorphic vector bundle centered around any such pair. We compute the transformation between the coordinates to second order at the center of the coordinates. We conclude that they agree to second order, but not to third order at the center.

#### Jørgen Ellegaard Andersen and Jens-Jakob Kratmann Nissen

Asymptotic aspects of the Teichmüller TQFT (pp. 41–95)

We calculate the knot invariant coming from the Teichmüller TQFT introduced by Andersen and Kashaev a couple of years ago. Specifically we calculate the knot invariant for the complement of the knot  $6_1$  both in Andersen and Kashaev's original version of Teichmüller TQFT but also in their new formulation of the Teichmüller TQFT for the one-vertex H-triangulation of  $(S^3, 6_1)$ . We show that the two formulations give equivalent answers. Furthermore we apply a formal stationary phase analysis and arrive at the Andersen-Kashaev volume conjecture.

Furthermore we calculate the first examples of knot complements in the new formulation showing that the new formulation is equivalent to the original one in all the special cases considered.

Finally, we provide an explicit isomorphism between the Teichmüller TQFT representation of the mapping class group of a once punctured torus and a representation of this mapping class group on the space of Schwartz class functions on the real line.

#### Jørgen Ellegaard Andersen and Simone Marzioni

Level N Teichmüller TQFT and Complex Chern–Simons Theory (pp. 97–146)

In this manuscript we review the construction of the Teichmüller TQFT due to Andersen and Kashaev. We further upgrading it to a theory dependent on an extra odd integer N using results developed by Andersen and Kashaev in their work on complex quantum Chern-Simons theory. We also describe how this theory is related with quantum Chern-Simons Theory at level N with gauge group PSL(2,  $\mathbb{C}$ ).

#### Jørgen Ellegaard Andersen and William Elbæk Petersen

Construction of Modular Functors from Modular Categories (pp. 147–211)

In this paper we follow the constructions of Turaev's book, "Quantum invariants of knots and 3-manifolds" closely, but with small modifications, to construct a modular functor, in the sense of Kevin Walker, from any modular tensor category. We further show that this modular functor has duality and if the modular tensor category is unitary, then the resulting modular functor is also unitary. We further introduce the notion of a fundamental symplectic character for a modular tensor category. In the cases where such a character exists we show that compatibilities between the structures in a modular functor can be made strict in a certain sense. Finally we establish that the modular tensor categories which arise from quantum groups of simple Lie algebras all have natural fundamental symplectic characters.

### Jørgen Ellegaard Andersen, Hiroyuki Fuji, Robert C. Penner, and Christian M. Reidys

The boundary length and point spectrum enumeration of partial chord diagrams using cut and join recursion (pp. 213–232)

We introduce the boundary length and point spectrum, as a joint generalization of the boundary length spectrum and boundary point spectrum introduced by Alexeev, Andersen, Penner and Zograf. We establish by cut-and-join methods that the number of partial chord diagrams filtered by the boundary length and point spectrum satisfies a recursion relation, which combined with an initial condition determines these numbers uniquely. This recursion relation is equivalent to a second order, non-linear, algebraic partial differential equation for the generating function of the numbers of partial chord diagrams filtered by the boundary length and point spectrum.

#### Jørgen Ellegaard Andersen, Hiroyuki Fuji, Masahide Manabe, Robert C. Penner, and Piotr Sułkowski

Partial Chord Diagrams and Matrix Models

(pp. 233-283)

In this article, the enumeration of partial chord diagrams is discussed via matrix model techniques. In addition to the basic data such as the number of backbones and chords, we also consider the Euler characteristic, the backbone spectrum, the boundary point spectrum, and the boundary length spectrum. Furthermore, we consider the boundary length and point spectrum that unifies the last two types of spectra. We introduce matrix models that encode generating functions of partial chord diagrams filtered by each of these spectra. Using these matrix models, we derive partial differential equations – obtained independently by cut-and-join arguments in an earlier work – for the corresponding generating functions.

## Jørgen Ellegaard Andersen, Hiroyuki Fuji, Masahide Manabe, Robert C. Penner, and Piotr Sułkowski

Enumeration of Chord Diagrams via Topological Recursion and Quantum Curve Techniques (pp. 285–323)

In this paper we consider the enumeration of orientable and non-orientable chord diagrams. We show that this enumeration is encoded in appropriate expectation values of the  $\beta$ -deformed Gaussian and RNA matrix models. We evaluate these expectation values by means of the  $\beta$ -deformed topological recursion, and – independently – using properties of quantum curves. We show that both these methods provide efficient and systematic algorithms for counting of chord diagrams with a given genus, number of backbones and number of chords. Travaux mathématiques, Volume 25 (2017), 9–39, © by the authors

### Coordinates for the Universal Moduli Space of Holomorphic Vector Bundles

by Jørgen Ellegaard Andersen and Niccolo Skovgård Poulsen<sup>1</sup>

#### Abstract

In this paper we provide two ways of constructing complex coordinates on the moduli space of pairs of a Riemann surface and a stable holomorphic vector bundle centred around any such pair. We compute the transformation between the coordinates to second order at the center of the coordinates. We conclude that they agree to second order, but not to third order at the center.

#### 1 Introduction

Fix g, n > 1 to be integers and let  $d \in \{0, \ldots n - 1\}$ . Let  $\Sigma$  be a closed oriented surface of genus g. Consider the universal moduli space  $\mathcal{M}$  consisting of equivalence classes of pairs  $(\phi : \Sigma \to X, E)$  where X is a Riemann Surface of genus g,  $\phi : \Sigma \to X$  is a diffeomorphism and E is a semi-stable bundle over X of rank nand degree d. Let  $\mathcal{M}^s$  be the open dense subset of  $\mathcal{M}$  consisting of equivalence classes of such pairs  $(\phi : \Sigma \to X, E)$  with E stable. The main objective of this paper is to provide coordinates in a neighbourhood of the equivalence class of any pair  $(\phi : \Sigma \to X, E)$  in  $\mathcal{M}^s$ . There is an obvious forgetful map

$$\pi_{\mathcal{T}}: \mathcal{M} \to \mathcal{T}$$

where  $\mathcal{T}$  is the Teichmüller space of  $\Sigma$ , whose fiber over  $[\phi : \Sigma \to X] \in \mathcal{T}$  is the moduli space of semi-stable bundles for that Riemann surface structure on  $\Sigma$ . Let  $\pi^s_{\mathcal{T}} : \mathcal{M}^s \to \mathcal{T}$  denote the restriction of  $\pi_{\mathcal{T}}$  to  $\mathcal{M}^s$ , and we denote a point  $[\phi : \Sigma \to X]$  in  $\mathcal{T}$  by  $\sigma$ .

We recall that locally around any  $\sigma \in \mathcal{T}$  there are the Bers coordinates [AhB]. Further, for any point [E] in some fiber  $(\pi_{\mathcal{T}}^s)^{-1}(\sigma)$  we have the Zograf and Takhtadzhyan coordinates near [E] along that fiber of  $\pi_{\mathcal{T}}$  [TZ1].

In order to describe our coordinates on  $\mathcal{M}^s$  we recall the Narasimhan-Seshadri theorem. Let  $\tilde{\pi}_1(\Sigma)$  be the universal central  $\mathbb{Z}/n\mathbb{Z}$  extension of  $\pi_1(\Sigma)$  and let M

<sup>&</sup>lt;sup>1</sup>Work supported in part by the center of excellence grant "Center for Quantum Geometry of Moduli Spaces" from the Danish National Research Foundation (DNRF95).

be the moduli space of representations of  $\tilde{\pi}_1(\Sigma)$  to  $\mathbf{U}(n)$  such that the central generator goes to  $e^{2\pi i d/n}$ Id. Let M' be the subset of M consisting of equivalence classes of irreducible representations. The Narasimhan-Seshadri theorem gives us a diffeomorphism

$$\Psi: \mathcal{T} \times M' \to \mathcal{M}^s$$

which we use to induce a complex structure on  $\mathcal{T} \times M'$  such that  $\Psi$  is complex analytic. We will now represent a point in  $\mathcal{T}$  by a representation

$$\rho_0 : \tilde{\pi}_1(\Sigma) \rightarrow \mathbf{PSL}(2)$$

and denote the corresponding point in Teichmüller space by  $X_{\rho_0}$ . Here  $\rho_0$  is really a representation of  $\pi_1(\Sigma)$  pulled back to  $\tilde{\pi}_1(\Sigma)$ . A point in M' will be represented by a representation

$$\rho_E : \tilde{\pi}_1(\Sigma) \to \mathbf{U}(n)$$

which corresponds to the stable holomorphic bundle E on  $X_{\rho_0}$ .

We build complex analytic coordinates around any such  $(\rho_0, \rho_E) \in \mathcal{T} \times M'$  by providing a complex analytic isomorphism from a small neighbourhood around 0 in the vector space  $H^{0,1}(X_{\rho_0}, TX_{\rho_0}) \oplus H^{0,1}(X_{\rho_0}, \text{End}E)$  to a small open subset containing  $(\rho_0, \rho_E)$  in  $\mathcal{T} \times M'$ .

The coordinates are given by constructing a certain family

(1.1) 
$$\Phi^{\mu \oplus \nu} : \mathbf{H} \times \mathbf{GL}(n, \mathbb{C}) \to \mathbf{H} \times \mathbf{GL}(n, \mathbb{C})$$

of bundle maps of the trivial  $\mathbf{GL}(n,\mathbb{C})\text{-principal bundles over }\mathbf{H}$  indexed by pairs of sufficiently small elements

$$\mu \oplus \nu \in H^{0,1}(X_{\rho_0}, TX_{\rho_0}) \oplus H^{0,1}(X_{\rho_0}, \operatorname{End} E).$$

These bundle maps will uniquely determine representations  $(\rho^{\mu}, \rho_E^{\mu\oplus\nu}) \in \mathcal{T} \times M'$  such that

(1.2) 
$$\rho^{\mu}(\gamma) \times \rho_{E}^{\mu \oplus \nu}(\gamma) = \Phi^{\mu \oplus \nu} \circ (\rho_{0}(\gamma) \times \rho_{E}(\gamma)) \circ (\Phi^{\mu \oplus \nu})^{-1}$$

for all  $\gamma \in \tilde{\pi}_1(X)$  by the following theorem. Pick a base point  $z_0 \in \mathbf{H}$  and let  $p_{\mathbf{GL}(n,\mathbb{C})}$  be the projection onto  $\mathbf{GL}(n,\mathbb{C})$  of the trivial bundle  $\mathbf{H} \times \mathbf{GL}(n,\mathbb{C})$ .

**Theorem 1.1.** For all sufficiently small  $\mu \oplus \nu \in H^{0,1}(X_{\rho_0}, TX_{\rho_0}) \oplus H^{0,1}(X_{\rho_0}, \text{End}E)$ there exist a unique bundle map  $\Phi^{\mu \oplus \nu}$  such that

1.  $\Phi^{\mu\oplus\nu}$  solves

(1.3) 
$$\bar{\partial}_{\mathbf{H}} \Phi^{\mu \oplus \nu} = \partial \Phi^{\mu \oplus \nu} (\mu \oplus \nu)$$

where  $\nu$  is considered a left-invariant vector field on  $\mathbf{GL}(n, \mathbb{C})$  at each point in  $\mathbf{H}$ .

Coordinates for the Universal Moduli Space of Holomorphic Vector Bundles 11

- 2. The base map extends to the boundary of **H** and fixes 0, 1 and  $\infty$ .
- The pair of representations (ρ<sup>μ</sup>, ρ<sup>μ⊕ν</sup><sub>E</sub>) defined by equation (1.2) represents a point in T × M'.
- 4.  $p_{\mathbf{GL}(n,\mathbb{C})}(\Phi^{\mu\oplus\nu}(z_0,e))$  has determinant 1 and is positive definite.

From this theorem we easily derive our main theorem of this paper.

**Theorem 1.2.** Mapping all sufficiently small pairs

$$\mu \oplus \nu \in H^{0,1}(X_{\rho}, TX_{\rho}) \oplus H^{0,1}(X_{\rho}, EndE)$$

to

$$(\rho^{\mu}, \rho_E^{\mu \oplus \nu}) \in \mathcal{T} \times M'$$

provides local analytic coordinates centered at  $(\rho_0, \rho_E) \in \mathcal{T} \times M'$ .

Our second coordinate construction provides fibered coordinates, which along  $\mathcal{T}$  uses Bers' coordinates, [AhB], and which uses Zograf and Takhtadzhyan's coordinates [TZ1] along the fibers. We refer to section 4 for the precise description of these fibered coordinates.

Finally, we compare the two sets of coordinates by computing the infinitesimal transformation of the coordinates up to second order at the center of both coordinates.

**Theorem 1.3.** The fibered coordinates and the universal coordinates agree to second order, but not the third order at the center of the coordinates.

We refer to Theorem 5.5, for the details of how the two set of coordinates differ at third-order.

With these new coordinates we get a new tool to analyse the metric and the curvature of the moduli space. Here we have taken the first step in understanding the curvature by calculating the second variation of the metric in local coordinates, at the center point. We intend to return to the full calculation of the curvature in these coordinates in a future publication.

**Remark 1.4.** If we perform our construction using elements of  $\mathcal{H}^{0,1}(X, (\operatorname{End}_0 E))$  where  $(\operatorname{End}_0 E)$  is the subspace of traceless endomorphisms, we get coordinates on the universal  $\mathbf{SU}(n)$  moduli space in a completely similar way.

The coordinate construction presented in this paper has been used and modified by the authors of this paper in [AP1], so as to give Kähler coordinates on the universal moduli space. In these modified coordinates, it was possible to verify an explicit expression for the Ricci potential for universal moduli space. Further in the paper [AP2], the result of this paper and that of [AP1] was used to compute explicitly the curvature of the Hitchin connection. This is of current great interest, since the Hitchin connection is know to be equivalent to the TUY connection [TUY] by the work of Laszlo [L] and by the work of Andersen and Ueno [AU1, AU2, AU3, AU4] the modular functor which underlies the conformal field theories studied by Tsuchiya, Ueno and Yamada [TUY] is equivalent to the Witten-Reshetikhin-Turaev TQFT [T, RT1, RT2]. This connection has been exploited significantly by Andersen and collaborators in a series of papers to obtain deep results about the WRT-TQFT [A1, A2, AGr1, AH, AMU, A3, A4, A5, AGa1, AB1, A6, AGL, AHi, A7].

#### Acknowledgements

We would like to thank Peter Zograf for many interesting discussions.

#### 2 The Complex Structure on $\mathcal{M}^s$ from a Differential Geometric Perspective

Recall that we endow the space  $\mathcal{T} \times M$  with the structure of a complex manifold by using the Narasimhan-Seshadri theorem to provide us with the diffeomorphism

$$\Psi: \mathcal{T} \times M' \to \mathcal{M}^s$$

and then declaring it to be complex analytic. There is the following alternative construction of this complex manifold structure.

Recall the general setting of [AGL] in the context of geometric quantization and the Hitchin connection, namely  $\tilde{\mathcal{T}}$  is a general complex manifold and  $(\tilde{M}, \omega)$  is a general symplectic manifold. In that paper a construction of a complex structure on  $\tilde{\mathcal{T}} \times \tilde{M}$  is provided via the following proposition. But first we need the following definition.

**Definition 2.1.** A family of Kähler structures on  $(\tilde{M}, \omega)$  parametrized by  $\tilde{\mathcal{T}}$  is called holomorphic if it satisfies:

$$V'[J] = V[J]' \quad V''[J] = V[J]''$$

for all vector fields V on  $\tilde{\mathcal{T}}$ . Here the single prime on V denotes projection on the (1, 0)-part and the double prime on V denotes projection on the (0, 1)-part of the vector field V. Further  $V[J] \in T_{\sigma} \otimes (\bar{T}_{\sigma})^* \oplus \bar{T}_{\sigma} \otimes T_{\sigma}^*$  and we let V[J]' denote the projection on the first, and V[J]'' the projection on the second factor.

**Proposition 2.2** ([AGL, Proposition 6.2]). The family  $J_{\sigma}$  of Kähler structures on  $\tilde{M}$  is holomorphic, if and only if the almost complex structure J given by

$$J(V \oplus X) = IV \oplus J_{\sigma}X, \quad \forall (\sigma, [\rho_E]) \in \tilde{\mathcal{T}} \times \tilde{M}, \forall (V, X) \in T_{\sigma, [\rho_E]}(\tilde{\mathcal{T}} \times \tilde{M}),$$

is integrable.

The family of complex structures on M' considered in [H] see also [AGL], [A6] and [AGa1], given by the Hodge star,  $-\star_{\sigma}$ ,  $\sigma \in \mathcal{T}$ , fulfills the requirements of the proposition with respect to the Atiyah-Bott symplectic form  $\omega$  on M'. We will denote the complex structure which  $\mathcal{T} \times M'$  has by J.

**Proposition 2.3.** We have that the map

$$\Psi : (\mathcal{T} \times M', J) \to \mathcal{M}^s$$

is complex analytic, e.g. J is in fact the complex analytic structure this space gets from the Narasimhan-Seshadri diffeomorphism  $\Psi$ .

*Proof.* In order to understand the complex structure of  $\mathcal{T} \times M'$  from the algebraic geometric perspective we want to construct holomorphic horizontal sections of  $\mathcal{T} \times M' \to \mathcal{T}$ . We will use the universal property of the space of holomorphic bundles to show that the sections  $\mathcal{T} \to \mathcal{T} \times \{\rho_E\} \subset \mathcal{T} \times M'$  are holomorphic for all  $[\rho_E]$  in M'.

Our first objective is to construct a holomorphic family of vector bundles over Teichmüller space, where each bundle corresponds to the same unitary representation of  $\tilde{\pi}_1(\Sigma)$ . We start from the universal curve  $\mathcal{T} \times \Sigma$  and its universal cover  $\mathcal{T} \times \tilde{\Sigma}$ . Both of these spaces are complex analytic, and we get the universal curve  $\mathcal{T} \times \Sigma$  as the quotient of  $\mathcal{T} \times \tilde{\Sigma}$  by the holomorphic  $\pi_1(\Sigma)$  action.

This allows us to construct the vector bundles over  ${\mathcal T}$  as the sheaf theoretic quotient of

 $\mathcal{T} \times \tilde{\Sigma} \times \mathbb{C}^n$ 

by the  $\tilde{\pi}_1(\Sigma)$  action, given by the  $\pi_1(\Sigma)$ -action on  $\mathcal{T} \times \tilde{\Sigma}$ , and the unitary action on  $\mathbb{C}^n$  given by our fixed representation  $\rho_E : \tilde{\pi}_1(\Sigma) \to \mathbf{U}(n)$  (see [MS] for details on this construction, where we simply just compose representations with the natural quotient map from  $\pi_1(\Sigma - \{p\})$  to  $\tilde{\pi}_1(\Sigma)$  to match up the setting of this paper to a special case of the setting in [MS]). The action is of course holomorphic, and so the quotient (fiberwise invariant sections over  $\mathcal{T}$ ) is a family of Riemann surfaces with a holomorphic vector bundle over it of rank n and degree d. The universal property of  $\mathcal{M}^s$  implies that this family therefore induced a holomorphic section

$$\iota_{\rho_E}: \mathcal{T} \to \mathcal{M}^s$$

by the universality of the moduli space  $\mathcal{M}^s$ . This shows that the horizontal sections are holomorphic submanifolds, and so the tangent space must split at every point as  $I \oplus J_{\sigma}$ . Here  $J_{\sigma}$  must be  $-\star_{\sigma}$  since it comes from the structure of the fibers.

The conclusion is, that the algebraic complex structure on the moduli space of pairs of a Riemann surface and a holomorphic vector bundle over it and the complex structure from [AGL] on  $\mathcal{T} \times M'$  are the same.

#### 3 Coordinates for the Universal Moduli Space of Holomorphic Vector Bundles

In this section we prove Theorem 1.1.

We will need the composition of the map  $\Phi^{\mu\oplus\nu}$  with the projection on each of the two factors, which we denote as follows:

$$\Phi_1^{\mu\oplus\nu}: \mathbf{H} \times \mathbf{GL}(n, \mathbb{C}) \to \mathbf{H},$$
$$\Phi_2^{\mu\oplus\nu}: \mathbf{H} \times \mathbf{GL}(n, \mathbb{C}) \to \mathbf{GL}(n, \mathbb{C}).$$

In fact  $\Phi_1^{\mu\oplus\nu}$  is the projection onto **H** followed by the induced map on the base by (3.3) below.

The equation (1.3) is equivalent to the following two equations on  $\Phi_i^{\mu\oplus\nu}$ :

(3.1) 
$$\bar{\partial}_{\mathbf{H}} \Phi_1^{\mu \oplus \nu}(z,g) = \mu \partial_{\mathbf{H}} \Phi_1^{\mu \oplus \nu}(z,g)$$

(3.2) 
$$\bar{\partial}_{\mathbf{H}} \Phi_2^{\mu \oplus \nu}(z,g) = \mu \partial_{\mathbf{H}} \Phi_2^{\mu \oplus \nu}(z,g) + \partial_{\mathbf{GL}(n,\mathbb{C})} \Phi_2^{\mu \oplus \nu}(z,g) \nu$$

since  $\partial_{\mathbf{GL}(n,\mathbb{C})} \Phi_1^{\mu \oplus \nu}(z,g) = 0$ . With this simplification the first equation is exactly Bers's equation for

(3.3) 
$$\Phi_1^{\mu}(z) = \Phi_1^{\mu \oplus \nu}(z, g),$$

and so we can solve it using the techniques in [AhB], and we obtain a Riemann surface  $X_{\rho_{\mu}}$  corresponding to a representation  $\rho_{\mu}$ .

The second equation (3.2) we solve in two steps. First, we identify  $\nu$  with an endomorphism valued 1-form using the standard identification of left invariant vector fields and elements of the Lie algebra. To solve the equation we consider the anti-holomorphic solution of the equation

$$\bar{\partial}_{\mathbf{H}} \Phi^{\nu}_{-}(z,e) = \partial_{\mathbf{GL}(n,\mathbb{C})} \Phi^{\nu}_{-}(z,g)(\nu)|_{g=e} = \Phi^{\nu}_{-}(z,e) \cdot \nu$$

and extend it equivariantly to the rest of  $\mathbf{H} \times \mathbf{GL}(n, \mathbb{C})$ . We observe that  $\partial_{\mathbf{H}} \Phi_{-}^{\nu} = 0$  since it is anti-holomorphic. And so it follows, by adding 0 to the defining equation of  $\Phi_{-}^{\nu}$  that:

$$\bar{\partial}_{\mathbf{H}}\Phi^{\nu}_{-}(z,g) = \partial_{\mathbf{GL}(n,\mathbb{C})}\Phi^{\nu}_{-}(z,g)(\nu) + \mu\partial_{\mathbf{H}}\Phi^{\nu}_{-}(z,g).$$

The vector bundle on  $X_{\rho_{\mu}}$  corresponding to the representation

$$\chi^{\nu}(\gamma) = \Phi^{\nu}_{-}(\rho_{0}(\gamma)z, e)\rho^{0\oplus 0}_{E}(\gamma)(\Phi^{\nu}_{-}(z, e))^{-1}$$

is stable, if  $\mu \oplus \nu$  is small enough. This means, we can find a holomorphic gauge transformation on the universal cover of  $X_{\rho\mu}$ ,  $\Phi^{\mu\oplus\nu}_+$ :  $\mathbf{H} \to \mathbf{GL}(n, \mathbb{C})$ , such that

(3.4) 
$$\rho_E^{\mu\oplus\nu}(\gamma) = \Phi_+^{\mu\oplus\nu}(\rho_\mu(\gamma)z)\chi^{\mu\oplus\nu}(\gamma)(\Phi_+^{\mu\oplus\nu}(z))^{-1}$$

is an admissible  $\mathbf{U}(n)$ -representation and independent of z by the Narasimhan-Seshadri theorem [NSNS]. Now we use the basemap to define  $\tilde{\Phi}_{+}^{\mu\oplus\nu} = \Phi_{+}^{\mu\oplus\nu} \circ \Phi_{1}^{\mu}$ . The following computation shows that the map  $\tilde{\Phi}_{+}^{\mu\oplus\nu}$  is in the kernel of  $\bar{\partial}_{\mathbf{H}} -$ 

The following computation shows that the map  $\Phi_{+}^{\mu\sigma\nu}$  is in the kernel of  $\partial_{\mathbf{H}} - \mu \partial_{\mathbf{H}}$ :

$$\begin{split} (\bar{\partial}_{\mathbf{H}} - \mu \partial_{\mathbf{H}}) \tilde{\Phi}_{+}^{\mu \oplus \nu} &= (\bar{\partial}_{\mathbf{H}} \Phi_{+}^{\mu \oplus \nu}) \circ \Phi_{1}^{\mu} \bar{\partial}_{\mathbf{H}} \bar{\Phi}_{1}^{\mu} + (\partial_{\mathbf{H}} \Phi_{+}^{\mu \oplus \nu}) \circ \Phi_{1}^{\mu} \bar{\partial}_{\mathbf{H}} \Phi_{1}^{\mu} \\ &- \mu (\bar{\partial}_{\mathbf{H}} \Phi_{+}^{\mu \oplus \nu}) \circ \Phi_{1}^{\mu} \partial_{\mathbf{H}} \bar{\Phi}_{1}^{\mu} - \mu (\partial_{\mathbf{H}} \Phi_{+}^{\mu \oplus \nu}) \circ \Phi_{1}^{\mu} \partial_{\mathbf{H}} \Phi_{1}^{\mu} \end{split}$$

We then use the differential equation  $\bar{\partial}\Phi_1^{\mu} = \mu\partial\Phi_1^{\mu}$  and that  $\bar{\partial}_{\mathbf{H}}\Phi_+^{\mu\oplus\nu} = 0$  to get that

$$(\bar{\partial}_{\mathbf{H}} - \mu \partial_{\mathbf{H}}) \tilde{\Phi}_{+}^{\mu \oplus \nu} = \partial_{\mathbf{H}} \Phi_{+}^{\mu \oplus \nu} \circ \Phi_{1}^{\mu} \mu \partial_{\mathbf{H}} \Phi_{1}^{\mu} - \mu \partial_{\mathbf{H}} \Phi_{+}^{\mu \oplus \nu} \circ \Phi_{1}^{\mu} \partial_{\mathbf{H}} \Phi_{1}^{\mu} = 0$$

Define  $\Phi_2^{\mu\oplus\nu}(z,g) = \tilde{\Phi}_+^{\mu\oplus\nu}(z,g)\Phi_-^{\nu}(z,g)$ . We see that  $\Phi_2^{\mu\oplus\nu}$  fulfills equation (3.2) by the following calculation

$$\begin{split} \bar{\partial}_{\mathbf{H}} \Phi_2^{\mu \oplus \nu} &= (\bar{\partial}_{\mathbf{H}} \tilde{\Phi}_+^{\mu \oplus \nu}) (\Phi_-^{\nu}) + (\tilde{\Phi}_+^{\mu \oplus \nu}) (\bar{\partial}_{\mathbf{H}} \Phi_-^{\nu}) \\ &= (\bar{\partial}_{\mathbf{H}} \tilde{\Phi}_+^{\mu \oplus \nu}) (\Phi_-^{\nu}) + (\tilde{\Phi}_+^{\mu \oplus \nu}) (\partial_{\mathbf{GL}(n,\mathbb{C})} \Phi_-^{\nu} \nu) \end{split}$$

since  $\tilde{\Phi}^{\mu\oplus\nu}_+ \in \ker(\bar{\partial}_{\mathbf{H}} - \mu\partial_{\mathbf{H}})$  we get that

$$\bar{\partial}_{\mathbf{H}} \Phi_2^{\mu \oplus \nu} = (\mu \partial_{\mathbf{H}} \tilde{\Phi}_+^{\mu \oplus \nu}) (\Phi_-^{\nu}) + (\tilde{\Phi}_+^{\mu \oplus \nu}) (\partial_{\mathbf{GL}(n,\mathbb{C})} \Phi_-^{\nu} \nu).$$

To finish the calculation we use that  $\Phi_+$  and  $\Phi_1$  are independent of the  $\mathbf{GL}(n, \mathbb{C})$  factor, and therefore so is  $\tilde{\Phi}_+^{\mu\oplus\nu}$ . Also  $\Phi_-^{\mu\oplus\nu}$  is antiholomorphic so we have that

$$\begin{split} \bar{\partial}_{\mathbf{H}} \Phi_{2}^{\mu \oplus \nu} &= \mu \partial_{\mathbf{H}} (\tilde{\Phi}_{+}^{\mu \oplus \nu} \Phi_{-}^{\nu}) + \partial_{\mathbf{GL}(n,\mathbb{C})} (\tilde{\Phi}_{+}^{\mu \oplus \nu} \Phi_{-}^{\nu}) \nu \\ &= \mu \partial_{\mathbf{H}} \Phi_{2}^{\mu \oplus \nu} + (\partial_{\mathbf{GL}(n,\mathbb{C})} \Phi_{2}^{\mu \oplus \nu}) \nu \end{split}$$

To show that we still get an admissible representation, we use that (3.4) is independent of which z we choose. This lets us conclude that

$$\begin{split} \rho_E^{\mu\oplus\nu}(\gamma) &= \Phi_+^{\mu\oplus\nu}(\rho_\mu(\gamma)\Phi_1^{\mu}(z))\chi^{\mu\oplus\nu}(\gamma)(\Phi_+^{\mu\oplus\nu}(\Phi_1^{\mu}(z)))^{-1} \\ &= \Phi_+^{\mu\oplus\nu}(\Phi_1^{\mu}(\rho_0(\gamma)(\Phi_1^{\mu})^{-1}(\Phi_1^{\mu}(z)))\chi^{\mu\oplus\nu}(\gamma)(\Phi_+^{\mu\oplus\nu}(\Phi_1^{\mu}(z)))^{-1} \\ &= \tilde{\Phi}_+^{\mu\oplus\nu}(\rho_0(\gamma)z)\chi^{\mu\oplus\nu}(\gamma)(\tilde{\Phi}_+^{\mu\oplus\nu}(z))^{-1}, \end{split}$$

and so

$$\rho_E^{\mu\oplus\nu}(\gamma) = \Phi_2^{\mu\oplus\nu}(\rho_0(\gamma)z, g)\rho_E^{0\oplus0}(\gamma)(\Phi_2^{\mu\oplus\nu}(z, g))^{-1}$$

is an admissible  $\mathbf{U}(n)$ -representation. Finally, the requirement that  $\Phi_2^{\mu\oplus\nu}(z_0, e)$  is a positive definite matrix of determinant 1 fixes all remaining indeterminacy as in [TZ1].

#### 3.1 The Tangent Map from Kodaira-Spencer Theory

We will now analyse the tangential map of our coordinates. The only problematic part is what happens in the tangent directions parallel to the fibers. We can calculate the Kodaira-Spencer map of the family of representations  $\rho_E^{\mu\oplus\nu+t\bar{\mu}\oplus\bar{\nu}}$ ,  $t \in \mathbb{C}$ . However, to ease the computation we first prove the following lemma.

**Lemma 3.1.** We let  $X_{\rho_0}$  be a Riemann surface and  $\rho_0$  the corresponding representation of  $\pi_1(X_{\rho_0})$ . For a family of representations of  $R_t : \tilde{\pi}_1(X_{\rho_0}) \to \mathbf{U}(n)$ , where

$$R_t(\gamma) = \Upsilon(t, \rho_0(\gamma)z)\rho_E(\gamma)\Upsilon(t, z)^{-1}$$

with both  $\rho_0$  and  $\rho_E$  independent of t and  $\Upsilon$  any smooth map

$$\Upsilon: \mathbb{C} \times \mathbf{H} \to \mathbf{GL}(n, \mathbb{C}),$$

we have that the Kodaira-Spencer class's harmonic representative of the family  $R_t$  at t = 0 is:

$$P^{0,1}_{\rho_0,E}\left(Ad\Upsilon(0,z)\left(\left.\frac{d}{dt}\right|_{t=0}\Upsilon(t,z)^{-1}\bar{\partial}_{\mathbf{H}}\Upsilon(t,z)\right)\right)\in H^{0,1}(X_0,EndE_{R_0}).$$

Here  $P^{0,1}_{\rho_0,E}$  denotes the projection on the harmonic forms on  $X_{\rho_0}$  with values in  $\operatorname{End} E_{R_0}$ .

*Proof.* To compute the Kodaira-Spencer map we first consider  $\frac{d}{dt}\Big|_{t=0} R_t$  and note, this is an element of  $H^1(X, \operatorname{End}(E))$ . However, this cohomology group is isomorphic to  $H^{0,1}(X, \operatorname{End} E)$ . The isomorphism is constructed by finding a Čech chain with values in the sheaf  $\Omega^1(\operatorname{End} E)$ , say  $\varphi_i$ , such that

$$\delta^*(\phi)_{ij} = \phi_i - \phi_j = \left. \frac{d}{dt} \right|_{t=0} R_t(\gamma_{ij}),$$

for open sets  $U_i \cap U_j \neq \emptyset$  which are related by the transformation  $\gamma_{ij} \in \tilde{\pi}_1(\Sigma)$ on the universal cover. Once  $\phi_i$  has been found,  $P^{0,1}_{\rho_0,E}(\bar{\partial}_{\mathbf{H}}\phi_i)$  will give a harmonic representative of the Kodaira-Spencer class.

We can now calculate that

$$\begin{aligned} \frac{d}{dt}\Big|_{t=0} R_t(\gamma_{ij}) &= \left. \frac{d}{dt} \right|_{t=0} \Upsilon(t,\rho_0(\gamma)z)\rho_E(\gamma)\Upsilon(t,z)^{-1} \\ &= \left. \frac{d}{dt} \right|_{t=0} \Upsilon(t,\rho_0(\gamma)z)\rho_E(\gamma)\Upsilon(0,z)^{-1} \\ &+ \Upsilon(0,\rho_0(\gamma)z)\rho_E(\gamma) \left. \frac{d}{dt} \right|_{t=0} \Upsilon(t,z)^{-1} \\ &= \left. \frac{d}{dt} \right|_{t=0} (\Upsilon(t,\rho_{\mathbf{H}}(\gamma_{ij})z)\Upsilon(0,\rho_0(\gamma_{ij})z)^{-1})R_0(\gamma_{ij}) \end{aligned}$$

$$-R_0(\gamma_{ij}) \frac{d}{dt}\Big|_{t=0} (\Upsilon(t,z)\Upsilon(0,z)^{-1})$$
$$=R_0(\gamma_{ij})\delta(\frac{d}{dt}\Big|_{t=0} (\Upsilon(t,z)\Upsilon(0,z)^{-1}))_{ij}$$

The Kodaira-Spencer class is then:

$$\bar{\partial}_{\mathbf{H}} \left. \frac{d}{dt} \right|_{t=0} \left( \Upsilon(t,z) \Upsilon(0,z)^{-1} \right) = \mathrm{Ad}\Upsilon(0,z) \left( \left. \frac{d}{dt} \right|_{t=0} \Upsilon(t,z)^{-1} \bar{\partial}_{\mathbf{H}} \Upsilon(t,z) \right).$$

We compose with the harmonic projection to get the harmonic representative.  $\hfill\square$ 

We have the following proposition.

**Proposition 3.2.** The Kodaira-Spencer map of  $\rho_E^{\mu\oplus\nu+t\tilde{\mu}\oplus\tilde{\nu}}, t \in \mathbb{C}$  at  $\mu \oplus \nu \in H^{0,1}(X,TX) \oplus H^{0,1}(X,\operatorname{End} E)$ ,

$$KS_{\mu\oplus\nu}: H^{0,1}(X,TX)\oplus H^{0,1}(X,\operatorname{End} E) \to H^{0,1}(X_{\rho_{\mu}},TX_{\rho_{\mu}})\oplus H^{0,1}(X_{\rho_{\mu}},\operatorname{End} E_{\rho_{E}^{\mu\oplus\nu}})$$

is given by

$$KS_{\mu\oplus\nu}(\tilde{\mu}\oplus\tilde{\nu}) = P_{\mu}\tilde{\mu}^{\mu}\oplus P_{\mu\oplus\nu}^{0,1}\left((\Phi_{1}^{\mu})_{*}^{-1}\left(Ad\Phi_{2}^{\mu\oplus\nu}\left(\tilde{\mu}(\Phi_{2}^{\mu\oplus\nu})^{-1}\partial\Phi_{2}^{\mu\oplus\nu}+\tilde{\nu}\right)\right)\right).$$

Here  $\tilde{\mu}^{\mu} = \left(\frac{\tilde{\mu}}{1-|\mu|^2}\frac{\partial \Phi_1^{\mu}}{\partial \Phi_1^{\mu}}\right) \circ \left(\Phi_1^{\mu}\right)^{-1}$  and  $P_{\mu}$  and  $P_{\mu \oplus \nu}^{0,1}$  are the L<sup>2</sup>-projections on the harmonic forms  $H^{0,1}(X_{\rho_{\mu}}, TX_{\rho_{\mu}})$  respectively  $H^{0,1}(X_{\rho_{\mu}}, \text{End}E_{\rho_E^{\mu \oplus \nu}})$ .

*Proof.* By using that the defining equation (1.2) for  $\rho_E^{\mu\oplus\nu+t\tilde{\mu}\oplus\tilde{\nu}}$  is independent of z, we get that

$$\begin{split} \rho_E^{\mu \oplus \nu + t\bar{\mu} \oplus \bar{\nu}} &= \Phi_2^{\mu \oplus \nu + t\bar{\mu} \oplus \bar{\nu}}(\rho_0(\gamma)z), e) \rho_E^{0 \oplus 0}(\gamma) \Phi_2^{\mu \oplus \nu}(z, e)^{-1} \\ &= \Phi_2^{\mu \oplus \nu + t\bar{\mu} \oplus \bar{\nu}}((\Phi_1^{\mu})^{-1}(\rho_\mu(\gamma)z), e) \rho_E^{0 \oplus 0}(\gamma) \\ &\cdot \Phi_2^{\mu \oplus \nu}((\Phi_1^{\mu})^{-1}(z), e)^{-1}. \end{split}$$

And so to find the Kodaira-Spencer class, by Lemma 3.1 we only need to calculate:

$$\begin{aligned} \operatorname{Ad}\Phi_{2}^{\mu\oplus\nu} \circ (\Phi_{1}^{\mu})^{-1} \frac{d}{dt}|_{t=0} (\Phi_{2}^{\mu\oplus\nu+t\tilde{\mu}\oplus\tilde{\nu}} \circ (\Phi_{1}^{\mu})^{-1})^{-1} \bar{\partial} (\Phi_{2}^{\mu\oplus\nu+t\tilde{\mu}\oplus\tilde{\nu}} \circ (\Phi_{1}^{\mu})^{-1}) \\ = \operatorname{Ad}\Phi_{2}^{\mu\oplus\nu} \circ (\Phi_{1}^{\mu})^{-1} \\ \cdot \frac{d}{dt}|_{t=0} ((t\tilde{\mu}(\Phi_{2}^{\mu\oplus\nu+t\tilde{\mu}\oplus\tilde{\nu}})^{-1}\partial\Phi_{2}^{\mu\oplus\nu+t\tilde{\mu}\oplus\tilde{\nu}}) \circ \Phi_{1}^{\mu})^{-1} \bar{\partial} (\bar{\Phi}_{1}^{\mu})^{-1} \\ + \operatorname{Ad}\Phi_{2}^{\mu\oplus\nu} \circ (\Phi_{1}^{\mu})^{-1} \frac{d}{dt}|_{t=0} (\nu+t\tilde{\nu}) \circ (\Phi_{1}^{\mu})^{-1} \bar{\partial} (\bar{\Phi}_{1}^{\mu})^{-1} \\ = (\Phi_{1}^{\mu})_{*}^{-1} \left(\operatorname{Ad}\Phi_{2}^{\mu\oplus\nu} \left(\tilde{\mu}(\Phi_{2}^{\mu\oplus\nu})^{-1}\partial\Phi_{2}^{\mu\oplus\nu} + \tilde{\nu}\right)\right). \end{aligned}$$

Now to get the Kodaira-Spencer map we project on the harmonic (0, 1)-forms and remark that in the Teichmüller directions we can apply the usual arguments from the classical case of Bers's coordinates.

We see the map is injective and complex linear in both  $\tilde{\mu}$  and  $\tilde{\nu}$ . Since we know  $\mathcal{T} \times M'$  is a manifold the Implicit Function Theorem now implies that the coordinates we constructed are in fact holomorphic coordinates in a small neighbourhood. This completes the proof of Theorem 1.2.

#### 4 The Fibered Coordinates

In this section we will fuse Zograf and Takhtadzhyan's coordinates with Bers's coordinates in a kind of fibered manner in order also to produce coordinates on  $\mathcal{T} \times M'$ , which are complex analytic with respect to J.

Since we trough any stable bundle have a copy of  $\mathcal{T}$  embedded as a complex submanifold, we can construct fibered coordinates, once we identify the tangent spaces in the fiber direction locally along these copies of  $\mathcal{T}$ . We identify them by the maps

$$H^{0,1}(X_{\rho_0}, \operatorname{End} E_{\rho_E^0}) \ni \nu \to \nu^{\mu} = P^{0,1}_{\mu}((\Phi_1^{\mu})^{-1}_*(\nu)) \in H^{0,1}(X_{\rho_{\mu}}, \operatorname{End} E_{\rho_E^{0^{\mu}}}).$$

This identification gives us coordinates taking  $(\mu, \nu)$  to

$$(\rho_{\mu}, \rho_{E}^{\nu^{\mu}}) = \left(\Phi_{1}^{\mu} \circ \rho_{0}(\gamma) \circ (\Phi_{1}^{\mu})^{-1}, f^{\nu^{\mu}}(\rho_{\mu}(\gamma)z)\rho_{E}^{0^{0}}(\gamma)(f^{\nu^{\mu}}(z))^{-1}\right).$$

These are complex coordinates, since  $\nu^{\mu}$  are local holomorphic sections of the tangent bundle.

Before we calculate the Kodaira-Spencer maps for these coordinate curves, we will need to understand the derivatives of  $(\Phi_{1}^{\mu})^{-1}$ .

**Lemma 4.1.** We have the following two identities for  $(\Phi_1^{\mu})^{-1} : \mathbf{H} \to \mathbf{H}$ 

(4.1) 
$$\bar{\partial}(\Phi_1^{\mu})^{-1} = -\mu \circ (\Phi_1^{\mu})^{-1} \bar{\partial} \overline{(\Phi_1^{\mu})^{-1}}.$$

(4.2) 
$$\overline{\partial}(\overline{\Phi_1^{\mu}})^{-1} = \left(\frac{1}{1-|\mu|^2}\frac{1}{\overline{\partial}\overline{\Phi_1^{\mu}}}\right) \circ (\Phi_1^{\mu})^{-1}.$$

*Proof.* We consider the identity  $\Phi_1^{\mu} \circ (\Phi_1^{\mu})^{-1}(z) = z$ . And we use the differential equation for  $\Phi_1^{\mu}$  which is

$$\bar{\partial}\Phi_1^\mu = \mu \partial \Phi_1^\mu$$

to calculate:

$$\begin{split} 0 &= \bar{\partial} (\Phi_1^{\mu} \circ (\Phi_1^{\mu})^{-1}) = (\bar{\partial} \Phi_1^{\mu}) \circ (\Phi_1^{\mu})^{-1} \bar{\partial} (\overline{\Phi_1^{\mu}})^{-1} + (\partial \Phi_1^{\mu}) \circ (\Phi_1^{\mu})^{-1} \bar{\partial} (\Phi_1^{\mu})^{-1} \\ &= (\mu \partial \Phi_1^{\mu}) \circ (\Phi_1^{\mu})^{-1} \bar{\partial} (\overline{\Phi_1^{\mu}})^{-1} + (\partial \Phi_1^{\mu}) \circ (\Phi_1^{\mu})^{-1} \bar{\partial} (\Phi_1^{\mu})^{-1}. \end{split}$$

Now  $\partial \Phi_1^{\mu} \neq 0$  for  $\mu$  small, since  $\Phi_1$  is a continuous perturbation of the identity map Id :  $\mathbf{H} \to \mathbf{H}$ . We then calculate that

$$-\mu \circ (\Phi_1^{\mu})^{-1} \bar{\partial} \overline{(\Phi_1^{\mu})^{-1}} = \bar{\partial} (\Phi_1^{\mu})^{-1},$$

which is (4.1). We can use (4.1) to describe  $\overline{\partial}(\overline{\Phi_1^{\mu}})^{-1}$  by differentiating  $\Phi_1^{\mu} \circ (\Phi_1^{\mu})^{-1}(z) = z$ :

$$\begin{split} 1 &= \partial (\Phi_1^{\mu} \circ (\Phi_1^{\mu})^{-1}) = (\bar{\partial} \Phi_1^{\mu}) \circ (\Phi_1^{\mu})^{-1} \partial \overline{(\Phi_1^{\mu})^{-1}} + (\partial \Phi_1^{\mu}) \circ (\Phi_1^{\mu})^{-1} \partial (\Phi_1^{\mu})^{-1} \\ &= -\mu (\partial \Phi_1^{\mu}) \circ (\Phi_1^{\mu})^{-1} \overline{\mu} \overline{\partial} \overline{(\Phi_1^{\mu})^{-1}} + (\partial \Phi_1^{\mu}) \circ (\Phi_1^{\mu})^{-1} \partial (\Phi_1^{\mu})^{-1} \\ &= ((1 - |\mu|^2) (\partial \Phi_1^{\mu})) \circ (\Phi_1^{\mu})^{-1} \partial (\Phi_1^{\mu}))^{-1}, \end{split}$$

and so conjugating and isolating  $\bar{\partial} \overline{(\Phi_1^{\mu})^{-1}}$  we find:

$$\bar{\partial}(\overline{\Phi_1^{\mu}})^{-1} = \left(\frac{1}{1 - |\mu|^2} \frac{1}{\overline{\partial \Phi_1^{\mu}}}\right) \circ (\Phi_1^{\mu})^{-1}$$

which proves (4.2).

Let  $\kappa_\mu$  be an (n,m)-tensor with values in the holomorphic bundle  $E_{\rho_E^{o\mu}}$  on the Riemann surface  $X_{\rho_\mu}$  i.e.

$$\kappa_{\mu} \in C^{\infty}(X_{\rho_{\mu}}, T^{-n}X_{\rho_{\mu}} \otimes \overline{T}^{-m}X_{\rho_{\mu}} \otimes \operatorname{End} E_{\rho_{E}^{0\mu}}).$$

Then we have that  $(\Phi_1^{\mu})_*(\kappa_{\mu}) = (\kappa_{\mu} \circ \Phi_1^{\mu})(\partial \Phi_1^{\mu})^n (\overline{\partial \Phi_1^{\mu}})^m$  and so

$$(\Phi_1^{\mu})_*^{-1}(\kappa_0) = (\kappa_0 \circ (\Phi_1^{\mu})^{-1})(\partial \Phi_1^{\mu})^{-n} (\overline{\partial \Phi_1^{\mu}})^{-m}.$$

We have the families of unbounded operators

$$\bar{\partial}_{\mu,E_{\rho_E^{0^{\mu}}}} : L^2(X_{\rho_{\mu}}, \operatorname{End} E_{\rho_E^{0^{\mu}}}) \to L^2(X_{\rho_{\mu}}, T^{0,1} \otimes \operatorname{End} E_{\rho_E^{0^{\mu}}}),$$
$$\bar{\partial}_{\mu,E_{\rho_E^{0^{\mu}}}}^* : L^2(X_{\rho_{\mu}}, T^{0,1} \otimes \operatorname{End} E_{\rho_E^{0^{\mu}}}) \to L^2(X_{\rho_{\mu}}, \operatorname{End} E_{\rho_E^{0^{\mu}}}),$$
$$\Delta_{\mu,E_{\rho_E^{0^{\mu}}}} = \bar{\partial}^* \bar{\partial} : L^2(X_{\rho_{\mu}}, \operatorname{End} E_{\rho_E^{0^{\mu}}}) \to L^2(X_{\rho_{\mu}}, \operatorname{End} E_{\rho_E^{0^{\mu}}}),$$

and the finite range operator

$$P^{0,1}_{\mu,E_{\rho_E^{0^{\mu}}}}: L^2(X_{\rho_{\mu}}, T^{0,1} \otimes \operatorname{End} E_{\rho_E^{0^{\mu}}}) \to L^2(X_{\rho_{\mu}}, T^{0,1} \otimes \operatorname{End} E_{\rho_E^{0^{\mu}}})$$

given by

$$P^{0,1}_{\mu,E_{\rho_E^{0^{\mu}}}} = I - \bar{\partial}_{\mu,E_{\rho_E^{0^{\mu}}}} \Delta_{0,\mu,E_{\rho_E^{0^{\mu}}}}^{-1} \bar{\partial}^*_{\mu,E_{\rho_E^{0^{\mu}}}},$$

where  $\Delta_{0,\mu,E_{\rho_E^{0^{\mu}}}}$  is the restriction of  $\Delta_{\mu,E_{\rho_E^{0^{\mu}}}}$  to the orthogonal complement of the subspace consisting of constant functions tensor the identity, and  $P^{0,1}$  is the projection on the harmonic (0, 1)-forms. We will also need the following results of Takhtajan and Zograf.

**Lemma 4.2** ([TZ3]). We have the following variational formulae for the derivative at  $X_{\rho_0}$ 

$$\begin{aligned} \frac{d}{dt}|_{t=0}(\Phi_1^{t\tilde{\mu}})_*\bar{\partial}_{t\tilde{\mu},E_{\rho_E^{0\mu}}}(\Phi_1^{t\tilde{\mu}})_*^{-1} &= -\tilde{\mu}\partial_{0,E} \quad \frac{d}{d\tilde{t}}|_{t=0}(\Phi_1^{t\tilde{\mu}})_*\bar{\partial}_{t\tilde{\mu},E_{\rho_E^{0\mu}}}(\Phi_1^{t\tilde{\mu}})_*^{-1} &= 0\\ \frac{d}{dt}|_{t=0}(\Phi_1^{t\tilde{\mu}})_*\bar{\partial}_{t\tilde{\mu},E_{\rho_E^{0\mu}}}(\Phi_1^{t\tilde{\mu}})_*^{-1} &= 0 \quad \frac{d}{dt}|_{t=0}(\Phi_1^{t\tilde{\mu}})_*\bar{\partial}_{t\tilde{\mu},E_{\rho_E^{0\mu}}}(\Phi_1^{t\tilde{\mu}})_*^{-1} &= -\partial_{0,E}^*\bar{\mu}.\end{aligned}$$

We further have at  $(X_{\rho_{\mu}}, E_{\rho_{E}^{0^{\mu}}})$ , that

$$\begin{split} \frac{d}{dt}|_{t=0} (\Phi_1^{\mu+t\tilde{\mu}})_* P_{\mu+t\tilde{\mu},E_{\rho_E^{0\mu+t\tilde{\mu}}}}^{0,1} (\Phi_1^{\mu+t\tilde{\mu}})_*^{-1} \\ &= (\Phi_1^{\mu})_* P_{\mu,E_{\rho_E^{0\mu}}}^{0,1} (\Phi_1^{\mu})_*^{-1} \frac{d}{dt}|_{t=0} (\Phi_1^{\mu+t\tilde{\mu}})_* \bar{\partial}_{\mu+t\tilde{\mu},E_{\rho_E^{0\mu+t\tilde{\mu}}}} (\Phi_1^{\mu+t\tilde{\mu}})_*^{-1} \\ &\quad (\Phi_1^{\mu})_* \Delta_{0,\mu,E_{\rho_E^{0\mu}}}^{-1} \bar{\partial}_{\mu,E_{\rho_E^{0\mu}}}^* (\Phi_1^{\mu})_*^{-1} \\ &\quad + (\Phi_1^{\mu})_* \bar{\partial}_{\mu,E_{\rho_E^{0\mu}}} \Delta_{0,\mu,E_{\rho_E^{0\mu}}}^{-1} (\Phi_1^{\mu})_*^{-1}|_{t=0} (\Phi_1^{\mu+t\tilde{\mu}})_* \bar{\partial}_{\mu+t\tilde{\mu},E_{\rho_E^{0\mu+t\tilde{\mu}}}} (\Phi_1^{\mu+t\tilde{\mu}})_*^{-1} \\ &\quad \frac{d}{dt} (\Phi_1^{\mu})_* P_{\mu,E_{\rho_E^{0\mu+t\tilde{\mu}}}}^{0,1} (\Phi_1^{\mu})_*^{-1}. \end{split}$$

*Proof.* The first identities are proven in [TZ3, Equation (2.6)] (without the End*E* factor, which makes no difference), the last statement is seen straightforwardly as follows

$$\begin{split} \frac{d}{dt}|_{t=0} (\Phi_1^{\mu+t\tilde{\mu}})_* P^{0,1}_{\mu+t\tilde{\mu},E_{\rho_E^{0^{\mu+t\tilde{\mu}}}}} (\Phi_1^{\mu+t\tilde{\mu}})_*^{-1} &= \frac{d}{dt}|_{t=0} (\Phi_1^{\mu+t\tilde{\mu}})_* \bar{\partial}_{\mu+t\tilde{\mu},E_{\rho_E^{0^{\mu+t\tilde{\mu}}}}} (\Phi_1^{\mu+t\tilde{\mu}})_*^{-1} \\ & (\Phi_1^{\mu+t\tilde{\mu}})_* \Delta_{0,\mu+t\tilde{\mu},E_{\rho_E^{0^{\mu+t\tilde{\mu}}}}}^{-1} (\Phi_1^{\mu+t\tilde{\mu}})_*^{-1} (\Phi_1^{\mu+t\tilde{\mu}})_* \bar{\partial}_{\mu+t\tilde{\mu},E_{\rho_E^{0^{\mu+t\tilde{\mu}}}}}^{+1} (\Phi_1^{\mu+t\tilde{\mu}})_*^{-1}. \end{split}$$

We can then use the following identities

$$\begin{split} \frac{d}{dt}|_{t=0}(\Phi_{1}^{\mu+t\tilde{\mu}})_{*}\Delta_{0,\mu+t\tilde{\mu},E_{\rho_{E}^{0\mu+t\tilde{\mu}}}}^{-1}(\Phi_{1}^{\mu+t\tilde{\mu}})_{*}^{-1} &= -(\Phi_{1}^{\mu})_{*}\Delta_{0,\mu,E_{\rho_{E}^{0\mu}}}^{-1}(\Phi_{1}^{\mu})_{*}^{-1} \\ & \frac{d}{dt}|_{t=0}(\Phi_{1}^{\mu+t\tilde{\mu}})_{*}\Delta_{0,\mu+t\tilde{\mu},E_{\rho_{E}^{0\mu+t\tilde{\mu}}}}(\Phi_{1}^{\mu+t\tilde{\mu}})_{*}^{-1}(\Phi_{1}^{\mu})_{*}\Delta_{0,\mu,E_{\rho_{E}^{0\mu}}}^{-1}(\Phi_{1}^{\mu})_{*}^{-1}, \\ \frac{d}{dt}|_{t=0}(\Phi_{1}^{\mu+t\tilde{\mu}})_{*}\Delta_{0,\mu+t\tilde{\mu},E_{\rho_{E}^{0\mu+t\tilde{\mu}}}}(\Phi_{1}^{\mu+t\tilde{\mu}})_{*}^{-1} \\ &= \frac{d}{dt}|_{t=0}(\Phi_{1}^{\mu+t\tilde{\mu}})_{*}\bar{\partial}_{\mu+t\tilde{\mu},E_{\rho_{E}^{0\mu+t\tilde{\mu}}}}(\Phi_{1}^{\mu+t\tilde{\mu}})_{*}^{-1}(\Phi_{1}^{\mu})_{*}\bar{\partial}_{\mu,E_{\rho_{E}^{0\mu}}}^{*}(\Phi_{1}^{\mu})_{*}^{-1}. \end{split}$$

Now, putting this together and using that  $P^{0,1}_{\mu,E_{\rho_E^{0^{\mu}}}} = I - \bar{\partial}_{\mu,E_{\rho_E^{0^{\mu}}}} \Delta^{-1}_{0,\mu,E_{\rho_E^{0^{\mu}}}} \bar{\partial}^*_{\mu,E_{\rho_E^{0^{\mu}}}}$  we have the last identity.

20

**Proposition 4.3.** The Kodaira-Spencer map of the curve  $\rho_E^{(\nu+t\bar{\nu})^{\mu+t\bar{\mu}}}$  at t=0 is

$$\begin{split} KS_{\nu^{\mu}}(\tilde{\mu} \oplus \tilde{\nu}) = & P_{\mu}\tilde{\mu}^{\mu} \oplus P_{\nu^{\mu}}^{0,1} \left( Ad(f^{\nu^{\mu}})((f^{\nu^{\mu}})^{-1} \cdot (\partial f^{\nu^{\mu}})\tilde{\mu}^{\mu} + \tilde{\nu}^{\mu}) \right. \\ & \left. + Adf^{\nu^{\mu}} \left( \Phi_{1}^{\mu} \right)_{*}^{-1} \frac{d}{dt} |_{t=0} (\Phi_{1}^{\mu+t\tilde{\mu}})_{*}P_{\mu+t\tilde{\mu}}^{0,1} (\Phi_{1}^{\mu+t\tilde{\mu}})_{*}^{-1}(\nu))(1 - |\mu \circ (\Phi_{1}^{\mu})^{-1}|^{2}) \right) \right) \end{split}$$

with  $\tilde{\mu}^{\mu} = (\frac{\tilde{\mu}}{1-|\mu|^2} \frac{\partial \Phi_1^{\mu}}{\partial \Phi_1^{\mu}}) \circ (\Phi_1^{\mu})^{-1}$  and  $P_{\mu}$  and  $P_{\nu^{\mu}}^{0,1}$  the  $L^2$ -projections on the harmonic forms  $\mathcal{H}^{0,1}(X_{\rho_{\mu}}, TX_{\rho_{\mu}})$  respectively  $\mathcal{H}^{0,1}(X_{\rho_{\mu}}, \operatorname{End} E_{\rho_{E}^{\nu^{\mu}}})$ 

*Proof.* First, we observe that the Teichmüller direction is unchanged from the classical case. Now we want to use Lemma 3.1, and so using that  $\rho_E^{(\nu+t\bar{\nu})^{\mu+t\bar{\mu}}}$  is independent of z we find that

$$\begin{split} \rho_E^{(\nu+t\bar{\nu})^{\mu+t\bar{\mu}}}(\gamma) &= f^{(\nu+t\bar{\nu})^{\mu+t\bar{\mu}}}(\rho_{\mu+t\bar{\mu}}(\gamma)z)\rho_E(\gamma)(f^{(\nu+t\bar{\nu})^{\mu+t\bar{\mu}}}(z))^{-1} \\ &= f^{(\nu+t\bar{\nu})^{\mu+t\bar{\mu}}}((\Phi_1^{\mu+t\bar{\mu}}((\Phi_1^{\mu})^{-1}(\rho_{\mu}(\gamma)z)))\rho_E(\gamma) \\ &\quad (f^{(\nu+t\bar{\nu})^{\mu+t\bar{\mu}}}((\Phi_1^{\mu+t\bar{\mu}}((\Phi_1^{\mu})^{-1}(z)))^{-1}. \end{split}$$

Next we have to calculate

$$\begin{split} \operatorname{Ad}(f^{\nu^{\mu}}) & \frac{d}{dt}|_{t=0} \left( \left( \left( f^{(\nu+t\bar{\nu})^{\mu+t\bar{\mu}}} \circ \Phi_{1}^{\mu+t\bar{\mu}} \right) \circ (\Phi_{1}^{\mu})^{-1} \right) \bar{\partial} \left( \left( f^{\nu^{\mu+t\bar{\mu}}} \right)^{-1} \circ \Phi_{1}^{\mu+t\bar{\mu}} \circ (\Phi_{1}^{\mu})^{-1} \right) \right) \\ &= \operatorname{Ad}(f^{\nu^{\mu}}) \frac{d}{dt}|_{t=0} \left( \left( \left( \nu + t\tilde{\nu} \right)^{\mu+t\bar{\mu}} \circ \Phi_{1}^{\mu+t\bar{\mu}} \circ (\Phi_{1}^{\mu})^{-1} \right) \left( \overline{\partial} (\Phi_{1}^{\mu+t\bar{\mu}} \circ (\Phi_{1}^{\mu})^{-1}) \right) \right) \\ &+ \operatorname{Ad}(f^{\nu^{\mu}}) \frac{d}{dt}|_{t=0} \left( \left( f^{(\nu+t\bar{\nu})^{\mu+t\bar{\mu}}} \circ \Phi_{1}^{\mu+t\bar{\mu}} \circ (\Phi_{1}^{\mu})^{-1} \right)^{-1} \\ &\cdot \left( \partial f^{\nu^{\mu}} \right) \circ \Phi_{1}^{\mu+t\bar{\mu}} \circ (\Phi_{1}^{\mu})^{-1} (\overline{\partial} (\Phi_{1}^{\mu+t\bar{\mu}} \circ (\Phi_{1}^{\mu})^{-1})) \right). \end{split}$$

For the first term we find that

$$\begin{split} \operatorname{Ad}(f^{\nu^{\mu}}) \frac{d}{dt}|_{t=0} \left( \left( \left( \nu + t\tilde{\nu} \right)^{\mu + t\tilde{\mu}} \right) \circ \Phi_{1}^{\mu + t\tilde{\mu}} \circ (\Phi_{1}^{\mu})^{-1} \right) \left( \overline{\partial} (\Phi_{1}^{\mu + t\tilde{\mu}} \circ (\Phi_{1}^{\mu})^{-1}) \right) \right) = \\ = \frac{d}{dt}|_{t=0} (\operatorname{Ad}(f^{\nu^{\mu}}) ((\nu + t\tilde{\nu})^{\mu + t\tilde{\mu}}) \circ \Phi_{1}^{\mu + t\tilde{\mu}} \circ (\Phi_{1}^{\mu})^{-1} \\ \cdot \overline{\left( (\partial \Phi_{1}^{\mu + t\tilde{\mu}} \circ (\Phi_{1}^{\mu})^{-1}) \partial (\Phi_{1}^{\mu})^{-1} + (\bar{\partial} \Phi_{1}^{\mu + t\tilde{\mu}} \circ (\Phi_{1}^{\mu})^{-1}) \partial (\bar{\Phi}_{1}^{\mu})^{-1} \right)} \end{split}$$

We can now rewrite the last factor using (4.1) and (4.2) and their conjugates to get that

$$\begin{split} (\partial \Phi_1^{\mu+t\bar{\mu}} \circ (\Phi_1^{\mu})^{-1}) \partial (\Phi_1^{\mu})^{-1} &+ (\bar{\partial} \Phi_1^{\mu+t\bar{\mu}} \circ (\Phi_1^{\mu})^{-1}) \partial (\bar{\Phi}_1^{\mu})^{-1} \\ &= (\partial \Phi_1^{\mu+t\bar{\mu}} \circ (\Phi_1^{\mu})^{-1}) \partial (\Phi_1^{\mu})^{-1} \\ &+ (((\mu+t\bar{\mu})\partial \Phi_1^{\mu+t\bar{\mu}}) \circ (\Phi_1^{\mu})^{-1}) (-\bar{\mu} \circ (\Phi_1^{\mu})^{-1} \partial (\Phi_1^{\mu})^{-1}). \\ &= (\partial \Phi_1^{\mu+t\bar{\mu}} \circ (\Phi_1^{\mu})^{-1}) \partial (\Phi_1^{\mu})^{-1} (1 - (((\mu+t\bar{\mu})\bar{\mu}) \circ (\Phi_1^{\mu})^{-1})). \end{split}$$

Using that  $(\Phi_1^{\mu})_*(\nu) = \nu \circ \Phi_1^{\mu} \overline{\partial \Phi_1^{\mu}}$  in the first term we find that

Here we have used the result from Lemma 4.2 to calculate the derivative of the projection.

For the second term we rewrite

$$\begin{aligned} \overline{\partial} (\Phi_1^{\mu+t\bar{\mu}} \circ (\Phi_1^{\mu})^{-1}) \\ &= (\bar{\partial} \Phi_1^{\mu+t\bar{\mu}}) \circ (\Phi_1^{\mu})^{-1} \overline{\partial} \overline{(\Phi_1^{\mu})^{-1}} + (\partial \Phi_1^{\mu+t\bar{\mu}}) \circ (\Phi_1^{\mu})^{-1} \overline{\partial} (\Phi_1^{\mu})^{-1} \\ (4.3) &= ((\mu+t\bar{\mu})\partial \Phi_1^{\mu+t\bar{\mu}}) \circ (\Phi_1^{\mu})^{-1} \overline{\partial} \overline{(\Phi_1^{\mu})^{-1}} + (\partial \Phi_1^{\mu+t\bar{\mu}}) \circ (\Phi_1^{\mu})^{-1} \overline{\partial} (\Phi_1^{\mu})^{-1} \end{aligned}$$

using (4.2) and (4.1) in (4.3) and find that

$$\overline{\partial}(\Phi_1^{\mu+t\tilde{\mu}} \circ (\Phi_1^{\mu})^{-1}) = \left(\frac{t\tilde{\mu}}{1-|\mu|^2} \frac{\partial \Phi_1^{\mu+t\tilde{\mu}}}{\partial \Phi_1^{\mu}}\right) \circ (\Phi_1^{\mu})^{-1},$$

which implies that

$$\begin{split} \operatorname{Ad}(f^{\nu^{\mu}}) \frac{d}{dt}|_{t=0} \Big( \left( f^{(\nu+t\bar{\nu})^{\mu+t\bar{\mu}}} \circ \Phi_{1}^{\mu+t\bar{\mu}} \circ (\Phi_{1}^{\mu})^{-1} \right)^{-1} \\ & \cdot (\partial f^{\nu^{\mu}}) \circ \Phi_{1}^{\mu+t\bar{\mu}} \circ (\Phi_{1}^{\mu})^{-1} (\overline{\partial}(\Phi_{1}^{\mu+t\bar{\mu}} \circ (\Phi_{1}^{\mu})^{-1})) \Big) \\ &= \operatorname{Ad}(f^{\nu^{\mu}}) ((f^{\nu^{\mu}})^{-1} \cdot (\partial f^{\nu^{\mu}}) (\tilde{\mu}^{\mu}), \end{split}$$

where  $\tilde{\mu}^{\mu} = \left(\frac{\tilde{\mu}}{1-|\mu|^2} \frac{\partial \Phi_1^{\mu}}{\partial \Phi_1^{\mu}}\right) \circ (\Phi_1^{\mu})^{-1}$ . And so we have that

$$\begin{split} \operatorname{Ad}(f^{\nu^{\mu}}) & \frac{d}{dt}|_{t=0} \left( \left( (f^{(\nu+t\bar{\nu})^{\mu+t\bar{\mu}}} \circ \Phi_{1}^{\mu+t\bar{\mu}}) \circ (\Phi_{1}^{\mu})^{-1} \right) \bar{\partial} \left( (f^{\nu^{\mu+t\bar{\mu}}})^{-1} \circ \Phi_{1}^{\mu+t\bar{\mu}} \circ (\Phi_{1}^{\mu})^{-1} \right) \right) \\ &= \operatorname{Ad}(f^{\nu^{\mu}}) \left( (\Phi_{1}^{\mu})_{*}^{-1} \frac{d}{dt}|_{t=0} (\Phi_{1}^{\mu+t\bar{\mu}})_{*} P^{0,1}_{\mu+t\bar{\mu},E_{\rho_{E}^{0}}^{\mu+t\bar{\mu}}} (\Phi_{1}^{\mu+t\bar{\mu}})_{*}^{-1}(\nu))) \right) \\ & (4.4) \\ & (1-|\mu|^{2}) \circ (\Phi_{1}^{\mu})^{-1}) \right) + \operatorname{Ad}(f^{\nu^{\mu}}) \tilde{\nu}^{\mu} + \operatorname{Ad}(f^{\nu^{\mu}}) ((f^{\nu^{\mu}})^{-1} \cdot (\partial f^{\nu^{\mu}}) (\tilde{\mu}^{\mu})). \end{split}$$

We have thus shown that composing with the projection gives us the harmonic representative.  $\hfill \Box$ 

#### 4.1 Comparison of the Two Tangent Maps and a proof of the first part of Theorem 1.3

We compare

$$P^{0,1}_{\mu\oplus\nu,}\left(\left(\Phi_1^{\mu}\right)_*^{-1}\left(\mathrm{Ad}\Phi_2^{\mu\oplus\nu}\left(\tilde{\mu}(\Phi_2^{\mu\oplus\nu})^{-1}\partial\Phi_2^{\mu\oplus\nu}+\tilde{\nu}\right)\right)\right)$$

and

$$\begin{split} & P^{0,1}_{\nu^{\mu}} \left( \mathrm{Ad}(f^{\nu^{\mu}}) ((f^{\nu^{\mu}})^{-1} \cdot (\partial f^{\nu^{\mu}}) \tilde{\mu}^{\mu} + \tilde{\nu}^{\mu}) \right. \\ & + \mathrm{Ad}f^{\nu^{\mu}} \left( \Phi^{\mu}_{1} \right)^{-1}_{*} \frac{d}{dt} |_{t=0} (\Phi^{\mu+t\tilde{\mu}}_{1})_{*} P^{0,1}_{\mu+t\tilde{\mu},E_{\rho^{\mu+t\tilde{\mu}}_{E}}} (\Phi^{\mu+t\tilde{\mu}}_{1})^{-1}_{*}(\nu)) (1 - |\mu \circ (\Phi^{\mu}_{1})^{-1}|^{2}) \right). \end{split}$$

First we observe that

$$\mathrm{Ad}f^{\nu^{\mu}}(\Phi_{1}^{\mu})_{*}^{-1}\frac{d}{dt}|_{t=0}(\Phi_{1}^{\mu+t\tilde{\mu}})_{*}P^{0,1}_{\mu+t\tilde{\mu},E_{\rho_{E}^{0\mu+t\tilde{\mu}}}}(\Phi_{1}^{\mu+t\tilde{\mu}})_{*}^{-1}(\nu))(1-|\mu\circ(\Phi_{1}^{\mu})^{-1}|^{2})$$

vanishes to first order in  $\nu$  and  $\mu$  at the center, since we either differentiate with respect to  $\mu$  and set  $\nu = 0$  or we differentiate with respect to  $\nu$  and then we find, when we evaluate at  $\mu = 0$ , that  $\bar{\partial}_{0,E}^* \nu = 0$ , from the expression in Lemma 4.2.

Next we compare

$$(\Phi_1^{\mu})_*^{-1} \left( \mathrm{Ad} \Phi_2^{\mu \oplus \nu} \left( \tilde{\mu} (\Phi_2^{\mu \oplus \nu})^{-1} \partial \Phi_2^{\mu \oplus \nu} \right) \right)$$

with

$$\operatorname{Ad}(f^{\nu^{\mu}})((f^{\nu^{\mu}})^{-1} \cdot (\partial f^{\nu^{\mu}})\tilde{\mu}^{\mu})$$

We observe, that since  $\partial I = 0$  both  $(\Phi_2^{\mu\oplus\nu})^{-1}\partial\Phi_2^{\mu\oplus\nu}$  and  $(f^{\nu^{\mu}})^{-1} \cdot (\partial f^{\nu^{\mu}})$  vanish unless we differentiate it with respect to the moduli space direction or the Teichmüller direction. If we differentiate with respect to  $\mu$  we get  $\frac{\partial}{\partial \epsilon} \nu^{\epsilon \mu}$ , but at  $\nu = 0$  this is 0. This means we can compare the two after evaluating  $\mu = 0$ , and then we have  $f^{\nu^0} = \Phi^{0\oplus\nu}$ , and so they agree to first order.

The last terms to consider are  $(\Phi_1^{\mu})_*^{-1} \left( \operatorname{Ad} \Phi_2^{\mu \oplus \nu}(\tilde{\nu}) \right)$  and  $\operatorname{Ad}(f^{\nu\mu}) \tilde{\nu}^{\mu}$ . Now, if we put  $\mu = 0$  the terms agree. If we differentiate with respect to  $\mu$ , we can put  $\nu = 0$  first. We are differentiating a term of the form  $\bar{\partial}_{\mu,E_{\rho_E^{\mu\mu}}} \Delta_{0,\mu,E_{\rho_E^{\mu\mu}}}^{-1} \bar{\partial}_{\mu,E_{\rho_E^{\mu\mu}}}^{+1} \bar{\partial}_{\mu,E_{\rho_E^{\mu\mu}}}^{+1} \bar{\partial}_{\mu,E_{\rho_E^{\mu\mu}}}^{+1} \bar{\partial}_{\mu,E_{\rho_E^{\mu\mu}}}^{+1} \Delta_{0,\mu,E_{\rho_E^{\mu\mu}}}^{-1} \bar{\partial}_{\mu,E_{\rho_E^{\mu\mu}}}^{+1} \bar{\partial}_{\mu,E_$ 

#### 5 Variation of the Metric

In order to prove that our new coordinates are not the same as the fibered coordinates discussed above, we shall consider the variation of the metric in both set of coordinates and use the resulting formulae to demonstrate that they are not identical to third order.

#### 5.1 Variation in the Universal Coordinates

In this section will calculate the second variation of the metric using the coordinates from Theorem 1.1. In the next section we will do the same for the fibered coordinates, and use this to show that the two sets of coordinates differ at third order. So first we consider the function  $(\bar{\Phi}_2^{\epsilon(\mu\oplus\nu)})^T \bar{\Phi}_2^{\epsilon(\mu\oplus\nu)})$ . This transforms as a function on X with values in EndE, our reference point. Now, to further understand this function, we look at  $\frac{d}{d\epsilon}|_{\epsilon=0}(\bar{\Phi}_2^{\epsilon(\mu\oplus\nu)})^T \Phi_2^{\epsilon(\mu\oplus\nu)})$ . Then we find that

$$\begin{split} \Delta \frac{d}{d\epsilon}|_{\epsilon=0} (\bar{\Phi}_2^{\epsilon(\mu\oplus\nu)})^T \Phi_2^{\epsilon(\mu\oplus\nu)} &= \Delta (\frac{d}{d\epsilon}|_{\epsilon=0} (\bar{\Phi}_+^{\epsilon(\mu\oplus\nu)})^T + \frac{d}{d\epsilon}|_{\epsilon=0} \Phi_-^{\epsilon(\mu\oplus\nu)} \\ &+ \frac{d}{d\epsilon}|_{\epsilon=0} (\bar{\Phi}_-^{\epsilon(\mu\oplus\nu)})^T + \frac{d}{d\epsilon}|_{\epsilon=0} \Phi_+^{\epsilon(\mu\oplus\nu)}), \end{split}$$

and since  $\Phi_{-}^{\epsilon(\mu\oplus\nu)}$  is antiholomorphic we get that

$$\Delta \frac{d}{d\epsilon}|_{\epsilon=0} (\bar{\Phi}_2^{\epsilon(\mu\oplus\nu)})^T \Phi_2^{\epsilon(\mu\oplus\nu)} = \frac{d}{d\epsilon}|_{\epsilon=0} \Delta (\bar{\Phi}_+^{\epsilon(\mu\oplus\nu)})^T + \frac{d}{d\epsilon}|_{\epsilon=0} \Delta \Phi_+^{\epsilon(\mu\oplus\nu)})$$

We now use that  $(\bar{\partial} - \epsilon \mu \partial) \Phi^{\epsilon(\mu \oplus \nu)}_+ = 0$  and  $\Delta = y^{-2} \partial \bar{\partial}$  to see that

$$\Delta \frac{d}{d\epsilon}|_{\epsilon=0}(\bar{\Phi}_2^{\epsilon(\mu\oplus\nu)})^T \Phi_2^{\epsilon(\mu\oplus\nu)} = y^{-2}\bar{\mu}\bar{\partial}\bar{\partial}(\bar{\Phi}_+^{0(\mu\oplus\nu)})^T + y^{-2}\mu\partial\partial\Phi_+^{0(\mu\oplus\nu)}) = 0,$$

since  $\Phi_+^0 = I$ , and so the derivative is 0. This allows us to conclude, that  $\frac{d}{d\epsilon}|_{\epsilon=0}(\bar{\Phi}_2^{\epsilon(\mu\oplus\nu)})^T \Phi_2^{\epsilon(\mu\oplus\nu)})$  is a constant multiple of the identity element in  $\operatorname{End} E$ , and because of the determinant criteria in Theorem 1.1 we have

$$\begin{split} 0 &= \frac{d}{d\epsilon}|_{\epsilon=0} (\det(\bar{\Phi}_2^{\epsilon(\mu\oplus\nu)})^T \Phi_2^{\epsilon(\mu\oplus\nu)}) \\ &= \operatorname{tr}|_{\epsilon=0} ((\bar{\Phi}_2^0)^T \Phi_2^0)^{-1} \frac{d}{d\epsilon}|_{\epsilon=0} (\bar{\Phi}_2^{\epsilon(\mu\oplus\nu)})^T \Phi_2^{\epsilon(\mu\oplus\nu)} = \operatorname{tr} \frac{d}{d\epsilon}|_{\epsilon=0} (\bar{\Phi}_2^{\epsilon(\mu\oplus\nu)})^T \Phi_2^{\epsilon(\mu\oplus\nu)}, \end{split}$$

and so  $\frac{d}{d\epsilon}|_{\epsilon=0}(\bar{\Phi}_2^{\epsilon(\mu\oplus\nu)})^T \Phi_2^{\epsilon(\mu\oplus\nu)} = 0$ . We see that this immediately implies that  $\frac{d}{d\epsilon}|_{\epsilon=0}\partial \Phi_+^{\epsilon(\mu\oplus\nu)} = -\overline{\frac{d}{d\epsilon}|_{\epsilon=0}\bar{\partial}\Phi_-^{\epsilon(\mu\oplus\nu)}}^T = 0$ . We can study

$$\frac{d}{d\bar{\epsilon}}|_{\epsilon=0}(\bar{\Phi}_2^{\epsilon(\mu\oplus\nu)})^T\Phi_2^{\epsilon(\mu\oplus\nu)}$$

similarly and conclude that

$$\frac{d}{d\bar{\epsilon}}|_{\epsilon=0}\partial\Phi_+^{\epsilon(\mu\oplus\nu)} = -\overline{\frac{d}{d\epsilon}}|_{\epsilon=0}\bar{\partial}\Phi_-^{\epsilon(\mu\oplus\nu)}{}^T = -\bar{\nu}^T.$$

Now, we want to understand the variation of the  $\bar{\partial}_{\mu,E_{\rho(\mu\oplus\nu)}}$ -operator on functions and  $\bar{\partial}^*_{\mu,E_{\rho(\mu\oplus\nu)}}$ -operator on (0, 1)-forms, since they play a central role in understanding the tangent spaces over the universal moduli space. We work on the universal cover and pull back our family of differential operators from the universal cover of  $(X_{\rho_{\mu}}, E_{\rho^{(\mu\oplus\nu)}})$  to that of  $(X_{\rho_0}, E)$ , in terms of representations. Then  $\bar{\partial}_{\mu, E_{\rho^{(\mu\oplus\nu)}}}$  is just represented by  $\bar{\partial}$  on **H** 

$$\begin{aligned} \frac{d}{d\epsilon}|_{\epsilon=0} \mathrm{Ad}\Phi_{2}^{\epsilon\mu\oplus\nu}(\Phi_{1}^{\epsilon\mu})_{*}\bar{\partial}(\Phi_{1}^{\epsilon\mu})_{*}^{-1}(\mathrm{Ad}\Phi_{2}^{\epsilon\mu\oplus\nu})^{-1} \\ &= \frac{d}{d\epsilon}|_{\epsilon=0} \mathrm{Ad}\Phi_{2}^{\epsilon\mu\oplus\nu}\frac{1}{1-|\epsilon\mu|^{2}}(\bar{\partial}-\mu\partial)(\mathrm{Ad}\Phi_{2}^{\epsilon\mu\oplus\nu})^{-1} \\ &= \frac{d}{d\epsilon}|_{\epsilon=0}\frac{1}{1-|\epsilon\mu|^{2}}(\mathrm{Ad}\Phi_{2}^{\epsilon\mu\oplus\nu}(\mathrm{ad}(\bar{\partial}-\epsilon\mu\partial)\Phi_{2}^{\epsilon\mu\oplus\nu})(\mathrm{Ad}\Phi_{2}^{\epsilon\mu\oplus\nu})^{-1} \\ &\quad + (\bar{\partial}-\epsilon\mu\partial)) \\ &= \frac{d}{d\epsilon}|_{\epsilon=0}\frac{1}{1-|\epsilon\mu|^{2}}(\epsilon\mathrm{ad}\mathrm{Ad}\Phi_{2}^{\epsilon\mu\oplus\nu}\nu+(\bar{\partial}-\epsilon\mu\partial)) \\ &= \mathrm{ad}\nu-\mu\partial. \end{aligned}$$
(5.1)

Likewise we find that the variation of  $\bar{\partial}^* = -\rho^{-1}\partial$ , where also the first derivative of density  $\rho$  is zero at the center point of our coordinates([W]). We begin by observing that on (0, 1)-forms we have that

$$\begin{split} (\Phi_1^{\epsilon\mu})_*\partial(\Phi_1^{\epsilon\mu})_*^{-1}\alpha &= (\Phi_1^{\epsilon\mu})_*\partial(\alpha \circ (\Phi_1^{\epsilon\mu})^{-1}\frac{1}{\bar{\partial}\overline{\Phi}_1^{\epsilon\mu} \circ (\Phi_1^{\epsilon\mu})^{-1}})\\ &= (\Phi_1^{\epsilon\mu})_*((\partial\alpha) \circ (\Phi_1^{\epsilon\mu})^{-1}\frac{\partial(\Phi_1^{\epsilon\mu})^{-1}}{\bar{\partial}\overline{\Phi}_1^{\epsilon\mu} \circ (\Phi_1^{\epsilon\mu})^{-1}}\\ &+ (\bar{\partial}\alpha) \circ (\Phi_1^{\epsilon\mu})^{-1}\frac{\partial(\overline{\Phi}_1^{\epsilon\mu})^{-1}}{\bar{\partial}\overline{\Phi}_1^{\epsilon\mu} \circ (\Phi_1^{\epsilon\mu})^{-1}}\\ &- \alpha \circ (\Phi_1^{\epsilon\mu})^{-1}\frac{\overline{(\partial(\bar{\partial}\Phi_1^{\epsilon\mu})) \circ (\Phi_1^{\epsilon\mu})^{-1}(\bar{\partial}\overline{\Phi}_1^{\epsilon\mu})^{-1}}}{(\bar{\partial}\overline{\Phi}_1^{\epsilon\mu} \circ (\Phi_1^{\epsilon\mu})^{-1})^2})\\ &- \alpha \circ (\Phi_1^{\epsilon\mu})^{-1}\frac{\overline{(\partial(\bar{\partial}\Phi_1^{\epsilon\mu}) \circ (\Phi_1^{\epsilon\mu})^{-1}(\bar{\partial}\Phi_1^{\epsilon\mu})^{-1}}}{(\bar{\partial}\overline{\Phi}_1^{\epsilon\mu} \circ (\Phi_1^{\epsilon\mu})^{-1})^2})\\ &= \frac{1}{1 - |\epsilon\mu|^2}(\partial - \bar{\epsilon}\bar{\mu}\bar{\partial} - \bar{\epsilon}(\bar{\partial}\bar{\mu})) = \frac{1}{1 - |\epsilon\mu|^2}(\partial - \bar{\epsilon}\bar{\partial}\mu)). \end{split}$$

And so we find that

$$\begin{split} \frac{d}{d\bar{\epsilon}}|_{\epsilon=0} \mathrm{Ad}\Phi_{2}^{\epsilon\mu\oplus\nu}(\Phi_{1}^{\epsilon\mu})_{*}\bar{\partial}^{*}(\Phi_{1}^{\epsilon\mu})_{*}^{-1}(\mathrm{Ad}\Phi_{2}^{\epsilon\mu\oplus\nu})^{-1} \\ &= \frac{d}{d\bar{\epsilon}}|_{\epsilon=0} \mathrm{Ad}\Phi_{2}^{\epsilon\mu\oplus\nu}\frac{-\rho^{-1}}{1-|\epsilon\mu|^{2}}(\partial-\bar{\epsilon}\bar{\partial}\bar{\mu})(\mathrm{Ad}\Phi_{2}^{\epsilon\mu\oplus\nu})^{-1} \\ &= \frac{d}{d\bar{\epsilon}}|_{\epsilon=0}\frac{-\rho^{-1}}{1-|\epsilon\mu|^{2}}(\mathrm{Ad}\Phi_{2}^{\epsilon\mu\oplus\nu}(\mathrm{Ad}(\partial-\bar{\epsilon}\bar{\partial}\bar{\mu})\Phi_{2}^{\epsilon\mu\oplus\nu})^{-1} \\ &\quad + (\bar{\partial}-\bar{\epsilon}\bar{\partial}\bar{\mu})) \end{split}$$

$$(5.2) = \frac{d}{d\bar{\epsilon}}|_{\epsilon=0} \frac{-\rho^{-1}}{1-|\epsilon\mu|^2} (\operatorname{ad}(\operatorname{Ad}(\Phi_2^{\epsilon(\mu\oplus\nu)})^{-1}\partial\Phi_2^{\epsilon(\mu\oplus\nu)}) - \bar{\epsilon}\bar{\mu}\rho^{-1}\operatorname{ad}\bar{\partial}\Phi_2^{\epsilon(\mu\oplus\nu)} + (\partial - \bar{\epsilon}\bar{\partial}\bar{\mu})) = -\ast\operatorname{ad}\nu \ast + \bar{\mu}\bar{\partial}\rho^{-1},$$

where the equality follows from the equation  $\partial \mu = 2y^{-1}\mu$ , and  $\rho^{-1} = y^2$ .

This is the first step in understanding the metric on the universal moduli space of pairs of a Riemann surface and a holomorphic bundle over it, given at a point (X, E) by identifying the tangent space with  $\mathcal{H}^{0,1}(X, TX) \oplus \mathcal{H}^{0,1}(X, \text{End}E)$ . Two elements  $\mu_1 \oplus \nu_1$  and  $\mu_2 \oplus \nu_2$  can be paired as follows

$$g(\mu_1 \oplus \nu_1, \mu_2 \oplus \nu_2) = \int_{\Sigma} (\rho_X \mu_1 \bar{\mu}_2 + i \mathrm{tr} \nu_1 \wedge \star \bar{\nu}_2^T),$$

where  $\rho_X$  is the density of the hyperbolic metric corresponding to the complex structure on X. Since the term  $\int_{\Sigma} \rho_X \mu_1 \bar{\mu}_2$ , is independent of the bundle, nothing has changed compared to the situation on Teichmüller space. Let us examine the term  $\int_{\Sigma} \text{tr} \nu_1 \wedge \bar{(} - \star) \nu_2^T$ . Since we are evaluation the metric on tangent vectors,  $-\star$  will act by -i and so we replace it in the following to avoid confusion.

In coordinates around (X, E) we have, using Proposition 3.2, that the metric is given by

$$g_{\epsilon(\mu\oplus\nu)}^{VB}(\mu_{1}\oplus\nu_{1},\mu_{2}\oplus\nu_{2})$$

$$= -i\int_{\Sigma} \operatorname{tr} P_{\epsilon(\mu\oplus\nu)}^{0,1}(\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*}^{-1}\operatorname{Ad}(\Phi_{2}^{\epsilon(\mu\oplus\nu)})\nu_{1}$$

$$\wedge \overline{P_{\epsilon(\mu\oplus\nu)}^{0,1}((\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*}^{-1}\operatorname{Ad}\Phi_{2}^{\epsilon(\mu\oplus\nu)})\nu_{2}}^{T}$$

$$- i\int_{\Sigma} \operatorname{tr} P_{\epsilon(\mu\oplus\nu)}^{0,1}(\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*}^{-1}\operatorname{Ad}(\Phi_{2}^{\epsilon(\mu\oplus\nu)})\mu_{1}(\Phi_{2}^{\epsilon(\mu\oplus\nu)})^{-1}\partial\Phi_{2}^{\epsilon(\mu\oplus\nu)}$$

$$\wedge \overline{P_{\epsilon(\mu\oplus\nu)}^{0,1}((\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*}^{-1}\operatorname{Ad}\Phi_{2}^{\epsilon(\mu\oplus\nu)})\nu_{2}}^{T}$$

$$- i\int_{\Sigma} \operatorname{tr} P_{\epsilon(\mu\oplus\nu)}^{0,1}(\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*}^{-1}\operatorname{Ad}(\Phi_{2}^{\epsilon(\mu\oplus\nu)})\nu_{1}$$

$$\wedge \overline{P_{\epsilon(\mu\oplus\nu)}^{0,1}(\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*}^{-1}(\operatorname{Ad}\Phi_{2}^{\epsilon(\mu\oplus\nu)})\mu_{2}(\Phi_{2}^{\epsilon(\mu\oplus\nu)})\partial\Phi_{2}^{\epsilon(\mu\oplus\nu)}}^{T}$$

$$- i\int_{\Sigma} \operatorname{tr} P_{\epsilon(\mu\oplus\nu)}^{0,1}(\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*}^{-1}\operatorname{Ad}(\Phi_{2}^{\epsilon(\mu\oplus\nu)})\mu_{1}(\Phi_{2}^{\epsilon(\mu\oplus\nu)})^{-1}\partial\Phi_{2}^{\epsilon(\mu\oplus\nu)}}^{T}.$$
(5.3)

Now we can use that  $P^{0,1}_{\epsilon(\mu\oplus\nu)}$  is self-adjoint with respect to the metric to rewrite the terms as follows

(5.4) 
$$\int_{\Sigma} \operatorname{tr} P^{0,1}_{\epsilon(\mu \oplus \nu)} (\Phi_1^{\epsilon(\mu \oplus \nu)})_*^{-1} \operatorname{Ad}(\Phi_2^{\epsilon(\mu \oplus \nu)}) \nu_1 \wedge \overline{P^{0,1}_{\epsilon(\mu \oplus \nu)} ((\Phi_1^{\epsilon(\mu \oplus \nu)})_*^{-1} \operatorname{Ad}\Phi_2^{\epsilon(\mu \oplus \nu)})} \nu_2^{-1}$$

26

Coordinates for the Universal Moduli Space of Holomorphic Vector Bundles 27

$$= \int_{\Sigma} \operatorname{trAd}(\overline{\Phi_{2}^{\epsilon(\mu\oplus\nu)}}^{T} \Phi_{2}^{\epsilon(\mu\oplus\nu)})(1 - |\epsilon\mu|^{2}) \operatorname{Ad}(\Phi_{2}^{\epsilon(\mu\oplus\nu)})^{-1} (\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*} P_{\epsilon(\mu\oplus\nu)}^{0,1} (\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*}^{-1} \operatorname{Ad}(\Phi_{2}^{\epsilon(\mu\oplus\nu)})\nu_{1} \wedge \overline{\nu_{2}}^{T}$$

Since  $(\Phi_1^{\epsilon\mu})_*(d\bar{z} \wedge dz) = (|\partial \Phi_1^{\epsilon\mu}|^2 - |\bar{\partial} \Phi_1^{\epsilon\mu}|^2)d\bar{z} \wedge dz$ . Also recall that

$$(\Phi_1^{\epsilon\mu})_*^{-1}\nu = \left(\frac{\nu}{\bar{\partial}\bar{\Phi}_1^{\epsilon\mu}}\right) \circ (\Phi_1^{\epsilon\mu})^{-1}$$

and  $(\Phi_1^{\epsilon\mu})_* P^{0,1}_{\epsilon(\mu\oplus\nu)}h = (\bar{\partial}\bar{\Phi}_1^{\epsilon\mu})(P^{0,1}_{\epsilon(\mu\oplus\nu)}h) \circ \Phi_1^{\epsilon\mu}$ . From this it follows that

**Lemma 5.1.** In the coordinates around (X, E) given by Theorem 1.1 we have that

$$\frac{d}{d\epsilon}|_{\epsilon=0}g^{VB}_{\epsilon(\mu\oplus\nu)}(\mu_1\oplus\nu_1,\mu_2\oplus\nu_2) = i\int_X tr((\bar{\mu}_2\nu_1)\wedge\nu)$$
$$\frac{d}{d\bar{\epsilon}}|_{\epsilon=0}g^{VB}_{\epsilon(\mu\oplus\nu)}(\mu_1\oplus\nu_1,\mu_2\oplus\nu_2) = i\int_X tr((\mu_1\bar{\nu}^T)\wedge\bar{\nu}_2^T).$$

*Proof.* We calculate each term gathering the terms like (5.4). We have already seen that  $\frac{d}{d\epsilon}|_{\epsilon=0}\overline{\Phi_2^{\epsilon(\mu\oplus\nu)}}^T \Phi_2^{\epsilon(\mu\oplus\nu)} = 0$ , and so these terms don't contribute. Now we consider the operators, where we have left out subscripts from the calculation as it should be clear where they live. The derivative of the projection is a sum of terms starting with an operator ending with  $\bar{\partial}^*$  and ones which starts with  $\bar{\partial}$  as is seen from the following calculation

$$\begin{split} \frac{d}{d\epsilon}|_{\epsilon=0} \mathrm{Ad}(\Phi_{2}^{\epsilon(\mu\oplus\nu)})^{-1}(\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*}P^{0,1}(\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*}^{-1}\mathrm{Ad}(\Phi_{2}^{\epsilon(\mu\oplus\nu)}) \\ &= \frac{d}{d\epsilon}|_{\epsilon=0} \mathrm{Ad}(\Phi_{2}^{\epsilon(\mu\oplus\nu)})^{-1}(\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*}(-\bar{\partial}\Delta_{0}^{-1}\bar{\partial}^{*})(\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*}^{-1}\mathrm{Ad}(\Phi_{2}^{\epsilon(\mu\oplus\nu)}) \\ &= \frac{d}{d\epsilon}|_{\epsilon=0} \mathrm{Ad}(\Phi_{2}^{\epsilon(\mu\oplus\nu)})^{-1}(\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*}(-\bar{\partial})(\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*}^{-1}\mathrm{Ad}(\Phi_{2}^{\epsilon(\mu\oplus\nu)}) \\ &\quad \mathrm{Ad}(\Phi_{2}^{\epsilon(\mu\oplus\nu)})^{-1}(\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*}(\bar{\partial}^{*})(\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*}^{-1}\mathrm{Ad}(\Phi_{2}^{\epsilon(\mu\oplus\nu)}) \\ &\quad \mathrm{Ad}(\Phi_{2}^{\epsilon(\mu\oplus\nu)})^{-1}(\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*}(\bar{\partial}^{*})(\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*}^{-1}\mathrm{Ad}(\Phi_{2}^{\epsilon(\mu\oplus\nu)}) \\ &= \frac{d}{d\epsilon}|_{\epsilon=0} \mathrm{Ad}(\Phi_{2}^{\epsilon(\mu\oplus\nu)})^{-1}(\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*}(-\bar{\partial})(\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*}^{-1}\mathrm{Ad}(\Phi_{2}^{\epsilon(\mu\oplus\nu)})\Delta_{0}^{-1}\bar{\partial}^{*} \\ &\quad + \bar{\partial}\frac{d}{d\epsilon}|_{\epsilon=0} \mathrm{Ad}(\Phi_{2}^{\epsilon(\mu\oplus\nu)})^{-1}(\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*}(\bar{\partial}_{0}^{-1})(\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*}^{-1}\mathrm{Ad}(\Phi_{2}^{\epsilon(\mu\oplus\nu)})\bar{\partial}^{*} \\ &\quad + \bar{\partial}\Delta_{0}^{-1}\frac{d}{d\epsilon}|_{\epsilon=0} \mathrm{Ad}(\Phi_{2}^{\epsilon(\mu\oplus\nu)})^{-1}(\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*}(\bar{\partial}^{*})(\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*}^{-1}\mathrm{Ad}(\Phi_{2}^{\epsilon(\mu\oplus\nu)})\bar{\partial}^{*} \\ &\quad + \bar{\partial}\Delta_{0}^{-1}\frac{d}{d\epsilon}|_{\epsilon=0} \mathrm{Ad}(\Phi_{2}^{\epsilon(\mu\oplus\nu)})^{-1}(\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*}(\bar{\partial}^{*})(\Phi_{1}^{\epsilon(\mu\oplus\nu)})_{*}^{-1}\mathrm{Ad}(\Phi_{2}^{\epsilon(\mu\oplus\nu)})\bar{\partial}^{*} \end{split}$$

The first two terms are orthogonal to  $\nu \in \mathcal{H}^{0,1}(X, \operatorname{End} E)$ , and the second one applied to a harmonic from is 0. This means the contribution form the first term in (5.3) is 0.

Now all the remaining terms contain a  $\partial \Phi_2^{\epsilon(\mu\oplus\nu)}$  which is 0 at  $\epsilon = 0$ . Hence the only contributions to the derivative arise when we derive these, and then we have that  $\frac{d}{d\epsilon}\partial \Phi_2^{\epsilon(\mu\oplus\nu)} = -\bar{\nu}^T$  and  $\frac{d}{d\epsilon}\partial \Phi_2^{\epsilon(\mu\oplus\nu)} = 0$ . Inserting this and setting  $\epsilon = 0$ we find the formulas in the lemma.

We proceed to calculate the second order derivatives of the metric. To do so, we need to calculate  $\frac{d^2}{d\epsilon_1 d\epsilon_2}|_{\epsilon=0} (\overline{\Phi_2^{\epsilon(\mu\oplus\nu)}})^T \Phi_2^{\epsilon(\mu\oplus\nu)}, \frac{d^2}{d\epsilon_1 d\epsilon_2}|_{\epsilon=0} (\Phi_2^{\epsilon(\mu\oplus\nu)})^{-1} \partial \Phi_2^{\epsilon(\mu\oplus\nu)}$  and the contribution from  $\frac{d^2}{d\epsilon_1 d\epsilon_2}|_{\epsilon=0} \operatorname{Ad}(\Phi_2^{\epsilon(\mu\oplus\nu)})^{-1} P^{0,1} \operatorname{Ad} \Phi_2^{\epsilon(\mu\oplus\nu)}$ . For the last term, we only need it when applied to a harmonic form and also it should not be orthogonal to a harmonic form.

We now calculate the three terms. For the first term, we begin by applying the Laplace operator on  ${\bf H}$  to the expression.

$$\begin{split} &\Delta \frac{d^2}{d\epsilon_1 d\bar{\epsilon}_2}|_{\epsilon=0} (\overline{\Phi_2^{\epsilon(\mu \oplus \nu)}})^T \Phi_2^{\epsilon(\mu \oplus \nu)} \\ &= y^2 \bar{\partial} \partial \frac{d^2}{d\epsilon_1 d\bar{\epsilon}_2}|_{\epsilon=0} (\overline{\Phi_2^{\epsilon(\mu \oplus \nu)}})^T \Phi_2^{\epsilon(\mu \oplus \nu)} \\ &= \frac{d^2}{d\epsilon_1 d\bar{\epsilon}_2}|_{\epsilon=0} y^2 \bar{\partial} \partial ((\overline{\Phi_+^{\epsilon(\mu \oplus \nu)}} \overline{\Phi_-^{\epsilon(\mu \oplus \nu)}})^T \Phi_+^{\epsilon(\mu \oplus \nu)} \Phi_-^{\epsilon\mu}) \\ &= \frac{d^2}{d\epsilon_1 d\bar{\epsilon}_2}|_{\epsilon=0} y^2 (\overline{\partial} \overline{\Phi_+^{\epsilon\mu}}^T \overline{\partial} \overline{\Phi_+^{\epsilon(\mu \oplus \nu)}}^T \Phi_+^{\epsilon(\mu \oplus \nu)} \Phi_-^{\epsilon\mu} + \overline{\partial} \overline{\Phi_-^{\epsilon\mu}}^T \overline{\Phi_+^{\epsilon(\mu \oplus \nu)}}^T \bar{\partial} \Phi_+^{\epsilon(\mu \oplus \nu)} \Phi_-^{\epsilon\mu} \\ &\quad + \overline{\partial} \overline{\Phi_-^{\epsilon\mu}}^T \overline{\Phi_+^{\epsilon(\mu \oplus \nu)}}^T \Phi_+^{\epsilon(\mu \oplus \nu)} \bar{\partial} \Phi_-^{\epsilon\mu} + \overline{\Phi_-^{\epsilon\mu}}^T \overline{\partial} \overline{\Phi_+^{\epsilon(\mu \oplus \nu)}}^T \bar{\partial} \partial \Phi_+^{\epsilon(\mu \oplus \nu)} \Phi_-^{\epsilon\mu} \\ &\quad + \overline{\Phi_-^{\epsilon\mu}}^T \overline{\Phi_+^{\epsilon(\mu \oplus \nu)}}^T \partial \Phi_+^{\epsilon(\mu \oplus \nu)} \bar{\partial} \Phi_-^{\epsilon\mu} + \overline{\Phi_-^{\epsilon\mu}}^T \overline{\partial} \overline{\Phi_+^{\epsilon(\mu \oplus \nu)}}^T \bar{\partial} \Phi_+^{\epsilon(\mu \oplus \nu)} \bar{\partial} \Phi_-^{\epsilon\mu} \\ &\quad + \overline{\Phi_-^{\epsilon\mu}}^T \overline{\partial} \overline{\partial} \Phi_+^{\epsilon(\mu \oplus \nu)}^T \Phi_+^{\epsilon(\mu \oplus \nu)} \Phi_-^{\epsilon\mu} + \overline{\Phi_-^{\epsilon\mu}}^T \overline{\partial} \overline{\Phi_+^{\epsilon(\mu \oplus \nu)}}^T \Phi_-^{\epsilon(\mu \oplus \nu)} \bar{\partial} \Phi_-^{\epsilon\mu} ). \end{split}$$

For all the terms where two different factors are differentiated we are only able to match the  $\epsilon$ -derivatives in one way that is nonzero. We also have that  $\bar{\partial}\partial\Phi_{+}^{\epsilon(\mu\oplus\nu)} = \partial\epsilon\mu\partial\Phi_{+}^{\epsilon(\mu\oplus\nu)}$ , and so we need to derive it with respect to  $\epsilon$  and  $\bar{\epsilon}$  to get a nonzero contribution. For the same reason  $\bar{\partial}\Phi_{+}^{\epsilon(\mu\oplus\nu)}$  needs to be differentiated twice to be nonzero. Since  $\bar{\partial}\Phi_{+}^{\epsilon(\mu\oplus\nu)}$  is always paired with another term, we need to differentiate these terms and hence they will not contribute, thus we get that

$$\begin{split} \Delta_0 \frac{d^2}{d\epsilon_1 d\bar{\epsilon}_2} |_{\epsilon=0} (\overline{\Phi_2^{\epsilon(\mu \oplus \nu)}})^T \Phi_2^{\epsilon(\mu \oplus \nu)} &= y^2 (\overline{(\nu_2)}^T \overline{(-\bar{\nu}_1^T)}^T + 0 + \overline{(\nu_2)}^T \nu_1 + \overline{(-\bar{\nu}_1)}^T \\ &\cdot (-\bar{\nu}_2^T) + (-\bar{\nu}_2)^T \nu_1 - \mu_1 \bar{\nu}_2^T - \bar{\mu}_2 \nu_1 + 0) \\ &= y^2 ([\nu_1, \bar{\nu}_2^T] - \partial \mu_1 \bar{\nu}_2^T - \partial \bar{\mu}_2 \nu_1). \end{split}$$

We conclude that

$$\frac{d^2}{d\epsilon_1 d\bar{\epsilon}_2}|_{\epsilon=0} (\overline{\Phi_2^{\epsilon(\mu\oplus\nu)}})^T \Phi_2^{\epsilon(\mu\oplus\nu)} = \Delta_0^{-1} ((-\star) \mathrm{ad}\nu_2 \star \nu_1 - \star \partial \mu_1 \bar{\nu}_2^T - \star \bar{\partial} \bar{\mu}_2 \nu_1) + cI$$

for some constant c. Since the kernel of  $\Delta_0$  is the constant multiples of I. In what remains this term will not contribute, as we will be looking at  $\operatorname{ad}(\frac{d^2}{d\epsilon_1 d\bar{\epsilon}_2}|_{\epsilon=0}(\overline{\Phi_2^{\epsilon(\mu\oplus\nu)}})^T \Phi_2^{\epsilon(\mu\oplus\nu)}).$ 

Next we calculate the second term  $\frac{d^2}{d\epsilon_1 d\epsilon_2}|_{\epsilon=0} (\Phi_2^{\epsilon(\mu \oplus \nu)})^{-1} \partial \Phi_2^{\epsilon(\mu \oplus \nu)}$ . The calculation tion follows directly from the previous computation.

$$\begin{split} \bar{\partial}\Delta_0^{-1}((-\star)\mathrm{ad}\nu_2\star\nu_1-\star\mu_1\bar{\nu}_2^T-\star\bar{\mu}_2\nu_1) &= \bar{\partial}\frac{d^2}{d\epsilon_1d\bar{\epsilon}_2}|_{\epsilon=0}(\overline{\Phi_2^{\epsilon(\mu\oplus\nu)}})^T\Phi_2^{\epsilon(\mu\oplus\nu)}\\ &= \frac{d^2}{d\epsilon_1d\bar{\epsilon}_2}|_{\epsilon=0}(\overline{\Phi_2^{\epsilon(\mu\oplus\nu)}})^T\bar{\partial}\Phi_2^{\epsilon(\mu\oplus\nu)} + \frac{d^2}{d\epsilon_1d\bar{\epsilon}_2}|_{\epsilon=0}(\overline{\partial\Phi_2^{\epsilon(\mu\oplus\nu)}})^T\Phi_2^{\epsilon(\mu\oplus\nu)}\\ &= \frac{d^2}{d\epsilon_1d\bar{\epsilon}_2}|_{\epsilon=0}(\overline{\Phi_2^{\epsilon(\mu\oplus\nu)}})^T(\Phi_2^{\epsilon(\mu\oplus\nu)}\epsilon\nu + \epsilon\mu\partial\Phi_2^{\epsilon(\mu\oplus\nu)})\\ &\quad + \frac{d^2}{d\epsilon_1d\bar{\epsilon}_2}|_{\epsilon=0}(\overline{(\Phi_2^{\epsilon(\mu\oplus\nu)})^{-1}\partial\Phi_2^{\epsilon(\mu\oplus\nu)}})^T\overline{\Phi_2^{\epsilon(\mu\oplus\nu)}}^T\Phi_2^{\epsilon(\mu\oplus\nu)}\\ &= -\mu_1\bar{\nu}_2^T + \frac{d^2}{d\epsilon_1d\bar{\epsilon}_2}|_{\epsilon=0}(\overline{(\Phi_2^{\epsilon(\mu\oplus\nu)})^{-1}\partial\Phi_2^{\epsilon(\mu\oplus\nu)}})^T, \end{split}$$

since  $\frac{d}{d\epsilon}|_{\epsilon=0} (\overline{\Phi_2^{\epsilon(\mu\oplus\nu)}})^T \Phi_2^{\epsilon(\mu\oplus\nu)} = 0.$ Finally we need to calculate the third term

$$\frac{d^2}{d\epsilon_1 d\bar{\epsilon}_2}|_{\epsilon=0} \operatorname{Ad}(\Phi_2^{\epsilon(\mu \oplus \nu)})^{-1} (\Phi_1^{\epsilon\mu})_* P^{0,1} (\Phi_1^{\epsilon\mu})_*^{-1} \operatorname{Ad}\Phi_2^{\epsilon(\mu \oplus \nu)},$$

but only where both  $\bar{\partial}$  and  $\bar{\partial}^*$  in  $\bar{\partial}\Delta_0^{-1}\bar{\partial}^*$  has been differentiated. This is simplified by the fact that  $\bar{\partial}$  only depending on  $\epsilon$  and not  $\bar{\epsilon}$  (see (5.1)). Using this and (5.2) we have that

$$\begin{aligned} \frac{d}{d\epsilon_1}|_{\epsilon=0} \operatorname{Ad}(\Phi_2^{\epsilon(\mu\oplus\nu)})^{-1}(\Phi_1^{\epsilon\mu})_* \bar{\partial}(\Phi_1^{\epsilon\mu})_*^{-1} \operatorname{Ad}\Phi_2^{\epsilon(\mu\oplus\nu)} \Delta_0^{-1} \\ \frac{d}{d\bar{\epsilon}_2}|_{\epsilon=0} \operatorname{Ad}(\Phi_2^{\epsilon(\mu\oplus\nu)})^{-1}(\Phi_1^{\epsilon\mu})_* \bar{\partial}^*(\Phi_1^{\epsilon\mu})_*^{-1} \operatorname{Ad}\Phi_2^{\epsilon(\mu\oplus\nu)} \\ &= (-\mu_1\partial + \operatorname{ad}\nu_1) \Delta_0^{-1} (\partial^*\bar{\mu}_2 - \operatorname{\starad}\nu_2 \star). \end{aligned}$$

Now we are ready to prove that

**Theorem 5.2.** Consider the second variation of the metric in the coordinates on the universal moduli space of pairs of a Riemann surface and a holomorphic bundle on it. Then we have this second variation at the center is

$$\frac{d^2}{d\epsilon_1 d\bar{\epsilon}_2}\Big|_{\epsilon=0} g_{\epsilon(\mu\oplus\nu)}\mu_3 \oplus \nu_3, \mu_4 \oplus \nu_4) = \\ -i\int_{\Sigma} tr((-\mu_1\partial + ad\nu_1)\Delta_0^{-1}(\partial^*\bar{\mu}_2 - \star ad\nu_2\star)\nu_3 \wedge \bar{\nu}_4^T$$

$$\begin{split} &-i\int_{\Sigma} tr(ad\Delta_{0}^{-1}((-\star)ad\nu_{2}\star\nu_{1}-\star(\partial\mu_{1}\bar{\nu}_{2}^{T})-\star(\bar{\partial}\bar{\mu}_{2}\nu_{1}))\nu_{3}\wedge\bar{\nu}_{4}^{T})\\ &+i\int_{\Sigma}\mu_{1}\bar{\mu}_{2}tr\nu_{3}\wedge\bar{\nu}_{4}^{T}\\ &-i\int_{\Sigma}tr(ad\nu_{1}+\mu_{1}\partial)\Delta_{o}^{-1}\bar{\partial}^{*}\mu_{3}\bar{\nu}_{2}^{T}\wedge\bar{\nu}_{4}^{T}\\ &-i\int_{\Sigma}tr\mu_{3}(\partial\Delta_{0}^{-1}(\star[\star\nu_{1}\nu_{2}]-\star(\partial\mu_{1}\bar{\nu}_{2}^{T})-\star(\bar{\partial}\bar{\mu}_{2}\nu_{1}))\wedge\bar{\nu}_{4}^{T}\\ &-i\int_{\Sigma}tr\bar{\mu}_{2}\mu_{3}\nu_{1}\wedge\bar{\nu}_{4}^{T}\\ &-i\int_{\Sigma}tr\bar{\partial}\Delta_{o}^{-1}(-\star ad\nu_{2}\star+\partial^{*}\mu_{2})\nu_{3}\wedge\bar{\mu}_{4}\nu_{1}\\ &-i\int_{\Sigma}tr\nu_{3}\wedge\overline{\mu_{4}}(\partial\Delta_{0}^{-1}(\star[\star\nu_{2}\nu_{1}]-\star(\partial\mu_{2}\bar{\nu}_{1}^{T})-\star(\bar{\partial}\bar{\mu}_{1}\nu_{2}))^{T}\\ &-i\int_{\Sigma}tr\nu_{3}\wedge\overline{\mu_{1}}\mu_{4}\nu_{2}^{T}-i\int_{\Sigma}tr\mu_{3}\nu_{1}\wedge\overline{\mu_{4}}\nu_{2}^{T}.\end{split}$$

Proof. Since we already have computed all the ingredients, we gather the results here.

$$\begin{split} -i\frac{d^2}{d\epsilon_1d\bar{\epsilon}_2} vert_{\epsilon=0} & \int_{\Sigma} \operatorname{tr} P^{0,1}(\Phi_1^{\epsilon(\mu\oplus\nu)})_*^{-1} \operatorname{Ad}(\Phi_2^{\epsilon(\mu\oplus\nu)})\nu_3 \\ & \wedge \overline{P^{0,1}((\Phi_1^{\epsilon(\mu\oplus\nu)})_*^{-1} \operatorname{Ad}\Phi_2^{\epsilon(\mu\oplus\nu)})\nu_4}^T \\ &= -i\frac{d^2}{d\epsilon_1d\bar{\epsilon}_2}|_{\epsilon=0} \int_{\Sigma} \operatorname{tr} \operatorname{Ad}(\overline{\Phi_2^{\epsilon(\mu\oplus\nu)}}^T \Phi_2^{\epsilon(\mu\oplus\nu)})(1 - |\epsilon\mu|^2))\operatorname{Ad}(\Phi_2^{\epsilon(\mu\oplus\nu)})^{-1} \\ & (\Phi_1^{\epsilon(\mu\oplus\nu)})_* P^{0,1}(\Phi_1^{\epsilon(\mu\oplus\nu)})_*^{-1} \operatorname{Ad}(\Phi_2^{\epsilon(\mu\oplus\nu)})\nu_3 \wedge \overline{\nu_4}^T \\ &= -i\int_{\Sigma} \operatorname{tr} \frac{d^2}{d\epsilon_1d\bar{\epsilon}_2}|_{\epsilon=0} \operatorname{Ad}(\overline{\Phi_2^{\epsilon(\mu\oplus\nu)}}^T \Phi_2^{\epsilon(\mu\oplus\nu)})\nu_3 \wedge \overline{\nu_4}^T \\ &- i\int_{\Sigma} \operatorname{tr} \frac{d^2}{d\epsilon_1d\bar{\epsilon}_2}|_{\epsilon=0} \operatorname{Ad}(\Phi_2^{\epsilon(\mu\oplus\nu)})^{-1}(\Phi_1^{\epsilon(\mu\oplus\nu)})_* \\ & P^{0,1}(\Phi_1^{\epsilon(\mu\oplus\nu)})_*^{-1} \operatorname{Ad}(\Phi_2^{\epsilon(\mu\oplus\nu)})\nu_3 \wedge \overline{\nu_4}^T \\ &- i\int_{\Sigma} \frac{d^2}{d\epsilon_1d\bar{\epsilon}_2}|_{\epsilon=0}(1 - |\epsilon\mu|^2)\operatorname{tr}\nu_3 \wedge \overline{\nu_4}^T \\ &= -i\int_{\Sigma} \operatorname{tr}(\operatorname{ad}(\Delta_0^{-1}((-\star)\operatorname{ad}\nu_2 \star \nu_1 - \star(\partial\mu_1\bar{\nu}_2^T) - \star(\bar{\partial}\bar{\mu}_2\nu_1)))\nu_3 \wedge \bar{\nu_4}^T \\ &- i\int_{\Sigma} \operatorname{tr}((-\mu_1\partial + \operatorname{ad}\nu_1)\Delta_0^{-1}(\partial^*\bar{\mu}_2 - \star\operatorname{ad}\nu_2\star)\nu_3 \wedge \overline{\nu_4}^T \\ &+ i\int_{\Sigma} \mu_1\bar{\mu}_2\operatorname{tr}\nu_3 \wedge \bar{\nu_4}^T. \end{split}$$
Now for the second term we have that

And similarly

$$\begin{split} -i \int_{\Sigma} \mathrm{tr} P^{0,1} (\Phi_{1}^{\epsilon(\mu \oplus \nu)})_{*}^{-1} \mathrm{Ad} (\Phi_{2}^{\epsilon(\mu \oplus \nu)}) \nu_{3} \\ & \wedge \overline{P^{0,1} (\Phi_{1}^{\epsilon(\mu \oplus \nu)})_{*}^{-1} (\mathrm{Ad} \Phi_{2}^{\epsilon(\mu \oplus \nu)}) \mu_{4} (\Phi_{2}^{\epsilon(\mu \oplus \nu)}) \partial \Phi_{2}^{\epsilon(\mu \oplus \nu)}}^{T} \\ &= -i \int_{\Sigma} \mathrm{tr} \nu_{3} \wedge \overline{P^{0,1} \mu_{4} (\partial \Delta_{0}^{-1} ((-\star) \mathrm{ad} \nu_{1} \star \nu_{2} - \star (\bar{\partial} \bar{\mu}_{1} \nu_{2}) - \star (\partial \mu_{2} \bar{\nu}_{1}^{T})) + \bar{\mu}_{1} \nu_{2})}^{T} \\ & - i \int_{\Sigma} \mathrm{tr} \bar{\partial} \Delta_{0}^{-1} (-\partial^{*} \mu_{2} - \star \mathrm{ad} \nu_{2} \star) \nu_{3} \wedge \overline{\mu_{4}} \nu_{1}. \end{split}$$

Finally there is not much choice in how to differentiate the following term

$$-i\frac{d^2}{d\epsilon_1 d\bar{\epsilon}_2}|_{\epsilon=0} \int_{\Sigma} \operatorname{tr} P^{0,1}(\Phi_1^{\epsilon(\mu\oplus\nu)})_*^{-1} \operatorname{Ad}(\Phi_2^{\epsilon(\mu\oplus\nu)})\mu_3(\Phi_2^{\epsilon(\mu\oplus\nu)})^{-1}\partial\Phi_2^{\epsilon(\mu\oplus\nu)})$$
  
 
$$\wedge \overline{P^{0,1}((\Phi_1^{\epsilon(\mu\oplus\nu)})_*^{-1} \operatorname{Ad}\Phi_2^{\epsilon(\mu\oplus\nu)})\mu_4(\Phi_2^{\epsilon(\mu\oplus\nu)})^{-1}\partial\Phi_2^{\epsilon(\mu\oplus\nu)}}^T$$
  
$$= -i\int_{\Sigma} \operatorname{tr} P^{0,1}(\mu_3 \frac{d}{d\bar{\epsilon}_2}|_{\epsilon=0} \partial\Phi_2^{\epsilon(\mu\oplus\nu)} \wedge \overline{P^{0,1}\mu_4 \frac{d}{d\bar{\epsilon}_1}}|_{\epsilon=0} \partial\Phi_2^{\epsilon(\mu\oplus\nu)} \overset{T}{=} -i\int_{\Sigma} \operatorname{tr} \mu_3 \bar{\nu}_2^T \wedge \bar{\mu}_4 \nu_1.$$

Collect all these results and we have the conclusion.

#### 5.2 The Variation of the Metric in Fibered Coordinates

Now for the fibered coordinates we can do the same computations. From the calculation of the Kodaira-Spencer map (Proposition 4.3) we know, that the metric

in the moduli space of bundles direction is given by

$$\begin{split} g_{VB}^{\psi^{\mu}}(\mu_{1} \oplus \nu_{1}, \mu_{2} \oplus \nu_{2}) &= -i \int_{\Sigma} P_{e\nu^{e\mu}}^{0,1} \mathrm{Adf}^{e\nu^{e\mu}} \nu_{1}^{e\mu} \wedge \overline{P_{e\nu^{e\mu}}^{0,1} \mathrm{Adf}^{e\nu^{e\mu}} \nu_{2}^{e\mu}}^{T} \\ &- i \int_{\Sigma} P_{e\nu^{e\mu}}^{0,1} \mathrm{Adf}^{e\nu^{e\mu}} \nu_{1}^{e\mu} \wedge \overline{P_{e\nu^{e\mu}}^{0,1} \mathrm{Ad(f}^{e\nu^{e\mu}})} (\mu_{2}^{e\mu}(f^{e\nu^{e\mu}})^{-1} \partial f^{e\nu^{e\mu}})^{T} \\ &- i \int_{\Sigma} P_{e\nu^{e\mu}}^{0,1} \mathrm{Adf}^{e\nu^{e\mu}} \nu_{1}^{e\mu} \\ &\overline{P_{e\nu^{e\mu}}^{0,1} \mathrm{Ad(f}^{e\nu^{e\mu}})} ((\Phi_{1}^{e\mu})_{*}^{-1}(1 - |\epsilon\mu|^{2}) \frac{d}{dt}|_{t=0} (\Phi_{1}^{e\mu^{+\mu}\mu_{2}}) P^{0,1}(\Phi_{1}^{e\mu^{+\mu}\mu_{2}})^{-1} \nu)^{T} \\ &- i \int_{\Sigma} P_{e\nu^{e\mu}}^{0,1} \mathrm{Ad(f}^{e\nu^{e\mu}}) (\mu_{1}^{e\mu}(f^{e\nu^{e\mu}})^{-1} \partial f^{e\nu^{e\mu}}) \wedge \overline{P_{e\nu^{e\mu}}^{0,1} \mathrm{Adf}^{e\nu^{e\mu}}} \nu_{2}^{e\mu}^{T} \\ &- i \int_{\Sigma} P_{e\nu^{e\mu}}^{0,1} \mathrm{Ad(f}^{e\nu^{e\mu}}) (\mu_{1}^{e\mu}(f^{e\nu^{e\mu}})^{-1} \partial f^{e\nu^{e\mu}}) \wedge \overline{P_{e\nu^{e\mu}}^{0,1} \mathrm{Adf}^{e\nu^{e\mu}}} \nu_{2}^{e\mu}^{T} \\ &- i \int_{\Sigma} P_{e\nu^{e\mu}}^{0,1} \mathrm{Ad(f}^{e\nu^{e\mu}}) (\mu_{1}^{e\mu}(f^{e\nu^{e\mu}})^{-1} \partial f^{e\nu^{e\mu}}) \wedge \overline{P_{e\nu^{e\mu}}^{0,1} \mathrm{Ad(f}^{e\nu^{e\mu}})} (\mu_{2}^{e\mu}(f^{e\nu^{e\mu}})^{-1} \partial f^{e\nu^{e\mu}})^{T} \\ &- i \int_{\Sigma} P_{e\nu^{e\mu}}^{0,1} \mathrm{Ad(f}^{e\nu^{e\mu}}) ((\Phi_{1}^{e\mu})_{*}^{-1}(1 - |\epsilon\mu|^{2}) \frac{d}{dt}|_{t=0} (\Phi_{1}^{e\mu^{+\mu\mu}}) P^{0,1}(\Phi_{1}^{e\mu^{+\mu\mu}})^{-1} \nu) \\ &\wedge \overline{P_{e\nu^{e\mu}}^{0,1} \mathrm{Ad(f}^{e\nu^{e\mu}})} ((\Phi_{1}^{e\mu})_{*}^{-1}(1 - |\epsilon\mu|^{2}) \frac{d}{dt}|_{t=0} (\Phi_{1}^{e\mu^{+\mu\mu}}) P^{0,1}(\Phi_{1}^{e\mu^{+\mu\mu}})^{-1} \nu) \\ &\wedge \overline{P_{e\nu^{e\mu}}^{0,1} \mathrm{Ad(f}^{e\nu^{e\mu}})} ((\Phi_{1}^{e\mu})_{*}^{-1}(1 - |\epsilon\mu|^{2}) \frac{d}{dt}|_{t=0} (\Phi_{1}^{e\mu^{+\mu\mu}}) P^{0,1}(\Phi_{1}^{e\mu^{+\mu\mu}})^{-1} \nu) \\ &\wedge \overline{P_{e\nu^{e\mu}}^{0,1} \mathrm{Ad(f}^{e\nu^{e\mu}})} ((\Phi_{1}^{e\mu})_{*}^{-1}(1 - |\epsilon\mu|^{2}) \frac{d}{dt}|_{t=0} (\Phi_{1}^{e\mu^{+\mu\mu}}) P^{0,1}(\Phi_{1}^{e\mu^{+\mu\mu}})^{-1} \nu) \\ &\wedge \overline{P_{e\nu^{e\mu}}^{0,1} \mathrm{Ad(f}^{e\nu^{e\mu}})} ((\Phi_{1}^{e\mu})_{*}^{-1}(1 - |\epsilon\mu|^{2}) \frac{d}{dt}|_{t=0} (\Phi_{1}^{e\mu^{+\mu\mu}}) P^{0,1}(\Phi_{1}^{e\mu^{+\mu\mu}})^{-1} \nu) \\ &\wedge \overline{P_{e\nu^{e\mu}}^{0,1} \mathrm{Ad(f}^{e\nu^{e\mu}})} ((\Phi_{1}^{e\mu})_{*}^{-1}(1 - |\epsilon\mu|^{2}) \frac{d}{dt}|_{t=0} (\Phi_{1}^{e\mu^{+\mu\mu}}) P^{0,1}(\Phi_{1}^{e\mu^{+\mu\mu}})^{-1} \nu) \\ &\wedge \overline{P_{e\nu^{e\mu}}^{0,1} \mathrm{Ad(f}^{e\nu^{e\mu}})} ((\Phi_{1}^{e\mu})_{*}^{-1}(1 - |\epsilon\mu|^{2}) \frac{d}{dt}|_{t=0} (\Phi_{1}^{e\mu^{+\mu\mu}})^$$

While these nine terms look intimidating, we can discard three of the terms, because  $P^{0,1}_{\epsilon\nu^{\epsilon\mu}} \operatorname{Ad}(f^{\epsilon\nu^{\epsilon\mu}})^{-1}(1-|\epsilon\mu|^2) \frac{d}{dt}|_{t=0} (\Phi_1^{\epsilon\mu+t\mu_2}) P^{0,1}(\Phi_1^{\epsilon\mu+t\mu_2})^{-1}\nu)$  vanishes to second-order and  $P^{0,1}_{\epsilon\nu^{\epsilon\mu}} \operatorname{Ad}(f^{\epsilon\nu^{\epsilon\mu}})(\mu_2^{\epsilon\mu}(f^{\epsilon\nu^{\epsilon\mu}})^{-1}\partial f^{\epsilon\nu^{\epsilon\mu}})$  vanishes to first-order, so terms containing both kind of factors or only the first kind of factors will vanish to higher order, than we are interested in. Now the first variation will be the same as in Section 5, but to calculate it we will have to work with slightly different expressions.

First we consider

$$\frac{d}{d\epsilon}|_{\epsilon=0}((\overline{f^{\epsilon\nu^{\epsilon}\mu}})^T f^{\epsilon\nu^{\epsilon}\mu}) \circ \Phi_1^{\epsilon\mu} = 0$$

and

$$\frac{d}{d\bar{\epsilon}}|_{\epsilon=0}((\overline{f^{\epsilon\nu^{\epsilon}\mu}})^T f^{\epsilon\nu^{\epsilon}\mu}) \circ \Phi_1^{\epsilon\mu} = 0$$

Both of these follow from the computations in [TZ1], where it was shown that  $\frac{d}{d\epsilon}|_{\epsilon=0}((\overline{f^{\epsilon\nu}})^T f^{\epsilon\nu}) = 0$ . Now composing with  $\nu \to \nu^{\epsilon\mu}$  won't change it, and if we differentiate  $\Phi_1^{\epsilon\mu}$  then we can set  $\epsilon = 0$  in the rest of the terms and calculate  $\frac{d}{d\epsilon}I \circ \Phi_1^{\epsilon\mu} = 0$ . Now for a projection, the first derivative will either have harmonic forms in it's kernel or the image is in the orthogonal complement, hence the only contributions are from the terms

$$\int_{\Sigma} P^{0,1}_{\epsilon\nu^{\epsilon\mu}} \mathrm{Ad} f^{\epsilon\nu^{\epsilon\mu}} \nu_1^{\epsilon\mu} \overline{P^{0,1}_{\epsilon\nu^{\epsilon\mu}} \mathrm{Ad}(f^{\epsilon\nu^{\epsilon\mu}})} (\mu_2^{\epsilon\mu} (f^{\epsilon\nu^{\epsilon\mu}})^{-1} \partial f^{\epsilon\nu^{\epsilon\mu}})^T$$

and

$$\int_{\Sigma} P^{0,1}_{\epsilon\nu^{\epsilon\mu}} \mathrm{Ad}(f^{\epsilon\nu^{\epsilon\mu}}) (\mu_1^{\epsilon\mu} (f^{\epsilon\nu^{\epsilon\mu}})^{-1} \partial f^{\epsilon\nu^{\epsilon\mu}}) \overline{P^{0,1}_{\epsilon\nu^{\epsilon\mu}} \mathrm{Ad} f^{\epsilon\nu^{\epsilon\mu}} \nu_2^{\epsilon\mu}}^T$$

And so we have, completely analogues to the previous section the following lemma.

**Lemma 5.3.** In the fibered coordinates around (X, E) we have that:

$$\begin{aligned} \frac{d}{d\epsilon}|_{\epsilon=0} g^{VB}_{\epsilon\nu^{\epsilon\mu}}(\mu_1 \oplus \nu_1, \mu_2 \oplus \nu_2) &= i \int_X \bar{\mu}_2 tr(\nu_1 \nu) \\ \frac{d}{d\bar{\epsilon}}|_{\epsilon=0} g^{VB}_{\epsilon\nu^{\epsilon\mu}}(\mu_1 \oplus \nu_1, \mu_2 \oplus \nu_2) &= i \int_X \mu_1 tr(\bar{\nu}^T \bar{\nu}_2^T) \end{aligned}$$

Now for the second variation of the metric we need to calculate the two terms

$$\frac{d^2}{d\epsilon_1 d\bar{\epsilon}_2}|_{\epsilon=0} ((\overline{f^{\epsilon\nu^{\epsilon\mu}}})^T f^{\epsilon\nu^{\epsilon\mu}}) \circ \Phi_1^{\epsilon\mu}$$

and

$$\frac{d^2}{d\epsilon_1 d\bar{\epsilon}_2}|_{\epsilon=0} ((f^{\epsilon\nu^{\epsilon\mu}})^{-1} \partial f^{\epsilon\nu^{\epsilon\mu}}) \circ \Phi_1^{\epsilon\mu}.$$

We calculate these the same way as we did with the previous set of coordinates.

$$\begin{split} \Delta_0 \frac{d^2}{d\epsilon_1 d\bar{\epsilon}_2} |_{\epsilon=0} ((\overline{f^{\epsilon\nu^{\epsilon\mu}}})^T f^{\epsilon\nu^{\epsilon\mu}}) \circ \Phi_1^{\epsilon\mu} &= y^2 \partial \bar{\partial} \frac{d^2}{d\epsilon_1 d\bar{\epsilon}_2} |_{\epsilon=0} ((\overline{f^{\epsilon\nu^{\epsilon\mu}}})^T f^{\epsilon\nu^{\epsilon\mu}}) \circ \Phi_1^{\epsilon\mu} \\ &= \frac{d^2}{d\epsilon_1 d\bar{\epsilon}_2} |_{\epsilon=0} y^2 \partial \bar{\partial} ((\overline{f^{\epsilon\nu^{\epsilon\mu}}})^T (\overline{f^{\epsilon\nu^{\epsilon\mu}}})^T f^{\epsilon\nu^{\epsilon\mu}}_+ f^{\epsilon\nu^{\epsilon\mu}}_-) \circ \Phi_1^{\epsilon\mu}. \end{split}$$

Now we use

$$\begin{split} \bar{\partial}\partial(h\circ\Phi_{1}^{\epsilon\mu}) &= \partial\Phi_{1}^{\epsilon\mu}\partial\bar{\Phi}_{1}^{\epsilon\mu}(\partial\partial h)\circ\Phi_{1}^{\epsilon\mu} + \bar{\partial}\bar{\Phi}_{1}^{\epsilon\mu}\partial\Phi_{1}^{\epsilon\mu}(\partial\bar{\partial}h)\circ\Phi_{1}^{\epsilon\mu} \\ &+ \bar{\partial}\Phi_{1}^{\epsilon\mu}\partial\bar{\Phi}_{1}^{\epsilon\mu}(\bar{\partial}\partial h)\circ\Phi_{1}^{\epsilon\mu} + \partial\bar{\Phi}_{1}^{\epsilon\mu}\bar{\partial}\bar{\Phi}_{1}^{\epsilon\mu}(\bar{\partial}\bar{\partial}h)\circ\Phi_{1}^{\epsilon\mu} \end{split}$$

Jørgen E. Andersen and Niccolo S. Poulsen

$$= (\partial \Phi_1^{\epsilon\mu} \epsilon \mu \bar{\partial} \bar{\Phi}_1^{\epsilon\mu} (\partial \partial h) \circ \Phi_1^{\epsilon\mu} + \bar{\partial} \bar{\Phi}_1^{\epsilon\mu} \partial \Phi_1^{\epsilon\mu} (\partial \bar{\partial} h) \circ \Phi_1^{\epsilon\mu} \\ + |\epsilon\mu|^2 \partial \Phi_1^{\epsilon\mu} \partial \bar{\Phi}_1^{\epsilon\mu} (\bar{\partial} \bar{\partial} h) \circ \Phi_1^{\epsilon\mu} + \bar{\epsilon} \mu \bar{\partial} \bar{\Phi}_1^{\epsilon\mu} \bar{\partial} \bar{\Phi}_1^{\epsilon\mu} (\bar{\partial} \bar{\partial} h) \circ \Phi_1^{\epsilon\mu}$$

for  $h = (\overline{f^{\epsilon\nu^{\epsilon\mu}}})^T f^{\epsilon\nu^{\epsilon\mu}}$ , and since we know that  $\partial f^{\epsilon\nu^{\epsilon\mu}}$  and  $\overline{\partial} f^{\epsilon\nu^{\epsilon\mu}}$  vanish to first-order in  $\epsilon$ , we only have the surviving terms

$$\Delta_0 \frac{d^2}{d\epsilon_1 d\bar{\epsilon}_2} |_{\epsilon=0} ((\overline{f^{\epsilon\nu^{\epsilon\mu}}})^T f^{\epsilon\nu^{\epsilon\mu}}) \circ \Phi_1^{\epsilon\mu} = y^2 (-\partial\mu_1 \bar{\nu}_2^T - \bar{\partial}\bar{\mu}_2 \nu_1 + [\nu_1, \bar{\nu}_2^T]),$$

which is exactly like in the previous case. We proceed on to calculate  $\frac{d^2}{d\epsilon_1 d\epsilon_2}|_{\epsilon=0}((f^{\epsilon\nu^{\epsilon\mu}})^{-1}\partial f^{\epsilon\nu^{\epsilon\mu}}) \circ \Phi_1^{\epsilon\mu}$ , and so we study

$$\begin{split} \bar{\partial}\Delta_0^{-1}y^2(-\mu_1\partial\bar{\nu}_2^T - \bar{\mu}_2\bar{\partial}\nu_1 + [\nu_1,\bar{\nu}_2^T]) &= \bar{\partial}\frac{d^2}{d\epsilon_1d\bar{\epsilon}_2}|_{\epsilon=0}((\overline{f^{\epsilon\nu^{\epsilon\mu}}})^T f^{\epsilon\nu^{\epsilon\mu}}) \circ \Phi_1^{\epsilon\mu} \\ &= \frac{d^2}{d\epsilon_1d\bar{\epsilon}_2}|_{\epsilon=0}(\bar{\partial}((\overline{f^{\epsilon\nu^{\epsilon\mu}}})^T f^{\epsilon\nu^{\epsilon\mu}})) \circ \Phi_1^{\epsilon\mu}\bar{\partial}\bar{\Phi}_1^{\epsilon\mu} \\ &+ \frac{d^2}{d\epsilon_1d\bar{\epsilon}_2}|_{\epsilon=0}(\partial((\overline{f^{\epsilon\nu^{\epsilon\mu}}})^T f^{\epsilon\nu^{\epsilon\mu}})) \circ \Phi_1^{\epsilon\mu}\bar{\partial}\Phi_1^{\epsilon\mu} \\ &= \frac{d^2}{d\epsilon_1d\bar{\epsilon}_2}|_{\epsilon=0}(((\overline{\partial}\overline{f^{\epsilon\nu^{\epsilon\mu}}})^T f^{\epsilon\nu^{\epsilon\mu}})) \circ \Phi_1^{\epsilon\mu}\bar{\partial}\Phi_1^{\epsilon\mu} \\ &+ \frac{d^2}{d\epsilon_1d\bar{\epsilon}_2}|_{\epsilon=0}(((\overline{f^{\epsilon\nu^{\epsilon\mu}}})^T f^{\epsilon\nu^{\epsilon\mu}})) \circ \Phi_1^{\epsilon\mu}\bar{\partial}\Phi_1^{\epsilon\mu} \\ &+ \frac{d^2}{d\epsilon_1d\bar{\epsilon}_2}|_{\epsilon=0}(((\overline{f^{\epsilon\nu^{\epsilon\mu}}})^T \bar{\partial}f^{\epsilon\nu^{\epsilon\mu}})) \circ \Phi_1^{\epsilon\mu}\bar{\partial}\Phi_1^{\epsilon\mu} \\ &+ \frac{d^2}{d\epsilon_1d\bar{\epsilon}_2}|_{\epsilon=0}(((\overline{f^{\epsilon\nu^{\epsilon\mu}}})^T \bar{\partial}f^{\epsilon\nu^{\epsilon\mu}})) \circ \Phi_1^{\epsilon\mu}\bar{\partial}\Phi_1^{\epsilon\mu} \end{split}$$

Here the second and fourth term cancel, as is seen by using that two of the factors vanish to first-order in  $\epsilon$ , which then give  $\mu_1 \bar{\nu}_2^T$  and  $-\mu_1 \bar{\nu}_2^T$  respectively.

$$\begin{split} \frac{d^2}{d\epsilon_1 d\bar{\epsilon}_2}|_{\epsilon=0} (((\overline{\partial} f^{\epsilon\nu^{\epsilon\mu}})^T f^{\epsilon\nu^{\epsilon\mu}})) \circ \Phi_1^{\epsilon\mu} \bar{\partial} \bar{\Phi}_1^{\epsilon\mu} \\ &+ \frac{d^2}{d\epsilon_1 d\bar{\epsilon}_2}|_{\epsilon=0} (((\overline{f^{\epsilon\nu^{\epsilon\mu}}})^T \bar{\partial} f^{\epsilon\nu^{\epsilon\mu}})) \circ \Phi_1^{\epsilon\mu} \bar{\partial} \bar{\Phi}_1^{\epsilon\mu} \\ &= \frac{d^2}{d\epsilon_1 d\bar{\epsilon}_2}|_{\epsilon=0} (((\overline{f^{\epsilon\nu^{\epsilon\mu}}})^{-1} \partial f^{\epsilon\nu^{\epsilon\mu}}^T f^{\epsilon\nu^{\epsilon\mu}})) \circ \Phi_1^{\epsilon\mu} \bar{\partial} \bar{\Phi}_1^{\epsilon\mu} \\ &+ \frac{d^2}{d\epsilon_1 d\bar{\epsilon}_2}|_{\epsilon=0} (((\overline{f^{\epsilon\nu^{\epsilon\mu}}})^T f^{\epsilon\nu^{\epsilon\mu}} \epsilon\nu^{\epsilon\mu})) \circ \Phi_1^{\epsilon\mu} \bar{\partial} \bar{\Phi}_1^{\epsilon\mu} \\ &= \frac{d^2}{d\epsilon_1 d\bar{\epsilon}_2}|_{\epsilon=0} (((\overline{f^{\epsilon\nu^{\epsilon\mu}}})^{-1} \partial f^{\epsilon\nu^{\epsilon\mu}}^T) \circ \Phi_1^{\epsilon\mu} \bar{\partial} \bar{\Phi}_1^{\epsilon\mu} \\ &+ \frac{d^2}{d\epsilon_1 d\bar{\epsilon}_2}|_{\epsilon=0} (\epsilon\nu^{\epsilon\mu})) \circ \Phi_1^{\epsilon\mu} \bar{\partial} \bar{\Phi}_1^{\epsilon\mu} \end{split}$$

Coordinates for the Universal Moduli Space of Holomorphic Vector Bundles 35

$$=\frac{d^2}{d\epsilon_1 d\bar{\epsilon}_2}|_{\epsilon=0}((\overline{(f^{\epsilon\nu^{\epsilon\mu}})^{-1}\partial f^{\epsilon\nu^{\epsilon\mu}}}^T)\circ\Phi_1^{\epsilon\mu}\bar{\partial}\bar{\Phi}_1^{\epsilon\mu}+\bar{\partial}\Delta_0^{-1}\partial^*\bar{\mu}_2\nu_1$$

Now this is different from the previous coordinates. We need however to consider two things more. The first is the second variation of the harmonic projection. Since the only relevant part is the contribution where both  $\bar{\partial}$  and  $\bar{\partial}^*$  have been differentiated in  $\bar{\partial}\Delta_0^{-1}\bar{\partial}^*$  and the coordinates agree to second-order nothing will have changed and we have it gives the following term

$$(-\mu_1\partial + \mathrm{ad}\nu_1)\Delta_0^{-1}(\partial^*\bar{\mu}_2 - \star \mathrm{ad}\nu_2\star).$$

The final term to consider is the new term in the formula for the metric, which is

$$\frac{d^2}{d\epsilon_1 d\bar{\epsilon}_2}|_{\epsilon=0} P^{0,1}_{\epsilon\nu^{\epsilon\mu}} \mathrm{Ad}(f^{\epsilon\nu^{\epsilon\mu}})((\Phi_1^{\epsilon\mu})^{-1}_*(1-|\epsilon\mu|^2)\frac{d}{dt}|_{t=0}(\Phi_1^{\epsilon\mu+t\bar{\mu}})P^{0,1}(\Phi_1^{\epsilon\mu+t\bar{\mu}})^{-1}\epsilon\nu),$$

as  $\frac{d}{dt}|_{t=0}(\Phi_1^{\epsilon\mu+t\tilde{\mu}})P^{0,1}(\Phi_1^{\epsilon\mu+t\tilde{\mu}})^{-1}\epsilon\nu)$  vanishes to second-order this has to be differentiated twice

$$\begin{split} \frac{d}{d\bar{\epsilon}_{2}}|_{\epsilon=0} \frac{d}{dt}|_{t=0} (\Phi_{1}^{\epsilon\mu+t\bar{\mu}})P^{0,1}(\Phi_{1}^{\epsilon\mu+t\bar{\mu}})^{-1}\nu_{1}) \\ &= -(\frac{d}{d\bar{\epsilon}_{2}}|_{\epsilon=0} \frac{d}{dt}|_{t=0} (\Phi_{1}^{\epsilon\mu+t\bar{\mu}})\bar{\partial}(\Phi_{1}^{\epsilon\mu+t\bar{\mu}})^{-1} \\ & \Delta_{0}^{-1} \frac{d}{d\bar{\epsilon}_{2}}|_{\epsilon=0} \Phi_{1}^{\epsilon\mu+t\bar{\mu}})\bar{\partial}^{*}(\Phi_{1}^{\epsilon\mu+t\bar{\mu}})^{-1}\nu_{1}) \\ &- \bar{\partial} \frac{d}{d\bar{\epsilon}_{2}}|_{\epsilon=0} \frac{d}{dt}|_{t=0} (\Phi_{1}^{\epsilon\mu+t\bar{\mu}})\Delta_{0}^{-1}\bar{\partial}^{*}(\Phi_{1}^{\epsilon\mu+t\bar{\mu}})^{-1})\nu_{1} \\ &= -\tilde{\mu}\partial\Delta_{0}^{-1}\partial^{*}\bar{\mu}_{2}\nu_{1} - \bar{\partial} \frac{d}{d\bar{\epsilon}_{2}}|_{\epsilon=0} \frac{d}{dt}|_{t=0} (\Phi_{1}^{\epsilon\mu+t\bar{\mu}})\Delta_{0}^{-1}\bar{\partial}^{*}(\Phi_{1}^{\epsilon\mu+t\bar{\mu}})^{-1}\nu_{1}. \end{split}$$

We are now ready to gather all the contributions in the following theorem.

**Theorem 5.4.** We have the following for the second variation of the metric at (X, E) in the fibered coordinates:

$$\begin{split} \frac{d^2}{d\epsilon_1 d\bar{\epsilon}_2}|_{\epsilon=0} g^{VB}_{\epsilon\nu^{\epsilon\mu}}(\mu_3 \oplus \nu_3, \mu_4 \oplus \nu_4) = \\ & \int_{\Sigma} tr((-\mu_1 \partial + ad\nu_1) \Delta_0^{-1}((-\star) ad\nu_2 \star + \partial^* \bar{\mu}_2)\nu_3) \wedge \bar{\nu}_4^T \\ & -i \int_{\Sigma} tr(ad\Delta_0^{-1}((-\star) ad\nu_2 \star \nu_1 - \star (\partial \mu_1 \bar{\nu}_2^T) - \star (\bar{\partial} \bar{\mu}_2 \nu_1))\nu_3 \wedge \bar{\nu}_4^T) \\ & -i \int_{\Sigma} tr(ad\nu_1 - \mu_1 \partial) \Delta_o^{-1} \bar{\partial}^* \mu_3 \bar{\nu}_2^T \wedge \bar{\nu}_4^T \\ & +i \int_{\Sigma} \mu_1 \bar{\mu}_2 tr \nu_3 \wedge \bar{\nu}_4^T \end{split}$$

$$\begin{split} &-i\int_{\Sigma} tr\mu_{3}(\partial\Delta_{0}^{-1}(\star[\star\nu_{1}\nu_{2}]-\star(\partial\mu_{1}\bar{\nu}_{2}^{T})-\star(\bar{\partial}\bar{\mu}_{2}\nu_{1}))\wedge\bar{\nu}_{4}^{T} \\ &+i\int_{\Sigma} tr\mu_{3}\partial\Delta_{0}^{-1}\bar{\partial}^{*}\mu_{1}\bar{\nu}_{2}^{T}\wedge\bar{\nu}_{4}^{T} \\ &-i\int_{\Sigma} tr\bar{\partial}\Delta_{o}^{-1}(-\star ad\nu_{2}\star+\partial^{*}\bar{\mu}_{2})\nu_{3}\wedge(-\bar{\mu}_{4}\nu_{1}) \\ &-i\int_{\Sigma} tr\nu_{3}\wedge\overline{\mu_{4}}(\partial\Delta_{0}^{-1}(\star[\star\nu_{2}\nu_{1}]-\star(\partial\mu_{2}\bar{\nu}_{1}^{T})-\star(\bar{\partial}\bar{\mu}_{1}\nu_{2}))^{T} \\ &+i\int_{\Sigma} tr\nu_{3}\wedge\overline{\mu_{4}}\partial\Delta_{0}^{-1}\bar{\partial}^{*}\mu_{2}\bar{\nu}_{1}^{T}^{T}-i\int_{\Sigma} tr\mu_{3}\nu_{1}\wedge\overline{\mu_{4}}\bar{\nu}_{2}^{T} \\ &+i\int_{\Sigma} \mu_{3}\partial\Delta_{0}^{-1}\partial^{*}\bar{\mu}_{2}\nu_{1}\wedge\bar{\nu}_{4}^{T}+i\int_{\Sigma} \nu_{3}\wedge\overline{\mu_{4}}\partial\Delta_{0}^{-1}\partial^{*}\bar{\mu}_{1}\nu_{2}^{T} \end{split}$$

Comparing this to the previous coordinates (Theorem 5.2) we see that there are four terms here which we didn't have before and two we no longer have. The new terms are

$$i \int_{\Sigma} \mu_3 \partial \Delta_0^{-1} \partial^* \bar{\mu}_2 \nu_1 \wedge \bar{\nu}_4^T,$$
  
$$i \int_{\Sigma} \nu_3 \wedge \overline{\mu_4 \partial \Delta_0^{-1} \partial^* \bar{\mu}_1 \nu_2}^T,$$
  
$$i \int_{\Sigma} \operatorname{tr} \mu_3 \partial \Delta_0^{-1} \bar{\partial}^* \mu_1 \bar{\nu}_2^T \wedge \overline{\nu}_4^T$$

and

$$i \int_{\Sigma} \mathrm{tr} \nu_3 \wedge \overline{\mu_4 \partial \Delta_0^{-1} \bar{\partial}^* \mu_2 \bar{\nu}_1^T}^T,$$

While the ones we no longer have are

$$-i\int_{\Sigma}\mathrm{tr}\nu_{3}\wedge\overline{\mu_{1}\overline{\mu}_{4}\nu_{2}}^{T}$$

and

$$-i\int_{\Sigma}\mathrm{tr}\bar{\mu}_{2}\mu_{3}\nu_{1}\wedge\overline{\nu_{4}}^{T}.$$

Now we have that for  $\nu_1 = \nu_4$  and  $\mu_2 = \mu_3$  and the rest 0 the difference between the two expressions are

$$-i\int_{\Sigma}\mu_{3}\partial\Delta_{0}^{-1}\partial^{*}\bar{\mu}_{2}\nu_{1}\wedge\bar{\nu}_{4}^{T}-i\int_{\Sigma}\mathrm{tr}\bar{\mu}_{2}\mu_{3}\nu_{1}\wedge\bar{\nu}_{4}^{T}$$
$$=-i\int_{\Sigma}\Delta_{0}^{-1}\partial^{*}\bar{\mu}_{2}\nu_{1}\wedge\overline{\partial^{*}\bar{\mu}_{2}\nu_{1}}^{T}\rho-i\int_{\Sigma}\mathrm{tr}\bar{\mu}_{2}\mu_{2}\nu_{1}\wedge\bar{\nu}_{1}^{T}$$

Since  $\Delta$  is a positive operator we have that  $-i \int_{\Sigma} \Delta_0^{-1} \partial^* \bar{\mu}_2 \nu_1 \wedge \overline{\partial^* \bar{\mu}_2 \nu_1}^T \rho \ge 0$  and for obvious reasons  $-i \int_{\Sigma} \operatorname{tr} \bar{\mu}_2 \mu_2 \nu_1 \wedge \overline{\nu_1}^T > 0$ , if  $\nu_1 \neq 0$  and  $\mu_2 \neq 0$ .

Coordinates for the Universal Moduli Space of Holomorphic Vector Bundles 37

**Theorem 5.5.** The coordinates of 1.1 and the Fibered coordinates agree to second order, but differ at third order in the derivatives at the center point.

# References

- [AhB] L. Ahlfors and L. Bers, Riemann's mapping theorem for variable metrics, Annals of Mathematics, 72(2):385–404, 1960.
- [A1] J.E. Andersen, Deformation Quantization and Geometric Quantization of Abelian Moduli Spaces., Comm. of Math. Phys. 255:727–745, 2005.
- [A2] J. E. Andersen. Asymptotic faithfulness of the quantum SU(n) representations of the mapping class groups. Annals of Mathematics. **163**:347–368, 2006.
- [AH] J.E. Andersen & S.K. Hansen. Asymptotics of the quantum invariants for surgeries on the figure 8 knot. *Journal of Knot theory and its Ramifications*, 15:479–548, 2006.
- [AGr1] J.E. Andersen & J. Grove. Automorphism Fixed Points in the Moduli Space of Semi-Stable Bundles. *The Quarterly Journal of Mathematics*, 57:1– 35, 2006.
- [AMU] J.E. Andersen, G. Masbaum & K. Ueno. Topological Quantum Field Theory and the Nielsen-Thurston classification of M(0,4). Math. Proc. Cambridge Philos. Soc. 141:477–488, 2006.
- [AU1] J. E. Andersen & K. Ueno. Abelian Conformal Field theories and Determinant Bundles. *International Journal of Mathematics*. 18:919–993, 2007.
- [AU2] J. E. Andersen & K. Ueno, Constructing modular functors from conformal field theories. *Journal of Knot theory and its Ramifications*. 16(2):127–202, 2007.
- [A3] J. E. Andersen. The Nielsen-Thurston classification of mapping classes is determined by TQFT. math.QA/0605036. J. Math. Kyoto Univ. 48(2):323– 338, 2008.
- [A4] J.E. Andersen. Asymptotics of the Hilbert-Schmidt Norm of Curve Operators in TQFT. Letters in Mathematical Physics 91:205–214, 2010.
- [A5] J.E. Andersen. Toeplitz operators and Hitchin's projectively flat connection. in *The many facets of geometry: A tribute to Nigel Hitchin*, Edited by O. García-Prada, Jean Pierre Bourguignon, Simon Salamon, 177–209, Oxford Univ. Press, Oxford, 2010.

- [AB1] J.E. Andersen & J. L. Blaavand, Asymptotics of Toeplitz operators and applications in TQFT, *Traveaux Mathématiques*, 19:167–201, 2011.
- [AGa1] J.E. Andersen & N.L. Gammelgaard. Hitchin's Projectively Flat Connection, Toeplitz Operators and the Asymptotic Expansion of TQFT Curve Operators. Grassmannians, Moduli Spaces and Vector Bundles, 1–24, Clay Math. Proc., 14, Amer. Math. Soc., Providence, RI, 2011.
- [AU3] J. E. Andersen & K. Ueno. Modular functors are determined by their genus zero data. *Quantum Topology*. 3:255–291, 2012.
- [A6] J. E. Andersen. Hitchin's connection, Toeplitz operators, and symmetry invariant deformation quantization. *Quantum Topol.* 3(3-4):293–325, 2012.
- [AGL] J.E. Andersen, N.L. Gammelgaard & M.R. Lauridsen, Hitchin's Connection in Metaplectic Quantization, *Quantum Topology* 3:327–357, 2012.
- [AHi] J. E. Andersen & B. Himpel. The Witten-Reshetikhin-Turaev invariants of finite order mapping tori II Quantum Topology. 3:377–421, 2012.
- [A7] J. E. Andersen. The Witten-Reshetikhin-Turaev invariants of finite order mapping tori I. Journal f
  ür Reine und Angewandte Mathematik. 681:1–38, 2013.
- [AU4] J. E. Andersen & K. Ueno. Construction of the Witten-Reshetikhin-Turaev TQFT from conformal field theory. *Invent. Math.* 201(2):519–559, 2015.
- [AP1] J. E. Andersen & N. S. Poulsen, An explicit Ricci potential for the Universal Moduli Space of Vector Bundles, arXiv:1609.01242, 2016.
- [AP2] J. E. Andersen & N. S. Poulsen, The Curvature of the Hitchin Connection, arXiv:1609.01243, 2016.
- [H] N. J. Hitchin, Flat connections and geometric quantization, Comm. Math. Phys., 131(2):347–380, 1990.
- [L] Y. Laszlo. Hitchin's and WZW connections are the same. J. Diff. Geom. 49(3):547–576, 1998.
- [MS] V.B. Mehta and C.S. Seshadri, Moduli of vector bundles on curves with parabolic structures, *Math. Ann.*, 248,(3):205–239, 1980.
- [NSNS] M. S. Narasimhan, R. R. Simha, R. Narasimhan and C. S. Seshadri, *Riemann surfaces*, volume 1 of *Mathematical Pamphlets*, Tata Institute of Fundamental Research, Bombay.
- [RT1] N. Reshetikhin & V. Turaev. Ribbon graphs and their invariants derived from quantum groups *Comm. Math. Phys.* 127:1–26, 1990.

- [RT2] N. Reshetikhin & V. Turaev. Invariants of 3-manifolds via link polynomials and quantum groups *Invent. Math.* 103:547–597, 1991.
- [TZ1] L. A. Takhtadzhyan and P. G. Zograf, The geometry of moduli spaces of vector bundles over a Riemann surface, *Izv. Akad. Nauk SSSR Ser. Mat.*, 53(4):753–770, 911, 1989.
- [TZ2] L. A. Takhtajan and P. G. Zograf, The first Chern form on moduli of parabolic bundles, *Math. Ann.*, **341**(1):113–135, 2008.
- [TZ3] L.A. Takhtajan and P. G.Zograf, A local index theorem for families of operators on punctured Riemann surfaces and a new Kähler metric on their moduli spaces, *Comm. Math. Phys.*, **137**(2):399–426, 1991.
- [TUY] A. Tsuchiya, K. Ueno & Y. Yamada. Conformal Field Theory on Universal Family of Stable Curves with Gauge Symmetries Advanced Studies in Pure Mathematics. 19:459–566, 1989.
- [T] V. G. Turaev. Quantum invariants of knots and 3-manifolds, volume 18 of de Gruyter Studies in Mathematics. Walter de Gruyter & Co., Berlin, 1994.
- [W] S. A. Wolpert, Chern forms and the Riemann tensor for the moduli space of curves, *Invent. Math.*, 85(1):119–145, 1986.
- [W1] E. Witten. Quantum field theory and the Jones polynomial. Comm. Math. Phys. 121:351–98, 1989.

Jørgen Ellegaard Andersen and Niccolo Skovgård Poulsen Center for Quantum Geometry of Moduli Spaces Department of Mathematics University of Aarhus DK-8000, Denmark Travaux mathématiques, Volume 25 (2017), 41-95, © by the authors

# Asymptotic aspects of the Teichmüller TQFT

by Jørgen Ellegaard Andersen and Jens-Jakob Kratmann Nissen<sup>1</sup>

#### Abstract

We calculate the knot invariant coming from the Teichmüller TQFT introduced by Andersen and Kashaev a couple of years ago. Specifically we calculate the knot invariant for the complement of the knot  $6_1$  both in Andersen and Kashaev's original version of Teichmüller TQFT but also in their new formulation of the Teichmüller TQFT for the one-vertex Htriangulation of  $(S^3, 6_1)$ . We show that the two formulations give equivalent answers. Furthermore we apply a formal stationary phase analysis and arrive at the Andersen-Kashaev volume conjecture.

Furthermore we calculate the first examples of knot complements in the new formulation showing that the new formulation is equivalent to the original one in all the special cases considered.

Finally, we provide an explicit isomorphism between the Teichmüller TQFT representation of the mapping class group of a once punctured torus and a representation of this mapping class group on the space of Schwartz class functions on the real line.

# 1 Introduction

Since discovered and axiomatised by Atiyah [At], Segal [S] and Witten [W], Topological Quantum Field Theories (TQFT's) have been studied extensively. The first constructions of such theories in dimension 2+1 was given by Reshetikhin and Turaev [T, RT1, RT2] who obtained TQFT's through surgery and the combinatorial framework of Kirby calculus, and by Turaev and Viro [TV] using the framework of triangulations and Pachner moves. In both constructions the central algebraic ingredients comes from the category of finite dimensional representation of the quantum group  $U_q(\mathfrak{sl}(2,\mathbb{C}))$  at roots of unity. Subsequently Blanchet, Habegger, Masbaum and Vogel gave a pure topological construction using Skein theory [BHMV1, BHMV2]. Recently it has been established by the first author and Ueno that this TQFT is equivalent to the one coming from conformal field theory [AU1, AU2, AU3, AU4] and further by the work of Laszlo [L] in the higher genus

<sup>&</sup>lt;sup>1</sup>Work supported in part by the center of excellence grant "Center for Quantum Geometry of Moduli Spaces" from the Danish National Research Foundation (DNRF95).

case with no marked point and the first author and Egsgaard [AE] in genus zero with marked points (for certain labels), that these TQFT's can be studied from the point of view of geometric quantization of the compact moduli space of flat SU(2) connections. The first author has extensively studied the asymptotics of this TQFT using this quantization of moduli spaces approach to this theory [A1, A2, AGr1, AH, AMU, A3, A4, A5, AGa1, AB1, A6, AGL, AHi, A7, AHJMMc].

A new line of development was initiated by Kashaev in [K1] where a state sum invariant of links in 3-manifolds was defined by using the combinatorics of charged triangulations. Here the charges are algebraic versions of dihedral angles of ideal hyperbolic tetrahedra in finite cyclic groups. The approach was subsequently developed further by Baseilhac, Benedetti and by Geer, Kashaev and Turaev [BB, GKT].

New challenges appear when one tries to construct combinatorial versions of Chern–Simons theory with non-compact gauge group such as  $PSL(2, \mathbb{R})$ , which is the isometry group of 2-dimensional hyperbolic space. When one considers the corresponding classical moduli space of flat  $PSL(2, \mathbb{R})$ -connections on a two dimensional surface, a connected component is identified with Teichmüller space, hence this Chern–Simons theory deserves the name Teichmüller TQFT.

Quantum Teichmüller theory corresponds to a specific classes of unitary mapping class representations on infinite dimensional Hilbert spaces [K2, FC]. Based on quantum Teichmüller theory several formal state-integral partition functions have been studied by Hikami, Dimofte, Gukov, Lenells, Zagier, Dijkgraaf, Fuji, Manabe [H1, H2, DGLZ, DFM] with the view to approach the Teichmüller TQFT. The question about convergence of the studied integrals however remained open until a mathematical rigorous version of Teichmüller TQFT was suggested by the first author and Kashaev in [AK1]. See also [AK1a, AK1b]. The convergence property of the Teichmüller TQFT is a property of the underlying combinatorial setting. An extra structure on the triangulations called a shape structure is imposed where each tetrahedron carries dihedral angles of an ideal hyperbolic tetrahedron. The dihedral angles provide absolute convergence and moreover they implement the complete symmetry with respect to change of edge orientation. The positivity condition of dihedral angles seems to impose restrictions on the construction of topologically invariant partition functions. In [KLV] Kashaev, Luo and Vartanov suggests a TQFT of Turaev–Viro type based on the combinatorics of shaped triangulations. As the absolute convergence of the partition function in this model is also based on positivity of dihedral angles, it is similar to the Teichmüller TQFT. A consequence is that as in the case of the Teichmüller TQFT the 2-3 Pachner move is not immediately always applicable. However, in this model no other topological restrictions are needed. A new formulation of the Teichmüller TQFT was suggested in [AK2]. In the new formulation of Teichmüller TQFT both the 2-3 and 3-2 Pachner moves are applicable and as in the case of the TQFT of Turaev–Viro type [KLV] no other topological restrictions are needed.

Recently in [AK3] the first author of this paper and Kashaev have constructed quantum Chern–Simons theory for  $PSL(2, \mathbb{C})$  for all non-negative integer levels kand furthermore understood how it relates to geometric quantization of  $PSL(2, \mathbb{C})$ moduli spaces. They have proposed a general scheme which just requires a Pontryagin self-dual locally compact group, which is expected to lead to the construction of the  $SL(n, \mathbb{C})$  quantum Chern-Simons theory for all non-negative integer levels k. From the geometric quantization of moduli spaces viewpoint, the corresponding representations of the mapping class groups have been constructed in [AG]. This work is closely related to the work of Dimofte [Di] on the physics side. The Teichmüller TQFT is the complex quantum Chern–Simons TQFT for  $PSL(2, \mathbb{C})$  at level k = 1. See also [AM] in this volume.

### Outline

We will review the construction of the charged tetrahedral operators which originates from Kashaev's quantization of Teichmüller space. The main ingredients in this theory are Penner's cell decomposition of decorated Teichmüller space and the associated Ptolemy groupoid [P] and Faddeev's quantum dilogarithm [F] which allows us to change polarization on Teichmüller space.

We recall how the partition function from the Teichmüller TQFT is defined using tetrahedral operators in both the original version [AK1] and in the new formulation [AK2].

We will then prove the equivalence of the two versions by direct calculations in several cases and elaborate on the Andersen-Kashaev *volume conjecture* arising in [AK1, Conj. 1.].

Following this, we will investigate the Teichmüller TQFT representation of the genus one, one parked point, mapping class group and prove that it is equivalent to an action of this same mapping class group acting on the space of Schwartz class functions on the real line.

#### Acknowledgements

We would like to thank Rinat Kashaev for many interesting discussions.

# 2 Teichmüller Space

As mentioned in the introduction quantum Chern–Simons theory with non-compact gauge group is of great interest. The gauge group in Teichmüller theory is  $PSL(2, \mathbb{R})$  which is the isometry group of 2 dimensional hyperbolic space.

Let M be a 3-manifold. Recall that the classical phase space of Chern–Simons theory with gauge group  $PSL(2, \mathbb{R})$  is given by the moduli space of flat connections

$$\mathcal{M} = \operatorname{Hom}(\pi_1(M), \operatorname{PSL}(2, \mathbb{R}))/\operatorname{PSL}(2, \mathbb{R}).$$

It is natural to start from  $M = \Sigma \times \mathbb{R}$ , where  $\pi_1(M) = \pi_1(\Sigma)$ , so we can talk about the moduli space of flat connections on the surface

$$\mathcal{M}_{\Sigma} = \operatorname{Hom}(\pi_1(\Sigma), \operatorname{PSL}(2, \mathbb{R}))/\operatorname{PSL}(2, \mathbb{R}).$$

We can write the moduli space as the disjoint union of connected components

$$\mathcal{M}_{\Sigma} = \bigsqcup_{-\chi(\Sigma) \le k \le \chi(\Sigma)} (\mathcal{M}_{\Sigma})_k.$$

Teichmüller space is given by the connected component with the maximal index

$$\mathcal{T}_{\Sigma} = (\mathcal{M}_{\Sigma})_{-\chi(\Sigma)}.$$

Let  $\Sigma = \Sigma_{g,s}$  be a surface of finite type, i.e.  $\Sigma$  is an oriented genus g surface with s boundary components or punctures. Then, topologically, Teichmüller space is an open ball of dimension 6g - 6 + 2s, i.e.

$$\mathcal{T}_{\Sigma} \cong \mathbb{R}^{6g-6+2s}.$$

Recall that  $\mathcal{T}_{\Sigma}$  is a symplectic space, where the symplectic structure is given by the Weil–Petersson symplectic form.

# 2.1 Penner coordinate system on $\tilde{\mathcal{T}}_{\Sigma}$

Let  $\Sigma$  be an oriented genus g surface with s > 0 punctures and Euler characteristic 2 - 2g - s < 0. We denote the set of punctures

$$V := \{P_1, \dots, P_s\}.$$

**Definition 2.1.** A homotopy class of a path running between  $P_i$  and  $P_j$  is called an ideal arc. A set of ideal arcs obtained by taking a family X of disjointly embedded ideal arcs in  $\Sigma$  running between punctures and subject to the condition that each component of  $\Sigma \setminus X$  is a triangle is called an ideal triangulation. Let  $\Delta_{\Sigma}$ denote the set of all ideal triangulations.

Now take an ideal triangulation  $\tau \in \Delta_{\Sigma}$  and calculate all  $\lambda$ -lengths with respect to a fixed configuration of horocycles. Let  $E(\tau)$  denote the set of edges in  $\tau$ . We impose the equivalence relation

$$\lambda \sim \lambda' : E(\tau) \to \mathbb{R}_{>0},$$

Asymptotic aspects of the Teichmüller TQFT

if there exists  $f: V \to \mathbb{R}_{>0}$  such that

$$\lambda'(e) = f(v_1)f(v_2)\lambda(e) \quad e \in E(\tau),$$

where  $v_1$  and  $v_2$  are endpoints of e. Counting the number of edges and vertices in an ideal triangulation establishes the following

$$\mathbb{R}^{E(\tau)}_{>0}/\mathbb{R}^{V}_{>0} \cong \mathbb{R}^{6g-6+2s}.$$

The  $\lambda$ -lengths parametrizes the decorated Teichmüller space  $\tilde{\mathcal{T}}_{\Sigma}$ , which is a principal  $\mathbb{R}^s_{>0}$  foliated fibration  $\phi: \tilde{\mathcal{T}}_{\Sigma} \to \mathcal{T}_{\Sigma}$ , where the fiber over a point of  $\mathcal{T}_{\Sigma}$  is the space of all horocycles about the punctures of  $\Sigma$ .

**Theorem 2.2** (Penner). (a) As a topological space the decorated Teichmüller space is homeomorphic to the set of positive numbers on edges given by  $\lambda$ -lengths

$$\tilde{\mathcal{T}}_{\Sigma} \cong \mathbb{R}^{E(\tau)}_{>0}.$$

(b) Using the map φ which forgets the horocycles we can pull back the Weil-Petersson symplectic form to the decorated Teichmüller space. The pull-back satisfies the formula

$$\phi^*\omega_{WP} = \sum_{\substack{a \\ c \\ c}} \frac{da \wedge db}{ab} + \frac{db \wedge dc}{bc} + \frac{dc \wedge da}{ca}.$$

(c) The mapping class group is contained in the groupoid generated by Ptolemy transformations. Suppose  $a, b, c, d, e \in \tau \in \Delta_{\Sigma}$  are such that  $\{a, b, e\}$  and  $\{c, d, e\}$  bound distinct triangles. The operation that changes the ideal triangulation  $\tau$  into  $\tau^e$ , which consists of the ideal arcs of  $\tau$  except e, which is replaced by e' such that triangles  $\{a, b, e\}$  and  $\{c, d, e\}$  are replaced by  $\{b, c, e'\}$ and  $\{a, d, e'\}$ , is called an elementary move (see Figure 1). The six  $\lambda$ -lengths are related by one single equation

$$(2.1) ee' = ac + bd.$$

Due to positivity this is a global coordinate change between parametrizations associated to two ideal triangulations. Two ideal triangulations are related through a sequence of flips. Composing the relations on Ptolemy transformations one obtains the relations between two coordinate systems.



Figure 1: Elementary move



Figure 2: Ratio coordinates

#### 2.2 Ratio Coordinates

**Definition 2.3.** An ideal triangulation with a choice of distinguished corner for each triangle is called a *decorated ideal triangulation* (d.i.t).

For an ideal triangle with sides having  $\lambda$ -lengths a, b, c we assign ratio coordinates according to Figure 2, where  $t = \left(\frac{a}{c}, \frac{b}{c}\right) = (t_1, t_2)$ . The pull back of the Weil–Petersson symplectic 2-form is then written in the very simple way

$$\phi^* \omega_{WP} = \sum_{\underline{\land t}} \frac{dt_1 \wedge dt_2}{t_1 t_2} =: \sum_{\underline{\land t}} \omega_t,$$

where the sum is over all triangles.

The d.i.t.  $\tau_t$  obtained from  $\tau$  by a change of distinguished corner of triangle t as indicated in Figure 3 is said to be obtained from  $\tau$  by the *elementary change of decoration* in triangle t. The d.i.t.  $\tau^e$  obtained from the d.i.t.  $\tau$  by the elementary move along the i.a. e, where distinguished corners are as indicated in Figure 4, is said to be obtained from  $\tau$  by the *decorated elementary move* along the i.a. e.



Figure 3: Elementary change of decoration.

It is easily seen that the coordinates u, v are related to the coordinates x, y. The relation is given by the two functions in x and y.

$$u = x \cdot y = (x_1 y_1, x_1 y_2 + x_2),$$



Figure 4: Decorated elementary move

$$v = x * y = \left(\frac{x_2y_1}{x_1y_2 + x_2}, \frac{y_2}{x_1y_2 + x_2}\right).$$

We now observe that

$$\omega_x + \omega_y = \omega_u + \omega_v$$

so that the change of coordinate with respect to this transformation

$$T: (x, y) \mapsto (u, v)$$

is a symplectomorphism of  $\mathbb{R}^4_{>0}$ .

# 3 Tetrahedral operator from quantum Teichmüller theory

We recall the main algebraic ingredients of quantum Teichmüller theory, following the approach of [K2, K3, K4]. Consider the canonical quantization of  $T^*\mathbb{R}^n$  with the standard symplectic structure in the position representation. The Hilbert space we get is  $L^2(\mathbb{R}^n)$ . Position coordinates  $q_i$  and momentum coordinates  $p_i$ on  $T^*\mathbb{R}^n$  upon quantization becomes selfadjoint unbounded operators  $\mathbf{q}_i$  and  $\mathbf{p}_i$ acting on  $L^2(\mathbb{R}^n)$  via the formulae

$$\mathbf{q}_j(f)(t) = t_j f(t), \quad \mathbf{p}_j(f)(t) = \frac{1}{2\pi i} \frac{\partial}{\partial t_j} f(t), \quad \forall t \in \mathbb{R}^n,$$

satisfying the Heisenberg commutation relations

(3.1) 
$$[\mathbf{q}_j, \mathbf{q}_k] = [\mathbf{p}_j, \mathbf{p}_k] = 0, \quad [\mathbf{p}_j, \mathbf{q}_k] = \frac{1}{2\pi i} \delta_{j,k}.$$

By the spectral theorem, one defines the operators

$$\mathbf{u}_i = e^{2\pi \,\mathbf{b}\,\mathbf{q}_i}, \quad \mathbf{v}_i = e^{2\pi \,\mathbf{b}\,\mathbf{p}_i}.$$

The commutation relations for  $\mathbf{u}_i$  and  $\mathbf{v}_j$  takes the form

$$[\mathbf{u}_j, \mathbf{u}_k] = [\mathbf{v}_j, \mathbf{v}_k] = 0, \quad \mathbf{u}_j \mathbf{v}_k = e^{2\pi \mathbf{b}^2 \, \delta_{j,k}} \mathbf{v}_k \mathbf{u}_j.$$

J. E. Andersen J. J. K. Nissen

Consider the operations for  $\mathbf{w}_j = (\mathbf{u}_j, \mathbf{v}_j), j \in \{1, 2\}$ ,

$$\mathbf{w}_1 \cdot \mathbf{w}_2 := (\mathbf{u}_1 \mathbf{u}_2, \mathbf{u}_1 \mathbf{v}_2 + \mathbf{v}_1),$$

(3.3) 
$$\mathbf{w}_1 * \mathbf{w}_2 := (\mathbf{v}_1 \mathbf{u}_2 (\mathbf{u}_1 \mathbf{v}_2 + \mathbf{v}_1)^{-1}, \mathbf{v}_2 (\mathbf{u}_1 \mathbf{v}_2 + \mathbf{v}_1)^{-1})$$

**Proposition 3.1** (Kashaev). Let  $\psi(z)$  be some solution to the functional equation

(3.4) 
$$\psi\left(z+\frac{i\,\mathrm{b}}{2}\right) = \psi\left(z-\frac{i\,\mathrm{b}}{2}\right)(1+e^{2\pi\,\mathrm{b}\,z}), \quad z \in \mathbb{C}.$$

Then, the operator

(3.5) 
$$\mathbf{T} = \mathbf{T}_{12} := e^{2\pi i \mathbf{p}_1 \mathbf{q}_2} \psi \left( \mathbf{q}_1 - \mathbf{q}_2 + \mathbf{p}_2 \right),$$

defines a continuous linear map from  $\mathcal{S}(\mathbb{R}^4)$  to  $\mathcal{S}(\mathbb{R}^4)$ , which satisfies the equations

$$\mathbf{w}_1 \cdot \mathbf{w}_2 \mathbf{T} = \mathbf{T} \mathbf{w}_1, \quad \mathbf{w}_1 * \mathbf{w}_2 \mathbf{T} = \mathbf{T} \mathbf{w}_2.$$

For a proof of this proposition see [AK1] and Appendix B. One particular solution of (3.4) is given by Faddeev's quantum dilogarithm [F]

(3.7) 
$$\psi(z) = \frac{1}{\Phi_{\rm b}(z)}.$$

The most important property of the operator (3.5) is the pentagon identity in  $L^2(\mathbb{R}^3)$ 

$$\mathbf{T}_{12}\mathbf{T}_{13}\mathbf{T}_{23} = \mathbf{T}_{23}\mathbf{T}_{12},$$

which follows from the five term identity (A.17) satisfied by Faddeev's quantum dilogarithm. The indices in (3.8) has the standard meaning. For example  $\mathbf{T}_{13}$  is obtained from  $\mathbf{T}_{12}$  by replacing  $\mathbf{q}_2$  and  $\mathbf{p}_2$  with  $\mathbf{q}_3$  and  $\mathbf{p}_3$  respectively and so forth.

#### 3.1 Oriented triangulated pseudo 3-manifolds

Consider the disjoint union of finitely many copies of the standard 3-simplices in  $\mathbb{R}^3$ , each having totally ordered vertices. The order of the vertices induces an orientation on edges. Identify some codimension-1 faces of this union in pairs by vertex order preserving and orientation reversing affine homeomorphisms called gluing homeomorphisms. The quotient space X is a specific CW-complex with oriented edges which will be called an oriented triangulated pseudo 3-manifold. For  $i \in \{0, 1, 2, 3\}$ , we denote by  $\Delta_i(X)$  the set of *i*-dimensional cells in X. For any i > j, we denote

$$\Delta_i^j(X) = \{(a,b) \mid a \in \Delta_i(X), b \in \Delta_j(a)\}$$

with natural projection maps

$$\phi_{i,j}: \Delta_i^j(X) \to \Delta_i(X), \quad \phi^{i,j}: \Delta_i^j(X) \to \Delta_j(X).$$

We also have canonical boundary maps

$$\partial_i : \Delta_j(X) \to \Delta_{j-1}(X), \quad 0 \le i \le j,$$

which in the case of a *j*-dimensional simplex  $S = [v_0, v_1, \ldots, v_j]$  with ordered vertices  $v_0, v_1, \ldots, v_j$  in  $\mathbb{R}^3$  takes the form

 $\partial_i S = [v_0, \dots, v_{i-1}, v_{i+1}, \dots, v_j], \quad i \in \{0, \dots, j\}.$ 

#### 3.2 Shaped 3-manifolds

Let X be an oriented triangulated pseudo 3-manifold.

**Definition 3.2.** A *shape structure* on X is an assignment to each edge of each tetrahedron of X a positive number called the dihedral angle,

$$\alpha_X : \Delta^1_3(X) \to \mathbb{R}_+$$

so that the sum of the three angles at the edges from each vertex of each tetrahedron is  $\pi$ . An oriented triangulated pseudo 3-manifold with a shape structure will be called a shaped pseudo 3-manifold.

It is straightforward to see that the dihedral angles at opposite edges are equal.



Figure 5: Labeling of edges by dihedral angles.

**Definition 3.3.** To each shape structure on X, we associate a Weight function

$$\omega_X: \Delta_1(X) \to \mathbb{R}_+,$$

which to each edge of X associates the total sum of dihedral angles around it

$$\omega_X(e) = \sum_{(T,e)\in\Delta_3^1(X)} \alpha_X(T,e)$$

An edge of a shaped pseudo 3-manifold X will be called *balanced* if it is internal and  $\omega_X(e) = 2\pi$ . We call a shaped pseudo 3-manifold *fully balanced* if all edges of X are balanced.

#### 3.3 Shape gauge transformation

In the space of shape structures on a pseudo 3-manifold there is a gauge group action. The gauge group is generated by the total dihedral angles around internal edges acting through the Neumann–Zagier Poisson bracket. See [AK1] for further details.

#### 3.4 Geometric interpretation of the five term identity

For an operator **T** we denote the integral kernel of the operator as  $\langle x_0, x_2 | \mathbf{T} | x_1, x_3 \rangle$ . Then the pentagon identity can be written in the following way

(3.9) 
$$\langle x, y, z \mid \mathbf{T}_{12}\mathbf{T}_{13}\mathbf{T}_{23} \mid u, v, w \rangle = \langle x, y, z \mid \mathbf{T}_{23}\mathbf{T}_{12} \mid u, v, w \rangle$$

Decomposition of unity gives for the left hand side of (3.9)

$$\langle x, y, z \mid \mathbf{T}_{12} \mathbf{T}_{13} \mathbf{T}_{23} \mid u, v, w \rangle = \int \langle x, y \mid \mathbf{T} \mid \alpha_1, \alpha_2 \rangle \langle \alpha_1, z \mid \mathbf{T} \mid u, \beta_3 \rangle$$

$$\langle \alpha_2, \beta_3 \mid \mathbf{T} \mid v, w \rangle d\alpha_1 d\alpha_2 d\beta_3.$$

Decomposing of unity for the right hand side gives

$$\langle x, y, z \mid \mathbf{T}_{23}\mathbf{T}_{12} \mid u, v, w \rangle = \int \langle y, z \mid \mathbf{T} \mid \gamma_2, w \rangle \langle x, \gamma_2 \mid \mathbf{T} \mid u, v \rangle d\gamma_2.$$

To make the correspondence between the pentagon identity and the 3-2 Pachner move precise, we label each vertex of a tetrahedron T with a number  $i \in \{0, 1, 2, 3\}$ . The numbers on vertices induce an orientation on edges, i.e. we put arrows on the edges pointing in the direction from the smaller to the bigger label on vertices. The number at a vertex corresponds to the number of incoming edges, see Figure 6.



Figure 6: Interpretation of a positively oriented tetrahedron.



Figure 7: A tetrahedron in state x.

#### 3.5 States

A state of a tetrahedron T with totally ordered vertices  $\{0, 1, 2, 3\}$  is a map

$$x : \Delta_2(X) \to \mathbb{R}.$$

A tetrahedron in state x is illustrated in Figure 7, where  $x_i := x(\partial_i T)$ .

We identify a tetrahedron T in state x as in Figure 7 with the integral kernel  $\langle x_0, x_2 | \mathbf{T} | x_1, x_3 \rangle$ . This gives a geometric interpretation of the pentagon identity (3.9) as the 2-3 Pachner move as illustrated in Figure 8 and Figure 9. The



Figure 8: Decomposition of tetrahedra in the 2-3 Pachner move.

integrations corresponds to gluing of faces as illustrated in Figure 9.



Figure 9: 3-2 Pachner move.

#### 3.6 Integral kernel

Let us calculate the integral kernel of the operator  $\mathbf{T}$ 

$$\langle x_0, x_2 \mid \mathbf{T} \mid x_1, x_3 \rangle \equiv \mathbf{T}f(x, y) = \int \langle x, y \mid \mathbf{T} \mid u, v \rangle f(u, v) du dv,$$

where  $\mathbf{T}$  is the operator given by (3.5).

$$\begin{split} \langle x_0, x_2 \mid \mathbf{T} \mid x_1, x_3 \rangle &= \langle x_0, x_2 \mid e^{2\pi i \mathbf{p}_1 \mathbf{q}_2} \psi \left( \mathbf{q}_1 - \mathbf{q}_2 + \mathbf{p}_2 \right) \mid x_1, x_3 \rangle \\ &= e^{x_2 \frac{\partial}{\partial x_0}} \langle x_0, x_2 \mid \psi \left( \mathbf{q}_1 - \mathbf{q}_2 + \mathbf{p}_2 \right) \mid x_1, x_3 \rangle \\ &= \langle x_0 + x_2, x_2 \mid \psi \left( \mathbf{q}_1 - \mathbf{q}_2 + \mathbf{p}_2 \right) \mid x_1, x_3 \rangle \\ &= \int \langle x_0 + x_2, x_2 \mid e^{2\pi i (\mathbf{q}_1 - \mathbf{q}_2 + \mathbf{p}_2) y} \mid x_1, x_3 \rangle \tilde{\psi}(y) \, dy \\ &= \int e^{2\pi i y x_1} \delta(x_1 - x_0 - x_2) \langle x_2 \mid e^{2\pi i (\mathbf{p}_2 - \mathbf{q}_2) y} \mid x_3 \rangle \tilde{\psi}(y) \, dy \\ &= \int e^{2\pi i x_1 y} \delta(x_1 - x_0 - x_2) \tilde{\psi}(y) e^{-2\pi i x_3 y} \langle x_2 + y \mid x_3 \rangle \, dy \\ &= \int e^{2\pi i x_1 y} \delta(x_1 - x_0 - x_2) \tilde{\psi}(y) e^{-2\pi i x_3 y} \delta(x_2 + y - x_3) e^{\pi i y^2} \, dy \\ &= e^{2\pi i x_1 (x_3 - x_2)} \delta(x_1 - x_0 - x_2) \tilde{\psi}(x_3 - x_2) e^{-2\pi i x_3 (x_3 - x_2) + \pi i (x_3 - x_2)^2} \\ &= \delta(x_1 - x_0 - x_2) \tilde{\psi}'(x_3 - x_2) e^{2\pi i x_0 (x_3 - x_2)}, \end{split}$$

where

$$\tilde{\psi}'(x) := \tilde{\psi}(x)e^{-\pi i x^2}$$
, and  $\tilde{\psi}(x) := \int_{\mathbb{R}} \psi(y)e^{2\pi i x y} dy$ .

## 3.7 Positively and negatively oriented tetrahedra

In an oriented triangulated 3-manifold there are two possibilities for the orientation of tetrahedra. The orientation follows from Figure 10. To a negatively oriented tetrahedron the integral kernel associated to it in the geometric interpretation is the complex conjugate of that of a positively oriented tetrahedron.

## 3.8 Charged Tetrahedral Operators and Pentagon Identity

To ensure that the Fourier integral is absolutely convergent charges on the operator **T** are introduced. For any positive real a and c such that  $b := \frac{1}{2} - a - c$  is also positive, define the charged **T**-operators

(3.10) 
$$\mathbf{T}(a,c) := e^{-\pi i c_{\mathrm{b}}^{2} (4(a-c)+1)/6} e^{4\pi i c_{\mathrm{b}} (c\mathbf{q}_{2}-a\mathbf{q}_{1})} \mathbf{T} e^{-4\pi i c_{\mathrm{b}} (a\mathbf{p}_{2}+c\mathbf{q}_{2})}$$





Positively oriented tetrahedron.

Negatively oriented tetrahedron.

Figure 10: Orientations on tetrahedra.

and

$$\bar{\mathbf{T}}(a,c) := e^{\pi i c_{\mathbf{b}}^2 (4(a-c)+1)/6} e^{-4\pi i c_{\mathbf{b}}(a\mathbf{p}_2 + c\mathbf{q}_2)} \bar{\mathbf{T}} e^{4\pi i c_{\mathbf{b}}(c\mathbf{q}_2 - a\mathbf{q}_1)}$$

where  $\bar{\mathbf{T}} := \mathbf{T}^{-1}$  and  $c_{\mathbf{b}} := \frac{i}{2}(\mathbf{b} + \mathbf{b}^{-1})$ . We have the equality

$$\mathbf{T}(a,c) = e^{2\pi i \mathbf{p}_1 \mathbf{q}_2} \psi_{a,c} (\mathbf{q}_1 - \mathbf{q}_2 + \mathbf{p}_2)$$

where

$$\psi_{a,c}(x) := \psi(x - 2c_{\rm b}(a+c))e^{-4\pi i c_{\rm b}a(x-c_{\rm b}(a+c))}e^{-\pi i c_{\rm b}^2(4(a-c)+1)/6}.$$

The Fourier transformation formula for Faddeev's quantum dilogarithm (A.19) leads to the identities

$$\begin{split} \tilde{\psi}'_{a,c}(x) &= e^{-\frac{\pi i}{12}} \psi_{c,b}(x), \\ \overline{\psi}_{a,c}(x) &= e^{-\frac{\pi i}{6}} e^{\pi i x^2} \psi_{c,a}(-x) = e^{-\frac{\pi i}{12}} \tilde{\psi}_{b,c}(-x), \\ \overline{\tilde{\psi}'_{a,c}}(x) &= e^{\frac{\pi i}{12}} \overline{\psi}_{c,b}(x) = e^{-\frac{\pi i}{12}} e^{\pi i x^2} \psi_{b,c}(-x). \end{split}$$

From the formulas above we obtain that

(3.11) 
$$\langle x_0, x_2 | \mathbf{T}(a,c) | x_1, x_3 \rangle = \delta(x_1 - x_0 - x_2) \tilde{\psi}'_{a,c}(x_3 - x_2) e^{2\pi i x_0(x_3 - x_2)},$$

(3.12) 
$$\langle x, y \mid \overline{\mathbf{T}}(a, c) \mid u, v \rangle = \overline{\langle u, v \mid \mathbf{T}(a, c) \mid x, y \rangle}$$

 $\label{eq:proposition 3.4} \mbox{ (Andersen-Kashaev [AK1]). The charged pentagon identity is satisfied }$ 

(3.13) 
$$\mathbf{T}_{12}(a_4, c_4) \mathbf{T}_{13}(a_2, c_2) \mathbf{T}_{23}(a_0, c_0) = e^{\pi i c_{\mathrm{b}}^2 P_e/3} \mathbf{T}_{23}(a_1, c_1) \mathbf{T}_{12}(a_3, c_3),$$

where

$$P_e = 2(c_0 + a_2 + c_4) - \frac{1}{2}$$

and  $a_0, a_1, a_2, a_3, a_4, c_0, c_1, c_2, c_3, c_4 \in \mathbb{R}$  are such that

$$a_1 = a_0 + a_2, a_3 = a_2 + a_4, c_1 = c_0 + a_4, c_3 = a_0 + c_4, c_2 = c_1 + c_3.$$

J. E. Andersen J. J. K. Nissen

#### 3.9 Partition function

For a tetrahedron  $T = [v_0, v_1, v_2, v_3]$  with ordered vertices  $v_0, v_1, v_2, v_3$ , we define its sign

$$sign(T) = sign(det(v_1 - v_0, v_2 - v_0, v_3 - v_0))$$

For  $(T, \alpha, x)$  an oriented tetrahedron with shape structure  $\alpha$  in state x, define the partition function taking values in the space of tempered distributions by the formula

(3.14) 
$$Z_{\hbar}(T, \alpha, x) := \begin{cases} \langle x_0, x_2 \mid \mathbf{T}(a, c) \mid x_1, x_3 \rangle, & \text{if } \operatorname{sign}(T) = 1 \\ \langle x_1, x_3 \mid \bar{\mathbf{T}}(a, c) \mid x_0, x_2 \rangle, & \text{if } \operatorname{sign}(T) = -1. \end{cases}$$

where

$$x_i := x(\partial_i T)$$

and

$$a = \frac{1}{2\pi} \alpha_T(T, e_{01}), \quad c = \frac{1}{2\pi} \alpha_T(T, e_{03}).$$

For a closed oriented triangulated pseudo 3-manifold X with shape structure  $\alpha$ , we associate the partition function

(3.15) 
$$Z_{\hbar}(X,\alpha) := \int_{x \in \mathbb{R}^{\Delta_2(X)}} \prod_{T \in \Delta_3(X)} Z_{\hbar}(T,\alpha,x) \, dx.$$

**Theorem 3.5** (Andersen–Kashaev [AK1]). If  $H_2(X \setminus \Delta_0(X), \mathbb{Z}) = 0$ , then the quantity  $|Z_h(X, \alpha)|$  is well defined in the sense that the integral is absolutely convergent, and

- 1. it depends on only the gauge reduced class of  $\alpha$ ;
- 2. it is invariant under 2-3 Pachner moves.

The definition of the partition function (3.15) can be extended to manifolds having boundary eventually giving rise to a TQFT, see [AK1].

#### 3.10 Invariants of knots in 3-manifolds

By considering one-vertex ideal triangulations of complements of hyperbolic knots in compact oriented closed 3-manifolds, we obtain knot invariants.

Another possibility is to consider a one-vertex Hamiltonian triangulation (Htriangulation) of pairs (a closed 3-manifold M, a knot K in M), i.e., a one-vertex triangulation of M, where the knot is represented by one edge, with degenerate shape structures, where the weight on the knot approaches zero and where simultaneously the weights on all other edges approach the balanced value  $2\pi$ . This limit by itself is divergent as a simple pole (after analytic continuation to complex

54

angles) in the weight of the knot, but the residue at this pole is a knot invariant which is a direct analogue of Kashaev's invariants [K1], which were at the origin of the hyperbolic volume conjecture.

In [AK1] the first author and Rinat Kashaev have set forth the following conjecture:

**Conjecture 3.6** (Andersen and Kashaev [AK1]). Let M be a closed oriented 3-manifold. For any hyperbolic knot  $K \subset M$ , there exists a smooth function  $J_{M,K}(h,x)$  on  $\mathbb{R}_{>0} \times \mathbb{R}$  which has the following properties.

(1) For any fully balanced shaped ideal triangulation X of the complement of K in M, there exists a gauge invariant real linear combination of dihedral angles λ, a (gauge non-invariant) real quadratic polynomial of dihedral angles φ such that

$$Z_{\hbar}(X) = e^{i\frac{\phi}{\hbar}} \int_{\mathbb{R}} J_{M,K}(\hbar, x) e^{-\frac{x\lambda}{\sqrt{\hbar}}} dx.$$

(2) For any one vertex shaped H-triangulation Y of the pair (M, K) there exists a real quadratic polynomial of dihedral angles  $\phi$  such that

$$\lim_{\omega_Y \to \tau} \Phi_{\rm b} \left( \frac{\pi - \omega_Y(K)}{2\pi i \sqrt{\hbar}} \right) Z_{\hbar}(Y) = e^{i\frac{\phi}{\hbar} - i\pi/12} J_{M,K}(\hbar, 0),$$

where  $\tau : \Delta_1(Y) \to \mathbb{R}$  takes the value 0 on the knot K and the value  $2\pi$  on all other edges.

(3) The hyperbolic volume of the complement of K in M is recovered as the limit

$$\lim_{\hbar \to 0} 2\pi\hbar \log |J_{M,K}(\hbar, 0)| = -\operatorname{vol}(M \setminus K).$$

**Theorem 3.7.** Conjecture 3.6 is true for the pair  $(S^3, 6_1)$  with

$$(3.16) J_{S^3,6_1}(\hbar, x) = \chi_{6_1}(x).$$

The function  $\chi_{6_1}(x)$  is defined to be:

$$\chi_{6_1}(x) = \int_{\mathbb{R}^2} \frac{e^{2\pi i (x^2 + \frac{1}{2}y^2 + 2xy)e^{4\pi i c_{\mathbf{b}}z}}}{\Phi_{\mathbf{b}}(x+y)\Phi_{\mathbf{b}}(x+z+c_{\mathbf{b}})\Phi_{\mathbf{b}}(y)\Phi_{\mathbf{b}}(z-x-y)} dy dz.$$

See [N] for a calculation of the invariant of an ideal triangulation of the complement of the knot  $6_1$ .

# 4 New formulation

In this section we recall the new formulation of the Teichmüller TQFT introduced by Andersen and Kashaev in [AK2].

#### 4.1 States and Boltzmann weights

Let  $T \subset \mathbb{R}^3$  be a tetrahedron with shape structure  $\alpha_T$  and vertex ordering mapping

$$v: \{0, 1, 2, 3\} \to \Delta_0(T).$$

A state of a tetrahedron T is a map  $x: \Delta_1(T) \to \mathbb{R}$ . Pictorially, a positive tetrahedron T in state x looks as follows



More generally, a state of a triangulated pseudo 3-manifold X is a map

$$y: \Delta_1(X) \to \mathbb{R}.$$

For any state y define the *Boltzmann weight* 

$$B(T, x) = g_{\alpha_1, \alpha_3}(y_{02} + y_{13} - y_{03} - y_{12}, y_{02} + y_{13} - y_{01} - y_{23})$$

if T is positive and complex conjugate otherwise. Here  $y_{ij} \equiv y(v_i v_j), \alpha_i \equiv \alpha_T(v_0 v_i)/2\pi$ ,

(4.1) 
$$g_{a,c}(s,t) = \sum_{m \in \mathbb{Z}} \tilde{\psi}'_{a,c}(s+m) e^{\pi i t (s+2m)}.$$

**Theorem 4.1** (Andersen–Kashaev [AK1]). Let X be a levelled shaped triangulated oriented pseudo 3-manifold. Then, the quantity

(4.2) 
$$Z_{\hbar}^{new}(X) := e^{\pi i l_X/4\hbar} \int_{[0,1]^{\Delta_1(X)}} \left( \prod_{T \in \Delta_3(X)} B\left(T, y\big|_{\Delta_1(T)}\right) \right) \, dx$$

admits an analytic continuation to a meromorphic function of the complex shapes, which is invariant under all shaped 2-3 and 3-2 Pachner moves (along balanced edges).

**Conjecture 4.2.** The proposed model in Theorem 4.1 is equivalent to the Teichmüller TQFT from [AK1].

**Theorem 4.3.** The new formulation of the Teichmüller TQFT is equivalent to the original formulation for the pairs  $(S^3, 3_1)$ ,  $(S^3, 4_1)$ ,  $(S^3, 5_2)$  and  $(S^3, 6_1)$ .

# 5 Calculations of specific knot complements

In the following calculations we encode an oriented triangulated pseudo 3-manifold X into a diagram where a tetrahedron T is represented by an element

$$(5.1) T =$$

where the vertical segments, ordered from left to right, correspond to the faces  $\partial_0 T, \partial_1 T, \ \partial_2 T, \partial_3 T$  respectively. When we glue tetrahedron along faces, we illustrate this by joining the corresponding vertical segments.

We will further use the notation

$$\nu_{a,c} := e^{-\pi i c_{b^2} (4(a-c)+1)/6}.$$

### 5.1 The complement of the figure-8-knot

Let X be the following oriented triangulated pseudo 3-manifold,

which represented the usual diagram for the complement of the figure eight knot. Choosing an orientation, the diagram consists of one positive tetrahedron  $T_+$  and one negative  $T_-$ .  $\partial X = \emptyset$  and combinatorially we have  $\Delta_0(X) = \{*\}, \Delta_1(X) = \{e_0, e_1\}$ . The gluing of the tetrahedra is vertex order preserving which means that edges are glued together in the following manner.

$$\begin{split} e_0 &= x_{01}^+ = x_{03}^+ = x_{23}^+ = x_{02}^- = x_{12}^- = x_{13}^- =: x, \\ e_1 &= x_{02}^+ = x_{13}^+ = x_{12}^+ = x_{01}^- = x_{03}^- = x_{23}^- =: y. \end{split}$$

That this diagram represents the complement of the figure eight know means that the topological space  $X \setminus \{*\}$  is homeomorphic to the complement of the figureeight knot. The set  $\Delta_3^1(X)$  consists of the elements  $(T_{\pm}, e_{j,k})$  for  $0 \le j < k \le 3$ . We fix a shape structure

$$\alpha_X : \Delta_3^1(X) \to \mathbb{R}_+$$

by the formulae

$$\alpha_X(T_{\pm}, e_{0,1}) = 2\pi a_{\pm}, \quad \alpha_X(T_{\pm}, e_{0,2}) = 2\pi b_{\pm}, \quad \alpha_X(T_{\pm}, e_{0,3}) = 2\pi c_{\pm},$$

where  $a_{\pm} + b_{\pm} + c_{\pm} = \frac{1}{2}$ . This result in the following weight functions

$$\omega_X(e_0) = 2a_+ + c_+ + 2b_- + c_-, \quad \omega_X(e_1) = 2b_+ + c_+ + 2a_- + c_-.$$

In the completely balanced case these equations correspond to

$$a_+ - b_+ = a_- - b_-$$

The Boltzmann weights are given by the functions

$$\begin{split} B\left(T_{+}, x_{|_{\Delta_{1}(T_{+})}}\right) &= g_{a_{+},c_{+}}(y-x,2(y-x)),\\ B\left(T_{-}, x_{|_{\Delta_{1}(T_{-})}}\right) &= \overline{g_{a_{-},c_{-}}(x-y,2(x-y))}. \end{split}$$

We calculate the partition function for the Teichmüller TQFT using the new formulation

$$\begin{split} Z_{\hbar}^{\text{new}}(X) &= \int_{[0,1]^2} \sum_{m,n\in\mathbb{Z}} \tilde{\psi}'_{a_+,c_+}(y-x+m) \overline{\tilde{\psi}'_{a_-,c_-}(x-y+n)} e^{4\pi i (y-x)(m+n)} \, dx dy \\ &= \int_{[0,1]} \sum_{m,n\in\mathbb{Z}} \tilde{\psi}'_{a_+,c_+}(y+m) \overline{\tilde{\psi}'_{a_-,c_-}(-y+n)} e^{4\pi i y(m+n)} dy \\ &= \sum_{m,n\in\mathbb{Z}} \int_{[m,m+1]} \tilde{\psi}'_{a_+,c_+}(y) \overline{\tilde{\psi}'_{a_-,c_-}(-y+m+n)} e^{4\pi i (y-m)(m+n)} dy \\ &= \sum_{p\in\mathbb{Z}} \int_{\mathbb{R}} \tilde{\psi}'_{a_+,c_+}(y) \overline{\tilde{\psi}'_{a_-,c_-}(-y+p)} e^{4\pi i y p} dy \\ &= e^{-\frac{\pi i}{6}} \sum_{p\in\mathbb{Z}} \int_{\mathbb{R}} \psi_{c_+,b_+}(y) \psi_{b_-,c_-}(y-p) e^{\pi i (y-p)^2} e^{4\pi i y p} dy \\ &= e^{-\frac{\pi i}{6}} \sum_{p\in\mathbb{Z}} \int_{\mathbb{R}} \psi(y-2c_{\mathbf{b}}(c_++b_+)) \psi(y-p-2c_{\mathbf{b}}(b_-+c_-)) e^{\pi i y^2} e^{\pi i p^2} e^{2\pi i y p} \\ &\times e^{-4\pi i c_{\mathbf{b}} c_+(y-c_{\mathbf{b}}(c_++b_+))} e^{-4\pi i c_{\mathbf{b}} b_-(y-p-c_{\mathbf{b}}(b_-+c_-))} \\ &\times e^{-\pi i (4(c_+-b_+)+1)/6} e^{-\pi i (4(b_--c_-)+1)/6} dy. \end{split}$$

We set  $Y = y - 2c_b(c_+ + b_+)$ . Assuming that we are in the completely balanced case we have that

$$-b_{-} - c_{-} + c_{+} + b_{+} = -b_{+} + b_{-}.$$

Furthermore we have  $y^2 = Y^2 + 4c_b^2(c_+ + b_+)^2 + 4c_bY(c_+ + b_+)$ . Implementing this we get the following expression

$$\begin{split} Z_{\hbar}^{\text{new}}(X) &= \nu_{c_{+},b_{+}}\nu_{b_{-},c_{-}}e^{-\frac{\pi i}{6}}\sum_{p\in\mathbb{Z}}\int_{\mathbb{R}}\psi(Y-p-2c_{\text{b}}(b_{+}-b_{-}))\psi(Y) \\ &\times e^{\pi i(Y^{2}+4c_{\text{b}}^{2}(c_{+}+b_{+})^{2}+4c_{\text{b}}Y(c_{+}+b_{+}))}e^{\pi ip^{2}} \\ &\times e^{2\pi i(Y+2c_{\text{b}}(c_{+}+b_{+}))p} \\ &\times e^{-4\pi ic_{\text{b}}c_{+}(Y+c_{\text{b}}(c_{+}+b_{+}))}e^{-4\pi ic_{\text{b}}b_{-}(Y-p-c_{\text{b}}(b_{-}+c_{-})+2c_{\text{b}}(c_{+}+b_{+}))}dY \\ &= \nu_{c_{+},b_{+}}\nu_{b_{-},c_{-}}e^{-\frac{\pi i}{6}}\sum_{p\in\mathbb{Z}}\int_{\mathbb{R}}\frac{1}{\Phi_{\text{b}}(Y-p-2c_{\text{b}}(b_{+}-b_{-}))}\frac{1}{\Phi_{\text{b}}(Y)} \\ &\times e^{\pi iY^{2}}e^{\pi ip^{2}}e^{-4\pi ic_{\text{b}}Y(-c_{+}-b_{+}+c_{+}+b_{-})}e^{2\pi iYp} \end{split}$$

Asymptotic aspects of the Teichmüller TQFT

$$\begin{array}{l} \times \ e^{-4\pi i c_{\rm b} p(-(c_++b_+)-b_-)} \\ \times \ e^{4\pi i c_{\rm b}^2((c_++b_+)^2-c_+(c_++b_+)-b_-(b_-+c_--2(c_++b_+))} dY. \end{array}$$

Now set

(5.3) 
$$u = 2c_{\rm b}(b_+ - b_-)$$

and

(5.4) 
$$v = 2b_{-} + c_{-} = 2b_{+} + c_{+},$$

and use the formula

$$\Phi_{\rm b}(z)\Phi_{\rm b}(-z) = \zeta_{inv}^{-1}e^{\pi i z^2},$$

together with the calculation

$$b_{-} + b_{+} + c_{+} = b_{-} + b_{+} - 2b_{+} + 2b_{-} + c_{-} = -(b_{+} - b_{-}) + (2b_{-} + c_{-})$$

to get that

$$\begin{split} Z_{\hbar}^{\text{new}}(X) &= \nu_{c_{+},b_{+}}\nu_{b_{-},c_{-}}\zeta_{inv}e^{-\frac{\pi i}{6}}\sum_{p\in\mathbb{Z}}\int_{\mathbb{R}}\frac{\Phi_{\text{b}}(p+u-Y)}{\Phi_{\text{b}}(Y)}e^{-\pi i(Y^{2}+u^{2}+p^{2}-2Yu-2Yp+2up)} \\ &\times e^{\pi iY^{2}}e^{\pi ip^{2}}e^{2\pi iYu}e^{2\pi iYp}dY \\ &\times e^{-2\pi ipu}e^{2\pi ipv} \\ &\times e^{4\pi ic_{\text{b}}^{2}((c_{+}+b_{+})^{2}-c_{+}(c_{+}+b_{+})-b_{-}(b_{-}+c_{-}-2(c_{+}+b_{+})))}. \end{split}$$

Using the balance condition and formulas (5.3) and (5.4) we get the equality

$$\begin{split} &-4\pi i c_{\rm b}^2 \{(c_++b_+)^2+b_-(-b_--c_-+2c_++2b_+)+c_+(c_++b_+)\} = \\ &-4\pi i c_{\rm b}^2 \{(-(b_+-b_1)^2)-c_+b_++c_+b_-+b_-c_+-b_-c_-\} = \\ &-2\pi i c_{\rm b} \{-(c_++2b_-)u\}+\pi i u^2 = \\ &-2\pi i c_{\rm b} \{-(2b_-+c_-)u+2(b_+-b_-))u\}+\pi i u^2 = \pi i (uv-u^2). \end{split}$$

We get the following expression for the partition function

$$\begin{split} Z_{\hbar}^{\text{new}}(X) &= \nu_{c_{+},b_{+}}\nu_{b_{-},c_{-}}\zeta_{inv}e^{-\frac{\pi i}{6}}\sum_{p\in\mathbb{Z}}\int_{\mathbb{R}}\frac{\Phi_{\text{b}}(p+u-Y)}{\Phi_{\text{b}}(Y)}e^{-\pi i u^{2}} \\ &\times e^{4\pi i Y u}e^{4\pi i Y p}e^{-4\pi i p u}e^{2\pi i p v}e^{\pi i (uv-u^{2})}dY \\ &= \nu_{c_{+},b_{+}}\nu_{b_{-},c_{-}}\zeta_{inv}e^{-\frac{\pi i}{6}}\sum_{p\in\mathbb{Z}}\int_{\mathbb{R}}\frac{\Phi_{\text{b}}(p+u-Y)}{\Phi_{\text{b}}(Y)} \\ &\times e^{4\pi i Y u}e^{4\pi i Y p}e^{-4\pi i p u}e^{2\pi i p v}e^{\pi i u v}e^{-2\pi i u^{2}}dY \end{split}$$

J. E. Andersen J. J. K. Nissen

$$=\nu_{c_{+},b_{+}}\nu_{b_{-},c_{-}}\zeta_{inv}e^{-\frac{\pi i}{6}}\sum_{p\in\mathbb{Z}}\int_{\mathbb{R}}\frac{\Phi_{\rm b}(p+u-Y)}{\Phi_{\rm b}(Y)}e^{2\pi i(u+p)(2Y-u-p)}dYe^{\pi iv(2p+u)}.$$

Using the Weil-Gel'fand-Zak transform we see that the partition function has the form

$$Z_{\hbar}^{\text{new}}(X) = \nu_{c_+,b_+} \nu_{b_-,c_-} \zeta_{inv} e^{-\frac{\pi i}{6}} W(\chi_{4_1})(u,v).$$

Where the function  $\chi_{4_1}(x) = \int_{\mathbb{R}^{-i0}} \frac{\Phi_{\mathrm{b}(x-y)}}{\Phi_{\mathrm{b}}(y)} e^{2\pi i x (2y-x)} dy$ . The function  $\chi_{4_1}(x)$  is exactly the function  $J_{S^3,4_1}(\hbar, x)$  from [AK1, Thm. 5]. It should be noted that this result is connected to Hikami's invariant. Andersen and Kashaev observes in [AK1] that the expression

$$\frac{1}{2\pi\,\mathrm{b}}\chi_{4_1}\left(-\frac{u}{\pi\,\mathrm{b}},\frac{1}{2}\right),\,$$

where  $\chi_{4_1}(x,\lambda) = \chi_{4_1}(x)e^{4\pi i c_b \lambda}$  is equal to the formal derived expression in [H2].

#### 5.2 One vertex H-triangulation of the figure-8-knot

Let X be represented by the diagram

where the figure-eight knot is represented by the edge of the central tetrahedron connecting the maximal and next to maximal vertices. Choosing an orientation, the diagram consists of two positive tetrahedra  $T_1, T_3$  and one negative  $T_2$ .  $\partial X = \emptyset$  and combinatorially we have  $\Delta_0(X) = \{*\}, \Delta_1(X) = \{x, y, z, x'\}$ . The gluing of the tetrahedra is vertex order preserving which means that edges are glued together in the following manner.

$$\begin{split} & x = x_{01}^{1} = x_{03}^{1} = x_{02}^{2} = x_{02}^{3} = x_{03}^{3}, \\ & y = x_{02}^{1} = x_{12}^{1} = x_{13}^{1} = x_{01}^{2} = x_{03}^{2} = x_{23}^{2} = x_{23}^{2}, \\ & z = x_{13}^{1} = x_{12}^{2} = x_{13}^{2} = x_{12}^{3} = x_{13}^{3}, \\ & x' = x_{01}^{3}. \end{split}$$

This results in the following equations for the dihedral angles when we balance all but one edge

 $b_1 + a_3 = b_2, \quad a_1 = a_2 + a_3.$ 

In the limit where we let  $a_3 \rightarrow 0$  we get the equations

$$b_1 = b_2, \quad a_1 = a_2.$$

60

The Boltzmann weights are given by the functions

$$B\left(T_{1}, x_{|_{\Delta_{1}(T_{1})}}\right) = g_{a_{1},c_{1}}(y - x, 2y - x - z),$$
  

$$B\left(T_{2}, x_{|_{\Delta_{1}(T_{2})}}\right) = \overline{g_{a_{2},c_{2}}(x - y, x + z - 2y)},$$
  

$$B\left(T_{3}, x_{|_{\Delta_{1}(T_{3})}}\right) = g_{a_{3},c_{3}}(0, x + z - x' - y).$$

So we get that

$$\begin{split} Z_{\hbar}^{\mathrm{new}}(X) &= \int_{[0,1]^4} \sum_{m,n,l \in \mathbb{Z}} \tilde{\psi}'_{a_1,c_1}(y-x+m) e^{\pi i (2y-x-z)(y-x+2m)} \\ &\overline{\psi}'_{a_2,c_2}(x-y+n) e^{-\pi i (x+z-2y)(x-y+2n)} \\ & \tilde{\psi}'_{a_3,c_3}(l) e^{2\pi i (x+z-x'-y)l} \, dx dy dz dx'. \end{split}$$

Integration over x' removes one of the sums since  $\int_0^1 e^{-2\pi i x' l} dx' = \delta(l)$ . Hence

$$\begin{split} Z_{\hbar}^{\mathrm{new}}(X) &= \tilde{\psi}'_{a_{3},c_{3}}(0) \int_{[0,1]^{3}} \sum_{m,n \in \mathbb{Z}} \tilde{\psi}'_{a_{1},c_{1}}(y-x+m) e^{\pi i(2y-x-z)(y-x+2m)} \\ &= \tilde{\psi}'_{a_{3},c_{3}}(0) \int_{[0,1]^{3}} \sum_{m,n \in \mathbb{Z}} \tilde{\psi}'_{a_{1},c_{1}}(y-x+m) \overline{\psi}'_{a_{2},c_{2}}(x-y+n) \\ &= e^{2\pi i(2y-x)(m+n)} e^{2\pi i z(m+n)} dxdydz. \end{split}$$

Now integration over z gives  $\int_0^1 e^{-2\pi i z(m+n)} dz = \delta(n+m).$  So the partition function takes the form

$$Z_{\hbar}^{\text{new}}(X) = \tilde{\psi}'_{a_3,c_3}(0) \int_{[0,1]^2} \sum_{m \in \mathbb{Z}} \tilde{\psi}'_{a_1,c_1}(y-x+m) \overline{\tilde{\psi}'_{a_2,c_2}(x-y-m)} \, dx dy,$$

We make the shift  $y \mapsto y + x$  to get the expression

$$\begin{split} Z_{\hbar}^{\text{new}}(X) &= \tilde{\psi}'_{a_3,c_3}(0) \int_{[0,1]^2} \sum_{m \in \mathbb{Z}} \tilde{\psi}'_{a_1,c_1}(y+m) \overline{\tilde{\psi}'_{a_2,c_2}(-y-m)} \, dx dy \\ &= \tilde{\psi}'_{a_3,c_3}(0) \int_{[0,1]} \sum_{m \in \mathbb{Z}} \tilde{\psi}'_{a_1,c_1}(y+m) \overline{\tilde{\psi}'_{a_2,c_2}(-y-m)} \, dy \\ &= \tilde{\psi}'_{a_3,c_3}(0) \int_{\mathbb{Z}} \tilde{\psi}'_{a_1,c_1}(y) \overline{\tilde{\psi}'_{a_2,c_2}(-y)} \, dy \\ &= e^{-\frac{\pi i}{6}} \tilde{\psi}'_{a_3,c_3}(0) \int_{\mathbb{Z}} \psi_{c_1,b_1}(y) \psi_{b_2,c_2}(y) e^{\pi i y^2}. \end{split}$$

We set  $Y = y - 2c_b(c_1 + b_1) = y - c_b(1 - 2a_1)$ . Assuming that we are in the case where all but one edge is balanced we have  $a_1 = a_2$ 

$$y^{2} = Y^{2} + c_{\rm b}^{2}(1 - 2a_{1})^{2} + 2c_{\rm b}Y(1 - 2a_{1}).$$

Implementing this we get the following expression

$$\begin{split} Z_{\hbar}^{\mathrm{new}}(X) &= e^{-\frac{\pi i}{6}} \tilde{\psi}'_{a_{3},c_{3}}(0) \int_{\mathbb{Z}} \psi(Y)\psi(Y) e^{\pi i(Y^{2}+c_{b}^{2}(1-2a_{1})^{2}+2c_{b}Y(1-2a_{1}))} \\ &\quad e^{-4\pi ic_{b}c_{1}(Y+c_{b}(1/2-a_{1}))} \nu_{c_{1},b_{1}} \\ &\quad e^{-4\pi ic_{b}b_{2}(Y+c_{b}(1/2-a_{1}))} \nu_{b_{2},c_{2}} dy \\ &= e^{-\frac{\pi i}{6}} \nu_{c_{1},b_{1}} \nu_{b_{2},c_{2}} \tilde{\psi}'_{a_{3},c_{3}}(0) \int_{\mathbb{Z}-0i} \frac{1}{\Phi(Y)^{2}} e^{\pi iY^{2}} dy \ e^{\frac{i\phi}{\hbar}}. \end{split}$$

This result corresponds exactly to the partition function in the original formulation, see [AK1, Chap. 11]. I.e. in the limit where  $a_3 \rightarrow 0$  we get the renormalised partition function

$$\tilde{Z}_{\hbar}^{\text{new}}(X) := \lim_{a_3 \to 0} \Phi_{\text{b}}(2c_{\text{b}}a_3 - c_{\text{b}}) Z_{\hbar}^{\text{new}}(X) = \frac{e^{-\pi i/12}}{\nu(c_3)} \chi_{4_1}(0).$$

#### 5.3 The complement of the knot 5<sub>2</sub>

Let X be represented by the diagram

Choosing an orientation the diagram consists of three positive tetrahedra. We denote  $T_1, T_2, T_3$  the left, the right an top tetrahedra respectively. The combinatorial data in this case are  $\Delta_0(X) = \{*\}, \ \Delta_1(X) = \{e_0, e_1, e_2\}, \ \Delta_2(X) = \{f_0, f_1, f_2, f_3, f_4, f_5\}$  and  $\Delta_3(X) = \{T_1, T_2, T_3\}$ .

The edges are glued in the following manner

$$\begin{split} e_0 &= x_{02}^1 = x_{12}^1 = x_{13}^2 = x_{23}^2 = x_{01}^3 = x_{23}^3 =: x, \\ e_1 &= x_{03}^1 = x_{23}^1 = x_{02}^2 = x_{03}^2 = x_{03}^3 = x_{13}^3 = x_{12}^3 =: y \\ e_2 &= x_{01}^1 = x_{13}^1 = x_{01}^2 = x_{12}^2 = x_{02}^3 =: z. \end{split}$$

We impose the condition that all edges are balanced which exactly corresponds to the two equations

$$2a_3 = a_1 + c_2, \quad b_3 = c_1 + b_2.$$

The Boltzmann weights are given by the equations

$$\begin{split} &B\left(T_{1}, x_{\mid \Delta_{1}(T_{1})}\right) = g_{a_{1},c_{1}}(z-y, x-y), \\ &B\left(T_{2}, x_{\mid \Delta_{1}(T_{2})}\right) = g_{a_{2},c_{2}}(x-z, y-z), \\ &B\left(T_{2}, x_{\mid \Delta_{1}(T_{2})}\right) = g_{a_{3},c_{3}}(z-y, z+y-2x). \end{split}$$

We calculate the following function

$$\begin{split} Z_{\hbar}^{\mathrm{new}}(X) &= \int_{[0,1]^3} \sum_{j,k,l \in \mathbb{Z}} g_{a_1,c_1}(z-y,x-y) g_{a_2,c_2}(x-z,y-z) \\ &\times g_{a_3,c_3}(z-y,z+y-2x) \; dx dy dz \\ &= \int_{[0,1]^3} \sum_{j,k,l \in \mathbb{Z}} \tilde{\psi}'_{a_1,c_1}(z-y+j) e^{\pi i (x-y)(z-y+2j)} \tilde{\psi}'_{a_2,c_2}(x-z+k) \\ &e^{\pi i (y-z)(x-z+2k)} \times \tilde{\psi}'_{a_3,c_3}(z-y+l) e^{\pi i (z+y-2x)(z-y+2l)} \; dx dy dz. \end{split}$$

Shift  $x \mapsto x + z$ ,

$$\begin{split} Z_{\hbar}^{\mathrm{new}}(X) = & \int_{[0,1]^3} \sum_{j,k,l \in \mathbb{Z}} \tilde{\psi}'_{a_1,c_1}(z-y+j) e^{\pi i (x+z-y)(z-y+2j)} \tilde{\psi}'_{a_2,c_2}(x+k) e^{\pi i (y-z)(x+2k)} \\ & \times \tilde{\psi}'_{a_3,c_3}(z-y+l) e^{\pi i (y-2x-z)(z-y+2l)} \; dx dy dz. \end{split}$$

Shift  $z \mapsto z + y$ 

$$\begin{split} Z_{\hbar}^{\mathrm{new}}(X) &= \int_{[0,1]^3} \sum_{j,k,l \in \mathbb{Z}} \tilde{\psi}'_{a_1,c_1}(z+j) e^{\pi i(x+z)(z+2j)} \tilde{\psi}'_{a_2,c_2}(x+k) e^{\pi i(-z)(x+2k)} \\ &\times \tilde{\psi}'_{a_3,c_3}(z+l) e^{\pi i(-2x-z)(z+2l)} \, dx dy dz \\ &= \int_{[0,1]^3} \sum_{j,k,l \in \mathbb{Z}} \tilde{\psi}'_{a_1,c_1}(z+j) \tilde{\psi}'_{a_2,c_2}(x+k) \tilde{\psi}'_{a_3,c_3}(z+l) \\ &\times e^{\pi i(x+z)(z+2j)} e^{\pi i(-z)(x+2k)} e^{\pi i(-2x-z)(z+2l)} \, dx dy dz \\ &= \int_{[0,1]^3} \sum_{j,k,l \in \mathbb{Z}} \tilde{\psi}'_{a_1,c_1}(z+j) \tilde{\psi}'_{a_2,c_2}(x+k) \tilde{\psi}'_{a_3,c_3}(z+l) \\ &\times e^{2\pi i (x(j-2l-z)+z(j-k-l)} \, dx dy dz. \end{split}$$

Integration over y contributes nothing. We now shift  $x\mapsto x-k$  and integrate over the interval [-k,-k+1].

$$Z_{\hbar}^{\mathrm{new}}(X) = \sum_{j,k,l \in \mathbb{Z}} \int_{[0,1]} \int_{[-k,-k+1]} \tilde{\psi}'_{a_1,c_1}(z+j) \tilde{\psi}'_{a_2,c_2}(x) \tilde{\psi}'_{a_3,c_3}(z+l)$$

J. E. Andersen J. J. K. Nissen

$$\begin{split} & \times e^{2\pi i ((x-k)(j-2l-z)+z(j-k-l)} \, dx dz \\ & = \sum_{j,k,l\in\mathbb{Z}} e^{2\pi i k(2l-j)} \int_{[0,1]} \int_{[-k,-k+1]} \tilde{\psi}'_{a_1,c_1}(z+j) \tilde{\psi}'_{a_2,c_2}(x) \tilde{\psi}'_{a_3,c_3}(z+l) \\ & \times e^{2\pi i (x(j-2l-z)+z(j-l))} \, dx dz \\ & = \sum_{j,l\in\mathbb{Z}} \int_{[0,1]} \tilde{\psi}'_{a_1,c_1}(z+j) \tilde{\psi}'_{a_3,c_3}(z+l) e^{2\pi i z(j-l)} \int_{\mathbb{R}} \tilde{\psi}'_{a_2,c_2}(x) e^{-2\pi i x(z+2l-j)} dx \, dz \\ & = e^{-\frac{\pi i}{12}} \sum_{j,l\in\mathbb{Z}} \int_{[0,1]} \tilde{\psi}'_{a_1,c_1}(z+j) \tilde{\psi}'_{a_3,c_3}(z+l) e^{2\pi i z(j-l)} \int_{\mathbb{R}} \psi_{c_2,b_2}(x) e^{-2\pi i x(z+2l-j)} dx \, dz \\ & = e^{-\frac{\pi i}{4}} \sum_{j,l\in\mathbb{Z}} \int_{[0,1]} \psi_{c_1,b_1}(z+j) \psi_{c_3,b_3}(z+l) \tilde{\psi}_{c_2,b_2}(z+2l-j) e^{2\pi i z(j-l)} dz. \end{split}$$

We set m = j - l.

$$\begin{split} Z_{\hbar}^{\text{new}}(X) = & e^{-\frac{\pi i}{4}} \sum_{l,m \in \mathbb{Z}} \int_{[0,1]} \psi_{c_{1},b_{1}}(z+l+m) \psi_{c_{3},b_{3}}(z+l) \tilde{\psi}_{c_{2},b_{2}}(z+l-m) e^{2\pi i z m} \, dz \\ = & e^{-\frac{\pi i}{4}} \sum_{l,m \in \mathbb{Z}} \int_{[l,l+1]} \psi_{c_{1},b_{1}}(z+m) \psi_{c_{3},b_{3}}(z) \tilde{\psi}_{c_{2},b_{2}}(z-m) e^{2\pi i z m} \, dz \\ = & e^{-\frac{\pi i}{3}} \sum_{m \in \mathbb{Z}} \int_{\mathbb{R}} \psi_{c_{1},b_{1}}(z+m) \psi_{c_{3},b_{3}}(z) \psi_{b_{2},a_{2}}(z-m) e^{\pi i (z-m)^{2}} e^{2\pi i z m} \, dz \\ = & e^{-\frac{\pi i}{3}} \sum_{m \in \mathbb{Z}} \int_{\mathbb{R}} \psi_{c_{1},b_{1}}(z+m) \psi_{c_{3},b_{3}}(z) \psi_{b_{2},a_{2}}(z-m) e^{\pi i (z^{2}+m^{2})} \, dz. \end{split}$$

$$\begin{split} Z_{\hbar}^{\mathrm{new}}(X) &= e^{-\frac{\pi i}{3}} \sum_{m \in \mathbb{Z}} \int_{\mathbb{R}}^{} \psi(z+m-c_{\mathrm{b}}(1-2a_{1}))e^{-4\pi i c_{\mathrm{b}}c_{1}\{(z+m)-c_{\mathrm{b}}(1/2-a_{1})\}} \\ & e^{-\pi i c_{\mathrm{b}}^{2}(4(c_{1}-b_{1})+1)/6} \\ & \psi(z-m-c_{\mathrm{b}}(1-2c_{2}))e^{-4\pi i c_{\mathrm{b}}c_{1}\{(z+m)-c_{\mathrm{b}}(1/2-c_{2})\}} \\ & e^{-\pi i c_{\mathrm{b}}^{2}(4(c_{1}-b_{1})+1)/6} \\ & \psi(z-c_{\mathrm{b}}(1-2a_{3}))e^{-4\pi i c_{\mathrm{b}}c_{3}\{(z+m)-c_{\mathrm{b}}(1/2-a_{3})\}} \\ & e^{-\pi i c_{\mathrm{b}}^{2}(4(c_{3}-b_{3})+1)/6}e^{\pi i z^{2}}e^{\pi i p^{2}} \, dz. \end{split}$$

Set  $w = z - c_{\rm b}(1 - 2a_3)$ 

$$Z_{\hbar}^{\text{new}}(X) = e^{-\frac{\pi i}{3}} \sum_{m \in \mathbb{Z}} \int_{\mathbb{R} - i0} \psi(w + m + 2c_{\text{b}}(a_1 - a_3))\psi(w - m + 2c_{\text{b}}(c_2 - a_3))\psi(w)$$
$$\times e^{\pi i p^2} e^{\pi i w^2} e^{4\pi i c_{\text{b}}^2 (1/2 - a_3)^2} e^{4\pi i c_{\text{b}} w(1/2 - a_3)}$$
$$e^{-4\pi i c_{\text{b}} c_1 \{w + p + c_{\text{b}}(1 - 2a_3) - c_{\text{b}}(1/2 - a_1)\}}$$

Asymptotic aspects of the Teichmüller TQFT

$$\begin{split} & e^{-4\pi i c_{\rm b} c_1 \{w-p+c_{\rm b}(1-2a_3)-c_{\rm b}(1/2-c_2)\}} \\ & e^{-4\pi i c_{\rm b} c_1 \{w+c_{\rm b}(1/2-a_3)\}} \nu_{c_1,b_1} \nu_{b_2,a_2} \nu_{c_3,b_3} \ dw. \end{split}$$

Simplify by setting  $u = 2c_b(a_1 - a_3)$ . Using  $c_1 + b_2 + c_3 + a_3 - 1/2 = 0$  we are left with

$$Z_{\hbar}^{\text{new}}(X) = e^{-\frac{\pi i}{3}} \sum_{m \in \mathbb{Z}} \int_{\mathbb{R}^{-i0}} \psi(w + m + u) \psi(w - m - u) \psi(w)$$
$$e^{\pi i w^2} e^{\pi i m^2} e^{4\pi i c_{\rm b}(b_2 - c_1)m}$$
$$e^{-4\pi i c_{\rm b}^2 \{-b_3^2 - b_3 c_3 + c_1(b_3 + c_3) + b_2(b_3 + c_3) + (c_1 - b_2)(a_1 - a_3)\}} dw$$
$$\nu_{c_1, b_1} \nu_{b_2, a_2} \nu_{c_3, b_3}.$$

Let  $v = 2c_b(a_1 - c_1 + b_2 - a_3)$ , then Note that

$$4\pi i c_{\rm b}(b_2 - c_1)p = 4\pi i c_{\rm b}(a_1 - c_1 + b_2 - a_3)p - 4\pi i c_{\rm b}(a_3 - a_1) = 2\pi i (vp - up),$$
$$-b_3^2 - b_3 c_3 + c_1(b_3 + c_3) + b_2(b_3 + c_3) = 0,$$

and

$$-4\pi i c_{\rm b}^2((c_1-b_2)(a_1-a_3)) = 4\pi i c_{\rm b}^2((a_1-c_1+b_2-a_3)(a_1-a_3)-(a_1-a_3)(a_1-a_3))$$
  
=  $\pi i (vu-u^2).$ 

$$\begin{split} Z_{\hbar}^{\text{new}}(X) = & e^{-\frac{\pi i}{3}} e^{\pi i u v} \sum_{m \in \mathbb{Z}} \int_{\mathbb{R} - i0} \frac{e^{\pi i w^2} e^{-\pi i m^2} e^{-\pi i u^2}}{\Phi_{\text{b}}(w + m + u) \Phi_{\text{b}}(w - m - u) \Phi_{\text{b}}(w)} dw e^{2\pi i v m} \\ & \nu_{c_1, b_1} \nu_{b_2, a_2} \nu_{c_3, b_3} \\ = & e^{-\frac{\pi i}{3}} e^{\pi i u v} \sum_{m \in \mathbb{Z}} \int_{\mathbb{R} - i0} \frac{e^{\pi i (w + (u + m))(w - (u + m))}}{\Phi_{\text{b}}(w + m + u) \Phi_{\text{b}}(w - m - u) \Phi_{\text{b}}(w)} dw e^{2\pi i v m} \\ & \nu_{c_1, b_1} \nu_{b_2, a_2} \nu_{c_3, b_3} \\ = & W \chi_{5_2}(u, v) \nu_{c_1, b_1} \nu_{b_2, a_2} \nu_{c_3, b_3}. \end{split}$$

Where  $\chi_{5_2}(u)$  is given by the formula

$$\chi_{5_2}(u) = e^{-\frac{\pi i}{3}} \int_{\mathbb{R}-i0} \frac{e^{\pi i (w-u)(w+u)}}{\Phi_{\rm b}(w+m+u)\Phi_{\rm b}(w-m-u)\Phi_{\rm b}(w)} \, dw.$$

Again the function  $\chi_{5_2}$  is that of [AK1], which again is related to Hikami's invariant, in particular Hikami's formally derived expression in [H2, (4.10)] is equal to  $e^{\pi i \frac{c_{b2}}{3}} \frac{1}{2\pi b} \chi_{5_2}(\frac{-u}{\pi b}, \frac{1}{2})$ , where  $\chi_{5_2} := \chi_{5_2}(x) e^{4\pi i c_b x \lambda}$ .

# **5.4** One vertex H-triangulation of $(S^3, 5_2)$

Let X be represented by the diagram

Choosing an orientation, the diagram consists of four positive tetrahedra  $T_0, T_1, T_2, T_3$ .  $\partial X = \emptyset$  and combinatorially we have  $\Delta_0(X) = \{*\}, \Delta_1(X) = \{x, y, z, w, x'\}$ . The gluing of the tetrahedra is vertex order preserving which means that edges are glued together in the following manner.

$$\begin{split} x &= x_{03}^0 = x_{13}^0 = x_{11}^1 = x_{12}^3 = x_{02}^3, \\ y &= x_{03}^1 = x_{12}^1 = x_{13}^1 = x_{02}^2 = x_{03}^2 = x_{03}^3 = x_{23}^3, \\ z &= x_{01}^0 = x_{02}^1 = x_{01}^2 = x_{12}^2 = x_{01}^3 = x_{13}^3, \\ v &= x_{02}^0 = x_{12}^0 = x_{13}^1 = x_{13}^2 = x_{23}^2, \\ x' &= x_{23}^0. \end{split}$$

This results in the following equations for the dihedral angles, when we balance all edges but one edge.

$$a_3 = a_1 - a_0 = c_2$$
,  $a_0 + b_1 = b_2 + c_3$ ,  $a_1 + a_2 + b_3 = \frac{1}{2} + c_1$ .

The Boltzmann weights are given by the functions

$$\begin{split} B\left(T_{0}, x_{\mid \Delta_{1}(T_{0})}\right) &= g_{a_{0},c_{0}}(0, v + x - z - x'),\\ B\left(T_{1}, x_{\mid \Delta_{1}(T_{1})}\right) &= g_{a_{1},c_{1}}(z - y, z + y - x - v),\\ B\left(T_{2}, x_{\mid \Delta_{1}(T_{2})}\right) &= g_{a_{2},c_{2}}(v - z, y - z),\\ B\left(T_{3}, x_{\mid \Delta_{1}(T_{3})}\right) &= g_{a_{3},c_{3}}(z - y, x - y). \end{split}$$

The partition function is represented by the integral

$$\begin{split} Z_{\hbar}^{\text{new}}(X) = \int_{[0,1]^5} \sum_{m,n,k,p \in \mathbb{Z}} \tilde{\psi}'_{a_0,c_0}(m) e^{\pi i (v+x-z-x')(2m)} \\ \tilde{\psi}'_{a_1,c_1}(z-y+n) e^{\pi i (z+y-x-v)(z-y+2n)} \\ \tilde{\psi}'_{a_2,c_2}(v-z+k) e^{\pi i (y-z)(v-z+2k)} \\ \tilde{\psi}'_{a_3,c_3}(z-y+p) e^{\pi i (x-y)(z-y+2p)} \, dx' dx dy dz dv \end{split}$$
Integration over x' removes one of the sums since  $\int_0^1 e^{-2\pi i x' m} dx' = \delta(m).$  Hence

Now integration over x gives  $\int_0^1 e^{-2\pi i x(n-p)} dx = \delta(n-p)$ . Implementing this and shifting the variable  $v \mapsto v + z$ , the partition function takes the form

We make the shift  $z \mapsto z + y$  to get the expression

which is independent of y so we can remove the integration over this variable. We integrate over the variable v.

$$\begin{split} \sum_{k\in\mathbb{Z}} \int_{[0,1]} \tilde{\psi}'_{a_2,c_2}(v+k) e^{-2\pi i v(z+n)} dv e^{-2\pi i zk} &= \sum_{k\in\mathbb{Z}} \int_k^{k+1} \tilde{\psi}'_{a_2,c_2}(v) e^{-2\pi i v(z+n)} dv \\ &e^{-2\pi i zk} e^{2\pi i k(z+n)} \\ &= e^{-\frac{\pi i}{12}} \int_{\mathbb{R}} \psi_{c_2,b_2}(v) e^{-2\pi i z(z+n)} dv \\ &= e^{-\frac{\pi i}{12}} \tilde{\psi}_{c_2,b_2}(z+n) \\ &= e^{-\frac{\pi i}{6}} e^{\pi i (z+n)^2} \psi_{b_2,a_2}(z+n). \end{split}$$

We therefore get the expression

$$Z_{\hbar}^{\text{new}}(X) = e^{-\frac{\pi i}{3}} \tilde{\psi}'_{a_0,c_0}(0) \int_{[0,1]} \sum_{n \in \mathbb{Z}} \psi_{c_1,b_1}(z+n) \psi_{b_2,a_2}(z+n) \psi_{c_3,b_3}(z+n) e^{\pi i (z+n)^2} dz$$
$$= e^{-\frac{\pi i}{3}} \tilde{\psi}'_{a_0,c_0}(0) \int_{\mathbb{R}} \psi_{c_1,b_1}(z) \psi_{b_2,a_2}(z) \psi_{c_3,b_3}(z) e^{\pi i (z)^2}$$

We set  $Z = z - 2c_b(c_1 + b_1) = y - c_b(1 - 2a_1)$ . Assuming that we are in the case where all but the edge representing the knot is balanced, i.e.  $a_0 \rightarrow 0$ , we have  $a_1 = c_2 = a_3$ .

$$z^{2} = Z^{2} + c_{\rm b}^{2}(1 - 2a_{1})^{2} + 2c_{\rm b}Z(1 - 2a_{1}).$$

Implementing this we get the expression.

$$Z_{\hbar}^{\text{new}}(X) = e^{-\frac{\pi i}{3}} \tilde{\psi}'_{a_0,c_0}(0) \int_{\mathbb{R}} \psi(Z) \psi(Z) \psi(Z) e^{\pi i (Z^2 + c_{\mathrm{b}}^2 (1 - 2a_1)^2 + 2c_{\mathrm{b}} Z(1 - 2a_1))} e^{-4\pi i c_{\mathrm{b}} c_1 (Z + c_{\mathrm{b}} (1/2 - a_1))} \nu_{c_1,b_1} e^{-4\pi i c_{\mathrm{b}} b_2 (Z + c_{\mathrm{b}} (1/2 - a_2))} \nu_{b_2,a_2} e^{-4\pi i c_{\mathrm{b}} c_2 (Z + c_{\mathrm{b}} (1/2 - a_3))} \nu_{c_3,b_2,3} dz.$$

$$\begin{split} Z_{\hbar}^{\text{new}}(X) &= \nu_{c_1,b_1} \nu_{b_2,a_2} \nu_{c_3,b_3} e^{-\frac{\pi i}{3}} e^{\frac{\phi i}{\hbar}} \tilde{\psi}'_{a_0,c_0}(0) \int_{\mathbb{R}} \psi(Z)^3 e^{\pi i Z^2} \ dz \\ &= \nu_{c_1,b_1} \nu_{b_2,a_2} \nu_{c_3,b_3} e^{-\frac{\pi i}{3}} e^{\frac{\phi i}{\hbar}} \tilde{\psi}'_{a_0,c_0}(0) \int_{\mathbb{R}} \frac{e^{\pi i Z^2}}{\Phi_{\mathrm{b}}(Z)^3} \ dz \end{split}$$

Because the combination of dihedral angles in front of Z sums to 0.

$$-4\pi i c_{\rm b} Z(c_1 + b_2 + c_3 - \frac{1}{2} + a_1) = -4\pi i c_{\rm b} Z(a_1 + b_1 + c_1 - \frac{1}{2}) = 0$$

This corresponds to the partition function in the original formulation, see [AK1].

In this case the renormalised partition function takes the form

$$\tilde{Z}_{\hbar}^{\text{new}}(X) = \lim_{a_0 \to 0} \Phi_{\text{b}} Z_{\hbar}^{\text{new}}(X) = \frac{e^{i\pi/4}}{\nu_{c_{0,0}}} \chi_{5_2}(0).$$

### **5.5** One-vertex H-triangulation of $(S^3, 6_1)$

Let X be represented by the diagram



This one vertex H-triangulation of  $(S^3, 6_1)$  consists of 5 tetrahedra  $T_1$  and  $T_3$  which are negatively oriented tetrahedra and  $T_2, T_4, T_5$  which are positively oriented tetrahedra. We denoted the tetrahedra as follows. In the bottom we have  $T_1, T_2, T_3$  from left to right and on top we have  $T_4, T_5$  from right to left.

In the diagram the knot  $6_1$  is represented by the edge connecting the maximal and next to maximal vertex of  $T_1$ . We impose a shape structure on the triangulation and balance all but the one edge representing the knot. We get the following equations on the shape parameters

$$a_3 = a_1 + c_2, \quad a_3 + a_4 = a_1 + a_5, \quad a_1 + c_2 = c_4 + c_5,$$
  
$$\frac{1}{2} + b_3 + c_5 = a_2 + a_3 + a_4, \quad 1 = a_2 + c_3 + c_4 + a_5.$$

We calculate the partition function for the Teichmüller TQFT using the original formulation of the theory. In this formulation the states are assigned to each face of each tetrahedron according to the diagram (5.8).

$$Z_{\hbar}(X) = \int_{\mathbb{R}^{10}} \overline{\langle w, t \mid T_{a_1,c_1} \mid u, t \rangle} \langle z, q \mid T_{a_2,c_2} \mid v, u \rangle \overline{\langle x, q \mid T_{a_3,c_3} \mid r, v \rangle} \\ \langle s, y \mid T_{a_4,c_4} \mid r, z \rangle \langle w, x \mid T_{a_5,c_5} \mid y, s \rangle \, d\bar{x}$$

$$\begin{split} Z_{\hbar}(X) &= \int_{\mathbb{R}^{10}} \delta(w+t-u) \delta(z+q-v) \delta(x+q-r) \delta(s+y-r) \delta(w+x-y) \\ & \overline{\tilde{\psi}'_{a_{1},c_{1}}(0)} e^{-2\pi i w(0)} \\ & \overline{\tilde{\psi}'_{a_{2},c_{2}}(u-q)} e^{2\pi i z(u-q)} \\ & \overline{\tilde{\psi}'_{a_{3},c_{3}}(v-q)} e^{-2\pi i x(v-q)} \\ & \overline{\tilde{\psi}'_{a_{3},c_{3}}(v-q)} e^{2\pi i s(z-y)} \\ & \overline{\tilde{\psi}'_{a_{5},c_{5}}(s-x)} e^{2\pi i w(s-x)} \, dq dr ds dt du dv dx dw dz dy \end{split}$$

Integrating over five variables t, v, r, y, w yields the expression

$$Z_{\hbar}(X) = \overline{\tilde{\psi}'_{a_{1},c_{1}}(0)} \int_{\mathbb{R}^{5}} \widetilde{\psi}'_{a_{2},c_{2}}(u-q)e^{2\pi i z(u-q)} \\ \times \overline{\tilde{\psi}'_{a_{3},c_{3}}(z)}e^{-2\pi i x z} \\ \times \widetilde{\psi}'_{a_{4},c_{4}}(z+s-x-q)e^{2\pi i s(z+s-x-q)} \\ \times \widetilde{\psi}'_{a_{5},c_{5}}(s-x)e^{2\pi i (q-s)(s-x)} \, dq ds du dx dz$$

We integrate over the variable u using the Fourier transform.

$$e^{-\frac{\pi i}{12}} \int_{\mathbb{R}} \psi_{c_2,b_2}(u-q) e^{2\pi i z(u-q)} du = e^{-\frac{\pi i}{12}} \tilde{\psi}_{c_2,b_2}(-z) = e^{-\frac{\pi i}{6}} e^{\pi i z^2} \psi_{b_2,a_2}(-z).$$

Using formulas from Section 3.8 we can write

$$Z_{\hbar}(X) = e^{-\frac{3\pi i}{12}} \overline{\psi}'_{a_1,c_1}(0) \int_{\mathbb{R}^4} \psi_{b_2,a_2}(-z) e^{\pi i z^2} \psi_{b_3,c_3}(-z) e^{\pi i z^2}$$

J. E. Andersen J. J. K. Nissen

$$\begin{split} \tilde{\psi}'_{a_4,c_4}(z+s-x-q)\tilde{\psi}'_{a_5,c_5}(s-x) \\ e^{2\pi i(sz-xz-qx)} \; dqdsdxdz. \end{split}$$

Integration over the variable q gives

$$\int_{\mathbb{R}} \tilde{\psi}'_{a_4,c_4}(z+s-x-q)e^{-2\pi i q x} dq = e^{-\frac{\pi i}{12}} \tilde{\psi}_{c_4,b_4}(-x)e^{2\pi i (x^2-xz-sx)}$$
$$= e^{-\frac{\pi i}{6}} \psi_{b_4,a_4}(-x)e^{2\pi i (\frac{3}{2}x^2-xz-sx)}$$

$$Z_{\hbar}(X) = e^{-\frac{5\pi i}{12}} \overline{\tilde{\psi}'_{a_1,c_1}(0)} \int_{\mathbb{R}^2} \psi_{b_2,a_2}(-z) \psi_{b_3,c_3}(-z) \psi_{b_4,a_4}(-x) \tilde{\psi}'_{a_5,c_5}(s-x) e^{2\pi i (sz-2xz+z^2+\frac{3}{2}x^2-sx)} dx ds dz$$

Integration over s now gives

$$e^{-\frac{\pi i}{12}} \int_{\mathbb{R}} \psi_{c_5, b_5}(s-x) e^{-2\pi i s(x-z)} ds = e^{-\frac{\pi i}{12}} \tilde{\psi}_{c_5, b_5}(x-z) e^{-2\pi i (x^2-xz)}$$
$$= e^{-\frac{\pi i}{6}} \psi_{b_5, a_5}(x-z) e^{\pi i (x-z)^2} e^{-2\pi i (x^2-xz)}.$$

So the partition function takes the form

$$Z_{\hbar}(X) = e^{-\frac{7\pi i}{12}} \tilde{\psi}'_{a_1,c_1}(0) \int_{\mathbb{R}^2} \psi_{b_2,a_2}(-z) \psi_{b_3,c_3}(-z) \psi_{b_4,a_4}(-x) \psi_{b_5,a_5}(x-z) e^{2\pi i (-2xz+\frac{3}{2}z^2+x^2)} \, dx dz.$$

which is equivalent to

(5.9) 
$$Z_{\hbar}(X) = e^{-\frac{7\pi i}{12}} \overline{\tilde{\psi}'_{a_1,c_1}(0)} \int_{\mathbb{R}^2} \psi_{b_2,a_2}(z) \psi_{b_3,c_3}(z) \psi_{b_4,a_4}(-x) \psi_{b_5,a_5}(x+z) e^{2\pi i (2xz+\frac{3}{2}z^2+x^2)} dx dz.$$

Set  $\tilde{z} = z - c_b(1 - 2c_2)$  and  $-\tilde{x} = -x - c_b(1 - 2c_4)$ . Then  $z - c_b(1 - 2a_3) = \tilde{z} + c_b(1 - 2c_2) - c_b(1 - 2a_3) = \tilde{z}$ ,

because  $a_3 \rightarrow c_2$  in the limit where  $a_1 \rightarrow 0$ . Furthermore we have  $x + z - c_b(1 - 2c_5) = \tilde{x} - c_b(1 - 2c_4) + \tilde{z} + c_b(1 - 2c_2) - c_b(1 - 2c_5) = \tilde{x} + \tilde{z} - c_b$ because

$$c_4 + c_5 - c_2 \to 0$$

when  $a_1 \rightarrow 0$ . We can now write the partition function in the following way

$$\begin{split} Z_{\hbar}(X) &= e^{-\frac{7\pi i}{12}} \int_{\mathbb{R}^{2}} \psi(\tilde{z})\psi(\tilde{z})\psi(-\tilde{x})\psi(\tilde{x}+\tilde{z}-c_{\rm b})\overline{\psi'_{a_{1},c_{1}}(0)} \\ & e^{-4\pi i(-\tilde{z}-c_{\rm b}(1-2c_{2}))(\tilde{x}-c_{\rm b}(1-2c_{4}))+3\pi i(-\tilde{z}-c_{\rm b}(1-2c_{2}))^{2}+2\pi i(\tilde{x}-c_{\rm b}(1-2c_{4}))^{2}} \\ & e^{-4\pi ic_{\rm b}b_{2}(\tilde{z}+c_{\rm b}(1/2-c_{2}))}\nu_{a_{2},b_{2}} \\ & e^{-4\pi ic_{\rm b}b_{3}(\tilde{z}+c_{\rm b}(1-2c_{2})-c_{\rm b}(1/2-a_{3}))}\nu_{b_{3},c_{3}} \\ & e^{-4\pi ic_{\rm b}b_{4}(-\tilde{x}-c_{\rm b}(1/2-c_{4}))}\nu_{b_{4},a_{4}} \\ & e^{-4\pi ic_{\rm b}b_{5}(\tilde{x}+\tilde{z}-c_{\rm b}(1-2c_{4})+c_{\rm b}(1-2c_{2})-c_{\rm b}(1/2-c_{5}))}\nu_{b_{5},a_{5}}d\tilde{x}d\tilde{z} \end{split}$$

In front of  $\tilde{z}$  in the exponent we have the factor

$$\begin{aligned} &4\pi i c_{\rm b} (-1+2c_4+3/2-3c_2-b_2-b_3-b_5) \\ &= 4\pi i c_{\rm b} (1/2+2c_4-2c_2-b_2-a_3-b_3-b_5) \\ &= 4\pi i c_{\rm b} (1/2+2c_4-c_2-1/2+a_2-1/2+c_3-1/2+a_5+c_5) \\ &= 4\pi i c_{\rm b} (-1+1+c_4-c_2+c_5) = 0. \end{aligned}$$

In front of  $\tilde{x}$  in the exponent we also have the factor 0 since

$$-b_5 + b_4 + 1 - 2c_2 - 1 + 2c_4 = -\frac{1}{2} + a_5 + c_5 + c_4 + b_4 + c_4 - 2c_2$$
$$= -\frac{1}{2} + a_5 + \frac{1}{2} - a_4 - c_2$$
$$= a_5 - a_4 - a_3 = 0.$$

This gives us the partition function

$$Z_{\hbar}(X) = e^{i\frac{\phi}{\hbar}} e^{-\frac{7\pi i}{12}} \overline{\psi}'_{a_1,c_1}(0) \int_{\mathbb{R}^2} \psi(\tilde{z})\psi(\tilde{z})\psi(-\tilde{x})\psi(\tilde{x}+\tilde{z}-c_{\rm b})$$
$$e^{2\pi i(\frac{3}{2}\tilde{z}^2+\tilde{x}^2+2\tilde{x}\tilde{z})} d\tilde{x}d\tilde{z},$$

where  $\phi$  is a real quadratic polynomial of dihedral angles. Finally, we do the shift  $\tilde{x} \mapsto \tilde{x} - \tilde{z} + c_{\rm b}$  and get the expression

$$Z_{\hbar}(X) = \zeta_{inv}^2 e^{2\pi i c_{\rm b}^2} e^{i\frac{\phi}{\hbar}} e^{-\frac{7\pi i}{12}} \overline{\psi}'_{a_1,c_1}(0) \int_{\mathbb{R}^2} \frac{\Phi_{\rm b}(\tilde{z}) \Phi_{\rm b}(\tilde{x})}{\Phi_{\rm b}(-\tilde{z}) \Phi_{\rm b}(\tilde{x}-\tilde{z}-c_{\rm b})} e^{\pi i \tilde{x}^2 + 4\pi c_{\rm b} x} d\tilde{x} d\tilde{z}.$$

which exactly corresponds to the result for an H-triangulation of the  $6_1$  knot in [KLV].

# 5.6 One vertex H-triangulation of $(S^3, 6_1)$ – New formulation

We here calculate the partition function for the H-triangulation of the knot  $6_1$  using the new formulation of the TQFT from quantum Teichmüller theory.

The gluing pattern of faces and edges in diagram (5.8) results in the following states

$$\begin{split} & x := x_{02}^1 = x_{03}^1 = x_{01}^2 = x_{02}^2 = x_{01}^3, \\ & y := x_{03}^2 = x_{13}^2 = x_{02}^3 = x_{03}^3 = x_{13}^3 = x_{02}^4 = x_{03}^4 = x_{03}^5, \\ & z := x_{23}^2 = x_{12}^3 = x_{12}^4 = x_{01}^5, \\ & v := x_{12}^1 = x_{13}^1 = x_{23}^3 = x_{23}^4 = x_{52}^5 = x_{53}^5, \\ & w := x_{23}^1 = x_{12}^2 = x_{01}^4 = x_{13}^4 = x_{02}^5 = x_{23}^5, \\ & x' := x_{01}^1. \end{split}$$

The Boltzmann weights for the five tetrahedron are given by

 $\overline{g_{a_1,c_1}(0,x+v-x'-w)}, \quad g_{a_2,c_2}(x-w,y-z), \quad \overline{g_{a_3,c_3}(y-z,2y-x-v)}, \\ g_{a_4,c_4}(w-z,y-v), \quad g_{a_5,c_5}(w-y,v-z+p).$ 

Integration over x' gives  $\delta(k)$ , which removes one of the sums

$$\begin{split} Z_{\hbar}^{\text{New}}(X) = \overline{\tilde{\psi}'_{a_{1},c_{1}}(0)} \int_{[0,1]^{5}} \sum_{l,m,n,p \in \mathbb{Z}} \tilde{\psi}'_{a_{2},c_{2}}(x-w+l) e^{\pi i (y-z)(x-w+2l)} \\ \overline{\tilde{\psi}'_{a_{3},c_{3}}(y-z+m)} e^{-\pi i (2y-x-v)(y-z+2m)} \\ \tilde{\psi}'_{a_{4},c_{4}}(w-z+n) e^{\pi i (y-v)(w-z+2n)} \\ \tilde{\psi}'_{a_{5},c_{5}}(w-y+p) e^{\pi i (v-z)(w-y+p)} dxdydzdvdw \end{split}$$

We do a shift  $x \to x + w$ 

$$\begin{split} Z_{\hbar}^{\mathrm{New}}(X) = \overline{\tilde{\psi}'_{a_{1},c_{1}}(0)} \int_{[0,1]^{5}} \sum_{l,m,n,p \in \mathbb{Z}} \tilde{\psi}'_{a_{2},c_{2}}(x+l) e^{\pi i (y-z)(x+2l)} \\ \overline{\tilde{\psi}'_{a_{3},c_{3}}(y-z+m)} e^{-\pi i (2y-x-w-v)(y-z+2m)} \\ \tilde{\psi}'_{a_{4},c_{4}}(w-z+n) e^{\pi i (y-v)(w-z+2n)} \\ \tilde{\psi}'_{a_{5},c_{5}}(w-y+p) e^{\pi i (v-z)(w-y+p)} dx dy dz dv dw. \end{split}$$

Note that

$$\sum_{l\in\mathbb{Z}} e^{2\pi i l(y-z)} \int_{[0,1]} \tilde{\psi}'_{a_2,c_2}(x+l) e^{-2\pi i x(z-y-m)} dx$$
  
$$= e^{-\frac{\pi i}{12}} \sum_{l\in\mathbb{Z}} \int_{l}^{l+1} \psi_{c_2,b_2}(x) e^{-2\pi i x(z-y-m)} dx$$
  
$$= e^{-\frac{\pi i}{12}} \int_{\mathbb{R}} \psi_{c_2,b_2}(x) e^{-2\pi i x(z-y-m)} dx$$
  
$$= e^{-\frac{\pi i}{12}} \tilde{\psi}_{c_2,b_2}(z-y-m) = e^{-\frac{\pi i}{6}} \psi_{b_2,a_2}(z-y-m) e^{\pi i (z-y-m)^2}.$$

Integration over v removes yet another sum. I.e. n = p + m. We shift  $z \mapsto z + y$ and  $w \mapsto w + y$  and see that the function is independent of y which yields the expression

$$Z_{\hbar}^{\text{New}}(X) = e^{-\frac{\pi i 5}{12}} \overline{\psi}'_{a_1,c_1}(0) \int_{[0,1]^3} \sum_{m,p \in \mathbb{Z}} \psi_{b_2,a_2}(z-m) \psi_{b_3,c_3}(z-m)$$
$$\psi_{c_4,b_4}(w-z+p+m) \psi_{c_5,b_5}(w+p)$$
$$e^{2\pi i (z-m)^2} e^{\pi i w (-z+2m)} e^{-\pi i z (w+2p)} dz dw.$$

$$Z_{\hbar}^{\text{New}}(X) = e^{-\frac{\pi i 5}{12}} \overline{\tilde{\psi}'_{a_1,c_1}(0)} \int_{\mathbb{R}^2} \psi_{b_2,a_2}(z) \psi_{b_3,c_3}(z) \psi_{c_4,b_4}(w-z) \psi_{c_5,b_5}(w) e^{2\pi i z^2} e^{\pi i (w-m)(-z+m)} e^{-\pi i (z+m)(w+p)} dz dw.$$

Now let  $f(z) := \psi_{b_2,a_2}(z)\psi_{b_3,c_3}(z)$ . We then calculate

$$\begin{split} Z_{\hbar}^{\text{New}}(X) &= e^{-\frac{\pi i 5}{12}} \tilde{\psi}'_{a_1,c_1}(0) \int_{\mathbb{R}^2} f(z) \psi_{c_4,b_4}(w-z) \psi_{c_5,b_5}(w) e^{2\pi i z^2} e^{-2\pi i w z} \, dz dw \\ &= e^{-\frac{\pi i 5}{12}} \overline{\tilde{\psi}'_{a_1,c_1}(0)} \int_{\mathbb{R}^2} f(z) \psi_{b_3,c_3}(z) \psi_{c_4,b_4}(w) \psi_{c_5,b_5}(w+z) e^{-2\pi i w z} \, dz dw \\ &= e^{-\frac{\pi i 5}{12}} \overline{\tilde{\psi}'_{a_1,c_1}(0)} \int_{\mathbb{R}^2} f(z) \psi_{c_4,b_4}(w) \tilde{\psi}_{c_5,b_5}(x) e^{2\pi i ((x-z)(w+z)+z^2)} \, dx dz dw \end{split}$$

J. E. Andersen J. J. K. Nissen

$$\begin{split} &= e^{-\frac{\pi i 5}{12}} \tilde{\psi}'_{a_1,c_1}(0) \int_{\mathbb{R}^2} f(z) \psi_{c_4,b_4}(w) \tilde{\psi}_{c_5,b_5}(x+z) e^{2\pi i (x(w+z)+z^2)} \, dx dz dw \\ &= e^{-\frac{\pi i 6}{12}} \overline{\psi}'_{a_1,c_1}(0) \int_{\mathbb{R}^2} f(z) \psi_{c_4,b_4}(w) \psi_{b_5,a_5}(x+z) e^{2\pi i (wx+2xz+\frac{3}{2}z^2+\frac{x^2}{2})} \, dx dz dw \\ &= e^{-\frac{\pi i 6}{12}} \overline{\psi}'_{a_1,c_1}(0) \int_{\mathbb{R}^2} f(z) \tilde{\psi}_{c_4,b_4}(-x) \psi_{b_5,a_5}(x+z) e^{2\pi i (2xz+\frac{3}{2}z^2+\frac{x^2}{2})} \, dz dw \\ &= e^{-\frac{\pi i 7}{12}} \overline{\psi}'_{a_1,c_1}(0) \int_{\mathbb{R}^2} f(z) \psi_{b_4,a_4}(-x) \psi_{b_5,a_5}(x+z) e^{2\pi i (2xz+\frac{3}{2}z^2+x^2)} \, dz dw \end{split}$$

This is the exact same expression as in (5.9) and the two formulations coincide.

# **5.7** Volume of $(S^3, 6_1)$

**Theorem 5.1.** The hyperbolic volume of the complement of  $6_1$  in  $S^3$  is recovered as the following limit

(5.10) 
$$\lim_{\hbar \to 0} 2\pi\hbar \log |J_{s^3, 6_1}(\hbar, 0)| = -\text{Vol}(S^3 \backslash 6_1).$$

*Proof.* We consider the expression

(5.11) 
$$J_{S^{3},6_{1}}(\hbar,0) = \int_{\mathbb{R}^{2}} \frac{\Phi_{\rm b}(x)\Phi_{\rm b}(z)}{\Phi_{\rm b}(-x)\Phi_{\rm b}(z-x-c_{\rm b})} e^{\pi i z^{2} - 4\pi i c_{\rm b} z} dx dz.$$

Using the quasi-classical asymptotic behaviour of Faddeev's quantum dilogarithm shown in Corollary A.6 we can approximate in the following manner. For b close to zero the integral in (A.12) is approximated by the double contour integral

$$J_{S^{3},6_{1}}(\hbar,0) = \frac{1}{(2\pi b)^{2}} \int_{\mathbb{R}^{2}} \frac{\Phi_{b}\left(\frac{x}{2\pi b}\right) \Phi_{b}\left(\frac{z}{2\pi b}\right)}{\Phi_{b}\left(\frac{z-x-c_{b}}{2\pi b}\right)} e^{-\frac{z^{2}}{4\pi i b^{2}} + \frac{y}{b^{2}}} dxdz$$
$$\sim \frac{1}{(2\pi b)^{2}} \int_{\mathbb{R}^{2}} e^{\frac{1}{2\pi i b^{2}}(2\operatorname{Li}_{2}(-e^{x}) + \operatorname{Li}_{2}(-e^{z}) - \operatorname{Li}_{2}(e^{z-x}) - \frac{1}{2}z^{2} + 2\pi i z + \frac{1}{2}x^{2})} dxdz$$
$$= \frac{1}{(2\pi b)^{2}} \int_{\mathbb{R}^{2}} e^{\frac{1}{2\pi i b^{2}}V(x,z)} dxdz,$$

where the potential V is given by

(5.12) 
$$V(x,z) = 2\operatorname{Li}_2(-e^x) + \operatorname{Li}_2(-e^z) - \operatorname{Li}_2(e^{z-x}) - \frac{1}{2}z^2 + 2\pi i z + \frac{1}{2}x^2.$$

It is easily seen that we can treat  $\mathbf{b}^2$  as  $\hbar.$  Therefore, we look for stationary points of the potential V

(5.13) 
$$\frac{\partial V(x,z)}{\partial x} = -2\log(1+e^x) - \log(1-e^{z-x}) + x = \log\frac{e^x}{(1+e^x)^2(1-e^{z-x})}.$$

Asymptotic aspects of the Teichmüller TQFT

(5.14) 
$$\frac{\partial V(x,z)}{\partial z} = -\log(1+e^z) + \log(1-e^{z-x}) - z + 2\pi i = \log\frac{1-e^{z-x}}{(1+e^z)e^z}$$

Stationary points are given by solutions to the equations

(5.15) 
$$e^x = (1 + e^x)^2 (1 - e^{z-x}),$$

(5.16) 
$$(1+e^z)e^z = 1-e^{z-x}.$$

From (5.16) we see that

$$e^x = \frac{1}{e^{-z}-1-e^z}.$$

Inserting in (5.15) we get the equation

$$\frac{1}{e^{-z} - 1 - e^z} = \left(\frac{e^{-z} - e^z}{e^{-z} - 1 - e^z}\right)^2 (1 + e^z)e^z \iff 1 - t - t^2 = (1 - t^2)^2 (1 + t),$$

where we set  $e^z = t$ .

Numerical solutions for the last equation are given by

$$\begin{aligned} t_1 &= -1,39923 - 0,32564i, \\ t_2 &= -1,39923 + 0,32564i, \end{aligned} \qquad \ t_3 &= 0,899232 - 0,400532i, \\ t_4 &= 0,899232 + 0,400532i. \end{aligned}$$

The maximal contribution to the integral comes from the point  $t_2$ . The saddle point method lets us obtain the following limit

$$\lim_{\hbar^2 \to 0} 2\pi \hbar |J_{S^3, 6_1}(\hbar, 0)| = -3.1632... \cdot I = -\operatorname{Vol}(S^3 \backslash 6_1)$$

# 6 The Teichmüller TQFT representation of the mapping class group $\Gamma_{1,1}$

We will here give a representation for the mapping class group of the once punctured torus by the use of the new formulation of the Teichmüller TQFT.

The framed mapping class group  $\Gamma_{1,1}$  is generated by the standard elements S and T. See e.g. Section 6 in [AU3] for a description of these elements (they of course maps to the standard S and T matrix once mapped to the mapping class group of the torus). Hence we just need to understand how these to elements are represented by the Teichmüller TQFT. To this end, we build a cobordism  $(M, \mathbb{T}^2, \mathbb{T}^2)$  from one triangulation of  $\mathbb{T}^2$  to the image of this triangulation under the action of S and likewise for the action of T. We triangulate the torus  $\mathbb{T}^2 = S^1 \times S^1$  according to Figure 11. In this triangulation opposite arrows are identified



Figure 11: Triangulated torus.

and this gives us a triangulation with two triangles and three edges. We build the cobordism for the action of S according to Figure 12 and the cobordism for T according to Figure 13. We see that on each boundary component we have three edges. The cobordisms that we build are given shaped triangulations. We can choose the dihedral angles such that they are all positive. And we are able to compose these cobordisms.



Figure 12: The cobordism for the operator S.



Figure 13: The cobordism for the operator T.

For each edge in these triangulations we assign a state variable. We abuse notation and label an edge and a state variable by the same letter. We assign a multiplier to each edge. As we will see below in Lemma 8.1 and Lemma 9.1 it turns out, that all internal edges have trivial multiplier. Further we emphasise that there is a direction on each of the two boundary tori where the multiplier is trivial.

The Teichmüller TQFT gives an operator between the vector spaces associated to each of the boundary components. We will see that we get representations

$$\rho: \Gamma_{1,1} \to \mathcal{B}(C^{\infty}(\mathbb{T}^3, \mathcal{L}')),$$

of the mapping class group  $\Gamma_{1,1}$  into bounded operators on the smooth sections  $C^{\infty}(\mathbb{T}^3, \mathcal{L}')$ . However, we will show below that we actually get representations into  $\mathcal{B}(\mathcal{S}(\mathbb{R}))$ , bounded operators on the Schwartz space  $\mathcal{S}(\mathbb{R})$ .

**Theorem 6.1.** The Teichmüller TQFT provides us with representations (dependent on h)

$$\tilde{\rho}: \Gamma_{1,1} \to \mathcal{B}(\mathcal{S}(\mathbb{R}))$$

of the mapping class group  $\Gamma_{1,1}$  into bounded operators on the Schwarz space  $\mathcal{S}(\mathbb{R})$ . In particular we get operators  $\tilde{\rho}(S), \tilde{\rho}(T) : \mathcal{S}(\mathbb{R}) \to \mathcal{S}(\mathbb{R})$  according to the diagram (6.1).

where  $\mathcal{L}' = \pi^* \mathcal{L}$ .

Proof. We know that the Weil–Gel'fand–Zak transformation gives an isomorphism from the Schwarz space to smooth sections of the complex line bundle  $\mathcal{L}$  over the 2-torus. If a section of  $C^{\infty}(\mathbb{T}^2, \mathcal{L})$  is pulled back to  $\pi^*(C^{\infty}(\mathbb{T}^2, \mathcal{L}))$  we show in Lemma 8.2 and Lemma 9.2 that the operators  $\rho(S), \rho(T)$  acting on  $C^{\infty}(\mathbb{T}^3, \mathcal{L}')$ take this pull back of a section in  $\pi^*(C^{\infty}(\mathbb{T}^2, \mathcal{L}))$ . In Lemma 8.1 and 9.1 we prove that the multipliers on internal edges are trivial. Further we show that the multipliers on the two boundary tori are trivial in the direction (1, 1, 1). We can therefore integrate over the fibre in this direction. We then use the inverse WGZ transformation. In other words we have shown that the operators  $\rho(S), \rho(T)$ induce operators  $\tilde{\rho}(S), \tilde{\rho}(T) : \mathcal{S}(\mathbb{R}) \to S(\mathbb{R})$  given by

$$\tilde{\rho}(S) = W^{-1} \circ \int_{F_{z'}} \circ \rho(S) \circ \pi^* \circ W,$$

J. E. Andersen J. J. K. Nissen

$$\tilde{\rho}(T) = W^{-1} \circ \int_{F_{z'}} \circ \rho(S) \circ \pi^* \circ W.$$

**Remark 6.2.** Above we obtained a representation for the mapping class group  $\Gamma_{1,1}$ . We do not in a similar manner get a representation for the mapping class group  $\Gamma_{1,0}$ . The reason is that not all edges in the cobordisms can be balanced without turning to negative angles.

# 7 Line bundles over the two boundary tori

Let us here describe how the line bundles we pull back looks like.

Let  $\pi : \mathbb{R}^3 \to \mathbb{R}^2$  be defined by  $\pi(x_1, x_2, x_3) = (ax_1 + bx_2 + cx_3, \alpha x_1 + \beta x_2 + \gamma x_3)$ . Recall that we have the relation on multipliers

(7.1) 
$$e_{\lambda}^{\pi^*}(x,y,z) = e_{\pi(\lambda)}(\pi(x,y,z)).$$

Note that the map  $\pi$  sends  $\lambda_{x_1} = (1,0,0), \ \lambda_{x_2} = (0,1,0), \ \lambda_{x_3} = (0,0,1)$  to the following elements of  $\mathbb{R}^2$ 

$$\pi(\lambda_{x_1}) = (a, \alpha), \quad \pi(\lambda_{x_2}) = (b, \beta), \quad \pi(\lambda_{x_3}) = (c, \gamma).$$

The equation (7.1) gives the following relations:

#### In the $\lambda_{x_1}$ -direction

$$e^{2\pi i (x_3 - x_2)} = e_{(1,0,0)(x_1, x_2, x_3)} = e_{(a,\alpha)} (ax_1 + bx_2 + cx_3, \alpha x_1 + \beta x_2 + \gamma x_3)$$
  

$$= e_{(a,0)} (ax_1 + bx_2 + cx_3, \alpha x_1 + \beta x_2 + \gamma x_3)$$
  

$$e_{(0,\alpha)} (ax_1 + bx_2 + cx_3, \alpha (x_1 + 1) + \beta x_2 + \gamma x_3)$$
  

$$= e^{-\pi i a (\alpha x_1 + \beta x_2 + \gamma x_3)} e^{\pi i (ax_1 + bx_2 + cx_3)}$$
  

$$= e^{-\pi i a (\alpha x_1 + \beta x_2 + \gamma x_3)} e^{\pi i (ax_1 + bx_2 + cx_3)}$$

#### In the $\lambda_{x_2}$ -direction

$$e^{2\pi i(x_1-x_3)} = e_{(0,1,0)(x_1,x_2,x_3)} = e^{\pi i((\beta a - \alpha b)x_1 + (\beta c - b\gamma)x_3)},$$

In the  $\lambda_{x_3}$ -direction

$$e^{2\pi i(x_2-x_1)} = e_{(0,0,1)(x_1,x_2,x_3)} = e^{\pi i((\gamma a - \alpha c)x_1 + (\gamma b - c\beta)x_2)}.$$

In other words we only need to solve the three equations

(7.2) 
$$\alpha b - a\beta = -2, \quad \alpha c - a\gamma = 2, \quad \beta c - b\gamma = -2.$$

One particular solution is  $a=-2, b=0, c=2, \alpha=0, \beta=-1, \gamma=1$  which gives the map

$$\pi(x_1, x_2, x_3) = (-2x_1 + 2x_3, -x_2 + x_3)$$

Asymptotic aspects of the Teichmüller TQFT

# 8 The operator $\rho(S)$

The operator  $\rho(S)$  can be viewed as the cobordism  $X_S$  which is triangulated into 6 tetrahedra  $T_1, \ldots, T_6$  where  $T_1, T_3, T_4, T_6$  have positive orientation and the tetrahedra  $T_2, T_5$  have negative orientation. See the gluing pattern in figure 14.



Figure 14: Gluing pattern for the operator  $X_S$ 

In the triangulation we have ten edges  $x_1, x_2 \dots, x_7, x'_1, x'_2, x'_3$ . To each of the edges on the boundary we associate the a weight function:

$$\begin{aligned} \omega_{X_S}(x_1) &= 2\pi(a_1 + a_5 + c_3), & \omega_{X_S}(x_2) &= 2\pi(a_4 + c_5 + a_6), & \omega_{X_S}(x_3) &= 2\pi(b_5 + b_6), \\ \omega_{X_S}(x_1') &= 2\pi(a_1 + c_2 + a_3), & \omega_{X_S}(x_2') &= 2\pi(a_2 + c_3 + a_4), & \omega_{X_S}(x_3') &= 2\pi(b_2 + b_3). \end{aligned}$$

and to the edges  $x_4, x_5, x_6, x_7$  we associate the weight functions:

$$\begin{split} \omega_{X_S}(x_4) &= 2\pi(a_1+c_2+b_4+c_5+c_6), \qquad \omega_{X_S}(x_5) = 2\pi(c_1+b_3+b_4+a_5+a_6), \\ \omega_{X_S}(x_6) &= 2\pi(b_1+a_2+a_3+c_4+b_6), \qquad \omega_{X_S}(x_7) = 2\pi(b_1+c_2+c_3+c_4+b_5). \end{split}$$

#### 8.1 Boltzmann weights

The Boltzmann weights assigned to the tetrahedra are

$$\begin{split} B\left(T_{1}, x_{|_{\Delta_{1}(T_{1})}}\right) &= g_{a_{1},c_{1}}(x_{7} + x_{6} - x_{4} - x_{5}, x_{7} + x_{6} - x_{1}' - x_{1}), \\ B\left(T_{2}, x_{|_{\Delta_{1}(T_{2})}}\right) &= \overline{g_{a_{2},c_{2}}(x_{3}' + x_{4} - x_{1}' - x_{7}, x_{3}' + x_{4} - x_{2}' - x_{6})}, \\ B\left(T_{3}, x_{|_{\Delta_{1}(T_{3})}}\right) &= g_{a_{3},c_{3}}(x_{3}' + x_{5} - x_{2}' - x_{7}, x_{3}' + x_{5} - x_{1}' - x_{6}) \\ B\left(T_{4}, x_{|_{\Delta_{1}(T_{4})}}\right) &= g_{a_{4},c_{4}}(x_{5} + x_{4} - x_{7} - x_{6}, x_{5} + x_{4} - x_{2}' - x_{2}) \\ B\left(T_{5}, x_{|_{\Delta_{1}(T_{5})}}\right) &= \overline{g_{a_{5},c_{5}}(x_{7} + x_{3} - x_{4} - x_{2}, x_{5} + x_{4} - x_{5} - x_{1})} \\ B\left(T_{6}, x_{|_{\Delta_{1}(T_{6})}}\right) &= g_{a_{6},c_{6}}(x_{6} + x_{3} - x_{4} - x_{1}, x_{6} + x_{3} - x_{5} - x_{2}). \end{split}$$

**Lemma 8.1.** The multipliers corresponding to the edges are calculated to be 1 for the internal edges  $x_4, x_5, x_6, x_7$ . And the multipliers for the remaining 6 edges are calculated to be

$$e_{\lambda_{x_1}}(\boldsymbol{x}) = e^{2\pi i (x_3 - x_2)}, \quad e_{\lambda_{x_2}}(\boldsymbol{x}) = e^{2\pi i (x_1 - x_3)}, \quad e_{\lambda_{x_3}}(\boldsymbol{x}) = e^{2\pi i (x_2 - x_1)},$$

J. E. Andersen J. J. K. Nissen

$$e_{\lambda_{x_1'}}(\pmb{x}) = e^{2\pi i (x_2' - x_3')}, \quad e_{\lambda_{x_2'}}(\pmb{x}) = e^{2\pi i (x_3' - x_1')}, \quad e_{\lambda_{x_3'}}(\pmb{x}) = e^{2\pi i (x_1' - x_2')},$$

where **x** denotes the tuple  $\mathbf{x} = (x_1, x_2, x_3, x'_1, x'_2, x'_3)$ .

*Proof.* Let us here just calculate the multiplier for the direction  $x_4$ . The rest follows by analogous calculations. The edge  $x_4$  is an edge in the tetrahedra  $T_1, T_2, T_4, T_5, T_6$  each contributing to the multiplier. The contribution from  $T_1$  corresponds to the multiplier

$$e_{\lambda_{x_4}}(x_1, x_2, \dots, x_7, x_1', x_2', x_3') = e_{-(1,0)}(x_5 + x_4 - x_7 - x_6, x_5 + x_4 - x_2' - x_2)$$
$$= e^{\pi i (x_7 + x_6 - x_1' - x_1)}$$

The contribution from  $T_2$  is

$$e_{\lambda_{x_4}}(x_1, x_2, \dots, x_7, x_1', x_2', x_3') = e_{(1,1)}(x_3' + x_4 - x_1' - x_7, x_3' + x_4 - x_2' - x_6)$$
  
=  $-e^{-\pi i (x_2' + x_6 - x_1' - x_7)}.$ 

The contribution from  $T_4$  is

$$e_{\lambda_{x_4}}(x_1, x_2, \dots, x_7, x_1', x_2', x_3') = e_{(1,1)}(x_5 + x_4 - x_7 - x_6, x_5 + x_4 - x_2' - x_2)$$
  
=  $-e^{\pi i (x_2' + x_2 - x_6 - x_7)}$ 

The contribution from  $T_5$  is

$$e_{\lambda_{x_{*}}}(x_1, x_2, \dots, x_7, x_1', x_2', x_3') = -e^{-\pi i (x_7 + x_3 - x_5 - x_1)}.$$

The contribution from  $T_6$  is

$$e_{\lambda_{x_4}}(x_1, x_2, \dots, x_7, x_1', x_2', x_3') = -e^{\pi i (x_6 + x_3 - x_5 - x_2)}$$

Multiplying these contributions gives  $e^0 = 1$ .

We remark that the multiplier on each boundary component in direction (1, 1, 1) is trivial.

We are interested in how the operator  $\rho(S)$  acts. We express the operator  $\rho(S)$  in terms of the integral kernel  $K_S$ . The operator  $\rho(S)$  acts on sections in the following manner

(8.1) 
$$\rho(S)(s)(x'_1, x'_2, x'_3) = \int_{[0,1]^3} K_S(x'_1, x'_2, x'_3, x_1, x_2, x_3) s(x_1, x_2, x_3) dx_1 dx_2 dx_3.$$

We want to show that the operator S takes the pull back of a section to the pull back of a section. Using integration by parts it is enough to check that the sum of partial derivatives disappear.

80

Asymptotic aspects of the Teichmüller TQFT

**Lemma 8.2.** The sum of the partial derivatives of  $K_S$  disappears. I.e.

$$\frac{\partial K_S}{\partial x_1'} + \frac{\partial K_S}{\partial x_2'} + \frac{\partial K_S}{\partial x_3'} + \frac{\partial K_S}{\partial x_1} + \frac{\partial K_S}{\partial x_2} + \frac{\partial K_S}{\partial x_3} = 0.$$

Proof. Let

$$I_{3}^{n,m,k,j}(x_{1},x_{2},x_{3},x_{2}',x_{3}') := \int_{[0,1]^{2}} \tilde{\psi}_{a_{1},c_{1}}'(x_{7}+k) \tilde{\psi}_{a_{4},c_{4}}'(-x_{7}+n) \\ e^{2\pi i x_{7}(x_{2}-x_{3}+x_{7}+x_{5}+2n-m+k-j)} \\ e^{2\pi i (x_{3}'-x_{2}'-x_{3}+x_{1}+k-j)} dx_{5} dx_{7}$$

The partial derivatives of  $I_3$  with respect to  $x_1, x_2, x_3, x_2^\prime, x_3^\prime$  are easily calculated to be

$$\begin{split} &\frac{\partial}{\partial x_1}I_3^{n,m,k,j}(x_1,x_2,x_3,x_2',x_3') = 2\pi i x_5 I_3(x_1,x_2,x_3,x_2',x_3') =: I_3'(x_1,x_2,x_3,x_2',x_3'),\\ &\frac{\partial}{\partial x_2}I_3^{n,m,k,j}(x_1,x_2,x_3,x_2',x_3') = 2\pi i x_7 I_3(x_1,x_2,x_3,x_2',x_3') =: I_3''(x_1,x_2,x_3,x_2',x_3'),\\ &\frac{\partial}{\partial x_3}I_3^{n,m,k,j}(x_1,x_2,x_3,x_2',x_3') = -I_3'(x_1,x_2,x_3,x_2',x_3') - I_3''(x_1,x_2,x_3,x_2',x_3'),\\ &\frac{\partial}{\partial x_2'}I_3^{n,m,k,j}(x_1,x_2,x_3,x_2',x_3') = -I_3'(x_1,x_2,x_3,x_2',x_3'),\\ &\frac{\partial}{\partial x_3'}I_3^{n,m,k,j}(x_1,x_2,x_3,x_2',x_3') = -I_3'(x_1,x_2,x_3,x_2',x_3'), \end{split}$$

The partial derivatives of  $I_2$  with respect to the variables  $x_2, x_3, x_1^\prime, x_3^\prime$  are

$$\begin{split} \frac{\partial}{\partial x_2} I_2^{k,l,n,p}(x_2, x_3, x_1', x_3') &= \frac{e^{2\pi i (x_1' - x_3' - x_3 + x_2 + 2(k,l,n,p))}(x_1' - x_3' - x_3 + x_2 + 2(k,l,n,p))}{(x_1' - x_3' - x_3 + x_2 + 2(k,l,n,p))^2} \\ &- \frac{(e^{2\pi i (x_1' - x_3' - x_3 + x_2 + 2(k,l,n,p)) - 1)}}{2\pi i (x_1' - x_3' - x_3 + x_2 + 2(k,l,n,p))^2} \\ &= : I_2'(x_2, x_3, x_1', x_3'), \\ \frac{\partial}{\partial x_3} I_2^{k,l,n,p}(x_2, x_3, x_1', x_3') &= - I_2'(x_2, x_3, x_1', x_3'), \\ \frac{\partial}{\partial x_1'} I_2^{k,l,n,p}(x_2, x_3, x_1', x_3') &= - I_2'(x_2, x_3, x_1', x_3'), \\ \frac{\partial}{\partial x_1'} I_2^{k,l,n,p}(x_2, x_3, x_1', x_3') &= - I_2'(x_2, x_3, x_1', x_3'). \end{split}$$

The partial derivatives of  $I_1$  with respect to the variables  $x_2, x_3, x_1^\prime, x_3^\prime$  are

$$\frac{\partial}{\partial x_2}I_1^{l,p}(x_2, x_3, x_1', x_3') = - \frac{e^{2\pi i (x_3' - x_1' - x_2 + x_3 + 2(m+q))}(x_3' - x_1' - x_2 + x_3 + 2(m+q))}{(x_3' - x_1' - x_2 + x_3 + 2(m+q))^2}$$

J. E. Andersen J. J. K. Nissen

$$\begin{split} &+ \frac{(e^{2\pi i (x_3'-x_1'-x_2+x_3+2(m+q))}-1)}{2\pi i (x_3'-x_1'-x_2+x_3+2(m+q))^2} \\ &= I_1'(x_2,x_3,x_1',x_3'), \\ &\frac{\partial}{\partial x_3} I_1^{l,p}(x_2,x_3,x_1',x_3') = - I_1'(x_2,x_3,x_1',x_3'), \\ &\frac{\partial}{\partial x_1'} I_1^{l,p}(x_2,x_3,x_1',x_3') = I_1'(x_2,x_3,x_1',x_3'), \\ &\frac{\partial}{\partial x_3'} I_1^{l,p}(x_2,x_3,x_1',x_3') = - I_1'(x_2,x_3,x_1',x_3'). \end{split}$$

The rest of the terms in  $K_S$  all depends on pairs of the variables  $x_1, x_2, x_3, x'_1, x'_2, x'_3$  with opposite sign, summing all contributions together therefore shows that the sum of the partial derivatives disappears.

# 9 The operator $\rho(T)$

The operator  $\rho(T)$  is the TQFT operator associated to the cobordism  $X_T$  which is triangulated into 6 tetrahedra  $T_1, \ldots, T_6$  where  $T_1, T_4, T_5$  have negative orientation and the tetrahedra  $T_2, T_3, T_6$  have positive orientation. See Figure 15.



Figure 15: Gluing pattern for the operator  $X_T$ .

In the triangulation we have ten edges  $x_1, x_2, \ldots, x_7, x'_1, x'_2, x'_3$ . The weight functions corresponding to this triangulation for the edges  $x_1, x_2, x_3, x'_1, x'_2, x'_3$  are

$$\begin{aligned} \omega_{Y_T}(x_1) &= 2\pi(c_3 + a_6), \qquad \omega_{Y_T}(x_2) = 2\pi(b_2 + a_3 + b_6), \qquad \omega_{Y_T}(x_3) = 2\pi(b_3 + b_5 + c_6), \\ \omega_{Y_T}(x_1') &= 2\pi(a_1 + c_4), \qquad \omega_{Y_T}(x_2') = 2\pi(b_1 + a_4 + b_5), \qquad \omega_{Y_T}(x_3') = 2\pi(c_1 + b_2 + b_4). \end{aligned}$$

and to the edges  $x_4, x_5, x_6, x_7$  we associate the weight functions

$$\omega_{Y_T}(x_4) = 2\pi(a_1 + c_2 + c_5 + a_6), \qquad \qquad \omega_{Y_T}(x_5) = 2\pi(b_1 + a_2 + b_3 + b_4 + a_5 + b_6), \\ \omega_{Y_T}(x_6) = 2\pi(c_1 + a_2 + a_3 + a_4 + a_5 + c_6), \qquad \omega_{Y_T}(x_7) = 2\pi(b_2 + c_3 + c_4 + c_5).$$

The Boltzmann weights assigned to the tetrahedra are

$$B\left(T_{1}, x_{|\Delta_{1}(T_{1})}\right) = \overline{g_{a_{1},c_{1}}(x_{5} + x_{2}' - x_{3}' - x_{6}, x_{5} + x_{2}' - x_{1}' - x_{4})},$$

$$\begin{split} B\left(T_{2}, x_{|_{\Delta_{1}(T_{2})}}\right) &= g_{a_{2},c_{2}}(x_{3}' + x_{2} - x_{7} - x_{4}, x_{3}' + x_{2} - x_{5} - x_{6}),\\ B\left(T_{3}, x_{|_{\Delta_{1}(T_{3})}}\right) &= g_{a_{3},c_{3}}(x_{5} + x_{3} - x_{7} - x_{1}, x_{5} + x_{3} - x_{6} - x_{2})\\ B\left(T_{4}, x_{|_{\Delta_{1}(T_{4})}}\right) &= \overline{g_{a_{4},c_{4}}(x_{3}' + x_{5} - x_{7} - x_{1}', x_{3} + x_{5} - x_{2}' - x_{6})}\\ B\left(T_{5}, x_{|_{\Delta_{1}(T_{5})}}\right) &= \overline{g_{a_{5},c_{5}}(x_{2}' + x_{3} - x_{7} - x_{4}, x_{2}' + x_{3} - x_{6} - x_{5})}\\ B\left(T_{6}, x_{|_{\Delta_{1}(T_{6})}}\right) &= g_{a_{6},c_{6}}(x_{2} + x_{5} - x_{3} - x_{6}, x_{2} + x_{5} - x_{4} - x_{1}). \end{split}$$

**Lemma 9.1.** The multipliers corresponding to the edges are calculated to be 1 for the internal edges  $x_4, x_5, x_6, x_7$ . And the multipliers for the remaining 6 edges are calculated to be

$$\begin{split} e_{\lambda_{x_1}}(\textbf{\textit{x}}) &= e^{2\pi i (x_3 - x_2)}, \quad e_{\lambda_{x_2}}(\textbf{\textit{x}}) = e^{2\pi i (x_1 - x_3)}, \quad e_{\lambda_{x_3}}(\textbf{\textit{x}}) = e^{2\pi i (x_2 - x_1)}, \\ e_{\lambda_{x_1'}}(\textbf{\textit{x}}) &= e^{2\pi i (x_2' - x_3')}, \quad e_{\lambda_{x_2'}}(\textbf{\textit{x}}) = e^{2\pi i (x_3' - x_1')}, \quad e_{\lambda_{x_3'}}(\textbf{\textit{x}}) = e^{2\pi i (x_1' - x_2')}, \end{split}$$

where **x** denotes the tuple  $\mathbf{x} = (x_1, x_2, x_3, x'_1, x'_2, x'_3)$ .

*Proof.* The proof is straight forward verification. The computations are analogue to the calculations in the proof of Lemma 8.1.  $\hfill \square$ 

Again, in order to check that the operator  $\rho(T)$  takes the pull back of a section to a pull back of a section we show the following Lemma.

**Lemma 9.2.** The sum of the partial derivatives of  $K_T$  disappears. I.e.

$$\frac{\partial K_T}{\partial x_1'} + \frac{\partial K_T}{\partial x_2'} + \frac{\partial K_T}{\partial x_3'} + \frac{\partial K_T}{\partial x_1} + \frac{\partial K_T}{\partial x_2} + \frac{\partial K_T}{\partial x_3} = 0$$

*Proof.* In each term of the expression for  $K_T$  there is an equal number of variables one half having positive coefficient and the other half having negative coefficient. Therefore the sum of the partial differentials must equal zero.

# Appendices

# A Faddeev's quantum dilogarithm

The quantum dilogarithm function  $\text{Li}_2(x;q)$ , studied by Faddeev and Kashaev [FK] and other authors, is the function of two variables defined by the series

(A.1) 
$$\operatorname{Li}_{2}(x;q) = \sum_{n=1}^{\infty} \frac{x^{n}}{n(1-q^{n})},$$

where  $x, q \in \mathbb{C}$ , with |x|, |q| < 1. It is connected to the classical Euler dilogarithm Li<sub>2</sub> given by Li<sub>2</sub> $(x) = \sum_{n=1}^{\infty} \frac{x^n}{n^2}$  in the sense that it is a *q*-deformation of the classical one in the following manner

(A.2) 
$$\lim_{\epsilon \to 0} \left( \epsilon \operatorname{Li}_2(x, e^{-\epsilon}) \right) = \operatorname{Li}_2(x), \quad |x| < 1.$$

Indeed using the expansion  $\frac{1}{1-e^{-t}} = \frac{1}{t} + \frac{1}{2} + \frac{t}{12} - \frac{t^3}{720} + \dots$  we obtain a complete asymptotic expansion

(A.3) 
$$\operatorname{Li}_2(x, e^{-\epsilon}) = \operatorname{Li}_2(x)\epsilon^{-1} + \frac{1}{2}\log\left(\frac{1}{1-x}\right) + \frac{x}{1-x}\frac{\epsilon}{12} - \frac{x+x^2}{(1-x)^3}\frac{\epsilon^3}{720} + \dots$$

as  $\epsilon \to 0$  with fixed  $x \in \mathbb{C}$ , |x| < 1.

The second quantum dilogarithm  $(x;q)_\infty$  defined for |q|<1 and all  $x\in\mathbb{C}$  is given as the function

(A.4) 
$$(x;q)_{\infty} = \prod_{i=0}^{\infty} (1 - xq^i).$$

This second quantum dilogarithm is related to the first by the formula

(A.5) 
$$(x;q)_{\infty} = \exp(-\operatorname{Li}_2(x;q)).$$

This is easily proven by a direct calculation (A.6)

$$-\log(x;q)_{\infty} = \sum_{i=0}^{\infty} \log(1-xq^i) = \sum_{i=0}^{\infty} \sum_{n=1}^{\infty} \frac{1}{n} x^n q^{in} = \sum_{n=1}^{\infty} \frac{x^n}{n(1-q^n)} = \operatorname{Li}_2(x;q).$$

**Proposition A.1.** The function  $(x;q)_{\infty}$  and its reciprocal have the Taylor expansions

(A.7) 
$$(x;q)_{\infty} = \sum_{n=0}^{\infty} \frac{(-1)^n}{(q)_n} q^{\frac{n(n-1)}{2}} x^n, \quad \frac{1}{(x;q)_{\infty}} \sum_{n=0}^{\infty} \frac{1}{(q)_n} x^n,$$

around x = 0, where

$$(q)_n = \frac{(q;q)_\infty}{(q^{n+1};q)_\infty} = (1-q)(1-q^2) \cdot (1-q^n).$$

The proofs of these formulas follows easily from the recursion formula  $(x; q)_{\infty} = (1-x)(qx; x)_{\infty}$ , which together with the initial value  $(0; q)_{\infty} = 1$  determines the power series for  $(x; q)_{\infty}$  uniquely.

Yet another famous result for the function  $(x;q)_{\infty}$ , which can be proven by use of the Taylor expansion and the identity  $\sum_{m-n=k} \frac{q^{mn}}{(q)_m(q)_n} = \frac{1}{(q)_{\infty}}$ , is the Jacobi triple product formula

(A.8) 
$$(q;q)_{\infty}(x;q)_{\infty}(qx^{-1};q)_{\infty} = \sum_{k\in\mathbb{Z}} (-1)^k q^{\frac{k(k-1)}{2}} x^k,$$

which relates the quantum dilogarithm function to the classical Jacobi theta-function.

The quantum dilogarithm functions introduced are related to yet another quantum dilogarithm function named after Faddeev.

#### Faddeev's quantum dilogarithm

**Definition A.2.** Faddeev's quantum dilogarithm function is the function in two complex arguments z and b defined by the formula

(A.9) 
$$\Phi_{\rm b}(z) = \exp\left(\int_C \frac{e^{-2izw}dw}{4\sinh(w\,\mathrm{b})\sinh(w/\,\mathrm{b})w}\right)$$

where the contour C runs along the real axis, deviating into the upper half plane in the vicinity of the origin.

**Proposition A.3.** Faddeev's quantum dilogarithm function  $\Phi_{\rm b}(z)$  is related to the function  $(x;q)_{\infty}$ , where |q| < 1, in the following sense. When  $\Im({\rm b}^2) > 0$ , the integral can be calculated explicitly

(A.10) 
$$\Phi_{\rm b}(z) = \frac{\left(e^{2\pi(z+c_{\rm b}){\rm b}};q^2\right)_{\infty}}{\left(e^{2\pi(z-c_{\rm b}){\rm b}};\tilde{q}^2\right)_{\infty}}$$

where  $q \equiv e^{i\pi b^2}$  and  $\tilde{q} \equiv e^{-\pi i b^{-2}}$ .

*Proof.* We consider the integrand of the integral  $I(z, \mathbf{b}) = \frac{1}{4} \int_C \frac{e^{-2izw}}{\sinh(w) \sinh(w/b)w} dw$ . The integrand has poles at  $w = \pi i n \mathbf{b}$  and  $w = \pi i n \mathbf{b}^{-1}$ . The residue at c of a fraction i.e.  $f(x) = \frac{g(x)}{h(x)}$  can be calculated as  $\operatorname{Res} f(c) = \frac{g(c)}{h'(c)}$  when c is a simple pole. Therefore we get by the residue theorem

$$\begin{split} I(z,\mathbf{b}) &= \frac{\pi i}{2} \sum_{n=1}^{\infty} \frac{e^{2\pi z \,\mathbf{b} \,n}}{\pi i n \,\mathbf{b}(-1)^n \sinh(\pi i n \,\mathbf{b}^2)} + \frac{e^{2\pi z \,\mathbf{b}^{-1} \,n}}{\pi i n \,\mathbf{b}(-1)^n \sinh(\pi i n \,\mathbf{b}^{-2})} \\ &= \sum_{n=1}^{\infty} \frac{e^{\pi i n} e^{2\pi z \,\mathbf{b} \,n}}{n (e^{\pi i n \,\mathbf{b}^2} - e^{-\pi i n \,\mathbf{b}^2})} + \frac{e^{\pi i n} e^{2\pi z \,\mathbf{b}^{-1} \,n}}{n (e^{\pi i n \,\mathbf{b}^2} - e^{-\pi i n \,\mathbf{b}^{-2}})} \\ &= \sum_{n=1}^{\infty} -\frac{\left(e^{2\pi z \,\mathbf{b} + \pi i + \pi i \,\mathbf{b}^2}\right)^n}{n (1 - e^{2\pi i \,\mathbf{b}^2 \,n})} + \frac{\left(e^{2\pi z \,\mathbf{b}^{-1} - \pi i - \pi i \,\mathbf{b}^{-2}}\right)^n}{n (1 - e^{-2\pi i \,\mathbf{b}^{-2} \,n})} \\ &= \sum_{n=1}^{\infty} -\frac{e^{2\pi (z + c_{\mathbf{b}}) \,\mathbf{b} \,n}}{n (1 - e^{2\pi i \,\mathbf{b}^2 \,n})} + \frac{e^{2\pi (z - c_{\mathbf{b}}) \,\mathbf{b}^{-1} \,n}}{n (1 - e^{-2\pi i \,\mathbf{b}^{-2} \,n})} \\ &= \log \left(e^{2\pi (z + c_{\mathbf{b}}) \,\mathbf{b}}; q^2\right)_{\infty} - \log \left(e^{2\pi (z - c_{\mathbf{b}}) \,\mathbf{b}}; \tilde{q}^2\right)_{\infty}. \end{split}$$

#### **Functional** equations

**Proposition A.4.** Faddeev's quantum dilogarithm function satisfies the two functional equations

(A.11) 
$$\frac{1}{\Phi_{\rm b}(z+i{\rm b}/2)} = \frac{1}{\Phi_{\rm b}(z-i{\rm b}/2)} \left(1+e^{2\pi{\rm b}z}\right),$$

(A.12) 
$$\Phi_{\rm b}(z)\Phi_{\rm b}(-z) = \zeta_{inv}^{-1} e^{i\pi z^2},$$

where  $\zeta_{inv} = e^{i\pi(1+2c_{\rm b}^2)/6}$ .

*Proof.* Let us first prove (A.11). We have

$$\begin{aligned} \frac{\Phi_{\rm b}(z-i{\rm b}/2)}{\Phi_{\rm b}(z+i{\rm b}/2)} &= \exp \int_C \frac{e^{-2i(z-i{\rm b}/2)w} - e^{-2i(z+i{\rm b}/2)w}}{4\sinh(w{\rm b})\sinh(w/{\rm b})w} \, dw \\ &= \exp \int_C \frac{e^{-2izw} \left(e^{-bw} - e^{bw}\right)}{4\sinh(w{\rm b})\sinh(w/{\rm b})w} \, dw \\ &= \exp\left(-\frac{1}{2}\int_C \frac{e^{-2izw}}{\sinh(w/{\rm b})w}\right) \, dw. \end{aligned}$$

Let a > 0. Let  $\varepsilon = 1$  if  $\Im(-2iz) \ge 0$  and  $\varepsilon = -1$  otherwise. Put  $\delta_a^- = [-a, i\varepsilon a]$ and  $\delta_a^- = [i\varepsilon a, a]$ . The integrals  $\int_{\delta_{a^{\pm}}} \frac{e^{-2izw}}{2\sinh(w/b)w} dw$  converge to zero as  $a \to \infty$ . Therefore

$$\int_C \frac{e^{-2izw}}{\sinh(w/b)w} \, dw = \epsilon 2\pi i \left( c_\epsilon + \sum_{n=1}^\infty \operatorname{Res}_{w=\epsilon i\pi bn} \left\{ \frac{e^{-2izw}}{\sinh(w/b)w} \right\} \right),$$

where  $c_1 = 0$  and  $c_{-1} = \operatorname{Res}_{w=0} \left\{ \frac{e^{2izw}}{\sinh(w/b)w} \right\} = -2izb$ . For  $n \in \mathbb{Z} \setminus \{0\}$  we have

$$\operatorname{Res}_{w=\pi inb\epsilon}\left\{\frac{e^{-2izw}}{\sinh(w/b)w}\right\} = \frac{(-1)^n e^{2z\pi b\epsilon n}}{\pi in}$$

 $\mathbf{SO}$ 

$$\int_C \frac{e^{-2izw}}{\sinh(w/b)w} dw = (\epsilon - 1)2\pi z b - 2\log(1 + e^{2z\pi b\epsilon}),$$

giving the first result.

To prove equation (A.12) let us choose the path  $C = (-\infty, -\epsilon] \cup \epsilon \exp([\pi i, 0]) \cup [\epsilon, \infty)$  and let  $\epsilon \to 0$ . The rest is just calculations

$$\log \Phi_{\rm b}(z)\Phi_{\rm b}(-z) = \frac{1}{2} \int_C \frac{\cos(2wz)}{\sinh(w\,\mathrm{b})\sinh(w/\,\mathrm{b})w} dw$$

Note that

$$\frac{1}{2}\int_{(-\infty,-\epsilon]}\frac{\cos(2wz)}{\sinh(w\,\mathbf{b})\sinh(w/\,\mathbf{b})w}dw = -\frac{1}{2}\int_{[\epsilon,\infty)}\frac{\cos(2wz)}{\sinh(w\,\mathbf{b})\sinh(w/\,\mathbf{b})w}dw.$$

i.e. it is enough to collect the half residue around w = 0 of the remaining integral

$$\frac{1}{2} \int_{\epsilon([\pi i,0])} \frac{\cos(2wz)}{\sinh(w\,\mathrm{b})\sinh(w/\,\mathrm{b})w} dw = \frac{\pi i}{2} \operatorname{Res}_{w=0} \frac{\cos(2wz)}{\sinh(w\,\mathrm{b})\sinh(w/\,\mathrm{b})w} = \frac{\pi i}{2} \left(\frac{b^2 + b^{-2}}{6} + 2z^2\right) = e^{-\pi i(1+2c_b^2)/6} e^{\pi i z^2}.$$

#### Zeros and poles

The functional equation (A.11) shows that  $\Phi_{\rm b}(z)$ , which in its initial domain of definition has no zeroes and poles, extends (for fixed b with  $\Im b^2 > 0$ ) to a meromorphic function in the variable z to the entire complex plane with essential singularity at infinity and with characteristic properties:

(A.13) 
$$(\Phi_{\mathbf{b}}(z))^{\pm 1} = 0 \iff z = \mp (c_{\mathbf{b}} + mi\,\mathbf{b} + ni\,\mathbf{b})$$

The behaviour at infinity depends on the direction along which the limit is taken

$$(A.14) \qquad \Phi_{\rm b}(z)\big|_{|z|\to\infty} \approx \begin{cases} 1 & |\arg z| > \frac{\pi}{2} + \arg {\rm b}, \\ \zeta_{inv}^{-1} e^{\pi i z^2} & |\arg z| < \frac{\pi}{2} - \arg {\rm b} \\ \frac{(q^2, q^2)_\infty}{\Theta({\rm i}\, b^{-1}\, z; - b^{-2})} & |\arg z - \frac{\pi}{2}| < \arg {\rm b} \\ \frac{\Theta({\rm i}\, b\, z; b^2)}{(q^2; q^2)_\infty} & |\arg z + \frac{\pi}{2}| < \arg {\rm b} \end{cases}$$

where

(A.15) 
$$\Theta(z;\tau) \equiv \sum_{n \in \mathbb{Z}} e^{\pi i \tau n^2 + 2\pi i z n}, \quad \Im \tau > 0.$$

#### Unitarity

When b is real or on the unit circle

(A.16) 
$$(1 - |\mathbf{b}|)\Im \mathbf{b} = 0 \quad \Rightarrow \quad \overline{\Phi_{\mathbf{b}}(z)} = \frac{1}{\Phi_{\mathbf{b}}(\overline{z})}.$$

#### Quantum Pentagon Identity

In terms of specifically normalised selfadjoint Heisenberg momentum and position operators acting as unbounded operators on  $L^2(\mathbb{R})$  by the formulae

$$\mathbf{q}f(x) = xf(x), \quad \mathbf{p}f(x) = \frac{1}{2\pi i}f(x),$$

the following pentagon identity for unitary operators is satisfied [FK]

(A.17) 
$$\Phi_{\rm b}(\mathbf{p})\Phi_{\rm b}(\mathbf{q}) = \Phi_{\rm b}(\mathbf{q})\Phi_{\rm b}(\mathbf{p}+\mathbf{q})\Phi_{\rm b}(\mathbf{p})$$

#### Fourier transformation formulae for Faddeev's quantum dilogarithm

The quantum pentagon identity (A.17) is equivalent to the integral identity

(A.18) 
$$\int_{\mathbb{R}+i\varepsilon} \frac{\Phi_{\rm b}(x+u)}{\Phi_{\rm b}(x-c_{\rm b})} e^{-2\pi i w x} \, dx = \frac{\Phi_{\rm b}(u)\Phi_{\rm b}(c_{\rm b}-w)}{\Phi_{\rm b}(u-w)} e^{\frac{\pi i}{12}(1-4c_{\rm b}^2)},$$

where  $\Im b^2 > 0$ . From here we get the Fourier transformation formula for the quantum dilogarithm formally sending  $u \to -\infty$  by the use of (A.10) and (A.16)

(A.19) 
$$\int_{\mathbb{R}+i\varepsilon} \Phi_{\rm b}(x+c_{\rm b})e^{2\pi iwx} = \frac{1}{\Phi_{\rm b}(-w-c_{\rm b})}e^{-\frac{\pi i}{12}(1-4c_{\rm b}^2)}.$$

#### Quasi-classical limit of Faddeev's quantum dilogarithm

**Proposition A.5.** For fixed x and  $b \rightarrow 0$  we have the following asymptotic expansion

(A.20) 
$$\log \Phi_{\rm b}\left(\frac{x}{2\pi\,{\rm b}}\right) = \sum_{n=0}^{\infty} (2\pi i\,{\rm b})^{2n-1} \frac{B_{2n}(1/2)}{(2n)!} \frac{\partial^{2n}\,{\rm Li}_2(-e^x)}{\partial x^{2n}},$$

where  $B_{2n}(1/2)$  are the Bernoulli polynomials  $B_{2n}$  evaluated at 1/2.

*Proof.* From (A.11) we have that

$$\log\left(\frac{\Phi_{\rm b}\left(\frac{x-i\pi\,{\rm b}^2}{2\pi\,{\rm b}}\right)}{\Phi_{\rm b}\left(\frac{x+i\pi\,{\rm b}^2}{2\pi\,{\rm b}}\right)}\right) = \log(1+e^x).$$

The left hand side yields

$$\log \Phi_{\rm b} \left( \frac{x - i\pi \,\mathrm{b}^2}{2\pi \,\mathrm{b}} \right) - \log \Phi_{\rm b} \left( \frac{x + i\pi \,\mathrm{b}^2}{2\pi \,\mathrm{b}} \right) = -2 \sinh(i\pi b^2 \partial/\partial x) \log \Phi_{\rm b} \left( \frac{x}{2\pi \,\mathrm{b}} \right),$$

where we have used the fact that

$$f(x+y) = e^{y\frac{\partial}{\partial x}}(f)(x),$$

which is just the Taylor expansion of f around x. While the right hand side can be written in the following manner

$$\log(1+e^x) = \frac{\partial}{\partial x} \int_{-\infty}^x \log(1+e^z) \, dz = -\frac{\partial}{\partial x} \operatorname{Li}_2(-e^x).$$

Using the expansion

$$\frac{z}{\sinh(z)} = \sum_{n=0}^{\infty} B_{2n}(1/2) \frac{(2z)^{2n}}{(2n)!}$$

gives exactly (A.20).

**Corollary A.6.** For fixed x and  $b \rightarrow 0$  one has

(A.21) 
$$\Phi_{\rm b}\left(\frac{x}{2\pi\,{\rm b}}\right) = \exp\left(\frac{1}{2\pi i b^2}\,{\rm Li}_2(-e^x)\right)\left(1+O(b^2)\right).$$

# **B** The Tetrahedral Operator

In order to prove Proposition 3.1 we make use of the following formulae

Lemma B.1. Suppose x and y are operators in an algebra such that

$$z = [x, y], [x, z] = 0$$

Then

$$f(x)y = yf(x) + zf'(x)$$
$$e^{x}f(x) = f(y+z)e^{x},$$

for every power series such that f(x), f'(x) and f(y + z) can be defined in the same operator algebra.

*Proof.* Let  $f(x) = \sum_{j=0}^{\infty} a_j x^j$ . Then,

$$[f(x), y] = \sum_{j=0}^{\infty} a_j [x^j, y] = \sum_{j=0}^{\infty} a_j \sum_{k=0}^{j-1} x^k [x, y] x^{j-k-1} = \sum_{j=0}^{\infty} a_j j z x^{j-1} = z f'(x).$$

which shows the first equation. The second equation follows from this when we set  $f(x) = e^x y^{l-1}$ 

$$e^{x}y^{l} = ye^{x}y^{l-1} = (y-z)e^{x}y^{l-1} = \dots = (y+z)^{l}e^{x},$$

and from here we get that

$$e^{x}f(y) = e^{x}\sum_{j=0}^{\infty}a_{j}y^{j} = \sum_{j=0}^{\infty}a_{j}(y+z)^{j}e^{x} = f(y+z)e^{x}.$$

*Proof of Proposition 3.1.* The equations in (3.6) follows from the system of equations

$$\begin{split} \mathbf{T}\mathbf{q}_1 &= (\mathbf{q}_1 + \mathbf{q}_2)\mathbf{T}, \\ \mathbf{T}(\mathbf{p}_1 + \mathbf{p}_2) &= (\mathbf{p}_1 + \mathbf{q}_2)\mathbf{T}, \\ \mathbf{T}(\mathbf{p}_1 + \mathbf{q}_2) &= (\mathbf{p}_1 + \mathbf{q}_2)\mathbf{T}, \\ \mathbf{T}e^{2\pi\,\mathbf{b}\,\mathbf{p}_1} &= (e^{2\pi\,\mathbf{b}\,\mathbf{p}_1} + e^{2\pi\,\mathbf{b}(\mathbf{q}_1 + \mathbf{p}_2)})\mathbf{T}, \end{split}$$

where  $\mathbf{T} = e^{2\pi i \mathbf{p}_1 \mathbf{q}_2} \psi(\mathbf{q}_1 - \mathbf{q}_2 + \mathbf{p}_2)$ . We prove them one by one below using Lemma B.1.

$$\begin{aligned} \mathbf{T}\mathbf{q}_{1} &= e^{2\pi i \mathbf{p}_{1}\mathbf{q}_{2}}\psi(\mathbf{q}_{1} + \mathbf{p}_{2} - \mathbf{q}_{2})\mathbf{q}_{1} \\ &= e^{2\pi i \mathbf{p}_{1}\mathbf{q}_{2}}\mathbf{q}_{1}\psi(\mathbf{q}_{1} + \mathbf{p}_{2} - \mathbf{q}_{2}) \\ &= (\mathbf{q}_{1}e^{2\pi i \mathbf{p}_{1}\mathbf{q}_{2}} + \mathbf{q}_{2}e^{2\pi i \mathbf{p}_{1}\mathbf{q}_{2}})\psi(\mathbf{q}_{1} + \mathbf{p}_{2} - \mathbf{q}_{2}) \\ &= (\mathbf{q}_{1} + \mathbf{q}_{2})\mathbf{T}. \end{aligned}$$

$$\begin{split} \mathbf{T}(\mathbf{p}_{1}+\mathbf{p}_{2}) &= e^{2\pi i \mathbf{p}_{1} \mathbf{q}_{2}} \psi(\mathbf{q}_{1}+\mathbf{p}_{2}-\mathbf{q}_{2})(\mathbf{p}_{1}+\mathbf{p}_{2}) \\ &= e^{2\pi i \mathbf{p}_{1}+\mathbf{q}_{2}}(\mathbf{p}_{1}+\mathbf{p}_{2})\psi(\mathbf{q}_{1}+\mathbf{p}_{2}-\mathbf{q}_{2}) \\ &= \left\{\mathbf{p}_{1}e^{2\pi i \mathbf{p}_{1} \mathbf{q}_{2}}+\mathbf{p}_{2}e^{2\pi i \mathbf{p}_{1} \mathbf{q}_{2}}-\mathbf{p}_{1}e^{2\pi i \mathbf{p}_{1} \mathbf{q}_{2}}\right\}\psi(\mathbf{q}_{1}+\mathbf{p}_{2}-\mathbf{q}_{2}) \\ &= \mathbf{p}_{2}\mathbf{T}, \end{split}$$

where the second equality is true since  $[\mathbf{q}_1 + \mathbf{p}_2 - \mathbf{q}_2, \mathbf{p}_1 + \mathbf{p}_2] = 0.$ 

$$\begin{split} \mathbf{T}(\mathbf{p}_1 + \mathbf{q}_2) &= e^{2\pi i \mathbf{p}_1 \mathbf{q}_2} \psi(\mathbf{q}_1 + \mathbf{p}_2 - \mathbf{q}_2)(\mathbf{p}_1 + \mathbf{q}_2) \\ &= e^{2\pi i \mathbf{p}_1 \mathbf{q}_2}(\mathbf{p}_1 + \mathbf{q}_2) \psi(\mathbf{q}_1 + \mathbf{p}_2 - \mathbf{q}_2) \\ &= (\mathbf{p}_1 + \mathbf{q}_2) \mathbf{T}, \end{split}$$

where second equality is true since  $[q_1 + p_2 - q_2, p_1 + q_2] = 0$ .

$$\begin{aligned} \mathbf{T}e^{2\pi \,\mathbf{b}\,\mathbf{p}_{1}} &= e^{2\pi i\mathbf{p}_{1}\mathbf{q}_{2}}\psi(\mathbf{q}_{1}-\mathbf{q}_{2}+\mathbf{p}_{2})e^{2\pi \,\mathbf{b}\,\mathbf{p}_{1}} \\ &= \psi(\mathbf{q}_{1}-\mathbf{p}_{1}+\mathbf{p}_{2})e^{2\pi i\mathbf{p}_{1}\mathbf{q}_{2}}e^{2\pi \,\mathbf{b}\,\mathbf{p}_{1}} \\ &= \psi(\mathbf{q}_{1}-\mathbf{p}_{1}+\mathbf{p}_{2})e^{2\pi \,\mathbf{b}\,\mathbf{p}_{1}}e^{2\pi i\mathbf{p}_{1}\mathbf{q}_{2}} \\ &= e^{2\pi \,\mathbf{b}\,\mathbf{p}_{1}}\psi(\mathbf{q}_{1}-\mathbf{p}_{1}+\mathbf{p}_{2}+i\,\mathbf{b})e^{2\pi i\mathbf{p}_{1}\mathbf{q}_{2}} \\ &= e^{2\pi \,\mathbf{b}\,\mathbf{p}_{1}}\left(1+e^{2\pi \,\mathbf{b}(\mathbf{q}_{1}-\mathbf{p}_{1}+\mathbf{p}_{2}+\frac{i\,\mathbf{b}}{2}}\right)\psi(\mathbf{q}_{1}-\mathbf{p}_{1}+\mathbf{p}_{2})e^{2\pi i\mathbf{p}_{1}\mathbf{q}_{2}} \\ &= \left(e^{2\pi \,\mathbf{b}\,\mathbf{p}_{1}}+e^{2\pi i\,\mathbf{b}(\mathbf{q}_{1}-\mathbf{p}_{1}+\mathbf{p}_{2}+\frac{i\,\mathbf{b}}{2}}\right)\Psi(\mathbf{q}_{1}-\mathbf{p}_{1}+\mathbf{p}_{2})e^{2\pi i\mathbf{p}_{1}\mathbf{q}_{2}} \end{aligned}$$

where in the last equality we use the Baker–Campbell–Hausdorff formula.  $\hfill \square$ 

### References

- [A1] J.E. Andersen, Deformation Quantization and Geometric Quantization of Abelian Moduli Spaces., Comm. of Math. Phys. 255:727–745, 2005.
- [A2] J. E. Andersen. Asymptotic faithfulness of the quantum SU(n) representations of the mapping class groups. Annals of Mathematics. **163**:347–368, 2006.
- [AH] J.E. Andersen & S.K. Hansen. Asymptotics of the quantum invariants for surgeries on the figure 8 knot. *Journal of Knot theory and its Ramifications*, 15:479–548, 2006.
- [AGr1] J.E. Andersen & J. Grove. Automorphism Fixed Points in the Moduli Space of Semi-Stable Bundles. *The Quarterly Journal of Mathematics*, 57:1– 35, 2006.
- [AMU] J.E. Andersen, G. Masbaum & K. Ueno. Topological Quantum Field Theory and the Nielsen-Thurston classification of M(0, 4). Math. Proc. Cambridge Philos. Soc. 141:477–488, 2006.
- [AU1] J. E. Andersen & K. Ueno. Abelian Conformal Field theories and Determinant Bundles. *International Journal of Mathematics*. 18:919–993, 2007.
- [AU2] J. E. Andersen & K. Ueno, Constructing modular functors from conformal field theories. Journal of Knot theory and its Ramifications. 16(2):127–202, 2007.
- [A3] J. E. Andersen. The Nielsen-Thurston classification of mapping classes is determined by TQFT. math.QA/0605036. J. Math. Kyoto Univ. 48(2):323– 338, 2008.
- [A4] J.E. Andersen. Asymptotics of the Hilbert-Schmidt Norm of Curve Operators in TQFT. Letters in Mathematical Physics 91:205–214, 2010.
- [A5] J.E. Andersen. Toeplitz operators and Hitchin's projectively flat connection. in *The many facets of geometry: A tribute to Nigel Hitchin*, Edited by O. García-Prada, Jean Pierre Bourguignon, Simon Salamon, 177–209, Oxford Univ. Press, Oxford, 2010.
- [AB1] J.E. Andersen & J. L. Blaavand, Asymptotics of Toeplitz operators and applications in TQFT, *Traveaux Mathématiques*, 19:167–201, 2011.
- [AGa1] J.E. Andersen & N.L. Gammelgaard. Hitchin's Projectively Flat Connection, Toeplitz Operators and the Asymptotic Expansion of TQFT Curve Operators. Grassmannians, Moduli Spaces and Vector Bundles, 1–24, Clay Math. Proc., 14, Amer. Math. Soc., Providence, RI, 2011.

- [AU3] J. E. Andersen & K. Ueno. Modular functors are determined by their genus zero data. Quantum Topology. 3:255–291, 2012.
- [A6] J. E. Andersen. Hitchin's connection, Toeplitz operators, and symmetry invariant deformation quantization. *Quantum Topol.* 3(3-4):293–325, 2012.
- [AGL] J.E. Andersen, N.L. Gammelgaard & M.R. Lauridsen, Hitchin's Connection in Metaplectic Quantization, *Quantum Topology* 3:327–357, 2012.
- [AHi] J. E. Andersen & B. Himpel. The Witten-Reshetikhin-Turaev invariants of finite order mapping tori II Quantum Topology. 3:377–421, 2012.
- [A7] J. E. Andersen. The Witten-Reshetikhin-Turaev invariants of finite order mapping tori I. Journal f
  ür Reine und Angewandte Mathematik. 681:1–38, 2013.
- [AG] J. E. Andersen and N. L. Gammelgaard. The Hitchin-Witten Connection and Complex Quantum Chern-Simons Theory. arXiv:1409.1035, 2014.
- [AK1] J. E. Andersen and R. Kashaev. A TQFT from Quantum Teichmüller Theory. Comm. Math. Phys. 330(3):887–934, 2014.
- [AK1a] J.E. Andersen & R.M. Kashaev, Quantum Teichmüller theory and TQFT. XVIIth International Congress on Mathematical Physics, World Sci. Publ., Hackensack, NJ, 684–692, 2014.
- [AK1b] J.E. Andersen & R.M. Kashaev. Faddeev's quantum dilogarithm and state-integrals on shaped triangulations. In *Mathematical Aspects of Quan*tum Field Theories, Editors D. Calaque and Thomas Strobl, *Mathematical Physics Studies*. XXVIII:133–152, 2015.
- [AHJMMc] J. E. Andersen, B. Himpel, S. F. Jørgensen, J. Martens and B. McLellan, "The Witten-Reshetikhin-Turaev invariant for links in finite order mapping tori I", Advances in Mathematics, 304:131–178, 2017.
- [AK2] J. E. Andersen and R. Kashaev. A new formulation of the Teichmüller TQFT. arXiv:1305.4291, 2013.
- [AK3] J. E. Andersen and R. Kashaev. Complex Quantum Chern-Simons. arXiv:1409.1208, 2014.
- [AM] J. E. Andersen and S. Marzioni, Level N Teichmüller TQFT and Complex Chern–Simons Theory, Travaux Mathématiques 25 (2017), 97–146, Preprint, 2016.
- [AU4] J. E. Andersen & K. Ueno. Construction of the Witten-Reshetikhin-Turaev TQFT from conformal field theory. *Invent. Math.* 201(2):519–559, 2015.

- [AE] J.E. Andersen & J.K. Egsgaard, The equivalence of the Hitchin connection and the Knizhnik–Zamolodchikov connection, In preparation.
- [At] M. Atiyah. Topological quantum field theories. Inst. Hautes Études Sci. Publ. Math., 68:175–186, 1988.
- [BB] S. Baseilhac and R. Benedetti. Quantum hyperbolic geometry. Algebr. Geom. Topol. 7:845–917, 2007.
- [BHMV1] C. Blanchet, N. Habegger, G. Masbaum & P. Vogel. Three-manifold invariants derived from the Kauffman Bracket. *Topology.* 31:685–699, 1992.
- [BHMV2] C. Blanchet, N. Habegger, G. Masbaum & P. Vogel. Topological Quantum Field Theories derived from the Kauffman bracket. *Topology.* 34:883– 927, 1995.
- [DFM] R. Dijkgraaf, H. Fuji, and M. Manabe. The volume conjecture, perturbative knot invariants, and recursion relations for topological strings. *Nuclear Phys. B*, 849(1):166–211, 2011.
- [Di] T. Dimofte, Complex Chern-Simons theory at level k via the 3d-3d correspondence. Comm. Math. Phys. 339(2):619–662, 2015.
- [DGLZ] T. Dimofte, S. Gukov, J. Lenells, and D. Zagier. Exact results for perturbative Chern-Simons theory with complex gauge group. *Commun. Number Theory Phys.*, 3(2):363–443, 2009.
- [FK] L. D. Faddeev and R. M. Kashaev. Quantum dilogarithm. Modern Phys. Lett. A. 9(5):427–434, 1994.
- [F] L. D. Faddeev. Discrete Heisenberg-Weyl group and modular group. Lett. Math. Phys. 34(3):249–254, 1995.
- [FC] V. V. Fock and L. O. Chekhov. Quantum Teichmüller spaces. Teoret. Mat. Fiz. 120(3):511–528, 1999.
- [GKT] N. Geer, R. Kashaev, and V. Turaev. Tetrahedral forms in monoidal categories and 3-manifold invariants. J. Reine Angew. Math. 673:69–123, 2012.
- [H1] K. Hikami. Hyperbolicity of partition function and quantum gravity. Nuclear Phys. B. 616(3):537–548, 2001.
- [H2] K. Hikami. Generalized volume conjecture and the A-polynomials: the Neumann-Zagier potential function as a classical limit of the partition function. J. Geom. Phys. 57(9):1895–1940, 2007.

- [K1] R. M. Kashaev. Quantum dilogarithm as a 6j-symbol. Modern Phys. Lett. A, 9(40):3757–3768, 1994.
- [K2] R. M. Kashaev. Quantization of Teichmüller spaces and the quantum dilogarithm. Lett. Math. Phys., 43(2):105–115, 1998.
- [K3] R. M. Kashaev. The Liouville central charge in quantum Teichmüller theory. *Tr. Mat. Inst. Steklova*, **226**(Mat. Fiz. Probl. Kvantovoi Teor. Polya):72–81, 1999.
- [K4] R. M. Kashaev. On the spectrum of Dehn twists in quantum Teichmüller theory. In *Physics and combinatorics*, 2000 (Nagoya), pages 63–81. World Sci. Publ., River Edge, NJ, 2001.
- [KLV] R. M. Kashaev, F. Luo, and G. Vartanov. A TQFT of Turaev-Viro type on shaped triangulations, 2012.
- [L] Y. Laszlo. Hitchin's and WZW connections are the same. J. Diff. Geom. 49(3):547–576, 1998.
- [N] J.-J. K. Nissen. The Andersen–Kashaev TQFT. PhD thesis, Aarhus University, 2014.
- [P] R. C. Penner. The decorated Teichmüller space of punctured surfaces. Comm. Math. Phys., 113(2):299–339, 1987.
- [RT1] N. Reshetikhin & V. Turaev. Ribbon graphs and their invariants derived from quantum groups *Comm. Math. Phys.* 127:1–26, 1990.
- [RT2] N. Reshetikhin & V. Turaev. Invariants of 3-manifolds via link polynomials and quantum groups *Invent. Math.* 103:547–597, 1991.
- [S] G. B. Segal. The definition of conformal field theory. In Differential geometrical methods in theoretical physics (Como, 1987), volume 250 of NATO Adv. Sci. Inst. Ser. C Math. Phys. Sci., pages 165–171. Kluwer Acad. Publ., Dordrecht, 1988.
- [T] V. G. Turaev. Quantum invariants of knots and 3-manifolds, volume 18 of de Gruyter Studies in Mathematics. Walter de Gruyter & Co., Berlin, 1994.
- [TV] V. G. Turaev and O. Y. Viro. State sum invariants of 3-manifolds and quantum 6*j*-symbols. *Topology*, **31**(4):865–902, 1992.
- [W] E. Witten. Topological quantum field theory. Comm. Math. Phys., 117(3):353–386, 1988.

Jørgen Ellegaard Andersen and Jens-Jakob Kratmann Nissen Center for Quantum Geometry of Moduli Spaces Department of Mathematics University of Aarhus DK-8000, Denmark

Travaux mathématiques, Volume 25 (2017), 97–146, © by the authors

# Level N Teichmüller TQFT and Complex Chern–Simons Theory

by Jørgen Ellegaard Andersen and Simone Marzioni<sup>1</sup>

#### Abstract

In this manuscript we review the construction of the Teichmüller TQFT due to Andersen and Kashaev. We further upgrading it to a theory dependent on an extra odd integer N using results developed by Andersen and Kashaev in their work on complex quantum Chern-Simons theory. We also describe how this theory is related with quantum Chern–Simons Theory at level N with gauge group PSL(2,  $\mathbb{C}$ ).

## 1 Introduction

In this paper we review Andersen and Kashaev's construction of the Teichmüller TQFT from [AK1] making it dependent on an extra odd integer N, called the level. The original work of [AK1] corresponds to the choice N = 1, and emerged as an extension to a 3-dimensional theory of the representations one obtains from Quantum Teichmüller Theory [K3]. In particular, it defines a class of quantum invariants for hyperbolic knots, dependent on a continuous parameter b. The level N Teichmüller TQFT is an analogously upgrade of representations in Quantum Teichmüller theory and it depends on a pair of parameters (b, N), one continuous and one discrete. Such quantum theory is related to the level N Chern-Simons theory with gauge group  $PSL(2, \mathbb{C})$  via the level N Weil-Gel'fand-Zak transform. Such a relation was proposed in [AK3], and here we show it in a more tight way for the four punctured sphere. One of the main ingredient in the construct of the Teichmüller TQFT is the quantum dilogarithm, that is a function  $D_{\rm b}: \mathbb{R} \times \mathbb{Z}/N\mathbb{Z} \to \mathbb{C}$ , satisfying some particular properties. Such functions were introduced in [AK3], which for N = 1 is Faddeev's original quantum dilogarithm. The theory we get has different and interesting unitarity behaviour depending on the pair of parameters (b, N): for level N = 1 the theory is unitary whenever b > 0 or |b| = 1 while for higher level N > 1 the unitarity is only manifest when  $|\mathbf{b}| = 1$  while in the case  $\mathbf{b} > 1$  the behaviour is more exotic. We will consider

<sup>&</sup>lt;sup>1</sup>Work supported in part by the center of excellence grant "Center for Quantum Geometry of Moduli Spaces" from the Danish National Research Foundation (DNRF95).

both situations here and we will use the setting b > 0 to present some asymptotic properties in the limit  $b \rightarrow 0$ . The Teichmüller TQFT can be used to define knot invariants starting from triangulations of their complements. In this presentation we update the examples presented in [AK1] to the level N setting together with their asymptotic analysis. For the simplest hyperbolic knot we show the appearance of the Baseilhac–Benedetti invariant from [BB] in such a limit.

It is an interesting challenge how the TQFT's which are reviewed in this paper are related to the Witten-Reshetikhin-Turaev TQFT's [W1, RT1, RT2, BHMV1, BHMV2, B] and in particular how they are related to the geometric construction of these TQFT's [ADW, Hit2, Las, TUY, AU1, AU2, AU3, AU4] and to Witten's proposal for the construction of the complex quantum Chern-Simons theory [W2], which can actually be constructed from a purely mathematical point of view [AG], resulting also in the mathematically well-defined Hitchin-Witten connection in the bundle of quantizations of the moduli space of flat  $SL(n, \mathbb{C})$ -connections over Teichmüller space. In the classical case of compact groups, the description of the representations of the mapping class groups via the monodromy of the Hitchin connections turned out to be very useful to prove deep properties about these representations [A1, A2, A3, A4, AH, AHJMMc], some of which also uses the theory of Toeplitz operators [BMS, KS]. Understanding how these kinds of results can be extended to the complex quantum theory discussed in this paper will be very interesting and most likely involve using Higgs bundles techniques [Hit1]. Certainly we have already seen the start of this with the Verlinde formula for Higgs bundle moduli space [AGP].

The paper is organised as follows. In section 2 we recall the definition of the (decorated) Ptolemy groupoid of punctured surfaces, which is the combinatorial foundation over which Quantum Teichmüller Theory is defined. In section 3 we recall the quantum dilogarithm  $D_b$ , and we list some of its properties. The function  $D_b$  was introduced in [AK3] for the first time, but some of its properties that we list here are not present in the literature. In section 5 we carry out the quantization of the moduli space of PSL(2,  $\mathbb{C}$ ) flat connections over a four punctured sphere with unipotent holonomies around the punctures. We follow the prescriptions of geometric quantisation, together with a choice of real polarisation, and we connect the resulting algebra of observables to the  $L^2(\mathbb{R} \times \mathbb{Z}/N\mathbb{Z})$  representations of quantum Teichmüller TQFT functor  $F_b^{(N)}$  mirroring the construction in [AK1] and we study some examples and their asymptotic behaviour. It would of course be interesting to go through all the examples treated in [AN] in this volume for the level N theory as well.

#### Acknowledgements

We would like to thank Rinat Kashaev for many interesting discussions.

# 2 Ptolemy Groupoid

Let  $\Sigma_{g,s}$  be a surface of genus g with s punctures, such that s > 0 and 2-2g+s < 0.

**Definition 2.1.** An *ideal arc*  $\alpha$  is the homotopy class relative endpoints of the embedding of a path in  $\Sigma_{g,s}$ , such that the endpoints are punctures of the surface. An *ideal triangle* is a triangle with the vertices removed, such that the edges are ideal arcs.

An *ideal triangulation*  $\tau$  of  $\Sigma_{g,s}$  is a collection of disjoint ideal arcs such that  $\Sigma_{g,s} \setminus \tau$  is a collection of interiors of ideal triangles.

Given an ideal triangulation  $\tau$ ,  $\Delta_j(\tau)$  will denote the set of its *j*-dimensional cells.

**Definition 2.2.** A decorated ideal triangulation of  $\Sigma_{g,s}$  is an ideal triangulation  $\tau$  up to isotopy relative to the punctures, together with the choice of a distinguished corner in each ideal triangle and a bijective ordering map

$$\overline{\tau}: \{1, \ldots, s\} \ni j \mapsto \overline{\tau}_j \in \Delta_2(\tau).$$

We denote the set of all decorated ideal triangulation as  $\dot{\Delta} = \dot{\Delta}(\Sigma_{g,s})$ .

When we say that  $\tau$  is a decorated ideal triangulation we mean that  $\tau$  is the set of decorated ideal triangles in the triangulation.

One of the main interests in quantizing moduli spaces is the consequent construction of representations of (central extensions of) the mapping class group of the surfaces [W2, Hit1, Las, A1, AU4, AG]. Quantum Teichmüller theory produce instead representations of a bigger object called the (decorated) *Ptolemy Groupoid* that we are going to introduce now.

Recall that, given a group G acting freely on a set X, we can define an associated groupoid  $\mathcal{G}$  as follows. The objects of  $\mathcal{G}$  are G-orbits in X while morphisms are G-orbits in  $X \times X$  with respect to the diagonal action. Then for any  $x \in X$  we can consider the object [x] and for any pair  $(x, y) \in X \times X$  we can consider the morphism [x, y]. When [y] = [u] there will be a  $g \in G$  so that gu = y and we can define the composition [x, y][u, v] = [x, gv]. The unit for [x] is given by [x, x]. If the action of G is transitive, we would get an actual group. We will abbreviate  $[x_1, x_2][x_2, x_3] \cdots [x_{n-1}, x_n]$  with  $[x_1, x_2, \ldots, x_n]$ .

We define the decorated Ptolemy groupoid  $\mathcal{G}(\Sigma_{g,s})$  of a punctured surface  $\Sigma_{g,s}$ following the above recipe. The set we consider is the set  $\dot{\Delta}$  of decorated triangulations  $\tau$  of  $\Sigma_{g,s}$ . The free group action is the one of the mapping class group MCG<sub>g,s</sub> acting on  $\dot{\Delta}$ . This action is not transitive, meaning that not all pairs of decorated ideal triangulations can be related by a mapping class group element. However in the language of groupoids, we can still describe generators and relations for the morphism groups. For  $\tau \in \dot{\Delta}$  there are three kind of generators  $[\tau, \tau^{\sigma}]$ ,  $[\tau, \rho_i \tau]$  and  $[\tau, \omega_{i,j} \tau]$ , where  $\tau^{\sigma}$  is obtained by applying the permutation  $\sigma \in \mathbb{S}_{|\tau|}$  to the ordering of triangles in  $\tau$ ,  $\rho_i \tau$  is obtained by changing the distinguished corner in the triangle  $\bar{\tau}_i \in \tau$  as in Figure 1, and  $\omega_{i,j}$  is obtained by applying a decorated diagonal exchange to the quadrilateral made of the two decorated ideal triangles  $\bar{\tau}_i$  and  $\bar{\tau}_j$  as in Figure 2.



Figure 1: Transformation  $\rho_i$ .



Figure 2: Transformation  $\omega_{ij}$ .

The relations are usually grouped in two sets, the first being

(2.1) 
$$[\tau, \tau^{\alpha}, (\tau^{\alpha})^{\beta}] = [\tau, \tau^{\alpha\beta}], \quad \alpha, \beta \in \mathbb{S}_{\tau},$$

(2.2) 
$$[\tau, \rho_i \tau, \rho_i \rho_i \tau, \rho_i \rho_i \tau] = \mathrm{id}_{[\tau]},$$

(2.3) 
$$[\tau, \omega_{i,j}\tau, \omega_{i,k}\omega_{i,j}\tau, \omega_{j,k}\omega_{i,k}\omega_{i,j}\tau] = [\tau, \omega_{j,k}\tau, \omega_{i,j}\omega_{j,k}\tau]$$

(2.4) 
$$[\tau, \omega_{i,j}\tau, \rho_i\omega_{i,j}\tau, \omega_{j,i}\rho_i\omega_{i,j}\tau] = [\tau, \tau^{(i,j)}, \rho_j\tau^{(i,j)}, \rho_i\rho_j\tau^{(i,j)}]$$

The first two relations are obvious, the third is called the Pentagon Relation and the fourth is called the Inversion Relation.

The second set of relations, are commutation relations

(2.5) 
$$[\tau, \rho_i \tau, \rho_i \tau^{\sigma}] = [\tau, \tau^{\sigma}, \rho_{\sigma^{-1}(i)} \tau^{\sigma}],$$

(2.6) 
$$[\tau, \omega_{i,j}\tau, (\omega_{i,j}\tau)^{\sigma}] = [\tau, \tau^{\sigma}, \omega_{\sigma^{-1}(i)\sigma^{-1}(j)}\tau^{\sigma}],$$

(2.7) 
$$[\tau, \rho_j \tau, \rho_j \rho_i \tau] = [\tau, \rho_i \tau, \rho_i \rho_j \tau],$$

(2.8) 
$$[\tau, \rho_i \tau, \omega_{j,k} \rho_i \tau] = [\tau, \omega_{j,k} \tau, \rho_i \omega_{j,k} \tau], \ i \notin \{j, k\},$$

(2.9) 
$$[\tau, \omega_{i,j}\tau, \omega_{k,l}\omega_{i,j}\tau] = [\tau, \omega_{k,l}\tau, \omega_{i,j}\omega_{k,l}\tau], \{i, j\} \cap \{k, l\} = \emptyset$$

Teichmüller TQFT at level N

To every decorated ideal triangulation  $\tau \in \dot{\Delta}$  it is possible to associate a simple symplectic space  $\mathcal{R}(\tau)$ , called the space of *Ratio Coordinates*. We summarize here its relation with the Ptolemy groupoid and refer to [K3] for a detailed introduction to ratio coordinates and their relation to the Teichmüller space. Let  $M \equiv 2g 2 + s = |\tau|$  be the number of ideal triangles, then  $\mathcal{R}(\tau) \equiv (\mathbb{R}_{>0} \times \mathbb{R}_{>0})^M$ . Let  $x^j \equiv (x_1^j, x_2^j) \in \mathbb{R}_{>0} \times \mathbb{R}_{>0}$ , for  $j = 1, \ldots, M$  be the coordinates associated to the ideal triangle  $\bar{\tau}_j \in \Delta_2(\tau)$ . The symplectic form that we consider on  $\mathcal{R}(\tau)$  is

(2.10) 
$$\omega_{\tau} \equiv \sum_{j=1}^{M} \frac{\mathrm{d}x_1^j}{x_1^j} \wedge \frac{\mathrm{d}x_2^j}{x_2^j}$$

Now we want to describe the action of  $\mathcal{G}(\Sigma_{g,s})$  as symplectomorphisms between these spaces. The morphisms  $[\tau, \tau^{\sigma}]$  act by permuting the coordinates in  $\mathcal{R}(\tau)$ . The morphism  $[\tau, \rho_i \tau]$  acts as the identity on any pair  $x = (x_1, x_2)$  corresponding to ideal triangles different from  $\bar{\tau}_i$  and as  $(x_1, x_2) = x \mapsto y = (\frac{x_2}{x_1}, \frac{1}{x_1})$  for the pair of coordinates corresponding to  $\bar{\tau}_i$ . Finally the action of  $[\tau, \omega_{i,j}\tau]$  is the identity on  $\bar{\tau}_k$  for  $k \neq i, j$  while letting  $x = (x_1, x_2)$  and  $y = (y_1, y_2)$  be the coordinates corresponding to the triangles  $\bar{\tau}_i$  and  $\bar{\tau}_j$  respectively, and letting  $u = (u_1, u_2)$  and  $v = (v_1, v_2)$  be the coordinates of the triangles  $\overline{\omega_{i,j}\tau}_i$  and  $\overline{\omega_{i,j}\tau}_j$ , then we have  $u = x \bullet y$  and v = x \* y where

(2.11) 
$$x \bullet y := (x_1y_1, x_1y_2 + x_2)$$
$$x * y := \left(\frac{y_1x_2}{x_1y_2 + x_2}, \frac{y_2}{x_1y_2 + x_2}\right)$$

Let  $\widetilde{\Delta}$  be the set of pairs  $(\tau, \mathcal{R}(\tau)), \tau \in \dot{\Delta}$ . Then the space  $\mathcal{R}(\Sigma_{g,s})$  is defined as the quotient of  $\widetilde{\Delta}$  by the action of  $\mathcal{G}(\Sigma_{g,s})$  as described above. This space of coordinates is now independent of the triangulation. For more details on the (decorated or not) Ptolemy groupoid see [P], [FK], [K6][K4].

# 3 Quantum Dilogarithm

In this section we recall the quantum dilogarithm  $D_b$  over  $\mathbb{A}_N$  and we state some of their properties.

**Definition 3.1** (q-Pochammer Symbol). Let  $x, q \in \mathbb{C}$ , such that |q| < 1. Define the q-Pochammer Symbol of x as

$$(x;q)_{\infty} := \prod_{i=0}^{\infty} (1 - xq^i)$$

**Theorem 3.2.** Let X, Y satisfying XY = qYX. Then the following five-term relation holds true

(3.1) 
$$(Y;q)_{\infty} (X;q)_{\infty} = (X;q)_{\infty} (-YX;q)_{\infty} (Y;q)_{\infty}.$$

**Definition 3.3** (Faddeev's Quantum Dilogarithm [F]). Let  $z, b \in \mathbb{C}$  be such that Re  $b \neq 0$ ,  $|\operatorname{Im}(z)| < |\operatorname{Im}(c_b)|$ , where  $c_b := i(b+b^{-1})/2$ . Let  $C \subset \mathbb{C}$ ,  $C = \mathbb{R} + i0$  be a contour equal to the the real line outside a neighborhood of the origin that avoid the singularity in 0 going in the upper half plane. Faddeev's quantum dilogarithm is defined to be

(3.2) 
$$\Phi_{\rm b}(z) = \exp\left(\int_C \frac{e^{-2izw} \mathrm{d}w}{4\sinh(wb)\sinh(wb^{-1})w}\right).$$

It is evident that  $\Phi_{\rm b}$  is invariant under the following changes of parameter

so that our choice of b can be restricted to the first quadrant

which implies

$$(3.5) Im(b2) \ge 0.$$

Faddeev's quantum dilogarithm has a lot of other interesting properties and applications, see for example [F],[FK],[FKV] and [V].

Let  $N \ge 1$  be a positive *odd* integer. Then, following [AK3], we can define a *quantum dilogarithm* over  $\mathbb{A}_N$  as follows

(3.6) 
$$D_{\rm b}(x,n) := \prod_{j=0}^{N-1} \Phi_{\rm b}\left(\frac{x}{\sqrt{N}} + (1-N^{-1})c_{\rm b} - i{\rm b}^{-1}\frac{j}{N} - i{\rm b}\left\{\frac{j+n}{N}\right\}\right)$$

where  $\{p\}$  is the fractional part of p, and  $\Phi_{\rm b}$  is the Faddeev's quantum dilogarithm. Of course for N = 1 we have just  $\Phi_{\rm b}(x)$ . The function  $D_{\rm b}$  was introduced in [AK3] only for  $|{\rm b}| = 1$ . It satisfies a series properties that we are going to list.

Lemma 3.4 (Inversion Relation [AK3]).

$$D_{\rm b}(x,n)D_{\rm b}(-x,-n) = e^{\pi i x^2} e^{-\pi i n(n+N)/N} \zeta_{N,inv}^{-1}$$

where

(3.7) 
$$\zeta_{N,inv} = e^{\pi i (N + 2c_{\rm b}^2 N^{-1})/6}.$$

Unitarity properties are different in the two situations  $|\mathbf{b}| = 1$  or  $\mathbf{b} \in \mathbb{R}$ .

Lemma 3.5 (Unitarity).

(3.8) 
$$\overline{\mathbf{D}_{\mathbf{b}}(x,n)} = \mathbf{D}_{\mathbf{b}}(\bar{x},n)^{-1} \qquad if \ |\mathbf{b}| = 1,$$

(3.9) 
$$\overline{\mathbf{D}_{\mathbf{b}}(x,n)} = \mathbf{D}_{\mathbf{b}}(\bar{x},-n)^{-1} \qquad if \, \mathbf{b} \in \mathbb{R}_{>0}$$
Remark 3.6. One can see that

(3.10) 
$$D_{b}(x, -n) = D_{b^{-1}}(x, n)$$

just by the Definition 3.6 for  $D_{b^{-1}}$  and carefully substituting  $j + n \mapsto j'$ . In particular the unitarity for b > 0 can be re-expressed as

(3.11) 
$$\overline{\mathbf{D}_{\mathbf{b}}(x,n)} = (\mathbf{D}_{\mathbf{b}^{-1}}(x,n))^{-1}$$

Lemma 3.7 (Faddeev's difference equations). Let

(3.12) 
$$\chi^{\pm}(x,n) \equiv e^{2\pi \frac{b^{\pm 1}}{\sqrt{N}}x} e^{\pm \frac{2\pi i n}{N}},$$

for every  $x, b \in \mathbb{C}$ ,  $\text{Im}(b) \neq 0$   $n, N \in \mathbb{Z}$  we have

(3.13) 
$$D_{b}\left(x+i\frac{b^{\pm 1}}{\sqrt{N}},n\pm 1\right) = D_{b}\left(x,n\right)\left(1+\chi^{\pm}(x,n)e^{-\pi i\frac{N-1}{N}}e^{\pi i\frac{b^{\pm 2}}{N}}\right)^{-1}$$

(3.14) 
$$D_{b}\left(x-i\frac{b^{\pm 1}}{\sqrt{N}}, n \mp 1\right) = D_{b}(x,n)\left(1+\chi^{\pm}(x,n)e^{\pi i\frac{N-1}{N}}e^{-\pi i\frac{b^{\pm 2}}{N}}\right)$$

**Proposition 3.8.** If Im(b) > 0 and Re(b) > 0 we have

(3.15) 
$$D_{\rm b}(x,n) = \frac{\left(\chi^+(x + \frac{c_{\rm b}}{\sqrt{N}}, n); q^2\omega\right)_{\infty}}{\left(\chi^-(x - \frac{c_{\rm b}}{\sqrt{N}}, n); \tilde{q}^2\overline{\omega}\right)_{\infty}}$$

where  $q = e^{i\pi \frac{\mathbf{b}^2}{N}}$ ,  $\tilde{q} = e^{-\pi i \frac{\mathbf{b}^{-2}}{N}}$ ,  $\omega = e^{\frac{2\pi i}{N}}$  and  $\chi^{\pm}(x,n) = e^{2\pi \frac{\mathbf{b}^{\pm 1}}{\sqrt{N}}x} e^{\pm \frac{2\pi i n}{N}}$ .

**Proposition 3.9.** The quantum dilogarithm  $D_b(x, n)$ , for Im(b) > 0 has poles

$$\begin{cases} x = \frac{c_{\rm b}}{\sqrt{N}} + i\frac{{\rm b}^{-1}}{\sqrt{N}}l + i\frac{{\rm b}}{\sqrt{N}}m\\ n = m - l \mod N \end{cases}$$

and zeros

$$\begin{cases} x = -\frac{c_{\rm b}}{\sqrt{N}} - i\frac{{\rm b}^{-1}}{\sqrt{N}}l - i\frac{{\rm b}}{\sqrt{N}}m\\ n = l - m \mod N \end{cases}$$

for  $l,m \in \mathbb{Z}_{>0}$ . Moreover its residue at  $(x_{l,m}, n_{l,m}) = \left(\frac{c_{\rm b}}{\sqrt{N}} + i\frac{b^{-1}}{\sqrt{N}}l + i\frac{b}{\sqrt{N}}m, m-l\right)$  is

(3.16) 
$$\frac{\sqrt{N}}{2\pi b^{-1}} \frac{(q^2\omega;q^2\omega)_{\infty}}{(\tilde{q}^2\overline{\omega};\tilde{q}^2\overline{\omega})_{\infty}} \frac{(-\tilde{q}^2\overline{\omega})^l(\tilde{q}^2\overline{\omega})^{l(l-1)/2}}{(q^2\omega;q^2\omega)_m (\tilde{q}^2\overline{\omega};\tilde{q}^2\overline{\omega})_l}$$

The following Summation Formula can be shown by a residue computation. It is well known for N = 1, i.e. for  $\Phi_{\rm b}$ , see [FKV] for example. **Theorem 3.10** (Summation Formula). Suppose Im(b) > 0 and N odd, and let  $u, v, w \in \mathbb{C}$  and  $a, b, c \in \mathbb{Z}/N\mathbb{Z}$  satisfy

(3.17) 
$$\operatorname{Im}\left(v + \frac{c_{\mathrm{b}}}{\sqrt{N}}\right) > 0, \quad \operatorname{Im}\left(-u + \frac{c_{\mathrm{b}}}{\sqrt{N}}\right) > 0, \quad \operatorname{Im}(v-u) < \operatorname{Im}(w) < 0.$$

Define

(3.18) 
$$\Psi(u, v, w, a, b, c) \equiv \int_{\mathbb{A}_N} \frac{\mathcal{D}_{\mathbf{b}}(x+u, a+d)}{\mathcal{D}_{\mathbf{b}}(x+v, b+d)} e^{2\pi i w x} e^{-2\pi i \frac{cd}{N}} \mathbf{d}(x, d)$$

Then we have that

 $\Psi(u,v,w,a,b,c)$ 

$$=\zeta_{0}\frac{\mathcal{D}_{\mathrm{b}}\left(v-u-w+\frac{c_{\mathrm{b}}}{\sqrt{N}},b-a-c\right)}{\mathcal{D}_{\mathrm{b}}\left(-w-\frac{c_{\mathrm{b}}}{\sqrt{N}},-c\right)\mathcal{D}_{\mathrm{b}}\left(v-u+\frac{c_{\mathrm{b}}}{\sqrt{N}},b-a\right)}e^{2\pi i w \left(\frac{c_{\mathrm{b}}}{\sqrt{N}}-u\right)}\omega^{ac}$$
$$=\zeta_{0}^{-1}\frac{\mathcal{D}_{\mathrm{b}}\left(w+\frac{c_{\mathrm{b}}}{\sqrt{N}},c\right)\mathcal{D}_{\mathrm{b}}\left(-v+u-\frac{c_{\mathrm{b}}}{\sqrt{N}},-b+a\right)}{\mathcal{D}_{\mathrm{b}}\left(-v+u+w-\frac{c_{\mathrm{b}}}{\sqrt{N}},-b+a+c\right)}e^{2\pi i w \left(-\frac{c_{\mathrm{b}}}{\sqrt{N}}-v\right)}\omega^{bc}$$

where  $\zeta_0 = e^{-\pi i (N - 4c_b^2 N^{-1})/12}$ .

**Remark 3.11.** Assumptions (3.17) even though sufficient are not optimal. Indeed they guarantee the theorem to hold true when the integration is performed along the real line, however we can deform the integration contour as long as

(3.19) 
$$|\arg(iz)| < \pi - \arg b$$
 z being one of  $\left\{ w, v - u - w, u - v - 2\frac{c_{\rm b}}{\sqrt{N}} \right\}$ 

Using the notation for the Fourier Kernels from (A.13) in Appendix A.2 we have that

**Proposition 3.12** (Fourier Transformation Formula, [AK3]). For N odd we have that

$$\begin{split} \int_{\mathbb{A}_{N}} \mathcal{D}_{\mathbf{b}}(x,n) \langle (x,n); (w,c) \rangle \mathrm{d}(x,n) &= \frac{e^{2\pi i w \frac{c_{\mathbf{b}}}{\sqrt{N}}}}{\mathcal{D}_{\mathbf{b}} \left( -w - \frac{c_{\mathbf{b}}}{\sqrt{N}}, -k \right)} e^{-\pi i (N - 4c_{\mathbf{b}}^{2}N^{-1})/12} \\ &= \mathcal{D}_{\mathbf{b}} \left( w + \frac{c_{\mathbf{b}}}{\sqrt{N}}, c \right) \overline{\langle (w,c) \rangle} e^{\pi i (N - 4c_{\mathbf{b}}^{2}N^{-1})/12} \\ \int_{\mathbb{A}_{N}} (\mathcal{D}_{\mathbf{b}}(x,n))^{-1} \langle (x,n); (w,c) \rangle \mathrm{d}(x,n) &= \frac{\langle (w,c) \rangle}{\mathcal{D}_{\mathbf{b}} \left( -w - \frac{c_{\mathbf{b}}}{\sqrt{N}}, -k \right)} e^{-\pi i (N - 4c_{\mathbf{b}}^{2}N^{-1})/12} \\ &= \mathcal{D}_{\mathbf{b}} \left( w + \frac{c_{\mathbf{b}}}{\sqrt{N}}, c \right) e^{-2\pi i w \frac{c_{\mathbf{b}}}{\sqrt{N}}} e^{\pi i (N - 4c_{\mathbf{b}}^{2}N^{-1})/12} \end{split}$$

**Proposition 3.13** (Integral Pentagon Relation). Let  $\widetilde{D_{b}}(x, n) \equiv \mathsf{F}_{N} \circ \mathcal{F}^{-1}(D_{b})(x, n)$ . We have the following integral relation

$$\begin{split} \langle (x,n);(y,m)\rangle \widetilde{\mathrm{D}_{\mathrm{b}}}(x,n) \widetilde{\mathrm{D}_{\mathrm{b}}}(y,m) \\ &= \int_{\mathbb{A}_{N}} \widetilde{\mathrm{D}_{\mathrm{b}}}(x-z,n-k) \widetilde{\mathrm{D}_{\mathrm{b}}}(z,k) \widetilde{\mathrm{D}_{\mathrm{b}}}(y-z,m-k) \langle (z,k) \rangle \mathrm{d}(z,k). \end{split}$$

Before we look at the asymptotic behavior of  $\Phi_b$  let us recall the classical dilogarithm function, defined on |z| < 1 by

(3.20) 
$$\operatorname{Li}_{2}(z) = \sum_{n \ge 1} \frac{z^{n}}{n^{2}}$$

and recall that it admits analytic continuation to  $\mathbb{C}\setminus [1,\infty]$  through the following integral formula

(3.21) 
$$\operatorname{Li}_{2}(z) = -\int_{0}^{z} \frac{\log(1-u)}{u} \mathrm{d}u$$

**Proposition 3.14.** We have the following behaviour when b > 0,  $b \rightarrow 0$  and x, n, N are fixed

(3.22) 
$$D_{\mathrm{b}}(\frac{x}{2\pi\mathrm{b}},n) = \mathrm{Exp}\left(\frac{\mathrm{Li}_{2}(-e^{x\sqrt{N}})}{2\pi\mathrm{i}b^{2}N}\right)\phi_{x}(n)(1+\mathcal{O}(b^{2}))$$

where  $\phi_x(n)$  is defined by  $\begin{cases} \phi_x(n+1) = \phi_x(n) \frac{(1-e^{x/\sqrt{N}}\overline{\omega}^{n+\frac{1}{2}})}{(1+e^{x\sqrt{N}})^{1/N}}\\ \phi_x(0) = (1+e^{x\sqrt{N}})^{-\frac{N-1}{2N}} \prod_{j=0}^{N-1} (1-e^{xN^{-\frac{1}{2}}}\overline{\omega}^{j+\frac{1}{2}})^{\frac{j}{N}} \end{cases}$ whenever N is odd.

**Remark 3.15.** The function  $\phi_x$  on the finite set  $\mathbb{Z}/N\mathbb{Z}$  is a cyclic quantum dilogarithm [FK],[K3], [K1]. Precisely  $\frac{1}{\phi_x}$  corresponds to the function  $\Psi_{\lambda}$  from Proposition 10 in [K3] with  $\lambda = e^{x/\sqrt{N}}$ .

The Hilbert space  $L^2(\mathbb{A}_N)$  is naturally isomorphic to the tensor product  $L^2(\mathbb{R})\otimes L^2(\mathbb{Z}/N\mathbb{Z}) \cong L^2(\mathbb{R})\otimes \mathbb{C}^N$ , see Appendix A.2. Let p and q two self-adjoint operators on  $L^2(\mathbb{R})$  satisfying

$$(3.23) \qquad \qquad [\mathsf{p},\mathsf{q}] = \frac{1}{2\pi i}$$

and let X and Y unitary operators satisfying

(3.24) 
$$YX = e^{2\pi i/N} XY, \quad X^N = Y^N = 1,$$

J.E. Andersen, S. Marzioni

together with the cross relations

$$(3.25) [p, X] = [p, Y] = [q, X] = [q, Y] = 0.$$

The equations in (3.24) imply that X and Y will have finite and the same spectrum, and this will be a subset of the set  $\mathbb{T}_N$  of all N-th complex roots of unity. Let

$$L_N: \mathbb{T}_N \longrightarrow \mathbb{Z}/N\mathbb{Z}$$

be the natural group isomorphism. We can define  $L_N(A)$ , by the spectral theorem, for any operator A of order N, such that it formally satisfies

$$A = e^{2\pi i \operatorname{L}_N(A)/N}$$

One has that

(3.26) 
$$\mathbf{L}_N(-e^{-\pi i/N}XY) = \mathbf{L}_N(X) + \mathbf{L}_N(Y).$$

For any function  $f : \mathbb{A}_N \longrightarrow \mathbb{C}$  recall the definition of  $\tilde{f}$  and the operator function  $f(\mathbf{x}, A) \equiv f(\mathbf{x}, \mathbf{L}_N(A))$  from Appendix A.2. The following Pentagon Identity for  $\mathbf{D}_{\mathbf{b}}$  was first proved in [AK3], where a projective ambiguity was undetermined and  $|\mathbf{b}| = 1$ .

**Lemma 3.16** (Pentagon Equation). Let p,q, X and Y be as above, then the following five-term relation holds

*Proof.* This is equivalent to the Integral Pentagon equation of Proposition 3.13. To see this we need to use equation (A.14) an all the five terms. Then we compare the coefficients of  $e^{2\pi i y q} Y^m e^{2\pi i x p} X^{-n}$  and get exactly the integral pentagon equation.

An alternative proof follows from the q-Pochammer presentation of  $D_b$  from Proposition 3.8.

#### 3.1 Charges

We are going to define a *charged* version of the dilogarithm. The charges will assume geometrical meaning in the construction of the partition function, however they already satisfy the purpose of turning all the conditional convergent integral relations of the dilogarithm  $D_b$  (e.g. Proposition 3.13 and 3.12) into absolutely convergent integrals.

Let a, b and c be three real positive numbers such that  $a + b + c = \frac{1}{\sqrt{N}}$ . We define the charged quantum dilogarithm

(3.28) 
$$\psi_{a,c}(x,n) := \frac{e^{-2\pi i c_{\mathrm{b}} a x}}{\mathrm{D}_{\mathrm{b}}(x - c_{\mathrm{b}}(a + c), n)}$$

106

From the Fourier transformation formula, Proposition 3.12, and the inversion formula in Lemma 3.4, we can deduce the following transformation formulas for  $\psi_{a,c}$  (recall notation (A.15) for the inverse Fourier transform)

**Lemma 3.17.** Suppose Im(b)(1 - |b|) = 0, then

(3.29) 
$$\overline{\psi}_{a,c}(x,k) = \psi_{c,b}(x,k) \langle x,k \rangle e^{-\pi i c_b^2 a(a+2c)} \zeta_0$$

(3.30) 
$$\overline{\psi_{a,c}(x,k)} = \psi_{c,a}(-x,\epsilon k) \langle x,k \rangle e^{\pi c_{\rm b}^2 (a+c)^2} \zeta_{N,in}$$

(3.31)  $\overline{\tilde{\psi}_{a,c}(x,k)} = \psi_{b,c}(-x,\epsilon k)e^{-2\pi ic_{\mathrm{b}}^2ab}\zeta_0$ 

where  $\zeta_0 = e^{-\pi i (N - 4c_b^2 N^{-1})/12}$  and  $\zeta_{N,inv} = \zeta_0^2 e^{-\pi i c_b^2/N}$  and  $\epsilon = +1$  if b > 0 or  $\epsilon = -1$  if |b| = 1.

**Remark 3.18.** The hypothesis on positivity of a, b and c assure that the Fourier integral of  $\tilde{\psi}_{a,c}$  is absolutely convergent.

**Theorem 3.19** (Charged Pentagon Equation). Let  $a_j$ ,  $c_j > 0$  such that  $\frac{1}{\sqrt{N}} - a_j - c_j > 0$  for j = 0, 1, 2, 3 or 4. Define  $\psi_j \equiv \psi_{a_j,b_j}$ . Suppose the following relations hold true

(3.32)

 $a_1 = a_0 + a_2$   $a_3 = a_2 + a_4$   $c_1 = c_0 + a_4$   $c_3 = a_0 + c_4$   $c_2 = c_1 + c_3$ 

and consider the operators on  $L^2(\mathbb{A}_N)$  defined to satisfy (3.23-3.24). We have the following charged pentagon relation

(3.33) 
$$\psi_1(\mathbf{q}, \mathbf{L}_N(X))\psi_3(\mathbf{p}, \mathbf{L}_N(Y))\xi(a, c) =$$
  
=  $\psi_4(\mathbf{p}, \mathbf{L}_N(X))\psi_2(\mathbf{p} + \mathbf{q}, \mathbf{L}_N(X) + \mathbf{L}_N(Y))\psi_0(\mathbf{q}, \mathbf{L}_N(Y))$ 

where  $\xi(a,c) = e^{2\pi i c_{\rm b}^2(a_0 a_2 + a_0 a_4 + a_2 a_4)} e^{\pi i c_{\rm b}^2 a_2^2}$ .

## 4 Quantum Teichmüller Theory

In this section we are going to quantize the space  $\mathcal{R}(\Sigma_{g,s})$  following [K3]. For any fixed  $\tau$  the quantization of  $\mathcal{R}(\tau)$  is just the canonical quantization in exponential coordinates of the space  $\mathbb{R}_{>0}^M \times \mathbb{R}_{>0}^M$ , where M = 2g - 2 + s, with symplectic form  $\omega_{\tau} = \sum_{j=1}^{M} d \log u_j \wedge d \log v_j$ . Formally, following the expectations from canonical quantization of  $\mathbb{R}^{2M}$ , we can quantize  $\mathcal{R}(\tau)$  and associate to it an algebra of operator

$$\begin{aligned} \mathcal{X}(\tau) \text{ generated by } \left\{ \hat{u}_j, \, \hat{v}_j \right\}, \text{ where } 0 \leq j < M, \text{ subject to the relations} \\ \hat{u}_j \hat{v}_l = q^{\delta(j-l)} \hat{v}_l \hat{u}_j \quad \hat{u}_j \hat{u}_l = \hat{u}_l \hat{u}_j \quad \hat{v}_j \hat{v}_l = \hat{v}_l \hat{v}_j \end{aligned}$$

where  $q \in \mathbb{C}^*$ . The algebra we mean here is the associative algebra of non commutative fractions of non commutative polynomials generated by these generators. In order to obtain a quantization of  $\mathcal{R}(\Sigma_{g,s})$  (i.e. triangulation independent) we have to look at the action of the  $\mathcal{G}(\Sigma_{g,s})$  generators on coordinates and translate it into an action on the algebras  $\mathcal{X}(\tau)$ . Precisely consider the set of the couples  $(\tau, \mathcal{X}(\tau))$  and let the generators  $[\tau, \tau^{\sigma}]$ ,  $[\tau, \rho_i \tau]$  and  $[\tau, \omega_{i,j} \tau]$  act on them. The action on the algebras is as follows. The elements  $[\tau, \tau^{\sigma}]$  just permutes the indexes of the generators according to the permutation  $\sigma$ . The change of decoration  $[\tau, \rho_i \tau]$ acts trivially on the operators  $(\hat{u}_i, \hat{v}_j)$  such that  $j \neq i$  and as follows on  $(\hat{u}_i, \hat{v}_i)$ 

(4.1) 
$$(\hat{u}_i, \hat{v}_i) \mapsto (q^{-1/2} \hat{v}_i \hat{u}_i^{-1}, \hat{u}_i^{-1}).$$

The most interesting generator  $[\tau, \omega_{i,j}\tau]$ , is again trivial in the triangles not involved in the diagonal exchange, but it maps the two couples of operators  $(\hat{u}_i, \hat{v}_i)$  and  $(\hat{u}_i, \hat{v}_j)$  to the two new couples (following formulas (2.11))

$$(4.2) \qquad (\hat{u}_i, \hat{v}_i) \bullet (\hat{u}_j, \hat{v}_j) \equiv (\hat{u}_i \hat{u}_j, \hat{u}_i \hat{v}_j + \hat{v}_i)$$

$$(4.3) \qquad (\hat{u}_i, \hat{v}_i) * (\hat{u}_j, \hat{v}_j) \equiv (\hat{u}_j \hat{v}_i (\hat{u}_i \hat{v}_j + \hat{v}_i)^{-1}, \hat{v}_j (\hat{u}_i \hat{v}_j + \hat{v}_i)^{-1}).$$

In order to get an actual quantization we need to provide a representation of  $\mathcal{X}(\tau)$ by operators acting on some vector space  $\mathcal{H}$ . In the original paper [K3], Kashaev proposed representations on the vector spaces  $L^2(\mathbb{R})$  and  $L^2(\mathbb{Z}/N\mathbb{Z}) \simeq \mathbb{C}^N$  for Nodd. The former was used to construct the Andersen-Kashaev invariants in [AK1], while the latter are related to the colored Jones polynomials ([K1], [MM]) and the Volume Conjecture [K2]. In the more recent work [AK3] a representation on the vector space  $L^2(\mathbb{A}_N) \equiv L^2(\mathbb{R} \times \mathbb{Z}/N\mathbb{Z}) \simeq L^2(\mathbb{R}) \otimes \mathbb{C}^N$  was implicitly proposed, or at least all the basics elements to construct it were presented. Here we describe the representations in  $L^2(\mathbb{A}_N)$ .

# 4.1 $L^2(\mathbb{A}_N)$ Representations

Fix N positive odd integer,  $\omega \equiv e^{\frac{2\pi i}{N}}$  and  $\mathbf{b} \in \mathbb{C}^*$ ,  $\operatorname{Re}(\mathbf{b}) > 0$ . To each decorated ideal triangle  $\overline{\tau}_j \in \tau$  we associate the Hilbert space  $\mathrm{L}^2(\mathbb{A}_N)$ . Then the Hilbert space associated to  $\mathcal{R}(\tau)$  will be  $\mathcal{H} = \mathrm{L}^2(\mathbb{A}_N)^{\otimes M} \cong \mathrm{L}^2(\mathbb{A}_N^M)$  where M = 2g - 2 + s is the number of triangles in  $\tau$ . For conventions and notation on the space  $\mathrm{L}^2(\mathbb{A}_N)$  see Appendix A.2. For every  $i = 0, \ldots, M$  let  $\mathbf{p}_i$ ,  $\mathbf{q}_i$  be self adjoint operator in  $\mathrm{L}^2(\mathbb{R})$  and  $X_i$ ,  $Y_i$  unitary operators in  $\mathrm{L}^2(\mathbb{Z}/N\mathbb{Z}) \simeq \mathbb{C}^N$  such that

(4.4) 
$$[p_i, q_j] = \frac{\delta_{ij}}{2\pi i}, \quad Y_i X_j = \omega^{\delta_{ij}} X_j Y_i, \quad X_i^N = Y_i^N = 1.$$

We can define the operators

(4.5) 
$$\mathbf{u}_{i} = e^{2\pi \frac{\mathbf{b}}{\sqrt{N}} \mathbf{q}_{i}} Y_{i}$$
  $\mathbf{u}_{i}^{*} = e^{2\pi \frac{\mathbf{b}^{-1}}{\sqrt{N}} \mathbf{q}_{i}} Y_{i}^{-1}$ 

(4.6) 
$$\mathbf{v}_i = e^{2\pi \frac{\mathbf{b}}{\sqrt{N}}\mathbf{p}_i} X_i$$
  $\mathbf{v}_i^* = e^{2\pi \frac{\mathbf{b}^{-1}}{\sqrt{N}}\mathbf{p}_i} X_i^{-1}$ 

satisfying

(4.7) 
$$\mathbf{u}_i \mathbf{v}_j = q^{\delta_{ij}} \mathbf{v}_j \mathbf{u}_i \qquad \mathbf{u}_i^* \mathbf{v}_j^* = \tilde{q}^{\delta_{ij}} \mathbf{v}_j^* \mathbf{u}_i^*$$

(4.8) 
$$\mathbf{u}_i \mathbf{v}_j^* = \mathbf{v}_j^* \mathbf{u}_i \qquad \mathbf{u}_i^* \mathbf{v}_j = \mathbf{v}_j \mathbf{u}_i^*$$

(4.9) 
$$q = e^{2\pi i \frac{D}{N}} \omega \qquad \qquad \tilde{q} = e^{2\pi i \frac{D}{N}} \omega.$$

The Quantum algebra  $\mathcal{X}(\tau)$  is generated by the  $u_j$ ,  $v_j$  for  $j = 0, \ldots M$ , and has a \*-algebra structure when extended to include  $u_i^*$  and  $v_i^*$ . We remark that the \* operator we are using here is the standard hermitian conjugation only if  $|\mathbf{b}| = 1$ . Explicitly let  $X_j, Y_j, \mathsf{p}_j, \mathsf{q}_j, j = 1, 2$  be operators acting on  $\mathcal{H} := L^2(\mathbb{A}^2_N)$  as follow

(4.10) 
$$\mathbf{p}_j f(\mathbf{x}, \mathbf{m}) = \frac{1}{2\pi i} \frac{\partial}{\partial x_j} f(\mathbf{x}, \mathbf{m}), \qquad \mathbf{q}_j f(\mathbf{x}, \mathbf{m}) = x_j f(\mathbf{x}, \mathbf{m})$$

 $X_1 f(\mathbf{x}, \mathbf{m}) = f(\mathbf{x}, (m_1 + 1, m_2)), \quad X_2 f(\mathbf{x}, \mathbf{m}) = f(\mathbf{x}, (m_1, m_2 + 1))$ (4.11)

(4.12) 
$$Y_j f(\mathbf{x}, \mathbf{m}) = \overline{\omega}^{m_j} f(\mathbf{x}, \mathbf{m})$$

where  $\mathbf{m} = (m_1, m_2) \in \mathbb{Z}_N^2$  and  $\mathbf{x} = (x_1, x_2) \in \mathbb{R}^2$ . These operators satisfy conditions (4.4). Let  $\psi_{\mathbf{b}}(x, n) \equiv \frac{1}{\mathbf{D}_{\mathbf{b}}(x, n)}$  and consider the operators

(4.13) 
$$\mathsf{D}_{12} \equiv e^{2\pi i \mathsf{q}_2 \mathsf{p}_1} \sum_{j,k=0}^{N-1} \omega^{jk} Y_2^j X_1^k$$

(4.14) 
$$\Psi_{12} \equiv \Psi_{\mathrm{b}}(\mathsf{q}_1 + \mathsf{p}_2 - \mathsf{q}_2, -e^{-\frac{\pi i}{N}}Y_1X_2\overline{Y_2})$$

(4.15) 
$$T_{12} \equiv D_{12} \Psi_{12}$$

One has

Lemma 4.1 (Tetrahedral Equations).

$$\begin{array}{ccc} (4.16) & \mathsf{T}_{12}\mathsf{u}_1 = \mathsf{u}_1\mathsf{u}_2\mathsf{T}_{12} \\ (4.17) & \mathsf{T} & \mathsf{T} & \mathsf{T} \end{array}$$

(4.17) 
$$T_{12}v_1v_2 = v_2T_{12}$$

(4.18) 
$$T_{12}v_1u_2 = v_1u_2T_{12}$$

(4.19) 
$$T_{12}v_1 = (u_1v_2 + v_1)T_{12}$$

$$(4.20) T_{12}T_{13}T_{23} = T_{23}T_{12}$$

Remark 4.2. If we define  $\tilde{\mathsf{T}}_{12}=\mathsf{D}_{12}\tilde{\Psi}_{12}$  where

(4.21) 
$$\Psi \equiv \Psi_{\mathbf{b}^{-1}}(\mathbf{q}_1 + \mathbf{p}_2 - \mathbf{q}_2, -e^{-\frac{\pi i}{N}}\overline{Y}_1\overline{X}_2Y_2)$$

then  $\tilde{\mathsf{T}}$  satisfies equations (4.16 – 4.20) with  $\mathsf{u}_i$  and  $\mathsf{v}_i$  substituted by  $\mathsf{u}_i^*$  and  $\mathsf{v}_i^*$ . However from Remark 3.6 we know that  $\Psi_{\mathbf{b}^{-1}}(x,n) = \Psi_{\mathbf{b}}(x,-n)$ , and

$$\left(-e^{-\frac{\pi i}{N}}Y_1X_2\overline{Y_2}\right)^{-1} = -e^{\frac{\pi i}{N}}\overline{Y}_1Y_2\overline{X}_2 = -e^{-\frac{\pi i}{N}}\overline{Y}_1\overline{X}_2Y_2$$

so that

$$\tilde{T} = T$$
.

From Lemma 4.1 we have the following implementations of equations (4.1 - 4.3).

**Proposition 4.3.** Let  $w_i \equiv (u_i, v_i)$  and  $w_i^* = (u_i^*, v_i^*)$ . Then we have

(4.22) 
$$\mathbf{w}_1 \bullet \mathbf{w}_2 \mathbf{T}_{12} = \mathbf{T}_{12} \mathbf{w}_1, \quad \mathbf{w}_1 * \mathbf{w}_2 \mathbf{T}_{12} = \mathbf{T}_{12} \mathbf{w}_2$$

(4.23)  $\mathbf{w}_1^* \bullet \mathbf{w}_2^* \mathsf{T}_{12} = \mathsf{T}_{12} \mathbf{w}_1^*, \quad \mathbf{w}_1^* * \mathbf{w}_2^* \mathsf{T}_{12} = \mathsf{T}_{12} \mathbf{w}_2^*.$ 

Proposition 4.4. Let

(4.24) 
$$\mathsf{A} \equiv e^{3\pi i \mathsf{q}^2} e^{\pi i (\mathsf{p}+\mathsf{q})^2} \sum_{j=0}^{N-1} \overline{\langle j \rangle}^3 Y^{3j} \sum_{l=0}^{N-1} \overline{\langle l \rangle} (-e^{-\pi i/N} YX)^l,$$

where  $\langle n \rangle = e^{-\pi i n (n+N)/N}$  and Y and X are as above. Then

(4.25) 
$$A(\mathbf{u}, \mathbf{v}) = (q^{-1/2}\mathbf{v}\mathbf{u}^{-1}, \mathbf{u}^{-1}) A(\mathbf{u}^*, \mathbf{v}^*) = (\tilde{q}^{-1/2}\mathbf{v}^*(\mathbf{u}^*)^{-1}, (\mathbf{u}^*)^{-1})$$

where q and  $\tilde{q}$  are defined by equation (4.9).

# 5 Quantization of the Model Space for Complex Chern-Simons Theory

In this Section we want to quantize the space  $\mathbb{C}^* \times \mathbb{C}^*$  with the complex symplectic form

$$\omega_{\mathbb{C}} = \frac{\mathrm{d}x \wedge \mathrm{d}y}{xy}.$$

We think of it as a model space for Complex Chern-Simons Theory because it is an open dense of the PSL(2,  $\mathbb{C}$ ) moduli space of flat connections on a four punctured sphere, with unipotent holonomy around the punctures, [AK3, K5, D, FG]. Tetrahedral operators are supposedly related to states in the quantization of the four punctured sphere. Since we want to construct knot invariants starting from tetrahedral ideal triangulations this is the space we need to quantize. We follow the ideas in Andersen and Kashaev [AK3] using a real polarization with contractible leaves. We will further show that the level N Weil-Gel'fand-Zak Transform relates this quantization with the  $L^2(\mathbb{A}_N)$  representations in Quantum Teichmüller theory. To use this transform to relates the Andersen–Kashaev invariants to complex Chern–Simons Theory was already proposed in [AK3]. However the relation between the two approaches was not as tight as the one present here.

110

Let  $t = N + is \in \mathbb{C}^*$  be the quantization constant, for  $N \in \mathbb{R}$  and  $s \in \mathbb{R} \sqcup i\mathbb{R}$ . Denote also  $\tilde{t} = N - is$ . Fix  $b \in \mathbb{C}$  such that  $s = -iN\frac{1-b^2}{1+b^2}$  and  $\operatorname{Re} b > 0$ . This substitution, for  $s \in i\mathbb{R}$ , is only possible when -N < is < N. Notice that

 $(5.1) s \in \mathbb{R} \iff |\mathbf{b}| = 1 \text{ and } \mathbf{b} \neq \pm i, s \in i\mathbb{R} \iff \mathrm{Im}\,\mathbf{b} = 0$ 

and notice the following useful expressions

(5.2) 
$$t = \frac{2N}{1+b^2}, \qquad \tilde{t} = \frac{2N}{1+b^{-2}}.$$

Consider the covering maps

(5.3) 
$$\zeta^{\pm} \colon \mathbb{R}^2 \longrightarrow \mathbb{C}^*$$
$$(z, n) \mapsto \exp\left(2\pi \mathbf{b}^{\pm 1} z \pm 2\pi i n\right)$$

and consequently

(5.4) 
$$\pi^{\pm} : \mathbb{R}^2 \times \mathbb{R}^2 \longrightarrow \mathbb{C}^* \times \mathbb{C}^*, \qquad \pi^{\pm} = (\zeta^{\pm}, \, \zeta^{\pm})$$

such that

(5.5) 
$$\mathbb{C}^* \times \mathbb{C}^* \ni (x, y) = \pi^+((z, n), (w, m)), \\ \mathbb{C}^* \times \mathbb{C}^* \ni (\tilde{x}, \tilde{y}) = \pi^-((z, n), (w, m)) \quad \text{for } ((z, n), (w, m)) \in \mathbb{R}^2 \times \mathbb{R}^2.$$

We remark that

(5.6) 
$$\overline{\zeta^+(z,n)} = \zeta^-(z,n) \qquad \Longleftrightarrow \qquad |\mathbf{b}| = 1$$

in this case  $\pi^- = \overline{\pi^+}$  and  $\tilde{x} = \overline{x}$ ,  $\tilde{y} = \overline{y}$ . In this sense  $x, y \tilde{x}$  and  $\tilde{y}$  are natural coordinate functions to quantize in  $\mathbb{C}^* \times \mathbb{C}^*$ . If  $\mathbf{b} \in \mathbb{R}$  they are still coordinates functions for the underlying real manifold, but we lose the complex conjugate interpretation. We will first consider the quantization of the covering  $\mathbb{R}^2 \times \mathbb{R}^2$ . Define the form

(5.7) 
$$\omega_t \equiv \frac{t}{4\pi} (\pi^+)^* (\omega_{\mathbb{C}}) + \frac{\tilde{t}}{4\pi} (\pi^-)^* (\omega_{\mathbb{C}}).$$

Lemma 5.1.

(5.8) 
$$\omega_t = 2\pi N (\mathrm{d}z \wedge \mathrm{d}w - \mathrm{d}n \wedge \mathrm{d}m).$$

In particular it is a real symplectic 2 form on  $\mathbb{R}^2 \times \mathbb{R}^2$ , independent of b.

Over  $\mathbb{R}^2 \times \mathbb{R}^2$  we take the trivial line bundle  $\widetilde{\mathcal{L}} = \mathbb{R}^2 \times \mathbb{R}^2 \times \mathbb{C}$ . On the *N*-th tensor power of this line bundle  $\widetilde{\mathcal{L}}^N$  we consider the connection

(5.9) 
$$\nabla^{(t)} \equiv \mathbf{d} - i\alpha_t$$

where

(5.10) 
$$\begin{aligned} \alpha_t &\equiv \frac{t}{4\pi} \alpha_{\mathbb{C}}^+ + \frac{\tilde{t}}{4\pi} \alpha_{\mathbb{C}}^-, \\ (5.11) \quad \alpha_{\mathbb{C}}^\pm &\equiv 2\pi^2 (\mathbf{b}^{\pm 1} z \pm in) \mathbf{d} (\mathbf{b}^{\pm 1} w \pm im) - 2\pi^2 (\mathbf{b}^{\pm 1} w \pm im) \mathbf{d} (\mathbf{b}^{\pm 1} z \pm in) \end{aligned}$$

In analogy to Lemma 5.1 we have

(5.12) 
$$\alpha_t = \pi N (z \mathrm{d}w - w \mathrm{d}z - n \mathrm{d}m + m \mathrm{d}n).$$

It is simple to see that

- (5.13)  $d\alpha_{\mathbb{C}}^{\pm} = (\pi^{\pm})^*(\omega_{\mathbb{C}}), \qquad \text{which implies}$
- (5.14)  $F_{\nabla^{(t)}} = -i\omega_t.$

Further, on  $\mathbb{R}^2 \times \mathbb{R}^2$  we have an action of  $\mathbb{Z} \times \mathbb{Z}$  compatible with the projection  $\pi^+$ , i.e.

(5.15) 
$$(\mathbb{Z} \times \mathbb{Z}) \times (\mathbb{R}^2 \times \mathbb{R}^2) \longrightarrow \mathbb{R}^2 \times \mathbb{R}^2$$
$$(\lambda_1, \lambda_2) \cdot ((z, n), (w, m)) \mapsto ((z, n + \lambda_1), (w, m + \lambda_2))$$

(5.16)

that satisfies

(5.17) 
$$\pi^{\pm}((z, n + \lambda_1), (w, m + \lambda_2)) = \pi^{\pm}((z, n), (w, m))$$

This action can be lifted to an action  $\widetilde{\mathcal{L}}^N$  in such a way that the quotient bundle  $\mathcal{L}^N \equiv \widetilde{\mathcal{L}}^N / (\mathbb{Z})^2 \to \mathbb{R}^4 / \mathbb{Z}^2$  has first Chern class  $c_1(\mathcal{L}^N) = \frac{1}{2\pi} [\omega_t]$  ( $\omega_t$  is evidently  $\mathbb{Z}^2$ -invariant). Such a condition gives the requirement (which is in fact the prequantum condition)  $\frac{1}{2\pi} [\omega_t] \in H^2((\mathbb{R}^2 \times \mathbb{R}^2)/(\mathbb{Z}^2), \mathbb{Z})$ , which boils down to the requirement  $N \in \mathbb{Z}$ . Explicitly the action of  $\mathbb{Z} \times \mathbb{Z}$  on  $\widetilde{\mathcal{L}}^N$  is given by the following two multipliers

(5.18) 
$$e_{(1,0)} = e^{-\pi N i m}, \qquad e_{(0,1)} = e^{\pi N i n}.$$

That means that we consider the space of sections

$$(5.19) \qquad (\mathbf{C}^{\infty}(\mathbb{R}^4, \widetilde{\mathcal{L}}^N))^{\mathbb{Z}^2}$$

of  $\mathbb{Z}^2$ -invariant, smooth sections of  $\widetilde{\mathcal{L}}^N$  . Explicitly

$$s \in (C^{\infty}(\mathbb{R}^4, \mathcal{\hat{L}}^N))^{\mathbb{Z}^2}$$
 if and only if  $s \in C^{\infty}(\mathbb{R}^4, \mathcal{\hat{L}}^N)$  and satisfies

(5.20) 
$$s((z, n+1), (w, m)) = e^{-\pi i N m} s((z, n), (w, m)),$$

(5.21)  $s((z,n),(w,m+1)) = e^{\pi i N n} s((z,n),(w,m))$ 

#### Lemma 5.2.

 $\nabla^{(t)}s \ \in \ (\mathbf{C}^{\infty}(\mathbb{R}^{4},\widetilde{\mathcal{L}}^{N}))^{\mathbb{Z}^{2}}, \qquad \textit{for any } s \ \in \ (\mathbf{C}^{\infty}(\mathbb{R}^{4},\widetilde{\mathcal{L}}^{N}))^{\mathbb{Z}^{2}}$ 

The following Hermitian structure on  $\widetilde{\mathcal{L}}^N$  is  $\mathbb{Z}^2$ -invariant and parallel with respect to  $\nabla^{(t)}$ .

(5.22) 
$$s \cdot s'(p) \equiv s(p)\overline{s'(p)}, \text{ for any } p \in \mathbb{R}^2 \times \mathbb{R}^2$$

Being parallel here means that

(5.23) 
$$\mathbf{d}(s \cdot s') = (\nabla^{(t)}s) \cdot s' + s \cdot (\nabla^{(t)}s'),$$

and this is a simple consequence of  $\alpha_t$  being a real 1-form. It follows that the following is a well defined inner product in the completion of  $\left(\left(\mathbf{L}^2 \cap \mathbf{C}^{\infty}\right) \left(\mathbb{R}^4, \widetilde{\mathcal{L}}^N\right)\right)^{\mathbb{Z}^2}$ 

(5.24) 
$$(s,s') \equiv \int_{\mathbb{R}} \mathrm{d}z \int_{\mathbb{R}} \mathrm{d}w \left( \int_{0}^{1} \mathrm{d}n \int_{0}^{1} \mathrm{d}m \ s \cdot s' \right)$$

**Lemma 5.3.** We have the following Hamiltonian vector field for the coordinates functions on  $\mathbb{R}^2 \times \mathbb{R}^2$ 

$$X_{z} = \frac{1}{2\pi N} \frac{\partial}{\partial w} \qquad \qquad X_{w} = -\frac{1}{2\pi N} \frac{\partial}{\partial z}$$
$$X_{n} = -\frac{1}{2\pi N} \frac{\partial}{\partial m} \qquad \qquad X_{m} = \frac{1}{2\pi N} \frac{\partial}{\partial n}$$

From the definition of the pre-quantum operator  $\hat{f}$  associated to the observable f, we have

$$(5.25)\qquad\qquad\qquad \hat{f} = -i\nabla_{X_f} + f$$

**Lemma 5.4** (Pre–Quantum operators). The following are the pre–quantum operators for the coordinate functions on  $\mathbb{R}^2 \times \mathbb{R}^2$ 

$$\begin{aligned} \hat{z} &= \frac{-i}{2\pi N} \nabla^{(t)}_w + z & \qquad \hat{w} &= \frac{i}{2\pi N} \nabla^{(t)}_z + w \\ \hat{n} &= \frac{i}{2\pi N} \nabla^{(t)}_w + n & \qquad \hat{m} &= \frac{-i}{2\pi N} \nabla^{(t)}_n + m \end{aligned}$$

and they satisfy the following canonical commutation relations

(5.26) 
$$[\hat{z}, \hat{w}] = \frac{1}{2\pi i N}$$
  $[\hat{n}, \hat{m}] = -\frac{1}{2\pi i N}$ 

(5.27)  $[\hat{z}, \hat{n}] = [\hat{z}, \hat{m}] = [\hat{w}, \hat{n}] = [\hat{w}, \hat{m}] = 0$ 

The Hermitian line bundle  $\mathcal{L}^N \to (\mathbb{R}^2 \times \mathbb{R}^2/\mathbb{Z}^2)$  together with the connection  $\nabla^{(t)}$  define a pre–Quantization of the theory. In order to finish the quantization program we need to choose a Lagrangian polarization.

Choose the following real Lagrangian polarization

(5.28) 
$$\widetilde{\mathcal{P}} \equiv \operatorname{Span}_{\mathbb{R}} \left\{ \frac{\partial}{\partial w} + \frac{\partial}{\partial n}, \frac{\partial}{\partial z} - \frac{\partial}{\partial m} \right\}.$$

The leaves of this polarization are all contractible after quotient by the action of  $\mathbb{Z}^2$  on  $\mathbb{R}^2 \times \mathbb{R}^2$ , so we do have polarized global sections. In particular the space  $T \subset \mathbb{R}^2 \times \mathbb{R}^2$ 

$$(5.29) T \equiv \{z = w = 0\}$$

is a transversal for the polarization. For any  $\psi \in (C^{\infty}(\mathbb{R}^4, \widetilde{\mathcal{L}}^N))^{\mathbb{Z}^2}$  polarized by  $\widetilde{\mathcal{P}}$ , the following two differential equations will determine  $\psi \equiv \psi((z, n), (w, m))$  by its value in (n, m)

(5.30) 
$$\nabla_w^{(t)}\psi = -\nabla_n^{(t)}\psi \qquad \nabla_z^{(t)}\psi = \nabla_m^{(t)}\psi.$$

The space  $T/\mathbb{Z}^2$  is of course  $\mathbb{T} \times \mathbb{T}$ , and the line bundle  $\mathcal{L}^N$  will restrict to a non trivial line bundle over  $\mathbb{T} \times \mathbb{T}$  that we shall call  $\mathcal{L}^N$  again. The quantum space that we obtain is then

(5.31) 
$$\mathcal{H}^{(N)} \equiv \mathbf{C}^{\infty} \left( \mathbb{T} \times \mathbb{T}, \mathcal{L}^{N} \right).$$

We consider the obvious inner product on  $\mathcal{H}^{(N)}$ 

(5.32) 
$$(\psi, \phi) = \int_0^1 \int_0^1 \psi \overline{\phi} \, \mathrm{d}n \mathrm{d}m$$

that is the standard inner product on the completion  $L^2(\mathbb{T} \times \mathbb{T}, \mathcal{L}^N)$ . Finally the quantum operators acts on polarized sections as

(5.33) 
$$\hat{x} \equiv \exp\left(2\pi \mathbf{b}\hat{z} + 2\pi i\hat{n}\right) = \exp\left(i\frac{\mathbf{b}}{N}\nabla_n^{(t)} - \frac{1}{N}\nabla_m^{(t)} + 2\pi in\right)$$

(5.34) 
$$\hat{y} \equiv \exp\left(2\pi b\hat{w} + 2\pi i\hat{m}\right) = \exp\left(i\frac{b}{N}\nabla_m^{(t)} + \frac{1}{N}\nabla_n^{(t)} + 2\pi im\right)$$

(5.35) 
$$\hat{x} \equiv \exp\left(2\pi b^{-1}\hat{z} - 2\pi i\hat{n}\right) = \exp\left(i\frac{b^{-1}}{N}\nabla_n^{(t)} + \frac{1}{N}\nabla_m^{(t)} - 2\pi in\right)$$

(5.36) 
$$\hat{y} \equiv \exp\left(2\pi b^{-1}\hat{w} - 2\pi i\hat{m}\right) = \exp\left(i\frac{b^{-1}}{N}\nabla_z^{(t)} - \frac{1}{N}\nabla_n^{(t)} - 2\pi im\right)$$

Now we are going to connect the quantization with the Quantum Teichmüller theory. Recall the operators u = u(b) and v = v(b) from equations (4.4 – 4.12), and recall that they depend on a parameter b. Define the rescaling operator

$$\mathcal{O}_{\sqrt{N}} \colon \mathrm{L}^2(\mathbb{A}_N) \longrightarrow \mathrm{L}^2(\mathbb{A}_N)$$

(5.37) 
$$\mathcal{O}_{\sqrt{N}}(\mathsf{f})(z,n) = \mathsf{f}\left(\sqrt{N}z,n\right)$$

and the following rescaled analogues of the Quantum Teichmüller Theory operators

(5.38) 
$$\hat{u} = \mathcal{O}_{\sqrt{N}} \circ \mathsf{u}(\mathbf{b}^{-1}) \circ \mathcal{O}_{\sqrt{N}}^{-1} \qquad \hat{v} = \mathcal{O}_{\sqrt{N}} \circ \mathsf{v}(\mathbf{b}^{-1}) \circ \mathcal{O}_{\sqrt{N}}^{-1}$$

(5.39) 
$$\hat{u}^* = \mathcal{O}_{\sqrt{N}} \circ \mathbf{u}^*(\mathbf{b}^{-1}) \circ \mathcal{O}_{\sqrt{N}}^{-1} \qquad \hat{v}^* = \mathcal{O}_{\sqrt{N}} \circ \mathbf{v}^*(\mathbf{b}^{-1}) \circ \mathcal{O}_{\sqrt{N}}^{-1}$$

which acts on  $f \in L^2(\mathbb{A}_N)$  as

(5.40) 
$$\hat{u}^* \mathsf{f}(z,l) = e^{2\pi i l/N} \mathsf{f}(z,l) \qquad \hat{u} \mathsf{f}(z,l) = e^{2\pi b^{-1} z} e^{-2\pi i l/N} \mathsf{f}(z,l)$$

(5.41) 
$$\hat{v}^* f(z,l) = f(z-i\frac{b}{N},l-1)$$
  $\hat{v} f(z,l) = f(z-i\frac{b^{-1}}{N},l+1)$ 

We make use of the level-N Weil-Gel'fand-Zak Transform, [AK3].

**Theorem 5.5.** Recall the line bundle  $\mathcal{L}^N$ . The following map  $Z^{(N)} \colon \mathcal{S}(\mathbb{A}_N) \longrightarrow C^{\infty}(\mathbb{T} \times \mathbb{T}, \mathcal{L}^N)$  is a an isomorphism

(5.42) 
$$Z^{(N)}(\mathbf{f})(n,m) = \frac{1}{\sqrt{N}} e^{\pi i N m n} \sum_{p \in \mathbb{Z}} \sum_{l=0}^{N-1} \mathbf{f}\left(n + \frac{p}{N}, l\right) e^{2\pi i m p} e^{2\pi i l p/N}$$

 $with \ inverse$ 

$$\overline{Z}^{(N)}(s)(x,j) = \frac{1}{\sqrt{N}} \sum_{l=0}^{N-1} e^{-2\pi i \frac{lj}{N}} \int_0^1 s\left(x - \frac{l}{N}, v\right) e^{-\pi i N (x + \frac{l}{N})v} \mathrm{d}v.$$

which preserves the inner products  $L^2(\mathbb{A}_N)$  and  $(\cdot, \cdot)$ , i.e.

$$\left(Z^{(N)}(\mathsf{f}), Z^{(N)}(\mathsf{g})\right) = \langle \mathsf{f}, \mathsf{g} \rangle$$

and so extends to an isometry between  $L^2(\mathbb{A}_N)$  and  $L^2(\mathbb{T} \times \mathbb{T}, \mathcal{L}^N)$ .

Proposition 5.6. We have

$$Z^{(N)} \circ \hat{u}^* \circ (Z^{(N)})^{-1} = \hat{y}^{-1} \qquad \qquad Z^{(N)} \circ \hat{v}^* \circ (Z^{(N)})^{-1} = \hat{x}^{-1}$$
  
$$Z^{(N)} \circ \hat{u} \circ (Z^{(N)})^{-1} = \hat{y}^{-1} \qquad \qquad Z^{(N)} \circ \hat{v} \circ (Z^{(N)})^{-1} = \hat{x}^{-1}$$

All together we have showed that the quantization for the model space of complex Chern-Simons theory is equivalent to the  $L^2(\mathbb{A}_N)$  representations of the quantum algebra defined from Quantum Teichmüller Theory. In the following section we will extend the  $L^2(\mathbb{A}_N)$  representations to knots invariants following the recipe given by Andersen and Kashaev in [AK1]. The previous discussion on the different quantizations serves to link such invariants to Complex Quantum Chern–Simons Theory.

# 6 The Andersen–Kashaev Teichmüller TQFT at Level N

#### 6.1 Angle Structures on 3-Manifolds

We present here shaped triangulated pseudo 3-manifolds, which are the combinatorial data underlying the Andersen-Kashaev construction of their invariant. Following strictly [AK1] we will describe the *categroid* of *admissible* oriented triangulated pseudo 3-manifolds, where the words admissible and categroid go together because admissibility is what will obstruct us to have a full category. See Appendix B for a definition of categorids.

**Definition 6.1** (Oriented Triangulated Pseudo 3-manifold). An Oriented Triangulated Pseudo 3-manifold X is a finite collection of 3-simplices (tetrahedra) with totally ordered vertices together with a collection of gluing homeomorphisms between some pairs of codimension 1 faces, so that every face is in, at most, one such pairs. By gluing homeomorphism we mean a vertex order preserving, orientation reversing, affine homeomorphism between the two faces.

The quotient space under the glueing homeomorphisms has the structure of CW-complex with oriented edges.

For  $i \in \{0, 1, 2, 3\}$  we denote by  $\Delta_i(X)$  the collection of *i*-dimensional simplices in X and, for i > j, we denote

$$\Delta_i^j(X) = \{(a,b) | a \in \Delta_i(X), b \in \Delta_j(a)\}.$$

We have projection maps

$$\phi_{i,j}: \Delta_i^j(X) \longrightarrow \Delta_i(X), \quad \phi^{i,j}: \Delta_i^j(X) \longrightarrow \Delta_j(X),$$

and boundary maps

 $\partial_i : \Delta_j(X) \longrightarrow \Delta_{j-1}(X), \quad \partial_i[v_0, \dots, v_j] \mapsto [v_0, \dots, v_{i-1}, v_{i+1}, \dots, v_j]$ 

where  $[v_0, \ldots, v_j]$  is the *j*-simplex with vertices  $v_0, \ldots, v_j$  and  $i \leq j$ .

**Definition 6.2** (Shape Structure). Let X be an oriented triangulated pseudo 3-manifold. A *Shape Structure* is a map

$$\alpha_X : \Delta^1_3(X) \longrightarrow \mathbb{R}_{>0},$$

so that, in every tetrahedron, the sum of the values of  $\alpha_X$  along three incident edges is  $\pi$ .

The value of the map  $\alpha_X$  in an edge *e* inside a tetrahedron *T* is called the *dihedral* angle of *T* at *e*. If we allow  $\alpha_X$  to take values in  $\mathbb{R}$  we talk about a *Generalized* Shape Structure. The set of shape structures supported by X is denoted S(X). The space of generalized shape structures is denoted by  $\tilde{S}(X)$ . X together with  $\alpha_X$  is called *Shaped Pseudo 3-manifold*.

**Remark 6.3** (Ideal Tetrahedron). A shape structure on a simplicial tetrahedron T as above defines an embedding of  $T \setminus \Delta_0(T)$  in the hyperbolic 3–space  $\mathbb{H}^3$  which extends to a map of T to  $\overline{\mathbb{H}^3}$ . In fact we can change a given embedding, so that it send the four vertices  $(v_0, v_1, v_2, v_3)$  to the four points  $(\infty, 0, 1, z) \in \mathbb{CP}^1 \simeq \partial \mathbb{H}^3$ , where

$$z = \frac{\sin \alpha_T([v_0, v_2])}{\sin \alpha_T([v_0, v_3])} \exp(i\alpha_T([v_0, v_1])).$$

This four points in  $\partial \mathbb{H}^3$  extend to a unique ideal tetrahedron in  $\mathbb{H}^3$ , by taking the geodesic convex hull, that has dihedral angles defined by  $\alpha_T$ .

**Remark 6.4.** In every tetrahedron, its orientation induces a cyclic ordering of all triples of edges meeting in a vertex. Such a cyclic ordering descends to a cyclic ordering of the pairs of opposite edges of the whole tetrahedron. Moreover, it follows from the definition that opposite edges share the same dihedral angle. Hence, we get a well defined cyclic order preserving projection  $p: \Delta_3^{1/P}(X) \longrightarrow \Delta_3^{1/P}(X)$  which identifies opposite edges.  $\alpha_X$  descends to a map from  $\Delta_3^{1/P}(X)$ and we can consider the following skew-symmetric functions

$$\varepsilon_{a,b} \in \{0, \pm 1\}, \quad \varepsilon_{a,b} = -\varepsilon_{b,a}, \quad a, b \in \Delta_3^{1/p}(X),$$

defined to be  $\varepsilon_{a,b} = 0$  if the underlying tetrahedra are distinct, and  $\varepsilon_{a,b} = +1$  if the underlying tetrahedra coincides and b cyclically follows a in the order induced on  $\Delta_{3}^{1/p}(X)$ .

**Definition 6.5.** To any shaped pseudo 3-manifold X, we associate a *Weight* function

$$\omega_X : \Delta_1(X) \longrightarrow \mathbb{R}_{>0}, \quad \omega_X(e) = \sum_{a \in (\phi^{3,1})^{-1}(e)} \alpha_X(a).$$

An edge e in X is called *balanced* if e is internal and  $\omega_X(e) = 2\pi$ . A shape structure is fully balanced if all its edges are balanced.

The shape structures of closed fully balanced 3-manifolds are called *Angle Structures* in the literature. For more details on them and their geometric admissibility see [Lac] and [LT].

**Definition 6.6.** A *leveled* (generalised) shaped pseudo 3-manifold is a pair  $(X, l_X)$  consisting of a (generalized) shaped pseudo 3-manifold X and a real number  $l_X \in \mathbb{R}$ , called the *level*. The set of all leveled (generalised) shaped pseudo 3-manifolds is denoted by LS(X) (respectively  $\widetilde{LS}(X)$ ).

There is a gauge action of  $\mathbb{R}^{\Delta_1(X)}$  on  $\widetilde{\mathrm{LS}}(X)$ .

**Definition 6.7.** Let  $(X, l_X)$  and  $(Y, l_Y)$  be two (generalized) leveled shaped pseudo 3-manifolds. They are said to be *gauge equivalent* if there exists an isomorphism  $h : X \longrightarrow Y$  of the underlying cellular structures, and a function  $g : \Delta_1(X) \longrightarrow \mathbb{R}$  such that

$$\begin{split} \Delta_1(\partial X) &\subset g^{-1}(0),\\ \alpha_Y(h(a)) &= \alpha_X(a) + \pi \sum_{b \in \Delta_3^1(X)} \varepsilon_{p(a), p(b)} g(\phi^{3,1}(b)), \quad \forall a \in \Delta_3^1(X), \text{ and}\\ l_Y &= l_X + \sum_{e \in \Delta_1(X)} g(e) \sum_{a \in (\phi^{3,1})^{-1}(e)} (\frac{1}{3} - \frac{\alpha_X(a)}{\pi}). \end{split}$$

We remark that  $\omega_X = \omega_Y \circ h$ .

**Definition 6.8.** Let  $(\alpha_X, l_X)$  and  $(\alpha_{X'}, l_{X'})$  be two (generalized) leveled shape structures of the oriented pseudo 3-manifold X. They are said *based gauge equivalent* if they are gauge equivalent as in Definition 6.7 if the isomorphism  $h: X \longrightarrow X$  is the identity.

Based gauge equivalence is an equivalence relation in the sets S(X), LS(X),  $\widetilde{S}(X)$ ,  $\widetilde{LS}(X)$  and the quotient sets are denoted (resp.)  $S_r(X)$ ,  $LS_r(X)$ ,  $\widetilde{S}_r(X)$ ,  $\widetilde{LS}_r(X)$ . We remark that  $S_r(X)$  is an open convex (possibly empty) subset of the space  $\widetilde{S}_r(X)$ . We will return to existence of shape structures later. Let us focus on  $\widetilde{S}(X)$  for now. Let

$$\tilde{\Omega}_X : \tilde{\mathrm{S}}(X) \longrightarrow \mathbb{R}^{\Delta_1(X)}$$

be the map which sends the shape structure  $\alpha_X$  to the corresponding weight function  $\omega_X$ . This map is gauge invariant, so it descends to a map

$$\tilde{\Omega}_{X,r}: \widetilde{S}_r(X) \longrightarrow \mathbb{R}^{\Delta_1(X)}$$

For fixed  $a \in \Delta_3^{1/p}(X)$  we can think of  $\alpha_a := \alpha_X(a)$  as an element of  $C^{\infty}\left(\widetilde{\mathbf{S}}(X)\right)$ .

**Definition 6.9** ([NZ]). The Neumann-Zagier symplectic structure on S(X) is the unique symplectic structure which induces the Poisson bracket  $\{\cdot, \cdot\}$  satisfying

$$\{\alpha_a, \alpha_b\} = \varepsilon_{a,b}$$

for all  $a, b \in \Delta_3^{1/p}(X)$ .

For a triangulated pseudo 3-manifold we have a symplectic decomposition

$$\widetilde{\mathbf{S}}(X) = \prod_{T \in \Delta_3(X)} \widetilde{\mathbf{S}}(T).$$

**Theorem 6.10** ([AK1]). The gauge action of  $\mathbb{R}^{\Delta_1(X)}$  on  $\widetilde{S}(X)$  is symplectic and  $\widetilde{\Omega}_X$  is a moment map for this action. It follows that  $\widetilde{S}_r(X) = \widetilde{S}(X)/\mathbb{R}^{\Delta_1(X)}$  is a Poisson manifold with symplectic leaves corresponding to the fibers of  $\widetilde{\Omega}_{X,r}$ .

Let  $N_0(X)$  be a sufficiently small neighbourhood of  $\Delta_0(X)$ , then  $\partial N_0(X)$  is a surface which inherits a triangulation from X, with a shape structure, if X has a shape structure. Notice that this surface can have boundary if  $\partial X \neq \emptyset$ .

Theorem 6.11 ([AK1]). The map

$$\tilde{\Omega}_{X,r}: \tilde{S}_r(X) \longrightarrow \mathbb{R}^{\Delta_1(X)}$$

is an affine  $\mathrm{H}^1(\partial N_0(X), \mathbb{R})$ -bundle. The Poisson structure of  $\widetilde{\mathrm{S}}_r(X)$  coincide with the one induced by the  $\mathrm{H}^1(\partial N_0(X), \mathbb{R})$ -bundle structure.

If  $h: X \longrightarrow Y$  is an isomorphisms of cellular structure, the induced morphism  $h^*: \widetilde{S}_r(Y) \longrightarrow \widetilde{S}_r(X)$  is compatible with all this structures, i.e. it is a Poisson affine bundle morphism which fiberwise coincide with the naturally induced group morphism  $h^*: H^1(\partial N_0(Y), \mathbb{R}) \longrightarrow H^1(\partial N_0(X), \mathbb{R})$ . Moreover  $h^*$  maps  $S_r(Y)$  to  $S_r(X)$ .

**Definition 6.12** (Shaped 3-2 Pachner moves). Let X be a shaped pseudo 3 manifold and let e be a balanced internal edge in it, shared exactly by three distinct tetrahedra  $t_1$ ,  $t_2$  and  $t_3$  with dihedral angles at e exactly  $\alpha_1$ ,  $\alpha_2$  and  $\alpha_3$ . Then the triangulated pseudo 3-manifold  $X_e$  obtained by removing the edge e, and substituting the three tetrahedra  $t_1$ ,  $t_2$  and  $t_3$  with other two new tetrahedra  $t_4$  and  $t_5$  glued along one face, is topologically the same space as X. In order to have the same weights of X on  $X_e$ , the dihedral angles of  $t_4$  and  $t_5$  are uniquely determined by the ones of  $t_1$ ,  $t_2$  and  $t_3$  as follows



Figure 3: A 3–2 Pachner move.

(6.1) 
$$\begin{aligned} \alpha_4 &= \beta_2 + \gamma_1 \quad \alpha_5 = \beta_1 + \gamma_2 \\ \beta_4 &= \beta_1 + \gamma_3 \quad \beta_5 = \beta_3 + \gamma_1 \\ \gamma_4 &= \beta_3 + \gamma_2 \quad \gamma_5 = \beta_2 + \gamma_3 \end{aligned}$$

where  $(\alpha_i, \beta_i, \gamma_i)$  are the dihedral angles of  $t_i$ . In this situation we say that  $X_e$  is obtained from X by a shaped 3-2 Pachner move.

We remark that the linear system, together with e being balanced, guarantees the positivity of the dihedral angles of  $t_4$  and  $t_5$  provided the positivity for  $t_1$ ,  $t_2$ and  $t_3$  but it does not provide any guarantees on the converse, i.e. the positivity of a shaped 2-3 Pachner moves. However, two different solutions for the angles for  $t_1$ ,  $t_2$  and  $t_3$  from the same starting angles for  $t_4$  and  $t_5$  are always gauge equivalent.

The system (6.1) define a map  $P^e: S(X) \longrightarrow S(X_e)$ , that extends to a map

$$\tilde{P}^e: \tilde{S}(X) \longrightarrow \tilde{S}(X_e).$$

For a balanced edge e, the latter restricts to the map

$$\tilde{P}_r: \tilde{\Omega}_{X,r}(e)^{-1}(2\pi) \longrightarrow \widetilde{S}_r(X_e),$$

and it can be noticed that  $\tilde{P}_r(\tilde{\Omega}_{X,r}(e)^{-1}(2\pi) \cap S_r(x)) \subset S_r(Y)$ .

We also say that a leveled shaped pseudo 3-manifold  $(X, l_X)$  is obtained from  $(Y, l_Y)$  by a *leveled shaped* 3-2 *Pachner move* if, for some balanced  $e \in \Delta_1(X)$ ,  $Y = X_e$  as above and

$$l_Y = l_X + \frac{1}{12\pi} \sum_{a \in (\phi^{3,1})^{-1}(e)} \sum_{b \in \Delta_3^1(X)} \varepsilon_{p(a), p(b)} \alpha_X(b).$$

**Definition 6.13.** A (leveled) shaped pseudo 3-manifold X is called a *Pachner re-finement* of a (leveled) shaped pseudo 3-manifold Y if there exists a finite sequence of (leveled) shaped pseudo 3-manifolds

$$X = X_1, X_2, \dots, X_n = Y$$

such that for any  $i \in \{1, ..., n-1\}$ ,  $X_{i+1}$  is obtained from  $X_i$  by a (leveled) shaped 3-2 Pachner move. Two (leveled) shaped pseudo 3-manifolds X and Y are called *equivalent* if there exist gauge equivalent (leveled) shaped pseudo 3-manifolds X' and Y' which are respective Pachner refinements of X and Y.

For technical reasons, which we will discuss later, we will restrict the category of triangulated 2 + 1 cobordisms, discussed so far, to a certain sub-categorid, as discussed below. This means that we will remove some morphisms as the following definition imposes.

**Definition 6.14** (Admissibility). An oriented triangulated pseudo 3-manifold is called *admissible* if

$$S_r(X) \neq \emptyset$$

and

$$H_2(X - \Delta_0(X), \mathbb{Z}) = 0$$

**Definition 6.15.** Two (leveled) admissible shaped pseudo 3-manifolds X and Y are said to be *admissibly equivalent* if there exists a gauge equivalence

$$h': X' \longrightarrow Y'$$

of (leveled) shaped 3-manifolds X' and Y' which are respective Pachner refinements of X and Y such that  $\Delta_1(X') = \Delta_1(X) \cup D_X$  and  $\Delta_1(Y') = \Delta_1(Y) \cup D_Y$ and the following holds

$$\left[h(\mathbf{S}_r(X) \cap \tilde{\Omega}_{X',r}(D_X)^{-1}(2\pi))\right] \cap \left[\tilde{\Omega}_{Y',r}(D_Y)^{-1}(2\pi)\right] \neq \emptyset$$

**Theorem 6.16** ([AK1]). Suppose two (leveled) shaped pseudo 3-manifolds X and Y are equivalent. Then there exist  $D \subset \Delta_1(X)$  and  $D' \subset \Delta_1(Y)$  and a bijection

$$i: \Delta_1(X) - D \to \Delta_1(Y) - D'$$

and a Poisson isomorphism

$$R: \tilde{\Omega}_{X,r}(D)^{-1}(2\pi) \to \tilde{\Omega}_{Y,r}(D')^{-1}(2\pi),$$

which is covered by an affine  $\mathbb{R}$ -bundle isomorphism from  $\widetilde{\mathrm{LS}}_r(X)|_{\tilde{\Omega}_{X,r}(D)^{-1}(2\pi)}$  to  $\widetilde{\mathrm{LS}}_r(Y)_{\tilde{\Omega}_{X,r}(D')^{-1}(2\pi)}$  and such that we get the following commutative diagram

$$\begin{split} \tilde{\Omega}_{X,r}(D)^{-1}(2\pi) & \xrightarrow{R} \tilde{\Omega}_{Y,r}(D')^{-1}(2\pi) \\ & \downarrow_{\text{proj } \circ \tilde{\Omega}_{X,r}} & \downarrow_{\text{proj } \circ \tilde{\Omega}_{Y,r}} \\ & \mathbb{R}^{\Delta_1(X)-D} \xrightarrow{i^*} & \mathbb{R}^{\Delta_1(Y)-D'}. \end{split}$$

Moreover, if X and Y are admissible and admissibly equivalent, the isomorphism R takes an open convex subset U of  $S_r(X) \cap \tilde{\Omega}_{X,r}(D)^{-1}(2\pi)$  onto an open convex subset U' of  $S_r(Y) \cap \tilde{\Omega}_{Y,r}(D)^{-1}(2\pi)$ .

We remark that in the previous notation  $D = \Delta_1(X) \cap h^{-1}(D_Y)$  and  $D' = \Delta_1(Y) \cap h(D_X)$ .

For a tetrahedron  $T = [v_0, v_1, v_2, v_3]$  in  $\mathbb{R}^3$  with ordered vertices  $v_0, v_1, v_2, v_3$ , we define its sign

$$sign(T) = sign(det(v_1 - v_0, v_2 - v_0, v_3 - v_0)),$$

as well as the signs of its faces

$$\operatorname{sign}(\partial_i T) = (-1)^i \operatorname{sign}(T), \text{ for } i \in \{0, \dots, 3\}.$$

For a pseudo 3-manifold X, the signs of faces of the tetrahedra of X induce a sign function on the faces of the boundary of X,  $\operatorname{sign}_X : \Delta_2(\partial X) \to \{\pm 1\}$ , which permits us to split the components of the boundary of X into two sets,  $\partial X = \partial_+ X \cup \partial_- X$ , where  $\Delta_2(\partial_\pm X) = \operatorname{sign}_X^{-1}(\pm 1)$ . Notice that  $|\Delta_2(\partial_+ X)| = |\Delta_2(\partial_- X)|$ .

**Definition 6.17** (Cobordism Categorid). The category  $\mathcal{B}$  is the category that has triangulated surfaces as objects, equivalence classes of (leveled) shaped pseudo 3-manifolds X as morphisms (so that  $X \in \text{Hom}_{\mathcal{B}}(\partial_{-}X, \partial_{+}X)$ ) and the composition given by glueing along boundary components, through edge orientation preserving and face orientation reversing CW-homeomorphisms.

The Categorid  $\mathcal{B}_a$  is the subcategorid of  $\mathcal{B}$  whose morphisms are restricted to be admissible equivalence classes of admissible (leveled) shaped pseudo 3-manifolds. In particular composition is possible only if the gluing gives an (leveled) admissible pseudo 3-manifold.

#### Remark 6.18. Admissible Shaped Pseudo 3-Manifolds in the real world.

Even though we will discuss the whole Andersen Kashaev construction of the Teichmüller TQFT functor in the general setting of the above defined cobordism categorid, the the main parts of this construction, we want to put our hands on in this paper, are invariants of links. We interpret Triangulated Pseudo 3manifolds X as ideal triangulations of the (non closed) manifold  $X \setminus \Delta_0(X)$ . This interpretation is enlighten in Remark 6.3. We shall ask ourself when a cusped 3manifold (cusped means non compact with finite volume here) admits a positive fully balanced shape structure. This requirement is weaker than asking for a full geometric structure on the manifold and in our language, this can be expressed by the fact that we did not required a precise gauge to be fixed. The problem of finding positive or generalized angle structures has been studied in [LT], where necessary and sufficient conditions for their existence are given. In the work [HRS] it is proved, among other things, that a particular class of manifolds Msupporting positive shape structures are complements in  $S^3$  of hyperbolic links. However the admissibility conditions kicks in here and further restrict us to just complements of hyperbolic knots. So, at the least, we know that the Andersen Kashaev construction will work on complements of hyperbolic knots, and that are the examples we will look a bit closer at below. Now we should clarify the equivalence relation in  $\mathcal{B}_a$ , in the context of knot complements. Combinatorially speaking, any two ideal triangulations of a knot complement are related by finite sequences of 3-2 or 2-3 Pachner moves. On the other hand it is not known (at least to the authors) if any such sequence can be realised as a sequence of shaped Pachner moves. For sure we know that 3–2 shaped Pachner moves are well defined in the category  $\mathcal{B}_a$  as we remarked when we defined them, and if a shaped Pachner 2–3 move is possible in some particular case, then it is an equivalence in the category  $\mathcal{B}_a$ . So the knot invariants that we will define starting from  $\mathcal{B}_a$  are not guaranteed to be topological invariants. There is however another construction of the Andersen–Kashaev invariant [AK2], that avoid this problem with analytic continuation properties of the partition function. The equivalence of the two constructions is still conjectural though.

In [AK1] another way to define knot invariants is suggested, by taking one vertex Hamiltonian triangulations of knots, that is, one vertex triangulations of  $S^3$  (or a general manifold M) where the knot is represented by a unique edge with a degenerating shape structure, meaning that we take a limit on the shapes, sending all the weights to be balanced except the weight of the knot that is sent to 0. The partition function is actually divergent but a residue can be computed as an invariant. We will show this in a couple of examples in subsection 6.4.

#### 6.2 The target Categorid $\mathcal{D}_N$

W recall all the relevant things regarding tempered distributions and the space  $\mathcal{S}(\mathbb{A}_N)$  in Appendix A. As always, here N is an odd positive integer and  $\mathbf{b} \in \mathbb{C}$  is fixed to satisfy  $\operatorname{Re}(\mathbf{b}) > 0$  and  $\operatorname{Im} \mathbf{b}(1 - |\mathbf{b}|) = 0$ .

**Definition 6.19.** The categorid  $\mathcal{D}_N$  has as objects finite sets and for two finite sets n, m the set of morphisms from n to m is

$$\operatorname{Hom}_{\mathcal{D}_N}(n,m) = \mathcal{S}'(\mathbb{A}_N^{n \sqcup m}) \simeq \mathcal{S}'(\mathbb{R}^{n \sqcup m}) \otimes \mathcal{S}((\mathbb{Z}/N\mathbb{Z})^{n \sqcup m}).$$

**Definition 6.20.** For  $\mathcal{A} \otimes A_N \in \text{Hom}_{\mathcal{D}_N}(n, m)$  and  $\mathcal{B} \otimes B_N \in \text{Hom}_{\mathcal{D}_N}(m, l)$ , such that  $\mathcal{A}$  and  $\mathcal{B}$  satisfy condition (A.3) and  $\pi^*_{n,m}(\mathcal{A})\pi^*_{m,l}(\mathcal{B})$  continuously extends to  $S(\mathbb{R}^{n \sqcup m \sqcup l})_m$ , we define

$$(\mathcal{A} \otimes A_N) \circ (\mathcal{B} \otimes B_N) = (\pi_{n,l})_* (\pi_{n,m}^*(\mathcal{A}) \pi_{m,l}^*(\mathcal{B})) \otimes A_N B_N \in \operatorname{Hom}_{\mathcal{D}_N}(n,l).$$

Where the product  $A_N B_N$  is just the matrix product.

We will frequently use the following notation in what follows. For any  $a \in \mathbb{A}_N$ ,  $a = (x, n) \in \mathbb{R} \times \mathbb{Z}/N\mathbb{Z}$  we will consider the b-dependent operator

$$\varepsilon \equiv \varepsilon(\mathbf{b}) \colon \mathbb{A}_N \to \mathbb{A}_N$$

defined by

(6.2) 
$$\varepsilon(x,n) \equiv \begin{cases} (x,n) & \text{if } |\mathbf{b}| = 1, \\ (x,-n) & \text{if } \mathbf{b} \in \mathbb{R} \end{cases}$$

For any  $\mathcal{A} \in \mathcal{L}(\mathcal{S}(\mathbb{R}^n), \mathcal{S}'(\mathbb{R}^m))$ , we have a unique adjoint  $\mathcal{A}^* \in \mathcal{L}(\mathcal{S}(\mathbb{R}^m), \mathcal{S}'(\mathbb{R}^n))$  defined by the formula

$$\mathcal{A}^*(f)(g) = \bar{f}(\mathcal{A}(\bar{g}))$$

for all  $f \in \mathcal{S}(\mathbb{R}^m)$  and all  $g \in \mathcal{S}(\mathbb{R}^n)$ .

**Definition 6.21** ( $\star_{\mathbf{b}}$  structure). Consider  $\mathbf{b} \in \mathbb{C}$  fixed as above and  $N \in \mathbb{Z}_{>0}$  odd. Let  $A_N \in \text{Hom}(\mathcal{S}((\mathbb{Z}/N\mathbb{Z})^m), \mathcal{S}((\mathbb{Z}/N\mathbb{Z})^n))$ . Recall the involution  $\varepsilon$  on  $\mathbb{Z}/N\mathbb{Z}$  form equation (6.2). Define  $A_N^{\star_{\mathbf{b}}}$  as

(6.3) 
$$\langle j_1, \dots j_m | A_N^{\star_b} | p_1, \dots, p_n \rangle = \overline{\langle \varepsilon p_1, \dots, \varepsilon p_n | A_N | \varepsilon j_1, \dots, \varepsilon j_m \rangle}$$

We can finally define the  $*_b$  operator as

$$(6.4) \qquad (\mathcal{A} \otimes A_N)^{*_{\mathbf{b}}} = \mathcal{A}^* \otimes A_N^{*_{\mathbf{b}}}$$

#### 6.3 Tetrahedral Partition Function

Recall the operators from Section 4.1,  $\mathbf{u}_j$ ,  $\mathbf{v}_j$ ,  $X_j$ ,  $Y_j$ ,  $\mathbf{p}_j$ ,  $\mathbf{q}_j$ , j = 1, 2 acting on  $\mathcal{H} := \mathcal{S}(\mathbb{A}_N^2)$ . Define the *Charged Tetrahedral Operator* as follows

**Definition 6.22.** Let a, b, c > 0 such that  $a + b + c = \frac{1}{\sqrt{N}}$ . Recall the Tetrahedral operator T defined in (4.15). Define the charged tetrahedral operator T(a, c) as follows

(6.5) 
$$\mathsf{T}(a,c) \equiv e^{-\pi i \frac{c_{\rm b}^2}{\sqrt{N}} \left(2(a-c) + \frac{1}{\sqrt{N}}\right)/6} e^{2\pi i c_{\rm b}(c\mathbf{q}_2 - a\mathbf{q}_1)} \mathsf{T}_{12} e^{-2\pi i c_{\rm b}(a\mathbf{p}_2 + c\mathbf{q}_2)}$$

Lemma 6.23. We have that

$$(6.6) \ \mathsf{T}(a,c) = e^{-\pi i \frac{c_{\mathrm{L}}^2}{\sqrt{N}} \left(2(a-c) + \frac{1}{\sqrt{N}}\right)/6} e^{\pi i c_{\mathrm{b}}^2 a(a+c)} \mathsf{D}_{12} \psi_{a,c}(\mathsf{q}_1 + \mathsf{p}_2 - \mathsf{q}_2, -e^{-\frac{\pi i}{N}} Y_1 X_2 \overline{Y_2})$$

were  $\psi_{a,c}(x,n)$  is the charged quantum dilogarithm from (3.28)

*Extra Notation*. Recall the notation for Fourier coefficients and Gaussian exponentials in  $\mathbb{A}_N$ . For a = (x, n) and a' = (y, m) in  $\mathbb{A}_N$  we write

$$\langle a, a' \rangle \equiv e^{2\pi i x y} e^{-2\pi i n m/N} \qquad \langle a \rangle \equiv e^{\pi i x^2} e^{-\pi i n (n+N)/N}$$

For  $a = (x, n) \in \mathbb{A}_N$ , define  $\delta(a) \equiv \delta(x)\delta(n)$  where  $\delta(x)$  is Dirac's delta distribution while  $\delta(n)$  is the Kronecker delta  $\delta_{0,n}$  between 0 and  $n \mod N$ . Define

(6.7) 
$$\varphi_{a,c}(x,n) \equiv \psi_{a,c}(x,-n).$$

Denote, for  $x, y \in \mathbb{R}$  and  $z \in \mathbb{A}_N$ 

(6.8) 
$$\nu(x) \equiv e^{-\pi i \frac{c_{\rm b}^2}{\sqrt{N}} \left(2x + \frac{1}{\sqrt{N}}\right)/6} \qquad \nu_{x,y} = \nu(x-y) e^{\pi i c_{\rm b}^2 x(x+y)}$$

The equations from Lemma 3.17 can be upgraded to

(6.9) 
$$\nu_{a,c}\tilde{\varphi}_{a,c}(z) = \nu_{c,b}\varphi_{c,b}(z)\langle z\rangle e^{-\pi iN/12}$$

(6.10) 
$$\nu_{a,c}\overline{\varphi_{a,c}}(z) = \nu_{c,a}\varphi_{c,a}(-\varepsilon z)\langle z\rangle e^{-\pi i N/6}$$

(6.11) 
$$\nu_{a,c}\overline{\tilde{\varphi}_{a,c}}(z) = \nu_{b,c}\varphi_{b,c}(-\varepsilon z)e^{-\pi i N/12}$$

Where  $\varepsilon$  was defined in (6.2). From the Charged Pentagon Equation (3.33) we get the following

**Proposition 6.24** (Charged Tetrahedral Pentagon equation). Let  $a_j$ ,  $c_j > 0$  such that  $\frac{1}{\sqrt{N}} - a_j - c_j > 0$  for j = 0, 1, 2, 3 and 4, which further satisfies the following relations

(6.12)

 $a_1 = a_0 + a_2$   $a_3 = a_2 + a_4$   $c_1 = c_0 + a_4$   $c_3 = a_0 + c_4$   $c_2 = c_1 + c_3$ .

Then we have that

(6.13) 
$$\mathsf{T}_{12}(a_4, c_4) \mathsf{T}_{1,3}(a_2, c_2) \mathsf{T}_{23}(a_0, c_0) = \mu \mathsf{T}_{23}(a_1, c_1) \mathsf{T}_{12}(a_3, c_3)$$

where

$$\mu = \exp \pi i \frac{c_{\rm b}^2}{6\sqrt{N}} \left( 2(c_0 + a_2 + c_4) - \frac{1}{\sqrt{N}} \right)$$

We have an integral kernel description for the charged tetrahedral operator. We use the Bra-Ket notation to denote integral kernels, see Appendix A.1.

Proposition 6.25. Let  $\overline{\mathsf{T}}(a,c) \equiv (\mathsf{T}(a,c))^{-1}$ .

$$\begin{aligned} \langle a_0, \, a_2 | \, \mathsf{T}_{12}(a,c) \, | a_1, \, a_3 \rangle \\ &= \nu(a-c) e^{\pi i c_b^2 a(a+c)} \langle a_3 - a_2, \, a_0 \rangle \overline{\langle a_3 - a_2 \rangle} \delta(a_0 + a_2 - a_1) \tilde{\varphi}_{a,c}(a_3 - a_2) \\ \langle a_0, \, a_2 | \overline{\mathsf{T}}(a,c) | a_1, \, a_3 \rangle \\ &= \nu(b-c) e^{\pi i c_b^2 b(b+c)} e^{-\pi i N/12} \langle a_3 - a_2, \, a_1 \rangle \langle a_3 - a_2 \rangle \delta(a_1 + a_3 - a_0) \varphi_{b,c}(a_3 - a_2) \end{aligned}$$

The appearance of  $\varepsilon$  is due to the non-unitarity of the theory for b > 0 and N > 1.

Let A and B two operators on  $L^2(\mathbb{A}_N)$  defined as bra-ket distributions by

$$\begin{array}{ll} (6.14) & \langle a_1, a_2 | \mathsf{A} \rangle = \delta(a_1 + a_2) \langle a_1 \rangle e^{\pi i N/12} & \langle a_1, a_2 | \mathsf{B} \rangle = \langle a_1 - a_2 \rangle \\ (6.15) & \langle \overline{\mathsf{A}} | a_1, a_2 \rangle = \overline{\langle \varepsilon a_1, \varepsilon a_2 | \mathsf{A} \rangle} & \langle \overline{\mathsf{B}} | a_1, a_2 \rangle = \overline{\langle \varepsilon a_1, \varepsilon a_2 | \mathsf{B} \rangle}. \end{array}$$

Lemma 6.26 (Fundamental Lemma). We have the following three relations

(6.16) 
$$\int_{\mathbb{A}_N^2} \langle \overline{\mathsf{A}} | v, s \rangle \langle x, s | \mathsf{T}(a, c) | u, t \rangle \langle t, y | \mathsf{A} \rangle \mathrm{d}s \mathrm{d}t = \langle x, y | \overline{\mathsf{T}}(a, b) \langle u, v \rangle$$

(6.17) 
$$\int_{\mathbb{A}_N^2} \langle \overline{\mathsf{A}} | u, s \rangle \langle s, x | \mathsf{T}(a, c) | v, t \rangle \langle t, y | \mathsf{B} \rangle \mathrm{d}s \mathrm{d}t = \langle x, y | \overline{\mathsf{T}}(b, c) \langle u, v \rangle$$

(6.18) 
$$\int_{\mathbb{A}_N^2} \langle \overline{\mathsf{B}} | u, s \rangle \langle s, y | \mathsf{T}(a, c) | t, v \rangle \langle t, x | \mathsf{B} \rangle \mathrm{d}s \mathrm{d}t = \langle x, y | \overline{\mathsf{T}}(a, b) \langle u, v \rangle.$$

#### 6.3.1 TQFT Rules, Tetrahedral Symmetries and Gauge Invariance

We consider oriented surfaces with cellular structure such that all 2-cells are either bigons or triangles. Not all the edge orientations will be admitted. We forbid cyclically oriented triangles. For the bigons, we consider only the *essential* ones, the others being contractible to an edge. These essential bigons are precisely the ones with cellular structure isomorphic to the unit disk with vertices  $\pm 1 \in \mathbb{C}$  and edges  $\{e_1 = e^{\pi i t}; e_2 = -e^{\pi i t}, \text{ for } t \in [0,1]\}$  or  $\{e_1 = -e^{-\pi i t}; e_2 = e^{-\pi i t}, \text{ for } t \in [0,1]\}$ . Given such an ideally triangulated surface  $\Sigma$  we will associate a copy of  $\mathbb{C}$ to any bigon and a copy of  $\mathcal{S}'(\mathbb{A}_N)$  to any triangle. Globally we associate to the surface the space  $\mathcal{S}'(\mathbb{A}_N^{\Delta_2(\Sigma)})$ . To a shaped tetrahedron T with ordered vertices  $\{v_0, v_1, v_2, v_3\}$  we associate the partition function  $Z_{\rm b}^{(N)}(T)$  through the Nuclear Theorem (A.6) as a ket distribution

(6.19) 
$$\langle x | \widetilde{Z_{b}^{(N)}(T)} \rangle = \begin{cases} \langle a_0, a_2 | \mathsf{T}(c(v_0v_1), c(v_0v_3)) | a_1, a_3 \rangle & \text{if } \operatorname{sign}(T) = 1; \\ \langle a_1, a_3 | \mathsf{T}(c(v_0v_1), c(v_0v_3)) | a_0, a_2 \rangle & \text{if } \operatorname{sign}(T) = -1 \end{cases}$$

where

$$\mathbb{A}_N \ni a_i := a(\partial_i T), \quad i \in \{0, 1, 2, 3\}$$

and

$$c := \frac{1}{\pi\sqrt{N}} \alpha_T : \Delta_1(T) \to \mathbb{R}_{>0}.$$

Having allowed bigons in triangulations of surfaces, we must also allow cones over such as cobordisms. From the 2 classes of bigons described above we have 4 isotopy classes of cellular structures of cones over them, described in the following as embedded in  $\mathbb{R}^3 \simeq \mathbb{C} \times \mathbb{R}$ . The bigon is identified with the unit disc embedded in  $\mathbb{C}$ . The apex of the cone will be the point  $(0,1) \in \mathbb{C} \times \mathbb{R}$ . The 1-cells will be either

$$\{e_{0\pm}^1(t) = (\pm e^{i\pi t}, 0), \ e_{1\pm}^1(t) = (\pm (1-t), t)\}$$

or

$$\{e_{0\pm}^1(t) = (\mp e^{-i\pi t}, 0), \ e_{1\pm}^1(t) = (\pm(1-t), t)\}$$

or

$$\{e_{0\pm}^1(t) = (\pm e^{i\pi t}, 0), \ e_{1\pm}^1(t) = (\pm t, 1-t)\}$$

or

$$\{e_{0\pm}^1(t) = (\mp e^{-i\pi t}, 0), \ e_{1\pm}^1(t) = (\pm t, 1-t)\}$$

We name these types of cones  $A_+$ ,  $A_-$ ,  $B_+$  and  $B_-$  respectively. We need TQFT rules for the gluing of these cones. We just need to consider their gluing to a tetrahedra. We assign a partition function to the cones as follows (6.20)

$$\langle a_1, a_2 | Z_{\rm b}^{(N)}(A_{\pm}) \rangle = \delta(a_1 + a_2) \langle a_1 \rangle^{\pm 1} e^{\pm \pi i N/12}, \quad \langle a_1, a_2 | Z_{\rm b}^{(N)}(B_{\pm}) \rangle = \langle a_1 - a_2 \rangle^{\pm 1}.$$

Tetrahedral symmetries are generated by permutation of the ordered vertices. Indeed the group of tetrahedral symmetries is identified with the symmetric group  $S_4$  and is generated by three transpositions. The three equations of the Fundamental Lemma 6.26 gain an interpretation as glueing of cones on the faces of a tetrahedron through definitions (6.20). These three glueing generates all the symmetries of a tetrahedron, and through this interpretation, the Fundamental Lemma assure that the partition function  $Z_b^{(N)}$  satisfies all the tetrahedral symmetries. For more details on tetrahedral symmetries and the cone's partition function see [AK1, GKT].

We can now formulate the main Theorem for the Teichmüller TQFT. This theorem was proved by Andersen and Kashaev for the case N = 1 in [AK1]. The statement that we have here is for every N odd, and it is strictly speaking not present as such in the literature.

**Theorem 6.27** (Level N Teichmüller TQFT, Andersen and Kashaev). For any  $b \in \mathbb{C}^*$  such that  $\operatorname{Im} b(|b| - 1) = 0$  and  $\operatorname{Re} b > 0$ , and for any  $N \in \mathbb{Z}_{>0}$  odd there exists a unique  $*_b$ -functor  $F_b^{(N)}$ :  $\mathcal{B}_a \to \mathcal{D}_N$  such that  $F_b^{(N)}(A) = \Delta_2(A), \forall A \in Ob \mathcal{B}_a$ , and for any admissible leveled shaped pseudo 3-manifold  $(X, l_X)$ , the associated morphism in  $\mathcal{D}_N$  takes the form

(6.21) 
$$F_{\mathbf{b}}^{(N)}(X, l_X) = Z_{\mathbf{b}}^{(N)}(X)e^{-\pi i \frac{l_X \circ_{\mathbf{b}}^2}{N}} \in \mathcal{S}'\left(\mathbb{A}_N^{\Delta_2(\partial X)}\right),$$

where  $Z_{\rm b}^{(N)}$  is defined in (6.19) for a tetrahedron.

Here  $*_{b}$ -functor means that  $F_{b}^{(N)}(X^{*}) = F_{b}^{(N)}(X)^{*_{b}}$ , where  $X^{*}$  is the oppositely oriented pseudo 3-manifold to X.

The discussion so far proves the theorem except for the gauge invariance and the convergence of the partition functions under glueings. We will not discuss the convergence here because it follows directly from the convergence in the case level N = 1, which was addressed in [AK1]. We just remark that the hypothesis of admissibility is used to prove the convergence of the partition function.

For the gauge invariance consider the suspension of an *n*-gone  $SP_n$  naturally triangulated into *n* tetrahedra sharing the only internal edge *e*. Every gauge transformation can be decomposed in a sequence of gauge transformations involving only one edge *e*, and every such gauge transformation can be understood in the example of the suspension. Suppose all the tetrahedra to be positive, and having vertex order such that the last two vertices are the endpoints of the internal common edge. After enumerating the tetrahedra in cyclic order, let  $a_i$ ,  $c_i$  be the two shape parameter of  $T_i$ , i = 0, ..., n, and  $\mathbf{a} = (a_0, ..., a_n)$ ,  $\mathbf{c} = (c_0, ..., c_n)$ . Notice that  $\sqrt{N\pi}a_i$  is the dihedral angle corresponding to the edge *e*. So a gauge transformation corresponding to *e* will affect the partition function of  $SP_N$ 

$$Z_{\rm b}^{(N)}(SP_N)({\rm a,c}) := {\rm Tr}_0({\sf T}_{01}(a_1,c_1){\sf T}_{02}(a_2,c_2)\cdots{\sf T}_{0n}(a_n,c_n))$$

by shifting c by an amount  $\lambda = (\lambda, ..., \lambda)$  say. One can show from the definitions and the discussion above, that

$$\mathsf{T}(a, c+\lambda) = e^{-2\pi i c_{\mathrm{b}}\lambda\mathsf{p}_{1}}\mathsf{T}(a, c)e^{2\pi i c_{\mathrm{b}}\lambda\mathsf{p}_{1}}e^{\pi i c_{\mathrm{b}}^{2}\left(\frac{1}{\sqrt{N}}-6a\right)\lambda/3}$$

which, after tracing, leads to the following

Proposition 6.28. [AK1]

$$Z_{\rm b}^{(N)}(SP_N)({\rm a,c}+\lambda) = Z_{\rm b}^{(N)}(SP_N)({\rm a,c})e^{\pi i c_{\rm b}^2 \left(\frac{n}{\sqrt{N}} - 6Q_e\right)\lambda/3}$$

where

$$Q_e = a_1 + a_2 + \dots a_n$$

#### 6.4 Knot Invariants: Computations and Conjectures

In this section we update the examples computed in [AK1] to the level  $N \ge 1$  setting. Similar results were obtained in [D].

*Notation.* In the examples we are going to use the following notation for quantum dilogarithms

(6.22) 
$$\varphi_{\mathbf{b}}(x,n) \equiv \mathbf{D}_{\mathbf{b}}(x,-n).$$

Moreover we will often abuse notation in favor of readability in the following ways. For  $z = (x, n) \in \mathbb{A}_N$  we will sometimes write  $e^{2\pi i c_b z \alpha}$  in place of  $e^{2\pi i c_b x \alpha}$ . Moreover sums of the form  $z + c_b a$  will always mean  $(x + c_b a, n)$ .

In the following examples we encode an oriented triangulated pseudo 3-manifold X into a diagram where a tetrahedron T is represented by an element

where the vertical segments, ordered from left to right, correspond to the faces  $\partial_0 T$ ,  $\partial_1 T$ ,  $\partial_2 T$ ,  $\partial_3 T$  respectively. When we glue tetrahedron along faces, we illustrate this by joining the corresponding vertical segments.

#### 6.4.1 Figure–Eight Knot 4<sub>1</sub>

Let X be represented by the diagram

(6.23)

XX

Choosing an orientation, it consists of one positive tetrahedron  $T_+$  and one negative tetrahedron  $T_-$  with four identifications

$$\partial_{2i+j}T_+ \simeq \partial_{2-2i+j}T_-, \quad i, j \in \{0, 1\}.$$

Combinatorially, we have  $\Delta_0(X) = \{*\}, \Delta_1(X) = \{e_0, e_1\}, \Delta_2(X) = \{f_0, f_1, f_2, f_3\},$ and  $\Delta_3(X) = \{T_+, T_-\}$  with the boundary maps

$$\begin{split} f_{2i+j} &= \partial_{2i+j} T_+ = \partial_{2-2i+j} T_-, \quad i, j \in \{0, 1\}, \\ \partial_i f_j &= \begin{cases} e_0, & \text{if } j - i \in \{0, 1\}; \\ e_1, & \text{otherwise}, \end{cases} \\ \partial_i e_j &= *, \quad i, j \in \{0, 1\}. \end{split}$$

The topological space  $X \setminus \{*\}$  is homeomorphic to the complement of the figure– eight knot, and indeed  $X \setminus \{*\}$  is an ideal triangulation of such a cuspidal manifold. The set  $\Delta_{3,1}(X)$  consists of the elements  $(T_{\pm}, e_{j,k})$  for  $0 \leq j < k \leq 3$ . We fix a shape structure

$$\alpha_X \colon \Delta_{3,1}(X) \to \mathbb{R}_{>0}$$

by the formulae

$$\alpha_X(T_{\pm}, e_{0,1}) = \pi \sqrt{N} a_{\pm}, \quad \alpha_X(T_{\pm}, e_{0,2}) = \pi \sqrt{N} b_{\pm}, \quad \alpha_X(T_{\pm}, e_{0,3}) = \pi \sqrt{N} c_{\pm},$$

where  $a_{\pm} + b_{\pm} + c_{\pm} = \frac{1}{\sqrt{N}}$ . The weight function

$$\omega_X \colon \Delta_1(X) \to \mathbb{R}_{>0}$$

takes the values

$$\omega_X(e_0) = \sqrt{N\pi(2a_+ + c_+ + 2b_- + c_-)} =: 2\pi w, \quad \omega_X(e_1) = 2\pi(2 - w).$$

As the figure–eight knot is hyperbolic, the completely balanced case w = 1 is accessible directly. We can state the balancing condition w = 1 as

$$(6.24) 2b_+ + c_+ = 2b_- + c_-.$$

The kernel representations for the operators  $\mathsf{T}(a_+, c_+)$  and  $\mathsf{T}(a_-, c_-)$  are as follows. Let  $z_j \in \mathbb{A}_N$ , j = 0, 1, 2, 3,

$$(6.25) \quad \langle z_0, z_2 | \mathsf{T}(a_+, c_+) | z_1, z_3 \rangle \\ = \nu_{a_+, c_+} \langle z_3 - z_2, z_0 \rangle \overline{\langle z_3 - z_2 \rangle} \delta(z_0 + z_2 - z_1) \tilde{\varphi}_{a_+, c_+}(z_3 - z_2) \\ (6.26) \quad \langle z_3, z_1 | \overline{\mathsf{T}}(a_+, c_-) | z_2, z_0 \rangle = \overline{\langle \varepsilon z_2, \varepsilon z_0 | \mathsf{T}(a_+, c_-) | \varepsilon z_3, \varepsilon z_1 \rangle} \\ = \overline{\nu_{a_-, c_-}} \langle z_0 - z_1, z_2 \rangle \langle z_1 - z_0 \rangle \delta(z_0 + z_2 - z_3) \overline{\tilde{\varphi}_{a_-, c_-}(\varepsilon z_1 - \varepsilon z_0)}$$

The Andersen–Kashaev invariant at level N for the complement of the figure–eight knot is then

$$Z_{\rm b}^{(N)}(X) = \int_{\mathbb{A}_N^4} \langle z_0, z_2 | \mathsf{T}(a_+, c_+) | z_1, z_3 \rangle \langle z_3, z_1 | \overline{\mathsf{T}}(a_+, c_-) | z_2, z_0 \rangle \mathrm{d}z_0 \mathrm{d}z_1 \mathrm{d}z_2 \mathrm{d}z_3$$

$$\begin{split} &= \int_{\mathbb{A}_{N}^{4}} \nu_{c_{+},b_{+}} \overline{\nu_{c_{-},b_{-}}} \varphi_{c_{+},b_{+}} (z_{3}-z_{2}) \overline{\varphi_{c_{-},b_{-}}(\varepsilon z_{1}-\varepsilon z_{0})} \delta(z_{0}+z_{2}-z_{1}) \times \\ &\times \delta(z_{0}+z_{2}-z_{3}) \langle z_{3}-z_{2},z_{0} \rangle \langle z_{0}-z_{1},z_{2} \rangle \mathrm{d}z_{0} \mathrm{d}z_{1} \mathrm{d}z_{2} \mathrm{d}z_{3} \\ &= \int_{\mathbb{A}_{N}^{3}} \nu_{c_{+},b_{+}} \overline{\nu_{c_{-},b_{-}}} \varphi_{c_{+},b_{+}} (z_{1}-z_{2}) \overline{\varphi_{c_{-},b_{-}}(\varepsilon z_{1}-\varepsilon z_{0})} \delta(z_{0}+z_{2}-z_{1}) \times \\ &\times \langle z_{1}-z_{2},z_{0} \rangle \langle z_{0}-z_{1},z_{2} \rangle \mathrm{d}z_{0} \mathrm{d}z_{1} \mathrm{d}z_{2} \\ &= \int_{\mathbb{A}_{N}^{2}} \nu_{c_{+},b_{+}} \overline{\nu_{c_{-},b_{-}}} \varphi_{c_{+},b_{+}} (z_{0}) \overline{\varphi_{c_{-},b_{-}}(\varepsilon z_{2})} \langle z_{0},z_{0} \rangle \langle -z_{2},z_{2} \rangle \mathrm{d}z_{0} \mathrm{d}z_{2} \\ &= \int_{\mathbb{A}_{N}} \nu_{c_{+},b_{+}} \overline{\varphi_{c_{+},b_{+}}} (z_{0}) \langle z_{0},z_{0} \rangle \mathrm{d}z_{0} \overline{\int_{\mathbb{A}_{N}} \nu_{c_{-},b_{-}} \varphi_{c_{-},b_{-}}(\varepsilon z_{2})} \langle z_{2},z_{2} \rangle \mathrm{d}z_{2} \\ &= \sigma_{c_{+},b_{+}} \overline{\sigma_{c_{-},b_{-}}} \end{split}$$

We can compute

$$\begin{split} \sigma_{c\pm,b\pm} &= \nu_{c\pm,b\pm} \int_{\mathbb{A}_N} \frac{e^{-2\pi i c_{\rm b} z c_{\pm}}}{\varphi_{\rm b} (z - c_{\rm b} (b_{\pm} + c_{\pm}))} \langle z \rangle^2 \mathrm{d}z \\ &= \nu_{c\pm,b\pm}' \int_{\mathbb{A}_N + di} \frac{e^{4\pi i c_{\rm b} z (2b_{\pm} + c_{\pm})}}{\varphi_{\rm b}(z)} \langle z \rangle^2 \mathrm{d}z \end{split}$$

where

(6.27) 
$$\nu_{c_{\pm},b_{\pm}}' = \nu_{c_{\pm},b_{\pm}} e^{4\pi i c_{\rm b}^2 (c_{\pm} b_{\pm} - b_{\pm}^2)}$$

and the domain of integration  $\mathbb{A}_N + di = (\mathbb{R} + di) \times \mathbb{Z}/N\mathbb{Z})$ . Note we have shifted the real integral to a contour integral in the complex plane, and  $d \in \mathbb{R}$  is such that the integral converges absolutely. We sometimes omit the contour shift in the computations but we state it in the results. Defining

$$\lambda \equiv 2b_+ + c_+ = 2b_- + c_-$$

we have

$$Z_{\mathbf{b}}^{(N)}(X) = \nu_{c_{+},b_{+}}' \overline{\nu_{c_{-},b_{-}}'} \int_{\mathbb{A}_{N}^{2}} \frac{e^{4\pi i c_{\mathbf{b}}\lambda(z_{0}+z_{2})}}{\varphi_{\mathbf{b}}(z_{0})\overline{\varphi_{\mathbf{b}}(\varepsilon z_{2})}} \langle z_{0} \rangle^{2} \overline{\langle z_{2} \rangle^{2}} \mathrm{d}z_{0} \mathrm{d}z_{2}$$
$$= \nu_{c_{+},b_{+}}' \overline{\nu_{c_{-},b_{-}}'} \int_{\mathbb{A}_{N}^{2}} \frac{\varphi_{\mathbf{b}}(z_{2})}{\varphi_{\mathbf{b}}(z_{0})} e^{4\pi i c_{\mathbf{b}}\lambda(z_{0}+z_{2})} \langle z_{0} \rangle^{2} \overline{\langle z_{2} \rangle^{2}} \mathrm{d}z_{0} \mathrm{d}z_{2}$$
$$= \nu_{c_{+},b_{+}}' \overline{\nu_{c_{-},b_{-}}'} \int_{\mathbb{A}_{N}^{2}} \frac{\varphi_{\mathbf{b}}(z_{2}-z_{0})}{\varphi_{\mathbf{b}}(z_{0})} e^{4\pi i c_{\mathbf{b}}\lambda z_{2}} \langle z_{0}, z_{2} \rangle^{2} \overline{\langle z_{2} \rangle^{2}} \mathrm{d}z_{0} \mathrm{d}z_{2}$$

130

that has the structure

$$\begin{array}{l} (6.28) \\ Z_{\rm b}^{(N)}(X) = e^{i\phi} \int_{\mathbb{A}_N + i0} \chi_{4_1}^{(N)}(x,\lambda) \mathrm{d}x, \\ (6.29) \\ \chi_{4_1}^{(N)}(x,\lambda) = \chi_{4_1}^{(N)}(x) e^{4\pi i c_{\rm b} \lambda x}, \qquad \qquad \chi_{4_1}^{(N)}(x) = \int_{\mathbb{A}_N - i0} \frac{\varphi_{\rm b}(x-y)}{\varphi_{\rm b}(y)} \langle x, y \rangle^2 \overline{\langle x \rangle^2} \mathrm{d}y \end{array}$$

where  $\phi$  is some constant quadratic combination of dihedral angles.

#### 6.4.2 The Complement of the Knot 52

Let X be the closed S.O.T.P. 3-manifold represented by the diagram



This triangulation has only one vertex \* and  $X \setminus \{*\}$  is topologically the complement of the knot 5<sub>2</sub>. We denote  $T_1, T_2, T_3$  the left, right, and top tetrahedra respectively. We choose the orientation so that all of them are positive. Balancing all the edges correspond to require the following equations to be true

$$(6.30) 2a_3 = a_1 + c_2, b_3 = c_1 + b_2.$$

The three integral kernels reads

$$\begin{split} \langle z, w | \mathsf{T}(a_1, c_1) | u, x \rangle &= \\ &= \nu_{a_1, c_1} \langle x - w, z \rangle \overline{\langle x - w \rangle} \delta(z + w - u) \tilde{\varphi}_{a_1, c_1}(x - w) \\ \langle x, v | \mathsf{T}(a_2, c_2) | y, w \rangle &= \\ &= \nu_{a_2, c_2} \langle w - v, x \rangle \overline{\langle w - v \rangle} \delta(x + v - y) \tilde{\varphi}_{a_2, c_2}(w - v) \\ \langle y, u | \mathsf{T}(a_3, c_3) | v, z \rangle &= \\ &= \nu_{a_3, c_3} \langle z - u, y \rangle \overline{\langle z - u \rangle} \delta(y + u - v) \tilde{\varphi}_{a_3, c_3}(z - u) \end{split}$$

Carrying out the computations, defining  $\lambda = -c_1 + b_2 - c_2 + a_3$ , one gets that

(6.31) 
$$Z_{\rm b}^{(N)}(X) = \int_{\mathbb{A}_N+i0} \chi_{5_2}^{(N)}(x,\lambda) \mathrm{d}x, \qquad \chi_{5_2}^{(N)}(x,\lambda) = \chi_{5_2}^{(N)}(x)e^{2\pi i c_{\rm b}\lambda x}$$
  
(6.32)  $\chi_{5_2}^{(N)}(x) = \int \frac{\overline{\langle x \rangle} \langle z \rangle}{\sqrt{(1-1)^2}} \mathrm{d}z$ 

(6.32) 
$$\chi_{5_2}^{(N)}(x) = \int_{\mathbb{A}_N - i0} \frac{\varphi_{\mathbf{b}}(z+x)\varphi_{\mathbf{b}}(z)\varphi_{\mathbf{b}}(z-x)}{\varphi_{\mathbf{b}}(z+x)\varphi_{\mathbf{b}}(z)\varphi_{\mathbf{b}}(z-x)} dx$$

#### 6.4.3 H-Triangulations

In this section we will look at one vertex H-triangulations of knots.

Let X be an H–Triangulation for the figure–eight knot, i.e. let X be given by the diagram

	4
$\supset$	<

where the figure-eight knot is represented by the edge of the central tetrahedron connecting the maximal and the next to maximal vertices. If we choosing central tetrahedron  $(T_0)$  to be positive, the left tetrahedron  $(T_+)$  will be positive and the right one  $(T_-)$  negative. The shape structure, in the limit  $a_0 \rightarrow 0$  satisfies  $2b_+ + c_+ = 2b_- + c_- =: \lambda$  The partition function satisfies the following limit formula

(6.33) 
$$\lim_{a_0 \to 0} \varphi_{\rm b}(c_{\rm b}a_0 - c_{\rm b}/\sqrt{N}) Z_{\rm b}^{(N)}(X) = \frac{e^{-\pi i N/12}}{\nu(c_0)} \chi_{4_1}^{(N)}(0)$$

Similarly let X be represented by the diagram

that is, the H-triangulation for the  $5_2$  knot. We denote  $T_0, T_1, T_2, T_3$  the central, left, right, and top tetrahedra respectively and we choose the orientation so that the central tetrahedron  $T_0$  is negative then all other tetrahedra are positive. The edge representing the knot  $5_2$  connects the last two edges of  $T_0$ , so that the weight on the knot is given by  $2\pi a_0$ . In the limit  $a_0 \to 0$ , all edges, except for the knot, become balanced under the conditions

$$a_1 = c_2 = a_3, \quad b_3 = c_1 + b_2,$$

which in particular imply (6.30). The partition function has the following expression

(6.34) 
$$Z_{\rm b}^{(N)}(X) = \Theta \frac{e^{-\pi i N/12}}{\varphi_{\rm b}(c_{\rm b}a - c_{\rm b}\sqrt{N})} \chi_{5_2}^{(N)}(c_{\rm b}(a_1 - a_3))$$

For some constant phase factor  $\Theta$ .

## 6.4.4 Asymptotics of $\chi_{4_1}^{(N)}(0)$

In this section we want to study the asymptotic behavior of the invariant of the figure–8 knot

$$\chi_{4_1}^{(N)}(0) = \int_{\mathbb{A}_N} \mathcal{D}_{\mathbf{b}}(-x, -k) \overline{\mathcal{D}_{\mathbf{b}}(x, -k)} \mathbf{d}(x, k)$$
$$= \frac{1}{2\pi \mathbf{b}\sqrt{N}} \sum_{k \in \mathbb{Z}/N\mathbb{Z}} \int_{\mathbb{R}^{-id}} \mathcal{D}_{\mathbf{b}}\left(\frac{-x}{2\pi \mathbf{b}}, -k\right) \overline{\mathcal{D}_{\mathbf{b}}\left(\frac{x}{2\pi \mathbf{b}}, -k\right)} \mathbf{d}x$$

132

when  $b \rightarrow 0$ . The analysis uses techniques similar to the one presented in [AK1] for N = 1, however higher level gives new informations that we will show here. The integration in the complex plane is a contour integral where d > 0 so that the integral is absolutely convergent. By means of the asymptotic formula for the quantum dilogarithm (3.22) we have that

$$\chi_{4_1}^{(N)}(0) = \frac{1}{2\pi b\sqrt{N}} \sum_{k \in \mathbb{Z}/N\mathbb{Z}} \int_{\mathbb{R}^{-id}} \operatorname{Exp}\left[\frac{\operatorname{Li}_2(-e^{-\sqrt{N}x}) - \operatorname{Li}_2(-e^{\sqrt{N}x})}{2\pi i b^2 N}\right] \\ \times \phi_{-x}(k) \overline{\phi_x(k)} (1 + \mathcal{O}(b^2)) \mathrm{d}x$$

We want to apply the steepest descent method to this integral to get an asymptotic formula for  $b \rightarrow 0$ . First we show the computation for the exact integral,

(6.35) 
$$\frac{1}{2\pi \mathrm{b}\sqrt{N}} \sum_{k \in \mathbb{Z}/N\mathbb{Z}} \int_{\mathbb{R}^{-id}} \mathrm{Exp}\left[\frac{\mathrm{Li}_{2}(-e^{-\sqrt{N}x}) - \mathrm{Li}_{2}(-e^{\sqrt{N}x})}{2\pi i \mathrm{b}^{2}N}\right] \phi_{-x}(k) \overline{\phi_{x}(k)} \mathrm{d}x$$

and then we will argue that the former one can be approximated by the latter when  $b \rightarrow 0$ .

Let  $h(x) := \text{Li}_2(-e^{-\sqrt{N}x}) - \text{Li}_2(-e^{\sqrt{N}x})$ . Its critical points are solutions to

$$\begin{cases} h'(x) = 0\\ h''(x) \neq 0 \end{cases}$$

which are  $S = \left\{ \pm \frac{2}{3} \frac{\pi i}{\sqrt{N}} + \frac{2\pi i k}{\sqrt{N}} : k \in \mathbb{Z} \right\}$ . We compute the value of  $\operatorname{Im} h$  at its critical points to be

(6.36) 
$$\operatorname{Im} h\left(\pm \frac{2}{3}\frac{\pi i}{\sqrt{N}} + \frac{2\pi i k}{\sqrt{N}}\right) = \pm 4\Lambda(\frac{\pi}{6})$$

where  $\Lambda$  is the Lobachevsky function

(6.37) 
$$\Lambda(\alpha) = -\int_0^\alpha \log|2\sin\varphi|\,\mathrm{d}\varphi$$

and we refer the reader to [Kir] for the expressions that relate Lobachevsky function to the classical dilogarithm.

We only remark that  $4\Lambda(\frac{\pi}{6}) = \text{Vol}(4_1)$ , where by  $\text{Vol}(4_1)$  we mean the hyper-

bolic volume of knot complement  $S^3 \setminus (4_1)$ . Fix  $\mathbb{C} \ni x_0 = -\frac{2}{3} \frac{\pi i}{\sqrt{N}}$ , which is accessible from the original contour without passing through other critical points, and consider the contour

$$\mathcal{C} = \{ z \in \mathbb{C} : \operatorname{Re}(h(z)) = \operatorname{Re}(h(x_0)), \operatorname{Im}(h(z)) \le \operatorname{Im}(h(x_0)) \}$$

which is asymptotic to  $\operatorname{Re}(z) + \operatorname{Im}(z) = 0$  for  $\operatorname{Re}(z) \to \infty$  and to  $\operatorname{Re}(z) - \operatorname{Im}(z) = 0$  for  $\operatorname{Re}(z) \to -\infty$ . Moreover

(6.38) 
$$\lim_{\operatorname{Re}(z)\to\pm\infty}\operatorname{Im}(h(z)) = \lim_{\operatorname{Re}(z)\to\pm\infty}\pm\operatorname{Re}(z)\operatorname{Im}(z) = -\infty$$

All together we have found a contour C along which the integral (6.35) can be computed with the steepest descent method (see [Won]), giving as the following approximation for  $b \to 0$ 

(6.39) 
$$e^{\frac{h(x_0)}{2\pi i b^2 N}} \frac{g_{4_1}\left(-\frac{2}{3}\frac{\pi i}{\sqrt{N}}\right)}{\sqrt{i N^{-1} h''(x_0)}} (1 + \mathcal{O}(b^2)$$

where

$$g_{4_1}(x) := \frac{1}{\sqrt{N}} \sum_{k=0}^{N-1} \phi_{-x}(k) \overline{\phi}_x(k).$$

We now go back to  $\chi_{4_1}^{(N)}(0)$ , and we write it as the following integral

(6.40) 
$$\chi_{4_1}^{(N)}(0) = \frac{1}{2\pi b\sqrt{N}} \sum_{k \in \mathbb{Z}/N\mathbb{Z}} \int_{\mathbb{R}^{-id}} f_b(x,k) d(x,k)$$

where

(6.41) 
$$f_{\rm b}(x,k) = \mathcal{D}_{\rm b}\left(\frac{-x}{2\pi {\rm b}},-k\right)\overline{\mathcal{D}_{\rm b}\left(\frac{x}{2\pi {\rm b}},-k\right)}.$$

Then consider the contour

(6.42) 
$$C_{\rm b} = \{ z \in \mathbb{C} : \arg f_{\rm b}(z) = \arg f_{\rm b}(z_{\rm b}), \ |f_{\rm b}(z)| = |f_{\rm b}(z_{\rm b})| \}$$

where  $z_{\rm b}$  is defined as the solution to

(6.43) 
$$\frac{\partial}{\partial x} \log f_{\rm b}(x) = 0$$

which minimize the absolute value of  $f_b$ . Using the asymptotic formula for  $f_b$  it is simple to show that the contours  $C_b$  approximates C as  $b \to 0$  and that the points  $z_b$ 's will converge to  $x_0$ . So, in the limit  $b \to 0$ , the integral (6.40) is approximated by the integral (6.35), for which we already have an asymptotic formula. We have proved the following

(6.44) 
$$\chi_{4_1}^{(N)}(0) = e^{\frac{h(x_0)}{2\pi i b^2 N}} \frac{g_{4_1}\left(-\frac{2}{3}\frac{\pi i}{\sqrt{N}}\right)}{\sqrt{iN^{-1}h''(x_0)}} (1 + \mathcal{O}(b^2)),$$

As we remarked above Im  $h(x_0) = -\operatorname{Vol}(4_1)$ . Next we look at the number  $g_{4_1}\left(-\frac{2}{3}\frac{\pi i}{\sqrt{N}}\right)$  which is a topological invariant of the knot in the formula above. We have that

$$\begin{split} \sqrt{N}g_{4_1}\left(-\frac{2}{3}\frac{\pi i}{\sqrt{N}}\right) &= \sum_{k=1}^N \phi_{\frac{2}{3}\frac{\pi i}{\sqrt{N}}}(k)\overline{\phi}_{-\frac{2}{3}\frac{\pi i}{\sqrt{N}}}(k) \\ &= \left|\prod_{j=1}^{N-1} \left(1 - e^{-\frac{\pi i}{3N}}e^{-\frac{2\pi i j}{N}}\right)^{\frac{j}{N}}\right| \sum_{k=0}^{N-1} \prod_{j=1}^k \frac{1}{\left|1 - e^{\frac{1}{3}\frac{-\pi i}{N}}e^{\frac{2\pi i j}{N}}\right|^2} \end{split}$$

The last expression allows us to make the following remark

(6.45) 
$$g_{4_1}\left(-\frac{2}{3}\frac{\pi i}{\sqrt{N}}\right) = \gamma_N \mathcal{H}_N^0(\overline{\rho_{comp}})$$

where  $H_N^0(\overline{\rho_{comp}})$  is the Baseilhac–Benedetti invariant for the figure–eight knot found in [BB], computed at the conjugate of the complete hyperbolic structure (meaning that the holonomies of the structure are all complex conjugated) and  $\gamma_N$  is a global rescaling given by

(6.46) 
$$\gamma_N = \left| \prod_{j=1}^{N-1} \left( 1 - e^{-\frac{2\pi i j}{N}} \right)^{\frac{j}{N}} \right|$$

**Remark 6.29.** The very same steps of the previous asymptotic computation for  $\chi_{4_1}^{(N)}(0)$  can be applied to  $\chi_{5_2}^{(N)}(0)$  up to the point of having an expression

(6.47) 
$$\chi_{5_2}^{(N)}(0) = e^{\frac{\phi(x_{5_2})}{2\pi i b^2 N}} \frac{g_{5_1}(x_{5_2})}{\sqrt{iN^{-1}h_{5_2}''(x_{5_2})}} (1 + \mathcal{O}(b^2)),$$

where  $x_{5_2}$  is the only critical point in the complex plane that contributes to the steepest descent and

(6.48) 
$$g_{5_1}(x) = \frac{1}{\sqrt{N}} \sum_{j=0}^{N-1} \overline{\phi_{-x}}(j) \phi_x(j) \overline{\phi_{-x}}(j).$$

The fact that  $\text{Im }\phi(x_{5_2}) = -\text{Vol}(5_2)$ , can be seen directly, see for example [AK1]. However this situation is already too complicated to allow us to check relations with other theories. The obvious guess is to look for the Baseilhac–Benedetti invariant, but no explicitly computed examples, other then  $4_1$ , are known to the authors.

The following conjecture was originally stated in [AK1] for N = 1. Here we restate it in the updated setting.

**Conjecture 6.30** ([AK1]). Let M be a closed oriented compact 3-manifold. For any hyperbolic knot  $K \subset M$ , there exist a two parameters (b, N) family of smooth functions  $J_{M,K}^{(b,N)}(x, j)$  on  $\mathbb{R} \times \mathbb{Z}/N\mathbb{Z}$  which has the following properties.

 For any fully balanced shaped ideal triangulation X of the complement of K in M, there exist a gauge invariant real linear combination of dihedral angles λ, a (gauge non-invariant) real quadratic polynomial of dihedral angles φ such that

$$Z_{\rm b}^{(N)}(X) = e^{ic_{\rm b}^2\phi} \frac{1}{\sqrt{N}} \sum_{j=0}^{N-1} \int_{\mathbb{R}} J_{M,K}^{({\rm b},N)}(x,j) e^{ic_{\rm b}x\lambda} \mathrm{d}x$$

2. For any one vertex shaped H-triangulation Y of the pair (M, K) there exists a real quadratic polynomial of dihedral angles  $\varphi$  such that

$$\lim_{\omega_Y \to \tau} \mathcal{D}_{\mathbf{b}} \left( c_{\mathbf{b}} \frac{\omega_Y(K) - \pi}{\pi \sqrt{N}}, 0 \right) Z_{\mathbf{b}}^{(N)}(Y) = e^{ic_{\mathbf{b}}^2 \varphi - i\frac{\pi N}{12}} J_{M,K}^{(\mathbf{b},N)}(0,0),$$

where  $\tau : \Delta_1(Y) \to \mathbb{R}$  takes the value 0 on the knot K and the value  $2\pi$  on all other edges.

3. The hyperbolic volume of the complement of K in M is recovered as the following limit

$$\lim_{\mathbf{b}\to 0} 2\pi \mathbf{b}^2 N \log |J_{M,K}^{(\mathbf{b},N)}(0,0)| = -\operatorname{Vol}(M \setminus K)$$

**Remark 6.31.** We have proved this extended conjecture for the knots  $(S^3, 4_1)$ and  $(S^3, 5_2)$ , see formulas (6.45), (6.47) and (6.48). Moreover we gave a more explicit expansion, showing the appearance of an extra interesting therm  $g_K$ , and a precise relation between  $g_{4_1}$  and a known invariant of hyperbolic knots, defined by Baseilhac–Benedetti in [BB], see equation (6.45). We could have been more bold and extend the conjecture declaring the appearance of  $g_{(M,K)}$  to be general, and it to be proportional to the Baseilhac–Benedetti invariant. However we feel that there are not enough evidences to state it as general conjecture.

# Appendices

## A Tempered Distributions

For standard references for the topics of this appendix see e.g. [Hör2, Hör1] and [RS1, RS2].

**Definition A.1.** The Schwartz space  $\mathcal{S}(\mathbb{R}^n)$  is the space of all the functions  $\phi \in C^{\infty}(\mathbb{R}^n, \mathbb{C})$  such that

$$||\phi||_{\alpha,\beta} := \sup_{x \in \mathbb{R}^n} |x^{\beta} \partial^{\alpha} \phi(x)| < \infty$$

for all multi-indeces  $\alpha$ ,  $\beta$ .

The space of Tempered Distributions  $\mathcal{S}'(\mathbb{R}^n)$  is the space of linear functionals on  $\mathcal{S}(\mathbb{R}^n)$  which are continuous with respect to all these semi-norms.

Both these spaces are stable under the action of the Fourier transform  $\mathcal{F}$  and we use the notation  $\hat{u} = \mathcal{F}(u)$ . Let  $Z_n$  be the zero section of  $T^*(\mathbb{R}^n)$ .

**Definition A.2.** For a temperate distribution  $u \in S'(\mathbb{R}^n)$ , we define its *Wave Front Set* to be the following subset of the cotangent bundle of  $\mathbb{R}^n$ 

$$WF(u) = \{(x,\xi) \in T^*(\mathbb{R}^n) - Z_{\mathbb{R}^n} | \xi \in \Sigma_x(u)\}$$

where

$$\Sigma_x(u) = \bigcap_{\phi \in C^\infty_x(\mathbb{R}^n)} \Sigma(\phi u).$$

Here

$$C_x^{\infty}(\mathbb{R}^n) = \{ \phi \in C_0^{\infty}(\mathbb{R}^n) | \phi(x) \neq 0 \}$$

and  $\Sigma(v)$  are all  $\eta \in \mathbb{R}^n - \{0\}$  having no conic neighborhood V such that

$$|\hat{v}(\xi)| \le C_N (1+|\xi|)^{-N}, \ N \in \mathbb{Z}_{>0}, \ \xi \in V.$$

**Lemma A.3.** Suppose u is a bounded density on a  $C^{\infty}$  sub-manifold Y of  $\mathbb{R}^n$ , then  $u \in \mathcal{S}'(\mathbb{R}^n)$  and

$$WF(u) = \{(x,\xi) \in T^*(\mathbb{R}^n) | x \in \operatorname{Supp} u, \ \xi \neq 0 \ and \ \xi(T_xY) = 0\}.$$

In particular if  $\operatorname{Supp} u = Y$ , then we see that  $\operatorname{WF}(u)$  is the co-normal bundle of Y.

**Definition A.4.** Let u and v be temperate distributions on  $\mathbb{R}^n$ . Then we define

$$WF(u) \oplus WF(v) = \{ (x, \xi_1 + \xi_2) \in T^*(\mathbb{R}^n) | (x, \xi_1) \in WF(u), (x, \xi_2) \in WF(v) \}.$$

**Theorem A.5.** Let u and v be temperate distributions on  $\mathbb{R}^n$ . If

$$WF(u) \oplus WF(v) \cap Z_n = \emptyset,$$

then the product of u and v exists and  $uv \in \mathcal{S}'(\mathbb{R}^n)$ .

**Definition A.6.** We denote by  $\mathcal{S}(\mathbb{R}^n)_m$  the set of all  $\phi \in C^{\infty}(\mathbb{R}^n)$  such that

$$\sup_{x \in \mathbb{R}^n} |x^\beta \partial^\alpha(\phi)(x)| < \infty$$

for all multi-indices  $\alpha$  and  $\beta$  such that if  $\alpha_i = 0$  then  $\beta_i = 0$  for  $n - m < i \leq n$ . We define  $\mathcal{S}'(\mathbb{R}^n)_m$  to be the continuous dual of  $\mathcal{S}(\mathbb{R}^n)_m$  with respect to these semi-norms.

138

We observe that if  $\pi : \mathbb{R}^n \longrightarrow \mathbb{R}^{n-m}$  is the projection onto the first n-m coordinates, then  $\pi^*(\mathcal{S}(\mathbb{R}^{n-m})) \subset \mathcal{S}(\mathbb{R}^n)_m$ . This means we have a well defined push forward map

$$\pi_*: \mathcal{S}'(\mathbb{R}^n)_m \longrightarrow \mathcal{S}'(\mathbb{R}^{n-m}).$$

**Proposition A.7.** Suppose Y is a linear subspace in  $\mathbb{R}^n$ , u a density on Y with exponential decay in all directions in Y. Suppose  $\pi : \mathbb{R}^n \longrightarrow \mathbb{R}^m$  is a projection for some m < n. Then  $u \in \mathcal{S}'(\mathbb{R}^n)_m$  and  $\pi_*(u)$  is a density on  $\pi(Y)$  with exponential decay in all directions of the subspace  $\pi(Y) \subset \mathbb{R}^m$ .

Tempered distributions can be thought of as functions of growth at most polynomial, thanks to the following

**Theorem A.8.** Let  $T \in \mathcal{S}'(\mathbb{R}^n)$ , then  $T = \partial^{\beta}g$  for some polynomially bounded continuous function g and some multi-index  $\beta$ . That is, for  $f \in \mathcal{S}(\mathbb{R}^n)$ ,

$$T(f) = \int_{\mathbb{R}^n} (-1)^{|\beta|} g(x) (\partial^{\beta} f)(x) \mathrm{d}x$$

In particular it is possible to show that  $\mathcal{S}(\mathbb{R}^n) \subset \mathcal{S}'(\mathbb{R}^n)$ , where  $\mathcal{S}(\mathbb{R}^n) \ni f \mapsto T_f \in \mathcal{S}'(\mathbb{R}^n)$  with  $T_f(g) = \int_{\mathbb{R}^n} f(x)g(x)dx$ .

Denoting by  $\mathcal{L}(\mathcal{S}(\mathbb{R}^n), \mathcal{S}^{i}(\mathbb{R}^m))$  the space of continuous linear maps from  $\mathcal{S}(\mathbb{R}^n)$  to  $\mathcal{S}'(\mathbb{R}^m)$ , we remark that we have an isomorphism

(A.1) 
$$\widetilde{\cdot} : \mathcal{L}(\mathcal{S}(\mathbb{R}^n), \mathcal{S}'(\mathbb{R}^m)) \to \mathcal{S}'(\mathbb{R}^{n \sqcup m})$$

determined by the formula

(A.2) 
$$\varphi(f)(g) = \tilde{\varphi}(f \otimes g)$$

for all  $\varphi \in \mathcal{L}(\mathcal{S}(\mathbb{R}^n), \mathcal{S}'(\mathbb{R}^m))$ ,  $f \in \mathcal{S}(\mathbb{R}^n)$ , and  $g \in \mathcal{S}(\mathbb{R}^m)$ . This is the content of the Nuclear theorem, see e.g. [RS2]. Since we can not freely multiply distributions we end up with a categorid instead of a category. The partially defined composition in this categorid is defined as follows. Let n, m, l be three finite sets and  $A \in \mathcal{S}'(\mathbb{R}^{n \sqcup m})$  and  $B \in \mathcal{S}'(\mathbb{R}^{m \sqcup l})$ . We have pull back maps

$$\pi_{n,m}^*:\mathcal{S}'(\mathbb{R}^{n\sqcup m})\to \mathcal{S}'(\mathbb{R}^{n\sqcup m\sqcup l}) \text{ and } \pi_{m,l}^*:\mathcal{S}'(\mathbb{R}^{m\sqcup l})\to \mathcal{S}'(\mathbb{R}^{n\sqcup m\sqcup l})$$

By what we summarised above, the product

$$\pi_{n,m}^*(A)\pi_{m,l}^*(B) \in \mathcal{S}'(\mathbb{R}^{n \sqcup m \sqcup l})$$

is well defined provided the wave front sets of  $\pi^*_{n,m}(A)$  and  $\pi^*_{m,l}(B)$  satisfy the condition

(A.3) 
$$(WF(\pi_{n,m}^*(A)) \oplus WF(\pi_{m,l}^*(B))) \cap Z_{n \sqcup m \sqcup l} = \emptyset$$

If we now further assume that  $\pi_{n,m}^*(A)\pi_{n,l}^*(B)$  continuously extends to  $\mathcal{S}(\mathbb{R}^{n\sqcup m\sqcup l})_m$ , then we obtain a well defined element

$$(\pi_{n,l})_*(\pi_{n,m}^*(A)\pi_{m,l}^*(B)) \in \mathcal{S}'(\mathbb{R}^{n \sqcup l}).$$
### A.1 Bra-Ket Notation

We often use the Bra-Ket notation to make computations with distributions. For  $\varphi \in \mathcal{S}'(\mathbb{R}^n)$  a density and  $x \in \mathbb{R}^n$  we will write

$$\langle x|\varphi\rangle := \varphi(x),$$

with distributional meaning

$$\varphi(f) = \int_{\mathbb{R}^n} \langle x | \varphi \rangle f(x) dx = \int_{\mathbb{R}^n} \varphi(x) f(x) dx$$

In particular if  $\varphi \in \mathcal{S}(\mathbb{R}^n) \subset \mathcal{S}'(\mathbb{R}^n)$ , then

$$\langle x|\varphi\rangle = \varphi(x) = \delta_x(\varphi)$$

The integral kernel of the operator  $\mathsf{T}$ , if it exists, is a distribution  $k_{\mathsf{T}}$  such that

(A.4) 
$$\mathsf{T}(\psi)(x) = \int_{\mathbb{R}^n} k_\mathsf{T}(x, y) \psi(y) \mathrm{d}y$$

Working with Schwartz functions, the nuclear theorem expressed by formula (A.2) guarantees that the kernel  $k_T$  exists and that it is a tempered distribution. We will usually write the kernel from equation (A.4), in Bra-Ket notation as follows

(A.5) 
$$\mathsf{T}(\psi)(x) = \int_{\mathbb{R}^n} \langle x | \mathsf{T} | y \rangle \psi(y) \mathrm{d} y$$

and the nuclear theorem morphism (A.2) can be read as

(A.6) 
$$\langle x|\mathsf{T}|y\rangle = \langle x, y|T\rangle.$$

# A.2 $L^2(\mathbb{A}_N)$ and $\mathcal{S}(\mathbb{A}_N)$

 $\mathbb{A}_N \equiv \mathbb{R} \times (\mathbb{Z}/N\mathbb{Z})$  has the structure of a locally compact abelian group, with the normalized Haar measure d(x, n) defined by

$$\int_{\mathbb{A}_N} f(x,n) \mathrm{d}(x,n) := \frac{1}{\sqrt{N}} \sum_{n \in \mathbb{Z}/N\mathbb{Z}} \int_{\mathbb{R}} f(x,n) \mathrm{d}x$$

where  $f : \mathbb{A}_N \longrightarrow \mathbb{C}$  is an integrable function. By definition  $L^2(\mathbb{A}_N)$  is the space of functions  $f : \mathbb{A}_N \longrightarrow \mathbb{C}$  such that

(A.7) 
$$\int_{\mathbb{A}_N} |\mathbf{f}(a)|^2 \, \mathrm{d}a \equiv \frac{1}{\sqrt{N}} \sum_{n=0}^{N-1} \int_{\mathbb{R}} |\mathbf{f}(x,n)|^2 \, \mathrm{d}x < \infty$$

with standard inner product

(A.8) 
$$\langle \mathbf{f}, \mathbf{g} \rangle \equiv \frac{1}{\sqrt{N}} \sum_{n=0}^{N-1} \int_{\mathbb{R}} \mathbf{f}(x, n) \overline{\mathbf{g}(x, n)} \mathbf{d}(x, n)$$

Finite square integrable sequences are just a finite dimensional vector space

$$\mathrm{L}^2(\mathbb{Z}/N\mathbb{Z})\simeq\mathbb{C}^N,$$

with a preferred basis given by mod N Kronecker delta functions

(A.9) 
$$\delta_j(n) \equiv \begin{cases} 1 & \text{if } j = n \mod N \\ 0 & \text{otherwise} \end{cases}$$

There is a natural isomorphism

(A.10) 
$$L^2(\mathbb{R}) \otimes L^2(\mathbb{Z}/N\mathbb{Z}) \simeq L^2(\mathbb{A}_N)$$

defined by

(A.11) 
$$f \otimes \delta_j(a) = f(x)\delta_j(n), \text{ for } a = (x, n) \in \mathbb{A}_N$$

with inverse

(A.12) 
$$\mathbb{A}_N \ni \mathsf{f} \mapsto \sum_{j=0}^{N-1} \mathsf{f}(\cdot, j) \otimes \delta_j \in \mathrm{L}^2(\mathbb{R}) \otimes \mathrm{L}^2(\mathbb{Z}/N\mathbb{Z})$$

Everything just said holds true substituting  $L^2$  with S, with the isomorphism  $S(\mathbb{A}_N) \simeq S(\mathbb{R}) \otimes \mathbb{C}^N$  and further also, the space of tempered distributions on  $\mathbb{A}_N$ , defined as linear continuous functionals over  $S(\mathbb{A}_N)$ , are simply  $S'(\mathbb{R}) \otimes \mathbb{C}^N$ . All the Bra-Ket notation extends trivially to  $S(\mathbb{A}_N)$ , including the nuclear theorem (A.6), substituting all the integrals over  $\mathbb{R}$  with integrals over  $\mathbb{A}_N$ .

We use a bracket notation for Fourier coefficients and Gaussian exponentials in  $\mathbb{A}_N$ , following the notation introduced in [AK3]

(A.13) 
$$\langle (x,n), (y,m) \rangle \equiv e^{2\pi i x y} e^{-2\pi i n m/N} \quad \langle (x,n) \rangle \equiv e^{\pi i x^2} e^{-\pi i n (n+N)/N}$$

For (x, n) and (y, m) in  $\mathbb{A}_N$ . The Fourier transform then takes the form

$$\mathcal{F}(f)(x,n) = \int_{\mathbb{A}_N} f(y,m) \langle (x,n), (y,m) \rangle \mathrm{d}(y,m)$$

For any operator A of order N, we can define the operator  $L_N(A)$  via the spectral theorem, such that it formally satisfies

$$A = e^{2\pi i \operatorname{L}_N(A)/N}.$$

140

We can define, for any function  $f : \mathbb{A}_N \longrightarrow \mathbb{C}$  the operator function  $f(\mathsf{x}, A) \equiv f(\mathsf{x}, L_N(A))$  for any commuting pair of operators  $\mathsf{x}$  and A, where the former is self adjoint and the latter is of order N. We have, for  $\mathsf{x}$  and A as above, that

(A.14) 
$$f(\mathbf{x}, A) = \int_{\mathbb{A}_N} \tilde{f}(y, m) e^{2\pi i y \mathbf{x}} A^{-m} \mathrm{d}(y, m)$$

where we use the following notation for the inverse Fourier transforms

(A.15) 
$$\tilde{f}(x,n) = \int_{\mathbb{A}_N} f(y,m) \overline{\langle (y,m); (x,n) \rangle} d(y,m).$$

# **B** Categroids

We need a notion which is slightly more general than categories to define the Teichmüller TQFT functor.

#### Definition B.1. [AK1]

A Categorid  $\mathcal{C}$  consist of a family of objects  $\operatorname{Obj}(\mathcal{C})$  and for any pair of objects A, B from  $\operatorname{Obj}(\mathcal{C})$  a set  $\operatorname{Mor}_{\mathcal{C}}(A, B)$  such that the following holds

**A** For any three objects A, B, C there is a subset  $K^{\mathcal{C}}_{A,B,C} \subset \operatorname{Mor}_{\mathcal{C}}(A, B) \times \operatorname{Mor}_{\mathcal{C}}(B, C)$ , called the composable morphisms and a *composition* map

$$\circ: K^{\mathcal{C}}_{A,B,C} \to \operatorname{Mor}_{\mathcal{C}}(A,C).$$

such that composition of composable morphisms is associative.

**B** For any object A we have an identity morphism  $1_A \in Mor_{\mathcal{C}}(A, A)$  which is composable with any morphism  $f \in Mor_{\mathcal{C}}(A, B)$  or  $g \in Mor_{\mathcal{C}}(B, A)$  and we have the equations

$$1_A \circ f = f$$
, and  $g \circ 1_A = g$ .

### References

- [A1] J. E. Andersen. Asymptotic faithfulness of the quantum SU(n) representations of the mapping class groups. Annals of Mathematics. 163:347–368, 2006.
- [A2] J. E. Andersen. The Nielsen-Thurston classification of mapping classes is determined by TQFT. J. Math. Kyoto Univ. 48(2):323–338, 2008.
- [A3] J. E. Andersen. The Witten-Reshetikhin-Turaev invariants of finite order mapping tori I. Journal f
  ür Reine und Angewandte Mathematik. 681:1–38, 2013.

- [A4] J. E. Andersen. Hitchin's connection, Toeplitz operators, and symmetry invariant deformation quantization. *Quantum Topol.* 3(3-4):293–325, 2012.
- [AG] J. E. Andersen and N. L. Gammelgaard. The Hitchin-Witten Connection and Complex Quantum Chern-Simons Theory. arXiv:1409.1035, 2014.
- [AGP] J. E. Andersen, S. Gukov, D. Pei. The Verlinde formula for Higgs bundles, arXiv:1608.01761, 2016.
- [AH] J. E. Andersen & B. Himpel. The Witten-Reshetikhin-Turaev invariants of finite order mapping tori II Quantum Topology. 3:377–421, 2012.
- [AHJMMc] J. E. Andersen, B. Himpel, S. F. Jørgensen, J. Martens, and B. McLellan. The Witten-Reshetikhin-Turaev invariant for links in finite order mapping tori I. Advances in Mathematics. 304:131–178, 2017.
- [AK1] J. E. Andersen and R. Kashaev. A TQFT from Quantum Teichmüller Theory. Comm. Math. Phys. 330(3):887–934, 2014.
- [AK2] J. E. Andersen and R. Kashaev. A new formulation of the Teichmüller TQFT. arXiv:1305.4291, 2013.
- [AK3] J. E. Andersen and R. Kashaev. Complex Quantum Chern-Simons. arXiv:1409.1208, 2014.
- [AN] J. E. Andersen and J.-J. K. Nissen, Asymptotic aspects of the Teichmüller TQFT, Travaux Mathématiques 25 (2017), 41–95, Preprint 2016.
- [AU1] J. E. Andersen & K. Ueno. Abelian Conformal Field theories and Determinant Bundles. *International Journal of Mathematics*. 18:919–993, 2007.
- [AU2] J. E. Andersen & K. Ueno, Constructing modular functors from conformal field theories. *Journal of Knot theory and its Ramifications*. 16(2):127–202, 2007.
- [AU3] J. E. Andersen & K. Ueno. Modular functors are determined by their genus zero data. *Quantum Topology*. 3:255–291, 2012.
- [AU4] J. E. Andersen & K. Ueno. Construction of the Witten-Reshetikhin-Turaev TQFT from conformal field theory. *Invent. Math.* 201(2):519–559, 2015.
- [ADW] S. Axelrod, S. Della Pietra, E. Witten. Geometric quantization of Chern Simons gauge theory. J.Diff.Geom. 33:787–902, 1991.
- [BB] S. Baseilhac and R. Benedetti. Quantum hyperbolic geometry. Algebr. Geom. Topol. 7:845–917, 2007.

- [B] C. Blanchet. Hecke algebras, modular categories and 3-manifolds quantum invariants. *Topology*. 39(1):193–223, 2000.
- [BHMV1] C. Blanchet, N. Habegger, G. Masbaum & P. Vogel. Three-manifold invariants derived from the Kauffman Bracket. *Topology*. 31:685–699, 1992.
- [BHMV2] C. Blanchet, N. Habegger, G. Masbaum & P. Vogel. Topological Quantum Field Theories derived from the Kauffman bracket. *Topology.* 34:883– 927, 1995.
- [BMS] M. Bordeman, E. Meinrenken & M. Schlichenmaier. Toeplitz quantization of Kähler manifolds and  $gl(N), N \to \infty$  limit Comm. Math. Phys. 165:281–296, 1994.
- [D] T. Dimofte, Complex Chern-Simons theory at level k via the 3d-3d correspondence. Comm. Math. Phys. 339(2):619–662, 2015.
- [F] L. D. Faddeev. Discrete Heisenberg-Weyl group and modular group. Lett. Math. Phys. 34(3):249–254, 1995.
- [FK] L. D. Faddeev and R. M. Kashaev. Quantum dilogarithm. Modern Phys. Lett. A. 9(5):427–434, 1994.
- [FKV] L. D. Faddeev, R. M. Kashaev, and A. Yu. Volkov. Strongly coupled quantum discrete Liouville theory. I. Algebraic approach and duality. *Comm. Math. Phys.* 219(1):199–219, 2001.
- [FG] V. Fock and A. Goncharov. Moduli spaces of local systems and higher Teichmüller theory. Publ. Math. Inst. Hautes Études Sci. 103:1–211, 2006.
- [FK] L. Funar and R. M. Kashaev. Centrally extended mapping class groups from quantum Teichmüller theory. Adv. Math. 252:260–291, 2014.
- [GKT] N. Geer, R. Kashaev, and V. Turaev. Tetrahedral forms in monoidal categories and 3-manifold invariants. J. Reine Angew. Math. 673:69–123, 2012.
- [Hik1] K. Hikami. Hyperbolicity of partition function and quantum gravity. Nuclear Phys. B. 616(3):537–548, 2001.
- [Hik2] K. Hikami. Generalized volume conjecture and the A-polynomials: the Neumann-Zagier potential function as a classical limit of the partition function. J. Geom. Phys. 57(9):1895–1940, 2007.
- [Hit1] N. J. Hitchin. The self-duality equations on a Riemann surface. Proc. London Math. Soc. 55(1):59–126, 1987.

- [Hit2] N. J. Hitchin. Flat connections and geometric quantization. Comm. Math. Phys. 131:347–380, 1990.
- [HRS] C. D. Hodgson, J. H. Rubinstein, and H. Segerman. Triangulations of hyperbolic 3-manifolds admitting strict angle structures. J. Topol. 5(4):887– 908, 2012.
- [Hör1] L. Hörmander. Linear partial differential operators. Third revised printing. Die Grundlehren der mathematischen Wissenschaften, Band 116. Springer-Verlag New York Inc., New York, 1969.
- [Hör2] L. Hörmander. The analysis of linear partial differential operators. I, Grundlehren der Mathematischen Wissenschaften [Fundamental Principles of Mathematical Sciences], 256. Springer-Verlag, Berlin, second edition, 1990.
- [KS] A. V. Karabegov & M. Schlichenmaier. Identification of Berezin-Toeplitz deformation quantization J. Reine Angew. Math. 540:49–76, 2001.
- [K1] R. M. Kashaev. Quantum dilogarithm as a 6j-symbol. Modern Phys. Lett. A. 9(40):3757–3768, 1994.
- [K2] R. M. Kashaev. The hyperbolic volume of knots from the quantum dilogarithm. Lett. Math. Phys. 39(3):269–275, 1997.
- [K3] R. M. Kashaev. Quantization of Teichmüller spaces and the quantum dilogarithm. Lett. Math. Phys. 43(2):105–115, 1998.
- [K4] R. Kashaev. The quantum dilogarithm and Dehn twists in quantum Teichmüller theory. In Integrable structures of exactly solvable two-dimensional models of quantum field theory. NATO Sci. Ser. II Math. Phys. Chem. 35:211–221. Kluwer Acad. Publ., Dordrecht, 2001.
- [K5] R. M. Kashaev. Coordinates for the moduli space of flat PSL(2, ℝ) connections. Math. Research Letters. 12:23–36, 2005.
- [K6] R. M. Kashaev. Discrete Liouville equation and Teichmüller theory. In Handbook of Teichmüller theory. Volume III, IRMA Lect. Math. Theor. Phys. 17:821–851. Eur. Math. Soc., Zürich, 2012.
- [Kir] A. N. Kirillov. Dilogarithm identities. Progress of Theoretical Physics Supplement. 118:61–142, 1995.
- [Lac] M. Lackenby. Word hyperbolic Dehn surgery. Invent. Math. 140(2):243– 282, 2000.
- [Las] Y. Laszlo. Hitchin's and WZW connections are the same. J. Diff. Geom. 49(3):547–576, 1998.

- [LT] F. Luo and S. Tillmann. Angle structures and normal surfaces. Trans. Amer. Math. Soc. 360(6):2849–2866, 2008.
- [M] S. Marzioni. Complex Chern-Simons Theory: Knot Invariants and Mapping Class Group Representations. *PhD Thesis*, Aarhus University, 2016.
- [MM] H. Murakami and J. Murakami. The colored Jones polynomials and the simplicial volume of a knot. Acta Math., 186(1):85–104, 2001.
- [NZ] W. D. Neumann and Don Zagier. Volumes of hyperbolic three-manifolds. *Topology*, 24(3):307–332, 1985.
- [P] R. C. Penner. Decorated Teichmüller theory. QGM Master Class Series. European Mathematical Society (EMS), Zürich. With a foreword by Yuri I. Manin, 2012.
- [RS1] M. Reed and B. Simon. Methods of modern mathematical physics. II. Fourier analysis, self-adjointness. Academic Press [Harcourt Brace Jovanovich, Publishers], New York-London, 1975.
- [RS2] M. Reed and B. Simon. Methods of modern mathematical physics. I. Academic Press, Inc. [Harcourt Brace Jovanovich, Publishers], New York, second edition, 1980.
- [RT1] N. Reshetikhin & V. Turaev. Ribbon graphs and their invariants derived from quantum groups *Comm. Math. Phys.* 127:1–26, 1990.
- [RT2] N. Reshetikhin & V. Turaev. Invariants of 3-manifolds via link polynomials and quantum groups *Invent. Math.* 103:547–597, 1991.
- [TUY] A. Tsuchiya, K. Ueno & Y. Yamada. Conformal Field Theory on Universal Family of Stable Curves with Gauge Symmetries Advanced Studies in Pure Mathematics. 19:459–566, 1989.
- [V] A. Y. Volkov. Noncommutative hypergeometry. Comm. Math. Phys. 258(2):257–273, 2005.
- [W1] E. Witten. Quantum field theory and the Jones polynomial. Comm. Math. Phys. 121:351–98, 1989.
- [W2] E. Witten. Quantization of Chern-Simons gauge theory with complex gauge group. Comm. Math. Phys. 137(1):29–66, 1991.
- [Won] R. Wong. Asymptotic approximations of integrals. Classics in Applied Mathematics, 34. Society for Industrial and Applied Mathematics (SIAM), Philadelphia, PA. Corrected reprint of the 1989 original, 2001.

Jørgen Ellegaard Andersen and Simone Marzioni, Center for Quantum Geometry of Moduli Spaces, Department of Mathematics, University of Aarhus, DK-8000, Denmark

146

Travaux mathématiques, Volume 25 (2017), 147-211, © by the authors

# Construction of Modular Functors from Modular Tensor Categories

by Jørgen Ellegaard Andersen and William Elbæk Petersen<sup>1</sup>

#### Abstract

In this paper we follow the constructions of Turaev's book, "Quantum invariants of knots and 3-manifolds" closely, but with small modifications, to construct a modular functor, in the sense of Kevin Walker, from any modular tensor category. We further show that this modular functor has duality and if the modular tensor category is unitary, then the resulting modular functor is also unitary. We further introduce the notion of a fundamental symplectic character for a modular tensor category. In the cases where such a character exists we show that compatibilities between the structures in a modular functor can be made strict in a certain sense. Finally we establish that the modular tensor categories which arise from quantum groups of simple Lie algebras all have natural fundamental symplectic characters.

# 1 Introduction

The axioms for a Topological Quantum Field Theory (TQFT) was proposed by Atiyah, Segal and Witten in the late 80'ties and further Witten proposed in his seminal paper [32], that quantum Chern-Simons gauge theory should provide examples of TQFT's. This was shortly thereafter demonstrated by Reshetikhin and Turaev in the fundamental papers [24, 25], where they used the representation theory of quantum groups to give a construction of these TQFT's for  $U_q(sl(2, \mathbb{C}))$ at a root of unity q, which is now known as the Witten-Reshetikhin-Turaev TQFT. This work further identified the needed categorical setup to construct a TQFT, which Turaev presented in his beautiful book [29], namely the notion of a Modular Tensor Category. Following these main events it was then shown that all quantum groups of simple Lie algebras at roots of unity gives examples of modular tensor categories (see e.g. the extensive reference list in the second edition of [29]). More topological constructions of these TQFT's, first for the  $U_q(sl(2, \mathbb{C}))$ -case were given in [17, 18] and then for the  $U_q(sl(n, \mathbb{C}))$ -case, the corresponding modular

<sup>&</sup>lt;sup>1</sup>Work supported in part by the center of excellence grant Center for Quantum Geometry of Moduli Spaces from the Danish National Research Foundation (DNRF95)

tensor category and its associated TQFT was given a topological construction in [16].

It was further conjectured by Witten in [32] that these TQFT's could also be constructed by *Conformal Field Theory* techniques. At the time there was a large body of work done on conformal field theory on the physics side and on the mathematical side as well, where we would like to highlight the works of Segal [26] and of Tsuchiya, Ueno and Yamada [27]. As it was later shown in [11, 12], the TUY construction of conformal field theory naturally leads to the construction of a Modular Functor in the sense of Kevin Walker [31]. It is well known that a modular functor in the sense of Walker also gives rise to a TQFT and that the TQFT is uniquely determined by the underlying modular functor (see e.g. [20]). The construction of a modular functor from a modular tensor category was provided by Turaev in his book [29]. He works however with similar, but not the same axioms for a modular functor, as Walker does. It is well-known to a broad range of researchers in the TQFT community that one can easily adapt the constructions in Turaev's book so as to provide the construction of a modular functor in the sense of Walker. Indeed, in this paper, we follow the construction from Turaev's book [29] very closely, and provide all details for how any modular tensor category  $\mathcal{V}$  gives rise to a modular functor  $Z_{\mathcal{V}}$  subject to the axioms formulated by Kevin Walker in [31] and used in [12, 13, 14]. The importance of the isomorphism provided in [11, 12, 13, 14] between the Witten-Reshetikhin-Turaev TQFT and the one coming from conformal field theory [27, 12] is that it provides a geometric construction of the WRT-TQFT's. When one further combines this isomorphism with Laszlo's isomorphism [23] between the covariant constant sections of the TUY connections with that of the Hitchin connection in the context of the geometric quantization of the moduli spaces of flat connections [21, 15, 19, 6, 7], one gets the full picture conjectured by Witten in [32]. This chain of isomorphisms has already been exploited by the first author of this paper, in part with collaborators, to obtain deep results about the asymptotics in terms of the above mentioned root of unity [1, 2, 3, 4, 5, 6, 8, 9, 10].

Let us now briefly review how one adjusts Turaev's construction of a modular functor so as to obtain one in the sense of Walker. A labeled marked surface is a closed oriented surface  $\Sigma$  endowed with a finite set of distinguished points equipped with a direction as well as a label from a finite set  $\Lambda$ . Moreover  $\Sigma$  is equipped with a Lagrangian subspace of its first homology group. A modular functor associates to any labeled marked surface  $\Sigma$  a module called its module of states. See section 4 below where we spell out Walker's axioms for a modular functor in all details.

Given a modular tensor category  $(\mathcal{V}, (V_i)_{i \in I})$ , Turaev constructs a 2-DMF in [29]. Taking the label set to be  $\Lambda = I$ , we simply use the modular functor provided by Turaev, and provide natural identifications between certain modules of states to make up for the differences between Turaev's axioms for a 2-DMF and Walker's

axioms for a 2-DMF. To do this one needs to fix isomorphisms

$$(1.1) q_i: V_{i^*} \to (V_i)^*$$

Existence of such a family of isomorphisms is guaranteed by the axioms of a modular tensor category as given in [29], but a specified choice is not part of these axioms. Indeed, some of the interesting results in this paper concerning duality are obtained by exploiting that these can be scaled. We first obtain the following result (Theorem 5.8 and 9.2).

**Theorem 1.1.** For any choice of the isomorphisms (1.1) we get a modular functor  $Z_{\mathcal{V}}$  satisfying Walker's axioms. For any two choices of the isomorphisms (1.1) we get quasi-isomorphic modular functors.

Here quasi-isomorphism refers to a notation which is exactly like isomorphism of modular functors, except that it allows for scalings of the glueing isomorphisms in a label dependent way, see Definition 9.1. Hence we see that there is a unique quasi-isomorphism class of modular functors associated to every modular tensor category. Two sets of isomorphisms  $q_i^{(j)} : V_{i^*} \to (V_i)^*, j = 1, 2$  give rise to two strictly isomorphic modular functors if the unique  $u_i \in K^*$  determined by  $q_i^{(2)} = u_i q_i^{(1)}$  satisfies that  $u_{i^*} = u_i$ .

**Remark 1.2.** We stress that for the rest of this paper, the term modular functor generally refers to Walker's axioms.

**Theorem 1.3.** For any choice of the isomorphisms (1.1) we get a duality structure on the modular functor  $Z_{\mathcal{V}}$ . If the modular tensor category is unitary, then we also get a unitary structure compatible with the rest of the structure of the modular functor.

This is the content of Theorem 11.4 and Theorem 13.1 below. We emphasize that we do not need to choose the same  $q_i$  for the glueing maps and for the duality, as discussed in section 14. Further, in the compatibility between glueing and duality, duality with itself and duality with the unitary structure, there are projective factors allowed, as detailed in the Definition 11.1 and Definition 12.1.

The existence of these projective factors in the compatibility between these structures naturally raises the question if one can actually normalise all quantities such that these projective factors disappear. Let us now address this question.

First we establish, that we can normalise the duality pairing and the unitary pairing, such that both are strictly compatible with glueing. This is done in section 14. From this scaling analysis, one sees that the scaling can be separated into a product of two factors, one which only depends on the genus of the surface (see Definition 14.1) and one, which is simply a product of contributions from each of the labels (see equation (14.12)). This provides us with what we call the *canonical symplectic scaling*, where (15.1) in Theorem 15.3 relate the two scalings

of the isomorphisms (1.1), which has the effect that the quantum invariant of the flat unknot labeled by *i* becomes  $\dim(V_i)$  (see equation (15.3), which is the corresponding normalisation for the unknot with one negative twist). The multiplicative factor in the compatibility of duality with duality and unitary pairing with duality becomes in this case negative one raised to the number of *symplectic self-dual labels* of a given labeled marked surface (see Definition 15.1 and 15.2 and Theorem 15.3 and 15.5).

In order to analyse if we can find a normalisation such that all projective factors in the compatibility between glueing and duality, duality with itself and duality with unitarity can be made unity, which we call *strict compatibility*, we introduce the *dual fundamental group*  $\Pi(\mathcal{V}, I)^*$  of a modular tensor category.

**Definition 1.4**  $(\Pi(\mathcal{V}, I)^*)$ . Let  $\Pi(\mathcal{V}, I)^*$  consist of the set of functions

 $\tilde{\mu}: I \to K^*$ 

that satisfies

and such that

 $\tilde{\mu}(i)\tilde{\mu}(j)\tilde{\mu}(k) \neq 1,$ 

 $\tilde{\mu}(i)\tilde{\mu}(i^*) = 1,$ 

implies

$$\operatorname{Hom}(\mathbf{1}, V_i \otimes V_j \otimes V_k) = \mathbf{0}.$$

We call it the dual of the fundamental group due to its similarity with the dual of the fundamental group of a simple Lie algebra as spelled out in section 18.

We make the following definition.

**Definition 1.5.** An element  $\tilde{\mu} \in \Pi(\mathcal{V}, I)^*$  with the property that  $\tilde{\mu}$  takes on the values  $\pm 1$  on the self-dual simple objects, in such way that  $\tilde{\mu}$  is -1 on the symplectic simple objects and 1 on the rest of the self-dual simple objects, is called a *fundamental symplectic character*.

We observe that if  $\mathcal{V}$  has no symplectic simple objects, then the identity in  $\Pi(\mathcal{V}, I)^*$  is a fundamental symplectic character. We ask the question if any modular tensor category has such a fundamental symplectic character.

**Theorem 1.6.** If  $\mathcal{V}$  has a fundamental symplectic character, then we can arrange that glueing and duality, duality with itself and duality with the unitary paring are strictly compatible.

This is proven in section 16. In section 17 we provide a fundamental symplectic character for the quantum SU(N) modular tensor category  $H_k^{SU(N)}$  at the root of unity  $q = e^{2\pi i/(k+N)}$  first constructed by Reshetikhin and Turaev for N = 2 [24, 25] and by Turaev and Wenzl for general N [28, 30]. See also [17, 18] for a skein theory

model of the N = 2 case and [16] for the general N. In section 18, we provide a fundamental symplectic character for any modular tensor category associated to the quantum group at a root of unity for any simple Lie algebra. Hence we have established

**Theorem 1.7.** Any quantum group at a root of unity gives a modular functor such that glueing and duality, duality with itself and duality with the unitary pairing are strictly compatible.

We thank Henning Haahr Andersen, Christian Blanchet, Jens Carsten Jantzen, Nicolai Reshetikhin and Vladimir Turaev for valuable discussion regarding this paper.

# 2 Axioms for a modular tensor category

For the axioms of a modular tensor category  $(\mathcal{V}, (V_i)_{i \in I})$  we refer to chapter II in [29]. For any modular tensor category, we have an induced involution  $^* : I \to I$ , determined by

$$(V_i)^* \cong V_{i^*}$$

Recall that the ground ring is K = End(1) in the notation of [29]. For an object V we have the important K-linear trace operation  $\text{tr} : \text{End}(V) \to K$ . We have the following definition  $\dim(V) := \text{tr}(\text{id}_V)$  and one gets the following identities for all objects V

$$\dim(V) = \dim(V^*),$$

We simply write  $\dim(V_i) = \dim(i)$  and so for all indices  $i \in I$ 

 $\dim(i) = \dim(i^*).$ 

# 3 Labeled marked surfaces, extended surfaces and marked surfaces

#### 3.1 A-Labeled marked surfaces

Let  $\Lambda$  be a finite set equipped with an involution  $^{\dagger} : \Lambda \to \Lambda$  and a preferred element  $0 \in \Lambda$  with  $0^{\dagger} = 0$ . We start by recalling that for a closed connected surface  $\Sigma$ , Poincare duality induces a non-degenerate skewsymmetric pairing

$$(\cdot, \cdot): H_1(\Sigma, \mathbb{Z}) \times H_1(\Sigma, \mathbb{Z}) \longrightarrow \mathbb{Z},$$

called the intersection pairing. For the rest of this paper,  $H_1(\Sigma)$  will mean the first integral homology group. We remark that we could just as well have considered  $H_1(\Sigma, \mathbb{R})$ . For any real vector space W, let  $P(W) := (W \setminus \{0\})/\mathbb{R}_+$ . We now define the objects of the category of  $\Lambda$ -labeled marked surfaces. **Definition 3.1** (A-marked surfaces). A A-marked surface is given by the following data:  $(\Sigma, P, V, \lambda, L)$ . Here  $\Sigma$  is a smooth oriented closed surface. P is a finite subset of  $\Sigma$ . We call the elements of P distinguished points of  $\Sigma$ . V assigns to any p in P an element  $v(p) \in P(T_p\Sigma)$ . We say that v(p) is the direction at p.  $\lambda$  is an assignment of labels from  $\Lambda$  to the points in P, e.g. it is a map  $P \to \Lambda$ . We say that  $\lambda(p)$  is the label of p. Assume  $\Sigma$  splits into connected components  $\{\Sigma_{\alpha}\}$ . L is a Lagrangian subspace of  $H_1(\Sigma)$  such that the natural splitting  $H_1(\Sigma) \simeq \bigoplus_{\alpha} H_1(\Sigma_{\alpha})$  induces a splitting  $L \simeq \bigoplus_{\alpha} L_{\alpha}$  where  $L_{\alpha} \subset H_1(\Sigma_{\alpha})$  is a Lagrangian subspace for each  $\alpha$ . By convention the empty set  $\emptyset$  is regarded as a  $\Lambda$ -labeled marked surface.

For the sake of brevity, we will refer to a  $\Lambda$ -labeled marked surface as a labeled marked surface, whenever there is no risk of ambiguities. Now we describe the morphisms of this category.

**Definition 3.2** (Morphisms). Let  $\Sigma_i$ , i = 1, 2 be two (non-empty)  $\Lambda$ -labeled marked surfaces. For i = 1, 2, write  $\Sigma_i = (\Sigma_i, P_i, V_i, \lambda_i, L_i)$ . A morphism is a pair  $\mathbf{f} = (f, s)$ , where s is an integer, and f is an equivalence class of orientation preserving diffeomorphisms  $\phi : \Sigma_1 \xrightarrow{\sim} \Sigma_2$  that restricts to a bijection of distinguished points  $P_1 \xrightarrow{\sim} P_2$  that preserves directions and labels. Two such diffeomorphisms  $\phi, \psi$  are said to be equivalent if they are related by an isotopy of such diffeomorphisms.

For a diffeomorphism such as  $\phi$ , we will write  $[\phi]$  for the equivalence class described above. Thus we will sometimes denote a morphism by ([f], s) if we want to stress that we are dealing with a pair where the isotopy class is the equivalence class of the diffeomorphism f. Let  $\sigma$  be Wall's signature cocycle for triples of Lagrangian subspaces. We now define composition.

**Definition 3.3** (Composition). Assume that we are given two composable morphisms  $\mathbf{f}_1 = (f_1, s_1) : \Sigma_1 \to \Sigma_2$  and  $\mathbf{f}_2 = (f_2, s_2) : \Sigma_2 \to \Sigma_3$ . We then define

$$\mathbf{f}_2 \circ \mathbf{f}_1 := (f_2 \circ f_1, s_2 + s_1 - \sigma((f_2 \circ f_1)_{\#}(L_1), (f_2)_{\#}(L_2), L_3)).$$

Using properties of Wall's signature cocycle we obtain that the composition operation is associative and therefore we obtain the category of  $\Lambda$ -labelled marked surfaces.

**Definition 3.4** (The category of  $\Lambda$ -labeled marked surfaces). The category  $\mathbf{C}(\Lambda)$  of  $\Lambda$ -labeled marked surfaces has  $\Lambda$ -labeled marked surfaces as objects and morphisms as described in definition 3.2 and composition as described in definition 3.3.

There is an easy way to make this category into a symmetric monoidal category.

**Definition 3.5** (The operation of disjoint union). Let  $\Sigma_1, \Sigma_2$  be two  $\Lambda$ -labeled marked surfaces. For i = 1, 2, write  $\Sigma_i = (\Sigma_i, P_i, V_i, \lambda_i, L_i)$ . We define their disjoint union  $\Sigma_1 \sqcup \Sigma_2$  to be

$$(\Sigma_1 \sqcup \Sigma_2, P_1 \sqcup P_2, V_1 \sqcup V_2, \lambda_1 \sqcup \lambda_2, L_1 \oplus L_2).$$

For morphisms  $\mathbf{f}_i : \Sigma_i \to \Sigma_3$  we define  $\mathbf{f_1} \sqcup \mathbf{f_2}$  to be

$$(f_1 \sqcup f_2, s_1 + s_2).$$

We have an obvious natural transformation

$$\operatorname{Perm}: \Sigma_1 \sqcup \Sigma_2 \to \Sigma_2 \sqcup \Sigma_1.$$

**Proposition 3.6** ( $\mathbf{C}(\Lambda)$  is a symmetric monoidal category). The category of  $\Lambda$ -labeled marked surfaces is a symmetric monoidal category with disjoint union as product, the empty surface as unit, and Perm as the braiding.

We now describe the operation of orientation reversal. For an oriented surface  $\Sigma$  we let  $-\Sigma$  be the oriented surface where we reverse the orientation on each component. For a map g with values in  $\Lambda$  we let  $g^{\dagger}$  be the map, with the same domain and codomain, given by  $g^{\dagger}(x) = g(x)^{\dagger}$ .

**Definition 3.7** (Orientation reversal). Let  $\Sigma = (\Sigma, P, V, \lambda, L)$  be a  $\Lambda$ -labeled marked surface. Then we define

$$-\Sigma := (-\Sigma, P, V, \lambda^{\dagger}, L).$$

We say that  $-\Sigma$  is obtained form  $\Sigma$  by reversal of orientation. For a morphism  $\mathbf{f} = (f, s)$  we let

$$-\mathbf{f} := (f, -s).$$

**Remark 3.8.** We note that we could also have defined the reversal of orientation to also involve changing the sign on the tangent vectors at the marked points. This gives complete equivalent theories, since there is a canonical morphism of labeled marked surfaces, which induces minus the identity at the marked points, and which is the identity on the complement of small disjoint neighbourhoods of the marked points and which locally around each marked point twist half a turn positively according to the surface orientation around the marked point, yet remains the identity near the boundary of the neighbourhood of the marked point.

Finally we describe the factorization procedure, where we obtain a  $\Lambda$ -labeled marked surface by cutting along an oriented simple closed curve  $\gamma$  whose homology class is in the distinguished Lagrangian subspace, and then collapse the resulting two boundary components to points, which get labeled by  $(i, i^{\dagger})$  as described below.

**Definition 3.9** (Factorization data). Factorization data is a triple  $(\Sigma, \gamma, i)$ , where  $\Sigma$  is a  $\Lambda$ -labeled marked surface and  $\gamma$  is a smooth, oriented, simple closed curve with a basepoint  $x_0$ , such that the homology class of  $\gamma$  lies in L. Further, i is an element of the label set  $\Lambda$ . We also say that the pair  $(\gamma, i)$  is a choice of factorization data for  $\Sigma$ .

**Definition 3.10** (Factorization). Let  $\Sigma = (\Sigma, P, V, \lambda, L)$  be a A-labeled marked surface with factorization data  $(\gamma, i)$ . We will define a A-labeled marked surface  $\Sigma^i_{\gamma}$ . We denote the underlying smooth surface by  $\Sigma_{\gamma}$ . Cutting along  $\gamma$  we get a smooth oriented surface  $\Sigma_{\gamma}$  with two boundary components  $\gamma_{-}$  and  $\gamma_{+}$ . The orientation of  $\gamma$  together with the orientation of  $\Sigma$  allows us to define  $\gamma_+$  to be the component whose induced Stokes orientation agrees with that of  $\gamma$ . The underlying smooth surface is given by  $\Sigma_{\gamma} := \tilde{\Sigma}_{\gamma} / \sim$  where we collapse  $\gamma_{-}$  to a point  $p_{-}$  and we collapse  $\gamma_+$  to a point  $p_+$ . We orient this surface such that  $\Sigma \setminus \gamma \hookrightarrow \Sigma_{\gamma} / \sim$ is orientation preserving. The set of distinguished points for  $\Sigma_{\gamma}$  is  $P \sqcup \{p_{-}, p_{+}\}$ . Identifying  $P(T_{p_{\pm}}(\tilde{\Sigma}_{\gamma}))$  with  $\gamma$ , we choose  $v(p_{\pm})$  to be  $x_0$ . We extend the labelling  $\lambda$  by labelling  $p_+$  by i and  $p_-$  by  $i^{\dagger}$ . There is a topological space X given by identifying  $p_{-}$  and  $p_{+}$ . Clearly this space is naturally homeomorphic to  $\Sigma/\sim$ , where we collapse  $\gamma$  to a point. Thus we have quotient maps  $q: \Sigma \to X$  and  $n: \Sigma_{\gamma} \to X$ . Define  $L_{\gamma} := (n_{\#})^{-1}(q_{\#})(L)$ . This yields a Lagrangian subspace of  $H_1(\Sigma_{\gamma})$  that respects the splitting induced by decomposing  $\Sigma_{\gamma}$  into connected components. We say that  $\Sigma_{\gamma}^{i}$  is obtained by factorizing  $\Sigma$  along  $(\gamma, i)$ .

There is an inverse procedure that we call glueing.

**Definition 3.11** (Glueing data). Glueing data consist of a triple  $(\Sigma, (p_0, p_1), c)$ . Here  $\Sigma = (\Sigma, P, V, \lambda, L)$  is a  $\Lambda$ -labeled marked surface with  $p_0, p_1 \in P$ , such that  $\lambda(p_0) = \lambda(p_1)^{\dagger}$  and  $c: P(T_{p_0}\Sigma) \xrightarrow{\sim} P(T_{p_+}\Sigma)$  is an orientation reversing projective linear isomorphism mapping  $v(p_0)$  to  $v(p_1)$ . We also say that  $(p_0, p_1, c)$  determine glueing data for  $\Sigma$  and that  $(p_0, p_1)$  is subject to glueing.

As we are dealing with ordered pairs  $(p_0, p_1)$  we will sometimes speak of  $p_0$  as the preferred point.

**Definition 3.12** (Glueing). Assume we are given a glueing data  $(\Sigma, (p_0, p_1), c)$ . We will define a  $\Lambda$ -labeled marked surface  $\Sigma_c^{p_0, p_1}$ . We denote the underlying smooth surface by  $\Sigma_c^{p_0, p_1}$ . Blow up  $\Sigma$  at  $p_0, p_1$  and glue in  $P(T_{p_0}\Sigma)$  and  $P(T_{p_1}\Sigma)$  to obtain a smooth oriented surface with boundary, that, as a set, can be naturally identified with

$$(\Sigma \setminus \{p_0, p_1\}) \sqcup P(T_{p_0}\Sigma) \sqcup P(T_{p_1}\Sigma).$$

Now identify the two boundary components through  $x \sim c(x)$ . This yields a smooth oriented surface, that will be the underlying surface of  $\Sigma_c^{p_0,p_1}$ . As distinguished points, directions and labels, we simply take those from  $\Sigma$ . Let X be the topological space obtained from  $\Sigma$  by identifying  $p_0$  with  $p_1$ . We have continuous maps  $q: \Sigma \to X$  and  $n: \Sigma_{c^{n,p_2}}^{n,p_2} \to X$ . Set  $L_{c,p_0,p_1} := (n_{\#})^{-1}(q_{\#})(L)$ . This is a Lagrangian subspace of  $H_1(\Sigma_{\gamma})$  that respects the splitting induced by decomposing  $\Sigma_{\gamma}$  into connected components.

Observe that the homology class of  $P(T_{p_0}\Sigma)$  lies in  $L_{c,p_0,p_1}$ .

**Proposition 3.13** (Consecutive glueing). Assume that two distinct pairs of points  $(p_1, p_2, c)$  and  $(q_1, q_2, d)$  are subject to glueing. Then there is a canonical diffeomorphism

$$s^{p_1,p_2,q_1,q_2}: (\Sigma_c^{p_1,p_2})_d^{q_1,q_2} \to (\Sigma_d^{q_1,q_2})_c^{p_1,p_2}.$$

In abuse of notation we will also write  $s^{p_1,p_2,q_1,q_2}$  for the induced morphism of labeled marked surfaces given by  $([s^{p_1,p_2,q_1,q_2}], 0)$ .

We recall that any two orientation reversing self-diffeomorphisms of  $S^1$  fixing a basepoint are isotopic among diffeomorphisms fixing this basepoint. Therefore we wish to detail the independence of the choice of c in the glueing construction.

**Proposition 3.14** (Glueing independent of c). Assume we are given a  $\Lambda$ -labeled marked surface  $\Sigma$  and two pairs of glueing data  $(p_0, p_1, c_1)$  and  $(p_0, p_1, c_2)$ . Then there is an orientation preserving diffeomorphism  $f : \Sigma \to \Sigma$  that induces the identity on  $(P, V, \lambda, L)$  and such that  $c_1 \circ df = df \circ c_2$ . Moreover f can be chosen to induce the identity morphism (id, 0) on  $\Sigma$ . Any two such f induces the same morphism of  $\Lambda$ -labeled marked surfaces, and therefore we have a canonical identification morphism  $\tilde{f}(c_1, c_2) : \Sigma_{c_1}^{p_0, p_1} \to \Sigma_{c_2}^{p_0, p_1}$  given by the pair ([f], 0).

It follows from this that in order to specify glueing, it will suffice to specify an ordered pair  $(p_0, p_1)$  with  $\lambda(p_0) = \lambda(p_1)^{\dagger}$ .

**Proposition 3.15** (Functoriality of glueing). Let  $\Sigma_i$  for i = 1, 2 be  $\Lambda$ -labeled marked surfaces. Assume  $(p_0^i, p_1^i)$  are subject to glueing for i = 1, 2. Consider any morphism  $\mathbf{f} = ([f], s) : \Sigma_1 \to \Sigma_2$  with  $f(p_0^1) = p_0^2$  and  $f(p_1^1) = p_1^2$ . Let  $c : P(T_{p_0^1}\Sigma_1) \to P(T_{p_1^1}\Sigma_1)$  be orientation reversing. Let  $c' := df \circ c \circ df^{-1} : P(T_{p_0^2}\Sigma_2) \to P(T_{p_1^2}\Sigma_2)$ . This data induces a morphism

$$\mathbf{f}' = ([f'], s) : (\mathbf{\Sigma}_1)_c^{p_0^1, p_1^2} \longrightarrow (\mathbf{\Sigma}_2)_{c'}^{p_0^2, p_1^2}$$

compatible with f.

#### 3.2 Extended surfaces

We now describe the category of extended surfaces following Turaev [29]. Observe that this is only defined relative to a modular tensor category  $(\mathcal{V}, (V_i)_{i \in I})$ . We recall that an orientation of a closed topological surface  $\Sigma$  is a choice of fundamental class in  $H^2(\Sigma_{\alpha}, \mathbb{Z})$  for each component  $\Sigma_{\alpha}$ . A degree 1-homeomorphism between oriented closed surfaces is a homeomorphism that respects this choice. We recall that an arc  $\gamma \subset \Sigma$  is a topological embedding of [0, 1]. **Definition 3.16** (Extended surfaces). An *e*-surface  $\Sigma$  is given by the following data  $(\Sigma, (\alpha_i), (W_i, \mu_i), L)$ . Here  $\Sigma$  is an oriented closed surface,  $(\alpha_i)$  is a finite collection of disjoint oriented arcs. To each arc  $\alpha_i$  we have an object  $W_i$  of  $\mathcal{V}$  and a sign  $\mu_i \in \{\pm 1\}$ . The pair  $(W_i, \mu_i)$  is called the marking of  $\alpha_i$ . Finally, L is a Lagrangian subspace of  $H_1(\Sigma, \mathbb{R})$ . By convention  $\emptyset$  is an *e*-surface.

We now describe the arrows.

**Definition 3.17** (Weak extended homeomorphisms and their composition). Let  $\Sigma_1, \Sigma_2$  be two *e*-surfaces. A weak *e*-homeomorphism  $f : \Sigma_1 \to \Sigma$  is a degree 1-homeomorphism between the underlying topological surfaces  $\Sigma_1 \to \Sigma_2$  that induces an orientation and marking preserving bijection between their distinguished arcs. An *e*-homeomorphism  $f : \Sigma_1 \to \Sigma$  is a weak *e*-homeomorphism that induces an isomorphism of distinguished Lagrangian subspaces  $f_{\#} : L_1 \to L_2$ . We observe that the class of weak *e*-homeomorphisms is closed under composition, and that this is also the case for *e*-homeomorphisms.

Thus we have the category of extended surfaces based on  $(\mathcal{V}, (V_i)_{i \in I})$ .

**Definition 3.18** (The category of extended surfaces based on  $\mathcal{V}$ ). The category of extended surfaces based on  $\mathcal{V}$  has *e*-surfaces as objects and weak *e*-homeomorphisms as morphisms. We denote it by  $\mathbf{E}(\mathcal{V})$ .

As above we wish to make this into a symmetric monoidal category with an orientation reversal.

**Definition 3.19** (Disjoint union of *e*-surfaces). Let  $\Sigma_1 = (\Sigma_1, (\alpha_i), (W_i, \mu_i), L)$ and  $\Sigma_2 = (\Sigma_2, (\beta_j), (Z_j, \eta_j), L')$  be two *e*-surfaces. We define  $\Sigma_1 \sqcup \Sigma_2$  to be

$$(\Sigma_1 \sqcup \Sigma_2, (\alpha_i \sqcup \beta_j), (W_i, \mu_i) \sqcup (Z_j, \eta_j), L \oplus L').$$

For a pair of (weak) morphisms  $f_i : \Sigma_i \to \Sigma_3$  we observe that  $f_1 \sqcup f_2$  is a (weak) morphism. We have an obvious natural transformation

$$\operatorname{Perm}: \Sigma_1 \sqcup \Sigma_2 \to \Sigma_2 \sqcup \Sigma_1.$$

**Proposition 3.20** ( $\mathbf{E}(\mathcal{V})$  is a symmetric monoidal category). The category of extended surfaces is a symmetric monoidal category with disjoint union as product, the empty surface as unit, and Perm as the braiding.

**Definition 3.21** (Orientation reversal for *e*-surfaces). Consider an extended surface  $\Sigma = (\Sigma, (\alpha_i), (W_i, \mu_i), L)$ . We define  $-\Sigma$  to be

$$(-\Sigma, (-\alpha_i), (W_i, -\mu_i), L).$$

That is, we reverse the orientation on each component, reverse the orientation of arcs, keep the labels, multiply all signs by -1, and keep the Lagrangian subspace. We observe that any (weak) *e*-homeomorphism  $f : \Sigma_1 \to \Sigma_2$  yields a (weak) morphism  $f : -\Sigma_1 \to -\Sigma_2$ .

#### 3.3 Marked surfaces

Finally we describe the category of marked surfaces<sup>2</sup>. This is defined relative to a monoidal class. That is, a class C together with a strictly associative operation  $C \times C \to C$  and a unit 1 for this operation. Again we here follow Turaev [29].

**Definition 3.22** (Marked surface over C). A marked surface (over C) is a compact oriented surface  $\Sigma$  endowed with a Lagrangian subspace of  $H_1(\Sigma, \mathbb{R})$  and such that each connected component X of  $\partial \Sigma$  is equipped with a basepoint, a sign  $\delta$ , and an element V of C called the label. The pair  $(V, \delta)$  is called the marking of X. By convention  $\emptyset$  is an *m*-surface.

Next we describe the morphisms.

**Definition 3.23** (Weak m-homeomorphisms). Let  $\Sigma_1, \Sigma_2$  be two marked surfaces. A weak *m*-homeomorphism  $f : \Sigma_1 \to \Sigma_2$  is an orientation preserving homeomorphism *f* that respects the marks of boundary components. An *m*-homeomorphism is a weak *m*-homeomorphism that also preserves the Lagrangian subspaces.

**Definition 3.24** (The category of marked surfaces over C). The category of marked surfaces over C has *m*-surfaces as objects and weak *m*-homeomorphisms as morphisms. We denote it  $\mathbf{M}(C)$ .

As above this naturally constitute a symmetric monoidal category with disjoint union as the product.

**Definition 3.25** (Disjoint union of marked surfaces). Let  $\Sigma_1, \Sigma_2$  be two *m*surfaces. Then we define the marked surface  $\Sigma_1 \sqcup \Sigma_2$  by declaring that the boundary components naturally inherit basepoints and markings, and equipping it with a Lagrangian subspace of  $H_1(\Sigma_1 \sqcup \Sigma_2, \mathbb{R})$ , by taking the direct sum of Lagrangian subspaces of  $\Sigma_1, \Sigma_2$ . If  $f_1, f_2$  are (weak) *m*-homeomorphisms, then  $f_1 \sqcup f_2$  is a (weak) *m*-homeomorphism. We have a natural transformation

Perm : 
$$\Sigma_1 \sqcup \Sigma_2 \to \Sigma_2 \sqcup \Sigma_1$$
.

**Proposition 3.26** ( $\mathbf{M}(C)$  is a symmetric monoidal category). The category  $\mathbf{M}(C)$  of marked surfaces (over C) is a symmetric monoidal category with disjoint union as product, the empty surface as unit, and Perm as the braiding.

**Definition 3.27** (Glueing). Let  $\Sigma$  be an *m*-surface. Assume that there are two components X, Y with the same label, but with opposite sign. We say that X, Y are subject to glueing. There is a (unique up to isotopy) basepoint preserving orientation-reversing homeomorphism  $c: X \to Y$ . The quotient  $\Sigma' = \Sigma / \sim$  where  $x \sim c(x)$  is naturally an oriented compact surface. The quotient map  $q: \Sigma \to \Sigma'$  yields a bijection  $\partial \Sigma' \sim \partial \Sigma \setminus X \cup Y$ . Using this, we equip each component of  $\partial \Sigma'$ 

<sup>&</sup>lt;sup>2</sup>Not to be confused with  $\Lambda$ -labeled marked surfaces.

with a basepoint and a marking. Finally, equip  $\Sigma'$  with the Lagrangian subspace that is the image of the Lagrangian subspace of  $\Sigma$  under  $q_{\#}$ . Denote the resulting *m*-surface by

$$\Sigma/[X = Y]_c$$

**Proposition 3.28** (Functorial property of glueing of *m*-surfaces). Let  $\Sigma$  be an *m*-surface. Assume  $X, Y \subset \partial \Sigma$  are two boundary components subject to glueing. Let  $x : X \to Y$  be basepoint preserving and orientation reversing. Let  $f : \Sigma \to \Sigma'$  be a (weak) *m*-homeomorphism. Then  $X' = f(X), Y' = f(Y) \subset \partial \Sigma'$  are subject to glueing and the map c' given by  $f \circ c \circ f^{-1} : X' \to Y'$  is orientation reversing and basepoint preserving. There is a unique (weak) homeomorphism  $f_c : \Sigma/[X = Y]_c \to \Sigma'/[X' = Y']_{c'}$  inducing a commutative diagram:



Here the vertical maps are the quotient maps.

When dealing with Turaev's 2-DMF, it is convenient to introduce a symmetric monoidal category  $\mathcal{M}'(C)$  very similar to  $\mathcal{M}(C)$  but with fever morphisms. See remark 3.30 below.

**Definition 3.29.** Let  $\mathcal{M}'(C)$  be the category with the same objects as  $\mathbf{M}(C)$ , but where morphisms are equivalence classes of weak *m*-homeomorphisms, where two parallel weak *m*-homeomorphisms are equivalent if and only if they are isotopic through weak *m*-homeomorphisms. The braiding and the permutation is defined similarly to those of  $\mathcal{M}(C)$ .

**Remark 3.30.** We recall that the 2-DMF  $\mathcal{H}_{\mathcal{V}}$  defined in chapter V of [29] descends to  $\mathcal{M}'(C)$  in the sense that if f, g are two equivalent weak *m*-homeomorphisms, then we have the identity  $\mathcal{H}(f) = \mathcal{H}(g)$ .

### 4 Axioms for a modular functor

We now recall Kevin Walker's axioms for a modular functor as they are given and used in [12, 13, 14]. For Turaev's axioms of a modular functor, we refer to chapter V in [29]. We assume familiarity with the notion of symmetric monoidal functors. Roughly speaking, a symmetric monoidal functor between symmetric monoidal categories  $(C, \otimes, e) \to (D, \otimes', e')$  is a triple  $(F, F_2, f)$  where  $F : C \to D$ is a functor,  $F_2$  is a family of morphisms  $F_2 : F(a) \otimes' F(b) \to F(a \otimes b)$  and f is a morphism  $f : e' \to F(e)$ . For the precise formulation of the axioms we refer to [22]. For brevity we will write  $F = (F, F_2, f)$ . If  $F_2, f$  are always isomorphisms, we say that F is a strong monoidal functor.

158

#### 4.1 The Walker axioms for a modular functor

Let  $\Lambda = (\Lambda,^{\dagger}, 0)$  be a label set. Let K be a commutative ring (with unit). Let  $\mathbf{P}(K) = \operatorname{Proj}(K)$  be the category of finitely generated projective K-modules. We recall that this is a symmetric monoidal category with the tensor product over K as product, and K as unit.

**Definition 4.1** (Modular functor V based on  $\Lambda$  and K). A modular functor based on a label set  $\Lambda$  and a commutative ring K is a strong monoidal functor V

$$V : \mathbf{C}(\Lambda) \to \mathbf{P}(K),$$

satisfying the glueing axiom, the one punctured sphere axiom and the twice punctured sphere axiom as these are described below.

The glueing axiom. Assume that  $(p_1, p_2, c)$  is a glueing data for a labeled marked surface  $\Sigma$ . For any  $\lambda \in \Lambda$ , let  $\Sigma(\lambda)$  be the labeled marked surface identical to  $\Sigma$  except for the fact that  $p_1$  is labeled with  $\lambda$  and  $p_2$  is labeled with  $\lambda^{\dagger}$ . Then  $(p_1, p_2, c)$  is a glueing data for  $\Sigma(\lambda)$ . We demand that there is a specified isomorphism

(4.1) 
$$g: \bigoplus_{\lambda \in \Lambda} V(\Sigma(\lambda)) \xrightarrow{\sim} V(\Sigma_c^{p_1, p_2}).$$

Let  $g_{\lambda}$  be the restriction of g to  $V(\Sigma(\lambda))$ . If the context is clear, we will simply write g for this restriction, and suppress  $\lambda$  from the notation. If we wish to stress the glueing map c, we will write  $g^c$ . The glueing isomorphism is subject to the four axioms below.

(i). The isomorphism should be associative in the following sense. Assume that  $(q_1, q_2, d)$  is another pair subject to glueing. For any pair  $(\lambda, \mu) \in \Lambda^2$  let  $\Sigma(\lambda, \mu)$  be the labeled marked surface identical to  $\Sigma$  except that  $p_1$  is labeled with  $\lambda, p_2$  is labeled with  $\lambda^{\dagger}, q_1$  is labeled with  $\mu$  and  $q_2$  is labeled with  $\mu^{\dagger}$ . Then the following diagram is commutative

Here  $s' = V(s^{p_1, p_2, q_1, q_2})$ , where  $s^{p_1, p_2, q_1, q_2}$  is as defined in Prop. 3.13.

(ii). The isomorphism should be compatible with glueing of morphisms in the following sense. Assume that  $\mathbf{f}: \Sigma_1 \to \Sigma_2$  is a morphism such that a pair  $(p_0, p_1)$  subject to glueing is taken to the pair  $(q_0, q_1)$ . Choosing *c* will induce a morphism  $\mathbf{f}': (\Sigma_1)_c^{p_0,p_1} \longrightarrow (\Sigma_2)_{c'}^{q_0,q_1}$  as in Prop 3.15. This should induce a commutative diagram:

(4.3) 
$$V(\Sigma_1) \xrightarrow{g} V((\Sigma_1)_c^{p_0,p_1})$$
$$V(\mathbf{f}) \qquad \qquad \downarrow V(\mathbf{f}')$$
$$V(\Sigma_2) \xrightarrow{g} V((\Sigma_2)_{c'}^{q_0,q_1})$$

(iii). The isomorphism should be compatible with disjoint union in the following way. Assume that  $(p_0, p_1, c)$  is a glueing data for  $\Sigma_1$ . For any  $\Sigma_2$ , we see that  $(p_0, p_1, c)$  is also a choice of glueing data for  $\Sigma_1 \sqcup \Sigma_2$ , and that there is a canonical morphism  $\iota = (\iota, 0) : (\Sigma_1)_c^{p_1, p_2} \sqcup \Sigma_2 \longrightarrow (\Sigma_1 \sqcup \Sigma_2)_c^{p_0, p_1}$ . This should induce a commutative diagram

(iv). The isomorphism should be independent of the glueing map c in the following way. Assume a pair of points  $(p_0, p_1)$  in  $\Sigma$  is subject to glueing. Assume that  $c_1, c_2 : P(T_{p_0}\Sigma) \to P(T_{p_1}\Sigma)$  are two glueing maps. Consider the identification morphism  $f(c_1, c_2) : \Sigma_{c_1}^{p_0, p_1} \to \Sigma_{c_2}^{p_0, p_2}$  as in Prop 3.14. This should induce a commutative diagram

The once punctured sphere axiom. For any  $\lambda \in \Lambda$  consider a sphere with one distinguished point  $\Sigma_{\lambda} = (S^2, \{p\}, \{v\}, \{\lambda\}, 0)$ . We demand that

(4.6) 
$$V(\Sigma_0) \simeq \begin{cases} K \text{ if } \lambda = 0\\ 0 \text{ if } \lambda \neq 0. \end{cases}$$

The twice punctured sphere axiom. For any ordered pair  $(\lambda, \mu)$  in  $\Lambda$ , consider a sphere with two distinguished points  $\Sigma_{\lambda,\mu} = (S^2, \{p_1, p_2\}, \{v_1, v_2\}, \{\lambda, \mu\}, 0)$ . We demand that

(4.7) 
$$V(\boldsymbol{\Sigma}_{\lambda,\mu}) \simeq \begin{cases} K \text{ if } \mu = \lambda^{\dagger} \\ 0 \text{ if } \mu \neq \lambda^{\dagger}. \end{cases}$$

**Remark 4.2.** We stress that the isomorphisms given in (4.6) and (4.7) are not part of the data of a modular functor. Only the existence of such isomorphisms are required. Below we will occasionally denote a modular functor by a pair (V,g) where V is the strong monoidal functor, and g is the glueing isomorphism 4.1.

# 5 Construction of a modular functor $Z_{\mathcal{V}}$ .

#### 5.1 The symmetric monoidal functor

From now on, we consider a modular tensor category  $(\mathcal{V}, (V_i)_{i \in I})$  and take  $\Lambda = I$ and  $^{\dagger} =^*$ . We let K be the commutative ring End(1), where 1 is the unit for the tensor product in  $\mathcal{V}$ .

**Proposition 5.1** (Existence of a strong monoidal functor  $\mathbf{C}(I) \to \mathbf{M}'(\mathcal{V})$ ). Consider a modular tensor category  $(\mathcal{V}, (V_i)_{i \in I})$ . Let  $\Lambda = I$ ,  $^{\dagger} =^*$  and let  $C = \mathcal{V}$  considered as a monoidal class. There is a strong monoidal functor from the category of I-labeled marked surface into the category  $\mathbf{M}'(C)$ 

$$\mathcal{G}: \mathbf{C}(I) \to \mathbf{M}'(C).$$

For an I-labeled marked surface  $\Sigma = (\Sigma, P, V, \lambda, L)$  the marked surface  $\mathcal{G}(\Sigma)$  is given as follows. For any distinguished point p, blow up  $\Sigma$  at p. That is, the underlying topological surface of  $\mathcal{G}(\Sigma)$  is denoted by  $\mathcal{G}(\Sigma)$  and is given as follows

$$\left(\Sigma \setminus P \bigsqcup_{p \in P} S_p^1\right) / \sim .$$

Here we glue in the circle  $S_p^1$  using smooth coordinates in a neigbourhood of p. The orientation agrees with that on  $\Sigma$ . The direction  $v_p$  yields a basepoint on  $S^1$ , the label  $i \in I$  yields a marking  $(V_i, 1)$ . Collapsing  $S_p^1$  to a point at all p yields a surface  $\Sigma'$ , that is canonically homeomorphic to  $\Sigma$ . Let  $\eta$  denote the natural homeomorphism  $\Sigma' \to \Sigma$ . Let q denote the quotient map that collapses any component to a point. The composition  $g := \eta \circ q : \mathcal{G}(\Sigma) \to \Sigma$  will be an isomorphism on homology, and this provides us with a Lagrangian subspace  $L' := g_{\#}^{-1}(L)$ . Given a morphism of labeled marked surfaces  $(f, s) : \Sigma_1 \to \Sigma_2$  any representative of fnaturally induces a weak m-homeomorphism  $\mathcal{G}(\Sigma_1) \to \mathcal{G}(\Sigma_2)$  and we let  $\mathcal{G}(f, s)$ be the corresponding equivalence class. We are now finally ready to define our modular functor. We recall that even though Turaev's axioms for a 2 - DMF as given in chapter V only requires functoriality with respect to *m*-homeomorphisms, it is also defined on weak *m*homeomorphisms. See section 4.3 in chapter V.

**Definition 5.2** (The definition of  $Z_{\mathcal{V}}$ ). Let  $\mathcal{H}_{\mathcal{V}}$  be the 2-DMF as defined in chapter V of [29] relative to  $(\mathcal{V}, (V_i)_{i \in I})$ . On the level of objects we define  $Z_{\mathcal{V}}$  to be

$$Z_{\mathcal{V}} := \mathcal{H}_{\mathcal{V}} \circ \mathcal{G} : \mathbf{C}(I) \longrightarrow \mathbf{P}(K).$$

For a morphism of labeled marked surfaces  $(f, s) : \Sigma \to \Sigma'$  we define

$$Z_{\mathcal{V}}(f,s) := (\Delta^{-1}D)^s \mathcal{H}_{\mathcal{V}}(\mathcal{G}(f,s)).$$

Here  $D, \Delta$  are invertible scalars in K to be introduced in section 7.1 below. We write  $Z = Z_{\mathcal{V}}$  and  $\mathcal{H} = \mathcal{H}_{\mathcal{V}}$ . We need to address the issue of functoriality. That is we must verify that  $Z(\mathbf{f}) \circ Z(\mathbf{g}) = Z(\mathbf{f} \circ \mathbf{g})$  for composable morphisms of labeled marked surfaces. Let V, V' be symplectic vector spaces. Recall that Walker's signature cocycle for an ordered triple  $(L_1, L_2, L_3)$  of Lagrangian subspaces  $L_i \subset V$  coincide with the Maslov index  $\mu(L_1, L_2, L_3)$ . Recall also that  $\mu(L_1, L_2, L_3) = \mu(f(L_1), f(L_2), f(L_3))$  for any symplectomorphism  $f : V \to V'$ . These facts together with remark 5.4 and lemma 6.3.2 in chapter IV of [29] easily imply functoriality.

We need to define a glueing isomorphism. We start by observing the following proposition.

**Proposition 5.3** ( $\mathcal{G}$  is compatible with glueing). Assume  $\Sigma$  is a labeled marked surface. Assume we are given glueing data (p,q,c). Assume p is labeled with i. Consider  $\Sigma' = \mathcal{G}(\Sigma)$ . If we replace the marking of  $X_q$  with  $(V_i, -1)$  to obtain a new marked surface  $\Sigma''$ , then  $X_p \subset \partial \Sigma''$  and  $X_q \subset \partial \Sigma''$  are subject to glueing. We observe

$$\mathcal{G}(\mathbf{\Sigma}_c^{p,q}) = \Sigma'' / [X_p \approx X_q].$$

We now compare the glueing isomorphism axiom of Walker and the splitting axiom of Turaev more closely. Turaev's modular functor is subject to the splitting axiom, which means that the glueing homomorphisms provide an isomorphism

$$g: \bigoplus_{i\in I} \mathcal{H}\left(\mathcal{G}(\Sigma), (V_i, 1), (V_i, -1)\right)) \xrightarrow{\sim} Z(\Sigma_c).$$

See chapter V, the splitting axiom on page 246 in [29]. Comparing with Walker's glueing axiom, we see that the summands are not the same, since there we need an isomorphism

$$g: \bigoplus_{i\in I} \mathcal{H} \left( \mathcal{G}(\Sigma), (V_i, 1), (V_{i^*}, 1) \right) \xrightarrow{\sim} Z(\Sigma_c).$$

162

Hence we need to provide isomorphisms between modules of states, where we exchange a marking  $(V_{i^*}, 1)$  with  $(V_{i_i}, -1)$ . To provide these identifications, we first recall that  $\mathcal{H}$  is compatible with the operator invariant  $\tau^e$ . For an explanation of this see the following remark.

**Remark 5.4.** Let  $\Sigma$  be a marked surface over  $\mathcal{V}$ . We recall that  $\mathcal{H}(\Sigma)$  is naturally isomorphic to  $\mathcal{T}^e(\overline{\Sigma})$ , where  $\overline{\Sigma}$  is an extended surface obtained from  $\Sigma$  by glueing in discs with preferred diameter, that are taken to be marked arcs. We can therefore use the operator invariant  $\tau^e$  to obtain morphisms between modules of states. The natural isomorphism  $\mathcal{H}(\Sigma) \simeq \mathcal{T}^e(\overline{\Sigma})$  also implies, that for a labeled marked surface  $\Sigma$  we could just as well define  $Z(\Sigma)$  as  $\mathcal{T}^e(\widetilde{\Sigma})$ , where  $\widetilde{\Sigma}$ is an extended surface naturally obtained from  $\Sigma$ . Similarly, we observe that a morphism (f, s) of *I*-labeled marked surfaces induces an equivalence class f' of weak *e*-homeomorphisms, and that  $\mathcal{H}(\mathcal{G}(f, s)) \sim \mathcal{T}^e(f')$ .

Now we provide the needed identifications.

**Lemma 5.5** (The natural transformation f). Let  $\Sigma$  be an *m*-surface with a boundary component  $X_{\alpha}$  marked with (V, 1). Assume that  $\Sigma'$  is obtained from  $\Sigma$  by replacing the marking (V, 1) with (W, 1). Assume that  $f : V \to W$  is a morphism. There is a K-linear morphism

$$\dot{f}: \mathcal{H}(\Sigma) \to \mathcal{H}(\Sigma'),$$

where  $\dot{f}$  is induced from the extended three manifold  $M = \overline{\Sigma} \times I$ . Here we think of the bottom as  $\overline{\Sigma}$ , the top as  $\overline{\Sigma'}$ , and we provide M with following ribbon graph. For each arc  $\beta$  different from the arc  $\alpha$  corresponding to  $X_{\alpha} \subset \partial \Sigma$ , we put in the identity strand  $\beta \times I$ . For  $\alpha$ , we put in a coupon colored with f.

**Lemma 5.6** (The natural transformation  $h_{\alpha}$ ). Let  $\Sigma$  be an *m*-surface with a boundary component  $X_{\alpha}$  marked with  $(V^*, 1)$ . Assume that  $\Sigma'$  is obtained from  $\Sigma$  by replacing the marking  $(V^*, 1)$  with the marking (V, -1). There is a K-linear morphism

$$h_{\alpha}: \mathcal{H}(\Sigma) \to \mathcal{H}(\Sigma').$$

The morphism  $h_{\alpha}$  is induced from the extended three manifold  $M = \overline{\Sigma} \times I$  where we think of the bottom as  $\overline{\Sigma}$ , the top as  $\overline{\Sigma'}$ , and we provide M with following ribbon graph. For each arc  $\beta$  different from the arc  $\alpha$  corresponding to  $X_{\alpha} \subset \partial \Sigma$ , we put in the identity strand  $\beta \times I$ . For  $\alpha$ , we put in a coupon colored with  $id_{V^*}$ .

If the relevant boundary component is understood, we will simply write  $h_{\alpha} = h$ . In section 8 below we will give all details of how these two lemmas follow directly from similar statements in [29].

#### 5.2 The glueing isomorphism

Let  $\Sigma_{\mathbf{c}}$  be an I-labeled marked surface obtained from  $\Sigma$  by glueing. We must provide an isomorphism

$$\bigoplus_{i\in I} Z(\mathbf{\Sigma},i,i^{\dagger}) \overset{\sim}{\longrightarrow} Z(\mathbf{\Sigma_c}).$$

For each  $i \in I$  fix an isomorphism  $q_i : V_{i^*} \to V_i^*$ . We define the glueing isomorphism as follows. For each i, consider the composition

$$(5.1) \quad Z(\mathbf{\Sigma}, i, i^*) \xrightarrow{q_i} \mathcal{H}(\mathcal{G}(\Sigma), (V_i, 1), (V_i^*, 1)) \xrightarrow{h} \mathcal{H}(\mathcal{G}(\Sigma), (V_i, 1), (V_i, -1)).$$

Using that  $\mathcal{H}$  satisfies the splitting axiom as defined in chapter V, we see that we have an isomorphism

$$g: \bigoplus_{i\in I} \mathcal{H} \left( \mathcal{G}(\Sigma), (V_i, 1), (V_i, -1) \right) \xrightarrow{\sim} Z(\Sigma_c).$$

Thus we can define our glueing isomorphism as follows.

**Definition 5.7** (The glueing isomorphism). We define

$$\tilde{g}(q) := g \circ (\bigoplus_{i \in I} h \circ \dot{q_i}) : \bigoplus_{i \in I} Z(\Sigma, \lambda, i, i^*) \xrightarrow{\sim} Z(\Sigma_c).$$

We write  $\tilde{g}(q)$  to stress that this depends on the choices of isomorphisms  $q_i$ .

#### 5.3 Main theorem

We are now ready to state our main theorem.

**Theorem 5.8** (Main Theorem). For any modular tensor category  $(\mathcal{V}, (V_i)_{i \in I})$  the symmetric modular functor  $Z_{\mathcal{V}}$  as given in definition 5.2 together with the glueing isomorphism  $\tilde{g}(q)$  as given in definition 5.7 satisfies Walker's axioms of a modular functor based on I and K as given in section 4.

We will sometimes write Z(q) for the modular functor  $Z_{\mathcal{V}}$  equipped with the glueing  $\tilde{g}(q)$ .

# 6 Proof of the main theorem

We first state more or less trivial statements about the K-linear morphisms coming from Lemma 5.5 and 5.6. Recalling the setting and notation of Lemma 5.5, it is clear that if  $g: \Sigma \to \tilde{\Sigma}$  is a weak *m*-homeomorphism, then so is  $g: \Sigma' \to \tilde{\Sigma}'$ , where  $\tilde{\Sigma}'$  is obtained from  $\tilde{\Sigma}$  by replacing the marking of  $g(X_{\alpha})$  with (W, 1). **Lemma 6.1** (The natural transformation  $\dot{f}$ ). For each such g,  $\dot{f}$  induces a commutative diagram



Moreover,  $\dot{f}$  is compatible with disjoint union in the following sense. Assume  $\Sigma = \Sigma_1 \sqcup \Sigma_2$ , and  $X_{\alpha} \subset \Sigma_2$ . Then the following diagram commutes

$$\begin{aligned} \mathcal{H}(\Sigma) & \longrightarrow \mathcal{H}(\Sigma_1) \otimes \mathcal{H}(\Sigma_2) \\ \downarrow^{j} & \qquad \qquad \downarrow^{id \otimes j} \\ \mathcal{H}(\Sigma') & \longrightarrow \mathcal{H}(\Sigma_1) \otimes \mathcal{H}(\Sigma'_2). \end{aligned}$$

Recalling the setting and notation of Lemma 5.6, we observe that if  $g: \Sigma \to \tilde{\Sigma}$ is a weak *m*-morphism, then so is  $g: \Sigma' \to \tilde{\Sigma}'$ , where  $\tilde{\Sigma}'$  is obtained from  $\tilde{\Sigma}$  by replacing the marking of  $g(X_{\alpha})$  with (V, -1).

**Lemma 6.2** (The natural transformation  $h_{\alpha}$ ). For each such g,  $h_{\alpha}$  induces a commutative diagram

Moreover,  $h_{\alpha}$  is compatible with disjoint union in the following sense. Assume  $\Sigma = \Sigma_1 \sqcup \Sigma_2$ , and  $X_{\alpha} \subset \Sigma_2$ . Then the following diagram commutes

$$\begin{aligned} \mathcal{H}(\Sigma) & \longrightarrow \mathcal{H}(\Sigma_1) \otimes \mathcal{H}(\Sigma_2) \\ & h_{\alpha} \\ & & \downarrow^{id \otimes h_{\alpha}} \\ \mathcal{H}(\Sigma') & \longrightarrow \mathcal{H}(\Sigma_1) \otimes \mathcal{H}(\Sigma'_2). \end{aligned}$$

We need to know how the morphisms  $\dot{f}$ , h relate to the glueing homomorphism provided by Turaev, and we need to know what happens if we apply the  $\dot{f}$ , hoperations consecutively to distinct boundary components. We call a morphism of type  $\dot{f}$  or h a coupon morphism. **Lemma 6.3** (Far commutativity). Assume that an m-surface  $\Sigma_2$  is obtained from an m-surface  $\Sigma_1$  by altering the markings of two distinct boundaries  $X_{\alpha}, X_{\beta}$ components in one of the two ways described above. Let q be the K-morphism  $\mathcal{H}(\Sigma_1) \to \mathcal{H}(\Sigma_2)$  that is obtained from composing the coupon morphism that alters  $X_{\alpha}$  with the coupon morphism that alters  $X_{\beta}$ . Let p be the K-morphism  $\mathcal{H}(\Sigma_1) \to \mathcal{H}(\Sigma_2)$  that is obtained from composing the coupon morphism that alters the labelling of  $X_{\beta}$  with the coupon morphism that alters the labelling of  $X_{\alpha}$ . Then we have

p = q.

**Lemma 6.4** (Compatibility of  $\dot{f}$ , h with glueing.). Assume that  $\Sigma_c$  is obtained from  $\Sigma$  by glueing. Consider a component  $X_\beta$  of  $\Sigma$ , that is not part of the glueing data. Assume the marking of  $\beta$  is altered either by using a morphism f of objects of  $\mathcal{V}$ , or by replacing  $(V^*, 1)$  with (V, -1). Then this operation applies to  $\Sigma_c$  as well. Let r denote the resulting isomorphisms of modules. Let g denote the glueing homomorphism. Then

$$r \circ g = g \circ r.$$

These lemmas will be proven in section 8 below.

Proof of the Main Theorem. Since  $\mathcal{H}$  is a strong monoidal functor, it is immediate that  $Z_{\mathcal{V}}$  is a symmetric monoidal functor, since it is a composition of strong monoidal functors. Thus it remains to verify the once punctured sphere axiom, the twice punctured sphere axiom, and the glueing axiom. The once punctured sphere axiom follows directly from Turaev's disc axiom, which is axiom 1.5.5 in chapter V. The twice punctured sphere axioms follows directly from the third normalization axiom 1.6.2 in chapter V of [29].

It remains to verify the glueing axiom. If  $\mathbf{f} = (f, n)$  is a morphism of labeled marked surfaces, we will abuse notation and write f for  $\mathcal{G}(\mathbf{f})$ .

(i) In the notation of definition 4.1 we must prove that the following diagram commutes

$$\begin{split} Z(\boldsymbol{\Sigma}(i,j)) & \xrightarrow{g_j} Z((\boldsymbol{\Sigma}_d^{q_1,q_2})(i)) \\ & & \bar{g}_i \\ & & \downarrow^{\tilde{g}_i} \\ Z((\boldsymbol{\Sigma}_c^{p_1,p_2})(j)) & \xrightarrow{s' \circ \tilde{g}_j} Z((\boldsymbol{\Sigma}_d^{q_1,q_2})_c^{p_1,p_2}). \end{split}$$

Here  $s' = Z(s^{p_1,p_2,q_1,q_2})$ , where  $s^{p_1,p_2,q_1,q_2}$  is as defined in proposition 3.13, and  $i = \lambda$  and  $j = \mu$ . Let  $\alpha$  be the relevant distinguished point labeled with  $j^*$ . Let  $\beta$  be the relevant distinguished point labeled with  $i^*$ . As above, let g be the glueing homomorphism provided by Turaev in chapter V of [29]. In the following

calculation we use that the integer associated to the morphism  $s^{p_1,p_2,q_1,q_2}$  is 0. Commutativity of the diagram above can be rewritten as the following equation

$$(6.1) g_i \circ h_\beta \circ \dot{q}_i \circ g_j \circ h_\alpha \circ \dot{q}_j = s' \circ g_j \circ h_\alpha \circ \dot{q}_j \circ g_i \circ h_\beta \circ \dot{q}_i.$$

Using lemma 6.4 and lemma 6.3 we see that

$$g_i \circ h_\beta \circ \dot{q}_i \circ g_j \circ h_\alpha \circ \dot{q}_j = g_i \circ g_j \circ h_\alpha \circ \dot{q}_j \circ h_\beta \circ \dot{q}_i$$

Using lemma 6.4 we then get that

$$s' \circ g_j \circ h_\alpha \circ \dot{q_j} \circ g_i \circ h_\beta \circ \dot{q_i} = s' \circ g_j \circ g_i \circ h_\alpha \circ \dot{q_j} \circ h_\beta \circ \dot{q_i}.$$

Using axiom 1.5.4(ii) in chapter V of [29] we see that

$$s' \circ g_j \circ g_i = g_i \circ g_j.$$

Therefore we see that equation (6.1) holds.

(ii) In the notation of definition 4.1 we must prove that the following diagram commutes

$$\begin{array}{ccc} Z(\boldsymbol{\Sigma}_1) & \xrightarrow{\bar{g}} & Z((\boldsymbol{\Sigma}_1)_c^{p_0,p_1}) \\ z(\mathbf{f}) & & & \downarrow \\ Z(\mathbf{f}) & & & \downarrow \\ Z(\boldsymbol{\Sigma}_2) & \xrightarrow{\bar{g}} & Z((\boldsymbol{\Sigma}_2)_{c'}^{q_0,q_1}). \end{array}$$

This amounts to proving

(6.2) 
$$\mathcal{H}(f') \circ g \circ h \circ \dot{q} = g \circ h \circ \dot{q} \circ \mathcal{H}(f).$$

Here we use that  $\mathbf{f}'$  is equipped with the same integer as  $\mathbf{f}$ . Equation (6.2) follows directly from lemma 5.6, lemma 5.5, and axiom 1.5.4(i) in chapter V of [29]. Here we use that even though the naturality condition is only formulated for *m*-homeomorphisms in this axiom, Turaev argues in section 4.6 of chapter V that it is also valid for weak *m*-homeomorphisms.

(iii) In the notation of definition 4.1 we must prove that the following diagram commutes

$$Z(\mathbf{\Sigma}_{1} \sqcup \mathbf{\Sigma}_{2}) \xrightarrow{g} Z((\mathbf{\Sigma}_{1} \sqcup \mathbf{\Sigma}_{2})_{c}^{p_{0},p_{1}})$$

$$Z_{2} \uparrow \qquad \uparrow Z(\iota) \circ Z_{2}$$

$$Z(\mathbf{\Sigma}_{1}) \otimes Z(\mathbf{\Sigma}_{2}) \xrightarrow{\tilde{g} \otimes 1} Z((\mathbf{\Sigma}_{1})_{c}^{p_{0},p_{1}}) \otimes Z(\mathbf{\Sigma}_{2}).$$

As the integer associated with the morphism  $\iota$  is zero, this takes the following equational form

(6.3) 
$$g \circ h \circ \dot{q} \circ \mathcal{H}_2 = \mathcal{H}(\iota) \circ \mathcal{H}_2 \circ (g \circ h \circ \dot{q} \otimes 1).$$

Rewrite the RHS as  $\mathcal{H}(\iota) \circ \mathcal{H}_2 \circ (g \otimes 1) \circ (h \circ \dot{q} \otimes 1)$ . Now use lemma 5.5 and lemma 5.6 to rewrite the LHS as  $g \circ H_2 \circ (h \circ \dot{q} \otimes 1)$ . Now axiom 1.5.4(*iii*) entails  $g \circ H_2 = \mathcal{H}(\iota) \circ \mathcal{H}_2 \circ (g \otimes 1)$ . This implies equation (6.3).

(iv) In the notation of definition 4.1 we must prove that the following diagram commutes

$$Z(\mathbf{\Sigma}) \xrightarrow{\tilde{g}^{c_1}} Z(\mathbf{\Sigma}_{c_1}^{p_0,p_1})$$

$$\downarrow Z(\tilde{f}_{(c_1,c_2)})$$

$$Z(\mathbf{\Sigma}_{c_2}^{p_0,p_1}).$$

As the integer associated with the morphism  $\tilde{f}(c_1, c_2)$  is zero, this takes the form

(6.4) 
$$\mathcal{H}(f(c_1, c_2))) \circ g^{c_1} \circ h \circ \dot{q} = g^{c_2} \circ h \circ \dot{q}.$$

**Lemma 6.5.** The morphisms  $\mathcal{H}(\tilde{f}(c_1, c_2))) \circ g^{c_1}$  and  $g^{c_2}$  are operator invariants of extended three manifolds that are naturally e-homeomorphic through an e-homeomorphism commuting with boundary parametrizations. In particular they coincide.

We see that lemma 6.5 implies equation (6.4). The lemma will be proven in section 8.  $\hfill \Box$ 

# 7 Review of the TQFT based on extended cobordisms

As observed above,  $\mathcal{H}$  is defined as

$$\mathcal{H}(\Sigma) = \mathcal{T}_{\mathcal{V}}^{e}(\overline{\Sigma}),$$

where  $\overline{\Sigma}$  is the associated extended surface, and  $\mathcal{T}_{\mathcal{V}}^e$  is the modular functor based on the category of extended surfaces and the modular tensor category  $\mathcal{V}$ . In this section we will give a quick review of the TQFT ( $\mathcal{T}_{\mathcal{V}}^e, \tau_{\mathcal{V}}^e$ ) based on the cobordism theory of extended cobordisms, as defined in chapter IV of [29]. We will assume familiarity with the axioms for a TQFT based on a cobordism theory as defined in chapter III of [29]. We will assume familiarity with the quantum invariant  $\tau(M, \Omega)$  of a closed oriented three manifold M containing a ribbon graph  $\Omega$  with colors in  $\mathcal{V}$ . This invariant is defined in chapter II of [29]. We will however provide the formula associated to a surgery presentation below, but we will not explain the Reshetikhin-Turaev functor  $F_{\mathcal{V}}$  as defined in chapter I of [29].

168

#### 7.1 Quantum invariants of 3-manifolds

We now recall the construction of  $\tau(\tilde{M}, \Omega) \in K$  where  $\tilde{M}$  is a closed oriented three manifold with a colored ribbon graph  $\Omega$  inside. Here  $\tau(\tilde{M}, \Omega)$  is called the quantum invariant of  $(\tilde{M}, \Omega)$ . We may assume  $\tilde{M} = \partial W$ , where W is a compact oriented four manifold obtained by performing surgery along a framed link  $L = \{L_1, ..., L_m\}$  in  $S^3 = \partial B^4$ . Let  $\sigma(L)$  be the signature of the intersection form on  $H_2(W, \mathbb{R})$ . Let  $\operatorname{Col}(L)$  be the set of all colorings of L by colors in  $(V_i)_{i \in I}$ . For any coloring  $\lambda$  we let  $\Gamma(L, \lambda)$  be the associated colored ribbon graph in  $S^3$ . Then  $\tau(\tilde{M}, \Omega)$  is given by

(7.1) 
$$\tau(\tilde{M},\Omega) = \Delta^{\sigma(L)} D^{-\sigma(L)-m-1} \sum_{\lambda \in \operatorname{Col}(L)} \dim(\lambda) F(\Gamma(L,\lambda) \cup \Omega).$$

Here  $\dim(\lambda) = \dim(\lambda_1) \cdots \dim(\lambda_m)$  and D is given by

$$D^2 = \sum_{i \in I} \dim(i)^2,$$

and  $\Delta$  is given by

$$\Delta := \sum_{i \in I} k_i^{-1} (\dim(i))^2 \in K,$$

where the  $k_i$  are the standard twists coefficients - Turaev denotes them  $v_i$  in [29].

#### 7.2 The TQFT based on decorated cobordisms

#### 7.2.1 Modules of states

Recall the notion of a *d*-surface and decorated type as defined in section 1.1 of chapter IV in [29]. Recall the notion of a standard *d*-surface and of a parametrized *d*-surface as in section 1.2 and section 1.3 of chapter IV in [29]. Assume  $\Sigma$  is a connected parametrized *d*-surface of topological type *t* given by  $(g; (W_i, \mu_i), ..., (W_m, \mu_m))$ . Recall the standard *d*-surface of type *t*. This is denoted by  $\Sigma_t$ . These notions can be found in sections 1.1 - 1.3 of chapter IV. For a decorated type *t* as above, and for  $i \in I^g$ , let

$$\Phi(t,i) := W_1^{\mu_1} \otimes \cdots \otimes W_m^{\mu_m} \bigotimes_{s=1}^g (V_{i_s} \otimes V_{i_s}^*).$$

Here  $W^1 = W$ , and  $W^{-1} = W^*$ . Recall that elements of  $\Phi(t, i)$  can be thought of as colorings of the ribbon graph  $R_t$  sitting inside  $\Sigma_t$ , as defined in section 1.2 of [29]. Moreover we define  $\mathcal{T}(\Sigma) := \Psi(t)$  where

$$\Psi(t) := \bigoplus_{i \in I^g} \operatorname{Hom}(\mathbf{1}, \Phi(t, i)).$$

Finally, if  $\Sigma$  is not connected, then we define  $\mathcal{T}(\Sigma)$  to be the unordered tensor product of the modules of states of the components of  $\Sigma$ .

#### 7.2.2 Operator invariants

We now describe the construction of  $\tau(M)$ , where M is a decorated cobordism. That is, M is a triple  $(M, \partial_- M, \partial_+ M)$  where  $\partial_{\pm} M$  are parametrized d-surfaces. For a general decorated type t let  $U_t$  be the standard decorated handlebody bounded by  $\Sigma_t$  as in section 1.7 of chapter IV in [29]. Equip it with the RH orientation. For an element  $x \in \Phi(t, i)$  consider the three manifold with boundary  $H(U_t, R_t, i, x)$ . Let  $U_t^-$  be the image of  $U_t$  under the reflection of  $\mathbb{R}^3$  in the plane  $\mathbb{R}^2 \times \{1/2\}$ . We denote this orientation reversing diffeomorphism by mir :  $\mathbb{R}^3 \to \mathbb{R}^3$ . Equip  $U_t^-$  with the RH orientation. We recall that they contain certain ribbon graphs denoted  $R_t, R_{-t}$  respectively. Let  $f: \Sigma_{t_0} \to \partial_- M$  be a parametrization of a component of  $\partial_{-}M$ . We glue in  $U_{t_0}$  by glueing  $\partial U_{t_0}$  to  $\Sigma_{t_1} \times \{0\}$  through f. We do this for all components of  $\partial_{-}M$ . Similarly, for any parametrized component  $g: \Sigma_{t_1} \to \partial_+ M$  we glue in  $U_{t_1}^-$  by glueing according to  $-g \circ \min : \partial(U_{t_1}^-) \to \Sigma_{t_1}$ . This produces a closed oriented three manifold  $\tilde{M}$  with a ribbon graph inside, such that choosing an element  $x \in \mathcal{T}(\partial_- M)$  and an element  $y \in \mathcal{T}(\partial_+ M)^*$  will produce a colored ribbon graph  $\Omega(x,y) \subset M$ . This descends to a K-linear map  $\mathcal{T}(\partial_{-}M) \otimes_{K} \mathcal{T}(\partial_{+}M)^{*} \longrightarrow K$  given by

$$x \otimes y \longrightarrow \tau(M, \Omega(x, y)),$$

where  $\tau$  is the quantum invariant defined in chapter II of [29]. This pairing induces a morphism  $j : \mathcal{T}(\partial_-M) \to \mathcal{T}(\partial_+M)$ . Finally, composing this with the map  $\eta : \mathcal{T}(\partial_+M) \to \mathcal{T}(\partial_+M)$  induced by multiplication by  $\mathcal{D}^{1-g}\dim(i)$  on  $\operatorname{Hom}(\mathbf{1}, \Phi(t_1; i))$ , we get the desired K-linear map

$$\tau(M) := \eta \circ j : \mathcal{T}(\partial_{-}M) \to \mathcal{T}(\partial_{+}M).$$

#### 7.3 The TQFT based on extended cobordisms

#### 7.3.1 Module of states

We start by describing the module of states for an *e*-surface. Start by assuming that  $\Sigma$  is a connected *e*-surface. Recall the notion of a parametrization of  $\Sigma$ . This is simply a weak e-homeomorphism  $\Sigma_t \to \Sigma$ . Given two parametrizations  $f: \Sigma_{t_0} \to \Sigma$  and  $g: \Sigma_{t_1} \to \Sigma$ , we wish to define an isomorphism  $\varphi(f, g)$  between  $\Psi(t_0)$  and  $\Psi(t_1)$ . We define

$$\varphi(f,g) := (D\Delta^{-1})^{-\mu((f_0)_*(\lambda(t_0)),\lambda(\Sigma),(g_0)_*(\lambda(t_1)))} \mathcal{E}(g^{-1}f) : \Psi(t_0) \to \Psi(t_1).$$

Here  $\mu$  is the Maslov index for triples of Lagrangian subspaces.  $\mathcal{E}$  is the morphism induced by the decorated three manifold  $\Sigma_{t_1} \times I$  where the bottom is parametrized by  $g^{-1} \circ f : \Sigma_{t_0} \to \Sigma_{t_1}$  and the top is parametrized by the identity. Turaev proves in [29] that

$$\varphi(f_1, f_2) \circ \varphi(f_0, f_1) = \varphi(f_0, f_2).$$

Now  $\mathcal{T}^{e}(\Sigma)$  is defined as the K-module of coherent sequences  $(x(t, f))_{(t,f)}$  where we index over all parametrizations. Finally, if  $\Sigma$  is not connected, then we define  $\mathcal{T}^{e}(\Sigma)$  to be the unordered tensor product of the modules of states of the components of  $\Sigma$ .

#### 7.3.2 Operator invariants

Consider an extended 3-manifold  $(M, \partial_- M, \partial_+ M)$ . Then any two parametrizations  $f: \Sigma_- \to \partial_- M$  and  $g: \Sigma_+ \to \partial_+ M$  makes M into a decorated cobordism  $\tilde{M}$ . We now define  $\tau^e(M)$  to be composition

$$\mathcal{T}^e(\partial_- M) \to \mathcal{T}(\Sigma_-) \xrightarrow{\lambda(M)\tau(M)} \mathcal{T}(\Sigma_+) \to \mathcal{T}^e(\partial_+ M).$$

Here  $\lambda(M)$  is an invertible element of K defined in section 6.5 of chapter IV of [29].

# 8 Proofs of lemmas

**Proposition 8.1** (The cobordism associated to a weak *e*-homomorphism). Let  $f: \Sigma_1 \to \Sigma_2$  be a weak *e*-homomorphism. There is an invertible scalar  $c \in K$  such that the operator invariant of the extended cobordism  $\Sigma_1 \times I \cup_f \Sigma_2 \times I$  coincide with  $c\mathcal{T}^e(f): \mathcal{T}^e(\Sigma_1) \to \mathcal{T}^e(\Sigma_2)$ . The scalar *c* depends only on the underlying continuous map of *f* and the Lagrangian subspaces  $L_i \subset H_1(\Sigma_i)$ .

*Proof.* This follows from theorem 7.1 of chapter VII in [29].

Here it is understood that the extended three manifold  $\Sigma_1 \times I \cup_f \Sigma_2 \times I$  is obtained by glueing the top of  $\Sigma_1 \times I$  to the bottom of  $\Sigma_2 \times I$  through f.

Proof of Lemmas 5.5 and 6.1. Using 8.1 and theorem 7.1 of chapter VII in [29], we see that both  $\mathcal{H}(g) \circ \dot{f}$  and  $\dot{f} \circ \mathcal{H}(g)$  are - up to the same scalar - induced by glueing certain extended three manifolds. Let  $M_1$  be the extended three manifold with  $\tau(M_1) = c\mathcal{H}(g)\circ \dot{f}$ , and let  $M_2$  be the extended three manifold with  $\tau^e(M_2) = c\dot{f} \circ \mathcal{H}(g)$  Clearly there is a homeomorphism of extended three manifolds taking  $M_1$  to  $M_2$ , commuting with boundary parametrizations. Therefore they induce the same morphism.

Proof of Lemmas 5.6 and 6.2. The proof is virtually identical to the proof of Lemma 5.5.  $\hfill \Box$ 

Proof of Lemma 6.3. The proof is virtually identical to the proof of Lemma 5.5.  $\hfill \Box$ 

Proof of Lemma 6.4. We start by recalling the definition of the glueing homomorphism provided by Turaev in sections 4.4 - 4.6 of chapter V in [29]. Let  $M_2$  be the extended three manifold obtained by attaching handles to  $\Sigma \times I$  as in section 4.4. of chapter V of [29]. The attachment uses the glueing data c. The operator invariant  $\tau^e(M_2)$  now yields a map  $g': \mathcal{T}^e(\Sigma) \to \mathcal{T}^e(\Sigma'_c)$ . Here  $\Sigma'_{c}$  is an *e*-surface canonically *e*-homeomorphic to  $\Sigma_{c}$ . Composing with the associated isomorphism of K-modules, we get the required glueing homomorphism  $g: \mathcal{T}^e(\Sigma) \to \mathcal{T}^e(\Sigma_c)$ . Similarly, if we let  $\widetilde{\Sigma}$  and  $\widetilde{\Sigma}'_c$  be the two same *e*-surfaces with the relevant change of markings, then the glueing g is obtained as the operator invariant of  $M_2$  with the obvious notation. Assume now that r = f for some morphism  $f: (V, +1) \to (W, +1)$ . Recalling the naturality property of  $\dot{f}$  we see that it is enough to argue that f commute with g'. Let  $M_1 = \Sigma \times I$  be the extended cobordism inducing  $\dot{f}$ . Then  $g' \circ \dot{f}$  is a multiple of  $\tau^e(M_2 \circ M_1)$ , where we glue the top of  $M_1$  to the bottom of  $M_2$  through the identity. To compute the relevant scalar we use theorem 7.1 in Chapter IV of [29]. Since the identity is an e-homeomorphism here, there is only one Maslow index to compute. In the notation of theorem 7.1 we have  $id_{\#}(N_1)_*(\lambda_-(M_1)) = id_{\#}\lambda_+(M_1)$ . Thus we see that

$$0 = \mu \left( \mathrm{id}_{\#}(N_1)_*(\lambda_{-}(M_1)), \mathrm{id}_{\#}\lambda_{+}(M_1), N_2^*(\lambda_{+}(\widetilde{M}_2)) \right)$$

See the proof of lemma 6.7.2 in chapter IV of [29]. Thus we get that

$$\tau^e(\widetilde{M}_2 \circ M_1) = g' \circ \dot{f}.$$

Clearly  $\widetilde{M_2} \circ M_1$  is *e*-homeomorphic to a cylinder with handles attached on the top, such that the  $\beta$ -band has a coupon colored with the *f*-coupon, and all other 'vertical' bands are colored with id. The exact same argument will yield a similar description of  $\dot{f} \circ g$ . Consider  $Q := \Sigma_c \times I$  as the extended cobordism inducing  $\dot{f} : \mathcal{T}^e(\Sigma_c) \xrightarrow{\rightarrow} \mathcal{T}^e(\widetilde{\Sigma}_c)$ . Arguing as above, wee see that  $\dot{f} \circ g'$  is given by the operator invariant  $\tau^e(Q \circ M_2)$ . But this is *e*-homeomorphic to  $\widetilde{M_2} \circ M_1$ . Therefore  $g \circ \dot{f} = \dot{f} \circ g'$ . Observe that a homomorphism of type *h* can be dealt with in exactly the same way.

*Proof of Lemma 6.5.* This is a consequence of the description of the glueing homomorphism given above, together with the existence of the proclaimed e-morphisms.

# 9 Uniqueness up to quasi-isomorphism

We observe that the construction of the glueing  $\tilde{g}$  map depended on a choice of isomorphisms  $q_i : V_{i^*} \xrightarrow{\sim} V_i^*$ . This dependence is not essential.

**Definition 9.1** (Quasi-isomorphism). Let (Z, g) and (Z', g') be two modular functors with the same label set  $\Lambda$ . These are said to be quasi-isomorphic if there is a pair  $(\Phi, \gamma)$  satisfying the following conditions.  $\Phi$  is an assignment, which for each labeled marked surface  $\Sigma$  gives an isomorphism

$$\Phi(\mathbf{\Sigma}): Z(\mathbf{\Sigma}) \xrightarrow{\sim} Z'(\mathbf{\Sigma})$$

This assignment is required to be natural with respect to morphisms of modules induced by morphisms of labeled marked surfaces. Similarly it is required to preserve the splitting into tensor products induced by disjoint union, as well as the permutation map.  $\gamma$  is an assignment  $\gamma : I \to K^*$  such that if  $\Sigma_c$  is obtained from  $\Sigma$  from glueing along an ordered pair (p, q) where p is labeled with  $\lambda$ , then the following diagram is commutative

Moreover we demand that  $\gamma(\lambda)\gamma(\lambda^*) = 1$  for all  $\lambda \in I$ .

This is easily seen to define an equivalence relation on modular functors with the same label set.

**Theorem 9.2** (Independence of  $(q_i)$  up to quasi-isomorphism). Let q, q' be two choices of isomorphisms  $V_{i^*} \xrightarrow{\sim} V_i^*$ . Then the two resulting modular functors Z(q) and Z(q') are quasi-isomorphic.

*Proof.* Write  $q'_i = f_i q_i$ . Then we have  $\tilde{g}_j(q') = f_j \tilde{g}_j(q)$ . We want to construct a pair  $(\Phi, \gamma)$ . Consider a labeled marked surface  $\Sigma$  with labels  $i_1, ..., i_k$ . We want to construct  $\Phi(\Sigma)$  to be of the form  $(\prod_i^k \alpha_{i_i}) \mathrm{Id}_{Z(\Sigma)}$  for some function  $\alpha : I \to K^*$ .

Assume  $\Sigma_c$  is obtained from  $\Sigma$  by glueing along an ordered pair (p, q) where p is labeled with i. Assume the labels of  $\Sigma$  are  $i_1, ..., i_k, i, i^*$ . Then equation (9.1) becomes

$$\prod_{l=1}^{k} \alpha(i_l) = \gamma(i) f_i \alpha(i) \alpha(i^*) \prod_{l=1}^{k} \alpha(i_l).$$

Thus we are forced to define

$$\gamma(i) := \frac{1}{\alpha(i)\alpha(i^*)f_i}.$$

We still have to ensure  $\gamma(i)\gamma(i^*) = 1$ . We see that this will follow for any choice of  $\alpha$  with

$$(\alpha(i)\alpha(i^*))^2 f_i f_{i^*} = 1,$$

which is easy to solve (by adjoining the needed square roots to K if needed).  $\Box$ 

### 10 Universal property

In this section we will describe how to apply Z, and how to use it in calculations. Let  $\Sigma$  be a connected labeled marked surface. Recall that a parametrization  $f: \Sigma_t \to \overline{\mathcal{G}}(\overline{\Sigma})$  is an orientation preserving homeomorphism that preserves all structure of extended surfaces, except possibly the Lagrangian subspaces in homology. These are also called weak *e*-homeomorphisms. Clearly the set of parametrizations is non-empty. Let f be a parametrization. This will induce an isomorphism

$$\Psi(t) \simeq Z(\Sigma).$$

For the definition of  $\Psi(t)$  see section 7. We now recall the definition of  $Z(\Sigma)$  and describe the isomorphism above. For any pair of parametrizations

$$f_i: \Sigma_{t_i} \to \overline{\mathcal{G}(\Sigma)}, i = 1, 2,$$

there is an isomorphism

$$\varphi(f_1, f_2) : \Psi_{t_1} \xrightarrow{\sim} \Psi_{t_2}$$

See section 7. With the obvious notation these isomorphisms satisfy

$$\varphi(f_1, f_3) = \varphi(f_2, f_3) \circ \varphi(f_1, f_2).$$

The module  $Z(\Sigma)$  is the module of coherent sequences. Hence an element of this module is an equivalence class of pairs (x, f) where f is a parametrization with domain  $\Sigma_t$  and x is an element of  $\Psi(t)$ . We have  $(x, f) \sim (y, g)$  if and only if  $\varphi(f, g)(x) = y$ . The isomorphism  $\Psi(t) \simeq Z_{\mathcal{V}}(\Sigma)$  induced from a parametrization is simply  $x \mapsto (x, f)$ .

Now we describe  $Z_{\mathcal{V}}(\mathbf{f})$  when  $\mathbf{f} = (f, s) : \Sigma_1 \to \Sigma_2$  is a morphism of connected labeled marked surfaces. Any representative f' of the equivalence class f will induce  $Z_{\mathcal{V}}(\mathbf{f})$  which is given by

$$(x,g) \mapsto ((\Delta^{-1}D)^s x, f' \circ g)$$

Here we also write f' for the induced *e*-homeomorphism  $\overline{\mathcal{G}(\Sigma_1)} \to \overline{\mathcal{G}(\Sigma_2)}$ .

# 11 The duality pairing

Consider a modular functor V based on a label set  $\Lambda$ . For a modular functor with duality we would like the operation of orientation reversal to correspond to the operation of taking the dual K-module. That is, we would like a perfect pairing  $V(\Sigma) \otimes V(-\Sigma) \to K$  that is compatible with the structure of V. Before we formulate the axioms, consider an arbitrary  $\Lambda$ -labeled marked surface  $\Sigma'$ . Observe that if  $p, q \in \Sigma'$  are subject to glueing then so are  $p, q \in -\Sigma'$ . Observe that if  $\Sigma$ is the result of glueing  $\Sigma$  along p, q then  $-\Sigma$  is the result of glueing  $-\Sigma'$  along the same ordered pair of points.
**Definition 11.1** (Duality). Let (V, g) be a modular functor based on  $\Lambda$  and K. A duality for V is a perfect pairing

$$(\cdot, \cdot)_{\Sigma}: V(\Sigma) \otimes V(-\Sigma) \to K,$$

subject to the following axioms.

Naturality. Let  ${\bf f}=(f,s):\Sigma_1\to \Sigma_2$  be a morphism between A-labeled marked surfaces. Then

(11.1) 
$$(V(\mathbf{f}), V(-\mathbf{f}))_{\Sigma_2} = (\cdot, \cdot)_{\Sigma_1}$$

Compatibility with disjoint union. Consider a disjoint union of  $\Lambda$ -labeled marked surfaces  $\Sigma = \Sigma_1 \sqcup \Sigma_2$ . The modular functor V provides an isomorphism

$$\eta: V(\mathbf{\Sigma}) \otimes V(-\mathbf{\Sigma}) \xrightarrow{\sim} V(\mathbf{\Sigma}_1) \otimes V(-\mathbf{\Sigma}_1) \otimes V(\mathbf{\Sigma}_2) \otimes V(-\mathbf{\Sigma}_2).$$

We demand that with respect to the natural isomorphism  $K\otimes K\simeq K$  we have that

(11.2) 
$$(\cdot, \cdot)_{\Sigma} = ((\cdot, \cdot)_{\Sigma_1} \otimes (\cdot, \cdot)_{\Sigma_2}) \circ \eta.$$

Compatibility with glueing. Let  $\Sigma$  be a  $\Lambda$  labeled marked surface obtained from glueing. Consider the glueing isomorphism

$$g: \bigoplus_{\lambda \in \Lambda} V(\boldsymbol{\Sigma}(\lambda)) \overset{\sim}{\longrightarrow} V(\boldsymbol{\Sigma}),$$

as described in definition 4.1. We have that

(11.3) 
$$(g,g)_{\Sigma} = \sum_{\lambda \in \Lambda} \mu_{\lambda} (\cdot, \cdot)_{\Sigma(\lambda)}$$

Here  $\mu_{\lambda} \in K$  is invertible and depends only on the isomorphism class of  $\Sigma(\lambda)$  for all  $\lambda$ .

Compatibility with orientation reversal. For a  $\Lambda$ -labeled marked surface  $\Sigma$  we demand that there is an invertible element  $\mu \in K^*$  that only depends on  $\Sigma$  such that for all  $(v, w) \in V(\Sigma) \times V(-\Sigma)$  the following equation holds

(11.4) 
$$\mu(w,v)_{-\Sigma} = (v,w)_{\Sigma}.$$

It is worth spelling out how we demand that the duality is compatible with glueing in a little more detail. Observe  $-(\Sigma(\lambda)) = (-\Sigma)(\lambda^{\dagger})$ . Thus the glueing isomorphism is a splitting

$$g': \bigoplus_{\lambda \in \Lambda} V(-\Sigma(\lambda^{\dagger})) \xrightarrow{\sim} V(-\Sigma).$$

This gives a decomposition

$$\bigoplus_{\lambda,\lambda'\in\Lambda} V(\mathbf{\Sigma}(\lambda)) \otimes V(-\mathbf{\Sigma}(\lambda')) \xrightarrow{g\otimes g} V(\mathbf{\Sigma}) \otimes V(-\mathbf{\Sigma}).$$

Then the statement is that  $g(V(\Sigma(\lambda)))$  and  $g(V(-\Sigma(\lambda')))$  are orthogonal w.r.t. the duality  $(\cdot, \cdot)_{\Sigma}$  unless  $\lambda = \lambda'$ . In this case we have

$$(g_{\lambda}, g_{\lambda^{\dagger}})_{\Sigma} = \mu_{\lambda}(\cdot, \cdot)_{\Sigma(\lambda)}.$$

### 11.1 Review of the duality for Turaev's modular functor based on extended surfaces

Let  $\Sigma$  be an *e*-surface. Recall the operation of oriental reversal as described in definition 3.21. We can think of  $\Sigma \times I$  as a morphism  $\Sigma \sqcup (-\Sigma) \to \emptyset$ . This induces a perfect pairing

 $\langle \cdot, \cdot \rangle_{\Sigma} : \mathcal{T}^e(\Sigma) \otimes \mathcal{T}^e(-\Sigma) \to K.$ 

See chapter III section 2 in [29]. The pairing is compatible with the action of *e*-homeomorphisms in the sense that for any *e*-homeomorphism  $f: \Sigma_1 \to \Sigma_2$  we have

 $\langle \mathcal{T}^e(f)(\cdot), \mathcal{T}^e(-f)(\cdot) \rangle_{\Sigma_2} = \langle \cdot, \cdot \rangle_{\Sigma_1}.$ 

It is proven in exercise 7.3 in chapter IV in [29] that the pairing is also natural with respect to weak *e*-homeomorphisms. The pairing is multiplicative with respect to disjoint union. Moreover the pairing is self-dual in the following sense

$$\langle \cdot, \cdot \rangle_{\Sigma} \circ \operatorname{Perm} = \langle \cdot, \cdot \rangle_{-\Sigma}$$

All these properties are stated in axiom 1.2.4 in section 1.2 of chapter III in [29].

#### 11.2 Construction of a duality pairing for Z

In order to induce Turaev's duality pairing, we will need an isomorphism

$$Z(-\Sigma) \xrightarrow{\sim} \mathcal{T}^e(-\overline{\mathcal{G}(\Sigma)})$$

It is very important to note that our choice made below can be scaled. See remark 11.3, and section 14.

Consider an *I*-labeled marked surface  $\Sigma = (\Sigma, P, V, (i_p)_{p \in P}, L)$ . Write

$$\overline{\mathcal{G}(\boldsymbol{\Sigma})} = (\boldsymbol{\Sigma}, (\alpha_p)_{p \in P}, (V_{i_p}, 1)_{p \in P}, L)$$

for the *e*-surface associated to the *m*-surface  $\mathcal{G}(\Sigma)$ . We have that

$$\overline{\mathcal{G}(-\boldsymbol{\Sigma})} = (\Sigma, (\alpha_p)_{p \in P}, (V_{i_p^*}, 1)_{p \in P}, L).$$

This is not quite  $-\overline{\mathcal{G}(\Sigma)}$ . However, let  $\dot{q}$  be the isomorphism of states that take all markings  $(V_{i^*}, 1)$  to  $(V_i^*, 1)$  and let  $\tilde{h}$  be the isomorphism of modules of states that exchange all markings  $(V_i^*, 1)$  with  $(V_i, -1)$ . Define

$$*\Sigma := (\widetilde{\Sigma}, (\alpha_p)_{p \in P}, (V_{i_p}, -1)_{p \in P}, L).$$

Now let r be the orientation-preserving diffeomorphism

$$r: *\Sigma \xrightarrow{\sim} -\overline{\mathcal{G}(\Sigma)},$$

that is given by twisting all arcs with a half-twist. This can of course be done in two different ways, the important thing for now is that it is done the same way for all arcs. We return to this choice in the proof of proposition 11.7. Then we have an isomorphism

$$\mathcal{T}^e(r) \circ \tilde{h} \circ \dot{q} : \mathcal{T}^e(\overline{\mathcal{G}(-\Sigma)}) \xrightarrow{\sim} \mathcal{T}^e(-\overline{\mathcal{G}(\Sigma)}).$$

This will allows us to define a perfect pairing. For notational convenience we will simply write  $r' = \mathcal{T}^e(r)$ . We will write

(11.5) 
$$\zeta = \mathcal{T}^e(r) \circ \tilde{h} \circ \dot{q} : Z(-\Sigma) \xrightarrow{\sim} \mathcal{T}^e(-\overline{\mathcal{G}(\Sigma)}).$$

One last notational definition will be convenient. For a decorated type t of the form  $(g; (V_{i_l}, \nu_l))$  with  $i_l \in I$  for all l, let  $t^*$  be the decorated type  $(g; (V_{i_1^*}, \nu_l))$ .

**Definition 11.2.** Consider an *I*-labeled marked surface  $\Sigma$ . We have a perfect pairing given by the composition

$$(\cdot, \cdot)_{\Sigma} = \langle \cdot, \zeta(\cdot) \rangle_{\overline{\mathcal{G}(\Sigma)}}.$$

**Remark 11.3.** We note that there are choices involved in defining  $\zeta$  and therefore choices involved in the pairing. In particular we here note that we can choose to scale the  $q_i$  used in the definition of  $\zeta$ , such that we use a different set of isomorphisms in the glueing and in the duality. We will examine this phenomenon in section 14. Moreover we have here resorted to a slight abuse of notation, since technically,  $Z(\Sigma)$  is not equal to  $\mathcal{T}^e(\overline{\mathcal{G}(\Sigma)})$ , but canonically isomorphic to it.

**Theorem 11.4** (Duality). The pairing  $(\cdot, \cdot)_{\Sigma}$  is a duality pairing for the modular functor  $Z_{\mathcal{V}}$ .

#### 11.3 Description of the glueing homomorphism

For the proof of theorem (11.4) we will use an explicit description of the glueing homomorphism in two cases.

#### 11.3.1 The two points lie on the same component

We will start by assuming that the points subject to glueing lie on the same component. Write  $\Sigma$  for the labeled marked surface resulting from glueing and write  $\Sigma(i)$  for the labeled marked surface with the two points subject to glueing where the preferred point is labeled with *i*. Due to the multiplicativity of the glueing we will assume that  $\Sigma(i)$  is connected. See equation (4.4). We will start by assuming  $\overline{\mathcal{G}(\Sigma(i))} = \Sigma_i$ , where

$$t = (g; (V_{i_1}, 1), \dots, (V_{i_k}, 1), (V_i, 1), (V_{i^*}, +1))$$

Hence  $\overline{\mathcal{G}(\Sigma)} = \Sigma_{t'}$  where t' is equal to the topological type  $(g+1, (V_{i_1}, 1), ..., (V_{i_k}, 1))$ . Moreover let

$$\tilde{t} = (g; (V_{i_1}, 1), ..., (V_{i_k}, 1), (V_i, 1), (V_i, -1)).$$

In Turaev's setup we see that  $\Sigma_{\tilde{t}}$  can be glued along the points labeled with  $(V_i, 1)$ and  $(V_i, -1)$  to obtain  $\Sigma_{t'}$ . Now the identity parametrizations induces isomorphisms

$$\begin{aligned} \mathcal{T}^{e}(\Sigma_{\tilde{t}}) &\simeq \bigoplus_{l \in I^{g}} \operatorname{Hom}(\mathbf{1}, \Phi(\tilde{t}, l)), \\ Z(\mathbf{\Sigma}) &\simeq \bigoplus_{l \in I^{g+1}} \operatorname{Hom}(\mathbf{1}, \Phi(t', l)), \\ Z(\mathbf{\Sigma}(i)) &\simeq \bigoplus_{l \in I^{g}} \operatorname{Hom}(\mathbf{1}, \Phi(t, l)). \end{aligned}$$

With respect to these isomorphisms, we see that our glueing homomorphism is the composition

(11.6) 
$$Z(\mathbf{\Sigma}(i)) \stackrel{h \circ q}{\to} \mathcal{T}^e(\Sigma_{\tilde{t}}) \hookrightarrow Z(\mathbf{\Sigma}).$$

Here the last map is the natural summandwise inclusion. This is proven in section 5.9 of chapter V of [29]. Thus we only need to describe the first map. Consider a summand in  $Z(\Sigma)$  and an element f

$$f \in \operatorname{Hom}\left(\mathbf{1}, \left(\otimes_{i=1}^{k} V_{i_{i}}\right) \otimes V_{i} \otimes V_{i^{*}} \otimes_{r=1}^{g} \left(V_{l_{r}} \otimes V_{l_{r}}^{*}\right)\right).$$

Let  $W = (\bigotimes_{j=1}^{k} V_{i_j}) \otimes V_i$ ,  $R = \bigotimes_{r=1}^{g} (V_{l_r} \otimes V_{l_r}^*)$  and  $q_i : V_{i^*} \xrightarrow{\sim} V_i^*$  be the isomorphism used to define the glueing. Post composing f with  $(1_W \otimes q_i \otimes 1_R)$  we get an element

$$(1_W \otimes q_i \otimes 1_R) \circ f \in \operatorname{Hom}\left(\mathbf{1}, (\bigotimes_{i=1}^k V_{i_i}) \otimes V_i \otimes V_i^* \otimes_{r=1}^g (V_{l_r} \otimes V_{l_r}^*)\right).$$

To see that  $h \circ \dot{q}(f) = (1_W \otimes q_i \otimes 1_R) \circ f$ , one can either go through the construction given in the review in section 7 above and use that the identity cylinder induces the identity, or one can use the techniques of section 2.3 in chapter IV of [29].

Now we describe the glueing in a slightly more general situation. We observe that  $\Sigma_r$  is naturally a labelled marked surface, for any type r where all marks are of type  $(V_i, 1)$ . Assume that we have a parametrization  $f : \Sigma_t \to \overline{\mathcal{G}(\Sigma(i))}$ . There is a natural homeomorphism  $\overline{\mathcal{G}(\Sigma(i))} \simeq \Sigma(i)$ . With respect to this identification we can think of f as a diffeomorphism between labeled marked surface that preserves all the data except possibly the Lagrangian subspaces in homology. There is also a natural homeomorphism of  $\Sigma_{t'}$  with the surface obtained from  $\Sigma_t$  by glueing the ordered pair corresponding under f to the relevant ordered pair of  $\Sigma(i)$ . We have a natural homeomorphism  $\overline{\mathcal{G}(\Sigma)} = \Sigma$ . As in proposition 3.15 we can choose a parametrization diffeomorphism  $F = z(f) : \Sigma_{t'} \to \overline{\mathcal{G}(\Sigma)}$  that is compatible with f. Now f, F induces a pair of isomorphisms

(11.7) 
$$Z(\Sigma) \simeq \bigoplus_{l \in I^{g+1}} \operatorname{Hom}(\mathbf{1}, \Phi(t', l)),$$

(11.8) 
$$Z(\mathbf{\Sigma}(i)) \simeq \bigoplus_{l \in I^g} \operatorname{Hom}(\mathbf{1}, \Phi(t, l))$$

With respect to these isomorphisms the description given by (11.6) is valid.

#### 11.3.2 The two points lie on distinct components

We assume that  $\Sigma(i) = \Sigma^+ \sqcup \Sigma^-$  where  $\Sigma^+$  and  $\Sigma^-$  are two spheres. Assume that  $(p, i) \in \Sigma^+$  and that  $(q, i^*) \in \Sigma^-$ . We will assume  $\Sigma^- = \Sigma_{t_-}$  and  $\Sigma^+ = \Sigma_{t_+}$  where  $t_+ = (0; (V_{i_1}, 1), ..., (V_{i_n}, 1), (V_i, 1))$  and  $t_- = (0; (V_{i^*}, 1), (V_{i_{n+1}}, 1), ..., (V_{i_m}, 1))$ . Thus we get  $\Sigma = \Sigma_t$  where

$$t = (0, (V_{i_1}, 1), ..., (V_{i_m}, 1)).$$

We get isomorphisms

$$Z(\Sigma(i)) \simeq \operatorname{Hom}(\mathbf{1}, \Phi(t_+)) \otimes \operatorname{Hom}(\mathbf{1}, \Phi(t_-)),$$
  
$$Z(\Sigma) \simeq \operatorname{Hom}(\mathbf{1}, \Phi(t)).$$

Let  $V_1 = V_{i_1} \otimes \cdots \otimes V_{i_n}$  and  $V_2 = V_{i_{n+1}} \otimes \cdots \otimes V_{i_m}$ . With respect to these isomorphisms the glueing homomorphism is given by

(11.9) 
$$Z(\mathbf{\Sigma}(i)) \ni x \otimes y \mapsto (\mathbf{1}_{V_1} \otimes \overline{d_{V_i}} \otimes \mathbf{1}_{V_2}) \circ (x \otimes q_i \circ y) \in Z(\mathbf{\Sigma}).$$

Here  $\overline{d_V}$  is given by  $F(\cap_V)$  where F is the Reshetikhin-Turaev functor and  $\cap_V$  is defined in Figure 2.6 in section 2.3 of chapter I in [29]. This formula is verified by using the description of  $h \circ \dot{q}$  given above and by arguing very similar to the reasoning in section 5.10 of chapter V in [29]. In the general case, where  $\Sigma^+, \Sigma^-$  are homeomorphic to spheres, we start with a parametrization of each component  $\overline{\mathcal{G}(\Sigma^{\pm})}$  and then we glue these two together to obtain a parametrization of  $\overline{\mathcal{G}(\Sigma)}$ . These will induce isomorphisms with respect to which the glueing is given by (11.9). If we start with parametrizations f, g we will write  $z'(f \otimes g)$  for the resulting parametrization.

#### 11.4 Proof of theorem 11.4

**Proposition 11.5.** The pairing  $(\cdot, \cdot)_{\Sigma}$  is functorial and is compatible with disjoint union.

*Proof.* This easily follows from the properties of  $\langle \cdot, \cdot \rangle$  and the functorial properties of  $\dot{q}, h$  and r. Observe that even though the axioms in section 1 of chapter II of [29] only ensure that  $\langle \cdot, \cdot \rangle$  is natural with respect to *e*-homeomorphisms, it is proven in exercise 7.3 in chapter IV that the pairing is also natural with respect to weak *e*-homeomorphisms.

Thus it remains to prove that it is compatible with glueing, and that it is self-dual. The proof of the main propositions needed for these results is based on an explicit ribbon graph presentation of  $\langle \cdot, \cdot \rangle_{\Sigma} : \mathcal{T}^e(\Sigma) \otimes \mathcal{T}^e(-\Sigma) \to K$ .

All of the proofs in this section are modifications of material appearing in section 10.4. in chapter IV of [29].

**Proposition 11.6** (Surgery presentation of  $\langle \cdot, \cdot \rangle_{\Sigma}$ ). Assume  $\Sigma$  is a connected esurface. For any parametrization  $f : \Sigma_t \to \Sigma$  there is an induced parametrization  $y(f) : \Sigma_{-t} \to -\Sigma$  such that with respect to the two induced isomorphisms

$$\Psi(t) \simeq \mathcal{T}^{e}(\Sigma),$$
  
$$\Psi(-t) \simeq \mathcal{T}^{e}(-\Sigma),$$

we have the following surgery presentation of  $\langle \cdot, \cdot \rangle_{\Sigma}$ .



Observe that in this proposition, orientation reversal is with respect to extended e-surfaces. Here the blue unknot's are the surgery link components. We depict here the genus 1 case. It is obvious how to generalize to higher genus.

We stress that the tangle is a presentation of a perfect pairing

(11.10) 
$$\Psi(t) \times \Psi(-t) \to K$$
,

and that we can only use it as a presentation of the duality pairing with respect to certain pairs of parametrizations  $f : \Sigma_t \to \Sigma$  and  $y(f) : \Sigma_{-t} \to -\Sigma$ . This will be explained in the proof. We will denote the pairing from (11.10) by  $\langle \cdot, \cdot \rangle_t$ .

*Proof.* This proof is a slight modification of the proof of theorem 10.4.1 in chapter IV of [29]. Observe that in this proof  $-\Sigma$  is the result of using the operation of orientation reversal of extended surfaces on  $\Sigma$ .

Choose a parametrization  $f: \Sigma_t \to \Sigma$ . This will induce a weak *e*-homeomorphism  $-f: -\Sigma_t \to -\Sigma$ . Consider the weak *e*-homeomorphism  $s: \Sigma_{-t} \to -\Sigma_t$  given by a reflection in y = 0 followed by counter clockwise half twists in the plane at the distinguished arcs - with respect to the usual identification  $\mathbb{R}^2 \times \mathbb{R} = \mathbb{R}^3$ . This yields a parametrization  $y(f) := (-f) \circ s : \Sigma_{-t} \to -\Sigma$ . These two parametrizations provides isomorphisms

$$\Psi(t) \simeq \mathcal{T}^{e}(\Sigma),$$
  
$$\Psi(-t) \simeq \mathcal{T}^{e}(-\Sigma).$$

Now let

$$x \in \operatorname{Hom}(\mathbf{1}, \Phi(t, i)) \subset \mathcal{T}^{e}(\Sigma), \quad y \in \operatorname{Hom}(\mathbf{1}, \Phi(-t, j)) \subset \mathcal{T}^{e}(-\Sigma)$$

Consider the standard handlebodies denoted by  $P(x) = H(U_t, R_t, i, x)$  and  $Q'(y) = H(U_{-t}, R_t, j, y)$ . We recall that  $\langle x, y \rangle$  is given by  $\tau(W(x, y))$ , where W(x, y) is the closed three manifold with a colored ribbon graph inside it, that is obtained by glueing  $P \sqcup Q'$  to  $\Sigma \times I$  through the orientation reversing homeomorphism

$$\partial(P(x)\sqcup Q'(y)) = \Sigma_t \sqcup \Sigma_{-t} \xrightarrow{f\sqcup y(f)} \Sigma \times \{0\} \sqcup \Sigma \times \{1\} = \partial(\Sigma \times I).$$

We now observe that the parametrization  $s : \Sigma_{-t} \to -\Sigma$  extends to an *e*-homeomorphism of three manifolds

$$Q'(y) \to Q(y),$$

where Q(y) is the same handlebody with the LH-orientation and the induced colored ribbon graph. We have a homeomorphism of extended three manifolds with colored ribbon graphs

$$P(x) \cup_{\mathrm{id}} Q(y) \xrightarrow{\sim} W(x, y).$$

Comparing  $P(x) \cup_{id} Q(y)$  with the three manifold  $M \sqcup_{id} - N$  as considered in the proof of theorem 10.4.1 in [29], we obtain the desired presentation.

Now we want to use this to provide a presentation of the induced duality pairing on  $Z_{\mathcal{V}}$ . Again, it should be stressed that this presentation is only valid with respect to certain parametrizations.

**Proposition 11.7** (Surgery presentation of  $(\cdot, \cdot)_{\Sigma}$ ). Let  $\Sigma$  be a connected *I*-labeled marked surface. For any parametrization  $f : \Sigma_t \to \overline{\mathcal{G}(\Sigma)}$  there is a parametrization  $u(f) : \Sigma_{t*} \to \overline{\mathcal{G}(-\Sigma)}$  such that with respect to the induced isomorphisms

$$\Psi(t) \simeq Z(\mathbf{\Sigma}),$$
  
$$\Psi(t^*) \simeq Z(-\mathbf{\Sigma}),$$

we have the following presentation of  $(\cdot, \cdot)_{\Sigma}$ .



*Proof.* Choose a parametrization  $f: \Sigma_t \to \overline{\mathcal{G}(\Sigma)}$ . Consider the induced parametrization  $y(f) = (-f) \circ s: \Sigma_{-t} \to -\overline{\mathcal{G}(\Sigma)}$  as in the previous proof. This provides isomorphisms

(11.11)  $\bigoplus_{i \in I^g} \operatorname{Hom}(\mathbf{1}, \Phi(t, i)) \simeq \mathcal{T}^e(\overline{\mathcal{G}(\Sigma)}),$ 

(11.12) 
$$\oplus_{i \in I^g} \operatorname{Hom}(\mathbf{1}, \Phi(-t, i)) \simeq \mathcal{T}^e(-\mathcal{G}(\Sigma))$$

Recall the isomorphism

(11.13) 
$$\mathcal{T}^{e}(r) \circ \tilde{h} \circ \dot{q} : \mathcal{T}^{e}(\overline{\mathcal{G}(-\Sigma)}) \xrightarrow{\sim} \mathcal{T}^{e}(-\overline{\mathcal{G}(\Sigma)}).$$

We can define r such that the half-twists cancel with those of s. More precisely there is a choice of convention such that the following holds. Let  $\tilde{s}$  be the same as

s but without the twists at the arcs. That is,  $\tilde{s}$  is only the reflection in the y = 0 plane. Observe that f induces a parametrization

$$(-f) \circ \tilde{s} : \Sigma_{t*} \to \overline{\mathcal{G}(-\Sigma)}.$$

Take  $u(f) = (-f) \circ \tilde{s}$ . We see that with respect to the parametrizations u(f) and y(f), the isomorphism

$$\zeta: \Psi(t*) \xrightarrow{\sim} \Psi(-t)$$

is given by post composing with q. Now combine the description of the pairing  $\Psi(t) \otimes \Psi(-t) \to K$  given in the previous proposition with the description of  $\tilde{h} \circ \dot{q}$  as post composition with q in each factor to obtain the desired presentation.  $\Box$ 

We show next that the pairing is compatible with glueing. That is, we prove that the formula holds and we explicitly calculate the  $\mu'_{\lambda}$ s. This calculation will depend on whether or not the two points subject to glueing are on the same connected component or not.

**Proposition 11.8.** Let  $\Sigma$  be a connected I-labeled marked surface obtained from glueing two points subject to glueing that lie on the same component. Consider the glueing isomorphism

$$\tilde{g}: \bigoplus_{i \in I} Z(\boldsymbol{\Sigma}(i)) \overset{\sim}{\longrightarrow} Z(\boldsymbol{\Sigma}),$$

as described in definition 4.1. We have

$$(\tilde{g}, \tilde{g})_{\Sigma} = \sum_{i \in I} D^4 dim(i)^{-1} (\cdot, \cdot)_{\Sigma(i)}.$$

Proof. We will use the description of the glueing homomorphism given in section 11.3 above. To do this we need to compare two parametrizations of  $-\overline{\mathcal{G}(\Sigma)}$ . Recall that to use (11.6) in general, we start with a parametrization  $f: \Sigma_t \to \overline{\mathcal{G}(\Sigma(i))}$  and provide a parametrization  $F: \Sigma_{t'} \to \overline{\mathcal{G}(\Sigma)}$  that agrees with f away from the points subject to glueing. Then (11.6) holds with respect to the isomorphisms (11.7),(11.8). We will write F = z(f). Recall that (11.6) is only valid with respect to a pair of parametrizations (r, y(r)). In order to use (11.6) for the pairing on  $\overline{\mathcal{G}(\Sigma)}$ , we use the pair of parametrizations (z(f), y(z(f))). For the pairing on  $\overline{\mathcal{G}(\Sigma)}$ , we use the pair (f, y(f)). Since (11.6) only holds with respect to isomorphisms induced by compatible parametrizations, we need to compare z(y(f)) with y(z(f)). We see that z(y(f)) is y(z(f)) followed by a Dehn twists at the attached handle. This implies that

(11.14) 
$$(k_l Y', z(y(f))) = (Y', y(z(f))),$$

for all  $Y' \in \text{Hom}(\mathbf{1}, \Phi(-t', l)) \subset \Psi(-t')$ , where l is such that the cap corresponding to the points subject to glueing is colored with  $V_l$  or  $V_{l^*}$ . To verify (11.14) recall the description of how to pass from one parametrization to another given in section 7, and use that a Dehn twists followed by a reflection in y = 0 is the same as the same reflection followed by the reverse Dehn twist. Using proposition 11.7, equation (11.14) and equation (11.6), we see that with respect to the isomorphisms  $\Psi(t) \simeq Z(\Sigma(i))$  and  $\Psi(-t) \simeq \mathcal{T}^e(-\overline{\mathcal{G}(\Sigma(i))})$  induced by the pair of parametrizations (f, y(f)), we have the following presentation of  $(\tilde{g}_i, k_l \tilde{g}_{l^*})$ 



Here the blue unknot's are the surgery link components. For the sake of notational simplicity we have assumed that

$$t = (0; (V_r, 1), (V_s, 1), (V_i, 1), (V_{i^*}, 1).$$

The proof in the general case is easily obtained from this, as it relies on a local argument involving the surgery links. As in the proof of theorem 10.4.1, we get the following local equality



Thus this pairing is zero unless  $l^* = i^*$ . If so, we conclude that the claimed equation holds.

**Proposition 11.9.** Let  $\Sigma$  be a connected *I*-labeled marked surface obtained from glueing two points subject to glueing that lie on two distinct components. Consider the glueing isomorphism

$$\tilde{g}: \bigoplus_{i \in I} Z(\Sigma(i)) \stackrel{\sim}{\longrightarrow} Z(\Sigma),$$

as described in definition 4.1. We have that

$$(\tilde{g}, \tilde{g})_{\Sigma} = \sum_{i \in I} \dim(i)^{-1} (\cdot, \cdot)_{\Sigma(i)}.$$

Proof. Let  $\Sigma_1$ , be the component containing the first point and let  $\Sigma_2$  be the component containing the second point. We may assume that both of these components are homeomorphic to spheres. To see this, let  $x \in Z(\Sigma_1)$  and let  $y \in Z(\Sigma_2)$ . We want to compare  $\langle x, y \rangle$  with  $\langle \tilde{g}(x), \tilde{g}(y) \rangle$ . We can reduce the genus by 1 on one of the components by factorization. That is, assume  $\Sigma_1 \sqcup \Sigma_2$  is obtained from  $\tilde{\Sigma}$ by glueing, where the glueing increase the genus. That is, the points subject to glueing lie on the same component. Then we may assume  $x = h(\tilde{x})$  and  $y = h(\tilde{y})$ , where h is the glueing homomorphism. Now the task is to identify a scalar  $\lambda$  such that

$$(h(\tilde{x}), h(\tilde{y})) = \lambda(\tilde{g} \circ h(\tilde{x}), \tilde{g} \circ h(\tilde{y})).$$

But we already know from the previous proposition, that there is a  $C \in K^*$  such that

$$(h(\tilde{x}), h(\tilde{y})) = C(\tilde{x}, \tilde{y})$$

and

$$(\tilde{g} \circ h(\tilde{x}), \tilde{g} \circ h(\tilde{y})) = (h \circ \tilde{g}(\tilde{x}), h \circ \tilde{g}(\tilde{y})) = C(\tilde{g}(\tilde{x}), \tilde{g}(\tilde{y})).$$

In the last equation we use that the pairing is compatible with morphisms and that glueing is associative. We now see that it will suffice to find  $\lambda$  such that

$$\lambda(\tilde{g}(\tilde{x}), \tilde{g}(\tilde{y})) = (\tilde{x}, \tilde{y}).$$

We have reduced the genus by 1 and can proceed inductively. Thus we can assume that we deal with spheres. Thus we can use the description of the glueing given above in section 11.3. For the sake of notational simplicity, we illustrate the case, where we have two spheres with three marked points. As in the previous proposition one starts by observing  $y(z'(f \otimes g))$  followed by a Dehn twists is  $z'(y(f) \otimes y(g))$ . This will allow us to adopt the same strategy. We consider  $\langle \tilde{g}_i(X \otimes Y), k_l \tilde{g}_{l^*}(X' \otimes Y') \rangle_{\Sigma}$ . The following presentations shows that the given presentation above naturally factors as a composition  $P(X, X') \circ Q(Y, Y')$  where P(X, X') is an element of  $\operatorname{Hom}(V_i \otimes V_l^*, \mathbf{1})$  and  $Q(Y, Y') \in \operatorname{Hom}(\mathbf{1}, V_i \otimes V_l^*)$ .



Now the orthogonality follows from the fact that  $\operatorname{Hom}(V_i \otimes V_l^*, \mathbf{1})$  is 0 if  $l \neq i$  and isomorphic to K otherwise. For l = i we note the following equation that holds for all  $f \in \operatorname{Hom}(V_i \otimes V_l^*, \mathbf{1})$ 



Applying this to P(X, X') and taking into account the twist that occurs when applying the isotopy to pull  $q_i$  to the right of  $q_s$ , we see that the claim holds.  $\Box$ 

**Corollary 11.10** (Compatibility of the pairing with glueing). The pairing  $(\cdot, \cdot)_{\Sigma}$  is compatible with glueing. It can be rescaled to a pairing  $\langle \cdot | \cdot \rangle_{\Sigma}$  according to topological types, such that

$$\langle \tilde{g} \mid \tilde{g} \rangle_{\Sigma} = \sum_{i \in I} \langle \cdot \mid \cdot \rangle_{\Sigma(i)}.$$

It is easily verified that the following normalization has the given property. Since the pairing is multiplicative with respect to disjoint union, it is enough to specify the normalization on connected  $\Sigma$ . Assume therefore that  $\Sigma$  is of genus g with labels  $i_1, ..., i_k$ . Then the normalization is given by

(11.15) 
$$\langle \cdot | \cdot \rangle_{\Sigma} = \left( D^{-4g} \prod_{l=1}^{k} \sqrt{\dim(i_l)} \right) (\cdot, \cdot)_{\Sigma}$$

It only remains to prove that the pairing is compatible with orientation reversal.

**Proposition 11.11** (Compatibility with orientation reversal). The two pairings  $\langle \cdot | \cdot \rangle$  and  $(\cdot, \cdot)$  are both compatible with orientation reversal.

Proof. Since the normalization factor is the same for  $\Sigma$  and  $-\Sigma$ , we see, that it is enough to consider  $(\cdot, \cdot)$ . For  $v \in Z(\Sigma)$  and  $w \in Z(-\Sigma)$  we want to find a scalar  $\mu$  such that  $\mu(v, w)_{\Sigma} = (w, v)_{-\Sigma}$ . For the moment let  $\Sigma'$  be an extended surface. Recall that in order to use the presentation of the pairing as given in proposition (11.6), we choose a parametrization f of  $\Sigma'$  and then we constructed a parametrization  $y(f) := (-f) \circ s$ . These give isomorphisms  $\mathcal{T}^e(\Sigma') \simeq \Psi(t_0)$ and  $\mathcal{T}^e(-\Sigma') \simeq \Psi(-t_0)$ . With respect to these parametrizations we can use the presentation of (11.6). For  $x \in \Psi(t_0)$  and  $y \in \Psi(-t_0)$  it is an easy exercise to verify

$$\langle x, y \rangle_{t_0} = \langle y, x \rangle_{-t_0}.$$

This identity is also necessary for self-duality, because if we take y(f) as the parametrization  $\Sigma_{-t_0} \to \Sigma'$  then we see that y(y(f)) = f. This follows from the fact that  $s^2 = id$ , which can be seen from the fact that counter clockwise

half twists at the arcs, followed by a reflection in the y = 0 plane is the same as a reflection in the y = 0 plane followed by clockwise half twists at the arcs. Now choose a parametrization  $f : \Sigma_t \to \overline{\mathcal{G}}(\Sigma)$ . This induces a parametrization  $u(f) = (-f) \circ \tilde{s} : \Sigma_{t*} \to \overline{\mathcal{G}}(-\Sigma)$ . Here  $\tilde{s}$  is simply the reflection in the y-plane. With respect to these isomorphisms we see, that  $\langle \cdot, \cdot \rangle_{\Sigma}$  is given as  $\langle \cdot, \dot{q} \rangle_t$ , where  $\dot{q}$  is given by post composing suitably in each factor of the tensor product. Now choose  $g = u(f) : \Sigma_{t*} \to \overline{\mathcal{G}}(-\Sigma)$ . Observe u(g) = f. Thus for  $(v, w) \in \Psi(t) \times \Psi(t*)$ we simply need to compare  $\langle v, \dot{q}(w) \rangle_t$  with  $\langle w, \dot{q}(v) \rangle_{t*}$ .

Assume the labeled marked points of  $\Sigma$  are  $i_1, ..., i_k$ . Let  $\mu(i) \in K^*$  be defined by the following equation



Let  $\mu = \mu_{i_1} \cdots \mu_{i_k}$ . Recall the fact that  $\langle v, \dot{q}(w) \rangle_t = \langle \dot{q}(w), v \rangle_{-t}$ . Now use the surgery presentation given in proposition (11.6). In the presentation of  $\langle \dot{q}(w), v \rangle_{-t}$  pull over the coupons colored with  $q_{i_l}$  from left to right to obtain

$$(v,w)_{\Sigma} = \mu(w,v)_{-\Sigma},$$

which finishes the proof.

**Remark 11.12.** We observe that if  $i \neq i^*$ , it is possible to consistently choose  $q_i$  and  $q_{i^*}$ , such that  $\mu(i)$  and  $\mu(i^*)$  takes any values, as long as  $\mu(i)\mu(i^*) = 1$ . This follows from the fact that turning a coupon upside down, and then turning the resulting morphism upside down will yield the original morphism. Call this operation F. Then the equation above reads  $F(q_i) = \mu(i)q_{i^*}$ . Similarly, it can be seen that if  $i^* = i$ , then we must have  $\mu(i)^2 = 1$ . Using the axioms for the unit object of a modular tensor category, it is also easily seen that  $\mu(0) = 1$ .

We note the following result, that allow us to define  $\mu$  on the self-dual objects independently of q.

**Proposition 11.13** ( $\mu$  is well defined on self-dual objects). Assume that  $i \in I$  satisfies  $i = i^*$ . Then  $\mu(i)$  is independent of  $q_i$ .

The fact that  $\mu(i)$  might be -1 for  $i = i^*$  leads us to consider the strict self-duality question in section 16, where we introduce a new algebraic concept associated to a modular tensor category. As will be clear below, this will in many cases produce a very interesting normalization of the duality pairing, that will be strictly self-dual.

## 12 Unitarity

Consider a complex vector space W with scalar multiplication  $(\lambda, w) \mapsto \lambda . w$  Let  $\overline{W}$  be the complex vector space with the same underlying Abelian group and scalar multiplication given by  $(\lambda, w) \mapsto \overline{\lambda} . w$ . Here  $\overline{\lambda}$  is the complex conjugate of  $\lambda$ .

**Definition 12.1** (Unitarity). Let (V, g) be a modular functor based on  $\Lambda$  and  $\mathbb{C}$ . A unitary structure on V is a positive definite hermitian form

$$(\,\cdot\,,\,\cdot\,)_{\Sigma}:V(\Sigma)\otimes V(\Sigma)\to\mathbb{C},$$

subject to the following axioms.

Naturality. Let  ${\bf f}=(f,s):\Sigma_1\to\Sigma_2$  be a morphism between A-labeled marked surfaces. Then

(12.1) 
$$(V(f), V(f))_{\Sigma_2} = (\cdot, \cdot)_{\Sigma_1}.$$

Compatibility with disjoint union. Consider a disjoint union of  $\Lambda$ -labeled marked surface  $\Sigma = \Sigma_1 \sqcup \Sigma_2$ . Composing with the permutation of the factors, the modular functor V provides an isomorphism

$$\eta: V(\mathbf{\Sigma}) \otimes \overline{V(\mathbf{\Sigma})} \xrightarrow{\sim} V(\mathbf{\Sigma}_1) \otimes \overline{V(\mathbf{\Sigma}_1)} \otimes V(\mathbf{\Sigma}_2) \otimes \overline{V(\mathbf{\Sigma}_2)}.$$

We demand that with respect to the natural isomorphism  $\mathbb{C}\otimes\mathbb{C}\simeq\mathbb{C}$  we have that

(12.2) 
$$(\cdot, \cdot)_{\Sigma} = ((\cdot, \cdot)_{\Sigma_1} \otimes (\cdot, \cdot)_{\Sigma_2}) \circ \eta.$$

Compatibility with glueing. Let  $\Sigma$  be a  $\Lambda$  labeled marked surface obtained from glueing. Consider the glueing isomorphism

$$g: \bigoplus_{\lambda \in \Lambda} V(\mathbf{\Sigma}(\lambda)) \xrightarrow{\sim} V(\mathbf{\Sigma}),$$

as described in the definition of a modular functor. Clearly g also induces an isomorphism

$$g: \bigoplus_{\lambda \in \Lambda} \overline{V(\Sigma(\lambda))} \xrightarrow{\sim} \overline{V(\Sigma)}.$$

We have

(12.3) 
$$(g,g)_{\Sigma} = \sum_{\lambda \in \Lambda} \mu_{\lambda} (\cdot, \cdot)_{\Sigma(\lambda)},$$

where  $\mu_{\lambda} \in \mathbb{R}_{>0}$  for all  $\lambda$ . We allow the  $\mu_{\lambda}$  to depend on the isomorphism class of  $(\Sigma, (p, q))$ .

If the modular functor (V, g) also has a duality pairing we demand the unitary structure and the duality is compatible in the following sense.

Compatibility with duality. For all labeled marked surfaces  $\Sigma$ , we demand that the following diagram is commutative up to a scalar  $\rho(\Sigma)$  depending only on the isomorphism class of  $\Sigma$ 

Here, the horizontal isomorphisms are induced by the duality pairing, whereas the vertical isomorphisms are induced by the unitary structure.

We now make explicit what the isomorphisms of the diagram (12.4) are. We start with the composition

$$\omega: V(\mathbf{\Sigma}) \xrightarrow{\simeq} V(-\mathbf{\Sigma})^* \xrightarrow{\simeq} \overline{V(-\mathbf{\Sigma})}.$$

Let  $\langle\,\cdot\,,\,\cdot\,\rangle$  be the duality pairing and  $(\,\cdot\,,\,\cdot\,)$  be the Hermitian form. The first map is given by

$$V(\mathbf{\Sigma}) \ni f \mapsto \langle \, \cdot \,, f \rangle_{-\mathbf{\Sigma}} : V(-\mathbf{\Sigma}) \to \mathbb{C}.$$

The second map is the inverse of the linear isomorphism  $\overline{V(-\Sigma)} \xrightarrow{\simeq} V(-\Sigma)^*$  given by

$$\overline{V(-\Sigma)} \ni u \mapsto (\,\cdot\,, u)_{-\Sigma} : V(-\Sigma) \to \mathbb{C}.$$

Thus  $\omega(f)$  is defined by

(12.5) 
$$\langle x, f \rangle_{-\Sigma} = (x, \omega(f))_{-\Sigma},$$

for all x in  $V(-\Sigma)$ . We now consider the composition

$$\phi: V(\mathbf{\Sigma}) \xrightarrow{\simeq} \overline{V(\mathbf{\Sigma})^*} \xrightarrow{\simeq} \overline{V(-\mathbf{\Sigma})}.$$

The first is the linear map

$$V(\Sigma) \ni f \mapsto (\cdot, f)_{-\Sigma} : V(\Sigma) \to \mathbb{C}.$$

The second map is the inverse of the linear isomorphism  $\overline{V(-\Sigma)} \xrightarrow{\simeq} \overline{V(\Sigma)^*}$  given by

$$\overline{V(-\Sigma)} \ni u \mapsto \langle \cdot, u \rangle_{-\Sigma} : V(\Sigma) \to \mathbb{C}.$$

Thus  $\phi(f)$  is defined by

(12.6) 
$$\langle y, \phi(f) \rangle_{\Sigma} = (y, f)_{\Sigma},$$

for all y in  $V(\Sigma)$ .

Projective commutativity of (12.4) can be reformulated as the existence of  $\rho(\mathbf{\Sigma})$  in  $\mathbb{C}$  with

(12.7) 
$$\phi = \rho(\Sigma)\omega$$

# 13 Unitary structure from a unitary MTC

Recall the definition of a unitary modular tensor category  $(\mathcal{V}, (V_i)_{i \in I})$  with conjugation  $f \mapsto \overline{f}$  as defined in section 5.5. of chapter V in [29]. Recall that  $K = \mathbb{C}$ in this case. Assume we are given a unitary modular tensor category. For an *e*-surface  $\Sigma$  let  $(\cdot, \cdot)_{\Sigma}$  be the Hermitian form on  $\mathcal{T}^e(\Sigma)$  as defined in section 10 of chapter IV in [29].

**Theorem 13.1** (Unitarity). Let  $(\mathcal{V}, (V_i)_{i \in I})$  be a unitary modular tensor category. Let  $\Sigma$  be an *I*-labeled marked surface. Consider the positive definite Hermitian form

$$(\cdot, \cdot)_{\Sigma} = (\cdot, \cdot)_{\overline{\mathcal{G}(\Sigma)}}.$$

This defines a unitary structure on  $Z_{\mathcal{V}}$  compatible with duality.

*Proof.* It is proven by Turaev, that the induced Hermitian form is natural with respect to weak *e*-morphisms, and that it is multiplicative with respect to disjoint union. As Turaev also proves that  $\overline{\Delta^{-1}D} = (\Delta^{-1}D)^{-1}$  these two properties carry over. All of this is proven in section 10 of chapter *IV* in [29].

Let us now prove that it is compatible with glueing. We first consider the case where the two points lies on the same component. Since the glueing as well as the Hermitian form is multiplicative with respect to disjoint union, as well as natural with respect to morphisms, we may assume that we are in the situation described in section 11.3. We adopt the the notation form the first subsection of this section. It follows directly from theorem 10.4.1 in section 10.4 of chapter

IV in [29] that if  $i \neq j$  then  $(\tilde{g}_i, \tilde{g}_j)_{\Sigma} = 0$ . Let  $x, y \in \text{Hom}(\mathbf{1}, \Phi(t, j)) \subset Z(\Sigma(i))$ . Using theorem 10.4.1, linearity of tr and  $\mathbb{C} \simeq \text{End}(V_{i^*})$ , we get that

$$(\tilde{g}_i(x), \tilde{g}_i(y))_{\Sigma} = D^g \left( \dim(i) \prod_{c=1}^g \dim(j_c) \right)^{-1} \operatorname{tr}(\tilde{g}_i(x) \circ \overline{\tilde{g}_i(y)}).$$

Here g is the genus of  $\Sigma$ . Unwinding the glueing formula and using properties of the conjugation as well as of the trace we get that

$$D^{g}((\dim(i)\dim(j))^{-1}\operatorname{tr}((1_{W}\otimes q_{i}\otimes 1_{R})\circ x\circ \overline{y}\circ (1_{W}\otimes \overline{q_{i}}\otimes 1_{R})))$$
  
=  $D^{g}((\dim(i)\dim(j))^{-1}\operatorname{tr}((1_{W}\otimes \overline{q_{i}}\circ q_{i}\otimes 1_{R})\circ x\circ \overline{y}))$   
=  $D\dim(i)^{-1}\lambda_{i}(x,y)_{\Sigma(\mathbf{i})}.$ 

Here  $\lambda_i \in \mathbb{C}$  is defined by

(13.1) 
$$\lambda_i 1_{V_{i^*}} = \overline{q_i} \circ q_i.$$

Thus we get that

(13.2) 
$$(\tilde{g}, \tilde{g})_{\Sigma} = \sum_{i \in I} D \dim(i)^{-1} \lambda_i(\cdot, \cdot)_{\Sigma(i)}$$

We now consider the case where the two points subject to glueing lie on distinct components. Using the result above, we may assume that these are homeomorphic to spheres. This can be argued as in the proof of Proposition (11.9). Using naturality and multiplicativity of both the glueing and the Hermitian form, we may assume that we are in the situation of the second subsection in section 11.3. An argument based on the surgery presentation of the form given in the proof of theorem 10.4.1 and based on the ideas of section 10.6 will show that in this case, we have that

(13.3) 
$$(\tilde{g}, \tilde{g})_{\Sigma} = \sum_{i \in I} \dim(i)^{-1} \lambda_i(\cdot, \cdot)_{\Sigma(i)}.$$

Finally we prove that the unitary structure is compatible with duality. This is done by considering a surgery presentations of the equations (12.5) and (12.6). Let  $f \in Z(\Sigma)$ . We may assume  $\Sigma$  is connected. Equation (12.5) is presented as an equation involving  $\overline{\omega(f)}$  and equation (12.6) is presented as an equation involving  $\phi(f)$ . Conjugating the surgery presentation of (12.5) we see that  $\phi(f) = \frac{1}{\sigma(i_1)\cdots\sigma(i_m)}\omega(f)$  where  $\sigma(i)$  is defined as  $\lambda_{i^*}\mu(i)$ . Thus, if  $\Sigma$  is an *I*-labeled marked surface (not necessarily connected) with labels  $i_1, \ldots, i_m$  we see that

(13.4) 
$$\rho(\Sigma) = \prod_{l=1}^{m} \sigma(i_l)^{-1},$$

which concludes the proof.

## 14 Scaling of the duality, unitarity and glueing

#### 14.1 A choice of isomorphisms $q_i$

For the remainder of section 14,15, and 16 we fix a choice  $q := (q_i)_{i \in I}$  where  $q_i : V_{i^*} \to V_i^*$  is an isomorphism. Recall the definition of  $\mu$  c.f. proposition 11.13. We can and will assume that  $\mu(i) = 1$  for all i with  $i \neq i^*$ . In addition, if  $\mathcal{V}$  is unitary, we can and will assume that  $\overline{q_i} \circ q_i = \operatorname{id}_{V_i^*}$ . Note however that all results from section 14.2 are true independently of these two extra assumptions. Let  $k_i$  be the twist coefficients of the modular tensor category. For the remainder of this article we fix for all i a choice of  $\sqrt{\dim(i)}$  and a choice of  $\sqrt{k_i}$ . We make these choices invariant under  $i \mapsto i^*$ . If  $K = \mathbb{C}$  and  $\dim(i)$  is positive, we of course choose the positive square root. This will be the case if  $\mathcal{V}$  is assumed to be unitary. We recall that if this is the case then  $k_i \in S^1$ .

The choice q gives us a modular functor with duality  $(Z, \tilde{g}, \langle \cdot, \cdot \rangle)$ . This is the content of theorem 5.8 and theorem 11.4. It serves as a set of reference isomorphisms, when dealing with scalings.

#### 14.2 The scaling analysis

As mentioned above, both the glueing  $\tilde{g}$  and the pairing  $\langle \cdot, \cdot \rangle$  depends on our choice of isomorphisms  $V_{i^*} \xrightarrow{\longrightarrow} V_i^*, i \in I$ . As we are considering isomorphisms between simple objects, any other choice  $\hat{q}_i$  will be of the form  $\hat{q}_i = u_i q_i$  for some  $u_i \in K^*$ , where  $K^*$  denotes the group of units in K. We now investigate how the different compatibilities of the modular functor with duality are affected by scaling. For  $u \in K^{*I}$  let  $q(u) := (u_i q_i)_{i \in I}$ . Let  $\tilde{g}_u$  be the glueing defined using q(u). To be precise, this means that in equation (5.1) we use  $u_i q_i$  instead of  $q_i$ . Let  $\langle \cdot, \cdot \rangle^u$  be the pairing defined using q(u). To be precise, this means that in the equation (11.5) we use  $u_i q_i$  instead of  $q_i$ . We observe that if  $\Sigma$  has labels  $i_1, ..., i_m$  then we have that

(14.1) 
$$\langle \cdot, \cdot \rangle_{\Sigma}^{u} = \left(\prod_{l=1}^{m} u_{i_l}\right) \langle \cdot, \cdot \rangle_{\Sigma}.$$

We will write

$$\langle \cdot, \cdot \rangle^u_{\Sigma} = u(\Sigma) \langle \cdot, \cdot \rangle_{\Sigma}.$$

Let  $u, w \in K^{*I}$ . Below we will consider what happens, if we use q(u) to define the glueing, and q(w) to define the pairing.

**Definition 14.1** (Genus normalized pairing). Let  $w \in K^{*I}$ . For a surface of genus g we consider the following normalization

$$\langle \cdot, \cdot \rangle_{*,\Sigma}^w := D^{-4g} \langle \cdot, \cdot \rangle^w.$$

Consider a general modular functor V with label set I. Assume V has a duality pairing  $\langle \cdot, \cdot \rangle$ . Consider  $S^2$  equipped with the Stokes orientation, where  $B^3$  is given the RHS orientation. Let  $(S^2, i, j)$  have the northpole colored with i and the southpole colored with j. There is a natural isotopy to the standard decorated surface of type  $(0; (V_i, 1), (V_j, 1))$ . This induces an isomorphism  $Z(S^2, i, j) \simeq \text{Hom}(1, V_i \otimes V_j)$ . Let  $\omega(i)$  be the unique vector in  $Z(S^2, i, i^*)$  that solves  $\tilde{g}(\omega(i) \otimes \omega(i)) = \omega(i)$ . Let  $\zeta(i) \in Z(-(S^2, i, i^*))$  be the unique vector that solves the analogous glueing problem. We define

(14.2) 
$$Z(i) := \langle \omega(i), \zeta(i) \rangle_{(\mathbf{S}^2, i, i^*)}.$$

We now return to  $Z_{\mathcal{V}}$ .

**Proposition 14.2.** Under the isomorphism  $Hom(\mathbf{1}, V_i \otimes V_{i^*}) \simeq Z(S^2, i, i^*)$  induced by the identity parametrization we see that  $\omega(i)$  is given as



**Proposition 14.3.** We have that

(14.3)  $\langle \omega(i), \zeta(i) \rangle = k_i^{-1} \dim(i).$ 

We start by observing that if  $\Sigma$  has labels  $i_1, ..., i_m$  then we have

(14.4) 
$$\langle \cdot, \cdot \rangle_{\Sigma}^{w} = \left(\prod_{l=1}^{m} w_{i_{l}}\right) \langle \cdot, \cdot \rangle_{\Sigma}$$

We will write

$$\langle \cdot, \cdot \rangle_{\Sigma}^{w} = w(\Sigma) \langle \cdot, \cdot \rangle_{\Sigma}.$$

**Proposition 14.4.** Assume  $i, i^*$  label points on the same component. Then we have that

(14.5) 
$$\langle \tilde{g}_u^i, \tilde{g}_u^{i^*} \rangle_{\boldsymbol{\Sigma}}^w = \frac{u_i u_{i^*}}{w_i w_{i^*}} D^4 \dim(i)^{-1} \langle \cdot, \cdot \rangle_{\boldsymbol{\Sigma}(i)}^w.$$

Assume i, i<sup>\*</sup> label points on distinct components. Then we have that

(14.6) 
$$\langle \tilde{g}_{u}^{i}, \tilde{g}_{u}^{i^{*}} \rangle_{\Sigma}^{w} = \frac{u_{i}u_{i^{*}}}{w_{i}w_{i^{*}}} dim(i)^{-1} \langle \cdot, \cdot \rangle_{\Sigma(i)}^{w}.$$

We want to know how scaling affects the self-duality scalar  $\mu$ .

Proposition 14.5. We have that

(14.7) 
$$\mu(i,w) = \frac{w_i}{w_{i^*}}\mu(i).$$

Let  $\omega(i, u) \in Z(S^2, i, i^*)$  be the unique vector that solves the analogous equation  $\tilde{g}_u(\omega(i, u) \otimes \omega(i, u)) = \omega(i, u)$ . Then

(14.8) 
$$\omega(i,u) = u_i^{-1}\omega(i)$$

Let  $\zeta(i)\in Z(-(S^2,i,i^*))=Z(-S^2,i^*,i)$  be the image of  $\omega(i,u)$  under Z(r). Again we have that

(14.9) 
$$\zeta(i, u) = u_i^{-1} \zeta(i).$$

We now investigate how this affect the compatibility of the unitary structure with glueing.

**Proposition 14.6.** Let  $\Sigma$  be a connected *I*-labeled marked surface obtained from glueing two points subject to glueing that lie on one and the same component. Consider the glueing isomorphism

$$\tilde{g}_u: \bigoplus_{i \in I} Z(\Sigma(i)) \xrightarrow{\sim} Z(\Sigma).$$

We have that

(14.10) 
$$(\tilde{g}_u, \tilde{g}_u)_{\mathbf{\Sigma}} = \sum_{i \in I} D^4 \dim(i)^{-1} \lambda_i u_i \overline{u_i} (\cdot, \cdot)_{\mathbf{\Sigma}(i)} .$$

**Proposition 14.7.** Let  $\Sigma$  be a connected *I*-labeled marked surface obtained from glueing two points subject to glueing that lie on two distinct components. Consider the glueing isomorphism

$$\widetilde{g}_u: \bigoplus_{i \in I} Z(\Sigma(i)) \xrightarrow{\sim} Z(\Sigma),$$

We have that

(14.11) 
$$(\tilde{g}_u, \tilde{g}_u)_{\Sigma} = \sum_{i \in I} \dim(i)^{-1} \lambda_i u_i \overline{u_i} (\cdot, \cdot)_{\Sigma(i)}.$$

### 14.3 Normalizations

For  $u, w \in K^{*I}$  we now consider the compatibility of the glueing  $\tilde{g}_u$  with the pairing  $\langle \cdot, \cdot \rangle^w$ .

Let  $\langle \cdot | \cdot \rangle^{u,w}$  be the normalized pairing given by

(14.12) 
$$\langle \cdot | \cdot \rangle_{\Sigma_g, i_1, \dots, i_k}^{u, w} = \left( D^{-4g} \prod_{l=1}^k s_{i_l}^{u, w} \right) \langle \cdot , \cdot \rangle_{\Sigma_g, i_1, \dots, i_k}^{w}$$

where

(14.13) 
$$s_i^{u,w} := \frac{\sqrt{w_i w_{i^*}} \sqrt{\dim(i)}}{\sqrt{u_i u_{i^*}}}.$$

From 14.4 we immediately get the following

**Corollary 14.8.** The pairing  $\langle \cdot | \cdot \rangle^{u,w}$  is strictly compatible with the glueing  $\tilde{g}_u$ . That is

(14.14) 
$$\langle \tilde{g}_u^i \mid \tilde{g}_u^{i^*} \rangle_{\Sigma}^{u,w} = \langle \cdot \mid \cdot \rangle_{\Sigma(\mathbf{i})}^{u,w}$$

We will write

$$\langle \cdot | \cdot \rangle^{u,w}_{\Sigma_g, i_1, \dots, i_k} := s(\mathbf{\Sigma}, u, w) \langle \cdot, \cdot \rangle.$$

If we want to stress the choice of square roots chosen for  $w_i w_i^*$  and  $u_i u_i^*$  we will write

$$\langle \cdot | \cdot \rangle^{u,w,S}.$$

Observe that

$$s(\boldsymbol{\Sigma}, \boldsymbol{u}, \boldsymbol{w}) = \left( D^{-4g} \prod_{l=1}^k s_{i_l}^{\boldsymbol{u}, \boldsymbol{w}} w_{i_l} \right),$$

when  $\Sigma = (\Sigma_g, i_1, ..., i_k)$ . We now consider a normalization of the Hermitian form. This normalized Hermitian form will be strictly compatible with the glueing  $\tilde{g}_u$ . First we note that

(14.15) 
$$(\cdot | \cdot)^{\underline{u}}_{\Sigma} = \left( D^{-4g} \prod_{l=1}^{k} r^{\underline{u}}_{i_l} \right) (\cdot, \cdot)_{\Sigma},$$

where

(14.16) 
$$r_i^u = \frac{\sqrt{\dim(i)}}{\sqrt{\lambda_i u_i \overline{u_i}}}.$$

**Corollary 14.9.** The Hermitian form  $(\cdot | \cdot)^u$  is strictly compatible with the glueing  $\tilde{g}_u$ . That is

(14.17) 
$$(\tilde{g}_u^i \mid \tilde{g}_u^i)_{\Sigma}^u = (\cdot \mid \cdot)_{\Sigma(\mathbf{i})}^u.$$

Recall the convention that the square root of any positive number is assumed to be chosen positive. Thus there is no ambiguity in choosing the  $r_i$ . Of course we have to choose them positively, if we want the pairing to remain positive definite. We will write

$$(\cdot \mid \cdot)^{u}_{\Sigma} = r(\Sigma, u)(\cdot, \cdot)_{\Sigma}$$

**Proposition 14.10.** Assume  $\Sigma$  is an I-labeled marked surface (not necessarily connected) with labels  $i_1, ..., i_m$ . With respect to the normalized duality  $\langle \cdot | \cdot \rangle^{u,w}$  and the normalized Hermitian form  $(\cdot | \cdot)^u$  we have the following equation

(14.18) 
$$\rho_N^{u,w}(\mathbf{\Sigma}) = \left(r(\mathbf{\Sigma}, u)\overline{r(-\mathbf{\Sigma}, u)}\right) \left(s(\mathbf{\Sigma}, u, w)\overline{s(-\mathbf{\Sigma}, u, w)}\right)^{-1} \prod_{l=1}^m \sigma(i_l)^{-1}.$$

With respect to the genus normalized pairing  $\langle \cdot, \cdot \rangle^w_*$  we have the following equation

(14.19) 
$$\rho_{g,N}^{u,w}(\Sigma) = \prod_{l=1}^{m} \left( r_{i_l}^u r_{i_l^*}^u \right) \left( \sigma(i_l) w_{i_l} \overline{w_{i_l^*}} \right)^{-1}$$

Observe that since r is always real (and positive) we have  $\overline{r(-\Sigma, u)} = r(-\Sigma)$ .

# 15 The canonical symplectic rescaling

Assume in the following that K is an integral domain. Recall that in accordance with the assumptions made in section 14.1 we have  $\mu(i) = 1$  for all *i* with  $i \neq i^*$ . In this section, we only consider scalings  $u: I \to K^*$  that satisfies  $u_i = u_{i^*}$ . Recall the definition of the normalization coefficients

$$s_i^{u,w} := \frac{\sqrt{w_i w_{i^*}} \sqrt{\dim(i)}}{\sqrt{u_i u_{i^*}}}$$

If u, w are invariant under  $i \mapsto i^*$  we see that we have canonical square roots given by  $\sqrt{X^2} = X$ . In the following, the normalization coefficients shall be interpreted according to this.

**Definition 15.1** (Symplectic labels). Let  $i \in I$  satisfy  $i = i^*$ . We say that i is symplectic if  $\mu(i) = -1$ .

**Definition 15.2** (Symplectic multiplicity). Let  $\Sigma$  be a labeled marked surface. Let  $\nu(\Sigma)$  denote the number of marked points on  $\Sigma$  labeled with symplectic labels. We call this number the symplectic multiplicity.

**Theorem 15.3** (Canonical symplectic scaling). Choose  $u, w \in (K^*)^I$  that solves

$$(15.1) u_i = s_i^{u,w} w_i$$

for all i. Then the modular functor  $Z_{\mathcal{V}}$  with glueing  $\tilde{g}(u)$  and genus normalized duality  $\langle \cdot, \cdot \rangle_*^u$  satisfies that glueing and duality are strictly compatible and the duality is self-dual up to a sign which is given by the symplectic multiplicity

(15.2) 
$$\mu = (-1)^{\nu}$$

We have that

(15.3) 
$$Z(i) = \frac{\dim(i)}{k_i}$$

Moreover, any two solutions (u, w) and (u', w') results in modular functors with duality that are isomorphic through an isomorphism that preserves the duality pairing.

**Remark 15.4.** We emphasize that equation (15.1) means that one uses the same scaling for the glueing isomorphism as one uses in the duality paring.

Before commencing the proof, we observe that up to a sign there is a preferred solution given by choosing  $w_i = 1$  for all *i* and solving

$$\frac{\sqrt{\dim(i)}}{u_i} = u_i$$

If there is no such  $u_i$  we may formally add it. If (u, w) is a solution, we will write  $Z^u$  for the resulting modular functor with duality  $(Z, \tilde{g}(u), \langle \cdot, \cdot \rangle_*^u)$ .

Proof. Let (u,w) be a solution. The fact that this is a solution to (15.1) implies that

$$\langle \cdot, \cdot \rangle^u_* = \langle \cdot | \cdot \rangle^{u,w}.$$

Since the bracket on the left is strictly compatible with  $\tilde{g}(u)$  by corollary 14.8 the first claim follows. Equation (15.2) is an easy consequence of  $w_i = w_{i^*}$ , proposition 14.5 and the proof of proposition 11.11. Equation (15.3) follows from the fact that a proof of proposition 14.3 only depends on the fact that the same set of isomorphisms  $V_{i^*} \xrightarrow{\sim} V_i^*$  is used for the duality as well as for the glueing, and that we always have  $\mu(i, w)\mu(i^*, w) = 1$  for all  $w : I \to K^*$ . The proof of proposition 14.3 is a straightforward calculation of the surgery presentation given above.

Finally we prove that if we have two solutions, then they are isomorphic as modular functors through an isomorphism that preserves the duality. Consider a function  $\alpha : I \to K^*$ . Let  $\Sigma$  be a labeled marked surface with labels  $i_1, ..., i_k$ . Then  $\alpha$  induces an automorphism  $\Phi_{\alpha} = \Phi$ ,

$$\Phi(\mathbf{\Sigma}): Z(\mathbf{\Sigma}) \xrightarrow{\sim} Z(\mathbf{\Sigma}),$$

given by  $\Phi(\Sigma) = \left(\prod_{l=1}^{k} \alpha(i_l)\right) \operatorname{id}_{Z(\Sigma)}$ . Since this is multiplicative with respect to labels, it is easily seen that  $\Phi$  is compatible with the action induced by morphisms of labeled marked surfaces, disjoint union and the permutation.

We will think of  $\Phi$  as a morphism of modular functors  $Z^u \to Z^{u'}$ . We now identify sufficient conditions for  $\Phi$  to be compatible with glueing and with the duality pairings. We start with glueing. Let  $\Sigma(\lambda)$  be obtained by glueing  $\Sigma(\lambda, i, i^*)$ . Compatibility with glueing is equivalent to the following equation

$$\tilde{g}_{u'}^i \circ \Phi(\lambda, i, i^*) = \Phi(\lambda) \circ \tilde{g}_u^i$$

This is equivalent to

(15.4) 
$$\alpha(i)\alpha(i^*) = \frac{u_i}{u_i}.$$

For  $\Phi$  to be compatible with duality, we must have that

$$\langle \cdot, \cdot \rangle_{*,u} = \langle \Phi(\Sigma)(\cdot), \Phi(-\Sigma)(\cdot) \rangle_{*,u'}.$$

For this equation to be satisfied, we see that equation (15.4) is sufficient. If this is so, then  $\beta = \alpha^{-1} : I \to K^*$  will satisfy

$$\beta(i)\beta(i^*) = \frac{u_i'}{u_i}.$$

Therefore  $\Phi_{\beta}$  will be an inverse morphism that preserves the duality. Thus we can choose any function  $\alpha: I \to K^*$  that satisfies (15.4), and then  $\Phi = \Phi_{\alpha}$  will be an isomorphism  $Z^u \xrightarrow{\sim} Z^{u'}$  that preserves the duality pairing. That such a function exists follows from the fact that u, u' are both invariant under  $i \mapsto i^*$ .

### 15.1 Unitarity

Assume now that  $K = \mathbb{C}$  and that  $\mathcal{V}$  comes equipped with a unitary structure  $\operatorname{Hom}(V, W) \ni f \mapsto \overline{f} \in \operatorname{Hom}(W, V)$ . Recall that by section 14.1 we have  $\overline{q_i} \circ q_i = \operatorname{id}_{V_*^*}$ . Thus  $\lambda_i = 1$  for all i.

**Theorem 15.5.** Assume (u, w) is a solution to (15.1) with  $|w_i| = 1$  for all *i*. Then the following holds. Up to a sign the genus normalized duality pairing  $\langle \cdot, \cdot \rangle_{*,u}$  is compatible with the normalized Hermitian form  $(\cdot | \cdot)^u$ . This sign is given by the parity of the symplectic multiplicity

$$\rho = (-1)^{\nu}.$$

Moreover any two solutions (u, w) and (u', w') to (15.1) yields modular functors  $Z^{u}, Z^{u'}$  that are isomorphic through an isomorphism that respects the duality pairing as well as the Hermitian form.

*Proof.* Consider a labeled marked surface  $\Sigma$  with labels  $i_1, ..., i_k$ . Recall the following formula from proposition 14.10

$$\rho_{g,N}^{u,u}(\mathbf{\Sigma}) = \prod_{l=1}^{m} \left( r_{i_l}^u r_{i_l^*}^u \right) \left( \sigma(i_l) u_{i_l} \overline{u_{i_l^*}} \right)^{-1}.$$

Recall that  $\sigma(i) = \lambda_{i^*}\mu(i) = \mu(i)$ . So in our situation we see that the product of the  $\sigma'_i$ s is equal to  $\mu(\Sigma)$ , which we already know is given by  $(-1)^{\nu(\Sigma)}$ . Since  $u_i = u_i^*$  and  $\lambda_i = 1$  for all *i* we get

$$r_{i_l}^u r_{i_l^*}^u = \frac{\dim(i)}{|u_i|^2}.$$

Thus  $\rho/\mu$  is seen to be a product of factors of the form

$$\frac{\dim(i)}{|u_i|^4}.$$

The equation

$$u_i^2 = w_i^2 \sqrt{\dim(i)},$$

implies that all of these factors are 1. Here we use that  $|w_i| = 1$  for all *i*.

Assume now that (u', w') is another solution. We recall that the isomorphism  $Z^u \xrightarrow{\sim} Z^{u'}$  from theorem 15.3 can be constructed by choosing a suitable function  $\alpha: I \to \mathbb{C}$  with  $\alpha(i)\alpha(i^*) = u_i/u'_i$  for all *i*. For any labeled marked surface  $\Sigma$  the isomorphism

$$\Phi_{\alpha}: Z^u(\Sigma) \xrightarrow{\sim} Z^{u'}(\Sigma),$$

will be multiplication by  $\alpha(i_1) \cdots \alpha(i_k)$  where  $i_1, \ldots, i_k$  are the labels of  $\Sigma$ . However, since  $|w_i| = |w'_i| = 1$  for all i we see that  $|u_i| = |u'_i|$  for all i. This implies the following two things. First  $r_i^u = r_i^{u'}$  for all i. Second it implies that we can choose  $\alpha(i) = \alpha(i^*)$  to be a square root of  $u_i/u'_i$ , which lies on the unit circle. Therefore  $\Phi_{\alpha}$  will be a Hermitian isomorphism.

# 16 The dual of the fundamental group of a modular tensor category

Recall the definition of the dual of the fundamental group and a fundamental symplectic character of a modular tensor category given in the introduction.

**Theorem 16.1.** Assume that a modular tensor category  $(\mathcal{V}, I)$  has a fundamental symplectic character. Then there exists  $u : I \to K^*$  such that the genus normalized duality pairing  $\langle \cdot, \cdot \rangle^u_*$  is strictly self-dual, strictly compatible with glueing and we have that

$$Z(u,i) = \frac{\dim(i)}{k_i}.$$

Moreover if  $\mathcal{V}$  is unitary and the image of  $\tilde{\mu}$  above is a subset of

$$S(K) = \{z \in K | z\bar{z} = 1\}$$

then we can choose u, such that  $\langle \cdot, \cdot \rangle^u_*$  is strictly compatible with the Hermitian form  $(\cdot \mid \cdot)^u$ .

Before commencing the proof, we remark that if  $\mu(i) = 1$  for all self-dual objects *i*, then the neutral element  $e \in \Pi(\mathcal{V}, I)^*$  is such an extension.

*Proof.* Partition  $I = I_{SD} \sqcup I_{NSD}$  such that  $i \in I_{SD}$  if and only if  $i^* = i$ . We further have the natural splitting

$$I_{SD} = I_{SD}^+ \sqcup I_{SD}^-,$$

where  $i \in I_{SD}^+$  if and only if  $\mu(i) = 1$ . Hence we have of course that  $i \in J_{SD}^-$  if and only if  $\mu(i) = -1$ .

Let us now pick a splitting

$$I_{NSD} = I^1_{NSD} \sqcup I^2_{NSD}$$

such that  $i \in I^1_{NSD}$  if and only if  $i^* \in I^2_{NSD}$ .

We start by describing the normalization. Recall that the choice made in section 14.1 implies that  $\mu(i) = \mu(i^*) = 1$  whenever  $i \neq i^*$ . Let  $w : I \to K^*$ . We consider the genus normalized pairing  $\langle \cdot, \cdot \rangle_{w}^{w}$ . We have that

$$\mu(i, w) = \frac{w_i}{w_i^*} \mu(i).$$

Since  $\tilde{\mu}(i) = \tilde{\mu}(i^*)^{-1}$  this implies that we can consistently choose  $w_i, w_i^*$  such that  $\mu(i, w) = \tilde{\mu}(i)$  whenever *i* is not self-dual. Since  $\tilde{\mu}$  is assumed to extend the  $\mu$  on the self-dual objects we conclude that we can normalize such that  $\mu(i, w) = \tilde{\mu}(i)$ . Assume that we can choose *w* such that  $\langle \cdot | \cdot \rangle_*^w$  is also strictly compatible with glueing. We want to argue that in this case, we have  $\mu(\Sigma) = 1$  unless  $Z(\Sigma) = 0$ .

Let  $\Sigma$  be a labeled marked surface. We recall that proving strict self-duality is the same as proving that for all  $(x, y) \in Z(\Sigma) \times Z(-\Sigma)$  we have

$$\langle x, y \rangle_{\Sigma} = \langle y, x \rangle_{-\Sigma}.$$

Let C be a collection of simple closed curves on  $\Sigma$ , whose homology classes are contained in the Lagrangian subspace of  $\Sigma$  and such that factorization along all of these will produce a disjoint union of spheres with one, two or three marked points. For the existence of such a collection see [31]. Let  $\lambda \in I^C$  and let  $\Sigma_C(\lambda)$ be the labeled marked surface obtained from factorization in C. Thus  $\Sigma_C(\lambda)$  is a disjoint union of labeled marked surfaces of genus zero with one, two or three labels. Write  $\Sigma_C(\lambda) = \sqcup_{l=1}^k \mathbf{S}_l(\lambda)$ . Let  $P_{\lambda} : Z(\Sigma) \to Z(\Sigma_C(\lambda))$  be the projection resulting from the factorization isomorphism. Let  $x \in Z(\Sigma)$  and let  $y \in Z(-\Sigma)$ . We can write  $P_{\lambda}(x)$  as a finite sum

$$\sum_{lpha\in(x,\lambda)}x(lpha,\lambda)^{(1)}\otimes\cdots\otimes x(lpha,\lambda)^{(l)},$$

with  $x(\alpha, \lambda)^{(i)} \in Z(\mathbf{S}_i(\lambda))$ . Here  $(x, \lambda)$  is a finite index set depending only on x and  $\lambda$ . In a similar way we write  $P_{\lambda^*}(y)$  as a finite sum of the form

$$\sum_{eta \in (y,\lambda^*)} y(eta,\lambda^*)^{(1)} \otimes \cdots \otimes y(eta,\lambda^*)^{(l)}$$

Recalling the following identity  $-(\Sigma(\lambda)) = (-\Sigma)(\lambda^*)$ , we have that

$$\langle x, y \rangle_{\mathbf{\Sigma}} = \sum_{\lambda \in I^C} \langle P_{\lambda}(x), P_{\lambda^*}(y) \rangle_{\mathbf{\Sigma}_C(\lambda)}$$

$$= \sum_{\lambda \in I^C} \sum_{\alpha \in (x,\lambda), \beta \in (y,\lambda^*)} \prod_{l=1}^k \langle x(\alpha,\lambda)^{(l)}, y(\beta,\lambda^*)^{(l)} \rangle_{\mathbf{S}_l(\lambda)}.$$

Similarly we see that

$$\langle y, x \rangle_{-\Sigma} = \sum_{\lambda^* \in I^C} \sum_{\alpha \in (x,\lambda), \beta \in (y,\lambda^*)} \prod_{l=1}^k \langle y(\beta,\lambda^*)^{(l)}, x(\alpha,\lambda)^{(l)} \rangle_{(-\mathbf{S}_l)(\lambda^*)}.$$

Therefore we see that it reduces to the case of spheres marked with one, two or three points. If a sphere is marked with one point its module of states is zero unless the point is labeled with 0, but we already saw that  $\mu(0, w) = 1$ . If it is marked with two points then we use that its associated module of states is zero unless its labels are  $i, i^*$ . If this is the case then the desired equality follows from  $\tilde{\mu}(i)\tilde{\mu}(i^*) = 1$ . Finally, for a sphere with three points labeled by i, j, k, we recall that the associated module of states is isomorphic to  $\operatorname{Hom}(\mathbf{1}, V_i \otimes V_j \otimes V_k)$  which is zero unless  $\tilde{\mu}(i)\tilde{\mu}(j)\tilde{\mu}(k) = 1$ .

Therefore it amounts to choosing (u, w) such that

(16.1) 
$$u_{i} = \frac{\sqrt{w_{i}w_{i}^{*}}\sqrt{\dim(i)}}{\sqrt{u_{i}u_{i^{*}}}}w_{i}.$$

and

(16.2) 
$$\mu(i)\frac{u_i}{u_{i^*}} = \tilde{\mu}(i).$$

Choose a square root of  $\tilde{\mu}(i)$  and a square root of  $\mu(i)$  for each *i*. This can be done consistently such that  $\sqrt{\tilde{\mu}(i^*)} = 1/\sqrt{\tilde{\mu}(i)}$ , for  $i \neq i^*$ . Now define  $\eta: I \to K^*$  by

$$\eta_i := \sqrt{\tilde{\mu}(i)} \sqrt{\mu(i)}.$$

With such a choice we have for all  $i \in I$  that

$$\eta_{i}\eta_{i^{*}} = 1.$$

This implies the important equation

(16.3) 
$$\frac{\eta_i}{\eta_{i^*}} = \tilde{\mu}(i)\mu(i)$$

Take  $w_i = \eta_i$  for all *i*. Consider  $i \in I^1_{NSD}$ . Fix a choice  $\sqrt{\tilde{\mu}(i^*)}$  and then solve

$$u_i^2 = \frac{\sqrt{\dim(i)}}{\sqrt{\tilde{\mu}(i^*)}} \eta_i.$$

Therefore, if we define  $u_{i^*} = u_i \tilde{\mu}(i^*)$  then equation (16.1) is true for *i*, since we may choose  $\sqrt{u_i u_{i^*}} = u_i \sqrt{\tilde{\mu}(i^*)}$  in this case. We now need to check that equation (16.1) is also true for *i*<sup>\*</sup>. We can choose  $\sqrt{u_i u_{i^*}} = u_{i^*} \sqrt{\tilde{\mu}(i)}$ , with  $\sqrt{\tilde{\mu}(i)} = 1/\sqrt{\tilde{\mu}(i^*)}$ . Then we must check that

$$u_{i^*}^2 = \frac{\sqrt{\dim(i)}}{\sqrt{\tilde{\mu}(i)}} \eta_{i^*}$$

We compute that

$$\begin{aligned} u_{i^*}^2 &= u_i^2 \tilde{\mu}(i^*)^2 \\ &= \tilde{\mu}(i^*) u_i^2 \tilde{\mu}(i)^{-1} \\ &= \frac{\tilde{\mu}(i^*)}{\sqrt{\tilde{\mu}(i^*)}} \sqrt{\dim(i)} \eta_i \tilde{\mu}(i)^{-1} \\ &= \sqrt{\tilde{\mu}(i^*)} \sqrt{\dim(i)} \eta_{i^*} \\ &= \frac{1}{\sqrt{\tilde{\mu}(i)}} \sqrt{\dim(i)} \eta_{i^*}. \end{aligned}$$

Thus (16.1) holds for all  $j \in I_{NSD}$ . For  $i \in I_{SD}$  we have  $\eta_i = 1$  and it is easy to choose  $u_i$  satisfying (16.1). That (16.2) holds is an easy consequence of equation (16.3). That

$$Z(i) = \frac{\dim(i)}{k_i},$$

follow as in the proof of theorem 15.3.

Finally, assume that  $(\mathcal{V}, I)$  is unitary. The choice made in section 14.1 ensures  $\lambda_i = 1$  for all *i*. Observe that  $\eta_i \in S^1$  for all *i*. Therefore  $|u_i| = |u_{i^*}| = \frac{1}{\dim(i)^4}$ . According to proposition 14.10 we know  $\rho$  is given by

$$\rho_{g,N}^{u,u}(\mathbf{\Sigma}) = \prod_{l=1}^m \left( r_{i_l}^u r_{i_l^*}^u \right) \left( \sigma(i_l) u_{i_l} \overline{u_{i_l^*}} \right)^{-1}$$

 $\square$ 

We have

$$r_i^u r_{i^*}^u = \frac{\dim(i)}{|u_i||u_{i^*}|}.$$

Using  $\sigma(i) = \mu(i)$  we see that

$$\sigma(i)u_i\overline{u_{i^*}}=\mu(i)\frac{u_i}{u_{i^*}}u_{i^*}\overline{u_{i^*}}=\tilde{\mu}(i)|u_{i^*}|^2.$$

Thus we get that

$$\rho_{g,N}^{u,u}(\mathbf{\Sigma}) = \prod_{l=1}^{m} \tilde{\mu}(i_l^*).$$

The argument given above proves that this is 1 unless  $Z(\Sigma) = 0$ .

**Corollary 16.2.** Assume  $\tilde{\mu}$  is as above. Assume a labeled marked  $\Sigma$  surface has labels  $i_1, ..., i_k$ . We see that  $\prod_{l=1}^{l} \tilde{\mu}(i_l) \neq 1$  implies  $Z(\Sigma) = \mathbf{0}$ .

# 17 The quantum SU(N) modular tensor categories

We refer to the papers [28], [30] and [16] (which uses the skein theory model for the SU(2) case build in [17], [18]) for the complete construction of the quantum SU(N) modular tensor category  $H_k^{\text{SU}(N)}$  at the root of unity  $q = e^{2\pi i/(k+N)}$ . For a short review see also [14]. The simple objects of this category are indexed by the following set of young diagrams

$$\Gamma_{N,k} = \{ (\lambda_1, \dots, \lambda_p) \mid \lambda_1 \le k, p < N \}.$$

The involution  $\dagger : \Gamma_{N,k} \to \Gamma_{N,K}$  is defined as follows. For a Young diagram  $\lambda$  in  $\Gamma_{N,k}$  we define  $\lambda^{\dagger} \in \Gamma_{N,k}$  to be the Young diagram obtained from the skew-diagram  $(\lambda_1^N)/\lambda$  by rotation as indicated in the following figure.



Let  $\mu = e^{2\pi i/N}$  and  $\zeta_N$  be the set of N'th roots of 1 in  $\mathbb{C}$ . We then consider the following map

 $\tilde{\mu}: \Gamma_{N,k} \to \zeta_N$ 

given by

$$\tilde{\mu}(\lambda) = \mu^{|\lambda|},$$

where  $|\lambda| = \lambda_1 + \ldots + \lambda_p$ .

Proposition 17.1. We have that

$$\tilde{\mu} \in \Pi(H_k^{\mathrm{SU}(\mathrm{N})}, \Gamma_{N,k})^*.$$

*Proof.* We have

$$\tilde{\mu}(\lambda)\tilde{\mu}(\lambda^{\dagger}) = 1,$$

since  $|\lambda| + |\lambda^{\dagger}| = N\lambda_1$  by construction. Now consider  $\lambda, \mu, \nu \in \Gamma_{N,k}$ . By the very definition  $H_k^{\mathrm{SU}(N)}(0, \lambda \otimes \mu \otimes \nu) = 0$  if

$$|\lambda| + |\mu| + |\nu| \notin N\mathbb{Z},$$

since the  $|\lambda| + |\mu| + |\nu|$  ingoing strands at the top of the cylinder over the disc can only disappear into coupons N at the time inside the cylinder, since we have the empty diagram at be bottom determined by the label 0.

Using the notation in [14], we will now fix isomorphisms

$$q_{\lambda} \in H^{\mathrm{SU}(\mathrm{N})}(\lambda^{\dagger}, \lambda^{*}),$$

as indicated in the figure below (illustrated for some particular element  $\lambda \in \Gamma_{6,k}$  for  $k \geq 5$ )



This gives us  $\mu: \Gamma_{N,k} \to \mathbb{C}^*$  such that

$$F(q_{\lambda}) = \mu(q)q_{\lambda^{\dagger}}.$$

**Proposition 17.2.** For N odd and any  $\lambda \in \Gamma_{N,k}$ , we have that

 $\mu(\lambda) = 1.$ 

For N even and any  $\lambda \in \Gamma_{N,k}$  we have that

$$\mu(\lambda) = (-1)^{|\lambda|}.$$

*Proof.* We observe that if we apply F to  $q_{\lambda}$ , top and bottom can by a half rotation be brought into the right position for comparison with  $q_{\lambda^{\dagger}}$  and the relevant computation for each coupons in between is the following



Here the last sign is a result of the following calculation in the notation of [16], using that

$$a = q^{-\frac{1}{2N}}, v = q^{-\frac{N}{2}}, s = q^{\frac{1}{2}},$$

namely, the braiding and the twist on top of the coupon contributes

$$(-a^{-1}s)^{nm+m(m-1)}(a^{-1}v)^m = (-1)^{Nm-m}$$

times the coupon with the strands in the original position again.

From this proposition, we observe that if N is odd or N is divisible by 4, then there are no self-dual symplectic objects in  $H_k^{\rm SU(N)}$ . If however, N is even, but N/2 is odd, then the self-dual objects are symplectic if and only if they have an odd number of boxes. Moreover, we observe that on these self-dual objects

$$\tilde{\mu}(\lambda) = -1.$$

In all cases, we see that  $\tilde{\mu}$  is a fundamental symplectic character.

# 18 The general quantum group modular tensor categories

We will now fix a simple Lie algebra **g** and we will consider the corresponding quantum group at the root of unity  $q = e^{2\pi i/(k+h)}$ . The associated modular tensor

category will be denoted  $(H_k^{\mathbf{g}}, \Lambda_k^{\mathbf{g}})$ . Let  $\mathcal{W}$  be the weight lattice and  $\mathcal{R}$  the root lattice for g. We recall that the fundamental group of g is  $\Pi(g) = \mathcal{W}/\mathcal{R}$ . We have three general facts about  $\Pi(\mathbf{g})$ . The first one is that

 $\lambda + \lambda^{\dagger} = 0 \mod \mathcal{R}.$ 

for all dominant weights  $\lambda \in \mathcal{W}_+$  and  $\lambda^{\dagger} = -w_0(\lambda)$ , where  $w_0$  is the longest element of the Weyl group, e.g.  $\lambda^{\dagger}$  is the highest weight vector of the dual of the irreducible representation  $V_{\lambda}$ , corresponding to  $\lambda$ . We further observe that if  $V_{\lambda}$ is self-dual, then  $2\lambda$  will be in  $\mathcal{R}$ .

The second fact is that if we know that for  $\lambda, \mu, \nu \in \mathcal{W}_+$ 

$$\operatorname{Hom}_{G}(0, V_{\lambda} \otimes V_{\mu} \otimes V_{\nu}) \neq 0,$$

then

 $\lambda + \mu + \nu \neq 0 \mod \mathcal{R}.$ 

We recall that

 $\operatorname{Hom}_{G}(0, V_{\lambda} \otimes V_{\mu} \otimes V_{\nu}) = 0,$ 

implies that

$$H_k^{\mathbf{g}}(0, V_\lambda \otimes V_\mu \otimes V_\nu) = 0.$$

The corresponding property for the modular functor coming from Conformal Field Theory (see [12]) is clear by construction.

We now recall that  $\Pi(\mathbf{g})$  is cyclic unless  $\mathbf{g} = D_n$ , where  $\Pi(\mathbf{g}) = \mathbb{Z}_2 \times \mathbb{Z}_2$ . In the last case, one knows that (1,0) and (0,1) are symplectic, but (1,1) is not. From this we conclude that in all cases, we see that there exist some even N and a homomorphism

$$\tilde{\mu}': \Pi(g) \to \zeta_N,$$

such that  $\tilde{\mu}'(\lambda) = -1$  if and only if  $\lambda$  is symplectic. But then we define

$$\tilde{\mu}: \Lambda_k^{\mathsf{g}} \to \zeta_N$$

to be the composite of the projection from  $\Lambda_k^{\mathsf{g}}$  to  $\Pi(\mathsf{g})$  followed by  $\tilde{\mu}'$ . We then see that  $\tilde{\mu} \in \Pi(H_k^{\mathsf{g}})^*$  and indeed it is a fundamental symplectic character.

We remark that the result of this section applied to  $\mathbf{g} = \mathbf{sl}(N)$  gives a second proof for the existence of a fundamental symplectic character in that case.

### References

[1] J. E. Andersen, "Asymptotic faithfulness of the quantum SU(n) representations of the mapping class groups", Annals of Mathematics, 163 (2006), 347 - 368.

- [2] J.E. Andersen, "The Nielsen-Thurston classification of mapping classes is determined by TQFT", J. Math. Kyoto Univ., 48 2 (2008), 323–338.
- [3] J.E. Andersen, "Asymptotics of the Hilbert-Schmidt Norm of Curve Operators in TQFT", *Letters in Mathematical Physics*, **91** (2010) 205– 214.
- [4] J.E. Andersen, "Toeplitz operators and Hitchin's projectively at connection", in *The many facets of geometry: A tribute to Nigel Hitchin*, Edited by O. Garcia-Prada, Jean Pierre Bourguignon, Simon Salamon, 177–209, Oxford Univ. Press, Oxford, 2010.
- [5] J.E. Andersen and N.L. Gammelgaard, "Hitchin' s Projectively Flat Connection, Toeplitz Operators and the Asymptotic Expansion of TQFT Curve Operators", *Grassmannians, Moduli Spaces and Vector Bundles, Clay Math. Proc.*, **14**, 1–24, Amer. Math. Soc., Providence, RI, 2011.
- [6] J.E. Andersen, "Hitchin's connection, Toeplitz operators, and symmetry invariant deformation quantization", *Quantum Topology*, 3(3-4) (2012), 293–325.
- [7] J.E. Andersen, N.L. Gammelgaard and M.R. Lauridsen, "Hitchin's Connection in Metaplectic Quantization", *Quantum Topology* 3 (2012), 327–357.
- [8] J.E. Andersen and B. Himpel, "The Witten-Reshetikhin-Turaev invariants of finite order mapping tori II", *Quantum Topology*, 3 (2012), 377–421.
- J.E. Andersen, "The Witten-Reshetikhin-Turaev invariants of finite order mapping tori I", Journal für Reine und Angewandte Mathematik, 681 (2013), 1–38.
- [10] J. E. Andersen, B. Himpel, S. F. Jørgensen, J. Martens and B. McLellan, "The Witten-Reshetikhin-Turaev invariant for links in finite order mapping tori I", Advances in Mathematics, **304** (2017), 131–178.
- [11] J.E. Andersen and K. Ueno, "Abelian Conformal Field Theories and Determinant Bundles", *International Journal of Mathematics*, 18 (2007), 919–993.
- [12] J.E. Andersen and K. Ueno, "Geometric Construction of Modular Functors from Conformal Field Theory", *Journal of Knot theory and* its Ramifications, 16(2) (2007), 127–202.

- [13] J.E. Andersen and K. Ueno, "Modular functors are determined by their genus zero data", *Quantum Topology*, 3 (2012), 255–291.
- [14] J.E. Andersen and K. Ueno, "Construction of the Witten-Reshetikhin-Turaev TQFT from conformal field theory", *Invent. Math.*, 201(2) (2015), 519–559.
- [15] S. Axelrod, S. Della Pietra, E. Witten, "Geometric quantization of Chern Simons gauge theory", J. Diff. Geom. 33 (1991), 787–902.
- [16] C. Blanchet, "Hecke algebras, modular categories and 3-manifolds quantum invariants", *Topology*, **39**(1) (2000), 193–223.
- [17] C. Blanchet, N. Habegger, G. Masbaum and P. Vogel, "Three-manifold invariants derived from the Kauffman bracket", *Topology*, **31**(4) (1992), 685–699.
- [18] C. Blanchet, N. Habegger, G. Masbaum and P. Vogel, "Topological quantum field theories derived from the Kauffman bracket", *Topology*, 34(4) (1995), 883–927.
- [19] B. Van Geemen and A. J. De Jong, "On Hitchin's connection", J. of Amer. Math. Soc., 11 (1998), 189–228.
- [20] J. Grove, "Constructing TQFT's from modular functors" J. of Knot theory and its Ramifications, 10 (2001), 1085–1131.
- [21] N. Hitchin, "Flat connections and geometric quantization", Comm. Math. Phys., 131 (1990), 347–380.
- [22] S. Mac Lane, "Categories for the Working Mathematician". Second Edition (1997), Springer; Berlin.
- [23] Y. Laszlo, "Hitchin's and WZW connections are the same", J. Diff. Geom., 49(3) (1998), 547–576.
- [24] N. Reshetikhin and V. G. Turaev, "Invariants of 3-manifolds via link polynomials and quantum groups", *Invent. Math.*, **103**(3) (1991), 547– 597.
- [25] N. Reshetikhin and V. G. Turaev, "Ribbon graphs and their invariants derived from quantum groups", Comm. Math. Phys., 127(1) (1990), 1–26.
- [26] G. Segal, "The definition of conformal field theory", Topology, geometry and quantum field theory. Proceedings of the symposium in honour of the 60<sup>th</sup> birthday of Graeme Segal held in Oxford, June 24-29, 2002.
London Mathematical society Lecture Note Series **308** (2004), Cambridge University Press, Cambridge.

- [27] A. Tsuchiya, K. Ueno and Y. Yamada, "Conformal field theory on Universal family of Stable Curves with Gauge Symmetries", Advanced Studies in Pure Mathematics, 10 (1989), 459–566.
- [28] V. G. Turaev and H. Wenzl, "Quantum invariants of 3-manifolds associated with classical simple Lie algebras", *International Journal of Mathematics*, 4 (1993), 323–358.
- [29] V.G. Turaev, Quantum invariants of knots and 3-manifolds. De Gruyter Studies in Mathematics, 18 (1994), Walter de Gruyter & Co., Berlin.
- [30] V. Turaev and H. Wenzl, "Semisimple and modular categories from link invariants", *Masthematische Annalen*, **309** (1997), 411–461.
- [31] K. Walker, "On Witten's 3-manifold invariants". Preprint (1991). Preliminary version #2 tqft.net/otherpapers/KevinWalkerTQFTNotes.pdf.
- [32] E. Witten, "Quantum field theory and the Jones polynomial", Commun. Math. Phys., 121 (1989), 351–98.

Jørgen Ellegaard Andersen Center for Quantum Geometry of Moduli Spaces University of Aarhus DK-8000, Denmark andersen@qgm.au.dk

William Elbæk Petersen Center for Quantum Geometry of Moduli Spaces University of Aarhus DK-8000, Denmark william@qgm.au.dk

### The boundary length and point spectrum enumeration of partial chord diagrams using cut and join recursion

by Jørgen Ellegaard Andersen, Hiroyuki Fuji, Robert C. Penner, and Christian M. Reidys<sup>1</sup>

#### Abstract

We introduce the boundary length and point spectrum, as a joint generalization of the boundary length spectrum and boundary point spectrum introduced by Alexeev, Andersen, Penner and Zograf. We establish by cut-and-join methods that the number of partial chord diagrams filtered by the boundary length and point spectrum satisfies a recursion relation, which combined with an initial condition determines these numbers uniquely. This recursion relation is equivalent to a second order, non-linear, algebraic partial differential equation for the generating function of the numbers of partial chord diagrams filtered by the boundary length and point spectrum.

## 1 Introduction

A partial chord diagram, is a special kind of graph, which can be specified as follows. The graph consists of a number of line segments (which we will also call backbones) arranged along the real line (hence they come with an ordering) with a number of vertices on each. A number of semi-circles (called chords) arranged in the upper half plan are attached at a subset of the vertices of the line segments, in such a way that no two chords have endpoints on the line segments in common. The vertices which are not attached to chord ends are called the marked points. A chord diagram is by definition a partial chord diagram with no marked points. Partial chord diagrams occur in many branches of mathematics, including topology [12, 15], geometry [8, 9, 2] and representation theory [13].

<sup>&</sup>lt;sup>1</sup>Keywords: chord diagram, fatgraph, cut and join equation AMS Classification: 05A15, 92-08

Acknowledgments: The authors thank Masahide Manabe and Piotr Sułkowski for useful comments. JEA and RCP are supported by the Centre for Quantum Geometry of Moduli Spaces which is funded by the Danish National Research Foundation. The research of HF is supported by the Grant-in-Aid for Research Activity Start-up [# 15H06453], Grant-in-Aid for Scientific Research(C) [# 26400079], and Grant-in-Aid for Scientific Research(B) [# 16H03927] from the Japan Ministry of Education, Culture, Sports, Science and Technology.

Furthermore, they play a very prominent role in macro molecular biology. Please see the introduction of [6] for a short review of these applications.

As documented in [17, 25, 24, 11, 10, 7, 3, 4, 1, 23, 22, 5, 6], the notion of a fatgraph [18, 19, 20, 21] is a useful concept when studying partial chord diagrams<sup>2</sup>. A fatgraph is a graph together with a cyclic ordering on each collection of half-edges incident on a common vertex. A partial linear chord diagram c has a natural fatgraph structure induced from its presentation in the plane. The fatgraph c has canonically a two dimensional surface with boundary  $\Sigma_c$  associated to it (e.g. see Figure 1).



Figure 1: The partial chord diagram c and the surface  $\Sigma_c$  associated to the fatgraph with marked points. This partial chord diagram has the type  $\{g, k, l; \{b_i\}; \{n_i\}; \{\ell_i\}\} = \{1, 6, 2; \{b_6 = 1, b_8 = 1\}; \{n_0 = 2, n_1 = 2\}; \{\ell_1 = 1, \ell_2 = 2, \ell_9 = 1\}\}$ . The boundary length-point spectra are  $\{m_{(1)} = 1, m_{(0,0)} = 2, m_{(0,0,0,0,1,0,0,0)} = 1\}$ .

We now recall the basic definitions from [1] for a partial chord diagram c.

- The number of chords, the number of marked points, and the number of backbones of c are denoted k, l, and b respectively.
- The Euler characteristic and the genus of Σ<sub>c</sub>, are denoted χ and g respectively. If n is the number of boundary components of Σ<sub>c</sub>, we have that

$$(1.1) \qquad \qquad \chi = 2 - 2g,$$

and q obeys Euler's relation

(1.2) 
$$2 - 2g = b - k + n$$

- The backbone spectrum  $\mathbf{b} = (b_0, b_1, b_2, ...)$  are assigned to c, if it has  $b_i$  backbones with precisely  $i \ge 0$  vertices (of degree either two or three);
- The boundary point spectrum  $\mathbf{n} = (n_0, n_1, ...)$  is assigned to c, if its boundary contains  $n_i$  connected components with i marked points;

 $<sup>^{2}\</sup>mathrm{In}$  [16, 14], the Schwinger-Dyson approach to the enumeration of chord diagrams is also discussed.

• The boundary length spectrum  $\ell = (\ell_1, \ell_2, ...)$  is assigned to c, if the boundary cycles of the diagram consist of  $\ell_K$  edge-paths of length  $K \ge 1$ , where the length of a boundary cycle is the number of chords it traverses counted with multiplicity (as usual on the graph obtained from the diagram by collapsing each backbone to a distinct point) plus the number of backbone undersides it traverses (or in other words, the number of traversed connected components obtained by removing all the chord endpoints from all the backbones).

We now introduce the combination of the boundary length spectrum and the boundary point spectrum, namely our new boundary length and point spectrum.

• The boundary length and point spectrum  $\mathbf{m} = (m_{(d_1,\ldots,d_K)})$  is assigned to c, if its boundary contains  $m_{(d_1,\ldots,d_K)}$  connected components of length K with marked point spectrum  $(d_1,\ldots,d_K)$ , meaning that there cyclically around the boundary components are  $d_1$  marked points, then a chord or a backbone underside, then  $d_2$  marked points, then a chord or a backbone underside, and so on all the way around the boundary component. In fact we will not need to distinguish which way around the boundary we go. Hence it is only the cyclic ordered tuple of the numbers  $d_1, \ldots, d_K$ , which we need and which we denote as  $\mathbf{d}_K = (d_1, \ldots, d_K)$ . We remark that some of the  $d_I$   $(1 \le I \le K)$  might be zero.

We have the following relations

$$\begin{split} b &= \sum_{i \geq 0} b_i, \ n = \sum_{i \geq 0} n_i = \sum_{K \geq 1} \ell_K = \sum_{K \geq 1} \sum_{\mathbf{d}_K} m_{\mathbf{d}_K}, \\ 2k + l &= \sum_{i > 0} i b_i, \ l = \sum_{i > 0} i n_i = \sum_{K \geq 1} \sum_{\mathbf{d}_K} |\mathbf{d}_K| m_{\mathbf{d}_K} \\ 2k + b &= \sum_{K \geq 1} K \ell_K = \sum_{K \geq 1} \sum_{\mathbf{d}_K} K m_{\mathbf{d}_K}, \end{split}$$

where  $|\boldsymbol{d}_{K}| = \sum_{I=1}^{K} d_{I}$ . For all K and i, we also have that

$$\ell_K = \sum_{\boldsymbol{d}_K} m_{\boldsymbol{d}_K}, \quad n_i = \sum_{K \ge 1} \sum_{i = |\boldsymbol{d}_K|} m_{\boldsymbol{d}_K}.$$

We define  $\mathcal{M}_{g,k,l}(\boldsymbol{b},\boldsymbol{m})$  to be the number of connected partial chord diagrams of type  $\{g, k, l; \boldsymbol{b}; \boldsymbol{m}\}$  taken to be zero if there is none of the specified type. In [1],  $\mathcal{N}_{g,k,l}(\boldsymbol{b},\boldsymbol{n},\boldsymbol{p})$  is defined as the number of distinct connected partial chord diagrams of type  $\{g, k, l; \boldsymbol{b}; \boldsymbol{n}; \boldsymbol{p}\}$ . We find the relation between these numbers by the following formula

$$\mathcal{N}_{g,k,l}(\boldsymbol{b},\boldsymbol{\ell},\boldsymbol{n}) = \sum_{\boldsymbol{m}\in M(\boldsymbol{\ell},\boldsymbol{n})} \mathcal{M}_{g,k,l}(\boldsymbol{b},\boldsymbol{m}),$$

where

$$M(\boldsymbol{\ell},\boldsymbol{n}) = \big\{ \boldsymbol{m} \, \big| \, \ell_K = \sum_{\boldsymbol{d}_K} m_{\boldsymbol{d}_K}, \, n_i = \sum_{K \ge 1} \sum_{i = |\boldsymbol{d}_K|} m_{\boldsymbol{d}_K} \big\}.$$

In particular, the numbers  $\mathcal{N}_{g,k,l}(\boldsymbol{b},\boldsymbol{n})$  and  $\mathcal{N}_{g,k,b}(\boldsymbol{\ell})$  are given by

$$\mathcal{N}_{g,k,l}(oldsymbol{b},oldsymbol{n}) = \sum_{oldsymbol{\ell}} \ \mathcal{N}_{g,k,l}(oldsymbol{b},oldsymbol{\ell},oldsymbol{n}), \quad \mathcal{N}_{g,k,b}(oldsymbol{\ell}) = \sum_{oldsymbol{n}} \sum_{\sum b_i = b} \ \mathcal{N}_{g,k,l} = 0(oldsymbol{b},oldsymbol{\ell},oldsymbol{n}),$$

For the index  $\boldsymbol{b} = (b_i)$ , we consider the variable  $\boldsymbol{t} = (t_i)$  and denote

$$\boldsymbol{t^b} = \prod_{i \ge 0} t_i^{b_i}.$$

And for the index  $\boldsymbol{d} = (\boldsymbol{d}_K)$ , we consider the variable  $\boldsymbol{u} = (u_{\boldsymbol{d}_K})$  and denote

$$\boldsymbol{u^m} = \prod_{K \ge 1} \prod_{\boldsymbol{d}_K} u_{\boldsymbol{d}_K}^{m_{\boldsymbol{d}_F}}$$

for any  $\boldsymbol{m} = (\boldsymbol{m}_{\boldsymbol{d}_K})$ . We define the orientable, multi-backbone, boundary length and point spectrum generating function  $H(x, y; \boldsymbol{t}; \boldsymbol{u}) = \sum_{b \geq 0} \mathcal{F}_b(x, y; \boldsymbol{t}; \boldsymbol{u})$ , where

(1.3) 
$$H_b(x,y;\boldsymbol{t};\boldsymbol{u}) = \frac{1}{b!} \sum_{g=0}^{\infty} \sum_{k=2g+b-1}^{\infty} \sum_{\substack{\sum_K \sum_{\boldsymbol{d}_K} m_{\boldsymbol{d}_K} \\ =k-2g-b+2}} \sum_{b_i=b} \mathcal{M}_{g,k,l}(\boldsymbol{b},\boldsymbol{m}) x^{2g} y^k \boldsymbol{t}^{\boldsymbol{b}} \boldsymbol{u}^{\boldsymbol{m}},$$

For an element  $\boldsymbol{p} = (p_{(d_1, \dots d_K)})$ , where each  $p_{(d_1, \dots d_K)} \in \mathbb{Z}$ , we write

$$\boldsymbol{p}=\boldsymbol{p}^+-\boldsymbol{p}^-$$

where  $p^+$  contains all the positive entries and  $p^-$  the absolute value of all the negative ones, which we assume to both be finite. We define the differential operator

$$D_{\boldsymbol{p}} = \prod_{\boldsymbol{d}} u_{\boldsymbol{d}}^{\boldsymbol{p}_{\boldsymbol{d}}^-} \prod_{\boldsymbol{d}} \left( \frac{\partial}{\partial u_{\boldsymbol{d}}} \right)^{\boldsymbol{p}_{\boldsymbol{d}}^+}$$

We now define  $s_{I,J,\ell,m}(\boldsymbol{d}_K)$ ,  $s_{I,\ell,m}(\boldsymbol{d}_K)$  and  $q_{I,J,\ell,m}(\boldsymbol{d}_K, \boldsymbol{f}_L)$  to be strings like  $\boldsymbol{p}$  given by the following formulae

$$s_{I,J,\ell,m}(\boldsymbol{d}_{K}) = \boldsymbol{e}_{\boldsymbol{d}_{K}} - \boldsymbol{e}_{(d_{1},...,d_{I-1},d_{I}-\ell-1,m,d_{J+1},...,d_{K})} - \boldsymbol{e}_{(\ell,d_{I+1},...,d_{J-1},d_{J}-m-1)},$$
  

$$s_{I,\ell,m}(\boldsymbol{d}_{K}) = \boldsymbol{e}_{\boldsymbol{d}_{K}} - \boldsymbol{e}_{(d_{1},...,d_{I-1},\ell,m,d_{I+1},...,d_{K})} - \boldsymbol{e}_{(d_{I}-\ell-m-2)},$$
  

$$q_{I,J,\ell,m}(\boldsymbol{d}_{K},\boldsymbol{f}_{L}) =$$
  

$$= \boldsymbol{e}_{\boldsymbol{d}_{K}} + \boldsymbol{e}_{\boldsymbol{f}_{L}} - \boldsymbol{e}_{(d_{1},...,d_{I-1},d_{I}-\ell-1,m,f_{J+1},...,f_{L},f_{I},...,f_{J-1},f_{J}-m-1,\ell,d_{I+1},...,d_{K})}.$$

216

where  $\boldsymbol{e}_{\boldsymbol{d}_{K}}$  denotes the sequence  $(0, \ldots, 0, 1, 0, \ldots)$  where the component 1 appears only at the entry indexed by  $\boldsymbol{d}_{K}$ . We further define the index  $c_{I,J,\ell,h}(\boldsymbol{d}_{K}, \boldsymbol{f}_{M})$  by the formula

$$c_{I,J,\ell,m}(\boldsymbol{d}_K, \boldsymbol{f}_L) = (d_1, \dots, d_{I-1}, d_I - \ell - 1, m, f_{J+1}, \dots, f_L, f_1, \dots, f_{J-1}, f_J - m - 1, \ell, d_{I+1}, \dots, d_K),$$

which is identical to the index on the last term of the above assignments.

**Theorem 1.1** (Enumeration of partial chord diagrams filtered by their boundary length and point spectrum).

Define the first and second order linear differential operators

(1.4) 
$$M_0 = \sum_{K \ge 1} \sum_{\mathbf{d}_K} \left( \sum_{1 \le J < I \le K} \sum_{\ell=0}^{d_I - 1} \sum_{m=0}^{d_J - 1} D_{s_{I,J,\ell,m}(\mathbf{d}_K)} + \sum_{I=1}^K \sum_{\ell,m=0}^{d_I - 1} D_{s_{I,\ell,m}(\mathbf{d}_K)} \right),$$

(1.5) 
$$M_2 = \frac{1}{2} \sum_{K,L \ge 1} \sum_{\boldsymbol{d}_K, \boldsymbol{f}_L} \sum_{I=1} \sum_{J=1} \sum_{\ell=0}^{I} \sum_{m=0}^{I} D_{q_{I,J,\ell,h}(\boldsymbol{d}_K)},$$

and the quadratic differential operator

(1.6) 
$$S(H) = \frac{1}{2} \sum_{K,L \ge 1} \sum_{\boldsymbol{d}_K, \boldsymbol{f}_L} \sum_{I=1}^K \sum_{J=1}^{L} \sum_{\ell=0}^{d_I-1} \sum_{m=0}^{f_L-1} u_{c_{I,J,\ell,m}(\boldsymbol{d}_K, \boldsymbol{f}_L)} D_{\boldsymbol{d}_K}(H) D_{\boldsymbol{f}_L}(H) .$$

Then the following partial differential equations hold

(1.7) 
$$\frac{\partial H_1}{\partial y} = (M_0 + x^2 M_2) H_1, \quad \frac{\partial H}{\partial y} = (M_0 + x^2 M_2 + S) H.$$

Together with the initial conditions

(1.8) 
$$H_1(x, y = 0; \boldsymbol{t} = (t_1); \boldsymbol{u}) = u_{(0)}t_1, \quad H(x, y = 0; \boldsymbol{t}; \boldsymbol{u}) = \sum_{i \ge 1} u_{(i)}t_i,$$

they determine the functions  $H_1$  and H uniquely.

In this article, we also consider the non-oriented analogue of partial chord diagrams. The generalization of the above analysis is straightforward, as we will now explain. A *non-oriented* partial chord diagrams, is a partial chord diagram together with a decoration of a binary variable at each chord, which indicates if the chord is *twisted* or not. When associating the surface  $\Sigma_c$ , to a non-oriented partial chord diagram, a twisted band is associated along twisted chords as indicated in Figure 2. By this construction,  $2^k$  orientable and non-orientable surfaces are obtained from one partial chord diagram with k chords, when we vary over all assignments of twisting or not to the k chords. In the non-oriented case, we have the following definition of the Euler characteristic.

• Euler characteristic  $\chi$ . The Euler characteristic of the two dimensional surface  $\Sigma_c$  is defined by the formula

$$(1.9) \qquad \qquad \chi = 2 - h,$$

where h is the number of cross-caps and we have Euler's relation

$$(1.10) 2 - h = b - k + n.$$

With this set-up, the enumeration of the non-oriented partial chord diagrams is considered in parallel to the oriented case discussed above with a small change for the boundary length and point spectrum  $\boldsymbol{m}$ . In this non-oriented case, there are now induced orientation on the boundaries of  $\Sigma_c$  and hence for an index  $\boldsymbol{d}_K = (d_1, \ldots, d_K)$  corresponding some boundary component of  $\Sigma_c$ , we not only need to consider this tuple up to cyclic permutation of the tuple, but also reversal of the order

$$\boldsymbol{d}_{K} = (d_{1}, d_{2}, \dots, d_{K}) = (d_{K}, \dots d_{2}, d_{1}).$$



Figure 2: The non-oriented surface constructed out of untwisted and twisted chords.

Let  $\widetilde{\mathcal{M}}_{h,k,l}(\boldsymbol{b},\boldsymbol{m})$  be the number of non-oriented partial chord diagrams of type  $\{h, k, l; \boldsymbol{b}; \boldsymbol{m}\}$ . In [1],  $\widetilde{\mathcal{N}}_{h,k,l}(\boldsymbol{b}, \boldsymbol{\ell}, \boldsymbol{n})$  is defined as the number of non-oriented connected partial chord diagrams of type  $\{h, k, l; \boldsymbol{b}; \boldsymbol{\ell}; \boldsymbol{n}\}$ . These numbers are related by the following formula

$$\widetilde{\mathcal{N}}_{h,k,l}(\boldsymbol{b},\boldsymbol{\ell},\boldsymbol{n}) = \sum_{\boldsymbol{m}\in M(\boldsymbol{\ell},\boldsymbol{n})} \widetilde{\mathcal{M}}_{h,k,l}(\boldsymbol{b},\boldsymbol{m}),$$

and the numbers  $\widetilde{\mathcal{N}}_{h,k,l}(\boldsymbol{b},\boldsymbol{n})$  and  $\widetilde{\mathcal{N}}_{h,k,b}(\boldsymbol{\ell})$  are given by

$$\widetilde{\mathcal{N}}_{h,k,l}(\boldsymbol{b},\boldsymbol{n}) = \sum_{\boldsymbol{\ell}} \ \widetilde{\mathcal{N}}_{h,k,l}(\boldsymbol{b},\boldsymbol{\ell},\boldsymbol{n}), \qquad \widetilde{\mathcal{N}}_{h,k,b}(\boldsymbol{\ell}) = \sum_{\boldsymbol{n}} \sum_{\sum b_i = b} \ \widetilde{\mathcal{N}}_{h,k,l=0}(\boldsymbol{b},\boldsymbol{\ell},\boldsymbol{n}).$$

218

We define the non-oriented generating function  $\widetilde{H}(x, y; t; u) = \sum_{b \ge 1} \widetilde{H}_b(x, y; t; u)$  to be given by

(1.11) 
$$\widetilde{H}_b(x,y;\boldsymbol{t};\boldsymbol{u}) = \frac{1}{b!} \sum_{h=0}^{\infty} \sum_{\substack{k=h+b-1 \ \sum_{K} \sum_{\boldsymbol{d}_K} m_{\boldsymbol{d}_K} \\ =k-h-b+2}} \sum_{\substack{\sum b_i=b \ \widetilde{\mathcal{M}}_{h,k,l}(\boldsymbol{b},\boldsymbol{m}) x^h y^k \boldsymbol{t}^{\boldsymbol{b}} \boldsymbol{u}^{\boldsymbol{m}}.}$$

We define  $s_{I,J,\ell,h}^{\times}(\boldsymbol{d}_K)$ ,  $s_{I,\ell,h}^{\times}(\boldsymbol{d}_K)$  and  $q_{I,J,\ell,h}^{\times}(\boldsymbol{d}_K, \boldsymbol{f}_L)$  to be by

$$\begin{aligned} s_{I,J,\ell,m}^{\times}(\boldsymbol{d}_{K}) &= \boldsymbol{e}_{\boldsymbol{d}_{K}} - \boldsymbol{e}_{(d_{1},...,d_{I-1},\ell,m,d_{J-1},...,d_{I+1},d_{J}-\ell-1,d_{J}-m-1,d_{J+1},...,d_{K})}, \\ s_{I,\ell,m}^{\times}(\boldsymbol{d}_{K}) &= \boldsymbol{e}_{\boldsymbol{d}_{K}} - \boldsymbol{e}_{(d_{1},...,d_{I-1},\ell,d_{I}-\ell-m-2,m,d_{I+1},...,d_{K})}, \\ q_{I,J,\ell,m}^{\times}(\boldsymbol{d}_{K},\boldsymbol{f}_{L}) \\ &= \boldsymbol{e}_{\boldsymbol{d}_{K}} + \boldsymbol{e}_{\boldsymbol{f}_{M}} - \boldsymbol{e}_{(f_{1},...,f_{J-1},f_{J}-m-1,\ell,d_{I-1},...,d_{1},d_{K},...,d_{I+1},d_{I}-\ell-1,m,f_{J+1},...,f_{L})} \end{aligned}$$

And we also define indices  $c^{\times}_{I,J,\ell,h}(\boldsymbol{d}_{K},\boldsymbol{f}_{M})$  by the formula

$$\begin{aligned} c_{I,J,\ell,h}^{\times}(\boldsymbol{d}_{K},\boldsymbol{f}_{L}) \\ &= (f_{1},\ldots,f_{J-1},f_{J}-m-1,\ell,d_{I-1},\ldots,d_{1},d_{K},\ldots,d_{I+1},d_{I}-\ell-1,m,f_{J+1},\ldots,f_{L}), \end{aligned}$$

which again, we note is identical to the index on the last term of the above assignments.

**Theorem 1.2** (Enumeration of non-oriented partial chord diagrams filtered by their boundary length and point spectrum).

Define the first and second order linear differential operators

(1.12) 
$$M_{1}^{\times} = \sum_{K \ge 1} \sum_{\boldsymbol{d}_{K}} \left( \sum_{1 \le J < I \le K} \sum_{\ell=0}^{d_{I}-1} \sum_{m=0}^{d_{J}-1} D_{s_{I,J,\ell,m}^{\times}(\boldsymbol{d}_{K})} + \sum_{I=1}^{K} \sum_{\ell,m=0}^{d_{I}-1} D_{s_{I,\ell,m}^{\times}(\boldsymbol{d}_{K})} \right),$$
  
(1.13) 
$$M_{2}^{\times} = \frac{1}{2} \sum_{K} \sum_{I} \sum_{K} \sum_{L} \sum_{I} \sum_$$

(1.13) 
$$M_2^{\times} = \frac{1}{2} \sum_{K,L \ge 1} \sum_{\boldsymbol{d}_K, \boldsymbol{f}_L} \sum_{I=1} \sum_{J=1} \sum_{\ell=0} \sum_{m=0} D_{q_{I,J,\ell,m}^{\times}(\boldsymbol{d}_K)},$$

and the quadratic differential operator

(1.14) 
$$S^{\times}(H) = \frac{1}{2} \sum_{K,L \ge 1} \sum_{\boldsymbol{d}_{K},\boldsymbol{f}_{L}} \sum_{I=1}^{K} \sum_{J=1}^{L} \sum_{\ell=0}^{d_{I}-1} \sum_{m=0}^{f_{L}-1} u_{c_{I,J,\ell,m}^{\times}(\boldsymbol{d}_{K},\boldsymbol{f}_{L})} D_{\boldsymbol{d}_{K}}(H) D_{\boldsymbol{f}_{L}}(H) .$$

Then the following partial differential equations hold

(1.15) 
$$\begin{aligned} \frac{\partial \widetilde{H}_1}{\partial y} &= (M_0 + xM_1^{\times} + x^2(M_2 + M_2^{\times}))\widetilde{H}_1, \\ \frac{\partial \widetilde{H}}{\partial y} &= (M_0 + xM_1^{\times} + x^2(M_2 + M_2^{\times}) + S + S^{\times})\widetilde{H}. \end{aligned}$$

Together with the following initial conditions

(1.16) 
$$\widetilde{H}_1(x, y = 0; \boldsymbol{t} = (t_1); \boldsymbol{u}) = u_{(0)}t_1, \quad \widetilde{H}(x, y = 0; \boldsymbol{t}; \boldsymbol{u}) = \sum_{i \ge 1} u_{(i)}t_i,$$

determines  $\widetilde{H}_1$  and  $\widetilde{H}$  uniquely.

This paper is organized as follows. Section 2 contains basic combinatorial results on the boundary length and point spectra of partial chord diagrams and derives the recursion relation of the number of diagrams (Proposition 2.1), by the cut-and-join method. This cut-and-join equation is rewritten as a second order, non-linear, algebraic partial differential equation for generating function of the number of partial chord diagrams filtered by the boundary length and point spectrum (Proposition 2.2). Section 3 extends these results to include the non-oriented analogues of the partial chord diagrams. The cut-and-join equation is extended to provide a recursion on the number of non-oriented partial chord diagrams (Proposition 3.1), and is also rewritten as partial differential equation (Proposition 3.2).

# 2 Combinatorial proof of the cut-and-join equation

In this section, we devote to prove Theorem 1.1. The partial differential equation (1.7) is equivalent to the following recursion relation for the numbers of connected partial chord diagrams.

**Proposition 2.1.** The numbers  $\mathcal{M}_{g,k,l}(\boldsymbol{b}, \boldsymbol{m})$  enumerating connected partial chord diagrams of type  $\{g, k, l; \boldsymbol{b}, \boldsymbol{m}\}$  obey the following recursion relation

$$\begin{split} k\mathcal{M}_{g,k,l}(\boldsymbol{b},\boldsymbol{m}) \\ = & \sum_{K \ge 1} \sum_{\boldsymbol{d}_{K}} (m_{\boldsymbol{d}_{K}} + 1) \bigg[ \sum_{1 \le I < J \le K} \sum_{\ell=0}^{d_{I}-1} \sum_{m=0}^{d_{J}-1} \mathcal{M}_{g,k-1,l+2}(\boldsymbol{b},\boldsymbol{m} + s_{I,J,\ell,m}(\boldsymbol{d}_{K})) \\ &+ \sum_{I=1}^{K} \sum_{\substack{\ell,m=0\\\ell+m \le d_{I}-2}}^{d_{I}-1} \mathcal{M}_{g,k-1,l+2}(\boldsymbol{b},\boldsymbol{m} + s_{I,\ell,m}(\boldsymbol{d}_{K})) \bigg] \\ &+ \frac{1}{2} \sum_{K \ge 1} \sum_{L \ge 1} \sum_{\boldsymbol{d}_{K}} \sum_{\boldsymbol{f}_{L}} (m_{\boldsymbol{d}_{K}} + 1)(m_{\boldsymbol{f}_{L}} + 1 - \delta_{\boldsymbol{d}_{K},\boldsymbol{f}_{L}}) \\ &\times \sum_{I=1}^{K} \sum_{J=1}^{L} \sum_{L=0}^{L} \sum_{m=0}^{d_{I}-1} \sum_{m=0}^{f_{J}-1} \mathcal{M}_{g-1,k-1,l+2}(\boldsymbol{b},\boldsymbol{m} + q_{I,J,\ell,m}(\boldsymbol{d}_{K},\boldsymbol{f}_{L})) \end{split}$$

$$(2.1) + \frac{1}{2} \sum_{K \ge 1} \sum_{L \ge 1} \sum_{d_K} \sum_{f_L} \sum_{g_1 + g_2 = g} \sum_{k_1 + k_2 = k - 1} \sum_{b^{(1)} + b^{(2)} = b} \sum_{k_1 = 1}^{K} \sum_{f_1 = 1}^{K} \sum_{f_2 = 0}^{d_1 - 1} \sum_{m = 0}^{f_2 - 1} \sum_{\substack{\boldsymbol{m}^{(1)} + \boldsymbol{m}^{(2)} \\ = \boldsymbol{m} + q_{I,J,\ell,m} (\boldsymbol{d}_K, \boldsymbol{f}_L)}} \sum_{\boldsymbol{m}^{(1)} = \boldsymbol{m} + q_{I,J,\ell,m} (\boldsymbol{d}_K, \boldsymbol{f}_L)} \times m_{\boldsymbol{d}_K}^{(1)} m_{\boldsymbol{f}_L}^{(2)} \underbrace{\frac{b!}{b^{(1)!}b^{(2)!}}}_{b^{(1)!} \mathcal{M}_{g_1,k_1,l_1}} (\boldsymbol{b}^{(1)}, \boldsymbol{m}^{(1)}) \mathcal{M}_{g_2,k_2,l_2} (\boldsymbol{b}^{(2)}, \boldsymbol{m}^{(2)})$$

This recursion relation is referred to as the *cut-and-join equation*, since it follows from a cut-and-join argument, which we shall now provide.

*Proof.* When one removes one chord from a partial chord diagram, there are essentially three distinct possible outcomes. First of all the diagram can stay connected and then there are two cases to consider. In the first one, the chord that is removed is adjacent to two different boundary components and in the second one it is adjacent to just one. The third case is when the chord diagram becomes disconnected.

In the first case, the genus of the partial chord diagram is not changed, but two boundary components join into one component. On the other hand, in the second case, the genus decreases by one, and one boundary component splits into two components.



Figure 3: Removal of a chord in case one. The chord is depicted as a band. After the removal of this chord, two boundary components join into one component. Left: The clusters of marked points  $(d_I - \ell - 1, m)$  and  $(d_J - m - 1, \ell)$  join into two clusters  $d_I$  and  $d_J$  Right: The clusters of marked points  $(\ell, m)$  and  $(d_J - m - 1)$ join into one cluster  $d_I$ .

In the first case, and let us say that after removing this chord, the two adjacent boundary components join into one component with the marked point spectrum  $d_K = (d_1, \ldots, d_K)$ . (See Figure 3.) Under this elimination, the numbers k and n change to k - 1 and n - 1, the genus g is not changed (c.f. Euler's relation 2 - 2g = b - k + n). The number of marked points l changes to l + 2, because the chord ends of the chord which is removed become new marked points. There are two distinct possible sub cases, namely either the chord ends belong to two distinct clusters of marked points  $d_I$  and  $d_J$  in the resulting chord diagram, or chord ends belong to the same cluster of marked points  $d_I$ .

We will consider the former kind of chord, and assume I < J without loss of generality. Before we remove the chord, the two boundaries adjacent to the chord needs to have the following two marked point spectra

$$(d_1, \ldots, d_{I-1}, d_I - \ell - 1, m, d_{J+1}, \ldots, d_K)$$
, and  $(\ell, d_{I+1}, \ldots, d_{J-1}, d_J - m - 1)$ ,  
 $0 \le \ell \le d_I - 1$ ,  $0 \le m \le d_J - 1$ ,

When removing the chord, we connect the clusters of marked points  $(d_I - \ell - 1, m)$ and  $(d_J - m - 1, \ell)$ . If the original partial chord diagram has the boundary lengthpoint spectrum  $\boldsymbol{m}$ , the resulting diagram has

$$m - e_{(d_1,...,d_{I-1},d_I - \ell - 1,m,d_{J+1},...,d_K)} - e_{(\ell,d_{I+1},...,d_{J-1},d_J - m - 1)} + e_{d_K}$$
  
=  $m + s_{I,J,\ell,m}(d_K).$ 

For the latter kind, we must have two boundary components with the marked point spectra

$$(d_1, \dots, d_{I-1}, \ell, m, d_{I+1}, \dots, d_K)$$
, and  $(d_I - \ell - m - 2)$ .  
 $0 \le \ell, m \le d_I - 1, \quad 0 \le \ell + m \le d_I - 2,$ 

and removing the chord connects the clusters of marked points  $(\ell, m)$  and  $(d_J - m - 1)$ . This manipulation changes the boundary length and point spectrum m into

$$m - e_{(d_1,...,d_{I-1},\ell,m,d_{I+1},...,d_K)} - e_{(d_I-\ell-m-2)} + e_{d_K} = m + s_{I,\ell,m}(d_K).$$

For both of these two kinds of removal, there are  $m_{d_{\kappa}} + 1$  possibilities to choose the boundary components in the partial chord diagram. Therefore, the number of possibilities for the first way of removal is

(2.2) 
$$\sum_{K \ge 1} \sum_{\boldsymbol{d}_{K}} (m_{\boldsymbol{d}_{K}} + 1) \bigg[ \sum_{1 \le I < J \le K} \sum_{\ell=0}^{d_{I}-1} \sum_{m=0}^{d_{J}-1} \mathcal{M}_{g,k-1,l+2} (\boldsymbol{b}, \boldsymbol{m} + s_{I,J,\ell,m} (\boldsymbol{d}_{K})) + \sum_{I=1}^{K} \sum_{\substack{\ell,m=0\\\ell+m \le d_{I}-2}}^{d_{I}-1} \mathcal{M}_{g,k-1,l+2} (\boldsymbol{b}, \boldsymbol{m} + s_{I,\ell,m} (\boldsymbol{d}_{K})) \bigg].$$

In the second case (see Figure 4), the removal changes the numbers k and n to k-1 and n+1 and the genus of the partial chord diagram decreases by one. For partial chord diagram with a boundary with marked point spectrum

$$(d_1, \dots, d_{I-1}, d_I - \ell - 1, m, f_{J+1}, \dots, f_L, f_1, \dots, f_{J-1}, f_J - m - 1, \ell, d_{I+1}, \dots, d_K), 0 \le \ell \le d_I - 1, \quad 0 \le m \le f_J - 1,$$

we remove the chord which connects the two clusters  $(f_J - m - 1, \ell)$  and  $(d_I - \ell - 1, m)$  of marked points. The boundary component then splits into two boundary components with marked point spectra  $\boldsymbol{d}_K = (d_1, \ldots, d_K)$  and  $\boldsymbol{f}_L = (f_1, \ldots, f_L)$ . If the original partial chord diagram has the boundary length and point spectrum  $\boldsymbol{m}$ , after removal of this chord, we find that

$$m - e_{(d_1,...,d_{I-1},d_I - \ell - 1,m,f_{J+1},...,f_L,f_1,...,f_{J-1},f_J - m - 1,\ell,d_{I+1},...,d_K)} + e_{d_K} + e_{f_L}$$
  
=  $m + q_{I,J,\ell,m}(d_K, f_L).$ 

The number of possibilities of this removal is  $(m_{\boldsymbol{d}_{K}} + 1)(m_{\boldsymbol{f}_{L}} + 1)$  for  $\boldsymbol{d}_{K} \neq \boldsymbol{f}_{L}$ . If  $\boldsymbol{d}_{K} = \boldsymbol{f}_{L}$ , the number of possibilities becomes  $m_{\boldsymbol{d}_{K}}(m_{\boldsymbol{d}_{K}} + 1)/2$ . In total, the number of possibilities for the second way of elimination is

(2.3) 
$$\frac{1}{2} \sum_{K=1}^{\infty} \sum_{L=1}^{\infty} \sum_{\boldsymbol{d}_{K}} \sum_{\boldsymbol{f}_{L}} (m_{\boldsymbol{d}_{K}} + 1) (m_{\boldsymbol{f}_{L}} + 1 - \delta_{\boldsymbol{d}_{K}, \boldsymbol{f}_{L}}) \\ \times \sum_{I=1}^{K} \sum_{J=1}^{L} \sum_{\ell=0}^{d_{I}-1} \sum_{h=0}^{f_{J}-1} \mathcal{M}_{g-1,k-1,l+2} (\boldsymbol{b}, \boldsymbol{m} + q_{I,J,\ell,h}(\boldsymbol{d}_{K}, \boldsymbol{f}_{L})) .$$

The factor 1/2 in front of the sum takes care of the over counting in the cases  $d_K \neq f_L$ .

In the third case, the partial chord diagram split into two connected components. We consider the case that the original diagram has the type  $\{g, k, l; \boldsymbol{b}, \boldsymbol{m}\}$ and the resulting two connected components have types  $\{g_1, k_1, l_1; \boldsymbol{b}^{(1)}, \boldsymbol{m}^{(1)}\}$  and  $\{g_2, k_2, l_2; \boldsymbol{b}^{(2)}, \boldsymbol{m}^{(2)}\}$ . These types are related such that

$$g = g_1 + g_2, \quad k - 1 = k_1 + k_2, \quad \boldsymbol{b} = \boldsymbol{b}^{(1)} + \boldsymbol{b}^{(2)}.$$

Since a boundary component also split into two components, the boundary length and point spectrum changes in the same manner as in the second case.

$$m + q_{I,J,\ell,m}(d_K, f_L) = m^{(1)} + m^{(2)}.$$

There are  $m_{\boldsymbol{d}_{K}}^{(1)} m_{\boldsymbol{f}_{L}}^{(2)}$  ways to choose the boundary components which are to be fused under the inverse operation of chord removal. And the number of different ordered splittings of a *b*-backbone diagram is  $\frac{b!}{b^{(1)}b^{(2)}!}$  where  $b^{(a)} = \sum_{i} b_{i}^{(a)}$  (a =1, 2). Therefore, the total number of possibilities of this case is

$$\frac{1}{2} \sum_{I=1}^{\infty} \sum_{J=1}^{\infty} \sum_{\boldsymbol{d}_K} \sum_{\boldsymbol{f}_L} \sum_{g_1+g_2=g} \sum_{k_1+k_2=k-1} \sum_{b^{(1)}+b^{(2)}=b}$$



Figure 4: The second and third way of elimination of a chord. After the elimination of this chord, a boundary component split into two different boundary components.

(2.4) 
$$\times \sum_{I=1}^{K} \sum_{J=1}^{L} \sum_{\ell=0}^{d_{I}-1} \sum_{m=0}^{f_{J}-1} \sum_{\substack{\boldsymbol{m}^{(1)} + \boldsymbol{m}^{(2)} \\ = \boldsymbol{m} + q_{I,J,\ell,m}(\boldsymbol{d}_{K}, \boldsymbol{f}_{L})}} }{m_{\boldsymbol{d}_{K}} m_{\boldsymbol{f}_{L}}^{(2)} \frac{b!}{b^{(1)} |\boldsymbol{h}^{(2)}|}} \mathcal{M}_{g_{1},k_{1},l_{1}}(\boldsymbol{b}^{(1)}, \boldsymbol{m}^{(1)}) \mathcal{M}_{g_{2},k_{2},l_{2}}(\boldsymbol{b}^{(2)}, \boldsymbol{m}^{(2)}). }$$

The factor 1/2 corrects for the over counting due to the ordering of the two connected components.

The sum of the contributions (2.2), (2.3), and (2.4) from the three different cases of chord removals equals  $k\mathcal{M}_{g,k,l}(\boldsymbol{b},\boldsymbol{m})$ , because there are k possibilities for the choice of the chord to be removed. This gives the cut-and-join equation (2.1).

**Proposition 2.2.** The generating function H(x, y; t, ; u) is uniquely determined by the differential equation

$$\frac{\partial H}{\partial y} = (M+S)H,$$

where  $M = M_0 + x^2 M_2$ . The generating function  $Z(x, y; t, ; u) = \exp[H]$  of the number of connected and disconnected partial chord diagrams satisfies

(2.5) 
$$\frac{\partial Z}{\partial y} = MZ,$$

and is as such determined by the initial conditions

$$H(x, y = 0; t; u) = \sum_{i \ge 1} t_i u_{(i)}, \quad Z(x, y = 0; t, ; u) = e^{\sum_{i \ge 1} t_i u_{(i)}}$$

*Proof.* It is straightforward to check that the differential equation  $\frac{\partial H}{\partial y} = (M+S)H$  is equivalent to the cut-and-join equation (2.1). The actions in the quadratic differential S on H can be rewritten by following relation

$$D_{\boldsymbol{d}_{K}}(H)D_{\boldsymbol{f}_{L}}(H) + D_{\boldsymbol{d}_{K}}D_{\boldsymbol{f}_{L}}H = \frac{1}{Z}D_{\boldsymbol{d}_{K}}D_{\boldsymbol{f}_{L}}Z.$$

The derivatives on the right hand side are contained in  $M_2$ , and the differential equation  $\frac{\partial Z}{\partial y} = MZ$  follows from that of H.

On the initial condition, every partial chord diagram of type  $\{g, k, l; \mathbf{b}; \mathbf{m}\}$  can be obtained from the disjoint collection of type  $\{0, 0, i; \mathbf{e}_i, \mathbf{e}_{(i)}\}$  with multiplicity  $b_i$  by connecting them with k chords. This implies  $H(x, y = 0; \mathbf{t}; \mathbf{u}) = \sum_{i \ge 1} t_i u_{(i)}$ . Since this is the first order differential equation of y, the coefficient of  $y^k$  is determined uniquely using this initial condition.

# 3 Non-oriented analogue of the cut-and-join equation

In this section, we will prove Theorem 1.2. We first establish the following proposition.

**Proposition 3.1.** The number  $\mathcal{M}_{g,k,l}(\boldsymbol{b},\boldsymbol{m})$  of connected non-oriented partial chord diagrams of type  $\{g, k, l; \boldsymbol{b}, \boldsymbol{m}\}$  obeys the following recursion relation

$$\begin{split} & k\mathcal{M}_{g,k,l}(\boldsymbol{b},\boldsymbol{m}) \\ &= \sum_{K \ge 1} \sum_{\boldsymbol{d}_{K}} (m_{\boldsymbol{d}_{K}} + 1) \\ & \times \left[ \sum_{I < J} \sum_{\ell=0}^{d_{I}-1} \sum_{m=0}^{d_{J}-1} \left\{ \widetilde{\mathcal{M}}_{h,k-1,l+2} \left( \boldsymbol{b}, \boldsymbol{m} + s_{I,J,\ell,m}(\boldsymbol{m}) \right) + \widetilde{\mathcal{M}}_{h-1,k-1,l+2} \left( \boldsymbol{b}, \boldsymbol{m} + s_{I,J,\ell,m}(\boldsymbol{m}) \right) \right\} \right] \\ & + \sum_{I=1}^{K} \sum_{\ell=0}^{K} \sum_{m=0}^{d_{I}-2} \left\{ \widetilde{\mathcal{M}}_{h,k-1,l+2} \left( \boldsymbol{b}, \boldsymbol{m} + s_{I,\ell,m}(\boldsymbol{m}) \right) + \widetilde{\mathcal{M}}_{h-1,k-1,l+2} \left( \boldsymbol{b}, \boldsymbol{m} + s_{I,\ell,m}^{\times}(\boldsymbol{m}) \right) \right\} \right] \\ & + \frac{1}{2} \sum_{K \ge 1} \sum_{L \ge 1} \sum_{d_{K}} \sum_{f_{L}} (m_{\boldsymbol{d}_{K}} + 1) (m_{\boldsymbol{f}_{L}} + 1 - \delta_{\boldsymbol{d}_{K},\boldsymbol{f}_{L}}) \\ & \times \sum_{I=1}^{K} \sum_{J=1}^{L} \sum_{\ell=0}^{L} \sum_{m=0}^{d_{I}-1} \sum_{m=0}^{f_{J}-1} \left\{ \widetilde{\mathcal{M}}_{h-2,k-1,l+2} \left( \boldsymbol{b}, \boldsymbol{m} + q_{I,J,\ell,m} (\boldsymbol{d}_{K}, \boldsymbol{f}_{L}) \right) \\ & + \widetilde{\mathcal{M}}_{h-2,k-1,l+2} \left( \boldsymbol{b}, \boldsymbol{m} + q_{I,J,\ell,m}^{\times} (\boldsymbol{d}_{K}, \boldsymbol{f}_{L}) \right) \right\} \\ & + \frac{1}{2} \sum_{K \ge 1} \sum_{L \ge 1} \sum_{L \ge 1} \sum_{d_{K}} \sum_{f_{L}} \sum_{h_{1}+h_{2}=h} \sum$$

$$\times \sum_{I=1}^{K} \sum_{J=1}^{L} \sum_{\ell=0}^{d_{I}-1} \sum_{m=0}^{f_{J}-1} \left( \sum_{\substack{\boldsymbol{m}^{(1)} + \boldsymbol{m}^{(2)} \\ = \boldsymbol{m} + q_{I,J,\ell,m}(\boldsymbol{d}_{K},\boldsymbol{f}_{L})} + \sum_{\substack{\boldsymbol{m}^{(1)} + \boldsymbol{m}^{(2)} \\ = \boldsymbol{m} + q_{K,J,\ell,m}^{\times}(\boldsymbol{d}_{K},\boldsymbol{f}_{L})} \right) \boldsymbol{m}_{\boldsymbol{d}_{K}}^{(1)} \boldsymbol{m}_{\boldsymbol{f}_{L}}^{(2)}$$

$$(3.1) \times \frac{b!}{b^{(1)}!b^{(2)}!} \widetilde{\mathcal{M}}_{h_{1},k_{1},l_{1}} \left(\boldsymbol{b}^{(1)}, \boldsymbol{m}^{(1)}\right) \widetilde{\mathcal{M}}_{h_{2},k_{2},l_{2}} \left(\boldsymbol{b}^{(2)}, \boldsymbol{m}^{(2)}\right).$$

*Proof.* If we remove a non-twisted chord, then we find the same recursive structure as for the numbers (2.2), (2.3), and (2.4) for  $\widetilde{\mathcal{M}}_{h,k,l}(\boldsymbol{b},\boldsymbol{m})$  in the oriented case. As we did in the proof of proposition 2.1, we also consider three cases, organised the same way, when removing a twisted chord.

In the first case (see Figure 5), there are again two possibilities, namely the twisted chord ends belong to two different or the same clusters of marked points on the boundary component in the resulting diagram after removal. Contrary to the case of non-twisted chords, the boundary cycle does not split, but the marked point spectrum changes due to the recombination of the boundary component. For both of these two cases, the numbers k and n change to k-1 and n, and the cross-cap number h decreases by one under this elimination (c.f. Euler's relation 2 - h = b - k + n). The chord ends become marked points and l changes to l+2.



Figure 5: Removal of a twisted chord from a non-oriented partial chord diagram. The chord is depicted as a twisted band. After the elimination of this chord, the boundary component is reconnected into one component with different marked point spectrum. Left: The clusters of marked points  $(d_I - \ell - 1, m)$  and  $(d_J - m - 1, \ell)$  join into two clusters  $d_I$  and  $d_J$ . Right: The clusters of marked points  $(\ell, m)$  and  $(d_J - m - 1)$  join into one cluster  $d_I$ .

In the former situation, we must have a boundary component with the marked point spectrum

$$(d_1, \ldots, d_{I-1}, \ell, m, d_{J-1}, d_{J-2}, \ldots, d_{I+1}, d_I - \ell - 1, d_J - m - 1, d_{J+1}, \ldots, d_K)$$

Partial chord diagrams and boundary length and point spectrum

$$I < J, \quad 0 \le \ell \le d_I - 1, \quad 0 \le m \le d_J - 1,$$

from which we remove one twisted chord with one end between the two clusters  $(\ell, m)$  and the other between  $(d_I - \ell - 1, d_J - m - 1)$ . Then the removal will result in a boundary component with the marked point spectrum  $\boldsymbol{d}_K$  and the boundary length and point spectrum  $\boldsymbol{m}$  is changed as follows

$$m{m} - e_{(d_1,...,d_{I-1},\ell,m,d_{J-1},...,d_{J+1},d_J-\ell-1,d_J-m-1,d_{J+1},...,d_K)} + e_{m{d}_K}$$
  
=  $m{m} + s_{I,J,\ell,m}^{\times}(m{d}_K).$ 

The possible number of choices for this kind of removal is  $m_{\mathbf{d}_{K}} + 1$ , and the total number of diagrams which can be obtained in this way is

(3.2) 
$$\sum_{K \ge 1} \sum_{\boldsymbol{d}_K} (m_{\boldsymbol{d}_K} + 1) \sum_{I < J} \sum_{\ell=0}^{d_I - 1} \sum_{m=0}^{d_J - 1} \widetilde{\mathcal{M}}_{h-1,k-1,l+2} (\boldsymbol{b}, \boldsymbol{m} + s_{I,J,\ell,m}^{\times}(\boldsymbol{m})).$$

For the removal of the latter kind of twisted chords, we must start with a diagram with a boundary component with the marked point spectrum

$$(d_1, \dots, d_{I-1}, \ell, d_I - \ell - m - 2, m, d_{I+1}, \dots, d_K),$$
  
 $0 \le \ell, m \le d_I, \quad \ell + m \le d_I - 2.$ 

from which we remove one twisted chords with one end between the two clusters  $(\ell, d_I - \ell - m - 2)$  and the other one between the two clusters  $(d_I - \ell - m - 2, m)$ . After removal, we obtain a boundary component with the marked point spectrum  $\boldsymbol{d}_K$ . Thus, the boundary length and point spectrum  $\boldsymbol{m}$  is changed to

$$m - e_{(d_1,...,d_{I-1},\ell,d_I-\ell-m-2,m,d_{I+1},...,d_K)} + e_{d_K} = m + s_{I,\ell,m}^{\times}(d_K).$$

The number of such chords to be removed is  $m_{d_{\kappa}} + 1$ , and the total number of partial chord diagrams obtained in this way is

(3.3) 
$$\sum_{K \ge 1} \sum_{\boldsymbol{d}_K} (m_{\boldsymbol{d}_K} + 1) \sum_{I=1}^K \sum_{\ell+m \le d_I - 2} \widetilde{\mathcal{M}}_{h-1,k-1,l+2} (\boldsymbol{b}, \boldsymbol{m} + s_{I,\ell,m}^{\times}(\boldsymbol{m})).$$

Next, we consider the second case (see Figure 6), where we must start with a non-oriented partial chord diagram with a boundary component with the marked point spectrum

$$(f_1, \dots, f_{J-1}, f_J - m - 1, \ell, d_{I-1}, \dots, d_1, d_K, \dots, d_{I+1}, d_I - \ell - 1, m, f_{J+1}, \dots, f_L), 0 \le \ell \le d_I - 1, \quad 0 \le m \le f_J - 1,$$

from which we remove a twisted chord with one end between the two clusters  $(f_J - m - 1, \ell)$  and the other end between the two clusters  $(d_I - \ell - 1, m)$ . After

removal of this chord, the boundary component has been split into two components with spectra  $d_K$  and  $f_L$ , and the cross-cap number h decreases by two. Then, the boundary length and point spectrum m changes to

$$m - e_{(f_1,...,f_{J-1},f_J - m - 1,\ell,d_{I-1},...,d_1,d_K,...,d_{I+1},d_I - \ell - 1,m,f_{J+1},...,f_L)} + e_{d_K} + e_{f_L}$$
  
=  $m + q_{I,J,\ell,m}^{\times}(d_K, f_L).$ 

The number of choices for the chord to be removed is  $(m_{d_K} + 1)(m_{f_L} + 1)$  for  $d_K \neq f_L$  and  $m_{d_K}(m_{d_K} + 1)/2$  for  $d_K = f_L$ , and the total number of partial chord diagrams obtained this way is

(3.4) 
$$\frac{1}{2} \sum_{K \ge 1} \sum_{L \ge 1} \sum_{\boldsymbol{d}_K} \sum_{\boldsymbol{f}_L} (m_{\boldsymbol{d}_K} + 1)(m_{\boldsymbol{f}_L} + 1 - \delta_{\boldsymbol{d}_K, \boldsymbol{f}_L}) \\ \times \sum_{I=1}^K \sum_{J=1}^L \sum_{\ell=0}^{d_I - 1} \sum_{m=0}^{f_J - 1} \widetilde{\mathcal{M}}_{h-2, k-1, l+2} (\boldsymbol{b}, \boldsymbol{m} + q_{I, J, \ell, m}^{\times}(\boldsymbol{d}_K, \boldsymbol{f}_L)).$$



Figure 6: The second case of a twisted chord removal. After the removal of this chord, the boundary component split into two distinct boundary components.

In case three partial chord diagram split into two connected components when we remove the chord. Assume that the original diagram has the type  $\{h, k, l; \boldsymbol{b}, \boldsymbol{m}\}$ and the resulting two connected components have types  $\{h_1, k_1, l_1; \boldsymbol{b}^{(1)}, \boldsymbol{m}^{(1)}\}$  and  $\{h_2, k_2, l_2; \boldsymbol{b}^{(2)}, \boldsymbol{m}^{(2)}\}$ . Then these types are related by

$$h = h_1 + h_2, \quad k - 1 = k_1 + k_2, \quad \boldsymbol{b} = \boldsymbol{b}^{(1)} + \boldsymbol{b}^{(2)}.$$

The marked point spectrum changes in the same way as the second case

$$\boldsymbol{m} + q_{L,L\ell,m}^{\times}(\boldsymbol{d}_{K},\boldsymbol{f}_{L}) = \boldsymbol{m}^{(1)} + \boldsymbol{m}^{(2)}$$

The total number of resulting diagrams is

$$(3.5) \qquad \frac{1}{2} \sum_{K \ge 1} \sum_{L \ge 1} \sum_{\boldsymbol{d}_{K}} \sum_{\boldsymbol{f}_{L}} \sum_{h_{1}+h_{2}=h} \sum_{k_{1}+k_{2}=k-1}^{N} \sum_{\boldsymbol{b}^{(1)}+\boldsymbol{b}^{(2)}=\boldsymbol{b}} \sum_{K \ge L} \sum_{I=1}^{L} \sum_{J=1}^{L} \sum_{\ell=0}^{d_{I}-1} \sum_{\boldsymbol{m}=0}^{f_{J}-1} \sum_{\substack{\boldsymbol{m}^{(1)}+\boldsymbol{m}^{(2)}\\ =\boldsymbol{m}+q_{I,J,\ell,m}^{(1)}(\boldsymbol{d}_{K},\boldsymbol{f}_{L})}} \sum_{\boldsymbol{m}^{(1)}+\boldsymbol{m}^{(2)}_{\boldsymbol{f}_{L}} \sum_{\boldsymbol{b}^{(1)}|\boldsymbol{b}^{(2)}|} \widetilde{\mathcal{M}}_{h_{1},k_{1},l_{1}}(\boldsymbol{b}^{(1)},\boldsymbol{m}^{(1)}) \widetilde{\mathcal{M}}_{h_{2},k_{2},l_{2}}(\boldsymbol{b}^{(2)},\boldsymbol{m}^{(2)})}$$

Therefore, in total, the number of possible partial chord diagrams obtained by removing a twisted or a non-twisted chord is the sum of (3.2) - (3.5) and of (2.2) - (2.4) for  $\widetilde{\mathcal{M}}_{h,k,l}(\boldsymbol{b},\boldsymbol{m})$ . This number gives the right hand side of equation (3.1), which we have just argued also gives the left side of equation (3.1).

Along the same line of arguments as the ones which proved Proposition 2.2, we obtain the proposition below.

**Proposition 3.2.** The generating function  $\widetilde{H}(x, y; t, ; u)$  is uniquely determined by the differential equation

$$\frac{\partial \widetilde{H}}{\partial y} = (\widetilde{M} + \widetilde{S})\widetilde{H},$$

where  $\widetilde{M} = M_0 + xM_1^{\times} + x^2(M_2 + M_2^{\times})$  and  $\widetilde{S} = S + S^{\times}$ . The generating function  $\widetilde{Z}(x, y; t, ; u) = \exp[\widetilde{H}]$  of the number of connected and disconnected partial chord diagrams filtered by the boundary length and point spectrum satisfies

(3.6) 
$$\frac{\partial \widetilde{Z}}{\partial y} = \widetilde{M}\widetilde{Z}$$

As such they are uniquely determined by the initial conditions

$$\widetilde{H}(x, y = 0; \boldsymbol{t}; \boldsymbol{u}) = \sum_{i \ge 1} t_i u_{(i)}, \quad \widetilde{Z}(x, y = 0; \boldsymbol{t}; ; \boldsymbol{u}) = e^{\sum_{i \ge 1} t_i u_{(i)}}.$$

### References

N. V. Alexeev, J. E. Andersen, R. C. Penner, and P. Zograf, *Enumeration of chord diagrams on many intervals and their non-orientable analogs*, Adv. Math. **289** (2016) 1056–1081, arXiv:1307.0967 [math.CO].

- [2] J. E. Andersen, A. J. Bene, J. -B. Meilhan, and R. C. Penner, *Finite type invariants and fatgraphs*, Adv. Math. **225** (2010), 2117–2161, arXiv:0907.2827 [math.GT].
- [3] J. E. Andersen, L. O. Chekhov, R. C. Penner, C. M. Reidys, and P. Sulkowski, *Topological recursion for chord diagrams, RNA complexes, and cells in moduli* spaces, Nucl. Phys. B866 (2013) 414-443, arXiv:1205.0658 [hep-th].
- [4] J. E. Andersen, L. O. Chekhov, R. C. Penner, C. M. Reidys, and P. Sulkowski, Enumeration of RNA complexes via random matrix theory, Biochem. Soc. Trans. 41 (2013) 652-655, arXiv:1303.1326 [q-bio.QM].
- [5] J. E. Andersen, H. Fuji, M. Manabe, R. C. Penner, and P. Sulkowski, Enumeration of chord diagrams via topological recursion and quantum curve techniques, Travaux Mathématiques, 25 (2017), 285–323, arXiv:1612.05839 [math-ph].
- [6] J. E. Andersen, H. Fuji, M. Manabe, R. C. Penner, P. Sulkowski, *Partial chord diagrams and matrix models*, Travaux Mathématiques, **25** (2017), 233–283, arXiv:1612.05840 [math-ph].
- [7] J. E. Andersen, F. W. D. Huang, R. C. Penner, and C. M. Reidys, *Topology of RNA-RNA interaction structures*, J. Comp. Biol. **19** (2012) 928-943, arXiv:1112.6194 [math.CO].
- [8] J. E. Andersen, J. Mattes, and N. Reshetikhin, *The Poisson structure on the moduli space of flat connections and chord diagrams*, Topology **35** (1996) 1069–1083.
- J. E. Andersen, J. Mattes, and N. Reshetikhin, Quantization of the algebra of chord diagrams, Math. Proc. Camb. Phil. Soc. 124 (1998) 451–467, arXiv:qalg/9701018.
- [10] J. E. Andersen, R. C. Penner, C. M. Reidys, R. R. Wang, *Linear chord diagrams on two intervals*, arXiv:1010.5857 [math.CO].
- [11] J. E. Andersen, R. C. Penner, C. M. Reidys, M. S. Waterman, *Enumeration of linear chord diagrams*, J. Math. Biol. **67** (2013) 1261–78, arXiv:1010.5614 [math.CO].
- [12] D. Bar-Natan, On the Vassiliev knot invariants, Topology 34 (1995) 423-475.
- [13] R. Campoamor-Stursberg, and V. O. Manturov, *Invariant tensor formulas via chord diagrams*, Jour. Math. Sci. **108** (2004) 3018–3029.
- [14] J. Courtiel, and K. Yeats, *Terminal chords in connected chord diagrams*, arXiv:1603.08596 [math.CO].

- [15] M. Kontsevich, Vassilievs knot invariants, Adv. Sov. Math. 16 (1993) 137– 150.
- [16] N. Marie, and K. Yeats, A chord diagram expansion coming from some Dyson-Schwinger equations, Comm. Numb. Theo. Phys. 7 (2013) 251–291, arXiv:1210.5457 [math.CO].
- [17] H. Orland, and A. Zee, RNA folding and large N matrix theory, Nucl. Phys. B620 (2002) 456-476, arXiv:cond-mat/0106359 [cond-mat.stat-mech].
- [18] R. C. Penner, The decorated Teichmüller space of punctured surfaces, Comm. Math. Phys. 113 (1987) 299–339.
- [19] R. C. Penner, Perturbative series and the moduli space of Riemann surfaces, J. Diff. Geom. 27 (1988) 35–53.
- [20] R. C. Penner, The simplicial compactification of Riemann's moduli space, Proceedings of the 37th Taniguchi Symposium, World Scientific (1996), 237– 252.
- [21] R. C. Penner, Cell decomposition and compactification of Riemann's moduli space in decorated Teichmüller theory, In Tongring, N. and Penner, R.C. (eds) Woods Hole Mathematics-Perspectives in Math and Physics, World Scientific, Singapore, arXiv:math/0306190 [math.GT].
- [22] R. C. Penner, Moduli spaces and macromolecules, Bull. Amer. Math. Soc. 53 (2016) 217–268.
- [23] C. M. Reidys, Combinatorial and computational biology of pseudoknot RNA, Springer, Applied Math series 2010.
- [24] G. Vernizzi, and H. Orland, Large-N random matrices for RNA folding, Acta Phys. Polon. B36 (2005) 2821-2827.
- [25] G. Vernizzi, H. Orland, and A. Zee, Enumeration of RNA structures by matrix models, Phys. Rev. Lett. 94 168103, arXiv:q-bio/0411004 [q-bio.BM].

Jørgen Ellegaard Andersen Centre for Quantum Geometry of Moduli Spaces, Department of Mathematics, Aarhus University, DK-8000 Aarhus C Denmark jea.qgm@gmail.com

Hiroyuki Fuji Faculty of Education, Kagawa University, Takamatsu 760-8522 Japan fuji@ed.kagawa-u.ac.jp

Robert C. Penner Institut des Hautes Études Scientifiques, 35 route de Chartres, 91440 Burs-sur-Yvette, France; Division of Physics, Mathematics and Astronomy, California Institute of Technology, Pasadena, CA 91125, USA rpenner@caltech.edu, rpenner@ihes.fr

Christian M. Reidys Biocomplexity Institute of Virginia Tech Blacksburg, VA 24061, USA duckcr@vbi.vt.edu

232

### Partial Chord Diagrams and Matrix Models

## by Jørgen Ellegaard Andersen, Hiroyuki Fuji, Masahide Manabe, Robert C. Penner, and Piotr Sułkowski<sup>1</sup>

#### Abstract

In this article, the enumeration of partial chord diagrams is discussed via matrix model techniques. In addition to the basic data such as the number of backbones and chords, we also consider the Euler characteristic, the backbone spectrum, the boundary point spectrum, and the boundary length spectrum. Furthermore, we consider the boundary length and point spectrum that unifies the last two types of spectra. We introduce matrix models that encode generating functions of partial chord diagrams filtered by each of these spectra. Using these matrix models, we derive partial differential equations – obtained independently by cut-and-join arguments in an earlier work – for the corresponding generating functions.

## 1 Introduction

A partial chord diagram is a special kind of graph, which is specified as follows. The graph consists of a number of line segments (which are called *backbones*) arranged along the real line (hence they come with an ordering), with a number of vertices on each. A number of semi-circles (called *chords*) arranged in the upper half plane is attached at a subset of the vertices of the line segments, in such a way that no two chords have endpoints at the same vertex. The vertices which are not attached to chord ends are called the marked points. A *chord diagram* is by

<sup>&</sup>lt;sup>1</sup>Keywords: chord diagram, fatgraph, matrix model,

AMS Classification: 05A15, 05A16, 81T18, 81T45, 92-08

Acknowledgments: JEA and RCP is supported in part by the center of excellence grant "Center for Quantum Geometry of Moduli Spaces" from the Danish National Research Foundation (DNRF95). The research of HF is supported by the Grant-in-Aid for Research Activity Start-up [# 15H06453], Grant-in-Aid for Scientific Research(C) [# 26400079], and Grant-in-Aid for Scientific Research(B) [# 16H03927] from the Japan Ministry of Education, Culture, Sports, Science and Technology. The work of MM and PS is supported by the ERC Starting Grant no. 335739 "Quantum fields and knot homologies" funded by the European Research Council under the European Union's Seventh Framework Programme. PS also acknowledges the support of the Foundation for Polish Science, and RCP acknowledges the kind support of Institut Henri Poincaré where parts of this manuscript were written.

definition a partial chord diagram with no marked points. Partial chord diagrams occur in many branches of mathematics, including topology [14, 30], geometry [9, 10, 3] and representation theory [16].

To each partial chord diagram c one can associate canonically a two dimensional surface with boundary  $\Sigma_c$ , see Figure 1. Moreover, as discussed in [56, 12, 2, 7], the notion of a *fatgraph* [42, 43, 44, 45] is a useful concept when studying partial chord diagrams. A fatgraph is a graph together with a cyclic ordering on each collection of half-edges incident on a common vertex. A partial linear chord diagram c has a natural fatgraph structure induced from its presentation in the plane.



Figure 1: The partial chord diagram (with marked points) c and the corresponding surface  $\Sigma_c$ . The type of this partial chord diagram reads  $\{g, k, l; \{b_i\}; \{n_i\}; \{p_i\}\} = \{1, 6, 2; \{b_6 = 1, b_8 = 1\}; \{n_0 = 2, n_1 = 2\}; \{p_1 = 1, p_2 = 2, p_9 = 1\}\}$ . The boundary length and point spectrum is  $\{n_{(1)} = 1, n_{(0,0)} = 2, n_{(0,0,0,0,1,0,0)} = 1\}$ .

The partial chord diagram c is characterized by various topological data, and we will consider the following five types of data, introduced in [2] and [7].

- The number of chords k in c and the number of backbones b in c.
- Euler characteristic  $\chi$  and genus g. Let  $\chi$  and g denote respectively the Euler characteristic and genus of  $\Sigma_c$ , which are related as follows

$$\chi = 2 - 2g.$$

Denoting by n the number of boundary components of  $\Sigma_c$ , the Euler relation can be written as

(1.1) 
$$2 - 2g = b - k + n.$$

• Backbone spectrum  $(b_0, b_1, \ldots)$ .

Let  $b_i$  denote the number of backbones with *i* trivalent (i.e. chord ends) or bivalent (i.e. marked points) vertices. The total number of backbones *b* is then

$$(1.2) b = \sum_{i>0} b_i,$$

and the total number m of trivalent (i.e. chord ends) and bivalent (i.e. marked points) vertices of the partial chord diagram c is

$$(1.3) m = \sum_{i \ge 1} i b_i.$$

 Boundary point spectrum (n<sub>0</sub>, n<sub>1</sub>,...). Let n<sub>i</sub> denote the number of boundary components containing i ≥ 0 marked points of Σ<sub>c</sub>. The total number n of boundary components is

(1.4) 
$$n = \sum_{i \ge 0} n_i,$$

and the total number l of marked points is

$$(1.5) l = \sum_{i \ge 1} i n_i.$$

These three numbers m, k and l satisfies

$$(1.6) m = 2k + l.$$

• Boundary length spectrum  $(p_1, p_2, \ldots)$ . Define the *length* of a boundary component to be the sum of the number of chords and the number of backbone undersides traversed by the boundary cycle. Let  $p_i$  be the number of boundary cycles with length  $i \ge 1$ . By definition, the following two relations hold

(1.7) 
$$n = \sum_{i \ge 1} p_i,$$

$$(1.8) 2k+b=\sum_{i\geq 1}ip_i.$$

The data  $\{g, k, l; \{b_i\}; \{n_i\}; \{p_i\}\}$  is called the *type* of a partial chord diagram c.

As a unification of the boundary length spectrum and the boundary point spectrum, we consider the *boundary length and point spectrum* introduced in [7]. Let us here recall its definition.

- Boundary length and point spectrum.
  - We associate a K-tuple of numbers  $\mathbf{i} = (i_1, \ldots, i_K)$  with a boundary component of length K, where  $i_L$   $(L = 1, \ldots, K)$  is the number of marked points between the L'th and (L+1)'th (taken modulo K) either chord or underpass of a backbone component (in either order) along the boundary.

Let  $n_i$  be the number of boundary components labeled in this way by i. The total number l of marked points is

(1.9) 
$$l = \sum_{K \ge 1} \sum_{i} \sum_{L=1}^{K} i_L n_{(i_1, \dots, i_K)},$$

and the total number n of boundary cycles is

(1.10) 
$$n = \sum_{i} n_{i}.$$

The data  $\{g, k, l; \{b_i\}, \{n_i\}\}$  stores more detailed information on the distribution of marked points on each boundary component. One can determine the previous two kinds of spectra from the boundary length and point spectrum by forgetting the partitions of marked points on the boundary cycles.

It is known that the enumeration of chord diagrams is intimately related to matrix models and cut-and-join equations [4, 5, 6, 20, 38]. In this paper, the enumeration of partial chord diagrams labeled by the boundary length and point spectrum with the genus filtration is studied using matrix model techniques. Let  $\mathcal{N}_{g,k,l}(\{b_i\}, \{n_i\})$  denote the number of connected chord diagrams labeled by the set of parameters  $(g, k, l; \{b_i\}; \{n_i\})$ . We define the generating function of these numbers

(1.11)  

$$\mathcal{F}(x, y; \{s_i\}; \{u_i\}) = \sum_{b \ge 1} \mathcal{F}_b(x, y; \{s_i\}; \{u_i\}),$$

$$\mathcal{F}_b(x, y; \{s_i\}; \{u_i\}) = \frac{1}{b!} \sum_{\sum_i b_i = b} \sum_{\{n_i\}} \mathcal{N}_{g,k,l}(\{b_i\}, \{n_i\}) x^{2g-2} y^k \prod_{i \ge 0} s_i^{b_i} \prod_{K \ge 1} \prod_{\{i_L\}_{L=1}^K} u_i^{n_i}.$$

Generating functions of disconnected and connected diagrams are related via the exponential relation

(1.12) 
$$\mathcal{Z}(x, y; \{s_i\}; \{u_i\}) = \exp\left[\mathcal{F}(x, y; \{s_i\}; \{u_i\})\right].$$

To analyze this enumeration further, we write the above generating function as a certain Hermitian matrix integral. Let  $\mathcal{Z}_N(y; \{s_i\}; \{u_i\})$  be the matrix integral over rank N Hermitian matrices  $\mathcal{H}_N$ 

$$\mathcal{Z}_N(y; \{s_i\}; \{u_i\}) =$$
(1.13) 
$$= \frac{1}{\operatorname{Vol}_N} \int_{\mathcal{H}_N} dM \exp\left[-N\operatorname{Tr}\left(\frac{M^2}{2} - \sum_{i\geq 0} s_i(y^{1/2}\Lambda_{\mathrm{L}}^{-1}M + \Lambda_{\mathrm{P}})^i\Lambda_{\mathrm{L}}^{-1}\right)\right],$$

where  $\Lambda_{\rm P}$  and  $\Lambda_{\rm L}$  are *external matrices* [29] of rank N, and the normalization factor Vol<sub>N</sub> is defined in (2.4). In this matrix integral representation, the counting

parameter  $u_{(i_1,\ldots,i_K)}$  is identified with the trace of the corresponding product of external matrices

(1.14) 
$$u_{(i_1,\ldots,i_K)} = \frac{1}{N} \operatorname{Tr} \left( \Lambda_{\mathrm{P}}^{i_1} \Lambda_{\mathrm{L}}^{-1} \Lambda_{\mathrm{P}}^{i_2} \Lambda_{\mathrm{L}}^{-1} \cdots \Lambda_{\mathrm{P}}^{i_K} \Lambda_{\mathrm{L}}^{-1} \right).$$

In Theorem 2.13 we show that

(1.15) 
$$\mathcal{Z}_N(y; \{s_i\}; \{u_i\}) = \mathcal{Z}(N^{-1}, y; \{s_i\}; \{u_i\}).$$



Figure 2: The cut-and-join manipulations on chord diagrams.

This matrix integral representation provides a new, matrix model proof of the cut-and-join equation found by combinatorial means in [7]. The cut-and-join equation can be written as

(1.16) 
$$\frac{\partial}{\partial y} \mathcal{Z}(x, y; \{s_i\}; \{u_i\}) = \mathcal{M} \mathcal{Z}_N(x, y; \{s_i\}; \{u_i\}),$$

where  $\mathcal{M}$  is the second order partial differential operator in variables  $u_i$  (see Theorem 3.11 for details). This cut-and-join equation can be regarded as the evolution equation in the variable y, and its formal solution reads

(1.17) 
$$\begin{aligned} \mathcal{Z}(x, y; \{s_i\}; \{u_i\}) &= \mathrm{e}^{y\mathcal{M}} \mathcal{Z}(x, 0; \{s_i\}; \{u_i\}), \\ \mathcal{Z}(x, 0; \{s_i\}; \{u_i\}) &= \mathrm{e}^{N^2 \sum_{i \ge 0} s_i u_{(i)}}. \end{aligned}$$

Expanding the operator  $e^{y\mathcal{M}}$  around y = 0, one determines the number of connected partial chord diagrams  $\mathcal{N}_{g,k,l}(\{b_i\}, \{n_i\})$  iteratively from this formal solution. The cut-and-join equation is a powerful method to systematically count partial chord diagrams of a given length and point spectrum.

In this work we also generalize the above analysis to non-oriented analogues of partial chord diagrams. By non-oriented partial chord diagrams we mean diagrams with all chords decorated by a binary variable, which indicates if they are *twisted* or not. When associating the surface  $\Sigma_c$  to a non-oriented partial chord diagram, twisted bands are associated along the twisted chords as indicated in Figure 3. This construction leads to  $2^k$  orientable or non-orientable surfaces associated to one particular partial chord diagram with k chords, if we consider all possible assignments of twisting or untwisting of k bands. In the non-oriented case the Euler characteristic is defined as follows. • Euler characteristic  $\chi$ . The Euler characteristic of the two dimensional surface  $\Sigma_c$  is defined by the formula

$$\chi = 2 - h$$
,

where h is the number of cross-caps. The Euler relation holds

(1.18) 2 - h = b - k + n.

With this setup, the enumeration of non-oriented partial chord diagrams can be considered analogously to the orientable case.



Figure 3: A non-oriented surface  $\Sigma_c$  associated to a non-oriented partial chord diagram c.

Let  $\widetilde{\mathcal{N}}_{h,k,l}(\{b_i\}, \{n_i\})$  denote the number of connected non-oriented partial chord diagrams with the cross-cap number h, k chords, the backbone spectrum  $\{b_i\}$ , l marked points, and the boundary length and point spectrum  $n_i$ . The generating function  $\widetilde{\mathcal{F}}(x, y; \{s_i\}; \{u_i\})$  is defined by

$$\begin{aligned} &(1.19)\\ \widetilde{\mathcal{F}}(x,y;\{s_i\};\{u_{\boldsymbol{i}}\}) = \sum_{b\geq 1} \widetilde{\mathcal{F}}_b(x,y;\{s_i\};\{u_{\boldsymbol{i}}\}),\\ &\widetilde{\mathcal{F}}_b(x,y;\{s_i\};\{u_{\boldsymbol{i}}\}) = \frac{1}{b!} \sum_{\sum_i b_i = b} \sum_{\{n_{\boldsymbol{i}}\}} \widetilde{\mathcal{N}}_{h,k,l}(\{b_i\},\{n_{\boldsymbol{i}}\}) x^{h-2} y^k \prod_{i\geq 0} s_i^{b_i} \prod_{K\geq 1} \prod_{\{i_L\}_{L=1}^K} u_{\boldsymbol{i}}^{n_{\boldsymbol{i}}}. \end{aligned}$$

We also define the generating function of the numbers of connected and disconnected non-oriented partial chord diagrams

(1.20) 
$$\widetilde{\mathcal{Z}}(x, y; \{s_i\}; \{u_i\}) = \exp\left[\widetilde{\mathcal{F}}(x, y; \{s_i\}; \{u_i\})\right].$$

In Theorem 4.5 we show that this generating function can be expressed as a real symmetric matrix integral with two external symmetric matrices  $\Omega_P$  and  $\Omega_L$ 

$$\widetilde{\mathcal{Z}}(N^{-1}, y; \{s_i\}; \{u_i\}) = \widetilde{\mathcal{Z}}_N(y; \{s_i\}; \{u_i\}),$$

238

Partial Chord Diagrams and Matrix Models

$$\begin{aligned} \widetilde{\mathcal{Z}}_N(y; \{s_i\}; \{u_i\}) &= \\ (1.22) \\ &= \frac{1}{\operatorname{Vol}_N(\mathbb{R})} \int_{\mathcal{H}_N(\mathbb{R})} dM \, \exp\bigg[ -N\operatorname{Tr}\bigg(\frac{M^2}{4} - \sum_{i\geq 0} s_i (y^{1/2}\Omega_{\mathrm{L}}^{-1}M + \Omega_{\mathrm{P}})^i \Omega_{\mathrm{L}}^{-1} \bigg) \bigg], \end{aligned}$$

where the normalization factor  $\operatorname{Vol}_N(\mathbb{R})$  is defined in (4.8), and  $\mathcal{H}_N(\mathbb{R})$  is the space of real symmetric matrices of rank N. The parameter  $u_{(i_1,\ldots,i_K)}$  is identified with a trace of the external matrices via the formula

(1.23) 
$$u_{(i_1,\ldots,i_K)} = \frac{1}{N} \operatorname{Tr} \left( \Omega_{\mathrm{P}}^{i_1} \Omega_{\mathrm{L}}^{-1} \Omega_{\mathrm{P}}^{i_2} \Omega_{\mathrm{L}}^{-1} \cdots \Omega_{\mathrm{P}}^{i_K} \Omega_{\mathrm{L}}^{-1} \right).$$

Using this matrix integral representation of the generating function, one can again prove the cut-and-join equation, established independently by combinatorial arguments in [7]

(1.24) 
$$\frac{\partial}{\partial y}\widetilde{\mathcal{Z}}_N(y;\{s_i\};\{u_i\}) = \widetilde{\mathcal{M}}\widetilde{\mathcal{Z}}_N(y;\{s_i\};\{u_i\}),$$

where  $\mathcal{M}$  is a second order partial differential operator in the variables  $u_i$ . The details of the differential operator  $\mathcal{M}$  and the matrix model derivation of the cut-and-join equation are presented in Theorem 4.10.

### 1.1 Motivation: RNA chains

One important motivation to study partial chord diagrams in this and the preceding work [2, 7] is a complicated problem of RNA structure prediction in molecular biology, which we now shortly review.

An RNA molecule is a linear polymer, referred to as the backbone, that consists of four types of nucleotides: adenine, cytosine, guanine, and uracil, denoted respectively **A**, **C**, **G**, and **U**. The backbone is endowed with an orientation from 5'-end to 3'-end, and the primary sequence is the sequence of nucleotides read with respect to this orientation. Between nucleotides hydrogen bonds are formed, resulting in the so-called Watson-Click pairs involving  $\mathbf{A} - \mathbf{U}$  or  $\mathbf{G} - \mathbf{C}$  nucleotides; in addition Wobble pairs  $\mathbf{U} - \mathbf{G}$  can be formed. The set of base pairs formed by such hydrogen bonds is referred to as the secondary structure.<sup>2</sup> Prediction of the secondary structure from the primary sequence is an outstanding problem that was initiated by the pioneering work of Michael Waterman [57] and has been studied intensively for last three decades.

Topologically, we can represent the base pairings for a given RNA structure by a partial chord diagram as follows. The backbone is represented as a disjoint union of horizontal straight line segments (arranged along the real line in the

<sup>&</sup>lt;sup>2</sup>There are other types of interactions in RNA secondary structure, which are however less common and we ignore them in this discussion.



Figure 4: Pseudoknot structures in RNA. The long curved line, blobs (i.e. marked points), and short lines represent the backbone, nucleotides, and base pairs, respectively.

plane), one for each backbone component, and each nucleotide is represented as a marked point on this union of line segments. The base pairs are represented by chords in the upper-half plane attached at two marked points corresponding to the bonded pair of nucleotides.

Note that a partial chord diagram has genus zero if no two of its chords cross each other. If however such crossings exist, then the structure is referred to as a pseudoknot, and its genus is non-zero. Considerable number of pseudoknot structures have been observed, e.g. tRNAs, RNAseP [31], telomerase RNA [53] and ribosomal RNAs [28]. According to the online database "RNA-strand" half of the known structures form pseudoknots [13]. There are various kinds of pseudoknots classified by the topology of the RNA [12], referred to as e.g. H-type [1], kissing hairpin [17, 51], etc.

In recent years, a combinatorial description of RNA structures in terms of linear chord diagrams has been developed in a series of works [41, 56, 55, 12, 11, 8, 4, 5, 2, 49, 46]. However, a large class of reasonable energy-based models that predict the secondary structure including pseudoknots are NP complete [32, 1], and a fully satisfactory energy model for RNA, including pseudoknot structures, has not been established yet.

In the search of a realistic energy function for RNA structures with pseudoknots, the boundary length and point spectrum should provide a useful tool that includes more detailed information about the location of marked points. In standard algorithms developed by Waterman [58], Nussinov et al. [40], Zucker and Stiegler [61], etc., dynamic programming (DP) has been used to predict most likely secondary structures. Indeed, in famous algorithms such as [60, 25], the (loop-based) energy in each configuration of RNA is considered. In these algorithms, the most probable secondary structure is determined as the minimum free energy configuration, and to make them more efficient the statistical mechanical



Figure 5: Partial chord diagrams unveil the difference in the topological structure of RNA molecules.

ensemble (i.e. the partition function algorithm) is implemented [34]. The application of these algorithms, which include pseudoknot structures stratified by  $\gamma$  structures, was studied in [50, 49]. Most of the energy functions essentially respect the boundary point and length spectra independently. In order to improve the energy model for RNA structure prediction with pseudoknots, it would be useful to explore energy parameters for more realistic and efficient energy function on the basis of the boundary length and point spectrum.

### 1.2 Plan of the paper

This paper is organized as follows. In Section 2 we construct Hermitian matrix models with external matrices, which encode generating functions of orientable partial chord diagrams labeled by the boundary point spectrum (in Subsection 2.1), the boundary length spectrum (in Subsection 2.2), and the boundary length and point spectrum (in Subsection 2.3). All these constructions are established by the correspondence between chord diagrams and Wick contractions via the Wick theorem. The matrix model encoding the boundary length and point spectrum is given in Theorem 2.13. In Section 3 we derive partial differential equations for matrix integrals found in Section 2. These partial differential equations coincide with the cut-and-join equations found combinatorially in [2, 7]. The cut-and-join equation for partial chord diagrams labeled by the boundary length and point spectrum is determined in Theorem 3.11. Section 4 is devoted to the analysis of non-oriented analogues of the results obtained in Section 2 and 3. In Subsection 4.1 we find real symmetric matrix models with external matrices, that encode generating functions of both orientable and non-orientable partial chord diagrams. The non-oriented analogue of the matrix integral from Theorem 2.13 is given in Theorem 4.5. Non-oriented analogues of cut-and-join equations from Section 3 are determined in Theorem 4.10. In Appendix A we derive a partial differential equation from Proposition 4.7 for a real symmetric matrix integral with external matrices. In Appendix B we prove Lemma 4.9.

# 2 Enumerating partial linear chord diagrams via matrix models

The enumeration problem of partial chord diagrams with respect to the genus filtration has been reformulated in terms of matrix integrals. Matrix model techniques for enumeration of the RNA structures with pseudoknots have been developed in a series of papers [41, 56, 55], and independently in [4, 5, 6]. Subsequently the analysis involving boundary point and length spectra of partial linear chord diagrams has been conducted in [2, 7]. In this section we develop a new perspective on this problem and construct a matrix model that enumerates partial chord diagrams labeled by the boundary length and point spectrum.

### 2.1 A matrix model enumerating partial chord diagrams

In the first step we construct a matrix model that counts partial chord diagrams labeled by the boundary point spectrum  $\{n_i\}$ .

**Definition 2.1.** Let  $\mathcal{N}_{g,k,l}(\{b_i\},\{n_i\},\{p_i\})$  denote the number of connected partial chord diagrams of type  $\{g,k,l;\{b_i\};\{n_i\};\{n_i\};\{p_i\}\}$ . In particular, focusing on the boundary point spectrum we define the following number of partial chord diagrams characterized by the data  $\{g,k,l;\{b_i\},\{n_i\}\}$ ,

$$\mathcal{N}_{g,k,l}(\{b_i\},\{n_i\}) = \sum_{\{p_i\}} \mathcal{N}_{g,k,l}(\{b_i\},\{n_i\},\{p_i\}).$$

We introduce the generating function<sup>3</sup> for the numbers  $\mathcal{N}_{g,k,l}(\{b_i\},\{n_i\})$ 

(2.1)  

$$F(x, y; \{s_i\}; \{t_i\}) = \sum_{b \ge 1} F_b(x, y; \{s_i\}; \{t_i\}),$$

$$F_b(x, y; \{s_i\}; \{t_i\}) = \frac{1}{b!} \sum_{\sum_i b_i = b} \sum_{\{n_i\}} \mathcal{N}_{g,k,l}(\{b_i\}, \{n_i\}) x^{2g-2} y^k \prod_{i \ge 0} s_i^{b_i} t_i^{n_i}.$$

The generating function for the numbers  $\widehat{\mathcal{N}}_{k,b,l}(\{b_i\},\{n_i\})$  of connected and disconnected partial chord diagrams arises in the usual way from the exponent

(2.2)  
$$Z^{P}(x, y; \{s_{i}\}; \{t_{i}\}) = \exp\left[F(x, y; \{s_{i}\}; \{t_{i}\})\right]$$
$$= \sum_{\{b_{i}\}} \sum_{\{n_{i}\}} \widehat{\mathcal{N}}_{k,b,l}(\{b_{i}\}, \{n_{i}\}) x^{-b+k-n} y^{k} \prod_{i \ge 0} s_{i}^{b_{i}} t_{i}^{n_{i}}.$$

<sup>3</sup>The parameters  $s_i$  and  $t_i$  in this article and in [2] are related by  $s_i \leftrightarrow t_i$ .

In the following we rewrite the generating function  $Z^{P}(x, y; \{s_i\}; \{t_i\})$  as a Hermitian matrix integral. To this end, we consider first Gaussian averages over Hermitian matrices.

**Definition 2.2.** Let  $\mathcal{O}(M)$  be a function of a rank N Hermitian matrix M. The Gaussian average  $\langle \mathcal{O}(M) \rangle_N^S$  is defined by the integral over the space  $\mathcal{H}_N$  of rank N Hermitian matrices with respect to the Haar measure dM with the Gaussian weight  $e^{-N \text{Tr} \frac{M^2}{2}}$ ,

(2.3) 
$$\langle \mathcal{O}(M) \rangle_N^{\mathrm{G}} = \frac{1}{\mathrm{Vol}_N} \int_{\mathcal{H}_N} dM \ \mathcal{O}(M) \,\mathrm{e}^{-N\mathrm{Tr}\frac{M^2}{2}},$$

where the normalization factor  $Vol_N$  takes form

(2.4) 
$$\operatorname{Vol}_{N} = \int_{\mathcal{H}_{N}} dM \ \mathrm{e}^{-N \operatorname{Tr} \frac{M^{2}}{2}} = N^{N(N+1)/2} \operatorname{Vol}(\mathcal{H}_{N}).$$

In particular for  $\mathcal{O}(M) = M_{\alpha\beta}M_{\gamma\epsilon}$   $(\alpha, \beta, \gamma, \epsilon = 1, ..., N)$ , the Gaussian average is

(2.5) 
$$\overrightarrow{M_{\alpha\beta}M_{\gamma\delta}} := \langle M_{\alpha\beta}M_{\gamma\epsilon}\rangle_N^{\rm G} = \frac{1}{N}\delta_{\alpha\epsilon}\delta_{\beta\gamma}$$

This quantity is called the *Wick contraction*. By definition, a multiple Wick contraction is a product of the Gaussian average of each Wick contracted pair.

It follows from the definition (2.3) that Gaussian averages of an odd number of matrix elements vanish. On the other hand, Gaussian averages of an even number of matrix elements are non-zero, and can be computed using the *Wick theorem* [15, 43, 37], as we now recall. Consider an ordered sequence

$$M_{\alpha_1\beta_1}M_{\alpha_2\beta_2}\cdots M_{\alpha_{2k}\beta_{2k}}$$

of 2k matrix elements  $M_{\alpha_n\beta_n}$   $(n = 1, \ldots, 2k)$ .

Let  $P_k$  denote a set of matchings by k Wick contractions among the 2k matrix elements in the above sequence.  $P_k$  is isomorphic to the following quotient of groups

$$P_k \simeq G_H/G_E$$
,  $G_H = S_{2k}$ ,  $G_E = S_k \rtimes (S_2)^k$ .

Here the elements of the permutation group  $S_{2k}$  permute 2k matrix elements. The factors  $S_k$  of  $G_E$  act by permuting k Wick contractions and  $(S_2)^k$  swaps matrix elements in each Wick contracted pair. The Wick theorem implies the following result.

**Theorem 2.3.** The Gaussian average of 2k matrix elements  $M_{\alpha_n\beta_n}$  (n = 1, ..., k) equals

(2.6)  
$$\langle M_{\alpha_1\beta_1} M_{\alpha_2\beta_2} \cdots M_{\alpha_{2k}\beta_{2k}} \rangle_N^{\mathcal{G}} = \sum_{\sigma \in P_k} \prod_{i=1}^k \overline{M_{\alpha_{\sigma(2i-1)}\beta_{\sigma(2i-1)}}} \overline{M_{\alpha_{\sigma(2i)}\beta_{\sigma(2i)}}} \\ = \frac{1}{N^k} \sum_{\sigma \in P_k} \prod_{i=1}^k \delta_{\alpha_{\sigma(2i-1)}\beta_{\sigma(2i)}} \delta_{\alpha_{\sigma(2i)}\beta_{\sigma(2i-1)}}.$$

#### 2.1.1 Chord diagrams and Wick contractions

Let c be a chord diagram. We now recall the explicit relation between a surface  $\Sigma_c$  associated to a chord diagram c and k-matchings or Wick contractions in the Gaussian average. To illustrate this correspondence we depict chord ends on backbones in  $\Sigma_c$  as trivalent vertices that consist of upright and horizontal line segments, see Figure 6. This correspondence is specified by the following four points C1–C4.



Figure 6: Bijective correspondence between chord diagrams and Wick contractions.

- C1 A matrix element  $M_{\alpha\beta}$  corresponds to a chord end on a backbone. Indices  $\alpha, \beta(=1, \ldots, N)$  are assigned to two upright line segments on the upper edge of the backbone.
- C2 If two matrix elements  $M_{\alpha_{j}\beta_{j}}M_{\alpha_{j+1}\beta_{j+1}}$  correspond to two adjacent chord ends on the same backbone, then the following quantity is assigned to the horizontal segment between these two chord ends on the upper edge of the backbone

c

$$\sum_{\alpha_{j+1},\beta_j=1}^N \delta_{\beta_j \alpha_{j+1}}.$$

This assignment encodes matrix multiplication of matrix elements corresponding to adjacent chord ends on the backbone.

C3 For the product of i matrix elements M

$$\sum_{\alpha_2,\dots,\alpha_i=1}^N \sum_{\beta_1,\dots,\beta_{i-1}=1}^N M_{\alpha_1\beta_1}\delta_{\beta_1\alpha_2}M_{\alpha_2\beta_2}\dots\delta_{\beta_{i-1}\alpha_i}M_{\alpha_i\beta_i} = (M^i)_{\alpha_1\beta_i}$$

which corresponds to a backbone with i chord ends, the following quantity is assigned to the bottom edge of the backbone

$$N\sum_{\alpha_1,\beta_i=1}^N \delta_{\beta_i\alpha_1}.$$

Thus, a backbone with i chord ends corresponds to a single trace of the i'th power of M, namely  $N \text{Tr} M^i$ .

**C4** The Wick contraction between  $M_{\alpha_j\beta_j}$  and  $M_{\alpha'_{j'}\beta'_{j'}}$  corresponds to a band connecting two chord ends. Each Wick contraction imposes a constraint  $\delta_{\alpha_j\beta'_{j'}}\delta_{\alpha'_{j'}\beta_j}$  on matrix indices assigned to the edges of the chord ends matched by the Wick contraction.

The above rules imply the following bijective correspondence

(2.7) 
$$W_N^{\mathcal{C}}(\{b_i\}) = \left\langle \prod_i \left( N \operatorname{Tr} M^i \right)^{b_i} \right\rangle_N^{\mathcal{G}}, \qquad \sum_i i b_i = 2k,$$

between matchings by k Wick contractions in the Gaussian average on one hand, and chord diagrams that consist of  $b_i$  backbones with i chord ends on the other hand, see Figure 7.



Figure 7: Chord diagrams and Wick contractions for  $\langle N \text{Tr} M^4 \rangle_{\text{N}}^{\text{G}}$ .

The Wick contractions (2.6) in  $W_N^C(\{b_i\})$  replace all matrix elements M's by products of  $\delta$ 's, and summing over matrix indices along a boundary cycle one finds a factor of N corresponding to each boundary cycle in a chord diagram. Therefore the overall N dependence following from the above rules amounts to assigning  $N^{b-k+n}$  factor to the term  $W_N^C(\{b_i\})$ , corresponding to a chord diagram with backbone spectrum  $\{b_i\}$  and n boundary cycles. Combing the Wick theorem and this bijective correspondence between matchings by k Wick contractions in the Gaussian average  $W_N^C(\{b_i\})$  and the set of chord diagrams with backbone spectrum  $\{b_i\}$ , the following proposition follows. **Proposition 2.4.** The Gaussian average  $W_N^c(\{b_i\})$  in equation (2.7) agrees with the generating function of chord diagrams with backbone spectrum  $\{b_i\}$ 

(2.8) 
$$W_N^{\mathcal{C}}(\{b_i\}) = \sum_{n \ge 0} \widehat{\mathcal{N}}_{k,b,n}(\{b_i\}) N^{b-k+n}.$$

Here  $\widehat{\mathcal{N}}_{k,b,n}(\{b_i\})$  is the number of chord diagrams that consist of  $b_i$  backbones with *i* trivalent vertices

(2.9) 
$$\widehat{\mathcal{N}}_{k,b,n}(\{b_i\}) = \sum_{\{p_i\}} \widehat{\mathcal{N}}_{k,b,l=0}(\{b_i\}, n_0 = n, \{n_i = 0\}_{i \ge 1}, \{p_i\}).$$

#### 2.1.2 Partial chord diagrams and Wick contractions

We now generalize the above bijective correspondence to partial chord diagrams. Let c be a partial chord diagram. On the boundary cycles of the surface  $\Sigma_c$ we add additional marked points, which correspond to those marked points on c which are not chord ends. These marked points are represented by *external matrices*  $\Lambda_P$  of rank N in the Gaussian average. The rules **P1–P5** below provide the correspondence between partial chord diagrams with backbone spectrum  $\{b_i\}$ and matchings with k Wick contractions in the Gaussian average.

- **P1** A matrix element  $M_{\alpha\beta}$  corresponds to a chord end on a backbone. The graphical rule is the same as the rule **C1**.
- **P2** A matrix element  $\Lambda_{P\alpha\beta}$  corresponds to a marked point on a backbone in  $\Sigma_c$ . Indices  $\alpha, \beta(=1, \ldots, N)$  are assigned to two upright line segments at each marked point on the upper edge of the backbone, see Figure 8.
- **P3** To a line segment (on the upper edge of the backbone) between adjacent chord ends or marked points (located on the same backbone), corresponding to matrix elements  $U_{\alpha_i\beta_i}$  and  $V_{\alpha_{i+1}\beta_{i+1}}$  (for U, V = M or  $\Lambda_P$ ), we assign

(2.10) 
$$\sum_{\beta_j,\alpha_{j+1}=1}^N \delta_{\beta_j\alpha_{j+1}},$$

just as in C2.

**P4** Let  $v_j, w_j \in \mathbb{Z}_{\geq 0}$  (j = 1, ..., i) with  $\sum_{j=1}^{i} (v_j + w_j) = i$ . For an ordered matrix product

(2.11) 
$$(M^{v_1}\Lambda_{\mathbf{P}}^{w_1}M^{v_2}\Lambda_{\mathbf{P}}^{w_2}\cdots M^{v_i}\Lambda_{\mathbf{P}}^{w_i})_{\alpha_1\beta_i},$$

corresponding to a backbone which is an ordered sequence of  $v_j$  chord ends and  $w_j$  marked points, we assign

$$N\sum_{\alpha_1,\beta_i=1}^N \delta_{\beta_i\alpha_1}$$
to the bottom edge of this backbone. It follows that the trace

(2.12) 
$$N \operatorname{Tr}(M^{v_1} \Lambda_{\mathbf{P}}^{w_1} M^{v_2} \Lambda_{\mathbf{P}}^{w_2} \cdots M^{v_i} \Lambda_{\mathbf{P}}^{w_i})$$

is assigned to this backbone.

**P5** The Wick contraction between  $M_{\alpha_j\beta_j}$  and  $M_{\alpha'_{j'}\beta'_{j'}}$  corresponds to a band connecting two chord ends, and it is represented in the same way as specified in **C4**.



Figure 8: Bijective correspondence between partial chord diagrams and matchings of Wick contractions

For a fixed backbone spectrum  $\{b_i\}$ , all possible sequences  $\{\alpha_j, \beta_j\}$  in the expression (2.12) are generated by the following product of traces

(2.13) 
$$\prod_{i\geq 0} \left(N \operatorname{Tr}(M + \Lambda_{\mathrm{P}})^{i}\right)^{b_{i}}.$$

Hence, by the above rules, all partial chord diagrams with the backbone spectrum  $\{b_i\}$  correspond bijectively to all matchings by Wick contractions among the M's in the expansion of the Gaussian average

(2.14) 
$$W_N^{\mathbf{P}}(\{b_i\},\{r_i\}) = \left\langle \prod_{i\geq 0} \left(N \operatorname{Tr}(M+\Lambda_{\mathbf{P}})^i\right)^{b_i} \right\rangle_N^{\mathbf{G}},$$

where we introduced the reverse Miwa times

(2.15) 
$$r_i = \frac{1}{N} \text{Tr} \Lambda_{\rm P}^i$$

If there are  $n_i$  boundary components containing *i* marked points, then one finds a trace factor  $(\text{Tr}\Lambda_{\rm P}^{\mathbf{j}})^{n_i}$  in the corresponding term in the Gaussian average (2.14), see Figure 9. Therefore, for partial chord diagrams with the backbone spectrum  $\{b_i\}$  and the boundary point spectrum  $\{n_i\}$ , the corresponding term in  $W_N^{\rm P}(\{b_i\}, \{r_i\})$  contributes the factor

$$N^{b-k+n}\prod_{i\geq 0}r_i^{n_i}.$$

Figure 9: Partial chord diagrams of types  $\{g = 0, k = 1, l = 2; b_4 = 1; n_0 = 1, n_2 = 1\}$  and  $\{g = 0, k = 1, l = 2; b_4 = 1; n_1 = 2\}$ , and the corresponding Wick contractions.

Therefore, from Wick theorem and the above bijective correspondence between partial chord diagrams and matchings by Wick contractions, one finds the following proposition.

**Proposition 2.5.** The Gaussian average (2.14) is the generating function for the numbers  $\widehat{\mathcal{N}}_{k,b,l}(\{b_i\}, \{n_i\})$  of partial chord diagrams with the backbone spectrum  $\{b_i\}$  and the boundary point spectrum  $\{n_i\}$ 

(2.16) 
$$W_N^{\mathrm{P}}(\{b_i\},\{r_i\}) = \sum_{\{n_i\}} \widehat{\mathcal{N}}_{k,b,l}(\{b_i\},\{n_i\}) N^{b-k+n} \prod_{i \ge 0} r_i^{n_i},$$

where the summation is constrained by  $\sum in_i = \sum ib_i - 2k$ .

Using this proposition, we consider the full generating function  $Z_N^{\mathrm{P}}(y; \{s_i\}; \{r_i\})$  for the numbers  $\widehat{\mathcal{N}}_{k,b,l}(\{b_i\}, \{n_i\})$  of partial chord diagrams weighted by

$$N^{b-k+n}y^k \prod_{i\geq 0} s_i^{b_i} r_i^{n_i}.$$

Since the contribution from a partial chord diagram is invariant under permutations of its backbones, the full generating function

$$Z_N^{\mathcal{P}}(y; \{s_i\}; \{r_i\}) = \sum_{\{b_i\}} \sum_{\{n_i\}} \widehat{\mathcal{N}}_{k,b,l}(\{b_i\}, \{n_i\}) N^{b-k+n} y^k \prod_{i \ge 0} s_i^{b_i} r_i^n$$

can be rewritten as a sum over all backbone spectra  $\{b_i\}$  of the terms

$$y^{\sum_i ib_i/2} W_N^{\mathrm{P}}(\{b_i\}, \{y^{-i/2}r_i\}) \prod_i \frac{s_i^{b_i}}{b_i!}$$

It follows that

$$Z_{N}^{\mathbf{P}}(y; \{s_{i}\}; \{r_{i}\}) = \sum_{\{b_{i}\}} \prod_{i \ge 0} \frac{s_{i}^{b_{i}} y^{ib_{i}/2}}{b_{i}!} \Big\langle \Big( N \operatorname{Tr}(M + y^{-1/2} \Lambda_{\mathbf{P}})^{i} \Big)^{b_{i}} \Big\rangle_{N}^{\mathbf{G}}$$

Performing the summation over  $b_i$ 's, one finds that the full generating function is given by the matrix integral

(2.17) 
$$Z_N^{\rm P}(y; \{s_i\}; \{r_i\}) = \frac{1}{\operatorname{Vol}_N} \int_{\mathcal{H}_N} dM \exp\left[-N\operatorname{Tr}\left(\frac{M^2}{2} - \sum_{i\geq 0} s_i(y^{1/2}M + \Lambda_{\rm P})^i\right)\right]$$

This matrix integral and  $Z^{\mathbf{P}}(x, y; \{s_i\}; \{t_i\})$  in equation (2.2) are identified by a change of variables. Since the reverse Miwa time for i = 0 yields  $r_0 = 1$ automatically, we need to introduce the parameter  $t_0$  by the following change of variables

$$N \to t_0 N, \quad y \to t_0 y, \quad s_i \to t_0^{-1} s_i, \quad r_i \to t_0^{-1} t_i.$$

As a result, we find the main theorem in this subsection.

**Theorem 2.6.** The generating function (2.2) is given by the matrix integral (2.17),

(2.18) 
$$Z^{\mathcal{P}}(N^{-1}, y; \{s_i\}; \{t_i\}) = Z^{\mathcal{P}}_{t_0N}(t_0y; \{t_0^{-1}s_i\}; \{t_0^{-1}t_i\}).$$

#### 2.2 A matrix model for the enumeration of chord diagrams

Next we turn to the enumeration of chord diagrams labeled by the backbone spectrum  $\{b_i\}$  and the boundary length spectrum  $\{p_i\}$ . The number  $\mathcal{N}_{g,k}(\{b_i\}, \{p_i\})$  of connected chord diagrams is given by

$$\mathcal{N}_{g,k}(\{b_i\},\{p_i\}) = \sum_{\{n_i\}} \mathcal{N}_{g,k,0}(\{b_i\},\{n_i\},\{p_i\}).$$

We introduce the following generating function of these numbers<sup>4</sup>

<sup>&</sup>lt;sup>4</sup>The parameters  $q_i$ 's in our paper correspond to  $s_i$ 's in [2].

**Definition 2.7.** Let  $G(x, y; \{s_i\}; \{q_i\})$  denote the generating function of chord diagrams labeled by the boundary length spectrum

(2.19)  

$$G(x, y; \{s_i\}; \{q_i\}) = \sum_{b \ge 1} G_b(x, y; \{s_i\}; \{q_i\}),$$

$$G_b(x, y; \{s_i\}; \{q_i\}) = \frac{1}{b!} \sum_{\sum b_i = b} \sum_{\{p_i\}} \mathcal{N}_{g,k}(\{b_i\}, \{p_i\}) x^{2g-2} y^k \prod_{i \ge 0} s_i^{b_i} \prod_{i \ge 1} q_i^{p_i}$$

In the same way as the generating function  $Z^{\mathcal{P}}(x, y; \{s_i\}; \{t_i\})$  in (2.2), the generating function for the numbers  $\widehat{\mathcal{N}}_{k,b}(\{b_i\}, \{p_i\})$  of connected and disconnected chord takes form

(2.20) 
$$Z^{L}(x, y; \{s_i\}; \{q_i\}) = \exp\left[G(x, y; \{s_i\}; \{q_i\})\right] \\ = \sum_{\{b_i\}} \sum_{\{p_i\}} \widehat{\mathcal{N}}_{k,b}(\{b_i\}; \{p_i\}) x^{-b+k-n} y^k \prod_{i \ge 0} s_i^{b_i} \prod_{i \ge 1} q_i^{p_i}.$$

#### 2.2.1 A matrix model for the boundary length spectrum

Let c be a chord diagram. The boundary length spectrum filters chord diagrams according to combinatorial length of each boundary cycle, i.e. the sum of the number of chords and backbone underpasses. This length can be determined by counting marked points of a new type, which we now introduce. We introduce marked points of a new type between all chord ends and backbone ends, see the left diagram in Figure 10. For chord diagram decorated in this way, we get new marked points on the boundaries of the surface  $\Sigma_c$  by sliding each new marked point along the boundary of  $\Sigma_c$  until it reaches the first chord or backbone underside midpoint, as indicated in the right hand side of Figure 10.



Figure 10: Decorating a chord diagram with new marked points for partitions.

In order to construct a Gaussian matrix integral which counts this type of chord diagrams we introduce another external matrix  $\Lambda_{\rm L}$ , which is an invertible rank N matrix that keeps track of new marked points. We introduce a new model model based on the following rules **L1–L5**, in which Wick contractions in the Gaussian average correspond bijectively to decorated chord diagrams.

L1 A matrix element  $M_{\alpha\beta}$  corresponds to a chord end on a backbone. This graphical rule is the same as the rule C1.

- L2 A matrix element  $(\Lambda_{L}^{-1})_{\alpha_{j}\beta_{j}}$  is adjacent to a matrix element  $M_{\alpha_{j+1}\beta_{j+1}}$  on an upper edge of a backbone in  $\Sigma_{c}$ . Without loss of generality, we can put  $\Lambda_{P}^{-1}$ 's on the left hand side of the M's. Indices  $\alpha_{j}, \beta_{j}(=1, \ldots, N)$  are assigned to two upright line segments nipping a marked point in the upper edge of the backbone, see Figure 11.
- **L3** If two matrix elements  $U_{\alpha_j\beta_j}$  and  $V_{\alpha_{j+1}\beta_{j+1}}$   $(U, V = M \text{ or } \Lambda_{\mathrm{L}}^{-1})$  on the same backbone are adjacent, we form a matrix product  $(UV)_{\alpha_j\beta_{j+1}}$ . This graphical rule is the same as the rule **C2**.
- L4 If a matrix product

$$(\Lambda_{\rm L}^{-1}M)^i_{\alpha_1\beta_i}$$

corresponds to a backbone with a marked point, we assign the expression

$$N\sum_{\alpha_1,\beta_i=1}^N (\Lambda_{\rm L}^{-1})_{\alpha_1\beta_i}$$

to the bottom edge of this backbone. This gives the contribution

$$N \operatorname{Tr}((\Lambda_{\mathrm{L}}^{-1}M)^{i}\Lambda_{\mathrm{L}}^{-1})$$

with i chord ends and therefore i + 1 new marked points.

**L5** The Wick contraction between  $M_{\alpha_j\beta_j}$  and  $M_{\alpha'_{j'}\beta'_{j'}}$  corresponds to a band connecting two chord ends. This graphical rule is the same as the rule **C4**.



Figure 11: Bijective correspondence between decorated chord diagrams and matchings of Wick contractions.

Repeating the same discussions as in the previous subsection, one finds that every chord diagram with the backbone spectrum  $\{b_i\}$  corresponds to matchings with  $k = \sum i b_i/2$  Wick contractions, which arise from the following Gaussian average

(2.21) 
$$W_N^{\mathrm{L}}(\{b_i\};\{q_i\}) = \left\langle \prod_{i\geq 0} \left( N \mathrm{Tr}(\Lambda_{\mathrm{L}}^{-1}M)^i \Lambda_{\mathrm{L}}^{-1} \right)^{b_i} \right\rangle_N^{\mathrm{G}},$$

where we introduced Miwa times

(2.22) 
$$q_i = \frac{1}{N} \operatorname{Tr} \Lambda_{\mathrm{L}}^{-i}.$$

$$N \sum_{\alpha_1,...,\alpha_9=1}^{N} \Lambda_{L}^{-1} M \Lambda_{L}^{-1} M \Lambda_{L}^{-1} M \Lambda_{L}^{-1}$$

$$N \sum_{\alpha_1,...,\alpha_9=1}^{N} \Lambda_{L\alpha_1\alpha_2}^{-1} M_{\alpha_2\alpha_3} \Lambda_{L\alpha_3\alpha_4}^{-1} M_{\alpha_4\alpha_5} \Lambda_{L\alpha_5\alpha_6}^{-1} M_{\alpha_6\alpha_7} \Lambda_{L\alpha_7\alpha_8}^{-1} M_{\alpha_8\alpha_9} \Lambda_{L\alpha_9\alpha_1}^{-1} = N^2 q_2 q_1^2$$

$$M \sum_{\alpha_1,...,\alpha_9=1}^{N} \Lambda_{L}^{-1} M \Lambda_{L}^{-1} M \Lambda_{L}^{-1} M \Lambda_{L}^{-1}$$

$$N \sum_{\alpha_1,...,\alpha_9=1}^{N} \Lambda_{L\alpha_1\alpha_2}^{-1} M_{\alpha_2\alpha_3} \Lambda_{L\alpha_3\alpha_4}^{-1} M_{\alpha_4\alpha_5} \Lambda_{L\alpha_5\alpha_6}^{-1} M_{\alpha_6\alpha_7} \Lambda_{L\alpha_7\alpha_8}^{-1} M_{\alpha_8\alpha_9} \Lambda_{L\alpha_9\alpha_1}^{-1} = q_5$$

Figure 12: Chord diagrams of types  $\{g, k; \{b_i\}; \{p_i\}\} = \{0, 2; b_5 = 1; p_1 = 2, p_3 = 1\}$  and  $\{g, k; \{b_i\}; \{p_i\}\} = \{1, 2; b_5 = 1; p_5 = 1\}.$ 

It follows from the rules **L1–L5** that  $i \Lambda_{\rm L}^{-1}$ 's are aligned along the boundary cycle with length *i*. Therefore, for chord diagrams with the backbone spectrum  $\{b_i\}$  and the boundary length spectrum  $\{p_i\}$ , the corresponding Wick contractions in  $W_N^{\rm L}(\{b_i\}; \{q_i\})$  involve the factor

$$N^{b-k+n}\prod_{i\geq 1}q_i^{p_i},$$

see Figure 12. The key proposition of this subsection follows.

**Proposition 2.8.** The Gaussian average  $W_N^L(\{b_i\}; \{q_i\})$  in eq.(2.21) is the generating function of the numbers  $\widehat{N}_{k,b}(\{b_i\}, \{p_i\})$  of chord diagrams with the backbone spectrum  $\{b_i\}$ 

(2.23) 
$$W_N^{\mathrm{L}}(\{b_i\};\{q_i\}) = \sum_{\{p_i\}} \widehat{N}_{k,b}(\{b_i\},\{p_i\}) N^{b-k+n} \prod_{i \ge 1} q_i^{p_i}.$$

We also consider the full generating function for the numbers  $\widehat{N}_{k,b}(\{b_i\},\{p_i\})$  of chord diagrams

$$Z_N^{\mathcal{L}}(y; \{s_i\}; \{q_i\}) = \sum_{\{b_i\}} \sum_{\{p_i\}} \widehat{N}_{k,b}(\{b_i\}, \{p_i\}) N^{b-k+n} y^k \prod_{i \ge 0} s_i^{b_i} \prod_{i \ge 1} q_i^{p_i}.$$

This full generating function is given by the sum of Gaussian averages (2.21), and in consequence by the following Hermitian matrix integral

(2.24)  
$$Z_{N}^{L}(y; \{s_{i}\}; \{q_{i}\}) = \sum_{\{b_{i}\}} \frac{1}{\prod_{i} b_{i}!} y^{k} W_{N}^{L}(\{b_{i}\}, \{y^{-ib_{i}/2}q_{i}\}) \prod_{i} s_{i}^{b_{i}}$$
$$= \frac{1}{\operatorname{Vol}_{N}} \int_{\mathcal{H}_{N}} dM \exp\left[-N\operatorname{Tr}\left(\frac{M^{2}}{2} - \sum_{i\geq 0} s_{i}y^{i/2} \left(\Lambda_{\mathrm{L}}^{-1}M\right)^{i}\Lambda_{\mathrm{L}}^{-1}\right)\right].$$

Comparing this matrix integral and the generating function  $Z_{L}^{L}(y; \{s_i\}; \{q_i\})$  in equation (2.20), we arrive at the main theorem of this subsection.

**Theorem 2.9.** The matrix integral (2.24) agrees with the generating function (2.20)

(2.25) 
$$Z_N^{\rm L}(y; \{s_i\}; \{q_i\}) = Z^{\rm L}(N^{-1}, y; \{s_i\}; \{q_i\}).$$

#### Specialization of the model

The cut-and-join equation for the numbers of chord diagrams is discussed in Subsection 3.2. For technical reasons, the partial differential equation for the generating function (2.20) with general parameter  $\{s_i\}$  cannot be written in a simple form. Therefore we consider the specialization of the generating function (2.20) defined by<sup>5</sup>

$$s_i = s$$
.

Under this specialization, the matrix integral (2.24) reduces to

(2.26) 
$$Z_N^{L}(y;s;\{q_i\}) = Z_N^{L}(y;\{s_i=s\};\{q_i\}) \\ = \frac{1}{\operatorname{Vol}_N} \int_{\mathcal{H}_N} dM \exp\left[-N\operatorname{Tr}\left(\frac{M^2}{2} - \frac{s}{1 - y^{1/2}\Lambda_{\mathrm{L}}^{-1}M}\Lambda_{\mathrm{L}}^{-1}\right)\right].$$

For  $Z^{L}(x, y; s; \{q_i\}) = Z^{L}(x, y; \{s_i = s\}; \{q_i\})$ , we find

(2.27) 
$$Z_N^{\rm L}(y;s;\{q_i\}) = Z^{\rm L}(N^{-1},y;s;\{q_i\})$$

In Subsection 3.2 we derive the cut-and-join equation for this specialized model, and show the agreement with the cut-and-join equation found by combinatorial means in [2].

 $<sup>{}^{5}</sup>$ In [2], the length spectrum generating function  $G_b(x, y; \{s_i\})$  is the same as in this specialized model.

#### The boundary length and point spectrum and the uni-2.3fied model

So far we have discussed separately the enumeration of chord diagrams and partial chord diagrams labeled by the boundary point spectrum and the boundary length spectrum. In this subsection we consider a unification of these two kinds of spectra, which is referred to as the boundary length and point spectrum. This unified spectrum was introduced and analyzed by cut-and-join methods in [7]. In what follows we construct a matrix model that encodes this new spectrum, and in Subsection 3.3 we show how the cut-and-join equation found in [7] follows from this matrix model.



Figure 13: Decorating a partial chord diagram with the boundary label i =(1, 0, 1, 4, 2) with marked points for partitions.

The boundary length and point spectrum  $\{n_i\}$  is defined as follows [7].

**Definition 2.10.** Let c be a partial chord diagram. We associate the K tuple of numbers  $\mathbf{i} = (i_1, i_2, \dots, i_K)$  to a boundary component of  $\Sigma_c$ , if we find the tuple i of marked points around this boundary component, once we record different numbers of marked points in between chord ends and backbone underpasses along the boundary in the cyclic order induced from the orientation of  $\Sigma_c$ . The boundary length and point spectrum  $\{n_i\}$  counts the number of boundary cycles indexed by i for the partial chord diagram c.

To enumerate the number of partial chord diagrams labeled by  $\{g, k, l; \{b_i\}; \{n_i\}\}$ , we consider the generating functions introduced in [7].

**Definition 2.11.** Let  $\mathcal{N}_{q,k,l}(\{b_i\},\{n_i\})$  denote the number of connected chord diagrams labeled by the set of parameters  $(g, k, l; \{b_i\}; \{n_i\})$  in the boundary length and point spectrum. The generating function for these numbers is defined as

$$(2.28)$$

$$\mathcal{F}(x, y; \{s_i\}; \{u_i\}) = \sum_{b \ge 1} \mathcal{F}_b(x, y; \{s_i\}; \{u_i\}),$$

$$\mathcal{F}_b(x, y; \{s_i\}; \{u_i\}) = \frac{1}{b!} \sum_{\sum_i b_i = b} \sum_{\{n_i\}} \mathcal{N}_{g,k,l}(\{b_i\}, \{n_i\}) x^{2g-2} y^k \prod_{i \ge 0} s_i^{b_i} \prod_{K \ge 1} \prod_{\{i_L\}_{L=1}^K} u_i^{n_i}.$$

Exponentiating this generating function, one obtains the full generating function for the numbers  $\widehat{\mathcal{N}}_{k,b,l}(\{b_i\},\{n_i\})$  of partial chord diagrams

(2.29) 
$$\begin{aligned} \mathcal{Z}(x,y;\{s_i\};\{u_i\}) &= \exp\left[\mathcal{F}(x,y;\{s_i\};\{u_i\})\right] \\ &= \sum_{\{b_i\}} \sum_{\{n_i\}} \widehat{\mathcal{N}}_{k,b,l}(\{b_i\},\{n_i\}) x^{-b+k-n} y^k \prod_{i \ge 0} s_i^{b_i} \prod_{K \ge 1} \prod_{\{i_L\}_{L=1}^K} u_i^{n_i} d_i^{n_i} d_i^{n$$

where l, k, and b obey

$$l = \sum_{K \ge 1} \sum_{\{i_L\}_{L=1}^K} \sum_{L=1}^K i_L n_{(i_1, \dots, i_K)}, \qquad 2k + l = \sum_{i \ge 1} ib_i, \qquad b = \sum_{i \ge 0} b_i.$$

The enumeration of partial chord diagrams decorated by the boundary length and point spectrum can also be expressed in terms of Gaussian averages over Hermitian matrices. To this end we again make use of extra marked points, just as in the previous section (concerning the length spectrum to mark the separation between marked points on the backbone, counted by the index i), see Figure 13. Indeed, the boundary length and point spectrum also encodes the length spectrum, simply as the number K of partitions of marked points on boundary cycles.

To represent the boundary length and point spectrum, we introduce two external matrices  $\Lambda_{\rm P}$  and  $\Lambda_{\rm L}$ . In order to faithfully represent the ordering between marked points and partitions on each boundary cycle, we assume that these two external matrices do not commute

$$[\Lambda_{\rm P}, \Lambda_{\rm L}] \neq 0.$$

The correspondence between partial chord diagrams with the backbone spectrum  $\{b_i\}$  and matchings by Wick contractions in a Gaussian average is given by a combination of the previous rules C1, C2, P2, L2, P4, L4, and L5. We summarize this correspondence in Table 1.

Table 1: The correspondence between partial chord diagrams with the backbone spectrum  $\{b_i\}$  and matchings by Wick contractions in the Gaussian average.

A partial chord diagram	Gaussian average
A chord end on a backbone	$\Lambda_{\rm L}^{-1}M$
A marked point on a backbone	$\Lambda_{ m P}$
An underside of a backbone	$N\Lambda_{\rm L}^{-1}$
A backbone	$\left  N \operatorname{Tr} \left( \Lambda_{\mathrm{L}}^{-1} \Lambda_{\mathrm{P}}^{\alpha_{1}} \Lambda_{\mathrm{L}}^{-1} \Lambda_{\mathrm{P}}^{\alpha_{2}} \Lambda_{\mathrm{L}}^{-1} \cdots \Lambda_{\mathrm{P}}^{\alpha_{K}} \Lambda_{\mathrm{L}}^{-1} \right) \right $
A chord	Wick contraction $MM$

Based on these rules, one finds a bijective correspondence between partial chord diagrams with the backbone spectrum  $\{b_i\}$  and matchings by Wick contractions in the Gaussian average

(2.30) 
$$\mathcal{W}_N(\{b_i\};\{u_i\}) = \left\langle \prod_{i\geq 0} \left( N \operatorname{Tr}(\Lambda_{\mathrm{L}}^{-1}M + \Lambda_{\mathrm{P}})^i \Lambda_{\mathrm{L}}^{-1} \right)^{b_i} \right\rangle_N^{\mathrm{G}},$$

where in order to represent trace factors  $\Lambda_{\rm P}$  and  $\Lambda_{\rm L}$  we introduced the *generalized Miwa times* 

(2.31) 
$$u_{(i_1,\ldots,i_K)} = \frac{1}{N} \operatorname{Tr} \left( \Lambda_{\mathrm{P}}^{i_1} \Lambda_{\mathrm{L}}^{-1} \Lambda_{\mathrm{P}}^{i_2} \Lambda_{\mathrm{L}}^{-1} \cdots \Lambda_{\mathrm{P}}^{i_K} \Lambda_{\mathrm{L}}^{-1} \right).$$

If a partial chord diagram c contains a boundary cycle labeled by  $\mathbf{i} = (i_1, \ldots, i_K)$ , one finds the following trace factor in the corresponding Wick contractions in  $\mathcal{W}_N(\{b_i\}; \{u_i\})$ 

$$\operatorname{Tr}\left(\Lambda_{\mathrm{P}}^{i_{1}}\Lambda_{\mathrm{L}}^{-1}\Lambda_{\mathrm{P}}^{i_{2}}\Lambda_{\mathrm{L}}^{-1}\cdots\Lambda_{\mathrm{P}}^{i_{K}}\Lambda_{\mathrm{L}}^{-1}\right).$$

Finally, combining Propositions 2.5 and 2.8, we obtain the key proposition.

**Proposition 2.12.** The Gaussian average  $\mathcal{W}_N(\{b_i\}; \{u_i\})$  in the equation (2.30) is the generating function for the numbers  $\widehat{\mathcal{N}}_{k,b,l}(\{b_i\}, \{n_i\})$  of partial chord diagrams

(2.32) 
$$\mathcal{W}_{N}(\{b_{i}\};\{u_{i}\}) = \sum_{\{n_{i}\}} \widehat{\mathcal{N}}_{k,b,l}(\{b_{i}\},\{n_{i}\}) N^{b-k+n} \prod_{K \ge 1} \prod_{\{i_{L}\}_{L=1}^{K}} u_{i}^{n_{i}}.$$

Repeating the same combinatorics as in the previous subsections, we find the main theorem of this section.

Theorem 2.13. The Hermitian matrix integral

(2.33) 
$$\mathcal{Z}_{N}(y; \{s_{i}\}; \{u_{i}\}) = \frac{1}{\operatorname{Vol}_{N}} \int dM \exp\left[-N\operatorname{Tr}\left(\frac{M^{2}}{2} - \sum_{i\geq 0} s_{i}(y^{1/2}\Lambda_{\mathrm{L}}^{-1}M + \Lambda_{\mathrm{P}})^{i}\Lambda_{\mathrm{L}}^{-1}\right)\right]$$

agrees with the generating function (2.29)

(2.34) 
$$\mathcal{Z}_N(y; \{s_i\}; \{u_i\}) = \mathcal{Z}(N^{-1}, y; \{s_i\}; \{u_i\})$$

### 3 Cut-and-join equations via matrix models

In Section 2 we discussed matrix models that enumerate partial chord diagrams filtered by the boundary point spectrum, the boundary length spectrum, and the boundary length and point spectrum. In this section we derive partial differential equations for these matrix models, and show that they agree with the cut-and-join equations found in [2, 7]. To derive these differential equations, it is useful to introduce the following matrix integral.

256

**Definition 3.1.** Let A and B denote invertible matrices of rank N. We define a formal matrix integral with parameters y,  $\{g_i\}_{i=-\infty}^{+\infty}$ , and matrices A and B, as follows

(3.1) 
$$Z_N(y; \{g_i\}; A; B) = \frac{1}{\operatorname{Vol}_N} \int_{\mathcal{H}_N} dM \exp\left[-N\operatorname{Tr}\left(\frac{1}{2}M^2 - \sum_{i\in\mathbb{Z}} g_i(y^{1/2}B^{-1}M + A)^i B^{-1}\right)\right].$$

By the following specializations of this matrix integral one finds matrix integrals discussed in Section 2  $\,$ 

$$\begin{split} &Z_N^{\rm P}(y;\{s_i\};\{r_i\}): \ g_{i<0}=0, \ g_{i\geq 0}=s_i, \ A=\Lambda_{\rm P}, \ B=I_N, \\ &Z_N^{\rm L}(y;\{s_i\};\{q_i\}): \ g_{i<0}=0, \ g_{i\geq 0}=s_i, \ A=0, \ B=\Lambda_{\rm L}, \\ &Z_N^{\rm L}(y;s;\{q_i\}): \ g_{i\neq -1}=0, \ g_{-1}=-s, \ A=\Lambda_{\rm L}, \ B=-I_N, \\ &\mathcal{Z}_N(y;\{s_i\};\{u_{\pmb{i}}\}): \ g_{i<0}=0, \ g_{i\geq 0}=s_i, \ A=\Lambda_{\rm P}, \ B=\Lambda_{\rm L}, \end{split}$$

where  $I_N$  is the rank N identity matrix.

The matrix integral (3.1) satisfies the following partial differential equation.

**Proposition 3.2.** The matrix integral  $Z_N(y; \{g_i\}; A; B)$  obeys a partial differential equation

(3.2) 
$$\left[\frac{\partial}{\partial y} - \frac{1}{2N} \operatorname{Tr}(B^{-1})^{\mathrm{T}} \frac{\partial}{\partial A} (B^{-1})^{\mathrm{T}} \frac{\partial}{\partial A}\right] Z_{N}(y; \{g_{i}\}; A; B) = 0,$$

where the trace in the second term is defined, for rank N matrices X and Y, as

$$\operatorname{Tr} X \frac{\partial}{\partial Y} = \sum_{\alpha,\beta=1}^{N} X_{\alpha\beta} \frac{\partial}{\partial Y_{\beta\alpha}}.$$

*Proof.* By a shift  $M = X - y^{-1/2}BA$ , the matrix integral (3.1) can be rewritten as

$$Z_N(y; \{g_i\}; A; B) =$$
(3.3)
$$= \frac{1}{\text{Vol}_N} \int_{\widetilde{\mathcal{H}}_N} dX \exp\left[-N\text{Tr}\left(\frac{1}{2}(X - y^{-1/2}BA)^2 - \sum_{i \in \mathbb{Z}} y^{i/2}g_i(B^{-1}X)^i B^{-1}\right)\right].$$

Here  $\widetilde{\mathcal{H}}_N$  is the space of shifted matrices  $X = M + y^{-1/2}BA$  with  $M \in \mathcal{H}_N$ . The invariance of this matrix integral under the infinitesimal scaling  $X_{\alpha\beta} \to (1+\epsilon)X_{\alpha\beta}$  leads to a constraint equation

(3.4) 
$$\left\langle N^2 - N \operatorname{Tr} X^2 + y^{-1/2} N \operatorname{Tr} BAX + N \sum_{i \in \mathbb{Z}} i y^{i/2} g_i \operatorname{Tr} (B^{-1} X)^i B^{-1} \right\rangle = 0,$$

where the first term  $N^2$  comes from the measure factor, as  $dX \to (1 + N^2 \epsilon) dX$ . Here we have defined the unnormalized average for an observable  $\mathcal{O}(X)$ 

$$\langle \mathcal{O}(X) \rangle = \int_{\widetilde{\mathcal{H}}_N} dX \ \mathcal{O}(X) \exp\left[-N \operatorname{Tr}\left(\frac{1}{2}(X - y^{-1/2}BA)^2 - \sum_{i \in \mathbb{Z}} y^{i/2}g_i(B^{-1}X)^i B^{-1}\right)\right].$$

Using

$$\frac{1}{N}y^{1/2}\sum_{\gamma=1}^{N}(B^{-1})_{\beta\gamma}^{\mathrm{T}}\frac{\partial}{\partial A_{\gamma\alpha}}Z_{N}(y;\{g_{i}\};A;B) = \left\langle X_{\alpha\beta} - y^{-1/2}N\sum_{\gamma=1}^{N}B_{\alpha\gamma}A_{\gamma\beta}\right\rangle,$$
$$\frac{1}{N}\frac{\partial}{\partial g_{i}}Z_{N}(y;\{g_{i}\};A;B) = y^{i/2}\left\langle \mathrm{Tr}(B^{-1}X)^{i}B^{-1}\right\rangle,$$

one finds that the constraint equation (3.4) yields

$$\left[-\frac{1}{N}y\mathrm{Tr}(B^{-1})^{\mathrm{T}}\frac{\partial}{\partial A}(B^{-1})^{\mathrm{T}}\frac{\partial}{\partial A}-\mathrm{Tr}A^{\mathrm{T}}\frac{\partial}{\partial A}+\sum_{i\in\mathbb{Z}}ig_{i}\frac{\partial}{\partial g_{i}}\right]Z_{N}(y;\{g_{i}\};A;B)=0.$$

It follows from (3.3) that the last two derivatives in the expression above can be replaced by  $2y\partial/\partial y$ , so that the partial differential equation (3.2) is obtained.  $\Box$ 

**Remark 3.3.** In the above proof of the constraint equation (3.4) we considered the infinitesimal scaling  $X_{\alpha\beta} \rightarrow (1 + \epsilon)X_{\alpha\beta}$ . More generally, matrix integral (3.3) is invariant under infinitesimal shifts

$$X_{\alpha\beta} \longrightarrow X_{\alpha\beta} + \epsilon (X^{n+1})_{\alpha\beta}, \quad n = -1, 0, 1, \dots$$

It is known that for the matrix integral without external matrices A and B this symmetry yields the Virasoro symmetry, and in particular the scaling  $X_{\alpha\beta} \rightarrow (1+\epsilon)X_{\alpha\beta}$  is related to the Virasoro generator  $L_0^{\rm Vir}$  [22, 19].<sup>6</sup>

#### 3.1 The boundary point spectrum

In Subsection 2.1 we showed that the matrix integral  $Z_N^{\mathbf{P}}(y; \{s_i\}; \{r_i\})$  in (2.17)

$$Z_N^P(y; \{s_i\}; \{r_i\}) = = \frac{1}{\text{Vol}_N} \int_{\mathcal{H}_N} dM \exp\left[-N\text{Tr}\left(\frac{M^2}{2} - \sum_{i\geq 0} s_i y^{i/2} (M + y^{-1/2}\Lambda_P)^i\right)\right],$$

enumerates partial chord diagrams labeled by the boundary point spectrum. By the specialization

 $g_{i<0}=0, \quad g_{i\geq 0}=s_i, \quad A=\Lambda_{\rm P}, \quad B=I_N={\rm identity\ matrix},$ 

 $<sup>^{6}\</sup>mathrm{In}$  [33, 18], the Schwinger-Dyson approach to the enumeration of chord diagrams is also discussed.

#### Partial Chord Diagrams and Matrix Models

of the matrix integral  $Z_N(y; \{g_i\}; A; B)$  in (3.1) we see that

(3.5) 
$$Z_N(y; s_{i<0} = 0, \{s_i\}_{i\geq 0}; \Lambda_{\mathrm{P}}; I_N) = Z_N^{\mathrm{P}}(y; \{s_i\}; \{r_i\}),$$

where the reverse Miwa times  $r_i$  are defined in (2.15). From (3.2) we obtain the partial differential equation satisfied by  $Z_N^P(y; \{s_i\}; \{r_i\})$ .

**Corollary 3.4.** The matrix integral  $Z_N^{\mathbf{P}}(y; \{s_i\}; \{r_i\})$  obeys the partial differential equation

(3.6) 
$$\left[\frac{\partial}{\partial y} - \frac{1}{2N} \operatorname{Tr} \frac{\partial^2}{\partial \Lambda_{\mathrm{P}}^2}\right] Z_N^{\mathrm{P}}(y; \{s_i\}; \{r_i\}) = 0.$$

This corollary implies the following theorem.

**Theorem 3.5.** Let  $L_0$  and  $L_2$  be the differential operators<sup>7</sup>

(3.7)  
$$L_{0} = \frac{1}{2} \sum_{i \geq 2} \sum_{j=0}^{i-2} ir_{j}r_{i-j-2} \frac{\partial}{\partial r_{i}},$$
$$L_{2} = \frac{1}{2} \sum_{i \geq 2} \sum_{j=1}^{i-1} j(i-j)r_{i-2} \frac{\partial^{2}}{\partial r_{i} \partial r_{i-j}}.$$

The matrix integral  $Z_N^{\mathbf{P}}(y; \{s_i\}; \{r_i\})$  obeys the cut-and-join equation

(3.8) 
$$\frac{\partial}{\partial y} Z_N^{\mathbf{P}}(y; \{s_i\}; \{r_i\}) = \mathcal{L} Z_N^{\mathbf{P}}(y; \{s_i\}; \{r_i\}),$$

where

$$\mathcal{L} = L_0 + \frac{1}{N^2} L_2.$$

The formal solution of this cut-and-join equation, which gives the matrix integral  $Z_{\rm P}^{\rm N}(y; \{s_i\}; \{r_i\})$ , is iteratively determined from the initial condition at y = 0,

(3.9) 
$$Z_N^{\mathbf{P}}(y; \{s_i\}; \{r_i\}) = e^{y\mathcal{L}} Z_N^{\mathbf{P}}(0; \{s_i\}; \{r_i\}) = e^{y\mathcal{L}} e^{N^2 \sum_{i \ge 0} s_i r_i}$$

This theorem follows from the lemma below by rewriting the derivative  $Tr\partial^2/\partial\Lambda_P^2$ in the partial differential equation (3.6).

**Lemma 3.6.** For a function  $f(\{r_i\})$  of the reverse Miwa times  $r_i$ , the derivative  $\mathrm{Tr}\partial^2/\partial\Lambda_{\mathrm{P}}^2$  can be rewritten as

(3.10) 
$$\frac{1}{2N} \operatorname{Tr} \frac{\partial^2}{\partial \Lambda_{\mathrm{P}}^2} f(\{r_i\}) = \left(L_0 + \frac{1}{N^2} L_2\right) f(\{r_i\}).$$

<sup>&</sup>lt;sup>7</sup>In [36] the differential operators  $L_0$  and  $L_2$  were denoted by  $W^{(3)}$ .

J. E. Andersen, H. Fuji, M. Manabe, R. C. Penner, and P. Sułkowski

*Proof.* Consider the derivative  $\partial/\partial \Lambda_{P\beta\alpha}$  of  $r_i$ ,

$$\frac{\partial r_i}{\partial \Lambda_{\mathrm{P}\beta\alpha}} = \frac{i}{N} \Lambda_{\mathrm{P}\alpha\beta}^{i-1}, \qquad \mathrm{Tr} \frac{\partial^2 r_i}{\partial \Lambda_{\mathrm{P}}^2} = iN \sum_{j=0}^{i-2} r_j r_{i-j-2}$$

Then the derivative  $\text{Tr}\partial^2/\partial\Lambda_{\rm P}^2$  of the function  $f(\{r_i\})$  is re-expressed as

$$\frac{1}{2N} \operatorname{Tr} \frac{\partial^2}{\partial \Lambda_{\mathrm{P}}^2} f(\{r_i\}) = \frac{1}{2N} \sum_{i \ge 0} \operatorname{Tr} \frac{\partial^2 r_i}{\partial \Lambda_{\mathrm{P}}^2} \frac{\partial f(\{r_i\})}{\partial r_i} + \frac{1}{2N} \sum_{i,j \ge 0} \sum_{\alpha,\beta=1}^N \frac{\partial r_i}{\partial \Lambda_{\mathrm{P}\beta\alpha}} \frac{\partial r_j}{\partial \Lambda_{\mathrm{P}\alpha\beta}} \frac{\partial^2 f(\{r_i\})}{\partial r_i \partial r_j} \\ = \frac{1}{2} \sum_{i \ge 2} \sum_{j=0}^{i-2} i r_j r_{i-j-2} \frac{\partial f(\{r_i\})}{\partial r_i} + \frac{1}{2N^2} \sum_{i,j \ge 1} i j r_{i+j-2} \frac{\partial^2 f(\{r_i\})}{\partial r_i \partial r_j}.$$

This coincides with the right hand side of (3.10).

The cut-and-join equation for the rescaled matrix integral (2.18) yields

(3.11) 
$$\frac{\partial}{\partial y} Z_{t_0 N}^{\mathrm{P}}(t_0 y; \{t_0^{-1} s_i\}; \{t_0^{-1} t_i\}) = \mathcal{L} Z_{t_0 N}^{\mathrm{P}}(t_0 y; \{t_0^{-1} s_i\}; \{t_0^{-1} t_i\}),$$

where  $\mathcal{L}$  is given by

(3.12) 
$$\mathcal{L} = L_0 + x^2 L_2, \quad x = N^{-1},$$
$$L_0 = \frac{1}{2} \sum_{i \ge 2} \sum_{j=0}^{i-2} i t_j t_{i-j-2} \frac{\partial}{\partial t_i}, \quad L_2 = \frac{1}{2} \sum_{i \ge 2} \sum_{j=1}^{i-1} j(i-j) t_{i-2} \frac{\partial^2}{\partial t_i \partial t_{i-j}}.$$

This cut-and-join equation agrees with the partial differential equation in Theorem 1 of [2], where it was proven combinatorially by the recursion relation for the number of partial chord diagrams.<sup>8</sup> This completes the proof of Theorem 2.6.

#### 3.2 The boundary length spectrum

In Subsection 2.2 we showed that the matrix integral  $Z_N^L(y; \{s_i\}; \{q_i\})$  in (2.24) enumerates chord diagrams labeled by the boundary length spectrum. By the specialization

$$g_{i<0} = 0, \quad g_{i\geq 0} = s_i, \quad A = 0, \quad B = \Lambda_{\rm L},$$

of the matrix integral  $Z_N(y; \{g_i\}; A; B)$  in (3.1) we see that

(3.13) 
$$Z_N(y; s_{i<0} = 0, \{s_i\}_{i\geq 0}; 0; \Lambda_L) = Z_N^L(y; \{s_i\}; \{q_i\}),$$

where the Miwa times  $q_i$  are defined in equation (2.22).

260

<sup>&</sup>lt;sup>8</sup>For the Grothendieck's dessin counting, a similar cut-and-join equation was found in [27].

Obviously, for A = 0 the partial differential equation (3.2) does not hold. Instead we consider the matrix integral (2.26) obtained by the specialization  $s_i = s$ 

$$Z_N^{\mathcal{L}}(y;s;\{q_i\}) = \frac{1}{\operatorname{Vol}_N} \int_{\mathcal{H}_N} dM \, \exp\left[-N\operatorname{Tr}\left(\frac{M^2}{2} + \frac{s}{y^{1/2}M - \Lambda_{\mathcal{L}}}\right)\right].$$

The same matrix integral can be obtained by the specialization

$$g_{i\neq-1}=0, \quad g_{-1}=-s, \quad A=\Lambda_{\rm L}, \quad B=-I_N,$$

and thus

(3.14) 
$$Z_N(y; s_i = -\delta_{i,-1}; \Lambda_L; B = -I_N) = Z_N^L(y; s; \{q_i\}).$$

Then from (3.2) we obtain a partial differential equation for  $Z_N^{L}(y; s; \{q_i\})$ .

**Corollary 3.7.** The matrix integral  $Z_N^{L}(y; s; \{q_i\})$  obeys the partial differential equation

(3.15) 
$$\left[\frac{\partial}{\partial y} - \frac{1}{2N} \operatorname{Tr} \frac{\partial^2}{\partial \Lambda_{\rm L}^2}\right] Z_N^{\rm L}(y; s; \{r_i\}) = 0.$$

This corollary implies the following theorem.

**Theorem 3.8.** Let  $K_0$  and  $K_2$  be the differential operators

(3.16)  

$$K_{0} = \frac{1}{2} \sum_{i \geq 3} \sum_{j=1}^{i-1} (i-2)q_{j}q_{i-j} \frac{\partial}{\partial q_{i-2}},$$

$$K_{2} = \frac{1}{2} \sum_{i \geq 2} \sum_{j=1}^{i-1} j(i-j)q_{i+2} \frac{\partial^{2}}{\partial q_{i}\partial q_{i-j}}$$

The matrix integral  $Z_N^{L}(y; s; \{q_i\})$  obeys the cut-and-join equation

(3.17) 
$$\frac{\partial}{\partial y} Z_N^{\mathrm{L}}(y;s;\{q_i\}) = \mathcal{K} Z_N^{\mathrm{L}}(y;s;\{q_i\}),$$

where

$$\mathcal{K} = K_0 + \frac{1}{N^2} K_2.$$

The formal solution of this cut-and-join equation, which gives the matrix integral  $Z_{\rm L}^{\rm N}(y;s;\{q_i\})$ , is iteratively determined from the initial condition at y=0,

(3.18) 
$$Z_N^{\rm L}(y;s;\{q_i\}) = e^{y\mathcal{K}} Z_N^{\rm L}(y=0;s;\{q_i\}) = e^{y\mathcal{K}} e^{N^2 s q_1}$$

The cut-and-join equation (3.17) was combinatorially proven in Theorem 2 of [2] for the generating function  $Z^{L}(x, y; s; \{q_i\})$  in (2.19), and thus Theorem 2.9 for  $s_i = s$  is reproved.

The claim of Theorem 3.8 is proven by rewriting the derivative  $\text{Tr}\partial^2/\partial\Lambda_{\rm L}^2$  in the partial differential equation (3.15) using the following lemma.

**Lemma 3.9.** For a function  $g(\{q_i\})$  of the Miwa times  $\{q_i\}$ , the derivative  $\text{Tr}\partial^2/\partial\Lambda_L^2$  can be rewritten as follows

(3.19) 
$$\frac{1}{2N} \operatorname{Tr} \frac{\partial^2}{\partial \Lambda_{\mathrm{L}}^2} g(\{q_i\}) = \left(K_0 + \frac{1}{N^2} K_2\right) g(\{q_i\}).$$

*Proof.* By acting  $\partial/\partial \Lambda_{\rm L}$  on the Miwa time  $q_i$  one obtains

$$\frac{\partial q_i}{\partial \Lambda_{\mathrm{L}\alpha\beta}} = -\frac{i}{N} \Lambda_{\mathrm{L}\beta\alpha}^{-i-1}, \qquad \mathrm{Tr} \frac{\partial^2 q_i}{\partial \Lambda_{\mathrm{L}}^2} = iN \sum_{j=1}^{i+1} q_j q_{i-j+2}.$$

Adopting this relation via the chain rule applied to the  $\Lambda_L$  derivatives, one finds that

$$\frac{1}{2N} \operatorname{Tr} \frac{\partial^2 g(\{q_i\})}{\partial \Lambda_{\mathrm{L}}^2} = \frac{1}{2N} \sum_{i \ge 0} \operatorname{Tr} \frac{\partial^2 q_i}{\partial \Lambda_{\mathrm{L}}^2} \frac{\partial g(\{q_i\})}{\partial q_i} + \frac{1}{2N} \sum_{i,j \ge 0} \sum_{\alpha,\beta=1}^N \frac{\partial q_i}{\partial \Lambda_{\mathrm{L}\alpha\beta}} \frac{\partial q_j}{\partial \Lambda_{\mathrm{L}\beta\alpha}} \frac{\partial^2 g(\{q_i\})}{\partial q_i \partial q_j} = \frac{1}{2} \sum_{i \ge 1} i q_j q_{i-j+2} \frac{\partial g(\{q_i\})}{\partial q_i} + \frac{1}{2N^2} \sum_{i,j \ge 1} i j q_{i+j+2} \frac{\partial^2 g(\{q_i\})}{\partial q_i \partial q_j}.$$

This coincides with the right hand side of (3.19).

#### 3.3 The boundary length and point spectrum

In Subsection 2.3 we showed that the matrix integral  $\mathcal{Z}_N(y; \{s_i\}; \{u_i\})$  in (2.29)

$$\begin{aligned} \mathcal{Z}_N(y; \{s_i\}; \{u_i\}) &= \\ &= \frac{1}{\operatorname{Vol}_N} \int_{\mathcal{H}_N} dM \, \exp\left[-N\operatorname{Tr}\left(\frac{M^2}{2} - \sum_{i\geq 0} s_i (y^{1/2}\Lambda_{\mathrm{L}}^{-1}M + \Lambda_{\mathrm{P}})^i \Lambda_{\mathrm{L}}^{-1}\right)\right] \end{aligned}$$

enumerates partial chord diagrams labeled by the boundary length and point spectrum. By the specialization

$$g_{i<0} = 0, \quad g_{i\geq 0} = s_i, \quad A = \Lambda_{\rm P}, \quad B = \Lambda_{\rm L},$$

of the matrix integral  $Z_N(y; \{g_i\}; A; B)$  in (3.1) we see that

(3.20) 
$$Z_N(y; s_{i<0} = 0, \{s_i\}_{i>0}; \Lambda_{\mathrm{P}}; \Lambda_{\mathrm{L}}) = \mathcal{Z}_N(y; \{s_i\}; \{u_i\}),$$

where the generalized Miwa times  $u_{(i_1,\ldots,i_K)}$  are defined in (2.31)

$$u_{(i_1,\dots,i_K)} = \frac{1}{N} \operatorname{Tr} \left( \Lambda_{\mathrm{P}}^{i_1} \Lambda_{\mathrm{L}}^{-1} \Lambda_{\mathrm{P}}^{i_2} \Lambda_{\mathrm{L}}^{-1} \cdots \Lambda_{\mathrm{P}}^{i_K} \Lambda_{\mathrm{L}}^{-1} \right)$$

From (3.2) we obtain a partial differential equation for  $\mathcal{Z}_N(y; \{s_i\}; \{u_i\})$ .

**Corollary 3.10.** The matrix integral  $\mathcal{Z}_N(y; \{s_i\}; \{u_i\})$  obeys the partial differential equation

(3.21) 
$$\left[\frac{\partial}{\partial y} - \frac{1}{2N} \operatorname{Tr}(\Lambda_{\mathrm{L}}^{-1})^{\mathrm{T}} \frac{\partial}{\partial \Lambda_{\mathrm{P}}} (\Lambda_{\mathrm{L}}^{-1})^{\mathrm{T}} \frac{\partial}{\partial \Lambda_{\mathrm{P}}}\right] \mathcal{Z}_{N}(y; \{s_{i}\}; \{u_{i}\}) = 0.$$

This corollary implies the following main theorem of this section.

**Theorem 3.11.** Let  $M_0$  and  $M_2$  be the following differential operators with respect to parameters  $u_i$ 

$$M_{0} = \frac{1}{2} \sum_{K \ge 1} \sum_{\{i_{1},...,i_{K}\}} \sum_{1 \le I \neq M \le K} \sum_{\ell=0}^{i_{I}-1} \sum_{m=0}^{i_{M}-1} u_{(i_{I}-\ell-1,i_{I+1},...,i_{K}-1,m)} u_{(i_{M}-m-1,i_{M+1},...,i_{I-1},\ell)} \frac{\partial}{\partial u_{(i_{1},...,i_{K})}} + \sum_{K \ge 1} \sum_{\{i_{1},...,i_{K}\}} \sum_{I=0}^{K} \sum_{\ell+m \le i_{I}-2} u_{(\ell,m,i_{I+1},...,i_{I-1})} u_{(i_{I}-\ell-m-2)} \frac{\partial}{\partial u_{(i_{1},...,i_{K})}}$$

$$M_{2} = \frac{1}{2} \sum_{K,L \ge 0} \sum_{\{i_{1},...,i_{K}\}} \sum_{\{j_{1},...,j_{L}\}} \sum_{I=0}^{K} \sum_{J=0}^{L} \sum_{\ell=0}^{L} \sum_{m=0}^{i_{I}-1} \sum_{m=0}^{j_{I}-1} u_{(i_{I}-\ell-1,i_{I+1},...,i_{I-1},\ell,j_{I}-m-1,j_{I+1},...,j_{I-1},m)} \frac{\partial^{2}}{\partial u_{(i_{1},...,i_{L})} \partial u_{(j_{1},...,j_{L})}},$$

where labels I, M's are defined modulo K, and the label J is defined modulo L. The matrix integral  $\mathcal{Z}_N(y; \{s_i\}; \{u_i\})$  obeys the cut-and-join equation

(3.23) 
$$\frac{\partial}{\partial y} \mathcal{Z}_N(y; \{s_i\}; \{u_i\}) = \mathcal{M} \mathcal{Z}_N(y; \{s_i\}; \{u_i\}),$$

where

$$\mathcal{M} = M_0 + \frac{1}{N^2} M_2$$

The formal solution of this cut-and-join equation, which gives the matrix integral  $\mathcal{Z}_N(y; \{s_i\}; \{u_i\})$ , is iteratively determined from the initial condition at y = 0,

(3.24) 
$$\mathcal{Z}_{N}(y; \{s_{i}\}; \{u_{i}\}) = e^{y\mathcal{M}}\mathcal{Z}_{N}(y=0; \{s_{i}\}; \{u_{i}\}) = e^{y\mathcal{M}}e^{N^{2}\sum_{i\geq 0}s_{i}u_{(i)}}$$

The partial differential equation (3.23) agrees with the cut-and-equation obtained combinatorially in Theorem 1.1 of [7]. Here we prove this theorem by rewriting the derivative in the second term of the partial differential equation (3.21), taking advantage of the following lemma.

**Lemma 3.12.** For a function  $h(\{u_i\})$  of the generalized Miwa times  $u_i$ , the derivative in the second term of the partial differential equation (3.21) can be rewritten as follows

(3.25) 
$$\frac{1}{2N} \operatorname{Tr}\left[ (\Lambda_{\mathrm{L}}^{-1})^{\mathrm{T}} \frac{\partial}{\partial \Lambda_{\mathrm{P}}} (\Lambda_{\mathrm{L}}^{-1})^{\mathrm{T}} \frac{\partial}{\partial \Lambda_{\mathrm{P}}} \right] h(\{u_{i}\}) = \left( M_{0} + \frac{1}{N^{2}} M_{2} \right) h(\{u_{i}\}).$$

*Proof.* By the chain rule, the derivative on the left hand side of (3.25) is rewritten as follows

$$\begin{split} \operatorname{Tr} & \left[ (\Lambda_{\mathrm{L}}^{-1})^{\mathrm{T}} \frac{\partial}{\partial \Lambda_{\mathrm{P}}} (\Lambda_{\mathrm{L}}^{-1})^{\mathrm{T}} \frac{\partial}{\partial \Lambda_{\mathrm{P}}} \right] h(\{u_{\boldsymbol{i}}\}) = \\ & = \sum_{K \geq 0} \sum_{(i_{1}, \dots, i_{K})} \operatorname{Tr} \left[ (\Lambda_{\mathrm{L}}^{-1})^{\mathrm{T}} \frac{\partial}{\partial \Lambda_{\mathrm{P}}} (\Lambda_{\mathrm{L}}^{-1})^{\mathrm{T}} \frac{\partial}{\partial \Lambda_{\mathrm{P}}} u_{(i_{1}, \dots, i_{K})} \right] \frac{\partial}{\partial u_{(i_{1}, \dots, i_{K})}} h(\{u_{\boldsymbol{i}}\}) \\ & + \sum_{K, L \geq 0} \sum_{(i_{1}, \dots, i_{K})} \sum_{(j_{1}, \dots, j_{L})} \operatorname{Tr} \left[ (\Lambda_{\mathrm{L}}^{-1})^{\mathrm{T}} \frac{\partial}{\partial \Lambda_{\mathrm{P}}} u_{(i_{1}, \dots, i_{K})} (\Lambda_{\mathrm{L}}^{-1})^{\mathrm{T}} \frac{\partial}{\partial \Lambda_{\mathrm{P}}} u_{(j_{1}, \dots, j_{L})} \right] \\ & \times \frac{\partial^{2}}{\partial u_{(i_{1}, \dots, i_{K})} \partial u_{(j_{1}, \dots, j_{L})}} h(\{u_{\boldsymbol{i}}\}). \end{split}$$

Each of the coefficients yields

a

$$\mathrm{Tr}\left[(\Lambda_{\mathrm{L}}^{-1})^{\mathrm{T}}\frac{\partial}{\partial\Lambda_{\mathrm{P}}}u_{(i_{1},\ldots,i_{K})}(\Lambda_{\mathrm{L}}^{-1})^{\mathrm{T}}\frac{\partial}{\partial\Lambda_{\mathrm{P}}}u_{(j_{1},\ldots,j_{L})}\right] =$$

Partial Chord Diagrams and Matrix Models

$$=\sum_{I=1}^{K}\sum_{J=1}^{L}\sum_{\ell=0}^{i_{I}-1}\sum_{m=0}^{j_{J}-1}\frac{1}{N^{2}}\mathrm{Tr}(\Lambda_{\mathrm{P}}^{i_{I}-\ell-1}\Lambda_{\mathrm{L}}^{-1}\Lambda_{\mathrm{P}}^{i_{I}+1}\Lambda_{\mathrm{L}}^{-1}\cdots\Lambda_{\mathrm{L}}^{i_{I}-1}\Lambda_{\mathrm{P}}^{-1}\Lambda_{\mathrm{L}}^{-1}\Lambda_{\mathrm{P}}^{\ell}\Lambda_{\mathrm{L}}^{-1}$$
$$\cdot\Lambda_{\mathrm{P}}^{j_{J}-m-1}\Lambda_{\mathrm{L}}^{-1}\Lambda_{\mathrm{P}}^{j_{J}+1}\Lambda_{\mathrm{L}}^{-1}\cdots\Lambda_{\mathrm{P}}^{j_{J}-1}\Lambda_{\mathrm{L}}^{-1}\Lambda_{\mathrm{P}}^{m}\Lambda_{\mathrm{L}}^{-1})$$
$$=\frac{1}{N}\sum_{I=1}^{K}\sum_{J=1}^{L}\sum_{\ell=0}^{i_{I}-1}\sum_{m=0}^{j_{J}-1}u_{(i_{I}-\ell-1,i_{I+1},\dots,i_{I-1},\ell,j_{J}-m-1,j_{J+1},\dots,j_{J-1},m).$$

In this way one obtains the right hand side of (3.25).

As a corollary of Theorem 3.11, one finds the cut-and-join equation for the 1-backbone generating function.  $^9$ 

**Corollary 3.13.** The 1-backbone generating function  $\mathcal{F}_1(x, y; \{s_i\}; \{u_i\})$  obtained by picking up the  $\mathcal{O}(s_i^1)$  terms in  $\mathcal{Z}_N(y; \{s_i\}; \{u_i\})$  as follows

(3.26) 
$$\mathcal{F}_{1}(N^{-1}, y; \{s_{i}\}; \{u_{i}\}) = \frac{1}{\operatorname{Vol}_{N}} \int_{\mathcal{H}_{N}} dM \ \mathrm{e}^{-N\operatorname{Tr}\frac{M^{2}}{2}} N \sum_{i \geq 0} s_{i} \operatorname{Tr}(y^{1/2} \Lambda_{\mathrm{L}}^{-1} M + \Lambda_{\mathrm{P}})^{i} \Lambda_{\mathrm{L}}^{-1},$$

obeys the cut-and-join equation

(3.27) 
$$\frac{\partial}{\partial y} \mathcal{F}_1(x, y; \{s_i\}; \{u_i\}) = \mathcal{M} \mathcal{F}_1(x, y; \{s_i\}; \{u_i\}),$$

where  $\mathcal{M} = M_0 + x^2 M_2$ . The solution is iteratively determined by

(3.28) 
$$\mathcal{F}_1(x, y; \{s_i\}; \{u_i\}) = e^{y\mathcal{M}} \mathcal{F}_1(x, y = 0; \{s_i\}; \{u_i\}) = e^{y\mathcal{M}} \left( x^{-2} \sum_{i \ge 0} s_i u_{(i)} \right).$$

### 4 Non-oriented analogues

In this section we consider the enumeration of both orientable and non-orientable (jointly called non-oriented) partial chord diagrams [2, 7]. To this end we generalize the matrix models introduced in Section 2. In Subsection 4.1, matrix models for the boundary point spectrum, the boundary length spectrum, and the boundary length and point spectrum are introduced, based on the corresponding Gaussian matrix integrals over the space of rank N real symmetric matrices. Subsequently, in Subsection 4.2, we derive cut-and-join equations for the generating functions of non-oriented partial chord diagrams, using analogous methods as those discussed in Section 3.

<sup>&</sup>lt;sup>9</sup>For  $\Lambda_{\rm L} = I_N$  (or  $s_i = s$  and  $\Lambda_{\rm P} = 0$ ) the cut-and-join equation for the 1-backbone generating function labeled by the boundary point spectrum (or boundary length spectrum) was proven combinatorially in [2].

#### 4.1 Non-oriented analogues of the matrix models

In this subsection we generalize matrix models found in Section 2, in order to enumerate both orientable and non-orientable partial chord diagrams [2, 7].

**Definition 4.1.** Let  $\widetilde{\mathcal{N}}_{h,k,l}(\{b_i\},\{n_i\},\{p_i\})$  denote the number of connected nonoriented partial chord diagrams of type  $\{h,k,l;\{b_i\};\{n_i\};\{p_i\}\}$ . Analogously as in the orientable case, we define

$$\begin{split} \widetilde{\mathcal{N}}_{h,k,l}(\{b_i\},\{n_i\}) &= \sum_{\{p_i\}} \widetilde{\mathcal{N}}_{h,k,l}(\{b_i\},\{n_i\},\{p_i\}),\\ \widetilde{\mathcal{N}}_{h,k}(\{b_i\},\{p_i\}) &= \sum_{\{n_i\}} \widetilde{\mathcal{N}}_{h,k,l=0}(\{b_i\},\{n_i\},\{p_i\}), \end{split}$$

and introduce generating functions

(4.1)  

$$\widetilde{F}(x, y; \{s_i\}; \{t_i\}) = \sum_{b \ge 1} \widetilde{F}_b(x, y; \{s_i\}; \{t_i\}),$$

$$\widetilde{F}_b(x, y; \{s_i\}; \{t_i\}) = \frac{1}{b!} \sum_{\sum_i b_i = b} \sum_{\{n_i\}} \widetilde{\mathcal{N}}_{h,k,l}(\{b_i\}, \{n_i\}) x^{h-2} y^k \prod_{i \ge 0} s_i^{b_i} t_i^{n_i},$$

and

(4.2)  

$$\widetilde{G}(x, y; \{s_i\}; \{q_i\}) = \sum_{b \ge 1} \widetilde{G}_b(x, y; \{s_i\}; \{q_i\}),$$

$$\widetilde{G}_b(x, y; \{s_i\}; \{q_i\}) = \frac{1}{b!} \sum_{\sum_i b_i = b} \sum_{\{p_i\}} \widetilde{\mathcal{N}}_{h,k}(\{b_i\}, \{p_i\}) x^{h-2} y^k \prod_{i \ge 0} s_i^{b_i} \prod_{i \ge 1} q_i^{p_i}.$$

Generating functions of connected and disconnected partial chord diagrams are related by

(4.3) 
$$\widetilde{Z}^{P}(x, y; \{s_i\}; \{t_i\}) = \exp\left[\widetilde{F}(x, y; \{s_i\}; \{t_i\})\right],$$

(4.4) 
$$\widetilde{Z}^{L}(x, y; \{s_i\}; \{q_i\}) = \exp\left[\widetilde{G}(x, y; \{s_i\}; \{q_i\})\right]$$

Furthermore, we introduce generating functions of non-oriented partial chord diagrams labeled by the boundary length and point spectrum.

**Definition 4.2.** Let  $\widetilde{\mathcal{N}}_{h,k,l}(\{b_i\},\{n_i\})$  denote the number of connected orientable and non-orientable partial chord diagrams of type  $\{h,k,l;\{b_i\};\{n_i\}\}$  with the boundary length and point spectrum  $n_i$ . We define the generating functions

$$\begin{aligned} &(4.5)\\ \widetilde{\mathcal{F}}(x,y;\{s_i\};\{u_{\boldsymbol{i}}\}) = \sum_{b\geq 1} \widetilde{\mathcal{F}}_b(x,y;\{s_i\};\{u_{\boldsymbol{i}}\}),\\ &\widetilde{\mathcal{F}}_b(x,y;\{s_i\};\{u_{\boldsymbol{i}}\}) = \frac{1}{b!} \sum_{\sum_i b_i = b} \sum_{\{n_{\boldsymbol{i}}\}} \widetilde{\mathcal{N}}_{h,k,l}(\{b_i\},\{n_{\boldsymbol{i}}\}) x^{h-2} y^k \prod_{i\geq 0} s_i^{b_i} \prod_{K\geq 1} \prod_{\{i_L\}_{L=1}^K} u_{\boldsymbol{i}}^{n_{\boldsymbol{i}}}. \end{aligned}$$

As usual, generating functions of connected and disconnected partial chord diagrams are related by

(4.6) 
$$\widetilde{\mathcal{Z}}(x,y;\{s_i\};\{u_i\}) = \exp\left[\widetilde{\mathcal{F}}(x,y;\{s_i\};\{u_i\})\right].$$

#### Non-oriented analogue of partial chord diagrams and Wick contractions

A non-oriented partial chord diagram is a partial chord diagrams with each chord decorated by a binary variable, which indicates if it is twisted or not. Such non-oriented partial chord diagrams are enumerated by real symmetric<sup>10</sup> matrix integrals. The Gaussian average  $\langle \mathcal{O}(M) \rangle_N^{\widetilde{G}}$  over the space  $\mathcal{H}_N(\mathbb{R})$  of real symmetric matrices is defined by

(4.7) 
$$\langle \mathcal{O}(M) \rangle_N^{\widetilde{G}} = \frac{1}{\operatorname{Vol}_N(\mathbb{R})} \int_{\mathcal{H}_N(\mathbb{R})} dM \ \mathcal{O}(M) \, \mathrm{e}^{-N \operatorname{Tr} \frac{M^2}{4}},$$

where

(4.8) 
$$\operatorname{Vol}_{N}(\mathbb{R}) = \int_{\mathcal{H}_{N}(\mathbb{R})} dM \ \mathrm{e}^{-N\operatorname{Tr}\frac{M^{2}}{4}} = N^{N(N+1)/2}\operatorname{Vol}(\mathcal{H}_{N}(\mathbb{R})),$$

For the choice of  $\mathcal{O}(M) = M_{\alpha\beta}M_{\gamma\epsilon}$   $(\alpha, \beta, \gamma, \epsilon = 1, ..., N)$ , the Wick contraction is defined as

(4.9) 
$$\overline{M_{\alpha\beta}}M_{\gamma\epsilon} := \langle M_{\alpha\beta}M_{\gamma\epsilon}\rangle_N^{\widetilde{G}} = \frac{1}{N}(\delta_{\alpha\epsilon}\delta_{\beta\gamma} + \delta_{\alpha\gamma}\delta_{\beta\epsilon}).$$

This Wick contraction consists of two terms, which encode the corresponding fatgraph as follows. The first term  $\frac{1}{N}\delta_{\alpha\epsilon}\delta_{\beta\gamma}$  is the same as in the Hermitian matrix integral (2.5), and it can be identified with an untwisted band in the two dimensional surface  $\Sigma_c$  associated to the partial chord diagram c. The second term  $\frac{1}{N}\delta_{\alpha\gamma}\delta_{\beta\epsilon}$  in (4.9) relates opposite matrix indices compared to the first term and can be identified with the twisted band in  $\Sigma_c$ , see Figure 14. Hence, for the real symmetric Gaussian average, the correspondence rules **C4**, **P5**, **L5** in Section 2 are replaced by the following rules [24, 54, 52, 26, 39, 23].

**N5** The Wick contraction between  $M_{\alpha_j\beta_j}$  and  $M_{\alpha'_{j'}\beta'_{j'}}$  corresponds to a band or a twisted band connecting two chord ends. Each Wick contraction imposes either the constraint  $\delta_{\alpha_j\beta'_{j'}}\delta_{\alpha'_{j'}\beta_j}$  or the constraint  $\delta_{\alpha_j\alpha'_{j'}}\delta_{\beta_j\beta'_{j'}}$  for matrix indices assigned to edges of chord ends matched by Wick contractions.

In order to construct matrix models that enumerate non-oriented partial chord diagrams, we introduce two external real symmetric matrices matrices  $\Omega_P$  and  $\Omega_L$ 

$$\Omega_{\rm P} = \Omega_{\rm P}^{\rm T}, \quad \Omega_{\rm L} = \Omega_{\rm L}^{\rm T},$$

<sup>&</sup>lt;sup>10</sup>The Gaussian matrix integral over the space of real symmetric matrix is also referred to as the *Gaussian orthogonal ensemble* [21, 35].

$$\overset{\alpha}{\underset{\beta}{\underbrace{\qquad}}} \overset{\alpha}{\underset{\gamma}{\underbrace{\qquad}}} \overset{\epsilon}{\underset{\gamma}{\underbrace{\qquad}}} + \overset{\alpha}{\underset{\beta}{\underbrace{\qquad}}} \overset{\epsilon}{\underset{\gamma}{\underbrace{\qquad}}} \overset{\epsilon}{\underset{\gamma}{\underbrace{\quad}}} \overset{\epsilon}{\underset{\gamma}{\underbrace{\atop}}} \overset{\epsilon}{\underset{\gamma}{\underbrace{\atop}}} \overset{\epsilon}{\underset{\gamma}{\underbrace{\atop}}} \overset{\epsilon}{\underset{\tau}{\underbrace{\atop}}} \overset{\epsilon}{\underset{\tau}{\underbrace{\atop}}} \overset{\epsilon}{\underset{\tau}{\underbrace{\atop}}} \overset{\epsilon}{\underset{\tau}{\underbrace{\atop}}} \overset{\epsilon}{\underset{\tau}{\underbrace{\atop}}} \overset{\epsilon}{\underset{\tau}{\underbrace{\tau}{\underbrace{\tau}}} \overset{\epsilon}{\underset{\tau}{\underbrace{\tau}} \overset{\epsilon}{\underset{\tau}{\underbrace{\tau}}} \overset{\epsilon}{\underset{\tau}{\underbrace{\tau}} \overset{\epsilon}{\underset{\tau}{\underbrace{\tau}}} \overset{\epsilon}{\underset{\tau}{\underbrace{\tau}}} \overset{\epsilon}{\underset{\tau}{\underbrace{\tau}}} \overset{\epsilon}{\underset{\tau}{\underbrace{\tau}}} \overset{\epsilon}{\underset{\tau}{\underbrace{\tau}} \overset{\epsilon}{\underset{\tau}{\underbrace{\tau}}} \overset{\epsilon}{\underset{\tau}{\underbrace{\tau}}} \overset{\epsilon}{\underset{\tau}{\underbrace{\tau}}} \overset{\epsilon}{\underset{\tau}{\underbrace{\tau}} \overset{\epsilon}{\underset{\tau}} \overset{\epsilon}{\underset{\tau$$

Figure 14: Wick contraction and the untwisted / twisted bands.

which take account of the fact that boundary cycles of non-oriented partial chord diagrams are not endowed with a specific orientation. To model the index structure i of the boundary length and point spectrum correctly, we assume these two matrices do not commute

$$[\Omega_{\rm P}, \Omega_{\rm L}] \neq 0.$$

Furthermore, we introduce corresponding generalized Miwa times

(4.10) 
$$u_{(i_1,\ldots,i_K)} = \frac{1}{N} \operatorname{Tr} \left( \Omega_{\mathrm{P}}^{i_1} \Omega_{\mathrm{L}}^{-1} \Omega_{\mathrm{P}}^{i_2} \Omega_{\mathrm{L}}^{-1} \cdots \Omega_{\mathrm{P}}^{i_K} \Omega_{\mathrm{L}}^{-1} \right),$$

which are invariant under the symmetry

$$u_{(i_1,\ldots,i_K)} = u_{(i_K,i_{K-1},\ldots,i_1)}.$$

This assignment implies the bijective correspondence (analogous to the orientable case discussed earlier) between non-oriented partial chord diagrams and Wick contractions, which is summarized in Table 2.

Table 2: The correspondence between partial chord diagrams and operator products in the real symmetric matrix integral.

Partial chord diagram	Gaussian average
A chord end on a backbone	$\Omega_{\rm L}^{-1}M$
A marked point on a backbone	$\Omega_{ m P}$
An underside of a backbone	$N\Omega_{ m L}^{-1}$
A backbone	$N \mathrm{Tr} \left( \Omega_{\mathrm{P}}^{\alpha_1} \Omega_{\mathrm{L}}^{-1} \Omega_{\mathrm{P}}^{\alpha_2} \Omega_{\mathrm{L}}^{-1} \cdots \Omega_{\underline{\mathrm{P}}}^{\alpha_K} \Omega_{\mathrm{L}}^{-1} \right)$
A Chord	A Wick contraction $\overline{MM}$

Using this correspondence, generating functions  $\widetilde{Z}^{P}(x, y; \{s_i\}; \{r_i\}), \widetilde{Z}^{L}(x, y; \{s_i\}; \{q_i\})$ , and  $\widetilde{Z}(x, y; \{s_i\}; \{u_i\})$  can be re-expressed in terms of matrix integrals. Repeating the same combinatorial arguments as for the orientable case in Section 2, we obtain the following three theorems.

268

**Theorem 4.3.** Let  $\widetilde{Z}_N^{\mathbf{P}}(y; \{s_i\}; \{r_i\})$  be the real symmetric matrix integral with the external symmetric matrix  $\Omega_{\mathbf{P}}$  of rank N

$$\widetilde{Z}_N^{\mathbf{P}}(y; \{s_i\}; \{r_i\}) = 11)$$

(4.11)  
= 
$$\frac{1}{\operatorname{Vol}_N(\mathbb{R})} \int_{\mathcal{H}_N(\mathbb{R})} dM \exp\left[-N\operatorname{Tr}\left(\frac{M^2}{4} - \sum_{i\geq 0} s_i y^{i/2} (M + y^{-1/2}\Omega_{\mathrm{P}})^i\right)\right],$$

where  $r_i$  are reverse Miwa times

(4.12) 
$$r_i = \frac{1}{N} \text{Tr}\Omega_{\rm P}^i.$$

This matrix integral agrees with the generating function (4.3)

(4.13) 
$$\widetilde{Z}_N^{\mathbf{P}}(y; \{s_i\}; \{r_i\}) = \widetilde{Z}^{\mathbf{P}}(N^{-1}, y; \{s_i\}, t_0 = 1, \{t_i = r_i\}_{i \ge 1}).$$

The  $t_0$ -dependence can be implemented by the following rescaling of parameters

(4.14) 
$$\widetilde{Z}_{t_0N}^{\mathrm{P}}(t_0y; \{t_0^{-1}s_i\}; \{t_0^{-1}t_i\}) = \widetilde{Z}^{\mathrm{P}}(N^{-1}, y; \{s_i\}; \{t_i = r_i\}).$$

**Theorem 4.4.** Let  $\widetilde{Z}_{N}^{L}(y; \{s_i\}; \{q_i\})$  be the real symmetric matrix integral with the external invertible symmetric matrix  $\Omega_{L}$  of rank N

$$\widetilde{Z}_{N}^{\mathrm{L}}(y; \{s_{i}\}; \{q_{i}\}) =$$

$$(4.15) \quad = \frac{1}{\mathrm{Vol}_{N}(\mathbb{R})} \int_{\mathcal{H}_{N}(\mathbb{R})} dM \exp\left[-N\mathrm{Tr}\left(\frac{M^{2}}{4} - \sum_{i \geq 0} s_{i}y^{i/2} \left(\Omega_{\mathrm{L}}^{-1}M\right)^{i}\Omega_{\mathrm{L}}^{-1}\right)\right],$$

where  $q_i$  are Miwa times

(4.16) 
$$q_i = \frac{1}{N} \text{Tr} \Omega_{\text{L}}^{-i}.$$

This matrix integral agrees with the generating function (4.4)

(4.17) 
$$\widetilde{Z}_{N}^{L}(y; \{s_{i}\}; \{q_{i}\}) = \widetilde{Z}^{L}(N^{-1}, y; \{s_{i}\}; \{q_{i}\})$$

As considered in (2.26) and Subsection 3.2, the specialization  $s_i = s$  of the matrix integral (4.15) gives the following reduced model

$$\widetilde{Z}_{N}^{\mathrm{L}}(y;s;\{q_{i}\}) = \widetilde{Z}_{N}^{\mathrm{L}}(y;\{s_{i}=s\};\{q_{i}\}) =$$

$$(4.18) \qquad \qquad = \frac{1}{\mathrm{Vol}_{N}(\mathbb{R})} \int_{\mathcal{H}_{N}(\mathbb{R})} dM \exp\left[-N\mathrm{Tr}\left(\frac{M^{2}}{4} + \frac{s}{y^{1/2}M - \Omega_{\mathrm{L}}}\right)\right].$$

The cut-and-join equation that follows from this reduced model is derived in the next subsection.

**Theorem 4.5.** Let  $\widetilde{Z}_N(y; \{s_i\}; \{u_i\})$  be the real symmetric matrix integral with the external invertible symmetric matrices  $\Omega_P$  and  $\Omega_L$  of rank N

$$\begin{aligned} \widetilde{\mathcal{Z}}_{N}(y; \{s_{i}\}; \{u_{i}\}) &= \\ (4.19) \\ &= \frac{1}{\operatorname{Vol}_{N}(\mathbb{R})} \int_{\mathcal{H}_{N}(\mathbb{R})} dM \exp\left[-N\operatorname{Tr}\left(\frac{M^{2}}{4} - \sum_{i\geq 0} s_{i}(y^{1/2}\Omega_{\mathrm{L}}^{-1}M + \Omega_{\mathrm{P}})^{i}\Omega_{\mathrm{L}}^{-1}\right)\right], \end{aligned}$$

and  $u_i$  be the generalized Miwa times defined in (4.10). This matrix integral agrees with the generating function (4.6)

(4.20) 
$$\widetilde{\mathcal{Z}}_N(y; \{s_i\}; \{u_i\}) = \widetilde{\mathcal{Z}}(N^{-1}, y; \{s_i\}; \{u_i\}).$$

### 4.2 Non-oriented analogues of cut-and-join equations

We derive now non-oriented analogues of cut-and-join equations discussed in Section 3. Analogously to the Hermitian matrix integral in (3.1), we introduce the following matrix integral.

**Definition 4.6.** Let  $U = U^{\mathrm{T}}$  and  $V = V^{\mathrm{T}}$  be rank N invertible symmetric matrices. We define a formal real symmetric matrix integral with parameters y,  $\{g_i\}_{i=-\infty}^{+\infty}$  as follows

$$\begin{aligned} &(4.21)\\ &\widetilde{Z}_{N}(y; \{g_{i}\}; U; V) = \\ &= \frac{1}{\operatorname{Vol}_{N}(\mathbb{R})} \int_{\mathcal{H}_{N}(\mathbb{R})} dM \, \exp\bigg[ -N\operatorname{Tr}\bigg(\frac{1}{4}M^{2} - \sum_{i \in \mathbb{Z}} g_{i}(y^{1/2}V^{-1}M + U)^{i}V^{-1}\bigg)\bigg]. \end{aligned}$$

The matrix integrals discussed in the previous subsection follow from this matrix integral by specializations

(4.22)  $\widetilde{Z}_{N}^{P}(y; \{s_{i}\}; \{r_{i}\}): g_{i<0} = 0, g_{i\geq0} = s_{i}, U = \Omega_{P}, V = I_{N},$ 

(4.23) 
$$Z_N^{\rm L}(y; \{s_i\}; \{q_i\}): g_{i<0} = 0, g_{i\geq 0} = s_i, U = 0, V = \Omega_{\rm L},$$

(4.24) 
$$\widetilde{Z}_{N}^{L}(y;s;\{q_{i}\}): g_{i\neq-1}=0, g_{-1}=-s, U=\Omega_{L}, V=-I_{N},$$

(4.25)  $\widetilde{\mathcal{Z}}_{N}(y; \{s_{i}\}; \{u_{i}\}): g_{i<0} = 0, g_{i>0} = s_{i}, U = \Omega_{\mathrm{P}}, V = \Omega_{\mathrm{L}},$ 

where  $I_N$  is the rank N identity matrix.

In Appendix A we prove the following proposition.

**Proposition 4.7.** The matrix integral  $\widetilde{Z}_N(y; \{g_i\}; U; V)$  in (4.21) obeys the partial differential equation

(4.26) 
$$\left[\frac{\partial}{\partial y} - \frac{1}{4N} \operatorname{Tr}(V^{-1})^{\mathrm{T}} \frac{\partial}{\partial A} (V^{-1})^{\mathrm{T}} \frac{\partial}{\partial A}\right] \widetilde{Z}_{N}(y; \{g_{i}\}; U; V) = 0,$$

where A is a matrix such that

$$U = A + A^{\mathrm{T}}.$$

From this proposition and by the specializations (4.22), (4.24), and (4.25) we find partial differential equations for the corresponding matrix integrals. For the specialization (4.23), because of U = 0 (and thus A = 0), the partial differential equation (4.26) cannot be reduced to a partial differential equation.

**Corollary 4.8.** The matrix integral  $\widetilde{Z}_N^{P}(y; \{s_i\}; \{r_i\})$  in (4.11),  $\widetilde{Z}_N^{L}(y; s; \{q_i\})$  in (4.18) and  $\widetilde{Z}_N(y; \{s_i\}; \{u_i\})$  in (4.19) obey partial differential equations

$$\begin{bmatrix} \frac{\partial}{\partial y} - \frac{1}{4N} \operatorname{Tr} \frac{\partial^2}{\partial \Lambda_{\rm P}^2} \end{bmatrix} \widetilde{Z}_N^{\rm P}(y; \{s_i\}; \{r_i\}) = 0,$$

$$\begin{bmatrix} \frac{\partial}{\partial y} - \frac{1}{4N} \operatorname{Tr} \frac{\partial^2}{\partial \Lambda_{\rm L}^2} \end{bmatrix} \widetilde{Z}_N^{\rm L}(y; s; \{q_i\}) = 0,$$

$$\begin{bmatrix} \frac{\partial}{\partial y} - \frac{1}{4N} \operatorname{Tr}(\Omega_{\rm L}^{-1})^{\rm T} \frac{\partial}{\partial \Lambda_{\rm P}} (\Omega_{\rm L}^{-1})^{\rm T} \frac{\partial}{\partial \Lambda_{\rm P}} \end{bmatrix} \widetilde{Z}_N(y; \{s_i\}; \{u_i\}) = 0,$$

where  $\Lambda_{\rm P}$  and  $\Lambda_{\rm L}$  are matrices satisfying

$$\Omega_{\rm P} = \Lambda_{\rm P} + \Lambda_{\rm P}^{\rm T}, \qquad \Omega_{\rm L} = \Lambda_{\rm L} + \Lambda_{\rm L}^{\rm T}.$$

From this corollary we obtain non-oriented analogues of cut-and-join equations, by rewriting the derivatives with respect to the external matrices  $\Lambda_{\rm P}$  and  $\Lambda_{\rm L}$  in Corollary 4.8 in terms of Miwa times  $r_i$  in (4.12),  $q_i$  in (4.16), and  $u_i$  in (4.10) as follows.

**Lemma 4.9.** Let  $L_1$ ,  $K_1$ ,  $M_1$ , and  $M_2^{\vee}$  denote differential operators

(4.28) 
$$L_1 = \frac{1}{2} \sum_{i \ge 1} i(i+1)r_i \frac{\partial}{\partial r_{i+2}},$$

(4.29) 
$$K_1 = \frac{1}{2} \sum_{i \ge 3} (i-2)(i-1)q_i \frac{\partial}{\partial q_{i-2}},$$

$$M_1 = \frac{1}{2} \sum_{K \ge 1} \sum_{\{i_1, \dots, i_K\}} \sum_{1 \le I \ne M \le K} \sum_{k=0}^{i_1 - 1} \sum_{m=0}^{M-1}$$

$$(4.30) \qquad u_{(m,i_{M-1},i_{M-2},\dots,i_{I+1},i_{I}-\ell-1,i_{M}-m-1,i_{M+1},\dots,i_{I-1},\ell)} \frac{\partial}{\partial u_{(i_{1},\dots,i_{K})}} \\ + \sum_{K \ge 1} \sum_{\{i_{1},\dots,i_{K}\}} \sum_{L=1}^{K} \sum_{\ell+m \le i_{I}-2} u_{(\ell,i_{I}-\ell-m-2,m,i_{I+1},\dots,i_{I-1})} \frac{\partial}{\partial u_{(i_{1},\dots,i_{K})}}$$

and

$$(4.31) M_2^{\vee} = \frac{1}{2} \sum_{K,L \ge 1} \sum_{\{i_1,\dots,i_K\}} \sum_{\{j_1,\dots,j_L\}} \sum_{I=1}^K \sum_{J=1}^L \sum_{\ell=0}^{i_I-1} \sum_{m=0}^{j_J-1} u_{\ell,i_{I-1},\dots,i_{I+1},i_I-\ell-1,j_J-m-1,j_{J+1},\dots,j_{J-1},m)} \frac{\partial^2}{\partial u_{(i_1,\dots,i_K)} \partial u_{(j_1,\dots,j_L)}}.$$

Then the derivatives with respect to  $\Lambda_P$  and  $\Lambda_L$  in Corollary 4.8 are rewritten as

$$(4.32) \qquad \begin{aligned} \frac{1}{4N} \operatorname{Tr} \frac{\partial^2}{\partial \Lambda_{\mathrm{P}}^2} f(\{r_i\}) &= \left(L_0 + \frac{1}{N}L_1 + \frac{2}{N^2}L_2\right) f(\{r_i\}), \\ \frac{1}{4N} \operatorname{Tr} \frac{\partial^2}{\partial \Lambda_{\mathrm{L}}^2} g(\{q_i\}) &= \left(K_0 + \frac{1}{N}K_1 + \frac{2}{N^2}K_2\right) g(\{q_i\}), \\ \frac{1}{4N} \operatorname{Tr} \left[(\Omega_{\mathrm{L}}^{-1})^{\mathrm{T}} \frac{\partial}{\partial \Lambda_{\mathrm{P}}} (\Omega_{\mathrm{L}}^{-1})^{\mathrm{T}} \frac{\partial}{\partial \Lambda_{\mathrm{P}}}\right] h(\{u_i\}) \\ &= \left(M_0 + \frac{1}{N}M_1 + \frac{1}{N^2} (M_2 + M_2^{\vee})\right) h(\{u_i\}), \end{aligned}$$

where  $f(\{r_i\})$ ,  $g(\{q_i\})$ , and  $h(\{u_i\})$  are functions of Miwa times  $r_i$ ,  $q_i$ , and  $u_i$ , respectively. Here  $L_{0,2}$ ,  $K_{0,2}$ , and  $M_{0,2}$  are defined in (3.7), (3.16), and (3.22), respectively.

The proof of this lemma is given in Appendix B. By combining Corollary 4.8 with Lemma 4.9 one arrives at the following theorem.

**Theorem 4.10.** The matrix integrals  $\widetilde{Z}_N^{\mathrm{P}}(y; \{s_i\}; \{r_i\})$  in (4.11),  $\widetilde{Z}_N^{\mathrm{L}}(y; s; \{q_i\})$  in (4.18), and  $\widetilde{Z}_N(y; \{s_i\}; \{u_i\})$  in (4.19) obey the cut-and-join equations

(4.33) 
$$\begin{aligned} \frac{\partial}{\partial y} \widetilde{Z}_{N}^{\mathrm{P}}(y; \{s_{i}\}; \{r_{i}\}) &= \widetilde{\mathcal{L}} \widetilde{Z}_{N}^{\mathrm{P}}(y; \{s_{i}\}; \{r_{i}\}), \\ \frac{\partial}{\partial y} \widetilde{Z}_{N}^{\mathrm{L}}(y; s; \{q_{i}\}) &= \widetilde{\mathcal{K}} \widetilde{Z}_{N}^{\mathrm{L}}(y; s; \{q_{i}\}), \\ \frac{\partial}{\partial y} \widetilde{\mathcal{Z}}_{N}(y; \{s_{i}\}; \{u_{i}\}) &= \widetilde{\mathcal{M}} \widetilde{\mathcal{Z}}_{N}(y; \{s_{i}\}; \{u_{i}\}) \end{aligned}$$

where

$$\begin{split} \widetilde{\mathcal{L}} &= L_0 + \frac{1}{N} L_1 + \frac{2}{N^2} L_2, \\ \widetilde{\mathcal{K}} &= K_0 + \frac{1}{N} K_1 + \frac{2}{N^2} K_2, \\ \widetilde{\mathcal{M}} &= M_0 + \frac{1}{N} M_1 + \frac{1}{N^2} (M_2 + M_2^{\vee}). \end{split}$$

272

Assuming certain initial conditions at y = 0, one can iteratively determine the above matrix integrals by solving the cut-and-join equations

$$\begin{aligned} \widetilde{Z}_{N}^{P}(y;\{s_{i}\};\{r_{i}\}) &= e^{y\widetilde{\mathcal{L}}}\widetilde{Z}_{N}^{P}(y=0;\{s_{i}\};\{r_{i}\}) = e^{y\widetilde{\mathcal{L}}}e^{N^{2}\sum_{i\geq0}s_{i}r_{i}}, \\ (4.34) \qquad \widetilde{Z}_{N}^{L}(y;s;\{q_{i}\}) &= e^{y\widetilde{\mathcal{K}}}\widetilde{Z}_{N}^{L}(y=0,s;\{q_{i}\}) = e^{y\widetilde{\mathcal{K}}}e^{N^{2}sq_{1}}, \\ \qquad \widetilde{Z}_{N}(y;\{s_{i}\};\{u_{i}\}) &= e^{y\widetilde{\mathcal{M}}}\widetilde{\mathcal{Z}}_{N}(y=0;\{s_{i}\};\{u_{i}\}) = e^{y\widetilde{\mathcal{M}}}e^{-N^{2}\sum_{i\geq0}s_{i}u_{(i)}} \end{aligned}$$

The cut-and-join equations (4.33) agree with those of [2, 7]. Finally, from Theorem 4.10 we find non-oriented analogues of cut-and-join equations for 1-backbone generating functions.

**Corollary 4.11.** The 1-backbone generating function  $\widetilde{\mathcal{F}}_1(x, y; \{s_i\}; \{u_i\})$  obtained by picking up the  $\mathcal{O}(s_i^1)$  term in  $\widetilde{\mathcal{Z}}_N(y; \{s_i\}; \{u_i\})$  is given by the following matrix integral

(4.35) 
$$\widetilde{\mathcal{F}}_{1}(N^{-1}, y; \{s_{i}\}; \{u_{i}\}) = \\ = \frac{1}{\operatorname{Vol}_{N}(\mathbb{R})} \int_{\mathcal{H}_{N}(\mathbb{R})} dM \, \mathrm{e}^{-N\operatorname{Tr}\frac{M^{2}}{4}} N \sum_{i \geq 0} s_{i} \operatorname{Tr}(y^{1/2}\Omega_{\mathrm{L}}^{-1}M + \Omega_{\mathrm{P}})^{i}\Omega_{\mathrm{L}}^{-1},$$

and it obeys the cut-and-join equation

(4.36) 
$$\frac{\partial}{\partial y}\widetilde{\mathcal{F}}_1(x,y;\{s_i\};\{u_i\}) = \widetilde{\mathcal{M}}\widetilde{\mathcal{F}}_1(x,y;\{s_i\};\{u_i\}),$$

where  $\widetilde{\mathcal{M}} = M_0 + xM_1 + x^2(M_2 + M_2^{\vee})$ . The solution is iteratively determined by

(4.37) 
$$\widetilde{\mathcal{F}}_1(x, y; \{s_i\}; \{u_i\}) = e^{y\widetilde{\mathcal{M}}} \widetilde{\mathcal{F}}_1(x, y = 0; \{s_i\}; \{u_i\}) = e^{y\widetilde{\mathcal{M}}} \left(x^{-2} \sum_{i \ge 0} s_i u_{(i)}\right).$$

# A Proof of Proposition 4.7

In this appendix we prove the Proposition 4.7, which states that the matrix integral

$$\begin{aligned} \widetilde{Z}_N(y; \{g_i\}; U; V) &= \\ &= \frac{1}{\operatorname{Vol}_N(\mathbb{R})} \int_{\mathcal{H}_N(\mathbb{R})} dM \, \exp\bigg[ -N\operatorname{Tr}\bigg(\frac{1}{4}M^2 - \sum_{i \in \mathbb{Z}} g_i(y^{1/2}V^{-1}M + U)^i V^{-1}\bigg) \bigg], \end{aligned}$$

obeys the partial differential equation

(A.1) 
$$\left[\frac{\partial}{\partial y} - \frac{1}{4N} \operatorname{Tr}(V^{-1})^{\mathrm{T}} \frac{\partial}{\partial A} (V^{-1})^{\mathrm{T}} \frac{\partial}{\partial A}\right] \widetilde{Z}_{N}(y; \{g_{i}\}; U; V) = 0,$$

where A is a matrix that satisfies  $U = A + A^{\mathrm{T}}$ .

*Proof.* In order to differentiate the matrix integral  $\widetilde{Z}_N(y; \{g_i\}; U; V)$  with respect to A we use the identities

$$\begin{aligned} \frac{\partial U_{\alpha\beta}}{\partial A_{\gamma\epsilon}} &= \delta_{\alpha\gamma}\delta_{\beta\epsilon} + \delta_{\alpha\epsilon}\delta_{\beta\gamma}, \\ \frac{\partial (y^{1/2}V^{-1}X)_{\alpha\beta}^{-1}}{\partial A_{\gamma\epsilon}} &= -(y^{1/2}V^{-1}X)_{\alpha\gamma}^{-1}(y^{1/2}V^{-1}X)_{\beta\epsilon}^{-1} - (y^{1/2}V^{-1}X)_{\alpha\epsilon}^{-1}(y^{1/2}V^{-1}X)_{\beta\gamma}^{-1}, \end{aligned}$$

where  $X = M + y^{-1/2}VU$ . Using this shifted variable X one obtains

$$\begin{split} \frac{1}{2N} \frac{\partial}{\partial A_{\alpha\beta}} \widetilde{Z}_N(y; \{g_i\}; U; V) &= \left\langle \sum_{i=0}^{\infty} y^{(i-1)/2} g_i \sum_{j=0}^{i-1} \left( (V^{-1}X)^j V^{-1} (V^{-1}X)^{i-j-1} \right)_{\alpha\beta} \right. \\ &\left. - \sum_{i=1}^{\infty} y^{-(i+1)/2} g_{-i} \sum_{j=0}^{i-1} \left( (V^{-1}X)^{-j-1} V^{-1} (V^{-1}X)^{-i+j} \right)_{\alpha\beta} \right\rangle_{\mathbb{R}}, \end{split}$$

where  $\langle \cdots \rangle_{\mathbb{R}}$  denotes the unnormalized average

$$\left\langle \mathcal{O}(X) \right\rangle_{\mathbb{R}} = \int_{\widetilde{\mathcal{H}}_{N}(\mathbb{R})} dX \ \mathcal{O}(X) \exp\left[ -N \operatorname{Tr}\left(\frac{1}{4} (X - y^{-1/2} V U)^{2} - \sum_{i \in \mathbb{Z}} y^{i/2} g_{i} (V^{-1} X)^{i} V^{-1} \right) \right].$$

Here  $\widetilde{\mathcal{H}}_N(\mathbb{R})$  is the space of shifted matrices  $X = M + y^{-1/2}VU$  with  $M \in \mathcal{H}_N(\mathbb{R})$ . It follows that

$$\begin{aligned} &\frac{1}{2N^2} \mathrm{Tr}(V^{-1})^{\mathrm{T}} \frac{\partial}{\partial A} (V^{-1})^{\mathrm{T}} \frac{\partial}{\partial A} \widetilde{Z}_N(y; \{g_i\}; U; V) = \\ &= \left\langle \sum_{i=0}^{\infty} y^{-1/2} g_i \sum_{j=0}^{i-1} \mathrm{Tr}(X - y^{-1/2} V U) V^{-1} (y^{1/2} V^{-1} X)^j V^{-1} (y^{1/2} V^{-1} X)^{i-j-1} \right. \end{aligned}$$

Partial Chord Diagrams and Matrix Models

(A.2)  

$$-\sum_{i=1}^{\infty} y^{-1/2} g_{-i} \sum_{j=0}^{i-1} \operatorname{Tr}(X - y^{-1/2}VU) V^{-1} (y^{1/2}V^{-1}X)^{-j-1} V^{-1} (y^{1/2}V^{-1}X)^{-i+j} \bigg\rangle_{\mathbb{R}}$$

On the other hand, by differentiating the matrix integral  $\widetilde{Z}_N(y; \{g_i\}; U; V)$  with respect to y one obtains the same expression as (A.2) times N/2, from which the partial differential equation (A.1) is obtained.

## B Proof of Lemma 4.9

In this appendix we prove the Lemma 4.9, which states that for functions  $f(\{r_i\})$ ,  $g(\{q_i\})$ , and  $h(\{u_i\})$  of Miwa times  $r_i$  in (4.12),  $q_i$  in (4.16), and  $u_i$  in (4.10), we find

(B.1) 
$$\frac{1}{4N} \operatorname{Tr} \frac{\partial^2}{\partial \Lambda_{\mathrm{P}}^2} f(\{r_i\}) = \left(L_0 + \frac{1}{N}L_1 + \frac{2}{N^2}L_2\right) f(\{r_i\}),$$

(B.2) 
$$\frac{1}{4N} \operatorname{Tr} \frac{\partial^2}{\partial \Lambda_{\mathrm{L}}^2} g(\{q_i\}) = \left(K_0 + \frac{1}{N} K_1 + \frac{2}{N^2} K_2\right) g(\{q_i\}),$$

(B.3) 
$$\frac{\frac{1}{4N} \operatorname{Tr} \left[ (\Omega_{\mathrm{L}}^{-1})^{\mathrm{T}} \frac{\partial}{\partial \Lambda_{\mathrm{P}}} (\Omega_{\mathrm{L}}^{-1})^{\mathrm{T}} \frac{\partial}{\partial \Lambda_{\mathrm{P}}} \right] h(\{u_{i}\}) \\ = \left( M_{0} + \frac{1}{N} M_{1} + \frac{1}{N^{2}} (M_{2} + M_{2}^{\vee}) \right) h(\{u_{i}\})$$

where  $L_{0,1,2}$ ,  $K_{0,1,2}$ ,  $M_{0,1,2}$ , and  $M_2^{\vee}$  are defined in (3.7), (3.16), (3.22), and in Lemma 4.9. Here the matrices  $\Lambda_{\rm P}$  and  $\Lambda_{\rm L}$  satisfy  $\Omega_{\rm P} = \Lambda_{\rm P} + \Lambda_{\rm P}^{\rm T}$  and  $\Omega_{\rm L} = \Lambda_{\rm L} + \Lambda_{\rm L}^{\rm T}$ .

*Proof.* First we prove (B.1). Consider the derivative  $\partial/\partial \Lambda_P$  of the reverse Miwa time  $r_i$ 

$$\frac{\partial r_i}{\partial \Lambda_{\mathrm{P}\beta\alpha}} = \frac{2i}{N} \Omega_{\mathrm{P}\alpha\beta}^{i-1}, \qquad \mathrm{Tr} \frac{\partial^2 r_i}{\partial \Lambda_{\mathrm{P}}^2} = 2Ni \sum_{j=1}^{i-1} r_{j-1} r_{i-j-1} + 2i(i-1)r_{i-2}.$$

Using these relations the left hand side of (B.1) is rewritten as

$$\begin{split} \frac{1}{4N} \mathrm{Tr} \frac{\partial^2}{\partial \Lambda_{\mathrm{P}}^2} f(\{r_i\}) &= \frac{1}{4N} \sum_{i \geq 1} \mathrm{Tr} \frac{\partial^2 r_i}{\partial \Lambda_{\mathrm{P}}^2} \frac{\partial f(\{r_i\})}{\partial r_i} + \frac{1}{4N} \sum_{i,j \geq 1} \mathrm{Tr} \frac{\partial r_i}{\partial \Lambda_{\mathrm{P}}} \frac{\partial r_j}{\partial \Lambda_{\mathrm{P}}} \frac{\partial^2 f(\{r_i\})}{\partial r_i \partial r_j} \\ &= \frac{1}{2} \sum_{i \geq 2} \sum_{j=1}^{i-1} i r_{j-1} r_{i-j-1} \frac{\partial f(\{r_i\})}{\partial r_i} + \frac{1}{2N} \sum_{i \geq 2} i(i-1) r_{i-2} \frac{\partial f(\{r_i\})}{\partial r_i} \\ &+ \frac{1}{N^2} \sum_{i,j \geq 1} i j r_{i+j-2} \frac{\partial}{\partial r_i} \frac{\partial}{\partial r_j} f(\{r_i\}). \end{split}$$

This agrees with the right hand side of (B.1).

Second, we prove (B.2). Using the identity

$$\frac{\partial \Omega_{L\gamma\epsilon}^{-1}}{\partial \Lambda_{L\alpha\beta}} = -\Omega_{L\beta\epsilon}^{-1} \Omega_{L\gamma\alpha}^{-1} - \Omega_{L\alpha\epsilon}^{-1} \Omega_{L\gamma\beta}^{-1},$$

one finds that

$$\frac{\partial q_i}{\partial \Lambda_{\mathrm{L}\alpha\beta}} = -2\frac{i}{N}\Omega_{\mathrm{L}\beta\alpha}^{-i-1}, \qquad \mathrm{Tr}\frac{\partial^2}{\partial \Lambda_{\mathrm{L}}^2}q_i = 2i(i+1)q_{i+2} + 2iN\sum_{j=0}^i q_{i+1}q_{j+1}.$$

Then the left hand side of (B.2) yields

$$\begin{split} \frac{1}{4N} \mathrm{Tr} \frac{\partial^2}{\partial \Lambda_{\mathrm{L}}^2} g(\{q_i\}) &= \frac{1}{2} \sum_{i \ge 1} \sum_{j=0}^i iq_{i+1}q_{j+1} \frac{\partial g(\{q_i\})}{\partial q_i} + \frac{1}{2N} \sum_{i \ge 1} i(i+1)q_{i+2} \frac{\partial g(\{q_i\})}{\partial q_i} \\ &+ \frac{1}{N^2} \sum_{i,j \ge 1} ijq_{i+j+2} \frac{\partial^2 g(\{q_i\})}{\partial q_i \partial q_j}. \end{split}$$

This agrees with the right hand side of (B.2).

Finally we prove (B.3). Using the chain rule, the derivative action on the left hand side of (B.3) is written as

$$\begin{split} &\operatorname{Tr}\left[(\Omega_{\mathrm{L}}^{-1})^{\mathrm{T}}\frac{\partial}{\partial\Lambda_{\mathrm{P}}}(\Omega_{\mathrm{L}}^{-1})^{\mathrm{T}}\frac{\partial}{\partial\Lambda_{\mathrm{P}}}\right]h(\{u_{\boldsymbol{i}}\}) = \\ &= \sum_{K\geq 1}\sum_{\{i_{1},\ldots,i_{K}\}}\operatorname{Tr}\left[(\Omega_{\mathrm{L}}^{-1})^{\mathrm{T}}\frac{\partial}{\partial\Lambda_{\mathrm{P}}}(\Omega_{\mathrm{L}}^{-1})^{\mathrm{T}}\frac{\partial}{\partial\Lambda_{\mathrm{P}}}u_{(i_{1},\ldots,i_{K})}\right]\frac{\partial}{\partial u_{(i_{1},\ldots,i_{K})}}h(\{u_{\boldsymbol{i}}\}) \\ &+ \sum_{K,L\geq 1}\sum_{\{i_{1},\ldots,i_{K}\}}\sum_{\{j_{1},\ldots,j_{L}\}}\operatorname{Tr}\left[(\Omega_{\mathrm{L}}^{-1})^{\mathrm{T}}\frac{\partial}{\partial\Lambda_{\mathrm{P}}}u_{(i_{1},\ldots,i_{K})}(\Omega_{\mathrm{L}}^{-1})^{\mathrm{T}}\frac{\partial}{\partial\Lambda_{\mathrm{P}}}u_{(j_{1},\ldots,j_{L})}\right] \\ &\times \frac{\partial^{2}}{\partial u_{(i_{1},\ldots,i_{K})}\partial u_{(j_{1},\ldots,j_{L})}}h(\{u_{\boldsymbol{i}}\}). \end{split}$$

Each of the coefficients yields

$$\begin{split} \operatorname{Tr} \left[ (\Omega_{\mathrm{L}}^{-1})^{\mathrm{T}} \frac{\partial}{\partial \Lambda_{\mathrm{P}}} (\Omega_{\mathrm{L}}^{-1})^{\mathrm{T}} \frac{\partial}{\partial \Lambda_{\mathrm{P}}} u_{(i_{1},\ldots,i_{K})} \right] = \\ &= 2 \sum_{1 \leq I \neq M \leq K}^{K} \sum_{\ell=0}^{i_{I}-1} \sum_{m=0}^{i_{M}-1} \frac{1}{N} \Big( \operatorname{Tr} (\Omega_{\mathrm{P}}^{i_{I}-\ell-1} \Omega_{\mathrm{L}}^{-1} \Omega_{\mathrm{P}}^{i_{I}+1} \Omega_{\mathrm{L}}^{-1} \cdots \Omega_{\mathrm{P}}^{i_{M-1}} \Omega_{\mathrm{L}}^{-1} \Omega_{\mathrm{P}}^{m} \Omega_{\mathrm{L}}^{-1}) \\ &\qquad \times \operatorname{Tr} (\Omega_{\mathrm{P}}^{i_{M}-m-1} \Omega_{\mathrm{L}}^{-1} \Omega_{\mathrm{P}}^{i_{M}+1} \Omega_{\mathrm{L}}^{-1} \cdots \Omega_{\mathrm{P}}^{i_{I}-1} \Omega_{\mathrm{L}}^{\ell} \Omega_{\mathrm{P}}^{\ell} \Omega_{\mathrm{L}}^{-1}) \\ &\qquad + \operatorname{Tr} (\Omega_{\mathrm{P}}^{m} \Omega_{\mathrm{L}}^{-1} \Omega_{\mathrm{P}}^{i_{M}-1} \Omega_{\mathrm{L}}^{-1} \Omega_{\mathrm{P}}^{i_{M}-2} \Omega_{\mathrm{L}}^{-1} \cdots \Omega_{\mathrm{P}}^{i_{I}+1} \Omega_{\mathrm{L}}^{-1} \Omega_{\mathrm{P}}^{\ell} \Omega_{\mathrm{L}}^{-1}) \\ &\qquad \cdot \Omega_{\mathrm{P}}^{i_{M}-m-1} \Omega_{\mathrm{L}}^{-1} \Omega_{\mathrm{P}}^{i_{M}+1} \Omega_{\mathrm{L}}^{-1} \Omega_{\mathrm{P}}^{i_{M}+2} \Omega_{\mathrm{L}}^{-1} \cdots \Omega_{\mathrm{P}}^{i_{I}-1} \Omega_{\mathrm{P}}^{\ell} \Omega_{\mathrm{L}}^{-1}) \Big) \end{split}$$

$$\begin{split} &+4\sum_{I=1}^{K}\sum_{\ell+m\leq i_{I}-2}\Bigl(\frac{1}{N}\mathrm{Tr}(\Omega_{\mathrm{P}}^{\ell}\Omega_{\mathrm{L}}^{-1}\Omega_{\mathrm{P}}^{m}\Omega_{\mathrm{L}}^{-1}\Omega_{\mathrm{P}}^{i_{I+1}}\Omega_{\mathrm{L}}^{-1}\cdots\Omega_{\mathrm{P}}^{i_{I-1}}\Omega_{\mathrm{L}}^{-1})\mathrm{Tr}(\Omega_{\mathrm{P}}^{i_{I}-\ell-m-2}\Omega_{\mathrm{L}}^{-1})\\ &+\mathrm{Tr}(\Omega_{\mathrm{P}}^{\ell}\Omega_{\mathrm{L}}^{-1}\Omega_{\mathrm{P}}^{m}\Omega_{\mathrm{L}}^{-1}\Omega_{\mathrm{P}}^{i_{I}-\ell-m-2}\Omega_{\mathrm{L}}^{-1}\Omega_{\mathrm{P}}^{i_{I+1}}\Omega_{\mathrm{L}}^{-1}\cdots\Omega_{\mathrm{P}}^{i_{I-1}}\Omega_{\mathrm{L}}^{-1})\Bigr)\\ &=2\sum_{1\leq I\neq M}\sum_{K}\sum_{\ell=0}^{i_{I}-1}\sum_{m=0}^{i_{M}-1}\Bigl(Nu_{(i_{I}-\ell-1,i_{I+1},\ldots,i_{M-1},m)}u_{(i_{M}-m-1,i_{M+1},\ldots,i_{I-1},\ell)}\\ &+u_{(m,i_{M-1},i_{M-2},\ldots,i_{I+1},i_{I}-\ell-1,i_{M}-m-1,i_{M+1},\ldots,i_{I-1},\ell)}\Bigr)\\ &+4\sum_{I=0}^{K}\sum_{\ell+m\leq i_{I}-2}\Bigl(Nu_{(\ell,m,i_{I+1},\ldots,i_{I-1})}u_{(i_{I}-\ell-m-2)}+u_{(\ell,i_{I}-\ell-m-2,m,i_{I+1},\ldots,i_{I-1},i_{I})}\Bigr),\end{split}$$

and

$$\begin{split} &\operatorname{Tr}\left[(\Omega_{\mathrm{L}}^{-1})^{\mathrm{T}}\frac{\partial}{\partial\Lambda_{\mathrm{P}}}u_{(i_{1},\ldots,i_{K})}(\Omega_{\mathrm{L}}^{-1})^{\mathrm{T}}\frac{\partial}{\partial\Lambda_{\mathrm{P}}}u_{(j_{1},\ldots,j_{L})}\right] = \\ &= \sum_{I=1}^{K}\sum_{J=1}^{L}\sum_{\ell=0}^{L}\sum_{m=0}^{i_{I}-1}\frac{2}{N^{2}}\bigg(\operatorname{Tr}(\Omega_{\mathrm{P}}^{i_{I}-\ell-1}\Omega_{\mathrm{L}}^{-1}\Omega_{\mathrm{P}}^{j_{I}+1}\Omega_{\mathrm{L}}^{-1}\cdots\Omega_{\mathrm{L}}^{i_{I}-1}\Omega_{\mathrm{P}}^{\ell}\Omega_{\mathrm{L}}^{-1} \\ &\quad \cdot\Omega_{\mathrm{P}}^{j_{J}-m-1}\Omega_{\mathrm{L}}^{-1}\Omega_{\mathrm{P}}^{j_{J}+1}\Omega_{\mathrm{L}}^{-1}\cdots\Omega_{\mathrm{P}}^{j_{J}-1}\Omega_{\mathrm{L}}^{-1}\Omega_{\mathrm{P}}^{m}\Omega_{\mathrm{L}}^{-1}) \\ &\quad +\operatorname{Tr}(\Omega_{\mathrm{P}}^{\ell}\Omega_{\mathrm{L}}^{-1}\Omega_{\mathrm{P}}^{i_{I}-1}\Omega_{\mathrm{L}}^{-1}\cdots\Omega_{\mathrm{P}}^{j_{I}-1}\Omega_{\mathrm{P}}^{-1}\Omega_{\mathrm{L}}^{-1}\Omega_{\mathrm{P}}^{m}\Omega_{\mathrm{L}}^{-1}) \\ &\quad \cdot\Omega_{\mathrm{P}}^{j_{J}-m-1}\Omega_{\mathrm{L}}^{-1}\Omega_{\mathrm{P}}^{j_{J}+1}\Omega_{\mathrm{L}}^{-1}\cdots\Omega_{\mathrm{P}}^{j_{J}-1}\Omega_{\mathrm{L}}^{-1}\Omega_{\mathrm{P}}^{m}\Omega_{\mathrm{L}}^{-1})\bigg) \\ &= \frac{2}{N}\sum_{I=1}^{K}\sum_{J=1}^{L}\sum_{\ell=0}^{i_{I}-1}\sum_{m=0}^{j_{J}-1}\bigg(u_{(i_{I}-\ell-1,i_{I+1},\ldots,i_{I-1},\ell,j_{J}-m-1,j_{J+1},\ldots,j_{J-1},m)} \\ &\quad +u_{(\ell,i_{I-1},\ldots,i_{I+1},i_{I}-\ell-1,j_{J}-m-1,j_{J+1},\ldots,j_{J-1},m)}\bigg). \end{split}$$

Then one obtains the right hand side of (B.3).

### References

- T. Akutsu, Dynamic programming algorithms for RNA secondary structure prediction with pseudoknots, Disc. Appl. Math. 104 (2000) 45–62.
- [2] N. V. Alexeev, J. E. Andersen, R. C. Penner, and P. Zograf, *Enumeration of chord diagrams on many intervals and their non-orientable analogs*, Adv. Math. **289** (2016) 1056–1081, arXiv:1307.0967 [math.CO].
- [3] J. E. Andersen, A. J. Bene, J. -B. Meilhan, and R. C. Penner, *Finite type invariants and fatgraphs*, Adv. Math. **225** (2010), 2117–2161, arXiv:0907.2827 [math.GT].

- [4] J. E. Andersen, L. O. Chekhov, R. C. Penner, C. M. Reidys, and P. Sulkowski, *Topological recursion for chord diagrams, RNA complexes, and cells in moduli* spaces, Nucl. Phys. B866 (2013) 414–443, arXiv:1205.0658 [hep-th].
- [5] J. E. Andersen, L. O. Chekhov, R. C. Penner, C. M. Reidys, and P. Sulkowski, Enumeration of RNA complexes via random matrix theory, Biochem. Soc. Trans. 41 (2013) 652–655, arXiv:1303.1326 [q-bio.QM].
- [6] J. E. Andersen, H. Fuji, M. Manabe, R. C. Penner, and P. Sulkowski, Enumeration of chord diagrams via topological recursion and quantum curve techniques, Travaux Mathématiques, 25 (2017), 285–323, arXiv:1612.05839 [math-ph].
- [7] J. E. Andersen, H. Fuji, R. C. Penner, and C. M. Reidys, *The boundary length and point spectrum enumeration of partial chord diagrams via cut and join recursion*, Travaux Mathématiques, **25** (2017), 213–232, arXiv:1612.06482 [math-ph].
- [8] J. E. Andersen, F. W. D. Huang, R. C. Penner, C. M. Reidys, *Topology of RNA-RNA interaction structures*, J. Comp. Biol. **19** (2012) 928–943, arXiv:1112.6194 [math.CO].
- [9] J. E. Andersen, J. Mattes, and N. Reshetikhin, The Poisson structure on the moduli space of flat connections and chord diagrams, Topology 35 (1996) 1069-1083.
- [10] J. E. Andersen, J. Mattes, and N. Reshetikhin, Quantization of the algebra of chord diagrams, Math. Proc. Camb. Phil. Soc. **124** (1998) 451–467, arXiv:qalg/9701018.
- [11] J. E. Andersen, R. C. Penner, C. M. Reidys, and R. R. Wang, *Linear chord diagrams on two intervals*, arXiv:1010.5857 [math.CO].
- [12] J. E. Andersen, R. C. Penner, C. M. Reidys, and M. S. Waterman, *Enumeration of linear chord diagrams*, J. Math. Biol. **67** (2013) 1261–78, arXiv:1010.5614 [math.CO].
- [13] M. Andronescu, V. Bereg, H. H. Hoos, and A. Condon, RNA STRAND: The RNA secondary structure and statistical analysis database, BMC Bioinformatics, 9 (2008) 340.
- [14] D. Bar-Natan, On the Vassiliev knot invariants, Topology 34 (1995) 423-475.
- [15] D. Bessis, C. Itzykson, and J. B. Zuber, Quantum field theory techniques in graphical enumeration, Adv. Appl. Math. 1 (1980) 109–157.

- [16] R. Campoamor-Stursberg, and V. O. Manturov, *Invariant tensor formulas via chord diagrams*, Jour. Math. Sci. **108** (2004) 3018–3029.
- [17] H. L. Chen, A. Condon, and H. Jabbari, An O(n<sup>5</sup>) Algorithm for MFE prediction of kissing hairpins and 4-chains in nucleic acids, J. Comput. Biol. 16 (2009) 803–815.
- [18] J. Courtiel, and K. Yeats, *Terminal chords in connected chord diagrams*, arXiv:1603.08596 [math.CO].
- [19] R. Dijkgraaf, H. L. Verlinde, and E. P. Verlinde, Loop equations and Virasoro constraints in nonperturbative 2-d quantum gravity, Nucl. Phys. B348, 435 (1991).
- [20] O. Dumitrescu, M. Mulase, B. Safnuk, and A. Sorkin, *The spectral curve of the Eynard-Orantin recursion via the Laplace transform*, Contemp. Math. 593, 263 (2013), arXiv:1202.1159 [math.AG].
- [21] F. J. Dyson, The S matrix in quantum electrodynamics, Phys. Rev. 75 (1949) 1736–1755.
- [22] M. Fukuma, H. Kawai, and R. Nakayama, Continuum Schwinger-dyson equations and universal structures in two-dimensional quantum gravity, Int. J. Mod. Phys. A6, 1385 (1991).
- [23] S. Garoufalidis, and M. Marino, Universality and asymptotics of graph counting problems in nonorientable surfaces, Jour. Comb. Thor. A117 (2010) 715– 740, arXiv:0812.1195 [math.CO].
- [24] I. P. Goulden, J. L. Harer, and D. M. Jackson, A geometric parametrization for the virtual Euler characteristics of the moduli spaces of real and complex algebraic curves, Trans. Amer. Math. Soc. 353 (2001), no. 11, 4405–4427.
- [25] I. L. Hofacker, W. Fontana, P. F. Stadler, L. S. Bonhoffer, M. Tacker, and P. Schuster, *Fast folding and comparison of RNA secondary structures*, Monatsh. Chem. **125** (1994) 167–188.
- [26] R. A. Janik, M. A. Nowak, G. Papp, and Z. Ismail, *Green's functions in non-hermitian random matrix models*, Physica E9 (2001) 456–462, arXiv: cond-mat/9909085.
- [27] M. Kazarian, and P. Zograf, Virasoro constraints and topological recursion for Grothendieck's dessin counting, Lett. Math. Phys. 105 (2015) 1057–1084, arXiv:1406.5976 [math.CO].

- [28] D. A. M. Konings, and R. R. Gutell A comparison of thermodynamic foldings with comparatively derived structures of 16s and 16s-like r RNAs, RNA 1 (1995) 559–574.
- [29] M. Kontsevich, Funk. Anal.&Prilozh. 25 (1991) 50 (in Russian); Intersection theory on the moduli space of curves and the matrix Airy function, Comm. Math. Phys. 147 (1992) 1–23.
- [30] M. Kontsevich, Vassiliev's knot invariants, Adv. Sov. Math. 16 (1993) 137– 150.
- [31] A. Loria, and T. Pan, Domain structure of the ribozyme from eubacterial ribonuclease, RNA 2 (1996) 551–563.
- [32] R. B. Lyngsø, and C. N. Pedersen, RNA pseudoknot prediction in energybased models, J. Comput. Biol. 7 (2000) 409–427.
- [33] N. Marie, and K. Yeats, A chord diagram expansion coming from some Dyson-Schwinger equations, Comm. Numb. Theo. Phys. 7 (2013) 251–291, arXiv:1210.5457 [math.CO].
- [34] J. S. McCaskill, The equilibrium partition function and base pair binding probabilities for RNA secondary structure, Biopolymers 29 (1990) 1105–1119.
- [35] M. L. Mehta, *Random matrices*, 2nd edition, Academic Press (1991).
- [36] A. Morozov, and Sh. Shakirov, Generation of matrix models by W-operators, JHEP 0904 (2009) 064, arXiv:0902.2627 [hep-th].
- [37] M. Mulase, Lectures on the asymptotic expansion of a hermitian matrix integral, Springer Lecture Notes in Physics vol. 502, H. Aratyn et al., Editors, (1998) 91–134, arXiv:math-ph/9811023.
- [38] M. Mulase, and P. Sulkowski, Spectral curves and the Schroedinger equations for the Eynard-Orantin recursion, Adv. Theor. Math. Phys. 19 (2015) 955– 1015, arXiv:1210.3006 [math-ph].
- [39] M. Mulase, and A. Waldron, Duality of orthogonal and symplectic matrix integrals and quaternionic Feynman graphs, Commun. Math. Phys. 240 (2003) 553–586, arXiv:math-ph/0206011.
- [40] R. Nussinov, G. Piecznik, J. R. Griggs, and D. J. Kleitman, Algorithms for loop matching, SIAM J. Appl. Math. 35 (1978) 68–82.
- [41] H. Orland, and A. Zee, RNA folding and large N matrix theory, Nucl. Phys. B620 (2002) 456–476, arXiv:cond-mat/0106359 [cond-mat.stat-mech].

- [42] R. C. Penner, The decorated Teichmüller space of punctured surfaces, Comm. Math. Phys. 113 (1987) 299–339.
- [43] R. C. Penner, Perturbative series and the moduli space of Riemann surfaces, J. Diff. Geom. 27 (1988) 35–53.
- [44] R. C. Penner, The simplicial compactification of Riemann's moduli space, Proceedings of the 37th Taniguchi Symposium, World Scientific (1996), 237– 252.
- [45] R. C. Penner, Cell decomposition and compactification of Riemann's moduli space in decorated Teichmüller theory, In Tongring, N. and Penner, R.C. (eds) Woods Hole Mathematics-Perspectives in Math and Physics, World Scientific, Singapore, arXiv:math/0306190 [math.GT].
- [46] R. C. Penner, Moduli spaces and macromolecules, Bull. Amer. Math. Soc. 53 (2016) 217–268.
- [47] R. C. Penner, and M. S. Waterman, Spaces of RNA secondary structures, Adv. Math. 101 (1993) 31–49.
- [48] M. Peskin, and D. V. Schroeder, An introduction to quantum field theory, Westview Press (1995).
- [49] C. M. Reidys, Combinatorial and computational biology of pseudoknot RNA, Springer, Applied Math series 2010.
- [50] C. M. Reidys, F. W. D. Huang, J. E. Andersen, R. C. Penner, P. F. Stadler, and M. E. Nebel, *Topology and prediction of RNA pseudoknots*, Bioinformatics **27** (2011) 1076–85.
- [51] E. Rivas, and S. R. Eddy, A dynamic programming algorithm for RNA structure prediction including pseudoknots, J Mol Biol. 285 (1999) 2053–2068.
- [52] P. G. Silvestrov, Summing graphs for random band matrices, Phys. Rev. E55 (1997), 6419-6432, arXiv:cond-mat/9610064 [cond-mat.stat-mech].
- [53] D. W. Staple, and S. E. Butcher, Pseudoknots: RNA structures with diverse functions, PLoS Biol 3 (6) (2005) 956–959.
- [54] J. J. M. Verbaarschot, H. A. Weidenmuller, and M. R. Zirnbauer, Grassmann integration in stochastic quantum physics: The case of compound nucleus scattering, Phys. Rept. 129 (1985) 367–438.
- [55] G. Vernizzi, and H. Orland, Large-N random matrices for RNA folding, Acta Phys. Polon. B36 (2005) 2821–2827.

- [56] G. Vernizzi, H. Orland, and A. Zee, Enumeration of RNA structures by matrix models, Phys. Rev. Lett. 94 168103, arXiv:q-bio/0411004 [q-bio.BM].
- [57] M. S. Waterman, Secondary structure of single-stranded nucleic acids, Adv. Math. (Suppl. Studies) 1 (1978) 167–212.
- [58] M. S. Waterman, and T. F. Smith, RNA secondary structure: A complete mathematical analysis, Math. Biosci. 42 (1978) 257-266.
- [59] E. Westhof, and L. Jaeger RNA pseudoknots, Curr. Opin. Chem. Biol. 2 (1992) 327–333.
- [60] M. Zucker, On finding all suboptimal foldings of an RNA molecule, Science **244** (1989) 48–52.
- [61] M. Zucker, and P. Stiegler, Optimal computer folding of larger RNA sequences using thermodynamics and auxiliary information, Nucleic Acids Res. 9 (1981) 133 - 148.
#### Partial Chord Diagrams and Matrix Models

Jørgen Ellegaard Andersen Centre for Quantum Geometry of Moduli Spaces, Department of Mathematics, Aarhus University, DK-8000 Aarhus C Denmark jea.qgm@gmail.com

Hiroyuki Fuji Faculty of Education, Kagawa University, Takamatsu 760-8522 Japan fuji@ed.kagawa-u.ac.jp

Masahide Manabe Faculty of Physics, University of Warsaw, ul. Pasteura 5, 02-093 Warsaw, Poland masahidemanabe@gmail.com

Robert C. Penner Institut des Hautes Études Scientifiques, 35 route de Chartres, 91440 Burs-sur-Yvette, France; Division of Physics, Mathematics and Astronomy, California Institute of Technology, Pasadena, CA 91125, USA rpenner@caltech.edu, rpenner@ihes.fr

Piotr Sułkowski
Faculty of Physics, University of Warsaw,
ul. Pasteura 5, 02-093 Warsaw, Poland;
Walter Burke Institute for Theoretical Physics,
California Institute of Technology, Pasadena, CA 91125, USA
psulkows@fuw.edu.pl

Travaux mathématiques, Volume 25 (2017), 285–323, © at the authors

## Enumeration of Chord Diagrams via Topological Recursion and Quantum Curve Techniques

## by Jørgen Ellegaard Andersen, Hiroyuki Fuji, Masahide Manabe, Robert C. Penner, and Piotr Sułkowski<sup>1</sup>

#### Abstract

In this paper we consider the enumeration of orientable and non-orientable chord diagrams. We show that this enumeration is encoded in appropriate expectation values of the  $\beta$ -deformed Gaussian and RNA matrix models. We evaluate these expectation values by means of the  $\beta$ -deformed topological recursion, and – independently – using properties of quantum curves. We show that both these methods provide efficient and systematic algorithms for counting of chord diagrams with a given genus, number of backbones and number of chords.

## 1 Introduction

A chord diagram is a graph which can be realised in the plane as follows. It is comprised of a collection of b line segments (called backbones) on the real axis with k semi-circles (called chords) in the upper-half plane attached to the line segments. All chords are attached at different points on the backbones. A chord diagram comes from its realisation in the plane with a natural fatgraph structure, namely, half edges incident to each trivalent vertex are endowed with a cyclic order induced from the orientation of the plane. For a chord diagram c the fatgraph structure allows us, in the usual way, to define a surface  $\Sigma_c$ , which is simply just a

 $<sup>^1</sup>Keywords:$  Chord diagram, fat<br/>graph,  $\beta$  -deformation, topological recursion, quantum curve AMS Classification: 05A15, 05A16, 81T18, 81T45, 92-08

Acknowledgments: JEA and RCP are supported by the Centre for Quantum Geometry of Moduli Spaces which is funded by the Danish National Research Foundation. The research of HF is supported by the Grant-in-Aid for Research Activity Start-up [# 15H06453], Grant-in-Aid for Scientific Research(C) [# 26400079], and Grant-in-Aid for Scientific Research(B) [# 16H03927] from the Japan Ministry of Education, Culture, Sports, Science and Technology. The work of MM and PS is supported by the ERC Starting Grant no. 335739 "Quantum fields and knot homologies" funded by the European Research Council under the European Union's Seventh Framework Programme. PS also acknowledges the support of the Foundation for Polish Science, and RCP acknowledges the kind support of Institut Henri Poincaré where parts of this manuscript were written.

small tubular neighbourhood of the realisation of the chord diagram in the plane, see left and middle diagrams in Figure 1. Let n be the number of boundary cycles, and g be the genus of the skinny surface  $\Sigma_c$ . Then the Euler characteristic  $\chi$  of  $\Sigma_c$  is given by



Figure 1: A chord diagram c, its skinny surface  $\Sigma_c$ , and the associated ribbon surface  $R_c$ .

To present a chord diagram c more simply and efficiently, we collapse each fattened backbone in  $\Sigma_c$  into a polyvalent fattened vertex, see the right side of Figure 1. We call the resulting surface  $R_c$  and we refer to it as the *Ribbon surface* associated to c. In order for c to be uniquely determined by the ribbon surface  $R_c$ , at each vertex we attach a tail at the place corresponding to the beginning of the backbone, see Figure 2.



Figure 2: Polyvalent fatten vertex with a tail

In this paper we consider the enumeration of chord diagrams with the topological filtration induced by the genus and the number of backbones, employing matrix model techniques. Chord diagrams are widely used objects in pure and applied mathematics, see e.g. [19, 64, 14, 15, 8, 29]. Furthermore, they are now used also in the biological context for characterisation of secondary and pseudoknot structures of RNA molecules [77, 76, 75, 86, 85, 17, 80].<sup>2</sup> In particular, motivated by the study of RNA pseudoknot structures, a matrix model for the enumeration of chord diagrams – which we refer to as the RNA-matrix model – was constructed in [9, 10]. In this paper we study the  $\beta$ -deformed version of this model and present how it encodes orientable and non-orientable chord diagrams.

<sup>&</sup>lt;sup>2</sup>The combinatorial aspects of interacting RNA molecules with the associated genus filtration are also discussed in [23, 78, 79, 87, 16, 17, 18, 4, 11, 12], and folding algorithms are studied in [22, 13, 81]. The matrix model approach to the enumeration of possible RNA structures is also studied in [20, 21, 47, 48, 49, 50, 51, 52, 53].

### 1.1 The RNA matrix model for orientable chord diagrams

Let  $c_{g,b}(k)$  denote the number of connected chord diagrams with genus g, b backbones, and k chords.<sup>3</sup> We consider the following generating function

(1.1) 
$$C_{g,b}(w) = \sum_{k=0}^{\infty} c_{g,b}(k) w^k.$$

In [17], the number  $c_{g,1}(k)$  of chord diagrams with 1 backbone was studied. In particular for the class of planar graphs which have genus g = 0, the number  $c_{0,1}(k)$  is shown to be equal to the Catalan number

$$C_k = \frac{(2k)!}{(k+1)!k!}.$$

We present explicitly the tailed ribbon surfaces with 1 backbone for k = 1, 2, 3, 4in Figure 3.



Figure 3: The planar ribbon surfaces with tails for k = 1, 2, 3, 4, whose counts agree with the Catalan numbers 1, 2, 5, 14.

In [9] the following theorem was established.

**Theorem 1.1** (RNA matrix model for orientable chord diagrams [9]). Let  $\mathcal{H}_N$  be the space of rank N Hermitian matrices. We consider the matrix integral with

<sup>&</sup>lt;sup>3</sup>Harer and Zagier found a remarkable formula for  $c_{g,1}(k)$ , referred to as the *Harer-Zagier formula*, in their computation of the virtual Euler characteristic of Riemann moduli space for punctured surfaces [58].

the potential

288

(1.2) 
$$V_{\rm RNA}(x) = \frac{x^2}{2} - \frac{stx}{1 - tx},$$

defined by

(1.3) 
$$Z_N(s,t) = \frac{1}{\operatorname{Vol}_N} \int_{\mathcal{H}_N} dM \ \mathrm{e}^{-\frac{1}{\hbar} \operatorname{Tr} V_{\mathrm{RNA}}(M)}$$

where

(1.4) 
$$\operatorname{Vol}_{N} = \int_{\mathcal{H}_{N}} dM \, \mathrm{e}^{-N \operatorname{Tr} \frac{M^{2}}{2}} = N^{N(N+1)/2} \operatorname{Vol}(\mathcal{H}_{N}).$$

Under the 't Hooft limit

(1.5)  $\hbar \to 0, \quad N \to \infty, \quad \mu = \hbar N,$ 

with the 't Hooft parameter  $\mu$  kept finite, the logarithm of the above matrix integral (called the free energy)  $F_N(s,t) = \log Z_N(s,t)$  has an asymptotic expansion

(1.6) 
$$F_N(s,t) = \sum_{g=0}^{\infty} \hbar^{2g-2} F_g(s,t).$$

Moreover, this free energy encodes generating functions (1.1) for the numbers  $c_{g,b}(k)$  of chord diagrams

(1.7) 
$$\mu^{2g-2}F_g(s,t) = \sum_{b=1}^{\infty} \frac{s^b}{\mu^b b!} C_{g,b}(\mu t^2) - \frac{s}{\mu} \delta_{g,0}.$$

By (1.7) we see that  $F_g(s, t)$  enumerates orientable chord diagrams with genus g, and s and  $t^2$  are generating parameters respectively for the number of backbones and chords.

## 1.2 RNA matrix model for non-oriented chord diagrams

As a natural generalization of the above enumerative problem, we consider the non-oriented<sup>4</sup> analogue of the enumeration of chord diagrams. A non-oriented chord diagram c is a chord diagram, where a binary quantity, twisted or untwisted, is assigned to each chord. There is a natural non-oriented fatgraph structure associated with a non-oriented chord diagram and the corresponding non-oriented surface  $\Sigma_c$ , where the binary quantity assigned to each chord determines if the band along the chord for the associated ribbon surface is twisted or not, as depicted in Figure 4.



Figure 4: A band and a twisted band.



Figure 5: Non-oriented ribbon surfaces with 1 backbone for k = 1, 2.

The non-oriented analogue of the ribbon surface is again constructed by collapsing the fattened backbones into polyvalent fattened vertices and, as before, adding a tail for each fattened vertex. Some examples of non-oriented ribbon surfaces are depicted in Figure 5 and in Appendix A.

Instead of the genus, for a surface  $\Sigma_c$  the cross-cap number (or the non-oriented genus) h is well-defined, and the Euler characteristic is given by

$$\chi = 2 - h = b - k + n.$$

Let  $c_{h,b}^{r}(k)$  denote the number of non-oriented chord diagrams with the crosscap number h, b backbones, and k chords. In analogy to the oriented case, we introduce the generating function  $C_{h,b}^{r}(w)$ 

(1.8) 
$$C_{h,b}^{\mathsf{r}}(w) = \sum_{k=0}^{\infty} c_{h,b}^{\mathsf{r}}(k) w^{k}$$

As proven in [4], the non-oriented analogue of Theorem 1.1 is straightforwardly obtained by replacing the integration over Hermitian matrices in (1.3) with the integration over real symmetric matrices.

**Theorem 1.2** (RNA matrix model for non-oriented chord diagrams). Let  $\mathcal{H}_N(\mathbb{R})$  be the space of rank N real symmetric matrices. We consider the real symmetric matrix integral with the potential (1.2) defined by

(1.9) 
$$Z_N^{\mathsf{r}}(s,t) = \frac{1}{\operatorname{Vol}_N(\mathbb{R})} \int_{\mathcal{H}_N(\mathbb{R})} dM \ \mathrm{e}^{-\frac{1}{2\hbar}\operatorname{Tr} V_{\mathrm{RNA}}(M)},$$

<sup>&</sup>lt;sup>4</sup>Non-oriented is a shorthand for the union of orientable and non-orientable.

where

(1.10) 
$$\operatorname{Vol}_{N}(\mathbb{R}) = \int_{\mathcal{H}_{N}(\mathbb{R})} dM \, \mathrm{e}^{-N\operatorname{Tr}\frac{M^{2}}{4}} = N^{N(N+1)/2} \operatorname{Vol}(\mathcal{H}_{N}(\mathbb{R}))$$

Under the 't Hooft limit

(1.11) 
$$\hbar \to 0, \quad N \to \infty, \quad \mu = \hbar N,$$

with the 't Hooft parameter  $\mu$  kept finite and fixed, the free energy  $F_N^r(s,t) = \log Z_N^r(s,t)$  has an asymptotic expansion

(1.12) 
$$F_N^{\mathsf{r}}(s,t) = \frac{1}{2} \sum_{h=0}^{\infty} \hbar^{h-2} F_h^{\mathsf{r}}(s,t)$$

This free energy encodes generating functions (1.8) for the numbers  $c_{h,b}^{r}(k)$  of non-oriented chord diagrams

(1.13) 
$$\mu^{h-2}F_h^r(s,t) = \sum_{b=1}^{\infty} \frac{s^b}{\mu^b b!} C_{h,b}^r(\mu t^2) - \frac{s}{\mu} \delta_{g,0}.$$

Note that matrix model techniques for the enumeration of non-oriented chord diagrams are also considered in [55, 84, 83, 60, 73, 54].

### **1.3** $\beta$ -deformed RNA matrix model as a unified model

In the context of matrix models, it is known that their  $\beta$ -deformation implements the enumeration of non-oriented chord diagrams [70], as we shall now recall<sup>5</sup>.

**Definition 1.3** ( $\beta$ -deformed RNA matrix model). The  $\beta$ -deformed eigenvalue integral with the potential (1.2) is defined by

(1.14) 
$$Z_N^\beta(s,t) = \frac{1}{\operatorname{Vol}_N^\beta} \int_{\mathbb{R}^N} \prod_{a=1}^N dz_a \Delta(z)^{2\beta} \mathrm{e}^{-\frac{\sqrt{\beta}}{\hbar} \sum_{a=1}^N V_{\mathrm{RNA}}(z_a)},$$

where  $\Delta(z)$  denotes the Vandermonde determinant

$$\Delta(z) = \prod_{a < b} (z_a - z_b),$$

and

(1.15) 
$$\operatorname{Vol}_{N}^{\beta} = \int_{\mathbb{R}^{N}} \prod_{a=1}^{N} dz_{a} \Delta(z)^{2\beta} \mathrm{e}^{-\frac{\sqrt{\beta}}{2\hbar} \sum_{a=1}^{N} z_{a}^{2}}.$$

290

<sup>&</sup>lt;sup>5</sup>The  $\beta$ -deformed Dyson's model is solved in various ways. See e.g. [61, 62, 63].

#### Enumeration of Chord Diagrams

In the cases of  $\beta = 1$  and  $\beta = 1/2$ , the  $\beta$ -deformed eigenvalue integral (1.14) reduces to the eigenvalue representation of the Hermitian matrix integral (1.3) and the real symmetric matrix integral (1.9) upon the redefinition  $\hbar \to \sqrt{2}\hbar$ , respectively. Here  $z_a \in \mathbb{R}$   $(a = 1, \dots, N)$  correspond to the eigenvalues of the matrix M in each matrix integral. The other special case is  $\beta = 2$ , for which the eigenvalue integral (1.14) represents the quaternionic matrix integral. Under the 't Hooft limit

(1.16) 
$$\hbar \to 0, \quad N \to \infty, \quad \mu = \beta^{1/2} \hbar N,$$

with the fixed 't Hooft parameter  $\mu$ , the free energy  $F_N^\beta(s,t) = \log Z_N^\beta(s,t)$  has the asymptotic expansion

(1.17) 
$$F_N^{\beta}(s,t) = \sum_{g,\ell=0}^{\infty} \hbar^{2g-2+\ell} \gamma^{\ell} F_{g,\ell}(s,t),$$

where

$$\gamma = \beta^{1/2} - \beta^{-1/2}.$$

The free energies (1.6) for  $\beta = 1$  and (1.12) for  $\beta = 1/2$  satisfy

(1.18) 
$$F_{g}(s,t) = F_{g,0}(s,t),$$
$$F_{h}^{\mathsf{r}}(s,t) = \sum_{\substack{g,\ell=0\\2g+\ell=h}}^{\infty} 2^{g}(-1)^{\ell} F_{g,\ell}(s,t).$$

Combining (1.7) in Theorem 1.1 for  $\beta = 1$  and (1.13) in Theorem 1.2 for  $\beta = 1/2$ , we obtain the following proposition.

**Proposition 1.4.** Let  $\widetilde{C}_{g,\ell,b}(w)$  be defined by

(1.19) 
$$(-\mu)^{2g-2+\ell}F_{g,\ell}(s,t) = \sum_{b=1}^{\infty} \frac{s^b}{\mu^b b!} \widetilde{C}_{g,\ell,b}(\mu t^2) - \frac{s}{\mu} \delta_{g,0} \delta_{\ell,0}.$$

Then we have the following relations

(1.20) 
$$C_{g,b}(w) = \widetilde{C}_{g,0,b}(w),$$
$$C_{h,b}^{\mathsf{r}}(w) = \sum_{\substack{g,\ell=0\\2g+\ell=h}}^{\infty} 2^{g} \widetilde{C}_{g,\ell,b}(w).$$

In the following sections, using  $\beta$ -deformed topological recursion and the theory of quantum curves, we will develop the computational techniques of determining the functions  $F_{g,\ell}$ .

## 1.4 $\beta$ -deformed topological recursion and Gaussian resolvents

In our first approach, we will employ an analytical method, referred to as the  $\beta$ -deformed topological recursion [33, 28, 24, 68, 67].  $\beta$ -deformed topological recursion is a powerful machinery for analyzing the asymptotic expansion of a matrix model free energy in the  $N \to \infty$  limit. Let  $Z_N^\beta(\{r_n\})$  and  $F_N^\beta(\{r_n\}) = \log Z_N^\beta(\{r_n\})$  denote the partition function and the free energy of the  $\beta$ -deformed matrix model with a general potential

(1.21) 
$$V(x) = \sum_{n=0}^{K} r_n x^n.$$

By the asymptotic expansion of the free energy  $F_N^\beta(\{r_n\})$  we mean the following expansion in the 't Hooft limit (1.16)

(1.22) 
$$F_N^{\beta}(\{r_n\}) = \sum_{g,\ell=0}^{\infty} \hbar^{2g-2+\ell} \gamma^{\ell} F_{g,\ell}(\{r_n\}).$$

The  $\beta$ -deformed topological recursion in fact can be formulated more generally, as a tool that assigns a series of multi-linear differentials to a given algebraic curve. In the context of matrix models these multi-linear differentials are identified with various matrix model correlators (and the topological recursion represents Ward identities between these correlators), while the underlying algebraic curve is identified with a *spectral curve* C (of the matrix model with  $\beta = 1$ )

(1.23) 
$$\mathcal{C} = \left\{ \begin{array}{l} (x,\omega) \in \mathbb{C}^2 \mid H(x,\omega) = \omega^2 - 2V'(x)\omega - f(x) = 0 \\ \\ f(x) = \lim_{\substack{N \to \infty, \hbar \to 0 \\ \beta \to 1}} -4\beta^{1/2}\hbar \left\langle \sum_{a=1}^N \frac{V'(x) - V'(z_a)}{x - z_a} \right\rangle_N^\beta, \end{array} \right.$$

where  $\langle \mathcal{O}(\{z_a\}) \rangle_N^{\beta}$  denotes the  $\beta$ -deformed eigenvalue integral

(1.24) 
$$\langle \mathcal{O}(\{z_a\})\rangle_N^\beta = \frac{1}{Z_N^\beta(\{r_n\})\operatorname{Vol}_N^\beta} \int_{\mathbb{R}^N} \prod_{a=1}^N dz_a \Delta(z)^{2\beta} \mathcal{O}(\{z_a\}) \mathrm{e}^{-\frac{\sqrt{\beta}}{\hbar}\sum_{a=1}^N V(z_a)}.$$

In particular for the RNA matrix model presented in Definition 1.3, the spectral curve takes the following form [9].

**Example 1.5.** For the potential  $V_{\text{RNA}}(x)$  given in (1.2), the defining equation for C in (1.23) takes form

(1.25) 
$$y_{\text{RNA}}^2 = M_{\text{RNA}}(x)^2(x-a)(x-b), \quad y_{\text{RNA}} = V_{\text{RNA}}'(x) - \omega,$$

where the branch points a < b are solutions of the equations

$$S = \frac{D^2 - 4\mu}{t(3D^2 - 4\mu)},$$
  
$$4s^2t^4(3D^2 - 4\mu)^6 = D^2(D^2 - 4\mu)^2 (4D^2 - t^2(3D^2 - 4\mu)^2)^3,$$

and  $\sigma$  and  $\delta$  are defined as

$$\sigma = St = \frac{a+b}{2}t, \quad \delta = Dt = \frac{a-b}{2}t$$

The function  $M_{\rm RNA}(x)$  is given by

$$M_{\rm RNA}(x) = \frac{(2tx - 2 + \sigma)^2 + \eta}{8(tx - 1)^2}, \quad \eta := \frac{\sigma(4 - 4\delta^2 - 7\sigma + 3\sigma^2)}{\sigma - 1}$$

The multi-linear differentials computed by the  $\beta$ -deformed topological recursion are defined as follows [33].

**Definition 1.6.** The connected *h*-point symmetric multi-linear differential  $W_h \in \mathcal{M}^1(\mathcal{C}^{\times h})^s$  is defined as

(1.26) 
$$W_h(x_1, \dots, x_h) = \beta^{h/2} \left\langle \prod_{i=1}^h \sum_{a=1}^N \frac{dx_i}{x_i - z_a} \right\rangle_{N,\beta}^{(c)},$$

where  $\langle \mathcal{O} \rangle_{N,\beta}^{(c)}$  denotes the connected part [46] of  $\langle \mathcal{O} \rangle_{N}^{\beta}$  introduced in (1.24). In the 't Hooft limit (1.16), *h*-point multi-linear differentials admit an asymptotic expansion

(1.27) 
$$W_h(x_1, \dots, x_h) = \sum_{g,\ell=0}^{\infty} \hbar^{2g-2+h+\ell} \gamma^\ell W_\ell^{(g,h)}(x_1, \dots, x_h).$$

For the class of genus 0 spectral curves of the form

(1.28) 
$$y(x)^2 = M(x)^2 \sigma(x),$$
$$\sigma(x) = (x-a)(x-b), \quad M(x) = c \prod_{i=1}^f (x-\alpha_i)^{m_i},$$

the 2-point multi-linear differential  $W_0^{(0,2)}(x_1,x_2)$  takes form

(1.29) 
$$W_0^{(0,2)}(x_1, x_2) = B(x_1, x_2) - \frac{dx_1 dx_2}{(x_1 - x_2)^2}$$

where the Bergman kernel  $B(x_1, x_2)$  is a bilinear differential

(1.30) 
$$B(x_1, x_2) = \frac{dx_1 dx_2}{2(x_1 - x_2)^2} \left[ 1 + \frac{x_1 x_2 - \frac{1}{2}(a+b)(x_1 + x_2) + ab}{\sqrt{\sigma(x_1)\sigma(x_2)}} \right].$$

An exact formula for  $W_1^{(0,1)}(x)$  is also known [32, 33, 28],

(1.31) 
$$W_1^{(0,1)}(x) = -\frac{dy(x)}{2y(x)} + \frac{dx}{2\sqrt{\sigma(x)}} \left[ 1 + \sum_{i=1}^f m_i \left( 1 + \frac{\sqrt{\sigma(\alpha_i)}}{x - \alpha_i} \right) \right].$$

All other differentials  $W_{\ell}^{(g,h)}(x_1,\ldots,x_h)$  in the asymptotic expansion (1.27) can be determined recursively by means of the topological recursion, as stated in the theorem below.

**Theorem 1.7** ([33, 31, 28]). The differentials  $W_{\ell}^{(g,h)}(x_H)$  for  $(g, h, \ell) \neq (0, 1, 0)$ , (0, 2, 0), (0, 1, 1) in the asymptotic expansion (1.27) obey the  $\beta$ -deformed topological recursion

$$\mathcal{W}_{0}^{(0,1)}(x) = 0, \quad \mathcal{W}_{0}^{(0,2)}(x_{1}, x_{2}) = W_{0}^{(0,2)}(x_{1}, x_{2}) + \frac{dx_{1}dx_{2}}{2(x_{1} - x_{2})^{2}},$$

$$\mathcal{W}_{\ell}^{(g,h)}(x_{H}) = W_{\ell}^{(g,h)}(x_{H}) \text{ for } (g,h,\ell) \neq (0,1,0), \quad (0,2,0),$$

$$W_{\ell}^{(g,h+1)}(x,x_{H}) = \oint_{\mathcal{A}} \frac{1}{2\pi i} \frac{dS(x,z)}{y(z)dz} \bigg[ W_{\ell}^{(g-1,h+2)}(z,z,x_{H}) + \sum_{k=0}^{g} \sum_{n=0}^{\ell} \sum_{\emptyset=J \subseteq H} \mathcal{W}_{\ell-n}^{(g-k,|J|+1)}(z,x_{J}) \mathcal{W}_{n}^{(k,|H|-|J|+1)}(z,x_{H\setminus J}) \bigg]$$

$$(32)$$

(1.32)

$$+ dz^2 \frac{\partial}{\partial z} \frac{W_{\ell-1}^{(g,h+1)}(z,x_H)}{dz} \bigg],$$

where  $H = \{1, 2, ..., h\} \supset J = \{i_1, i_2, ..., i_j\}, H \setminus J = \{i_{j+1}, i_{j+2}, ..., i_h\}, and \mathcal{A}$ is the counterclockwise contour around the branch cut [a, b]. Here  $dS(x_1, x_2)$  is the third type differential, which for the genus 0 spectral curve (1.28) takes form

(1.33) 
$$dS(x_1, x_2) = \frac{\sqrt{\sigma(x_2)}}{\sqrt{\sigma(x_1)}} \frac{dx_1}{x_1 - x_2}.$$

Our task in what follows is to determine generating functions of chord diagrams encoded in the free energy of the RNA matrix model (1.19). To this end we take advantage of the following trick [9]. The potential of the RNA matrix model  $V_{\text{RNA}}(x)$  can be separated into the Gaussian part  $V_{\text{G}}(x)$  and the rational part  $V_{\text{rat}}(x)$ 

(1.34) 
$$V_{\text{RNA}}(x) = V_{\text{G}}(x) + V_{\text{rat}}(x) + s$$
,  $V_{\text{G}}(x) = \frac{1}{2}x^2$ ,  $V_{\text{rat}}(x) = -\frac{t^{-1}s}{t^{-1}-x}$ .

Adopting this separation into the definition of  $Z_N^{\beta}(s,t)$  in (1.14), we find that the partition function of the RNA matrix model can be re-expressed as an expectation

value in the  $\beta$ -deformed Gaussian model

$$Z_{N}^{\beta}(s,t) = \mathrm{e}^{-\frac{\sqrt{\beta}}{\hbar}sN} \left\langle \exp\left[\frac{s\sqrt{\beta}}{t\hbar}\sum_{a=1}^{N}\frac{1}{t^{-1}-z_{a}}\right] \right\rangle_{N,\beta}^{\mathrm{G}},$$

where  $\langle \mathcal{O}(\{z_a\}) \rangle_{N,\beta}^{\mathrm{G}}$  denotes  $\langle \mathcal{O}(\{z_a\}) \rangle_N^{\beta}$  defined in (1.24) with the Gaussian potential  $V(x) = V_{\mathrm{G}}(x)$ . It follows that the right hand side of this equation is given by a sum of *h*-point multi-linear differentials with  $x_i = t^{-1}$   $(i = 1, \ldots, h)$  in the Gaussian model.

**Proposition 1.8.** Let  $\omega_b(x_1, \ldots, x_b)$  denote the connected b-resolvent in the  $\beta$ -deformed Gaussian model, which we also refer to as the Gaussian b-resolvent,

(1.35) 
$$\omega_b(x_1, \dots, x_b) = \frac{W_h(x_1, \dots, x_b)}{dx_1 \cdots dx_b} = \beta^{b/2} \left\langle \prod_{i=1}^b \sum_{a=1}^N \frac{1}{x_i - z_a} \right\rangle_{N,\beta}^{(c)}$$

Then the free energy  $F_N^\beta(s,t)$  of the  $\beta$ -deformed RNA matrix model is given by

(1.36) 
$$F_N^\beta(s,t) = -\frac{\sqrt{\beta}}{\hbar}sN + \sum_{b=1}^\infty \frac{s^b}{b!t^b\hbar^b}\omega_b(t^{-1},\dots,t^{-1}).$$

Combining the above relation with the form of the free energy in (1.19), we obtain the key theorem for our enumeration of chord diagrams.

**Theorem 1.9.** The coefficients  $\widetilde{C}_{g,\ell,b}(w)$  in (1.19) are obtained directly from the principal specialization of the Gaussian b-resolvents

(1.37) 
$$\omega_b(x,\ldots,x) = \frac{1}{x^b} \sum_{g=0,\ell=0}^{\infty} (\mu^{-1}\hbar)^{2g-2+b+\ell} (-\gamma)^\ell \widetilde{C}_{g,\ell,b}(\mu x^{-2}).$$

On the basis of this theorem, one can compute the generating function  $\widetilde{C}_{g,\ell,b}(w)$  of the numbers of non-oriented chord diagrams using the  $\beta$ -deformed topological recursion for the Gaussian model. We will present some details of such a computation in Section 2, and compare its results with other methods of enumeration.

Finally, in order to find the complete form of the asymptotic expansion, we also need to consider the *unstable* part of the free energy, which is not determined by the topological recursion and must be computed independently. The unstable part consists of four terms:  $F_{0,0}(s,t)$ ,  $F_{0,1}(s,t)$ ,  $F_{0,2}(s,t)$ , and  $F_{1,0}(s,t)$ . The two terms  $F_{0,0}(s,t)$  and  $F_{1,0}(s,t)$  are the same as the terms  $F_0(s,t)$  and  $F_{1,0}(s,t)$  in the asymptotic expansion (1.6) of the Hermitian matrix model. On the other hand,  $F_{0,1}(\{r_n\})$  and  $F_{0,2}(\{r_n\})$  can be computed using famous formulae, referred to respectively as the Dyson's formula and Wiegmann-Zabrodin formula, which for the genus 0 spectral curve (1.28) take the following form.

**Theorem 1.10.** For the class of genus 0 spectral curves (1.28) determined by the general potential (1.21), the unstable parts  $F_{0,1}(\{r_n\})$  and  $F_{0,2}(\{r_n\})$  of the free energy are given by

(1.38)

296

$$\begin{aligned} \frac{\partial}{\partial \mu} F_{0,1}(\{r_n\}) &= 1 + \log|c| + \frac{1}{2} \log\left(\frac{a-b}{4}\right)^2 + \sum_{i=1}^f m_i \log\left[\frac{1}{2}\left(\alpha_i - \frac{a+b}{2} + \sqrt{\sigma(\alpha_i)}\right)\right], \\ F_{0,2}(\{r_n\}) &= -\frac{1}{2} \sum_{i=1}^f m_i \log\left(1 - \mathfrak{s}_i^2\right) - \frac{1}{2} \sum_{i,j=1}^f m_i m_j \log\left(1 - \mathfrak{s}_i \mathfrak{s}_j\right) \\ (1.39) &+ \frac{1}{24} \log\left|M(a)M(b)(a-b)^4\right|, \end{aligned}$$

where  $\mathfrak{s}_i$ ,  $i = 1, \ldots, f$  are defined as

$$\alpha_i(\mathfrak{s}_i) = \frac{a+b}{2} - \frac{a-b}{4}(\mathfrak{s}_i + \mathfrak{s}_i^{-1}), \quad |\mathfrak{s}_i| < 1.$$

The formula (1.38) for  $F_{0,1}(\{r_n\})$  was found in [33, 31, 28]. On the other hand, the formula (1.39) for  $F_{0,2}(\{r_n\})$  is proven in appendix B.

#### 1.5 A recursion relation from the quantum curve

The main object that we consider in our second approach is the *wave-function* for the matrix model. The wave-function is the 1-point function defined as follows.

**Definition 1.11.** Let  $\epsilon_{1,2}$  denote the parameters

(1.40) 
$$\epsilon_1 = -\beta^{1/2} g_s, \quad \epsilon_2 = \beta^{-1/2} g_s, \quad g_s = 2\hbar$$

For the general type potential (1.21), the 1-point function  $Z_{\alpha}(x; \{r_n\})$  ( $\alpha = 1, 2$ )

(1.41) 
$$Z_{\alpha}(x; \{r_n\}) = \frac{1}{\operatorname{Vol}_N^{\beta}} \int_{\mathbb{R}^N} \prod_{a=1}^N dz_a \Delta(z)^{2\beta} \psi_{\alpha}(x) \mathrm{e}^{-\frac{\sqrt{\beta}}{h} \sum_{a=1}^N V(z_a)},$$

with the eigenvalue operator

$$\psi_{\alpha}(x) = \prod_{a=1}^{N} (x - z_a)^{\frac{\epsilon_1}{\epsilon_{\alpha}}},$$

is referred to as the wave-function. For the RNA matrix model we denote the wave-function by  $Z_{\alpha}(x; s, t)$ .

A prominent property of the wave-function [2, 1, 67, 34] is that it satisfies a partial differential equation, which is called the *quantum curve*<sup>6</sup> or the *timedependent Schrödinger equation*.

**Proposition 1.12** ([1, 67]). The wave-function  $Z_{\alpha}(x; \{r_n\})$  satisfies the partial differential equation

(1.42) 
$$\left[-\epsilon_{\alpha}^{2}\frac{\partial^{2}}{\partial x^{2}}-2\epsilon_{\alpha}V'(x)\frac{\partial}{\partial x}+\hat{f}(x)\right]Z_{\alpha}(x;\{r_{n}\})=0,$$

where  $\hat{f}(x)$  is the differential operator

$$\hat{f}(x) = g_s^2 \sum_{n=0}^K x^n \partial_{(n)}, \quad \partial_{(n)} = \sum_{k=n+2}^{K+2} k r_k \frac{\partial}{\partial r_{k-n-2}},$$

and we denoted  $\partial/\partial r_0 = -N/(2\epsilon_2)$ .

As its name suggests, the partial differential equation (1.42) is interpreted as the quantization of the spectral curve C of the matrix model. Promoting the parameters  $(x, \omega)$  in (1.23) to the non-commutative operators  $(\hat{x}, \hat{p})$ , such that

$$\hat{p} = \epsilon_{\alpha} \frac{\partial}{\partial x}, \qquad \hat{x} = x, \qquad [\hat{p}, \hat{x}] = \epsilon_{\alpha},$$

the partial differential equation (1.42) can be written in the form

$$\widehat{A}Z_{\alpha}(x; \{r_n\}) = 0, \qquad \widehat{A} \equiv A(\hat{x}, \hat{p}),$$

for an appropriate choice of  $A(\hat{x}, \hat{p})$ . The operator  $\hat{A}$  is interpreted as a quantization of the spectral curve C, and the equation (1.42) reduces to the defining equation of the spectral curve in the classical limit [1, 67].

Note that, more generally, to a given spectral curve one may associate an infinite family of wave-functions and corresponding quantum curves, which take form of Virasoro singular vectors [67, 34]. In particular the quantum curves in (1.42), for both values of  $\alpha = 1, 2$ , correspond to singular vectors at level 2. In this work we do not consider quantum curves at levels higher than 2.

In what follows we use the partial differential equation (1.42) as a tool to determine the partition function  $Z_N^{\beta}(s,t)$  of the  $\beta$ -deformed RNA matrix model. The point is that the partition function  $Z_N^{\beta}(\{r_n\})$  of the matrix model with the

<sup>&</sup>lt;sup>6</sup>The name "quantum curve" is also used for the ordinary differential equation for the wave function [36, 35, 59, 57, 88, 72, 71, 66, 41, 44, 39, 82, 37, 42, 74, 56, 43, 38, 26, 40, 25]. In this article, we also use this name for a partial differential equation which arises from the conformal field theoretical description of the matrix model [65].

general type potential (1.21) appears as the leading order term in the expansion of the wave-function  $Z_{\alpha}(x; \{r_n\})$  around  $x = \infty$ 

$$Z_{\alpha}(x; \{r_n\}) = x^{\epsilon_1 N/\epsilon_{\alpha}} \left( Z_N^{\beta}(\{r_n\}) + \mathcal{O}(x^{-1}) \right).$$

Therefore, from the expansion of the wave-function with  $\alpha=2$  for the RNA matrix model

$$Z_{\alpha=2}(x; s, t) = \exp\left[S(x, s, t)\right],$$
  
$$S(x, s, t) = -\frac{2\mu}{\epsilon_2}\log x + \sum_{b=0}^{\infty}\sum_{p=0}^{\infty}S_{b,p}(t)s^bx^{-p},$$

the free energy  $F_N^\beta(s,t)$  can be written as

$$F_N^{\beta}(s,t) = \sum_{b=0}^{\infty} S_{b,p=0}(t) s^b.$$

We use the partial differential equation (1.42) to determine each  $S_{b,p}(t)$ . The main result of this approach is summarized in the following theorem.

**Theorem 1.13.** The partial differential equation for the phase function S(x, s, t) for the RNA matrix model takes form

$$\epsilon_2^2 \left[ \frac{\partial^2}{\partial x^2} S(x,s,t) + \left( \frac{\partial}{\partial x} S(x,s,t) \right)^2 \right] + 2\epsilon_2 \left( x - \frac{st}{(1-tx)^2} \right) \frac{\partial}{\partial x} S(x,s,t)$$

$$(1.43)$$

$$+ 4\mu \left( 1 - \frac{st^2(2-tx)}{(1-tx)^2} \right) + \frac{\epsilon_1 \epsilon_2 st^2(2-tx)}{(1-tx)^2} \frac{\partial}{\partial s} S(x,s,t) + \frac{\epsilon_1 \epsilon_2 t^3}{1-tx} \frac{\partial}{\partial t} S(x,s,t) = 0.$$

This equation generates a hierarchy of differential equations for the coefficients  $S_{b,p}(t)$  of the phase function, and the phase function is determined recursively with respect to the backbone number b.

As an independent verification of this algorithm, we checked iterative computations for b = 1, 2, 3, 4, 5 up to  $\mathcal{O}(t^{12})$ , and confirmed that they agree with the results of the  $\beta$ -deformed topological recursion from the first approach.

## 2 Enumeration of chord diagrams via the topological recursion

In the introduction, we extended the RNA matrix model proposed in [9] to the  $\beta$ deformed RNA matrix model given by (1.14), that enumerates both orientable and non-orientable chord diagrams via Proposition 1.4. In this section we enumerate orientable and non-orientable chord diagrams by the formalism of the  $\beta$ -deformed topological recursion (1.32).

## 2.1 Enumeration of chord diagrams via Gaussian resolvents

In principle, by applying the  $\beta$ -deformed topological recursion formalism to the spectral curve (1.25) of the RNA matrix model, one can recursively compute the asymptotic expansion (1.17) of the free energy  $F_N^\beta(s,t)$ . In [9], using the topological recursion [5, 45, 46] for the  $\beta = 1$  (which does not involve  $W_{\ell\geq 1}^{(g,h)}(x_H)$  terms) RNA matrix model (1.3),  $F_2(s,t) = F_{2,0}(s,t)$  and  $F_3(s,t) = F_{3,0}(s,t)$  were explicitly computed. However, because of the complicated form of the curve (1.25), an explicit computation of  $W_{\ell\geq 1}^{(g,h)}(x_H)$  is not easy. In this section we consider instead the  $\beta$ -deformed Gaussian matrix model with the Gaussian potential  $V_G(x)$  in (1.34).

Using Theorem 1.9, one can compute the coefficients  $\widetilde{C}_{g,\ell,b}(w)$  in (1.19) of the free energy  $F_N^{\beta}(s,t)$  from the Gaussian *b*-resolvents (1.35). For the Gaussian matrix model, the spectral curve takes form

(2.1) 
$$y_{\rm G}(x)^2 = x^2 - 4\mu.$$

In order to apply the  $\beta$ -deformed topological recursion (1.32) to this Gaussian curve, it is convenient to introduce the Zhukovsky variable z as

$$x(\mathbf{z}) = \sqrt{\mu}(\mathbf{z} + \mathbf{z}^{-1}).$$

In this variable the branch points  $x = \pm 2\sqrt{\mu}$  are mapped to  $z = \pm 1$ . To express the  $\beta$ -deformed topological recursion (1.32) in this Zhukovsky variable, we define

(2.2)  

$$\begin{aligned} \widehat{y}_{G}(z)dz &= y_{G}(x(z))dx = \sqrt{\mu}(z-z^{-1})dz, \\ d\widehat{S}(z_{1},z_{2}) &= dS(x_{1}(z_{1}),x_{2}(z_{2})) = \frac{dz_{1}}{z_{1}-z_{2}} - \frac{dz_{1}}{z_{1}-z_{2}^{-1}}, \\ \widehat{W}_{\ell}^{(g,h)}(z_{1},\ldots,z_{h}) &= W_{\ell}^{(g,h)}(x_{1}(z_{1}),\ldots,x_{h}(z_{h})). \end{aligned}$$

We then obtain the  $\beta$ -deformed topological recursion (1.32) for the Gaussian spectral curve in the Zhukovsky variable. More generally, the  $\beta$ -deformed topological recursion for genus 0 spectral curves (1.28) in the Zhukovsky is discussed in detail in [68, 67].

**Corollary 2.1.** The differentials  $\widehat{W}_{\ell}^{(g,h)}(\mathbf{z}_H)$  for  $(g,h,\ell) \neq (0,1,0)$ , (0,2,0), (0,1,1) obey the  $\beta$ -deformed topological recursion in the Zhukovsky variable

(2.3) 
$$\widehat{W}_{\ell}^{(g,h+1)}(\mathbf{z},\mathbf{z}_{H}) = \oint_{\widetilde{\mathcal{A}}} \frac{1}{2\pi i} \frac{d\widehat{S}(\mathbf{z},\zeta)}{\widehat{y}_{\mathrm{G}}(\zeta)d\zeta} \operatorname{Rec}_{\ell}^{(g,h+1)}(\zeta,\mathbf{z}_{H}).$$

Here

$$\operatorname{Rec}_{\ell}^{(g,h+1)}(\zeta,\mathbf{z}_H) = \widehat{W}_{\ell}^{(g-1,h+2)}(\zeta,\zeta,\mathbf{z}_H)$$

J. E. Andersen, H. Fuji, M. Manabe, R. C. Penner, and P. Sułkowski

(2.4) 
$$+ \sum_{k=0}^{g} \sum_{n=0}^{\ell} \sum_{\emptyset=J \subseteq H} \widehat{\mathcal{W}}_{\ell-n}^{(g-k,|J|+1)}(\zeta, \mathbf{z}_J) \widehat{\mathcal{W}}_{n}^{(k,|H|-|J|+1)}(\zeta, \mathbf{z}_{H\setminus J}) + d\zeta^2 \left[ \frac{\partial}{\partial \zeta} + \frac{\partial^2 \zeta}{\partial w^2} \left( \frac{\partial w}{\partial \zeta} \right)^2 \right] \frac{\widehat{W}_{\ell-1}^{(g,h+1)}(\zeta, \mathbf{z}_H)}{d\zeta},$$

with  $w = \sqrt{\mu}(\zeta + \zeta^{-1}), \ |\zeta| > 1$ , and

$$\begin{aligned} \widehat{\mathcal{W}}_{0}^{(0,1)}(\mathbf{z}) &= 0, \quad \widehat{\mathcal{W}}_{0}^{(0,2)}(\mathbf{z}_{1},\mathbf{z}_{2}) = \widehat{W}_{0}^{(0,2)}(\mathbf{z}_{1},\mathbf{z}_{2}) + \frac{(\mathbf{z}_{1}^{2}-1)(\mathbf{z}_{2}^{2}-1)d\mathbf{z}_{1}d\mathbf{z}_{2}}{2(\mathbf{z}_{1}-\mathbf{z}_{2})^{2}(\mathbf{z}_{1}\mathbf{z}_{2}-1)^{2}}, \\ (2.5) \quad \widehat{\mathcal{W}}_{\ell}^{(g,h)}(\mathbf{z}_{H}) &= \widehat{W}_{\ell}^{(g,h)}(\mathbf{z}_{H}) \text{ for } (g,h,\ell) \neq (0,1,0), \ (0,2,0), \end{aligned}$$

where  $\widetilde{\mathcal{A}}$  is the contour surrounding the unit disk  $|\zeta| = 1$ . For the Gaussian model the initial data of the recursion (1.30) and (1.31) takes form

(2.6) 
$$\widehat{W}_0^{(0,2)}(\mathbf{z}_1, \mathbf{z}_2) = \frac{d\mathbf{z}_1 d\mathbf{z}_2}{(\mathbf{z}_1 \mathbf{z}_2 - 1)^2},$$

(2.7) 
$$\widehat{W}_1^{(0,1)}(z) = \left(\frac{1}{z} - \frac{1}{2(z-1)} - \frac{1}{2(z+1)}\right) dz.$$

From Theorem 1.9 the coefficients  $\widetilde{C}_{g,\ell,b}(w)$  of the free energy  $F_N^\beta(s,t)$  can now be computed. For example we find, for b = 1,

$$\begin{split} \widetilde{C}_{0,0,1}(w) &= \frac{1 - \sqrt{1 - 4w}}{2w}, \quad \widetilde{C}_{0,1,1}(w) = \frac{1 - \sqrt{1 - 4w}}{2(1 - 4w)}, \quad \widetilde{C}_{1,0,1}(w) = \frac{w^2}{(1 - 4w)^{5/2}}, \\ \widetilde{C}_{0,2,1}(w) &= \frac{w\left(1 + w - \sqrt{1 - 4w}\right)}{(1 - 4w)^{5/2}}, \quad \widetilde{C}_{1,1,1}(w) = \frac{w^2\left(1 + 30w - (1 + 6w)\sqrt{1 - 4w}\right)}{2(1 - 4w)^4}, \\ \widetilde{C}_{0,3,1}(w) &= \frac{5w^2\left(1 + 2w - (1 + w)\sqrt{1 - 4w}\right)}{(1 - 4w)^4}, \end{split}$$

for b = 2,

$$\begin{split} \widetilde{C}_{0,0,2}(w) &= \frac{w}{(1-4w)^2}, \quad \widetilde{C}_{0,1,2}(w) = \frac{w\left(1+18w-(1+4w)\sqrt{1-4w}\right)}{2(1-4w)^{7/2}}, \\ \widetilde{C}_{1,0,2}(w) &= \frac{w^3(21+20w)}{(1-4w)^5}, \quad \widetilde{C}_{0,2,2}(w) = \frac{w^2\left(8+98w+38w^2-(8+45w)\sqrt{1-4w}\right)}{(1-4w)^5}, \\ \widetilde{C}_{1,1,2}(w) &= \frac{w^3\left(21+1462w+2700w^2-(21+376w+240w^2)\sqrt{1-4w}\right)}{2(1-4w)^{13/2}}, \\ \widetilde{C}_{0,3,2}(w) &= \frac{w^3\left(117+1316w+1182w^2-(117+854w+292w^2)\sqrt{1-4w}\right)}{(1-4w)^{13/2}}, \end{split}$$

300

for 
$$b = 3$$
,  

$$\widetilde{C}_{0,0,3}(w) = \frac{2w^2(3+4w)}{(1-4w)^{9/2}},$$

$$\widetilde{C}_{0,1,3}(w) = \frac{w^2 \left(3 + 160w + 354w^2 - (3+50w+40w^2)\sqrt{1-4w}\right)}{(1-4w)^6},$$

$$\widetilde{C}_{1,0,3}(w) = \frac{12w^4(45+207w+68w^2)}{(1-4w)^{15/2}},$$

$$\widetilde{C}_{0,2,3}(w) = \frac{2w^3 \left(58 + 1797w + 5232w^2 + 1004w^3 - (58+977w+1416w^2)\sqrt{1-4w}\right)}{(1-4w)^{15/2}}$$

Then, by (1.20) in Proposition 1.4, we obtain the generating functions  $C_{g,b}(w)$  for orientable chord diagrams and  $C_{h,b}^{\mathsf{r}}(w)$  for non-oriented chord diagrams. For instance, we obtain

$$\begin{aligned} C_{0,1}(w) &= 1 + w + 2w^2 + 5w^3 + 14w^4 + 42w^5 + 132w^6 + 429w^7 + \mathcal{O}(w^8), \\ C_{1,1}(w) &= w^2 + 10w^3 + 70w^4 + 420w^5 + 2310w^6 + 12012w^7 + \mathcal{O}(w^8), \\ C_{0,2}(w) &= w + 8w^2 + 48w^3 + 256w^4 + 1280w^5 + 6144w^6 + \mathcal{O}(w^7), \\ C_{1,2}(w) &= 21w^3 + 440w^4 + 5440w^5 + 51840w^6 + 421120w^7 + \mathcal{O}(w^8), \\ C_{0,3}(w) &= 6w^2 + 116w^3 + 1332w^4 + 11880w^5 + 90948w^6 + \mathcal{O}(w^7), \\ C_{1,3}(w) &= 540w^4 + 18684w^5 + 350736w^6 + 4779720w^7 + \mathcal{O}(w^8), \end{aligned}$$

and

$$C_{1,1}^{\mathsf{r}}(w) = w + 5w^{2} + 22w^{3} + 93w^{4} + 386w^{5} + 1586w^{6} + 6476w^{7} + \mathcal{O}(w^{8}),$$

$$C_{2,1}^{\mathsf{r}}(w) = 5w^{2} + 52w^{3} + 374w^{4} + 2290w^{5} + 12798w^{6} + 67424w^{7} + \mathcal{O}(w^{8}),$$

$$C_{1,2}^{\mathsf{r}}(w) = 8w^{2} + 117w^{3} + 1084w^{4} + 8119w^{5} + 53640w^{6} + \mathcal{O}(w^{7}),$$

$$C_{2,2}^{\mathsf{r}}(w) = 111w^{3} + 2404w^{4} + 30442w^{5} + 295500w^{6} + \mathcal{O}(w^{7}),$$

$$C_{1,3}^{\mathsf{r}}(w) = 116w^{3} + 3204w^{4} + 49248w^{5} + 561782w^{6} + \mathcal{O}(w^{7}),$$

$$C_{2,3}^{\mathsf{r}}(w) = 2952w^{4} + 105300w^{5} + 2021396w^{6} + \mathcal{O}(w^{7}).$$

We note that the generating function  $C_{h,b}^{r}(w)$  with an even cross-cap number h = 2g enumerates both orientable and non-orientable chord diagrams, and therefore non-orientable chord diagrams are enumerated by

(2.10) 
$$C_{2a,b}^{\mathsf{r}}(w) - C_{g,b}(w).$$

## 2.2 The unstable part of the free energy

For a matrix model with the general potential (1.21), the unstable coefficients  $F_{0,0}(\{r_n\}), F_{0,1}(\{r_n\}), F_{1,0}(\{r_n\})$  and  $F_{0,2}(\{r_n\})$  in the asymptotic expansion (1.22)

of the free energy  $F_N^\beta(\{r_n\})$  must be computed separately. For the general potential (1.21) and the genus 0 spectral curve (1.28), the coefficients  $F_{0,0}(\{r_n\})$  [27, 69] and  $F_{1,0}(\{r_n\})$  [7, 6] (see [3, 30] for multi-cut solutions) are given by

(2.11) 
$$F_{0,0}(\{r_n\}) = -\mu \int_{[a,b]} dz \rho(z) V(z) + \mu^2 \int_{[a,b]^2} dz dz' \rho(z) \rho(z') \log |z - z'|,$$

(2.12) 
$$F_{1,0}(\{r_n\}) = -\frac{1}{24} \log \left| M(a) M(b) (a-b)^4 \right|$$

where  $\rho(z) = \lim_{N \to \infty} \frac{1}{N} \sum_{a=1}^{N} \delta(z - z_a)$  is the eigenvalue density given by

$$\rho(z) = \frac{1}{2\pi i \mu} \left( W_0^{(0,1)}(z - i\epsilon) - W_0^{(0,1)}(z + i\epsilon) \right) = \frac{1}{2\pi i \mu} y(z), \quad z \in [a,b].$$

In [9],  $F_{0,0}(s,t)$  and  $F_{1,0}(s,t)$  for the RNA matrix model (1.3) were computed using the above formulae.

Furthermore, the coefficients  $F_{0,1}(s,t)$  and  $F_{0,2}(s,t)$  for the genus 0 spectral curve (1.28) can be computed using Theorem 1.10. In particular, by Proposition 1.4 the generating function  $C_{2,b}^{\mathbf{r}}(w) - C_{1,b}(w)$  for the numbers of chord diagrams with the topology of the Klein bottle takes form

$$(2.13) \quad F_{1,0}(\{r_n\}) + F_{0,2}(\{r_n\}) = -\frac{1}{2} \sum_{i=1}^f m_i \log\left(1 - \mathfrak{s}_i^2\right) - \frac{1}{2} \sum_{i,j=1}^f m_i m_j \log\left(1 - \mathfrak{s}_i \mathfrak{s}_j\right).$$

Using the above formulae for the spectral curve (1.25) of the RNA matrix model, we determined the generating functions  $C_{0,b}(w)$ ,  $C_{1,b}(w)$ ,  $C_{1,b}^r(w)$ , and  $C_{2,b}^r(w)$ . We checked that the results coincide with the results obtained from the Gaussian *b*-resolvents discussed in the previous subsection. Compared with the method discussed in the previous subsection, the advantage of this method is that we find all order generating functions for the backbone number *b*.

## 3 Enumeration of chord diagrams via quantum curve techniques

In this section we consider a recursive computation of the numbers of chord diagrams, based on the quantum curve equation (1.42) for the wave-function of the RNA matrix model defined in Definition 1.11. For the  $\beta$ -deformed RNA matrix model the quantum curve equation reduces to a partial differential equation in three parameters x, s, and t. We solve this partial differential equation recursively and obtain the generating function for the numbers of chord diagrams as the leading term in the expansion of the wave-function near  $x \to \infty$ .

## 3.1 Differential equation for the wave-function from the quantum curve

The quantum curve for the  $\beta$ -deformed RNA matrix model is the key equation we take advantage of in this section.

**Proposition 3.1.** Let  $Z_{\alpha}(x;s,t)$  ( $\alpha = 1,2$ ) denote the wave-function for the  $\beta$ -deformed RNA matrix model

(3.1) 
$$Z_{\alpha}(x;s,t) = \frac{1}{\operatorname{Vol}_{N}^{\beta}} \int_{\mathbb{R}^{N}} \prod_{a=1}^{N} dz_{a} \Delta(z)^{2\beta} \psi_{\alpha}(x) \mathrm{e}^{-\frac{\sqrt{\beta}}{\hbar} \sum_{a=1}^{N} V_{\mathrm{RNA}}(z_{a})},$$

where  $\psi_{\alpha}(x) = \prod_{a=1}^{N} (x - z_a)^{\frac{\epsilon_1}{\epsilon_{\alpha}}}$  and the potential  $V(x) = V_{\text{RNA}}(x)$  is chosen as in equation (1.2). Then the partial differential equation (1.42) reduces to

(3.2) 
$$\begin{bmatrix} -\left(\epsilon_{\alpha}\frac{\partial}{\partial x}\right)^2 - 2\epsilon_{\alpha}\left(x - \frac{st}{(1-tx)^2}\right)\frac{\partial}{\partial x} - 4\mu\left(1 - \frac{st^2(2-tx)}{(1-tx)^2}\right) \\ -\frac{\epsilon_1\epsilon_2st^2(2-tx)}{(1-tx)^2}\frac{\partial}{\partial s} - \frac{\epsilon_1\epsilon_2t^3}{1-tx}\frac{\partial}{\partial t}\end{bmatrix} Z_{\alpha}(x;s,t) = 0,$$

where  $\mu$  denotes the 't Hooft parameter  $\mu = \beta^{1/2}\hbar N = -\epsilon_1 N/2$ .

*Proof.* The RNA matrix model has the potential (1.2), with the coefficients  $r_n$  in (1.21) that take form

$$r_0 = 0, \quad r_2 = \frac{1}{2} - st^2, \quad r_n = -st^n \ (n \neq 0, 2).$$

Adopting this choice of coefficients, the action of  $\hat{f}(x)$  in (1.42) on  $Z_{\alpha}(x; s, t)$  can be rewritten solely in terms of derivatives with respect to s and t

$$\begin{split} \hat{f}(x)Z_{\alpha}(x;s,t) &= -\frac{2}{\epsilon_2}g_s^2 N Z_{\alpha}(x;s,t) + g_s^2 \sum_{n=0}^{\infty} x^n t^{n+3} \frac{\partial}{\partial t} Z_{\alpha}(x;s,t) \\ &+ g_s^2 s \sum_{n=0}^{\infty} (n+2)x^n t^{n+2} \bigg(\frac{2}{\epsilon_2} N + \frac{\partial}{\partial s}\bigg) Z_{\alpha}(x;s,t) \\ &= g_s^2 \bigg[ -\frac{2}{\epsilon_2} N + \frac{t^3}{1-tx} \frac{\partial}{\partial t} + s \frac{t^2(2-tx)}{(1-tx)^2} \bigg(\frac{2N}{\epsilon_2} + \frac{\partial}{\partial s}\bigg) \bigg] Z_{\alpha}(x;s,t). \end{split}$$

Using the relation between N and  $\mu$  stated in Proposition 3.1, one obtains the partial differential equation (3.2).

For simplicity, in the following we choose  $\alpha = 2$  and consider the wave-function  $Z_2(x; s, t)$ . On the basis of Proposition 3.1 we describe an algorithm to compute

the free energy of the  $\beta$ -deformed RNA matrix model. To this end we consider the wave-function in the two following limits.

The first limit we consider is such that  $x \to \infty$ . By the definition of the wave-function, the partition function  $Z_N^{\beta}(s,t)$  is encoded in this limit as follows

(3.3) 
$$Z_2(x;s,t) = x^{\epsilon_1 N/\epsilon_2} \left( Z_N^\beta(s,t) + \mathcal{O}(x^{-1}) \right).$$

Equivalently, in the  $x \to \infty$  limit the free energy  $F_N^\beta(x; s, t) = \log Z_N^\beta(x; s, t)$  is found from the *phase function* 

$$(3.4) S(x,s,t) = \log Z_2(x;s,t)$$

304

Taking advantage of Proposition 3.1, we find the nonlinear partial differential equation (1.43) for the phase function S(x, s, t) in Theorem 1.13.

The second limit we consider is  $s \to 0$ . In this limit the RNA matrix model reduces to the Gaussian matrix model, and we denote by  $Z^{\rm G}_{\alpha}(x)$  the corresponding Gaussian wave-function

(3.5) 
$$Z_{\alpha}^{\mathrm{G}}(x) = \lim_{s \to 0} Z_{\alpha}(x;s,t) = \frac{1}{\operatorname{Vol}_{N}^{\beta}} \int_{\mathbb{R}^{N}} \prod_{a=1}^{N} dz_{a} \Delta(z)^{2\beta} \psi_{\alpha}(x) \mathrm{e}^{-\frac{\sqrt{\beta}}{2\hbar} \sum_{a=1}^{N} z_{a}^{2}}.$$

This Gaussian wave-function obeys an ordinary differential equation

(3.6) 
$$\left[-\left(\epsilon_{\alpha}\frac{\partial}{\partial x}\right)^2 - 2\epsilon_{\alpha}x\frac{\partial}{\partial x} - 4\mu\right]Z_{\alpha}^{\rm G}(x) = 0,$$

which is obtained in the  $s \to 0$  limit of equation (3.2). Here we denote the phase function of the Gaussian wave-function for  $\alpha = 2$  by

(3.7) 
$$S_0(x) = \log Z_2^G(x).$$

In order to determine the free energy  $F_N^{\beta}(s,t) = \log Z_N^{\beta}(s,t)$ , we consider now the expansion of the phase function S(x, s, t). First, we consider the expansion of the phase function with respect to the backbone parameter s

(3.8) 
$$S(x, s, t) = \sum_{b=0}^{\infty} S_b(x, t) s^b,$$

which has the following properties:

- The phase function  $S_0(x)$  agrees with that of the Gaussian model
  - (3.9)  $S_0(x) = S_0(x, t).$

#### Enumeration of Chord Diagrams

• The expansion of the phase function  $S_b(x,t)$  around  $x = \infty$  takes form

(3.10) 
$$S_b(x,t) = -\frac{2\mu}{\epsilon_2} \delta_{b,0} \log x + \sum_{p=0}^{\infty} S_{b,p}(t) x^{-p}$$

where an additional log x term for b = 0 appears from the limit of  $\psi_{\alpha}(x)$ . In particular  $S_{0,p} \equiv S_{0,p}(t)$  do not depend on t.

• The free energy  $F_N^{\beta}(s,t)$  is obtained as the generating function of  $S_{b,0}(t)$ 

(3.11) 
$$F_N^\beta(s,t) = \sum_{b=1}^\infty S_{b,0}(t) s^b$$

Applying the expansion (3.8) to the nonlinear partial differential equation (1.43), one finds a hierarchy of differential equations that determine the functions  $S_{b,p}(t)$  recursively. Although our main task is to determine the functions  $S_{b,0}(t)$  in (3.11), we need the extra data of the higher order terms  $S_{b,p\geq1}(t)$  to determine  $S_{b,0}(t)$ .<sup>7</sup> In the following we solve the equation (1.43) systematically in four steps.

#### 3.2 Solving the recursion relations in four steps

Now we solve the recursion relation for  $S_{b,p}(t)$  in the following steps.

- Step 1 Determine the hierarchy of differential equations by expanding the equation (1.43) in the parameter s.
- **Step 2** Solve the ordinary differential equation (3.6) for the Gaussian wavefunction.
- Step 3 Determine  $S_1(x, t)$  iteratively by solving the differential equation to order  $\mathcal{O}(s^1)$  in Step 1, with the initial data  $S_0(x)$  obtained in Step 2.
- Step 4 Repeat the same analysis for  $S_b(x,t)$ , by adopting the  $S_{b'(\leq b-1)}(x,t)$  as an input data.

Step 1: Hierarchy of differential equations for the phase function. We determine the form of the hierarchy of differential equations by substituting the expansion (3.8) in the differential equation (1.43). Picking up coefficients of  $s^0, s^1$ , and  $s^b$  ( $b \ge 2$ ) respectively, we obtain nonlinear partial differential equations for  $S_b(x, t)$ 

(3.12)

$$\epsilon_2^2 \frac{\partial^2}{\partial x^2} S_0(x) + \epsilon_2^2 \left(\frac{\partial}{\partial x} S_0(x)\right)^2 + 2\epsilon_2 x \frac{\partial}{\partial x} S_0(x) + 4\mu = 0,$$

<sup>&</sup>lt;sup>7</sup>It is not easy to reduce this recursion relation between  $S_{b,p}(t)$  to a simple recursion relation involving  $S_{b,0}(t)$  only.

J. E. Andersen, H. Fuji, M. Manabe, R. C. Penner, and P. Sułkowski

$$\epsilon_2^2 \frac{\partial^2}{\partial x^2} S_1(x,t) + 2\epsilon_2^2 \frac{\partial}{\partial x} S_0(x) \frac{\partial}{\partial x} S_1(x,t) + 2\epsilon_2 x \frac{\partial}{\partial x} S_1(x,t) - \frac{2\epsilon_2 t}{(1-tx)^2} \frac{\partial}{\partial x} S_0(x)$$

$$(3.13)$$

$$- \frac{4\mu t^2 (2-tx)}{(1-tx)^2} + \frac{\epsilon_1 \epsilon_2 t^2 (2-tx)}{(1-tx)^2} S_1(x,t) + \frac{\epsilon_1 \epsilon_2 t^3}{1-tx} \frac{\partial}{\partial t} S_1(x,t) = 0,$$

and

$$\epsilon_{2}^{2} \frac{\partial^{2}}{\partial x^{2}} S_{b}(x,t) + \epsilon_{2}^{2} \sum_{a=0}^{b} \frac{\partial}{\partial x} S_{a}(x,t) \frac{\partial}{\partial x} S_{b-a}(x,t) + 2\epsilon_{2}x \frac{\partial}{\partial x} S_{b}(x,t)$$

$$(3.14)$$

$$- \frac{2\epsilon_{2}t}{(1-tx)^{2}} \frac{\partial}{\partial x} S_{b-1}(x,t) + \frac{b\epsilon_{1}\epsilon_{2}t^{2}(2-tx)}{(1-tx)^{2}} S_{b}(x,t) + \frac{\epsilon_{1}\epsilon_{2}t^{3}}{1-tx} \frac{\partial}{\partial t} S_{b}(x,t) = 0.$$

The first differential equation (3.12) for  $S_0(x)$  is equivalent to the quantum curve equation (3.6) for the Gaussian model, and we find that (3.12)–(3.14) can be solved successively for  $S_b(x,t)$  (b = 1, 2, 3, ...).

Step 2: The Gaussian phase function. The Gaussian part of the wavefunction is necessary as an input data in order to solve the equation (3.13).<sup>8</sup> Substituting the expansion (3.10) for  $S_0(x)$  in the differential equation (3.12), we obtain the recursion relation for the *t*-independent (as follows from the *t*independence of  $S_0(x)$ ) coefficients  $S_{0,p} \equiv S_{0,p}(t)$ .

**Proposition 3.2.** The coefficients  $S_{0,p}$  in the 1/x expansion of the Gaussian phase function  $S_0(x)$  obey the recursion relation

$$S_{0,2p-3} = 0, \quad S_{0,2} = \frac{\mu}{2\epsilon_2}(2\mu + \epsilon_2),$$

$$S_{0,2p} = \frac{1}{2p} \Big\{ \epsilon_2(p-1)(2p-1)S_{0,2p-2} - 4\mu(p-1)S_{0,2p-2} \\ + 2\epsilon_2 \sum_{q=1}^{p-2} q(p-q-1)S_{0,2q}S_{0,2p-2q-2} \Big\},$$

where p is a positive integer with  $p \geq 2$ .

Solving this recursion relation iteratively, one finds the expansion

$$S_0(x) = -\frac{2\mu}{\epsilon_2} \log x + \frac{\mu}{2\epsilon_2 x^2} (2\mu + \epsilon_2) + \frac{\mu}{8\epsilon_2 x^4} (4\mu + 3\epsilon_2) (2\mu + \epsilon_2) + \frac{\mu}{24\epsilon_2 x^6} (2\mu + \epsilon_2) (15\epsilon_2^2 + 34\mu\epsilon_2 + 20\mu^2)$$

306

 $<sup>^{8}</sup>$ Any solution of the ordinary differential equation (3.6) can be expressed in terms of Hermite polynomials.

Enumeration of Chord Diagrams

$$+ \frac{\mu}{64\epsilon_2 x^8} (2\mu + \epsilon_2) (105\epsilon_2^3 + 310\mu\epsilon_2^2 + 316\mu^2\epsilon_2 + 112\mu^3) + \frac{\mu}{160\epsilon_2 x^{10}} (2\mu + \epsilon_2) (945\epsilon_2^4 + 3288\mu\epsilon_2^3 + 4424\mu^2\epsilon_2^2 + 2752\mu^3\epsilon_2 + 672\mu^4) (3.16) + \mathcal{O}(x^{-12}).$$

## Step 3: The 1-backbone phase function $S_1(x,t)$ .

In this step we continue our analysis of the 1-backbone phase function  $S_1(x,t)$ . For this purpose, we expand equation (3.13) with respect to  $x^{-1}$  and consider differential equations obtained for the coefficients of  $x^1, x^0, x^{-1}$ , and  $x^{-p}$   $(p \ge 2)$ .

**Corollary 3.3.** The coefficients  $S_{1,p}(t)$  (p = 0, 1, ...) obey the following differential equations in the parameter t:

(3.17) 
$$2S_{1,1}(t) + \epsilon_1 t(\Theta_t + 1)S_{1,0}(t) - \frac{4\mu t}{\epsilon_2} = 0,$$

$$(3.18) \quad 4tS_{1,2}(t) - 4S_{1,1}(t) + \epsilon_1 t^2 (\Theta_t + 1)S_{1,1}(t) - \epsilon_1 t (\Theta_t + 2)S_{1,0}(t) + \frac{8\mu t}{\epsilon_2} = 0,$$
  

$$6t^2 S_{1,3}(t) - 8tS_{1,2}(t) + \epsilon_1 t^3 (\Theta_t + 1)S_{1,2}(t) - \epsilon_1 t^2 (\Theta_t + 2)S_{1,1}(t)$$
  

$$(3.19) \quad -2\epsilon_2 t^2 S_{1,1}(t) - 4\mu t^2 S_{1,1}(t) + 2S_{1,1}(t) - \frac{4\mu t}{\epsilon_2} = 0,$$

and

$$2(p+2)t^{2}S_{1,p+2}(t) - 4(p+1)tS_{1,p+1}(t) + \epsilon_{1}t^{3}(\Theta_{t}+1)S_{1,p+1}(t) - \epsilon_{1}t^{2}(\Theta_{t}+2)S_{1,p}(t) - p(p+1)\epsilon_{2}t^{2}S_{1,p}(t) - 4p\mu t^{2}S_{1,p}(t) + 2pS_{1,p}(t) + 2(p-1)p\epsilon_{2}tS_{1,p-1}(t) + 8(p-1)\mu tS_{1,p-1}(t) - (p-2)(p-1)\epsilon_{2}S_{1,p-2}(t) - 4(p-2)\mu S_{1,p-2}(t) - 2(p-1)tS_{0,p-1} - 2\epsilon_{2}t^{2}\sum_{q=2}^{p-1}q(p-q)S_{0,q}S_{1,p-q}(t) (3.20) + 4\epsilon_{2}t\sum_{q=3}^{p-1}(q-1)(p-q)S_{0,q-1}S_{1,p-q}(t) - 2\epsilon_{2}\sum_{q=4}^{p-1}(q-2)(p-q)S_{0,q-2}S_{1,p-q}(t) = 0.$$

where  $\Theta_t = t\partial/\partial t$ , and  $S_{0,p}$  are solutions of the recursion relation (3.15).

In Step 2,  $S_{0,p}$  have been already determined iteratively. Therefore, substituting the solution (3.16) in equations (3.17)–(3.20), we obtain differential equations for  $S_{1,p}(t)$ . Furthermore, the following condition follows from the recursive structure

$$S_{1,p}(t) = \sum_{k=1}^{\infty} S_{1,p,k} t^k$$
,  $S_{1,p,k} = 0$  for  $p+k$  odd.

whose implementation accelerates the iteration. We have implemented this iteration on a computer and found iteratively solutions for  $S_{1,p}(t)$  (p = 0, 1, 2, ...), which are summarized in appendix C.

Step 4: The multi-backbone phase functions  $S_b(x,t)$   $(b \ge 2)$ .

To extend our analysis to the multi-backbone phase functions  $S_b(x, t)$   $(b \ge 2)$  we expand the differential equation (3.14) with respect to the parameter x, in the same way as in the previous steps.

**Corollary 3.4.** The coefficients  $S_{b,p}(t)$  of the multi-backbone phase function  $S_b(x, t)$  obey

$$2(p+2)t^{2}S_{b,p+2}(t) - 4(p+1)tS_{b,p+1}(t) + \epsilon_{1}t^{3}(\Theta_{t}+b)S_{b,p+1}(t) - \epsilon_{1}t^{2}(\Theta_{t}+2b)S_{b,p}(t) - p(p+1)\epsilon_{2}t^{2}S_{b,p}(t) - 4p\mu t^{2}S_{b,p}(t) + 2pS_{b,p}(t) + 2(p-1)p\epsilon_{2}tS_{b,p-1}(t) + 8(p-1)\mu tS_{b,p-1}(t) - (p-2)(p-1)\epsilon_{2}S_{b,p-2}(t) - 4(p-2)\mu S_{b,p-2}(t) - 2(p-1)tS_{b-1,p-1}(t) - \epsilon_{2}t^{2}\sum_{a=0}^{b}\sum_{q=1}^{p-1}q(p-q)S_{a,q}(t)S_{b-a,p-q}(t) + 2\epsilon_{2}t\sum_{a=0}^{b}\sum_{q=2}^{p-1}(q-1)(p-q)S_{a,q-1}(t)S_{b-a,p-q}(t) (3.21) - \epsilon_{2}\sum_{a=0}^{b}\sum_{q=3}^{p-1}(q-2)(p-q)S_{a,q-2}(t)S_{b-a,p-q}(t) = 0,$$

where  $S_{b,p}(t) = 0$  for  $p \leq -1$ .

The following conditions that follow from (3.21) again accelerate the iteration

$$S_{b,p}(t) = \sum_{k=b}^{\infty} S_{b,p,k} t^k, \quad S_{b,0,k} = 0 \text{ for } b \ge 2, \quad k \le 2b - 3,$$
$$S_{b,p,k} = 0 \text{ for } p + k \text{ odd.}$$

Programming this recursion on a computer, we determined the multi-backbone phase function  $S_b(x,t)$  iteratively. Computational results for  $S_{b,p}(t)$  b = 2, 3 are summarized in appendix C.

## 3.3 The free energy

Finally, we collect all  $S_{b,0}(t)$  obtained in the above four steps together, and substitute them into the equation (3.11). Rewriting  $\epsilon_{\alpha}$  ( $\alpha = 1, 2$ ) in (1.40) in terms of the parameters  $\hbar$  and  $\gamma = \beta^{1/2} - \beta^{-1/2}$ , we obtain the free energy of the  $\beta$ -deformed RNA matrix model

$$F_N^{\beta}(s,t) = \sum_{\ell=0}^{\infty} s^b F_b(\mu,\hbar,\gamma;t),$$

Enumeration of Chord Diagrams

$$\begin{split} F_{1}(\mu,\hbar,\gamma;t) &= \left(\frac{\mu^{2}}{\hbar^{2}} - \frac{\mu}{\hbar}\gamma\right)t^{2} + \left(\frac{2\mu^{3}}{\hbar^{2}} - \frac{5\mu^{2}}{\hbar}\gamma + \left(3\gamma^{2} + 1\right)\mu\right)t^{4} \\ &+ \left(\frac{5\mu^{4}}{\hbar^{2}} - \frac{22\mu^{3}}{\hbar}\gamma + \left(32\gamma^{2} + 10\right)\mu^{2} - \left(15\gamma^{3} + 13\gamma\right)\mu\hbar\right)t^{6} \\ &+ \left(\frac{14\mu^{5}}{\hbar^{2}} - \frac{93\mu^{4}}{\hbar}\gamma + \left(234\gamma^{2} + 70\right)\mu^{3} + \left(52\gamma^{3} + 43\gamma\right)\mu^{2}\hbar \\ &+ \left(105\gamma^{4} + 160\gamma^{2} + 1\right)\mu\hbar^{2}\right)t^{8} + \mathcal{O}(t^{10}), \end{split}$$

$$\begin{split} F_{2}(\mu,\hbar,\gamma;t) &= \frac{\mu}{2\hbar^{2}}t^{2} + \left(\frac{4\mu^{2}}{\hbar^{2}} - \frac{4\mu}{\hbar}\gamma\right)t^{4} \\ &+ \left(\frac{24\mu^{3}}{\hbar^{2}} - \frac{117\mu^{2}}{2\hbar}\gamma + \frac{\mu}{2}\left(69\gamma^{2} + 21\right)\right)t^{6} \\ &+ \left(\frac{128\mu^{4}}{\hbar^{2}} - \frac{542\mu^{3}}{\hbar}\gamma + \mu^{2}\left(762\gamma^{2} + 220\right) - \mu\hbar\left(348\gamma^{3} + 282\gamma\right)\right)t^{8} \\ &+ \mathcal{O}(t^{10}), \\ F_{3}(\mu,\hbar,\gamma;t) &= \frac{\mu}{\hbar^{2}}t^{4} + \left(\frac{58\mu^{2}}{3\hbar^{2}} - \frac{58\mu}{3}\gamma\right)t^{6} \\ &+ \left(\frac{222\mu^{3}}{\hbar^{2}} - \frac{534\mu^{2}}{\hbar}\gamma + \mu\left(312\gamma^{2} + 90\right)\right)t^{8} + \mathcal{O}(t^{10}). \end{split}$$

From (1.17) and (1.19) we obtain the coefficients  $\widetilde{C}_{g,\ell,b}(w)$  in  $F_b(\mu,\hbar,\gamma;t)$  via the formula

(3.22) 
$$F_b(\mu,\hbar,\gamma;t) = \sum_{g,\ell=0}^{\infty} \frac{(-\gamma)^{\ell} \hbar^{2g-2+\ell}}{\mu^{b-2+2g+\ell} b!} \widetilde{C}_{g,\ell,b}(\mu t^2) - \mu \hbar^{-2} \delta_{b,1}.$$

Taking into account the relations (1.20) for  $C_{g,b}(w)$  and  $C_{h,b}^{\mathsf{r}}(w)$ 

$$C_{g,b}(w) = \widetilde{C}_{g,0,b}(w), \qquad C_{h,b}^{\mathsf{r}}(w) = \sum_{\substack{g,\ell=0\\2g+\ell=h}}^{\infty} 2^g \widetilde{C}_{g,\ell,b}(w),$$

we find complete agreement with the computational results (2.8) and (2.9) obtained from the  $\beta$ -deformed topological recursion via Gaussian resolvents.

# **A** Enumeration of non-orientable fatgraphs with k = 3

For the case of one backbone (i.e. one vertex) and k = 3 chords, we list the non-oriented fatgraphs with  $\chi = 1$  in Figure 6 and  $\chi = 0$  in Figure 7.

309



Figure 6: Non-oriented tailed fatgraphs with b = 1, k = 3 and  $\chi = 1$ . The total number of graphs is 22.

Both of these numbers of graphs agree with the number computed from the cut-and-join method and the time-dependent Schrödinger equation.

# **B** The free energy $F_{0,2}$ in 1-cut $\beta$ -deformed models

In this appendix, we determine the unstable term  $F_{0,2}(\{r_n\})$  in the free energy (1.22) for the  $\beta$ -deformed eigenvalue integral (1.14) with the general potential (1.21)

(B.1) 
$$F_{0,2}(\{r_n\}) = F_{0,2}^I(\{r_n\}) + F_{0,2}^A(\{r_n\}).$$

We consider genus s - 1 spectral curves of the form

(B.2) 
$$y(x)^{2} = M(x)^{2}\sigma(x),$$
$$\sigma(x) = \prod_{i=1}^{2^{s}} (x - q_{i}), \quad M(x) = c \prod_{i=1}^{f} (x - \alpha_{i})^{m_{i}},$$

whose form is determined by the choice of the potential. The terms  $F_{0,2}^{I}(\{r_n\})$  and  $F_{0,2}^{A}(\{r_n\})$  are given by [33, 31]

(B.3) 
$$F_{0,2}^{I}(\{r_n\}) = -\frac{1}{8\pi^2} \oint_{\mathcal{A}} \frac{dy(z')}{y(z')} \int_{D} dS(z,z') \log |y(z)|,$$

(B.4) 
$$F_{0,2}^{A}(\{r_n\}) = -\frac{1}{12} \log \left| \prod_{i=1}^{2\mathsf{s}} M(\mathsf{q}_i) \cdot \prod_{1 \le i < j \le 2\mathsf{s}} (\mathsf{q}_i - \mathsf{q}_j) \right|,$$

where  $\mathcal{A} = \bigcup_{i=1}^{s} \mathcal{A}_{i}$  is the contour around the branch cut  $D = \bigcup_{i=1}^{s} D_{i}$ ,  $D_{i} = [\mathbf{q}_{2i-1}, \mathbf{q}_{2i}]$ . Then  $dS(x_{1}, x_{2})$  is the third type differential, which is a 1-form in  $x_{1}$ 



Figure 7: Non-oriented tailed fatgraphs with b = 1, k = 3 and  $\chi = 0$ . The total number of graphs is 42.

and a multivalued function of  $x_2$ , determined by the conditions

• 
$$dS(x_1, x_2) \underset{x_1 \to x_2}{\sim} \frac{dx_1}{x_1 - x_2} + \text{reg.}, \quad \bullet \, dS(x_1, x_2) \underset{x_1 \to \overline{x}_2}{\sim} - \frac{dx_1}{x_1 - x_2} + \text{reg.},$$
  
(B.5) •  $\oint_{x_2 \in \mathcal{A}_i} dS(x_1, x_2) = 0, \quad i = 1, \dots, \mathbf{s} - 1.$ 

Here 
$$\overline{x}$$
 is the conjugate point of a point x on the spectral curve (B.2), such that

(B.6) 
$$\sqrt{\sigma(x)} = -\sqrt{\sigma(\overline{x})}, \quad M(x) = M(\overline{x}).$$

In the following we consider the s = 1 case in (B.2)

(B.7) 
$$\sigma(x) = (x-a)(x-b), \quad a < b,$$

and by applying the method used to derive the unstable term  $F_{0,2}(\{r_n\})$  in a 1-cut matrix model with a hard edge [24], we prove the formula (1.39).

**Proposition B.1.** For the above genus 0 spectral curve, the unstable term  $F_{0,2}(\{r_n\})$  takes form

(B.8) 
$$F_{0,2}(\{r_n\}) = -\frac{1}{2} \sum_{i=1}^{f} m_i \log\left(1 - \mathfrak{s}_i^2\right) - \frac{1}{2} \sum_{i,j=1}^{f} m_i m_j \log\left(1 - \mathfrak{s}_i \mathfrak{s}_j\right) \\ + \frac{1}{24} \log\left|M(a)M(b)(a-b)^4\right|,$$

where  $\mathfrak{s}_i$ ,  $i = 1, \ldots, f$  are defined by

(B.9) 
$$\alpha_i(\mathfrak{s}_i) = \frac{a+b}{2} - \frac{a-b}{4}(\mathfrak{s}_i + \mathfrak{s}_i^{-1}), \quad |\mathfrak{s}_i| < 1.$$

*Proof.* In the above genus 0 case the third type differential  $dS(x_1, x_2)$  is given by

(B.10) 
$$dS(x_1, x_2) = \frac{\sqrt{\sigma(x_2)}}{\sqrt{\sigma(x_1)}} \frac{dx_1}{x_1 - x_2}$$

We introduce the Zhukovsky variable z by

(B.11) 
$$x(z) = \frac{a+b}{2} - \frac{a-b}{4}(z+z^{-1}).$$

Then the branch points x = a, b are mapped to z = -1, +1, and the first and second sheet of the spectral curve are mapped to the regions  $|z| \ge 1$  and  $|z| \le 1$ , respectively. Under this map we obtain

$$\sqrt{\sigma(x)} = \frac{b-a}{4}(z-z^{-1}), \quad \frac{dx}{\sqrt{\sigma(x)}} = \frac{dz}{z}, \quad \frac{\sqrt{\sigma(x_2)}}{x_1-x_2} = \frac{z_1}{z_1-z_2} - \frac{z_1}{z_1-z_2^{-1}},$$

and the third type differential (B.10) is rewritten as

(B.12) 
$$dS(x_1, x_2) = \frac{dz_1}{z_1 - z_2} - \frac{dz_1}{z_1 - z_2^{-1}}$$

Under the map (B.11), the zeros or poles  $\alpha_i$  of the moment function M(x) on the spectral curve (B.2) are mapped to 2f points  $\mathbf{s}_i^{\pm 1}$ ,  $i = 1, \ldots, f$ ,

$$\alpha_i(\mathfrak{s}_i) = \frac{a+b}{2} - \frac{a-b}{4}(\mathfrak{s}_i + \mathfrak{s}_i^{-1}),$$

and we obtain

$$x - \alpha_i = \frac{b - a}{4} \frac{(z - \mathfrak{s}_i)(z - \mathfrak{s}_i^{-1})}{z}$$

Without loss of generality, in this proof we can assume

 $|\mathfrak{s}_i| > 1.$ 

First, let us consider (B.3). Since the variable z of the integrand is on the branch cut D = [a, b], we can put  $z = e^{i\theta}$  in the Zhukovsky variable, and by (B.12)

$$dS(z, z') = -id\theta \left[ \frac{z}{z'} \frac{1}{1 - \frac{z}{z'}} + \frac{1}{1 - \frac{1}{zz'}} \right] = -id\theta \left[ 1 + \sum_{k=1}^{\infty} \frac{2}{z'^k} \cos k\theta \right]$$

is obtained, where we have used  $|\mathbf{z}'| > |\mathbf{z}| = 1.$  Then we obtain

$$\begin{split} \widehat{F}_{0,2}^{I}(z') &:= \int_{D} dS(z,z') \log |y(z)| \\ &= \int_{0}^{\pi} i d\theta \bigg[ 1 + \sum_{k=1}^{\infty} \frac{2}{z'^{k}} \cos k\theta \bigg] \bigg[ \log \bigg| c \frac{b-a}{4} \prod_{i=1}^{f} \Big( \frac{b-a}{4} \Big)^{m_{i}} \bigg| + \log(2\sin\theta) \\ &+ \log \prod_{i=1}^{f} |e^{i\theta} - \mathfrak{s}_{i}|^{m_{i}} |e^{i\theta} - \mathfrak{s}_{i}^{-1}|^{m_{i}} \bigg] \\ &= i\pi \log \bigg| c \frac{b-a}{4} \prod_{i=1}^{f} \Big( \frac{b-a}{4} \Big)^{m_{i}} \bigg| - i\pi \sum_{k=1}^{\infty} \frac{1}{kz_{2}^{2k}} \\ (B.13) \qquad + i \sum_{i=1}^{f} m_{i} \int_{0}^{\pi} d\theta \bigg[ 1 + \sum_{k=1}^{\infty} \frac{2}{z_{2}^{k}} \cos k\theta \bigg] \log |e^{i\theta} - \mathfrak{s}_{i}| |e^{i\theta} - \mathfrak{s}_{i}^{-1}|, \end{split}$$

where we have used that

$$\int_0^{\pi} d\theta \log(2\sin\theta) = 0,$$
  
$$\int_0^{\pi} d\theta \cos k\theta \log \sin\theta = \begin{cases} -\frac{\pi}{k} & \text{if } k \text{ is a nonzero even integer,} \\ 0 & \text{if } k \text{ is an odd integer.} \end{cases}$$

By

$$\begin{split} \log|e^{i\theta} - \mathfrak{s}_i| &= \log|\mathfrak{s}_i| - \frac{1}{2}\sum_{k=1}^{\infty}\frac{1}{k}\left(\frac{1}{\mathfrak{s}_i^k} + \frac{1}{\overline{\mathfrak{s}}_i^k}\right)\cos k\theta - \frac{i}{2}\sum_{k=1}^{\infty}\frac{1}{k}\left(\frac{1}{\mathfrak{s}_i^k} - \frac{1}{\overline{\mathfrak{s}}_i^k}\right)\sin k\theta,\\ \log|e^{i\theta} - \mathfrak{s}_i^{-1}| &= -\frac{1}{2}\sum_{k=1}^{\infty}\frac{1}{k}\left(\frac{1}{\mathfrak{s}_i^k} + \frac{1}{\overline{\mathfrak{s}}_i^k}\right)\cos k\theta + \frac{i}{2}\sum_{k=1}^{\infty}\frac{1}{k}\left(\frac{1}{\mathfrak{s}_i^k} - \frac{1}{\overline{\mathfrak{s}}_i^k}\right)\sin k\theta, \end{split}$$

(B.13) is written as

(B.14) 
$$\widehat{F}_{0,2}^{I}(z) = i\pi \left[ \log \left| c \frac{b-a}{4} \prod_{i=1}^{f} \left( \frac{b-a}{4} \mathfrak{s}_{i} \right)^{m_{i}} \right| - \sum_{k=1}^{\infty} \frac{1}{kz^{2k}} - \sum_{i=1}^{f} m_{i} \sum_{k=1}^{\infty} \frac{2}{kz^{k} \mathfrak{s}_{i}^{k}} \right].$$

Here we have used

$$\int_0^{\pi} d\theta \cos k\theta \cos \ell\theta = \frac{\pi}{2} \delta_{k,\ell}$$

and the fact that for an arbitrary  $\overline{\mathfrak{s}}_i$ , there exists an  $\mathfrak{s}_j$  such that  $\overline{\mathfrak{s}}_i = \mathfrak{s}_j$ . By

$$\frac{dy(z)}{y(z)} = \left[\frac{1}{z-1} + \frac{1}{z+1} - \frac{1}{z} + \sum_{i=1}^{f} m_i \left(\frac{1}{z-\mathfrak{s}_i} + \frac{1}{z-\mathfrak{s}_i^{-1}} - \frac{1}{z}\right)\right] dz,$$

and by (B.14), (B.3) is rewritten as

$$F_{0,2}^{I}(\{r_{n}\}) = \frac{1}{4} \oint_{\widetilde{\mathcal{A}}} \frac{dz}{2\pi i} \left[ \frac{1}{z-1} + \frac{1}{z+1} - \frac{1}{z} + \sum_{i=1}^{J} m_{i} \left( \frac{1}{z-\mathfrak{s}_{i}} + \frac{1}{z-\mathfrak{s}_{i}^{-1}} - \frac{1}{z} \right) \right] \\ \times \left[ \log \left| c \frac{b-a}{4} \prod_{i=1}^{f} \left( \frac{b-a}{4} \mathfrak{s}_{i} \right)^{m_{i}} \right| - \sum_{k=1}^{\infty} \frac{1}{kz^{2k}} - \sum_{i=1}^{f} m_{i} \sum_{k=1}^{\infty} \frac{2}{kz^{k} \mathfrak{s}_{i}^{k}} \right],$$

where  $\widetilde{\mathcal{A}}$  is the contour around the unit disk  $|\mathbf{z}| = 1$ . This contour can be changed as

$$\oint_{\widetilde{\mathcal{A}}} \longrightarrow -\sum_{i=1}^{J} \oint_{\widetilde{\mathcal{A}}_{i}} - \oint_{\widetilde{\mathcal{A}}_{\infty}},$$

where  $\widetilde{\mathcal{A}}_i$  and  $\widetilde{\mathcal{A}}_{\infty}$  are the contours around  $\mathfrak{s}_i$  and infinity, respectively. Then by

$$\sum_{k=1}^{\infty} \frac{u^k}{k} = -\log(1-u), \quad |u| < 1,$$

we obtain

(B.15) 
$$F_{0,2}^{I}(\{r_{n}\}) = -\frac{1}{2} \sum_{i=1}^{f} m_{i} \log\left(1 - \frac{1}{\mathfrak{s}_{i}^{2}}\right) - \frac{1}{2} \sum_{i,j=1}^{f} m_{i} m_{j} \log\left(1 - \frac{1}{\mathfrak{s}_{i}\mathfrak{s}_{j}}\right) + \frac{1}{8} \log\left|M(a)M(b)\left(\frac{a-b}{4}\right)^{2}\right|,$$

where we used

$$\log \left| c \prod_{i=1}^{f} \left( \frac{b-a}{4} \mathfrak{s}_i \right)^{m_i} \right| = -\sum_{i=1}^{f} m_i \log \left( 1 - \frac{1}{\mathfrak{s}_i^2} \right) + \frac{1}{2} \log \left| M(a) M(b) \right|.$$

With (B.4) for the genus 0 case, after changing  $\mathfrak{s}_i$  to  $\mathfrak{s}_i^{-1}$ , we finally obtain the formula (B.8), where we ignored the constant term  $-\log \sqrt{2}$ .

## C An iterative solution using the quantum curve

### The 1-backbone phase function $S_1(x,t)$

The solutions  $S_{1,p}(t)$  (p = 0, 1, 2) of the equations (3.17) – (3.20) are found successively as follows

$$S_{1,0}(t) = -t^3 \frac{2\mu(2\mu + \epsilon_1 + \epsilon_2)}{\epsilon_2 \epsilon_2} - t^4 \frac{\mu}{\epsilon_1 \epsilon_2} \left(5\epsilon_1 \epsilon_2 + 8\mu^2 + 10\mu(\epsilon_1 + \epsilon_2) + 3(\epsilon_1^2 + \epsilon_2^2)\right) - t^6 \frac{\mu}{2\epsilon_1 \epsilon_2} \left(40\mu^3 + 88\mu^2(\epsilon_1 + \epsilon_2) + 64\mu(\epsilon_1^2 + \epsilon_2^2) + 108\mu\epsilon_1\epsilon_2 + 15(\epsilon_1^3 + \epsilon_2^3)\right)$$

314

Enumeration of Chord Diagrams

$$+ 32\epsilon_{1}\epsilon_{2}(\epsilon_{1} + \epsilon_{1}))$$
  
-  $t^{8}\frac{\mu}{4\epsilon_{1}\epsilon_{2}} (224\mu^{4} + 744\mu^{3}(\epsilon_{1} + \epsilon_{2}) + 936\mu^{2}(\epsilon_{1}^{2} + \epsilon_{2}^{2}) + 1592\mu^{2}\epsilon_{1}\epsilon_{2}$   
+  $520\mu(\epsilon_{1}^{3} + \epsilon_{2}^{3}) + 1130\mu\epsilon_{1}\epsilon_{2}(\epsilon_{1} + \epsilon_{2}) + 260\epsilon_{1}\epsilon_{1}(\epsilon_{1}^{2} + \epsilon_{2}^{2})$   
+  $331\epsilon_{1}^{2}\epsilon_{2}^{2} + 105(\epsilon_{1}^{4} + \epsilon_{2}^{4})) + \mathcal{O}(t^{10}),$ 

$$S_{1,1}(t) = t \frac{2\mu}{\epsilon_2} + t^3 \frac{3\mu(2\mu + \epsilon_1 + \epsilon_2)}{\epsilon_2} + t^5 \frac{5\mu}{2\epsilon_2} (8\mu^2 + 10\mu(\epsilon_1 + \epsilon_2) + 3(\epsilon_1^2 + \epsilon_2^2) + 5\epsilon_1\epsilon_2) + t^7 \frac{7\mu}{4\epsilon_2} (40\mu^3 + 88\mu^2(\epsilon_1 + \epsilon_2) + 64\mu(\epsilon_1^2 + \epsilon_2^2) + 108\mu\epsilon_1\epsilon_2 + 15(\epsilon_1^3 + \epsilon_2^3) + 32\epsilon_1\epsilon_2(\epsilon_1 + \epsilon_2)) + \mathcal{O}(t^9),$$

$$S_{1,2}(t) = t^2 \frac{\mu(2\mu + \epsilon_2)}{\epsilon_2} + t^4 \frac{\mu(2\mu + \epsilon_2)(4\mu + 2\epsilon_1 + 3\epsilon_2)}{\epsilon_2} + t^6 \frac{3\mu(2\mu + \epsilon_2)(20\mu^2 + 2\mu(12\epsilon_1 + 17\epsilon_2) + 7\epsilon_1^2 + 15\epsilon_2^2 + 17\epsilon_1\epsilon_2)}{4\epsilon_2} + \mathcal{O}(t^8).$$

The 2-backbone phase function  $S_2(x,t)$ The solutions  $S_{2,p}(t)$  (p = 0, 1, 2) of equation (3.21) take form

$$S_{2,0}(t) = -t^2 \frac{2\mu}{\epsilon_1 \epsilon_2} - t^4 \frac{8\mu(2\mu + \epsilon_1 + \epsilon_2)}{\epsilon_1 \epsilon_2} - t^6 \frac{3\mu}{2\epsilon_1 \epsilon_2} (64\mu^2 + 78\mu(\epsilon_1 + \epsilon_2) + 23(\epsilon_1^2 + \epsilon_2)^2 + 39\epsilon_1 \epsilon_2) - t^8 \frac{\mu}{\epsilon_1 \epsilon_2} (512\mu^3 + 1084\mu^2(\epsilon_1 + \epsilon_2) + 762\mu(\epsilon_1^2 + \epsilon_2^2) + 1304\mu\epsilon_1 \epsilon_2 + 174(\epsilon_1^3 + \epsilon_2^3) + 381\epsilon_1 \epsilon_2(\epsilon_1 + \epsilon_2)) + \mathcal{O}(t^{10}),$$

$$S_{2,1}(t) = t^3 \frac{4\mu}{\epsilon_2} + t^5 \frac{24\mu(2\mu + \epsilon_1 + \epsilon_2)}{\epsilon_2} + t^7 \frac{6\mu}{\epsilon_2} (64\mu^2 + 78\mu(\epsilon_1 + \epsilon_2) + 23(\epsilon_1^2 + \epsilon_2^2) + 39\epsilon_1\epsilon_2) + \mathcal{O}(t^9),$$

$$S_{2,2}(t) = t^2 \frac{\mu}{\epsilon_2} + t^4 \frac{\mu(16\mu + 3\epsilon_1 + 8\epsilon_2)}{\epsilon_2} + t^6 \frac{3\mu}{4\epsilon_2} (192\mu^2 + 2\mu(61\epsilon_1 + 117\epsilon_2) + 13\epsilon_1^2 + 69\epsilon_2^2 + 61\epsilon_1\epsilon_2) + \mathcal{O}(t^8).$$

The 3-backbone phase function  $S_3(x,t)$ The solutions  $S_{3,p}(t)$  (p = 0, 1, 2) of equation (3.21) take form

$$S_{3,0}(t) = -t^4 \frac{4\mu}{\epsilon_1 \epsilon_2} - t^6 \frac{116\mu(2\mu + \epsilon_1 + \epsilon_2)}{3\epsilon_1 \epsilon_2}$$

$$-t^{8} \frac{6\mu}{\epsilon_{1}\epsilon_{2}} (148\mu^{2} + 178\mu(\epsilon_{1} + \epsilon_{2}) + 52(\epsilon_{1}^{2} + \epsilon_{2}^{2}) + 89\epsilon_{1}\epsilon_{2}) + \mathcal{O}(t^{10}),$$

$$S_{3,1}(t) = t^{5} \frac{14\mu}{\epsilon_{2}} + t^{7} \frac{174\mu(2\mu + \epsilon_{1} + \epsilon_{2})}{\epsilon_{2}} + \mathcal{O}(t^{9}),$$

$$S_{3,2}(t) = t^{4} \frac{4\mu}{\epsilon_{2}} + t^{6} \frac{2\mu(58\mu + 15\epsilon_{1} + 29\epsilon_{2})}{\epsilon_{2}} + \mathcal{O}(t^{8}).$$

## References

- M. Aganagic, M. C. N. Cheng, R. Dijkgraaf, D. Krefl, and C. Vafa, Quantum geometry of refined topological strings, JHEP **1211** (2012) 019, arXiv:1105.0630 [hep-th].
- [2] M. Aganagic, R. Dijkgraaf, A. Klemm, M. Marino, and C. Vafa, *Topological strings and integrable hierarchies*, Commun. Math. Phys. **261** (2006) 451, arXiv:hep-th/0312085.
- [3] G. Akemann, Higher genus correlators for the hermitian matrix model with multiple cuts, Nucl. Phys. B482 (1996) 403, arXiv:hep-th/9606004.
- [4] N. V. Alexeev, J. E. Andersen, R. C. Penner, and P. Zograf, *Enumeration of chord diagrams on many intervals and their non-orientable analogue*, Adv. Math. **289** (2016) 1056–1081, arXiv:1307.0967 [math.CO].
- [5] A. S. Alexandrov, A. Mironov, and A. Morozov, Partition functions of matrix models as the first special functions of string theory. 1. Finite size hermitean one matrix model, Int. J. Mod. Phys. A19 (2004) 4127, arXiv:hepth/0310113.
- J. Ambjorn, L. Chekhov, C. F. Kristjansen, and Yu. Makeenko, *Matrix model calculations beyond the spherical limit*, Nucl. Phys. B B404, 127 (1993)
   [Erratum-ibid. B449, 681 (1995)], arXiv:hep-th/9302014.
- [7] J. Ambjorn, L. Chekhov, and Y. Makeenko, *Higher genus correlators from the hermitian one-matrix model*, Phys. Lett. **B282** (1992) 341, arXiv:hep-th/9203009.
- [8] J. E. Andersen, A. J. Bene, J. -B. Meilhan, and R. C. Penner, *Finite type invariants and fatgraphs*, Adv. Math. **225** (2010) 2117–2161, arXiv:0907.2827 [math.GT].

316

- [9] J. E. Andersen, L. O. Chekhov, R. C. Penner, C. M. Reidys, and P. Sulkowski, Topological recursion for chord diagrams, RNA complexes, and cells in moduli spaces, Nucl. Phys. B866 (2013) 414–443, arXiv:1205.0658 [hep-th].
- [10] J. E. Andersen, L. O. Chekhov, R. C. Penner, C. M. Reidys, and P. Sulkowski, *Enumeration of RNA complexes via random matrix theory*, Biochem. Soc. Trans. **41** (2013) 652–655, arXiv:1303.1326 [q-bio.QM].
- [11] J. E. Andersen, H. Fuji, M. Manabe, R. C. Penner, and P. Sulkowski, *Partial chord diagrams and matrix models*, Travaux Mathématiques, **25** (2017), 233–283, arXiv:1612.05840 [math-ph].
- [12] J. E. Andersen, H. Fuji, R. C. Penner, and C. M. Reidys, *The boundary length and point spectrum enumeration of partial chord diagrams via cut and join*, Travaux Mathématiques, **25** (2017), 213–232, arXiv:1612.06482 [math-ph].
- [13] J. E. Andersen, F. W. D. Huang, R. C. Penner, and C. M. Reidys, *Topology of RNA-RNA interaction structures*, Jour. Comp. Biol. **19** (2012) 928–943, arXiv:1112.6194 [math.CO].
- [14] J. E. Andersen, J. Mattes, and N. Reshetikhin, The Poisson structure on the moduli space of flat connections and chord diagrams, Topology 35 (1996) 1069–1083.
- [15] J. E. Andersen, J. Mattes, and N. Reshetikhin, Quantization of the algebra of chord diagrams, Math. Proc. Camb. Phil. Soc. **124** (1998) 451–467, arXiv:qalg/9701018.
- [16] J. E. Andersen, R. C. Penner, C. M. Reidys, and R. R. Wang, *Linear chord diagrams on two intervals*, arXiv:1010.5857 [math.CO].
- [17] J. E. Andersen, R. C. Penner, C. M. Reidys, and M. S. Waterman, *Enumer*ation of linear chord diagrams, arXiv:1010.5614 [math.CO].
- [18] J. E. Andersen, R. C. Penner, C. M. Reidys, and M. S. Waterman, *Topological classification and enumeration of RNA structures by genus*, Jour. Math. Bio. 67 (2013) 1261–1278.
- [19] D. Bar-Natan, On the Vassiliev knot invariants, Topology 34 (1995) 423–475.
- [20] P. Bhadola, and N. Deo, Genus distribution and thermodynamics of a random matrix model of RNA with Penner interaction, Phys. Rev. E88 (2013) 032706.
- [21] P. Bhadola, and N. Deo, Matrix models with Penner interaction inspired by interacting ribonucleic acid, Pramana 84 (2015) 295–308.

- [22] M. Bon, and H. Orland, TT2NE: A novel algorithm to predict RNA secondary structures with pseudoknots, Nucl. Acids Res. 41 (2010) 1895–1900, arXiv:1010.4490 [q-bio.BM].
- [23] M. Bon, G. Vernizzi, H. Orland, and A. Zee, *Topological classification of RNA structures*, Jour. Mol. Bio. **379** (2008) 900–911, arXiv:q-bio/0607032 [q-bio.BM].
- [24] G. Borot, B. Eynard, S. Majumdar, and C. Nadal, Large deviations of the maximal eigenvalue of random matrices, J. Stat. Mech. 1111 (2011) P11024, arXiv:1009.1945 [math-ph].
- [25] V. Bouchard, N. K. Chidambaram, and T. Dauphinee, *Quantizing Weierstrass*, arXiv:1610.00225 [math-ph].
- [26] V. Bouchard, and B. Eynard, Reconstructing WKB from topological recursion, arXiv:1606.04498 [math-ph].
- [27] E. Brezin, C. Itzykson, G. Parisi, and J. B. Zuber, *Planar diagrams*, Comm. Math. Phys. **59** (1978) 35.
- [28] A. Brini, M. Marino, and S. Stevan, The uses of the refined matrix model recursion, J. Math. Phys. 52 (2011) 052305, arXiv:1010.1210 [hep-th].
- [29] R. Campoamor-Stursberg, and V. O. Manturov, Invariant tensor formulas via chord diagrams, Jour. Math. Sci. 108 (2004) 3018–3029.
- [30] L. Chekhov, Genus one correlation to multicut matrix model solutions, Theor. Math. Phys. 141 (2004) 1640, [Teor. Mat. Fiz. 141 (2004) 358] arXiv:hep-th/0401089.
- [31] L. Chekhov, Logarithmic potential beta-ensembles and Feynman graphs, arXiv:1009.5940 [math-ph].
- [32] L. Chekhov, and B. Eynard, Hermitean matrix model free energy: Feynman graph technique for all genera, JHEP 0603 (2006) 014, [hep-th/0504116].
- [33] L. Chekhov, and B. Eynard, Matrix eigenvalue model: Feynman graph technique for all genera, JHEP 0612 (2006) 026, arXiv:math-ph/0604014.
- [34] P. Ciosmak, L. Hadasz, M. Manabe, and P. Sulkowski, *Super-quantum curves from super-eigenvalue models*, JHEP**1610** (2016) 044, arXiv:1608.02596 [hep-th].
- [35] R. Dijkgraaf, L. Hollands, and P. Sulkowski, *Quantum curves and D-modules*, JHEP **0911** (2009) 047, arXiv:0810.4157 [hep-th].
- [36] R. Dijkgraaf, L. Hollands, P. Sulkowski, and C. Vafa, Supersymmetric gauge theories, intersecting branes and free fermions, JHEP 0802 (2008) 106, arXiv:0709.4446 [hep-th].
- [37] N. Do, A. Dyer, and D. Mathews, *Topological recursion and a quantum curve for monotone Hurwitz numbers*, arXiv:1408.3992 [math.GT].
- [38] N. Do, M. A. Koyama, and D. V. Mathews, *Counting curves on surfaces*, arXiv:1512.08853 [math.GT].
- [39] N. Do, and D. Manescu, Quantum curves for the enumeration of ribbon graphs and hypermaps, arXiv:1312.6869 [math.GT].
- [40] N. Do, and P. Norbury, Topological recursion on the Bessel curve, arXiv:1608.02781 [math-ph].
- [41] O. Dumitrescu, and M. Mulase, Quantum curves for Hitchin fibrations and the Eynard-Orantin theory, Lett. Math. Phys. 104 (2014) 635-671, arXiv:1310.6022 [math.AG].
- [42] O. Dumitrescu, and M. Mulase, Quantization of spectral curves for meromorphic Higgs bundles through topological recursion, arXiv:1411.1023 [math.AG].
- [43] O. Dumitrescu, and M. Mulase, Lectures on the topological recursion for Higgs bundles and quantum curves, arXiv:1509.09007 [math.AG].
- [44] P. Dunin-Barkowski, M. Mulase, P. Norbury, A. Popolitov, and S. Shadrin, *Quantum spectral curve for the Gromov-Witten theory of the complex projective line*, Journal für die reine und angewandte Mathematik. Published Online (2014), arXiv:1312.5336 [math-ph].
- [45] B. Eynard, Topological expansion for the 1-hermitian matrix model correlation functions, JHEP 0411 (2004) 031, arXiv:hep-th/0407261.
- [46] B. Eynard and N. Orantin, Invariants of algebraic curves and topological expansion, Commun. Num. Theor. Phys. 1 (2007) 347, math-ph/0702045.
- [47] I. Garg, P. Bhadola, and N. Deo, Non-linear interaction in random matrix models of RNA, arXiv:1103.5601 [cond-mat.stat-mech].
- [48] I. Garg, P. Bhadola, and N. Deo, Structure combinatorics and thermodynamics of a matrix model with Penner interaction inspired by interacting RNA, Nucl. Phys. B870 (2013) 384.
- [49] I. Garg, and N. Deo, Genus distributions for extended matrix models of RNA, arXiv:0802.2440 [cond-mat.stat-mech].

- [50] I. Garg, and N. Deo, A random matrix approach to RNA folding with interaction, Pramana 73 (2009) 533.
- [51] I. Garg, and N. Deo, RNA matrix models with external interactions and their asymptotic behavior, Phys. Rev. E79 (2009) 061903.
- [52] I. Garg, and N. Deo, Scaling, phase transition and genus distribution functions in matrix models of RNA with linear external interactions, arXiv:0911.3710 [q-bio.BM].
- [53] I. Garg, and N. Deo, Structural and thermodynamic properties of a linearly perturbed matrix model for RNA folding, Eur. Phys. J. E Soft Matter 33 (2010) 359–367.
- [54] S. Garoufalidis, and M. Marino, Universality and asymptotics of graph counting problems in nonorientable surfaces, Jour. Comb. Thor. A117 (2010) 715– 740, arXiv:0812.1195 [math.CO].
- [55] I. P. Goulden, J. L. Harer, and D. M. Jackson, A geometric parametrization for the virtual Euler characteristic for the moduli spaces of real and complex algebraic curves, Trans. Amer. Math. Soc. 353 (2001), 4405–4427, math/9902044 [math.AG].
- [56] J. Gu, A. Klemm, M. Mariño, and J. Reuter, Exact solutions to quantum spectral curves by topological string theory, arXiv:1506.09176 [hep-th].
- [57] S. Gukov, and P. Sulkowski, A-polynomial, B-model, and quantization, JHEP1202 (2012) 070, arXiv:1108.0002 [hep-th].
- [58] J. L. Harer, and D. Zagier, The Euler characteristic of the moduli space of curves, Invent. Math. 85 (1986), 457–485.
- [59] L. Hollands, Topological strings and quantum curves, arXiv:0911.3413 [hep-th].
- [60] R. A. Janik, M. A. Nowak, G. Papp, and Z. Ismail, Green's functions in non-hermitian random matrix models, Physica E9 (2001) 456–462, arXiv: cond-mat/9909085.
- [61] M. Katori, Determinantial martingales and noncolliding diffusion processes, Stoch. Proc. Appl. 124 (2014) 3724–3768, arXiv:1305.4412 [math.PR].
- [62] M. Katori, *Elliptic determinantal process of type A*, Probab. Theory Relat. Fields **162** (2015) 637-677, arXiv:1311.4146 [math.PR].
- [63] M. Katori, and H. Tanemura, Non-equilibrium dynamics of dysons model with an infinite number of particles, Comm. Math. Phys. 293 (2010) 469–497.

- [64] M. Kontsevich, Vassiliev's knot invariants, Adv. Sov. Math. 16 (1993) 137– 150.
- [65] I. Kostov, Conformal field theory techniques in random matrix models, arXiv:hep-th/9907060.
- [66] X. Liu, M. Mulase, and A. Sorkin, Quantum curves for simple Hurwitz numbers of an arbitrary base curve, arXiv:1304.0015 [math.AG].
- [67] M. Manabe, and P. Sulkowski, Quantum curves and conformal field theory, arXiv:1512.05785 [hep-th].
- [68] O. Marchal, One-cut solution of the β-ensembles in the Zhukovsky variable, J. Stat. Mech. **1201** (2012) 01011, arXiv:1105.0453 [math-ph].
- [69] M. Marino, Les Houches lectures on matrix models and topological strings, hep-th/0410165.
- [70] M. L. Mehta, Random matrices, 2nd edition, Academic Press (1991).
- [71] M. Mulase, S. Shadrin, and L. Spitz, The spectral curve and the Schrödinger equation of double Hurwitz numbers and higher spin structures, Comm. Numb. Theo. Phys. 7 (2013) 125–143, arXiv:1301.5580 [math.AG].
- [72] M. Mulase, and P. Sulkowski, Spectral curves and the Schroedinger equations for the Eynard-Orantin recursion, Adv. Theor. Math. Phys. 19 (2015) 955– 1015, arXiv:1210.3006 [math-ph].
- [73] M. Mulase, and A. Waldron, Duality of orthogonal and symplectic matrix Integrals and quaternionic Feynman graphs, Commun. Math. Phys. 240 (2003) 553–586, arXiv:math-ph/0206011.
- [74] P. Norbury, Quantum curves and topological recursion, arXiv:1502.04394 [math-ph].
- [75] H. Orland, and A. Zee, RNA folding and large N matrix theory, Nucl. Phys. B620 (2002) 456–476, arXiv:cond-mat/0106359 [cond-mat.stat-mech].
- [76] R. C. Penner, Cell decomposition and compactification of Riemann's moduli space in decorated Teichmüller theory, In Tongring, N. and Penner, R.C. (eds) Woods Hole Mathematics-Perspectives in Math and Physics, World Scientific, Singapore, arXiv:math/0306190 [math.GT].
- [77] R. C. Penner, M. S. Waterman, Spaces of RNA secondary structures, Adv. Math. 101 (1993) 31–49.

- [78] M. Pillsbury, H. Orland, and A. Zee, Steepest descent calculation of RNA pseudoknots, Phys. Rev. E72 (2010) 011911. arXiv:physics/0207110 [physics.bio-ph].
- [79] M. Pillsbury, J. A. Taylor, H. Orland, and A. Zee, An algorithm for RNA pseudoknots, arXiv:cond-mat/0310505.
- [80] C. M. Reidys, Combinatorial and computational biology of pseudoknot RNA, Applied Math series, Springer (2010).
- [81] C. M. Reidys, F. W. Huang, J. E. Andersen, R. C. Penner, P. F. Stadler, and M. E. Nebel, *Topology and prediction of RNA pseudoknots*, Bioinformatics 27 (2011) 1076–1085.
- [82] A. Schwarz, Quantum curves, Comm. Math. Phys. 338 (2015) 483–500, arXiv:1401.1574 [math-ph].
- [83] P. G. Silvestrov, Summing graphs for random band matrices, Phys. Rev. E55 (1997) 6419–6432, arXiv:cond-mat/9610064 [cond-mat.stat-mech].
- [84] J. J. M. Verbaarschot, H. A. Weidenmuller, and M. R. Zirnbauer, Grassmann integration in stochastic quantum physics: The case of compound nucleus scattering, Phys. Rept. 129 (1985) 367–438.
- [85] G. Vernizzi, and H. Orland, Large-N random matrices for RNA folding, Acta Phys. Polon. B36 (2005) 2821–2827.
- [86] G. Vernizzi, H. Orland, and A. Zee, *Enumeration of RNA structures by matrix models*, Phys. Rev. Lett. **94** 168103, arXiv:q-bio/0411004 [q-bio.BM].
- [87] G. Vernizzi, P. Ribecca, H. Orland, and A. Zee, *Topology of pseudoknotted homopolymers*, Phys. Rev. E73 (2006) 031902, arXiv:q-bio/0508042.
- [88] J. Zhou, Quantum mirror curves for C<sup>3</sup> and the resolved conifold, arXiv:1207.0598 [math.AG].

Jørgen Ellegaard Andersen Centre for Quantum Geometry of Moduli Spaces, Department of Mathematics, Aarhus University, DK-8000 Aarhus C Denmark jea.qgm@gmail.com

Hiroyuki Fuji Faculty of Education, Kagawa University, Takamatsu 760-8522 Japan fuji@ed.kagawa-u.ac.jp

Masahide Manabe Faculty of Physics, University of Warsaw, ul. Pasteura 5, 02-093 Warsaw, Poland masahidemanabe@gmail.com

Robert C. Penner Institut des Hautes Études Scientifiques, 35 route de Chartres, 91440 Burs-sur-Yvette, France; Division of Physics, Mathematics and Astronomy, California Institute of Technology, Pasadena, CA 91125, USA rpenner@caltech.edu, rpenner@ihes.fr

Piotr Sułkowski
Faculty of Physics, University of Warsaw,
ul. Pasteura 5, 02-093 Warsaw, Poland;
Walter Burke Institute for Theoretical Physics,
California Institute of Technology, Pasadena, CA 91125, USA
psulkows@fuw.edu.pl

### Table of Content

Foreword
Collection of Summaries
Jørgen Ellegaard Andersen and Niccolo Skovgård Poulsen. Coordinates for the Universal Moduli Space of Holomorphic Vector Bundles
Jørgen Ellegaard Andersen and Jens-Jakob Kratmann Nissen. Asymptotic aspects         of the Teichmüller TQFT       41–95
<i>Jørgen Ellegaard Andersen and Simone Marzioni</i> . Level N Teichmüller TQFT and Complex Chern–Simons Theory
Jørgen Ellegaard Andersen and William Elbæk Petersen. Construction of Modular Functors from Modular Categories
Jørgen Ellegaard Andersen, Hiroyuki Fuji, Robert C. Penner, and Christian M. Reidys. The boundary length and point spectrum enumeration of partial chord diagrams using cut and join recursion
Jørgen Ellegaard Andersen, Hiroyuki Fuji, Masahide Manabe, Robert C. Penner, and Piotr Sułkowski. Partial Chord Diagrams and Matrix Models . 233–283
Jørgen Ellegaard Andersen, Hiroyuki Fuji, Masahide Manabe, Robert C. Penner, and Piotr Sułkowski. Enumeration of Chord Diagrams via Topological Recursion and Quantum Curve Techniques

#### Instructions to authors

Manuscripts should be written in English, French or German, and printed in **Latex** style. The final Latex source files, on which the publication will be based, should be prepared by the authors in a prescribed format using the macro packages available at the webpage of the journal http://math.uni.lu/travaux-mathematiques

Each paper must include an **abstract** of not more than 200 words, which should contain a brief but informative summary of the contents of the paper. The abstract should be written in English.

Authors should include in their papers one or more **classification numbers**, following the AMS Mathematics Subject Classification. Details of this scheme can be found in each Annual Index of Mathematical Reviews or on the web at *http://www.ams.org/msc.* 

A few key words should also be indicated.

The manuscripts have to be submitted electronically to martin.schlichenmaier@uni.lu.

### Subscription

In order to subscribe to the journal or to order previous volumes separately, please write to the address .

University of Luxembourg Mathematics Research Unit, FSTC Maison du Nombre 6, avenue de la Fonte L-4364 Esch-sur-Alzette Grand-Duchy of Luxembourg

Electronic copies are also available on the above mentioned web-page of the journal.

## **University of Luxembourg**

Mathematics Research Unit

Maison du Nombre 6, avenue de la Fonte L-4364 Esch-sur-Alzette Grand-Duchy of Luxembourg

## University of Luxembourg

Multilingual. Personalised. Connected.

ISBN 978-2-87971-168-3 ISSN 1024-1833

# www.uni.lu

